UNIVERSITY OF SZEGED FACULTY OF HUMANITIES AND SOCIAL SCIENCES

MURAT IŞIK

DOCTORAL SCHOOL OF LINGUISTICS

THE BOOK OF LEVITICUS OF THE GÖZLEVE BIBLE (1841): A COMPREHENSIVE ANALYSIS

Doctoral dissertation written under the supervision of Dr. Zsuzsanna Olach

Contents

| Abbreviations | 6 |
|--|----|
| Preface | 11 |
| Acknowledgments | 12 |
| 1. Introduction | 13 |
| 1.1. General Remarks on the Aim, Scope, and Methodology of the Study | 13 |
| 1.2. Crimean Karaims | 15 |
| 1.2.1. Ethnic and Religious Heritage | 15 |
| 1.2.2. History | 17 |
| 1.2.3. Language | 20 |
| 1.2.3.1. Crimean Karaim and the Debates on Its Existence | 22 |
| 1.2.3.2. Crimean Karaim Varieties | 27 |
| 1.2.4. Written Sources | 30 |
| 1.2.4.1. Crimean Karaim Bible Translations and Associated Studies | 30 |
| 1.3. The Subject of the Dissertation | 33 |
| 2. The Linguistic Analysis of the Text | 35 |
| 2.1. Phonology | 35 |
| 2.1.1. Sound Inventory, Orthography and Transcription System | 36 |
| 2.1.1.1. Vowels | 37 |
| 2.1.1.1. Vowel System | 37 |
| 2.1.1.1.2. Orthography and Transcription of Vowel Sounds | 38 |
| 2.1.1.2. Consonants | 41 |
| 2.1.1.2.1. Consonant System | 41 |
| 2.1.1.2.2. Orthography and Transcription of Consonant Sounds | 42 |
| 2.1.2. Phonological Variations and Adaptations | 44 |
| 2.1.2.1. Vowels | 45 |
| 2.1.2.1.1. Vowel Assimilation | 45 |
| 2.1.2.1.2. Degree of Openness | 47 |
| 2.1.2.1.3. Degree of Roundness | 53 |
| 2.1.2.1.4. Addition of a Vowel | 54 |
| 2.1.2.2. Consonants | 56 |
| 2.1.2.2.1. Ottoman Turkish and Kipchak Consonantal Alternations | 56 |

| 2.1.2.2.2. Other Consonantal Features | 65 |
|--|-----|
| 2.1.2.2.1. Aspiration | 65 |
| 2.1.2.2.2. The Alternation Between t - ~ \check{c} - and t - ~ k | 66 |
| 2.1.2.2.3. Preservation and Devoicing of Final Lenes in Arabic and Persian Loans | |
| 2.1.2.2.2.4. Consonant Deletion | |
| 2.1.2.2.5. Addition of a Consonant | |
| 2.1.2.2.6. Replacement of Arabic Epiglottal Stop Ayin and Glottal Stop Aleph | |
| 2.1.2.2.7. Metathesis | |
| 2.1.2.3. Vowel-Consonant Sequences | |
| 2.1.2.3.1. Vowel-Consonantal Alternation | |
| 2.2. Morphology and Morphophonology | |
| 2.2.1. Morphophonological Characteristics | |
| 2.2.1.1. Intrasyllabic Vowel Harmony | |
| 2.2.1.2. Consonant Harmony | |
| 2.2.1.3. Intervocalic Lenition | |
| 2.2.1.4. Vowel Deletion | |
| 2.2.1.5. Consonant Deletion | 78 |
| 2.2.1.6. Metathesis | 79 |
| 2.2.1.7. The Alternation Between $\eta \sim n$ | 79 |
| 2.2.1.8. The Alternation Between $z \sim s$ | |
| 2.2.2. Non-Verbal Categories | 80 |
| 2.2.2.1. Nominal Formation | 80 |
| 2.2.2.1.1. Suffixation | 80 |
| 2.2.2.1.1.1. Denominal Derivations of Nominals | 80 |
| 2.2.2.1.1.2. Deverbal Derivations of Nominals | 86 |
| 2.2.2.1.2. Compounding | 92 |
| 2.2.2.2. Nominal Predicative Markers | 94 |
| 2.2.2.3. Nouns | 98 |
| 2.2.2.3.1. Inflectional Categories | 98 |
| 2.2.2.3.1.1. Number | 98 |
| 2.2.2.3.1.2. Possessive Markers | 100 |
| 2.2.2.3.1.3. Cases | |
| 2.2.2.4. Numerals | 110 |
| 2.2.2.4.1. Cardinal Numbers | 110 |

| | 2.2.2.4.2. Fractional Numerals | 111 |
|---|---|-----|
| | 2.2.2.4.3. Ordinal Numerals | 112 |
| | 2.2.2.4.4. Distributive Numerals | 113 |
| | 2.2.2.5. Pronouns | 113 |
| | 2.2.2.5.1. Personal Pronouns | 114 |
| | 2.2.2.5.2. Demonstrative Pronouns | 115 |
| | 2.2.2.5.3. Reflexive Pronouns | 117 |
| | 2.2.2.5.4. Interrogative Pronouns | 117 |
| | 2.2.2.5.5. Relative Pronouns | 118 |
| | 2.2.2.5.6. Indefinite Pronouns | 119 |
| | 2.2.2.6. Adjectives | 120 |
| | 2.2.2.7. Adverbs | 120 |
| 2 | 2.2.3. Verbal Categories | 121 |
| | 2.2.3.1. Verb Formation | 121 |
| | 2.2.3.1.1. Suffixation | 121 |
| | 2.2.3.1.1.1. Denominal Verb Formative Markers | 122 |
| | 2.2.3.1.1.2. Deverbal Verb Formative Marker | 125 |
| | 2.2.3.1.2. Compounding | 126 |
| | 2.2.3.2. Voice | 126 |
| | 2.2.3.2.1 Active Voice | 126 |
| | 2.2.3.2.2. Passive Voice | 127 |
| | 2.2.3.2.3. Causative Voice | 128 |
| | 2.2.3.2.4. Reflexive Voice | 131 |
| | 2.2.3.2.5. Reciprocal Voice | 131 |
| | 2.2.3.3. Tense | 132 |
| | 2.2.3.3.1. Definite Past Tense | 132 |
| | 2.2.3.3.2. Present Tense | 133 |
| | 2.2.3.3.3. Aorist | 136 |
| | 2.2.3.3.4. Future Tense | 137 |
| | 2.2.3.4. Mood | 138 |
| | 2.2.3.4.1. Imperative | 138 |
| | 2.2.3.4.2. Voluntative | 140 |
| | 2.2.3.4.3. Optative | 141 |
| | 2.2.3.4.4. Conditional | 143 |

| 2.2.3.5. Converbs | 145 |
|---|-----|
| 2.2.3.6. Participles | 147 |
| 2.2.3.7. Verbal Nouns | 150 |
| 2.2.4. Conjunctions | 151 |
| 2.3. Syntax and Morphosyntax | 152 |
| 2.3.1 Phrases | 152 |
| 2.3.1.1. Nominal Phrases | 152 |
| 2.3.1.1.1. Case Functions | 155 |
| 2.3.1.2. Verb Phrase | 167 |
| 2.3.1.2.1. Biblical Hebrew Infinitival Paronomasia | 167 |
| 2.3.1.3. Adjunct phrases | 169 |
| 2.3.1.3.1. Converbial Constructions | 169 |
| 2.3.1.3.2. Postpositional and Prepositional Constructions | 171 |
| 2.3.2. Sentences | 175 |
| 2.3.2.1. Simple Sentences | 175 |
| 2.3.2.2. Complex Sentences | 177 |
| 2.3.2.2.1. Coordinate Complex Sentences | 177 |
| 2.3.2.2.2. Subordinate Complex Sentences | 178 |
| 2.3.2.2.1. Relative Clauses | 178 |
| 2.3.2.2.2 Adverbial Clauses | 180 |
| 2.4. Lexicon | 184 |
| 2.4.1. Turkic Vocabulary | 185 |
| 2.4.2. Arabic Loanwords | 190 |
| 2.4.3. Persian Loanwords | 194 |
| 2.4.4. Biblical Hebrew Loanwords | 195 |
| 2.4.5. Loanwords from Other Languages | 198 |
| 2.4.6. Copies from Ottoman Bible Translations | 199 |
| 3. Conclusion | 204 |
| 3.1. Phonology | 204 |
| 3.2. Morphology and Morphophonology | 206 |
| 3.3. Syntax and Morphosyntax | 207 |
| 3.4. Lexicon | |
| 3.5. Final Remarks | 209 |
| References | 210 |

| Appendix | 221 |
|---------------|-----|
| Transcription | 221 |
| Translation | 282 |
| Facsimile | 343 |

Abbreviations

Biblical Books

Gen Book of Genesis

Exo Book of Exodus

Lev Book of Leviticus

Num Book of Numbers

Ruth Book of Ruth

Bible Translations

4B.Or.131-1 (**Kieffer 1827**; **Ot.Tur.B**): It represents a revision of Ali Bey's Ottoman Bible translation from 1665, republished by Jean Daniel Kieffer in 1827 across two volumes. This revised edition can be found within the collection of the Bayerische Staats bibliothek in Munich. The initial endeavor to revise Ali Bey's translation was undertaken by Baron H. F. von Diez, commissioned by the British Bible Society in 1814. Following his passing in 1817, Jean Daniel Kieffer became part of the project. By 1819, the translation of the New Testament was disseminated. Subsequently, in Kieffer's 1827 Bible edition, he incorporated the first four books of Ali Bey's Pentateuch, previously edited by Baron H. F. von Diez, along with a revised version of the 1819 New Testament translation (Privratsky 2014: 44; Işık 2021: 347).

ADub.III.73 (**T.Kar.B**): A mid-19th century Trakai Karaim Bible translation of the Torah and the Haftarah, available at:

https://middleturkic.lingfil.uu.se/manuscripts/middle-karaim/ADub.III.73

BSMS 288: A Crimean Karaim Bible translation (see 1.2.4.1.)

Cod. Or. 1101a-f (Ali Bey 1665): Situated in the Warner Collection at the Leiden University Library resides the meticulous secretary's fair copy of an edition. This constitutes an Ottoman Turkish translation of the Bible, produced by Ali Bey, who is also known under the names Ali Ufkî, Wojciech Bobowski, Albert Bobowski, and Albertus Bobovius. Crafted between the years 1662 and 1664 in Istanbul, this fair copy was subsequently printed in Leiden in 1665. While not the first-ever translation of the Bible into Ottoman Turkish, it is

distinguished as the first comprehensive translation, encompassing the entirety of the Old Testament, inclusive of the Apocrypha, as well as the New Testament (Pawlina 2006: 34; Işık 2021: 347).

Cr.Tat.B: a Crimean Tatar Bible translation from 2016, see: https://ibtrussia.org/en/text?m=CRT&l=Lev.1.1.1&g=0

H 170: A Crimean Karaim Bible translation (see 1.2.4.1.)

JSul.III.01: A mid-19th century Halich Karaim Bible translation of the Torah and the Haftarah, available at: https://middleturkic.lingfil.uu.se/manuscripts/middle-karaim/JSul.III.01

NAS 1977: New American Standard 1977 (via Bible Works 9)

RSO 1876: Russian Synodal Orthodox Version (via Bible Works 9)

Languages

Ar Arabic

B.Heb Biblical Hebrew

Cr.Kar Crimean Karaim

Cr.Tat Crimean Tatar

Cr.Tur Crimean Turkish

E.Kar Eastern Karaim

Eng English

H.Kar Halich Karaim

It Italian

Kip. Kipchak Turkic

Mong Mongolian

Ogh. Oghuz Turkic

Ot. Tur Ottoman Turkish

Per Persian

Sogd Sogdian

T.Kar Trakai Karaim

Trk Turkish

Tur. Turkic

W.Kar Western Karaim

Linguistics

1 First person

2 Second Person

3 Third Person

ABL Ablative

ABS Absolute state

ACC Accusative

AN Actional nominal

AOR Aorist

BOTH Common gender

CARD Cardinal numeral

CASE Case markers

CAUS Causative

CL Clitic

COMP Case Marked Complement

COND Conditional

CONN Connective

CONS Consect

CONST Construct state

CONV Converbial marker

COP Copula

DAT Dative marker

DEM Demonstrative

DER Derivational marker

DISTR Distributive

DUAL Dual

EQU Equative

FEM Feminine

GEN Genitive

HIP Hiphil

HOP Hophal

IMP Imperative

IMPT IMPERFECT

INF Infinitive

LOC Locative

MASC Masculine

N Noun

NEG Negative

NOM Nominative

O Object

onom Onomatopoeic

OPT Optative

ORD Ordinal numeral

PASS Passive

PAST Past tense

PERF Perfect

PL Plural

PN Participant Nominal

POSS Possessive

POST Postposition

PRD Predicative

PREP Preposition

PRES present/continuous

PRNM Pronominal

PRO Pronoun

PTCL Particle

RECP Reciprocal

REFL Reflexive

S Subject

SG Singular

SUBJ Subjunctive

TRANS transitivized

V Verb

VN Verbal noun marker

Additional

ch. chapter

chs. chapters

Preface

This dissertation undertakes a comprehensive analysis of the linguistic features present in a biblical book found in the so-called Gözleve Bible, published in 1841 in what is now known as Eupatoria. The printed edition comprises translations of the entire Tanakh, with the exception of the Chronicles, and is divided into four volumes, all written in Hebrew script. The language of the edition is generally recognized as Crimean Karaim. However, its precise language, and even the very existence of the Crimean Karaim language itself, have stirred significant scholarly debates. Previous investigations addressing this linguistic conundrum have primarily focused on discrete sections of this edition, often analyzing short fragments and rarely a specific entire book. Therefore, the objective of this dissertation is to provide a comprehensive review of an unstudied book from this edition, namely the Book of Leviticus. This review will include its transcription and translation, as well as a faithful reproduction of the original text in facsimile form. This examination will carry out a detailed linguistic assessment, comparing equivalent features found in the text with those in Crimean Tatar, Crimean Turkish, Ottoman Turkish, and all three Karaim dialects since distinctions between Ottoman Turkish and Kipchak languages are crucial in discerning the variants of Crimean Karaim. Moreover, this comparison will also contribute to the discussion about whether the edition was written in a superficial language, as suggested by certain scholars.

This dissertation begins with a succinct overview of the ethno-religious background, history, language, and written sources of the Crimean Karaims. It engages with ongoing discussions concerning the existence of the Crimean Karaim language and debates associated with the language used in this edition. The subsequent chapters delve into an in-depth linguistic analysis of the data, encompassing phonology, morphology, morphophonology, syntax, morphosyntax, and lexicon. Following the conclusion, an appendix will be provided that includes the transcription and translation.

Acknowledgments

First and foremost, I am profoundly grateful to the Tempus Public Foundation (Tempus Közalapítvány) for their support through the Stipendium Hungaricum scholarship, which facilitated my Ph.D. studies in Hungary. Throughout my studies, I have been fortunate to be surrounded by a network of supportive friends and family, all of whom have significantly contributed to my Ph.D. journey. While space constraints render it impossible to list each individual, I would be remiss not to recognize several key individuals who have provided unwavering support. Firstly, I would like to express my gratitude to my supervisor, Dr. Zsuzsanna Olach, for her support and for introducing me to the field of Karaim studies. I am also profoundly indebted to all my colleagues at the Department of Altaic Studies, including our former department head, Prof. Dr. István Zimonyi, and the current department head, Prof. Dr. Sándor Papp. Their consistent guidance and patience have been invaluable throughout the development and completion of my dissertation. I extend my appreciation to the opponents of my dissertation predefense, Assoc. Prof. Edina Dallos and Dr. Balázs Danka, for their invaluable contributions to the discussion. I also wish to express my gratitude to Assoc. Prof. Michał Németh and Prof. Emer. András Róna-Tas for their kind willingness to assist me whenever I had inquiries. My heartfelt thanks also go to Assoc. Prof. Zeynep Erk Emeksiz, and my dear friend Anita András, without whose efforts and support my Ph.D. journey in Hungary would have never commenced. Equally, I am immensely grateful for the unwavering and invaluable support I have received from my friends, Özgür Andaç and Kutse Altın. Finally, I am deeply grateful for the love and patience that I received from my family, whose unwavering support has been a beacon not only throughout my Ph.D. studies but also throughout my entire life. In addition, I would also like to acknowledge my little niece, Arven, whose birth not only brought joy to our family, but also has been a constant source of strength in my life.

Murat Işık

29.05.2023, Szeged/Hungary

1. Introduction

1.1. General Remarks on the Aim, Scope, and Methodology of the Study

The primary objective of this research is to undertake an exhaustive exploration of a book from a complete printed Tanakh translation (excluding the Chronicles), known as the Gözleve Bible or Eupatorian print (1841). While the language of this Bible translation is generally identified as Crimean Karaim, it continues to be a topic of debate due to its heterogeneous properties. A relatively small number of studies have investigated the language of the Gözleve Bible, typically focusing on selected portions (refer to section 1.3.). The analyses and conclusions regarding this edition also raise skepticism from certain scholars about the existence of Crimean Karaim, which is usually considered an extinct Eastern dialect of Karaim. Analyzing the limited written resources available could prove valuable in the ongoing discourse about Crimean Karaim and the historical tradition of Bible translations in Karaim. Thus, this dissertation seeks to unveil a hitherto untranscribed and thoroughly unstudied book of the Gözleve Bible, specifically, the third book of the Torah, commonly referred to as the Book of Leviticus. It should be noted that the outcomes of this research strictly relate to the linguistic attributes of this specific book and do not reflect the language of the entire edition.

This dissertation presents a comprehensive transcription of the Book of Leviticus, originally written in Hebrew. As detailed in Section 2.1.1, the differentiation of values among certain Karaim vowel pairs in non-Biblical Hebrew words is not denoted in Hebrew script due to orthographic complexities. In such instances, we have considered Karaim phonotactics and descriptions found in Karaim dictionaries and studies. In the transcription, which is included in the appendix, we carefully addressed specific errors or challenges, systematically comparing them with other Bible translations across the three different Karaim dialects and, occasionally, some Ottoman Bible translations² when equivalent portions existed in such translations. A more detailed description of the transcription notes can be found in the relevant section of the appendix.

An English translation of our text is also provided in the appendix. During the translation process, we adopted a philological approach, prioritizing the conveyance of the intended meaning. Concurrently, we aimed to mirror the structural elements of the original text in our

¹ It should be acknowledged that upon the completion of this dissertation, our transcription, along with the remaining books of the Torah translation from Göz. 1841, will be accessible through the following online portal: https://middleturkic.lingfil.uu.se/manuscripts/middle-karaim/JSul.IV.02A.

² For the descriptions of these Bible translations, please refer to the abbreviations section.

translated version, within the confines of the English language. This dual emphasis on meaning and structure defines our translation technique, which we term as 'meaning-based structural equivalence'. This term is introduced to characterize an approach that strives to strike a balance between semantic fidelity and structural preservation. It should be noted that in various examples where required, footnotes have been used to illustrate the literal meaning of certain instances. While this dissertation does not aim to conduct a detailed, systematic comparison with the Hebrew Bible, the literal meanings noted in the footnotes are systematically compared to Biblical Hebrew only in cases where discernible similarities exist, in order to show readers whether the discussed characteristics are possible calques from Biblical Hebrew or not. As for the translation notes, refer to the relevant part in the appendix.

This study employs a descriptive and comparative methodology for the linguistic analysis of the text. This involves a thorough examination of the phonological, morphological, morphophonological, syntactic, morphosyntactic, and lexical features of the text, with these categories being subsequently compared to their counterparts in Crimean Tatar, Crimean Turkish, Ottoman Turkish, and three dialects of Karaim (Crimean, Halich and Trakai), where available sources exhibit relevant features. In the course of our descriptive analysis, we highlight specific characteristics within the text, diligently selecting examples that represent these attributes. When relevant and needed, all instances embodying these features are included. However, only representative samples are cited in most sections, as the frequency of occurrence of such features does not hold substantial relevance to our investigation. Thus, only representative examples will be showcased in our description, unless specifically mentioned. Regarding the comparative aspect of this study, various sources were consulted. For Crimean Turkish, our resources were primarily limited to a short section written by Doerfer (1959a), and consequently, comparisons involving Crimean Turkish are sparse, as noted in the relevant sections. For linguistic features related to Crimean Tatar, primary sources included works by Kavitskaya (2010), Jankowski (2010), and Doerfer (1959b), and the KRUS dictionary was primarily used for lexical items. For Ottoman Turkish, which notably consists of three main phases - Old Ottoman (13th-15th century), Middle Ottoman (16th-18th century), and New Ottoman (19th century to 1928), as outlined by Kerslake (2022: 176) - descriptions pertaining to New Ottoman features by Kissling (1960) and Hagopian (1907) were predominantly employed. Dictionaries utilized included ETD, LET, and occasionally ÖTS and YTL, along with a dictionary pertaining to Middle Ottoman, TLO. In addition, an array of studies on Karaim was consulted, encompassing works by Prik (written in 1949 but published in 1976), Musaev (1964, 1977), Zajączkowski (1932), Gülsevin (2016), Çulha (2019), and Öztürk (2019). For lexical references, we primarily relied on two chief sources: CKED and KRPS. It is pivotal to note that when assessing the features of Crimean Karaim, our predominant reference was Prik's 1976 study, owing to its aptitude for comparing the Crimean Karaim Bible translations, which bear archaic features. This choice stands in contrast with Çulha's study, which is largely predicated on mejumas, and thereby predominantly showcases features heavily influenced by Crimean Turkish, often devoid of certain intrinsic Karaim properties. Moreover, in certain segments when needed, equivalent sections of the Crimean Karaim Bible translations (e.g., Jankowski 1997: 28–52, Németh 2016: 169–189, CrKB I: 165–217) and a Prayer Book from 1734 (e.g., Sulimowicz 1972: 56–64) were also utilized for comparative purposes.

It should be noted that in specific sections, selected examples from the Hebrew Bible and its translations in English, and occasionally Russian will be incorporated. These examples, collected from a software known as 'Bible Works 9', will be introduced where relevant. In addition, in syntax and morphosyntax section certain Bible translations from Crimean Tatar, Ottoman Turkish and Western Karaim were used in order to demonstrate the Karaim bible translation strategies which exhibit significant influence from Biblical Hebrew. However, while some of these influences are discussed in their respective sub-chapters, it is important to note, as mentioned earlier, that the study does not specifically focus on conducting an in-depth analysis of Hebrew influence. As a result, a separate section solely dedicated to this topic is not included.

1.2. Crimean Karaims

1.2.1. Ethnic and Religious Heritage

Karaite religion, also known as the Karaite Judaism, originated as a movement in the second half of the eighth century in Baghdad (Poznański 1915: 662; Nemoy 1978: 604). Anan ben David, who lived during the reign of a caliph of the Abbasid dynasty, is generally acknowledged as the principal founder and the first lawgiver of the Karaite movement³ (Zajączkowski 1961: 24). He drafted the initial document, a code of laws⁴ called *Sefer ha-Mitzvolt*, 'The Book of

³ It has also been discussed that Anan was not the founder of the Karaim religion, even though his thoughts influenced the movement (Nemoy 1950: 307–311; Poznański 1915: 662; Schur 1992: 20–24). For further discussion regarding Anan, see Harkavy 1904: 553–556; Nemoy 1952: 3–11.

⁴ It is worth noting that some of Anan's principles were later modified, mainly by Benjamin Nahawandi who had a significant impact on Karaite religion after Anan ben David (Kohler and Harkavy 1904: 438–439; Poznanski 1915: 664).

Precepts', written in Aramaic around the middle of the eighth century (Nemoy 1952: 8)⁵. This new movement⁶ emphasized the Tanakh⁷ as the sole source for interpretation, dismissing any supplements. Essentially, the central notion was to reject the Talmud, a written assembly of oral traditions deemed sacred by Rabbanites, followers of mainstream Judaism. Apart from this primary distinction between Karaites and Rabbanites, traces of Muslim law and philosophy are also apparent in the Karaite religion (Kohler and Harkavy 1904: 440; Ankori 1968: 3). These influences include avoiding anthropomorphic expressions in translations of the Tanakh, adopting a low prostration during prayers, removing shoes before entering a *kenesa*, and performing a form of ablution before praying⁸ (Kizilov 2009: 136; Zajączkowski 1961: 28–29; Besalel, 2001: 311).

The pinnacle of Karaite religious and scholarly activity occurred after its center moved from Iran and Iraq to Jerusalem and Cairo. From these hubs, it extended into Asia Minor, Africa, Cyprus, Spain, and the Byzantine Empire (Nemoy 1978: 604; Danon 1925: 288–289). By the second half of the twelfth century, post-second Crusade, new Karaite settlements had been established within the Byzantine Empire (Astren 2004: 124; Danon 1925: 290). It is generally acknowledged that the movement in Constantinople subsequently shifted to Crimea (Zajączkowski 1961: 36). As will be further discussed in the succeeding section (1.2.2.), through the Turkic Karaims, Karaite religion later expanded into Lithuania, Poland, Russia, and present-day Ukraine. At this point, it is worth distinguishing the Turkic (the Eastern European Karaites) from the non-Turkic believers of the Karaite religion, since the terms that have been used in the literature might be confusing. Anan's followers were first called 'Ananites' (Nemoy 1978: 604; Poznański 1915: 662), whereas some centuries later they were known as 'Karaim' or 'Karaite' in the literature⁹. In their Turkic vernacular, Karaims call themselves as *karajlar*

⁵ Only certain fragments of this book have survived (Poznański 1915: 663; Nemoy 1952: 8).

⁶ It is worth noting that the main principles of Karaism i.e., the rejection of the authority of the post-biblical oral tradition were not a brand-new approach in this area, since some similar ideas can be traced back to the early sects of Judaism. For instance, through the discovery of the Dead Sea Scrolls, a connection has been noted between the Karaites and the Essenes, who flourished in Palestine between the second century BC and first century AD (Csató 2006: 392). At the same time, Karaites are also considered to have borrowed some modifications from the Sadducees and the Boethusians (Kohler and Harkavy 1904: 438; Poznański 1915: 662; Astren 2003: 39).

⁷ The Tanakh, recognized by Christians as the 'Old Testament', comprises three constituent sections: the Torah (Pentateuch or five Books of Moses), the Nevi'im (Prophets) and the Ketuvim (Writings).

⁸ For further details regarding the principles of Karaim religion, see Kuzgun 2015: 277–301; Schur 1992: 13–57; Astren 2004: 23–123.

⁹ According to some scholars, the term 'Karaite' was first used by Benjamin Nahewendi (Lewis 1956: 315; Kohler and Harkavy 1904: 438)

(singular: *karaj*) which was derived from the Hebrew-Aramaic stem קרא 'to read'¹⁰ and stands for 'readers', referring to people who (only) read the Holy Scripture. In most European languages, the Indo-European linguistic milieu suffix {-it} has been also attached to this Hebrew-Aramaic stem and appears as Karaite to denote the believers of this religion (Harvianien 2003: 634). On the other hand, the word Karaim is the Hebrew plural form of the aforementioned word and is also widely used in other languages¹¹. It is worth noting that apart from *karay*, *karaylar*, the Turkic believers prefer the word Karaim, since the word Karaite sounds pejorative in Turkic languages: *kara* 'black' *it* 'dog' (Harvianien 2003: 635). In some studies that are written in English, there is a tendency to use the term Karaim referring the Turkic believers and their language while the word Karaite was used to described the non-Turkic or all adherents of Karaite religion. However, along with Crimean Karaims, it is also possible to see the term 'Crimean Karaites' in reference to the Turkic adherents of Karaism. Nevertheless, in this study, to avoid confusion the term Karaim will be used to refer only to Turkic speaking believers of the Karaite religion, who mainly lived in Eastern Europe¹², as well as to their Turkic vernacular language.

1.2.2. History

Circa 1180, Petahyah of Regensburg, a Rabbanite traveler, documented encountering a group of individuals in the Land of Kedar, present-day Ukraine, prior to his arrival in Crimea. His account portrayed these individuals not as Rabbanic Jews, but as heretics whose Sabbath eve was spent in darkness and whose prayers consisted solely of Psalms. Additionally, these individuals communicated to Petahyah their unfamiliarity with the Talmud. Generally, this account is considered the earliest reference that potentially pertains to the Karaites in the Crimean region, given the alignment of the described characteristics with those associated with Karaite customs. (Mann 1935: 288–290; Harvianien 2003: 636). Ankori (1968: 61–64) also concurs with the supposition that this group was likely composed of Crimean Karaites, albeit at a preliterate phase in their historical progression. He further proposes that the actual

¹⁰ The verb also means 'to call', and therefore it might refer to 'missionary' or 'summoner' as well (Lewis 1956: 315).

¹¹ Therefore, within the English language, the term 'Karaims' exhibits a double plural usage. Despite this, due to its wide acceptance in scholarly literature, we will also maintain its usage.

¹² It is worth noting that the Karaites living in Turkey have also been considered non-Turkic by some scholars (see Ankori 1968: 84–85). Along with the originally Greek-speaking community, there were also Turkish-speaking communities, as some Crimean Karaites migrated to the Ottoman Empire. However, in this study, the term 'Karaim' will be used exclusively to refer to the Turkic-speaking followers of Karaism residing in Eastern Europe, primarily in Crimea, Lithuania, Poland, and Russia.

migration from the Byzantine Empire to Crimea probably ensued once the Byzantine faction of Karaism had already attained its zenith, concurrent with the fourth Crusade (1204). Apart from this account, the initial reliable documentation of Karaims presence in the Crimean region traces back to the late thirteenth century (Harviainen 2003: 636, 639). Consequently, varied perspectives emerge concerning the historical backdrop of the Crimean adherents of the Karaite faith. A prominent theory suggests that the Karaims are the progeny of the Turkic Khazars, who embraced Karaitism around the mid-eighth century, thereby introducing the faith to the Crimean region following the fall of the Khazar Empire. This theory is predominantly endorsed by Karaim scholars (e.g., Zajączkowski 1961: 12–23; Szysman 1980: 73), who highlight the similarities between the Karaim language and the Codex Cumanicus as compelling evidence supporting the contention that the Turkic-speaking Karaims descend from the Khazars and Cumans¹⁴. Contrarily, some scholars argue that the Khazars spoke a language more akin to Bolgar Turkic, while the Karaim language exhibits no vestiges of Bolgar Turkic (Togan 1964: 402; Barthold 2004: 61; Jankowski 2004: 85). Moreover, the idea of discerning ethnicity based solely on linguistic evidence may be deemed misleading. This theory also contends that the Khazars practiced the Karaite variant of Judaism, which lacks any substantiated documentary support. In fact, Golden (2001: 44) maintains that the Khazars converted to Rabbinical Judaism.

During the period when Batu Khan, the founder of the Golden Horde, launched his invasion in the 1220s, the Crimea was inhabited by a variety of ethnic groups, including Greek, Italian and Frankish minorities, Armenians, and some nomadic Turkic tribes, primarily located in the northern regions. In the middle of the thirteenth century, Turkic people rose to political prominence over the aforementioned groups under the Golden Horde's rule. This period also saw some Anatolian Turks settle in the region (Fisher 1978: 1–2). Subsequently, the Tatars established the first Crimean Khanate in the early 1440s. From the fifteenth century until the Russian annexation of Crimea in 1783, the Ottoman Empire exerted control over the region. During this time, Crimean Karaims enjoyed a privileged status in the Crimean Khanate, both economically and judicially (Zajączkowski 1978: 608). Post the Crusades, as other Karaite centers began to wane in their influence, Crimea emerged as a principal hub of Karaism (Kuzgun 2015: 223). Crimean Karaims were predominantly located in Solkhat (Eski Qırım,

-

¹³ The question of whether the Turkic people in Crimea converted to the Karaim religion or if the non-Turkic Karaites began to speak the Turkic vernacular is another matter. For further examination, see Harviainen 2013: 636–643.

¹⁴ Zajączkowski (1961: 39–40) cites excerpts from the Lord's Prayer in the Codex Cumanicus, noting that the sentences are almost entirely comprehensible in Karaim.

Staryi Krym), Caffa (Kefe, Feodosia), Chufut-Kale (Qale, Qırkyer), and Mangup. The Muslim community later migrated to Bakhchysarai (Bahçesaray; lit. 'Garden Palace'), constructed by the khan Mengli Giray. Consequently, Chufut Kale (lit. 'Jewish Castle') transformed into a city inhabited primarily by non-Muslims, including Jews (predominantly Karaims) and Armenians, (Smętek 2012: 11–12). In subsequent centuries, Gözleve (also known as Kezlev, Közlev, or Eupatoria) and Karasuv Bazar (currently Bilohirsk, previously known as Belogorsk) also emerged as important centers for the Karaim community in Crimea (Shapira 2003: 726).

It is generally accepted that during the fourteenth century, communities of Karaims emerged in the Grand Duchy of Lithuania, particularly in Trakai. The presence of Karaims in Crimea is a matter of historical debate, as is their emergence in this northern region. According to numerous sources, Crimean Karaims migrated to the Grand Duchy of Lithuania, predominantly to Trakai, at the end of the fourteenth century, later disseminating to other Lithuanian towns such as Volhynia and Podolia, currently known as Halich (Poznański 1915: 669; Schur 1992: 107; Kowalski 1929: xvi-xviii; Dubiński 1991: 216; Wexler 1983: 29–30; Jankowski 2008: 165–166). In contrast, Kizilov (2009: 30–38) labels this narrative as Karaim mythology, attributing its origin to a falsified copy of a chronicle by Joseph Solomon Lutski. Nevertheless, Kizilov concedes that the Karaims most likely made their first appearance at the dawn of the fifteenth century in Trakai, Lutsk, and Lvov, cities that were crucial for trade with Crimea and eastwards. This perspective is echoed by Shapira, who argued that the Karaim community in the west did not migrate from Crimea, but were rather descendants of the Golden Horde. These shared views have led to further questions regarding the existence of a unique Crimean Karaim language (see 1.2.3.1.).

Regarding the following centuries, it must be mentioned that the Russian annexation of Crimea in 1783 instigated substantial transformations in the lives of the Crimean Karaims. The community initially strived to regain their privileges within the new Russian administration. Their religion was formally recognized in 1837, and by 1852, they were granted permission to establish residences anywhere within the Russian Empire. In 1863, they were awarded full privileges¹⁵ identical to those of the Christian inhabitants of Russian lands (Prohorov 2013: 209). Nonetheless, the upheavals of the 1917 revolution, along with the turbulence of World War II, compelled many Crimean Karaims to vacate their settlements (Gülsevin 2016: 15). The

¹⁵ These privileges marked the first legal distinction between Rabbanites and Crimean Karaims. However, Crimean Karaims sought these privileges largely for economic considerations, while they still regarded each other as brethren in faith (Miller 1993: 41), mirroring the dynamic from previous centuries.

advent of the Soviet regime brought about significant alterations to their lifestyle. Obligatory abdication of their religious rituals, closure of their prayer houses (*kenesa*), and the cessation of the long-standing tradition of Bible translation were among the stringent changes implemented. The so-called *Qaray Bitikligi*, known for its extensive collection of Karaim manuscripts, was also shuttered, and its invaluable collections were relocated to libraries in Russia, Poland, and Lithuania.

Currently, communities of Karaims are geographically dispersed, with primary population clusters identified in Russia (notably St. Petersburg and Moscow), Ukraine (Odessa and Kiev), Lithuania (Vilnius and Troki), and Poland (Warsaw, Wroclaw, Gdansk) (Smętek 2012: 14). Furthermore, Karaim settlements have been discovered in various countries such as Turkey, the United States, Switzerland, France, the United Kingdom, Belgium, Egypt, Israel, and Azerbaijan (Gülsevin 2013: 15). While precise population estimates remain challenging to ascertain, reports suggest that there are approximately 1,100 Karaims in Crimea, less than 300 in Lithuania and Poland, and globally the population probably does not exceed 5,000 (Jankowski 2015b: 452).

1.2.3. Language

The Karaim language is part of the Kipchak (Northwestern) branch of Turkic languages. A geographical delineation allows for the categorization of the language into Western and Eastern (Crimean) dialects. Western Karaim comprises two sub-branches: the Northwestern sub-branch, or Trakai Karaim, referring to the dialect of communities currently located in Lithuania and Poland; and the Southwestern sub-branch, represented by the extinct Łuck and Halich Karaim dialects previously spoken in modern-day Ukraine. In contrast, Eastern Karaim denotes the extinct dialect native to the Crimean region. The schema of the dialect classification, adapted from Németh (2011b: 11), is displayed below¹⁶.

¹⁶ For further details on the Crimean Karaim variants, see section 1.2.3.2.

Table 1: Breakdown of Karaim Dialects and Their Variants

| Karaim | Eastern | Crimean Kipchak Karaim | |
|--------|------------------|--|---------------|
| | (Crimean Karaim) | Crimean Tatar Karaim | |
| | | Crimean Turkish Karaim (in the Crimea) | |
| | | Crimean Turkish Karaim (in the Ottoman Empire) | |
| | Western | Southwestern Karaim | Halich Karaim |
| | | | Łuck Karaim |
| | | Northwestern Karaim = Troki = Trakai-Vilnius | |

In accordance with Prohorov's research (2013: 210), during 1897, 21% of Crimean Karaims designated Russian as their native language, whereas 70% identified Karaim as their mother tongue. However, by the mid-twentieth century, as indicated by Prik in her Crimean Karaim grammar book (originally written in 1949 but published in 1976), the utilization of the native language by the Karaims had significantly dwindled. This suggests that the language shift among the Crimean Karaims potentially accelerated from the onset of the twentieth century. By 2004, the number of individuals reporting some proficiency in this dialect had dwindled to a mere few (Jankowski 2004: 87–88), leading to the dialect's complete extinction today. In relation to Western Karaim, it has been recorded that in 2006, only 50 individuals were using the Trakai dialect in their daily interactions, while fewer than 10 were conversant in the Halich dialect (Csató 2006: 395). Presently, however, the Halich dialect has no known speakers (Jankowski 2015: 453), and the Trakai Karaim dialect is likewise on the brink of extinction

The distinction between the western and eastern branches of the Karaim language is significant, characterized by an array of differences attributable to regional influences that have evolved over centuries. Jankowski (2015b: 462) posits that the western and eastern Karaim variants exhibit such a multitude of disparities that they might be considered two distinct languages. Contrary to Eastern Karaim (Crimean Karaim dialect), the western branch has experienced prolonged isolation from other Turkic languages, resulting in the preservation of many archaic characteristics of Kipchak Turkic. These include several resemblances with the Cuman language, as documented in the Codex Cumanicus, as well as with Armeno Kipchak (Kowalski 1929: lix-lxv, lxvi-xxi; Zajączkowski 1961: 38–40).

The status of Crimean Karaim is a subject of considerable complexity and scholarly debate with respect to its linguistic characteristics. Some scholars argue that the dialect exhibits numerous archaic features of West Kipchak, while also displaying influences from Oghuzic due to the geographical context of the Crimean region (Prik 1976: 8; Jankowski 2015b: 453).

However, it is worth noting that the existence of the eastern dialect and its archaic traits have been contested by certain scholars who posit that Crimean Karaim never existed and that the Karaim language was originally confined to the western regions. The subsequent section will examine this matter in detail.

1.2.3.1. Crimean Karaim and the Debates on Its Existence

The Ottoman conquest of Caffa (present-day Feodosia) in 1475 is widely recognized as a significant event that introduced numerous Oghuzic characteristics into the Turkic languages spoken in the Crimean region (Doerfer 1959: 272–280; Schönig 2010: 107–119). Given the interaction between Crimean Karaims and various Turkic-speaking communities, such as Crimean Tatar, Crimean Turkish, Krimchak, Urum, and Nogay, the language naturally assimilated regional influences. Consequently, some scholars have argued that Crimean Karaim and Crimean Tatar were not distinct languages. Radloff (1896: xvi), for instance, shared written and oral materials and posited that Crimean Karaim was identical to Crimean Tatar or Crimean Turkish. However, Radloff's viewpoint was challenged by Samoylovič (2000: 116) in 1917, who asserted that Radloff's materials only represented literature modified under Ottoman and Crimean Karaim influences and could not accurately depict spoken Crimean Karaim. Prik, who extensively documented the grammar of the Crimean Karaim dialect based on a small number of speakers in 1949, also noted significant differences between spoken Karaim and Radloff's materials from 1896 (Prik 1976: 16). Furthermore, Prik observed numerous characteristics of the old canonical language in Crimean Karaim, which were preserved in the contemporary spoken western branch of Karaim (Prik 1976: 9–10) based on samples from Bible translations. Similar perspectives were shared by Pritsak (1959: 320–321) and Musaev (1964: 36), who did not include the Crimean Karaim dialect in their grammars but acknowledged its historical existence and proposed that it had been assimilated by Crimean Tatar, losing its distinctive features over time.

After a long hiatus, scholars have begun to study Karaim anew and analyze both eastern and western Karaim texts, thus providing further detail to ongoing debates. Among these scholars, Shapira has reignited previous discussions with his views, which have been contested by many others. Specifically, he has asserted (2003: 661–665; 2013: 155–157) that Crimean Karaim was a ghost dialect that never existed. According to Shapira, the Crimean Karaim texts exhibiting characteristics of the old canonical language were originally composed in the west and brought by northern Karaims (from Lutsk, Kiev, and Troki), which explains the archaic features found

in these texts, primarily in Bible translations (Shapira 2013: 150–157)¹⁷. Shapira based his arguments on the Book of Nehamia from the Gözleve Bible (henceforth referred to as Göz. 1841), the subject of our study. He compared this translation with equivalent portions of the Łuck-Karaim text. Below, we share a part of his comparison (Shapira 2013: 165), where he used a transliteration system for Karaim translations.

Table 2: Comparison between Göz. 1841 and Mid-Eighteenth Century Łuck-Karaim (Shapira 2013: 165)

| English Translation | The mid-18th century Łuck-Karaim | Göz. 1841 |
|-------------------------------|--|-------------------------------------|
| Honor/fame is to such a man, | sandır andıy k'išeg'e ki yür'um'es'e | san ol kišiga ki yurumadi kegaši |
| who does not walk with the | k'en'es'i bıla raša'larnın da yolunda | bilan rašaʻlarnig da yolinda |
| counsel of the wicked ones, | yazıqlırarnın tÿurmasa da oltÿurusunda | yaziqlilarnig turmadi da-oturašinda |
| and does not stand in the way | eliqcilarnin oltyurmasa | eriklavčilarnig oṭurmadi. |
| of the sinners, and does not | | |
| sit in the sitting(-place) of | | |
| mockers. | | |

In consequence of his analysis, he posits that these discrepancies can be primarily attributed to phonetic variations, largely influenced by the intrusion of local Slavic languages in the more contemporary translation. He hence posits that the disparities should not be ascribed to geographical diversities, but rather to chronological divergences. Additionally, he contends that the Western Karaim community did not originate from Crimea, but instead, were presumably progenies of the Golden Horde. The late 15th-century immigrants from the west, he suggests, brought their unique vernacular to Crimea (Shapira 2013: 152, 157). In this context, he postulates that the presence of Slavic lexical elements in what are purported to be Crimean Karaim texts lends further support to the hypothesis that the existence of a distinct Crimean Karaim language is improbable. The logic follows that if such a language did indeed exist, it would be devoid of Slavic influence. His interpretation of the Göz. 1841 provides additional substantiation for these assertions (Shapira 2003: 696):

'the so-called "Tirishqan translation" or "Gözleve Bible" was not a new translation, but rather a hasty attempt to Tatarize—or even vulgarize—earlier translations existing in

_

¹⁷ Furthermore, Shapira (2003: 662) misquoted Musaev (1964: 36–37) and Pritsak (1959: 320), falsely stating that they considered Crimean Karaim as a ghost dialect, thereby excluding it from their grammars. In reality, Musaev and Pritsak acknowledged the existence of Crimean Karaim but suggested that it had been assimilated over time, leading to its exclusion from their grammars. Notably, Shapira's erroneous statement has been refuted by scholars, including Jankowski (2008: 166) and Németh (2016: 209–210).

manuscripts. The main goal of this edition was to eradicate the Karaim-tinged "language of the educated (i.e., savants)" not fully understood by the Gözleve nouveaux riches, like Tirishqan himself or Simḥah Babowicz. Lacking genuine Crimean-Tatar manuscripts, the editors took Karaim ones brought apparently from Łuck, and changed some grammatical forms from Karaim to "Tatar," whatever this ambiguous term might mean, sometimes also substituting some Karaim words with their Tatar equivalents. The result was rather superficial Tatarization'.

Similar ideas have also been expressed by Kizilov. He argues that there is no concrete evidence supporting the belief that the initial Karaites in Crimea spoke a Kipchak Karaim language, despite its theoretical plausibility. Indeed, even if this were the case, Kizilov posits that the Crimean Karaims likely shifted their language to Crimean Tatar no later than the late fifteenth or early sixteenth century. Thus, he concurs that all known Karaite Turkic texts from the seventeenth to nineteenth centuries exhibit characteristics of different variants of Crimean Tatar and Crimean Turkish (Kizilov 2013: 247–248).

However, some scholars, including Jankowski (2003a, 2008, 2015a, 2015b), Aqtay (2009: 16-18), and Németh (2015a: 179-180; 2015d, 2016), have contested Shapira's ideas. The primary disagreement between these opposing viewpoints stems from the controversy surrounding the emergence of Karaim settlements in northern areas. Indeed, the concept that the northern settlement of the Karaims spread from Crimea is much more prevalent among scholars, as previously demonstrated (see 1.2.2.). Furthermore, Shapira's argument suggesting the existence of Slavic lexical elements in eastern texts as evidence that Crimean Karaim did not exist seems erroneous. Firstly, these texts do not contain a significant number of Slavic lexemes. Secondly, as Jankowski also noted (2008: 166–167), some Slavic loanwords were already present in Middle Turkic languages as far as Central Asia, as well as in the Codex Cumanicus. Echoing prior theories, Jankowski (2009: 502) asserts that after migrating to Crimea, the originally Greek-speaking Karaim believers adopted a Kipchak Turkic language that was quite similar to the Kipchak Turkic present in the Codex Cumanicus. Over time, however, they adapted their language to Crimean Tatar or Turkish, as these Turkic languages were used by their overlords (Jankowski 2015a: 200). Thus, considering the historical context, he delineates the progression of the language chain as follows: Arabic \rightarrow Greek \rightarrow Kipchak Turkic \rightarrow Crimean Turkish \rightarrow Russian (Jankowski 2015a: 201).

In 2014, Németh conducted a detailed examination and presentation of the oldest Western Karaim translation (expressed in a northern dialect), originating from the year 1720. This

manuscript, denoted as A.Dub.III.73, encompasses translations of the Torah and four other books from the Ketuvim: namely the Book of Ruth, the Book of Jeremiah, Ecclesiastes, and the Book of Esther (Németh 2014c). Subsequently, in a comprehensive study (Németh 2015c), he juxtaposed the language employed in the Book of Ruth in the A.Dub.III.73 manuscript with equivalent segments of the Göz. 1841. Through this comparison, he demonstrates that the variances observed between these Western and Eastern Karaim translations were dialectal, largely attributable to the Oghuzic influence in the Crimea, e.g., A.Dub.III.73 oltur- 'to sit; to dwell' vs. otur- 'id.' (Göz. 1841), tuv- 'to be born' vs. doy- 'id.' (Göz. 1841), the so-called negative infinitive, i.e. W.Kar {+maska} vs. E.Kar {+manakka}, etc. (Németh 2015c: 105). However, it merits noting that despite exhibiting Oghuzic traits, the Crimean translation predominantly manifests Kipchak characteristics, typical of Crimean Kipchak Karaim (see section 1.2.3.2.), e.g., ber- 'to give' (Ruth 1:6), bol- 'to be' (Ruth 1:2), kel- 'to come' (Ruth 1:2), the optative {-yay} (Németh 2015c: 105). Consequently, he elucidates a few Northern/Western Karaim attributes that were not previously discerned in Crimean Karaim. For example, W.Kar bašak¹⁸ 'ear of grain', and the usage of optative {-yay}, e.g., kilyaysin 'may you do'. Ultimately, Németh postulates that the Book of Ruth in the Göz. 1841 was potentially constructed based on a preceding Western Karaim version or another Eastern Karaim manuscript, which was, in turn, derived from a Western Karaim prototype.

Table 3: Comparison of the Book of Ruth in the A.Dub.III.73 and Göz. 1841 (Németh 2015c: 111)

| Ruth | English Standard Version | A.Dub.III.73 (1720, W.Kar) | Göz. 1841 (E.Kar) |
|------|---------------------------------|---|---|
| 2:10 | Then she fell on her face, | Da ťušťu juzĺari üsťuńa | Da tüštü jüzläri üstünä bašurdu |
| | bowing to the ground, and said | bašurdu jerģa da ajtty anar ne | jergä da ajtty aŋar ne učun taptym |
| | to him, "Why have I found | üčun taptym širinlik kozlarij <u>ďa</u> | širinlik közläriŋ <u>ä</u> tanyma meni da |
| | favor in your eyes, that you | tanyma meni da men jat <u>qatyn</u> | men jat. |
| | should take notice of me, since | men. | |
| | I am a foreigner? | | |

Subsequently, a year later, Németh (2016) introduced crucial new data that enriched the discussion and shed light on specific theories concerning the language of Göz. 1841. In his

¹⁸ Németh asserts that this word was not recorded in the most recognized Crimean Karaim dictionaries such as those by CKED, Çulha 2006, and Chafuz 1995. However, it bears emphasis that the word was listed for Ottoman Turkish (and remains in contemporary Turkish) in Meninski's dictionary (TLO I: 664), published in 1680. As such, the word could have been adopted from Ottoman Turkish, either directly or via Crimean Tatar

study, he unveiled the Book of Ruth from an Eastern Karaim translation, which was transcribed no later than 1687¹⁹, thus marking the oldest Eastern Karaim translation identified to date. In addition, he juxtaposed this translation with identical portions of the oldest Western translation (A.Dub.III.73) and the Göz. 1841. One remarkable outcome was the striking similarities between the JSul.III.02 and the Göz. 1841. The distinctions between Eastern and Western translations were, yet again, dialectal, encompassing aspects such as phonology, morphonology, morphology, syntax, and lexicon, e.g., the genitive case {+nin} (E.Kar) vs. {+nin} (W.Kar), the so-called 'negative infinitive' {+makka} vs. {+maska} (W.Kar) etc. (Németh 2016: 199–200). As such, Shapira's hypothesis, positing that the transcriber of the Göz. 1841 hastily altered the language to fit a phantom dialect, was refuted. This is because these adaptations are already evident in another Crimean Karaim Bible translation written 200 years prior to the Göz. 1841. Furthermore, Németh also mentions that the Crimean Tatar loanwords were not present in the Book of Ruth translations of the JSul.III.02 and Göz. 1841. However, it is possible to identify certain specific Karaim features that are absent in Crimean Tatar: for instance, *inir* 'evening', *tavus*- 'to complete', the {-yay} optative, or the {+ka} suffix in the negative infinitive form {+makka} (Németh 2016: 201). Hence, Németh (2016: 202– 203) claims that it is misleading to assert that the Göz. 1841 was 'Tatarized' or 'vulgarized'; rather, it was 'Ottomanized' in response to the linguistic trends of the era.

Table 4: Certain Differences Between Jsul.III.02, Göz. 1841 and III. 73 (Németh 2016: 203)

| Line nr | Jsul.III.02 (E.Kar) | Göz. 1841 (E.Kar) | III. 73 (W.Kar) |
|---------|---------------------|-------------------|------------------|
| 4 | edi | Boldy | edi |
| 6 | aty | ady | aty |
| 31 | asry | gajet | - |
| 43 | barča | <i>žümlä</i> | bar |
| 60 | keldi | keläjdir | keldi |
| 73 | qajdan | neden | qajdan |
| 84 | kibik | gibi | kibik |
| 106 | učramayajlar | učramasynlar | ki jadatmayajlar |

As for the large number of typical Western Karaim features found in Eastern texts, Németh posits that the Eastern Karaim translation of the Book of Ruth might have been prepared based on an earlier Western Karaim manuscript. However, as he also notes, given that the oldest

¹⁹ For the evidence of this theory, see Németh 2016: 166–167.

known Western Karaim Bible translation identified to date is actually more recent than the oldest Eastern Karaim Bible, this hypothesis remains uncertain. As an alternative explanation, he cites the idea of an Old Karaim linguistic heritage, a theory upheld by certain scholars such as Gordlevskij (1928: 87–88) and Dubiński (1993: 37–38). Lastly, he asserts that even if the Crimean texts were based on Western Karaim texts, this does not provide sufficient evidence to declare Crimean Karaim a ghost dialect, since the canonical language and the spoken language should not be regarded as identical (Németh 2016: 211).

Concerning these debates, it is crucial to acknowledge several pivotal considerations prior to discussing Crimean Karaim. Primarily, the language in supposed Crimean Karaim written sources was not designated as Karaim/Crimean Karaim, as Karaim communities, including those in the west, did not utilize a specific term²⁰ for their Turkic vernacular for an extended period. In the Hebrew script, their language was commonly referred to as lešon 'language' gedar 'Crimea/Black Sea region', for instance, Lešon Qedar, lašon Qedari, bisfat Qedar, or yišma 'eli 'Ishmailite' (Jankowski 2015b: 454). In fact, the label lešon Tatar was also applied to the Göz. 1841, the subject of our dissertation. According to Jankowski (2018: 52), the language of this edition would have been barely decipherable to Crimean or Volga Tatars in the nineteenth century, indicating that classifying written sources based on their labels may lead to misconceptions. Another pivotal consideration is that the geographical origins of the publications may not yield precise insights into the language of the Karaim dialects. Consequently, a text published in Crimea should not be designated as Crimean Karaim solely on this basis (Jankowski 2015a: 202). In terms of the spoken language, extant colloquial material is significantly limited, for instance, Prik 1976, Jankowski 2003a, considering Crimean Karaim now exists as an extinct dialect of Karaim. Hence, scholars can only evaluate written resources, which may occasionally fail to shed light on the intricacies of the colloquial language (Smetek 2012: 18).

1.2.3.2. Crimean Karaim Varieties

In this section, the varieties of Crimean Karaim will be succinctly described, with reference to Jankowski's paper (2015a) which stands as the most comprehensive study on the subject to

²⁰ However, drawing on certain sources (see, Jankowski 2018: 40), it is known that Crimean Karaims historically referred to their Turkic vernacular as 'Chaltai', a term that traces its origins to Chaghatai, a literary language used in Central Asia between the fifteenth and twentieth centuries.

date. As illustrated below, Németh (2011b) divides Crimean Karaim varieties into four subcategories:

- a) Crimean Kipchak Karaim
- b) Crimean Tatar Karaim
- c) Crimean Turkish Karaim (in the Crimea)
- d) Crimean Turkish Karaim (in the Ottoman Empire)

Given that the language and dialects were not standardized, it remains challenging to articulate the precise nature of these varieties. However, certain distinct features can be identified. Among these, Crimean Kipchak Karaim is the variety displaying the most prominent Kipchak characteristics, along with some Oghuzic influence. This variety, predominantly manifested in Bible translations, maintains more consistency compared to the other varieties²¹. Several common features are discernible in the Bible translations from Crimea. Drawing upon Jankowski's delineation (2015a: 202–204), these features can be listed as follows:

Table 5: Some features of Crimean Kipchak Karaim

| | Crimean Kipchak Karaim | |
|--|--|--|
| Phonological and | Rounded-unrounded vowel harmony is inconsistent, e.g., koyyïn 'put'. | |
| Morphohonological | The lenition of voiceless stops k , k , and p is present in verbs between vowels, e.g., $\ddot{c}i\gamma ar$ | |
| Features | Features 'to push out', except for -t-, e.g., etär 'she/he/it does'. | |
| The initial t- and k- are retained, along with b- in bar- 'to go'; bar 'there is/are'; ber- 't | | |
| | give', bol- 'to be; to become'. | |
| Morphological | The genitive is $\{+nI_{\mathfrak{I}}\}$, the accusative is $\{+nI_{\mathfrak{I}}\}$, and the dative is $\{+GA_{\mathfrak{I}}\}$. | |
| Features The imperative is {-QIn}, and the subjunctive is {-QAy}. | | |
| | Some actional forms such as kimilday turyan 'she/he/it is moving' are present. | |

It is important to note that some features of this dialect have evolved over time in response to the prevailing linguistic trends. However, Jankowski asserts that the Crimean Kipchak Karaim and Western Karaim translations (encompassing both southern and northern sub-branches) exhibit clear similarities. Despite this, there are many features found in Western Karaim Bible

²¹ Beyond Bible translations, there exist two other types of text in this dialect. The first is a prayer book, namely the *Targum Seliḥot*, which was published in Crimea in 1734 and likely translated by Simha Isaac Łucki. The second is a translation of the Passover Haggadah, referred to as the *Targum hallel haqqatan*, which was also published in Gözleve. The specifics concerning the date and translator of this latter translation remain unknown (Jankowski 2015a: 202–203).

translations that are absent in Crimean Kipchak Karaim, e.g., W.Kar *oltur*- 'to sit; to dwell' vs E.Kar *otur*- 'id.', *ošol* 'that over there' vs E.Kar *šol* 'id', W.Kar *-bïla* ~ *-bile* 'with' E.Kar *bilän*, W.Kar *-doyač* 'when ...' (Jankowski 2015a: 204).

Another variant, Crimean Tatar Karaim, is predominantly seen in literary works (e.g., short songs, poems, and the so-called mejumas), which were quite popular among the Crimean Turks and Tatars. According to Jankowski, the abundance of Crimean Tatar features makes it challenging to categorize this variant as Crimean Karaim. Notable examples are attested in various studies on mejumas, such as Aqtay 2009, Çulha 2010, and Jankowski 2013. Our understanding of Crimean Turkish Karaim is also limited. The initial examples of this dialect are found in a prayer book published in Venice in 1528/1529²². Subsequent attestations of this dialect are also presented in works by Shapira²³ (2001: 79–92), Poznański (1913), Jankowski (2012), and Aqtay (2009: 102–113, 224–294). As per Jankowski (2015b: 460–461), a variety of examples (primarily translations or adaptations) were penned in the ensuing centuries by Western Karaim authors who were active in both Turkey and the Crimea. Consequently, the language primarily consists of Turkish, along with Crimean and even Western Karaim elements. The most substantial of these is a Bible translation printed between 1832 and 1835 in Ortaköy (Istanbul) under the supervision of Abraham Firkovich (see section 1.2.4.1.).

Table 6: Some Features of Crimean Turkish Karaim (based on Jankowski 2015a: 204–205)

| | Crimean Turkish Karaim | | |
|-------------------|--|--|--|
| Phonological and | Initial t -, k - are voiced to d -, g -, which is also present for some words that do not show | | |
| Morphohonological | voicing in standard Turkish but do in Crimean Turkish, e.g., tut- vs dut- 'to hold, to | | |
| Features | grasp', taš vs daš 'stone'; kendi vs gendi '-self'. | | |
| | The deletion of the initial b- is evident in words such as ol- 'to be; to become', ilen | | |
| | 'with', or its spirantization, var- 'to go', ver- 'to give', and var 'there is/are'. | | |
| | Contrasting with Turkish, the change from $k \to b$ (most commonly denoted as h) is | | |
| | evident, e.g., $ka\check{c}an \rightarrow ha\check{c}an$ 'when', $yakin \rightarrow yahin$ 'near'. | | |
| Morphological | The genitive is $\{+(n)I\eta\}$, the accusative is $\{+(y)I\}$, and the dative is $\{+(y)A\}$. | | |
| Features | It is possible to find the perfect {-mIš}, and progressive {-yUr}. | | |
| | The usage of the converb {-DIQčAz} 'as long as' is present. | | |
| | The usage of the converb {-InčAz}, 'until' is attestable. | | |
| | The usage of the converb {-(y)AndA}, 'when' is present. | | |

²² The language of this prayer book was mistakenly described as Crimean Kipchak Karaim by Zajączkowski (1964: 793) and this reference was cited over time by many authors until Shapira (2003: 691–692), who describes the language as 'New Greek with some Turkish phrases not present in any kind of Karaim'.

²³ He defines this dialect as Crimean Tatar.

| | Crimean Turkish Karaim |
|---|---|
| | The present or progressive {-AyIr} and { -Uyır} is attestable. |
| Lexical Features The usage of čok instead of köp 'much, many'; kapu instead of ešik 'door'; gizle- instead | |
| | of yašīr- 'to hide'; millet instead of ulus 'nation'; hakka instead of kerti 'truly'. |

1.2.4. Written Sources

Written sources serve as critical materials that reflect the fundamental beliefs, traditions, rituals, and perspectives of their respective communities. The first study of Karaim written sources was undertaken by Poznański (1909, 1910, 1913), whose work did not provide a detailed classification of the dialects present in his sources. Subsequent studies, including those by Zajączkowski (1926), Shapira (2003a), Walfish (2003), and Sklare (2003), primarily built upon Poznański's foundational work.

Shapira (2003: 692) indicates that the earliest sources trace back to the mid-seventeenth century and encompass a variety of texts, including letters, notes, poems, glossaries, and fragments of Hebrew dictionaries. Until the advent of Karaim printing houses in the nineteenth century, comprehensive works were scarce. The nineteenth century, however, was considered a golden age of Crimean Karaim literature due to the proliferation of printed materials (Smętek 2009: 20). Yet, significant changes in their lifestyle in the twentieth century led to a decline in literary activities and the cessation of the longstanding tradition of Bible translations.

The available written materials can be broadly classified into two categories. The first group comprises compilations of literary materials, while the second primarily consists of translations of canonical texts and liturgies. Given that the present study is centered on the canonical materials, the following section will elaborate on these sources.

1.2.4.1. Crimean Karaim Bible Translations and Associated Studies

Given the nature of Karaim beliefs, it has always been essential for Karaims to read the Hebrew Bible in their vernacular. Consequently, Karaims have been translating the Holy Scripture into their Turkic vernacular for centuries²⁴.

The Hebrew Bible, also known as the Tanakh, is divided into three sections: the Torah, Nevi'im, and Ketuvim²⁵, each comprising various books of the Bible.

Table 7: The Books of Hebrew Bible (Tanakh)

-

²⁴ Additionally, they also employed their own language in liturgical practices (Zajączkowski 1961: 21).

²⁵ Indeed, the Hebrew Bible is commonly referred to as the Tanakh, an acronym derived from these three sections.

| Torah (Pentateuch; Five Books of Moses) | | Genesis, Exodus, Leviticus, Numbers, Deuteronomy |
|---|---------------------------|--|
| Neviim | The Former Prophets | Joshua, Judges, Samuel, Kings |
| | The Latter Prophets | Isaiah, Jeremiah, Ezekiel |
| | The Twelve Minor Prophets | Hosea, Joel, Amos, Obadiah, Jonah, Micah, Nahum, |
| | | Habakkuk, Zephaniah, Haggai, Zechariah, Malachi |
| Ketuvim | The Three Poetic Books | Psalms, Proverbs, Job |
| | The Five Megillot | Song of Songs (Song of Solomon), Ruth, Lamentations, |
| | | Ecclesiastes, Esther |
| | Other Books | Daniel, Ezra, Nehemiah, Chronicles |

The exact dates of the Bible translations remain elusive due to insufficient concrete evidence. The largest collection of Karaim manuscripts, which contain parts of Bible translations, resides at the Institute of Oriental Manuscripts of the Russian Academy of Sciences in St. Petersburg. A significant number of these manuscripts were transferred there in 1928 or 1929 from the Crimean Karaim Library (*Qaray Bitikligi*). Established in the nineteenth century, this library was later closed by the Soviet Union (Jankowski 2018: 48). As noted by Sklare (2003: 904), there are 413 manuscripts from Gözleve at this institute. The renowned Karaim-Russian-Polish Dictionary (KRPS) enumerates 30 of these manuscripts, inclusive of the Tanakh translations into Karaim (KRPS: 28–29), with seven of these described in detail by Jankowski (2018: 49–50). In addition to the collections at the aforementioned institute, Jankowski identified several significant manuscripts held in various other libraries, including BSMS 288 (Cambridge University Library), H 170 (John Rylands Library, Manchester), and Or. Ms. 629 (Edinburgh University Library). The remaining unspecified manuscripts are largely maintained in private collections.

Beyond manuscripts, it is known that two printed editions have also been identified²⁶. The first, Göz. 1841, has a translation of the Book of Leviticus and is the subject of the present study (see 1.3.). The second is a translation of the Torah published in Ortaköy/Istanbul between 1832 and 1835. This edition displays predominantly Turkish features interspersed with some Karaim characteristics (Jankowski 2009: 516).

One of the first studies on Crimean Karaim Bible translations was conducted in 1826 by Ebenezer Henderson, who published a brief fragment of the Tanakh in Crimean Karaim. He examined the manuscript BSMS 288, which was purchased in 1816 by Robert Pinkerton

²⁶ In addition, the so-called Jeremiah 1873, which was printed in Odessa, is listed in this group by Walfish and Kizilov (2011: 387). However, this source is not available for analysis and no further details have been provided (see Jankowski 2018: 52).

(Henderson 1816: 332). Henderson also identified certain translation strategies influenced by Aramaic Targums and rabbinic interpretations. Later, Steinschneider (1871: 38) presented the first two verses²⁷ of the Book of Genesis from the four-volume manuscript H 170, currently held at the John Rylands Library in Manchester. Subsequently, a short fragment was analyzed by Harkavy and Strack, who provided a fragment of Lev. 1:1–3 from the manuscript Evr. I 143 and 144 in their catalogue (Harkavy and Strack 1875: 168).

The research initiated by Henderson was continued by Gordlevskij (1928), who studied a Crimean Karaim Tanakh translation. Subsequently, Kowalski presented brief fragments from the Göz. 1841 (1929: 287–288). Additionally, a prayer book from 1734, known as the *Targum Selihot*²⁸, was also edited by Józef Sulimowicz (1972).

After an extended period of inactivity in the field, Jankowski revived the academic exploration of the Karaim Bible translations with a pivotal study in 1997. He produced both a transcription and transliteration of select portions of the H 170 manuscript, meticulously examining the text's grammar from a phonological, morphosyntactical, and lexical perspective. Notably, he characterized the manuscript's language as a northern dialect of Crimean Karaim.

In the subsequent years, Dan Shapira released several articles examining Crimean Karaim Tanakh translations. His 2006 study (264–270) presented comparative analysis of translations of Genesis 1:1-4 sourced from the H 170 manuscript shared by Jankowski (1997), the Göz. 1841 edition, the BSMS 288 introduced by Henderson (1826), and the Ortaköy 1832–1835 translation. More recent scholarship includes work by Shapira (2013) and Németh (2015c, 2016), who analyzed sections of the Göz. 1841. Alongside this, Olach (2016) conducted a linguistic comparison of the Song of Moses from the Karaim Bible translations, disseminating excerpts from the Göz. 1841 and BSMS 288. In a separate study, Smetek (2016) undertook lexical and phonological scrutiny of certain sections (the Book of Genesis, Leviticus, Job, and Psalms) of the manuscript known as BSMS 288. In 2018, Jankowski detailed the most recognized Tanakh translations of Crimean Karaim, additionally sharing fragments from the Evr. I 143, BSMS 288, Ortaköy 1832-1835, and Göz. 1841. In 2019, a critical edition of the Crimean Karaim Bible translation, encompassing the Torah, Five Scrolls, Book of Psalms, Proverbs, Job, Daniel, Ezra, and Nehemiah, from the BMS 288 manuscript, was published. This joint endeavor by academics, including Jankowski, Aqtay, Cegiołka (formerly Smętek), Çulha, and Németh, resulted in a comprehensive work spanning two volumes and including both

32

²⁷ For some concerns regarding his transcription, refer to Jankowski (2018: 44).

²⁸ Poznański (1913: 40) also provided a short fragment of this text.

transcription and translation (refer to CrKB in references). Furthermore, from 2018 to 2021, Işık has contributed several articles, each focusing on specific issues regarding the Lev of Göz 1841.

1.3. The Subject of the Dissertation

The so-called Göz. 1841 represents a comprehensive translation of the Tanakh (excluding the Chronicles) into Karaim. Published in four volumes²⁹ in Gözleve/Kezlev (modern-day Eupatoria) in 1841 (Jankowski 2018: 51), this translation was commissioned to commemorate the wedding of the future Russian Tsar Alexander II, and it simultaneously celebrated the newly granted administrative and religious rights of the Crimean Karaims (Shapira 2013: 134). Known also as the Eupatorian print or the Tirishqan translation, this edition was produced at the printing house of the renowned merchant Mordechaj Tirishkan in Eupatoria. The exact title of the translation is *Sefer Targum ha-Torah bilšon Tatar*, translating as 'The Targum Translation of the Torah in Tatar'.

According to Poznański (1916: 88), the Prophets and Writings in this publication were founded on the translation of Jacob b. Mordecai, completed in 1672. However, Shapira (2013: 135–151) contends that Jacob b. Mordecai was responsible for the copy, not the translation. Despite this, an unequivocal conclusion on the matter remains elusive. Additionally, the language of this translation underwent modernization by its editors, resulting in an inhomogeneous linguistic output. Alongside Karaim features, it displays certain Crimean Turkish traits identifiable in Crimean Tatar (Dubiński 1993: 37–38; Jankowski 2018: 52). However, as previous studies have focused solely on specific sections of the Göz. 1841, a comprehensive examination of the entire translation is warranted. In light of this, the present study aims to scrutinize the translation of the Book of Leviticus in Göz. 1841 to enrich the discussion. This text, representing the third book of the Torah, is referred to as a title that echoes the opening verse of the book, [Lev 1:1], 'And He (God) called'. Predominantly, it encapsulates laws and priestly rituals. For a detailed structural analysis, one can refer to the overview provided by Wenham (1979: 4).

Table 8: Structure of the Book of Leviticus

²⁹ טטר בלשון תורה תרגום ספר [The Book of the Targum of the Pentateuch into the Tatar Language] Vol. 1; נביאים [Translation of the Prophets] Vol. 2; מרגום ספר [Translation of the Hagiography] Vols. 3–4.

| I. Laws on Sacrifice | A. Instructions for the | 1. The burnt offering (ch. 1) | |
|--|---|--|--|
| (chs. 1:1–7:38) | Laity | 2. The cereal offering (ch. 2) | |
| | | 3. The peace offering (ch. 3) | |
| | (chs. 1:1–5:26) | 4. The purification offering (chs. 4:1–5:13) | |
| | | 5. The reparation offering (chs. 5:14–26) | |
| | B. Instructions for the | 1. The burnt offering (chs. 6:1–6) | |
| | Priests | 2. The cereal offering (chs. 6:7–11) | |
| | (chs. 6:1–7:38) | 3. The priest's cereal offering (chs. 6:12–16) | |
| | | 4. The purification offering (chs. 6:17–23) | |
| | | 5. The reparation offering (chs. 7:1–10) | |
| | | 6. The peace offering (chs. 7:11–36) | |
| | | 7. Summary (chs. 7:37–38) | |
| II. Institution of the | A. Ordination of Aaron and his Sons (ch. 8) | | |
| priesthood (chs. 8:1- | B. Aaron's First Sacrifices (ch. 9) | | |
| 10:20) | C. Judgement on Nadab and Abihu (ch. 10) | | |
| III. Uncleanness and | A. Unclean Animals (ch. 11) | | |
| its Treatment (chs. | B. Uncleanness of Childbirth (ch. 12) | | |
| 11:1–16:34) | C. Unclean Diseases (ch. 13) | | |
| | D. Cleansing of Diseases (ch. 14) | | |
| | E. Unclean Discharges (ch. 15) | | |
| | F. Purification of the Taber | rnacle from Uncleanness (ch. 16) | |
| IV. Prescriptions for | A. Basic Principles about Sacrifice and Food (ch. 17) | | |
| Practical Holiness | B. Basic Principles of Sexu | ual Behaviour (ch. 18) | |
| (chs. 17:1–27:34) C. Principles of Neighbourliness (ch.19) | | liness (ch.19) | |
| | D. Capital and other Grave Crimes (ch. 20) | | |
| | E. Rules for Priests (ch. 21) | | |
| | F. Rules about Eating Sacrifices (ch. 22) | | |
| | G. Religious Festivals (ch.23) | | |
| | H. Rules for the Tabernacle (ch. 24:1–9) | | |
| | I. A Case of Blasphemy (ch. 24:10–23) | | |
| | J. Sabbatical and Jubilee Years (ch. 25) | | |
| | K. Exhortation to Obey the Law: Blessing and Curse (ch. 26) | | |
| | L. Redemption of Votive C | Gifts (ch. 27) | |

The present study utilizes a printed copy, specifically JSul.IV.02A³⁰, which constitutes the first volume of Göz. 1841 and encompasses the five books of Moses. In this edition, the Book of Leviticus is contained within folios 93 r^o – 121 r^o, thus spanning 57 pages of the translation and comprising 27 chapters written in Hebrew script. It is noteworthy that one verse (Lev 26:38) is absent from this translation³¹. Consequently, on the 4th line of folio 120 r^o, verse 26:39 succeeds verse 26:37. The missing verse is exhibited below.

Table 9: Missing Verse in the Lev of the Göz. 1841

| Lev | Göz. 1841 | Eng. Bible (NAS 1977) |
|-------|--|--|
| 26:37 | Da sürünürlär har kiši ķardašï bilän tutki | They will therefore stumble over each other as if running |
| | aldından kiličnin da kuvuvči yoktir da | from the sword, although no one is pursuing; and you |
| | bolmaz sizgä turmaķ aldīna | will have no strength to stand up before your enemies. |
| | dušmanlariŋïznïŋ. | |
| 26:38 | - | But you will perish among the nations, and your |
| | | enemies' land will consume you. |
| 26:39 | Da ol kalyanlar sizdä čürirlar gunahlarï | So those of you who may be left will rot away because |
| | bilän yerlärindä dušmanlarïŋïznïŋ da | of their iniquity in the lands of your enemies; and also |
| | dayïn gunaḥlarï bilän atalarïnïŋ | because of the iniquities of their forefathers they will rot |
| | birgälärinä čürirlär. | away with them. |

2. The Linguistic Analysis of the Text

2.1. Phonology

In this section, a thorough examination of the phonological features within the text shall be conducted. It is important to note that the investigation of morphophonology, which pertains to the phonology of morphological elements, will be carried out concomitantly with the morphology section (refer to section 2.2.). Consequently, the current section will focus exclusively on the features of uninflected forms. As shall be demonstrated, the dataset encompasses a diverse array of phonological properties. In this context, the prevalence of these features will be delineated and juxtaposed with those observed in Crimean Tatar, Crimean

³⁰ The aforementioned copy is preserved in the private collection of Józef Sulimowicz. I express my profound gratitude to Anna Sulimowicz-Keruth, who graciously provided digital scans of the printed edition. Furthermore, my appreciation extends to Zsuzsanna Olach, who contributed digital photographs of another copy of Göz. 1841. Regrettably, limited information is available about this particular copy. I only referred to it on a few occasions when certain segments of the JSul.IV.02A were indecipherable.

³¹ It is noteworthy that this verse is also absent in the BSMS 288, whereas it does appear in the H 170: *Da tas bolursuz uluslarda da tavusur sizni yeri dušmanlarinizni* (CrKB I: 215).

Turkish, Ottoman Turkish, and the three dialects of Karaim. However, due to the constraints of accessible data, comparisons with Crimean Turkish will be based on only certain select features.

Finally, it is essential to acknowledge that, due to the orthographical challenge of vocalization (refer to section 2.1.1.1.2.), Crimean Karaim dictionaries were utilized in select instances to clarify specific vowel values.

2.1.1. Sound Inventory, Orthography and Transcription System

The investigation of phonemic variations in the data under analysis presents difficulties due to multiple factors, including the extinction of the Crimean Karaim language, the limited application of the Hebrew script in representing specific Turkic phonemic attributes, and the dearth of resources on spoken Crimean Karaim³². As a result, this research does not directly address phonetic concerns. Nevertheless, given that orthography occasionally pertains to certain phonemic variations, concise descriptions of these variations will be provided without an extensive examination. It can be posited that the phonemic inventory of Crimean Karaim demonstrates typical features of Turkic languages in Crimea, resulting in minimal deviation from languages such as Crimean Tatar, Crimean Turkish, and Ottoman Turkish. In contrast, it does not exhibit the unique phonemes of Western Karaim languages, which have evolved under the influence of a Slavonic linguistic milieu.

As previously mentioned, due to orthographic constraints, it is of paramount importance to elucidate the orthographic features of the text and their transcription within the context of our investigation. In the present printed edition, the traditional Hebrew script is utilized, written from right to left. The text is fully vocalized, except for some Hebrew incipits³³, and certain words appear within parentheses. Notably, the transcription system bears a close resemblance to that in CrKB I: XXIV–XXV. Since, in our text, Biblical Hebrew words were consistently written in their original forms, thereby displaying Biblical Hebrew orthographic features, taking into consideration Harvianien's work (2013: 453–457) was essential for further clarification.

³³ The Hebrew incipit is an unvocalized Hebrew word that precedes a Biblical verse, signifying the initial Biblical Hebrew word of the subsequent verse.

³² Prik's research (1976) represents the only study devoted to Crimean Karaim based on spoken material; however, the data was gathered in 1949 when only a limited number of speakers remained. Consequently, it may not accurately reflect a text written more than a century prior to the collection of this material.

2.1.1.1. Vowels

2.1.1.1.1. Vowel System

In Prik's (1976: 25) analysis of Crimean Karaim, eight vowel phonemes are identified: front unrounded /i/ and /ä/, front rounded /ü/ and /ö/, back unrounded /a/ and /ï/, and back rounded /o/ and /u/. These vowel phonemes are identical to that of Trakai Karaim, but differ from Halich Karaim, which possesses only six phonemes, as it lacks /ö/ and /ü/, which have been replaced by /ä/ and /i/, respectively (Musaev 1977: 9). In relation to Ottoman Turkish (Kissling 1960: 17) and Crimean Tatar (Kavitskaya 2010: 6), both languages display an identical set of eight phonemes, a characteristic also observed in our dataset.

As previously stated, the analysis of phonemic variations in the data under investigation presents difficulties owing to numerous factors. However, it is essential to explain a vowel phoneme variation that is apparent in the orthography. It is widely acknowledged that in Crimean Karaim Bible translations, the phoneme $\frac{\ddot{a}}{displays}$ variations, including \ddot{a}^{34} and a closed variant e^{35} . As illustrated in the following section (see 2.1.1.1.2.), in our text, the variations, \ddot{a} and e can be distinguished, with \ddot{a} being denoted through orthographic strategies akin to those for phoneme /a/, while being differentiated by the frontness and backness of the word. It is important to highlight that Doerfer (1959a: 273) also distinguishes ä from the closed e variant in Crimean Turkish and maintains that \(\bar{a}\) appears solely in non-initial syllables. A comparable phenomenon is also documented in Western Karaim, as the distribution of e and \ddot{a} as variants of /\(\bar{a}\)/ is determined by their position in Northwest Karaim, with \(\bar{a}\) primarily occurring in non-first syllables and e situated in first syllables (Berta and Csató 2022b: 320). In fact, Prik (1976: 26) also mentions the same system in spoken material of Crimean Karaim. Conversely, our analysis of the dataset reveals that, although the majority of the data adheres to the established rules, some fluctuations are observed where the vowel \(\bar{a}\) appears in initial syllables and e in non-initial syllables in Turkic words. It should be noted that we do not consider such examples as scribal errors and present them in their original forms. Notably, a similar fluctuation has also been mentioned by Jankowski (1997: 7; CrKB I: XXV). Consequently, it is worth emphasizing that the vowels \ddot{a} and e are phonemic variations of the phoneme /ä/ and can be used interchangeably within the same words (see 2.1.2.1.2.).

³⁴ It is important to note that this phoneme does not appear in Biblical Hebrew words.

³⁵ Regarding the transcription of Crimean Karaim vowels, Sulimowicz (1972: 42, 45) put forward a system that employed the letter \acute{a} to denote non-initial \ddot{a} sounds, which reflected a centralized front vowel. In contrast, our study employs a transcription system that is consistent with that utilized by Jankowski (1997) and CrKB I.

2.1.1.1.2. Orthography and Transcription of Vowel Sounds

Before presenting the transcription of vowels sounds in our study, it is essential to acknowledge the existence of two primary methods for denoting vowels in Hebrew script. The first approach involves employing a combination of consonants known as matres lectiones, such as he (7) aleph (x), waw (1) and yodh (1) along with vowel points (niqqud), such as patach (1), qamatz (1), tzere (), kubutz (), segol (), mobile shewa (); holam male (), interrupted patach (), interrupted gamatz (), interrupted segol (). The second method entails the sole usage of the aforementioned vowel points. It is essential to emphasize that the rationale behind the Hebrew script exhibiting this range of combinations is to indicate the position of vowels and various Biblical Hebrew vowel qualities, such as short, long, and interrupted (*chataf*). This comprehensive inventory is mirrored in Karaim, even though Karaim lacks the specific vowel values found in Biblical Hebrew. It is well known that in Karaim materials written in Hebrew script, Hebrew words are invariably written in their distinct forms, thus reflecting only orthographical characteristics rather than phonemic values. This is also the case for our text. However, similar to other Karaim materials written in Hebrew script, our text also presents alternative methods for denoting same vowel values in non-Hebrew words. It is crucial to understand that these different methods do not represent any phonemic variations; they are solely orthographical variations.

A significant observation to consider is that the Hebrew script does not allow for the distinction between particular vowel pairs, such as $\ddot{\imath}$ - $\dot{\imath}$, o- \ddot{o} , u- \ddot{u} , and a- \ddot{a} ³⁶. Comparable to the Arabic script, which has been utilized in a range of Turkic languages and persists in use among some Turkic languages today, the frontness-backness differentiation within a word is vital for identifying the back-front correspondences of these vowels. However, aside from consonant pairs like k-k (refer to section 2.1.1.2.1) that indicate whether the vowels in non-Hebrew words are articulated in the back or front³⁷, it is not feasible to ascertain whether the word contains back vowels or front vowels. In such cases, the mentioned vowel pairs can only be distinguished by context, and therefore the use of dictionaries is crucial, but it is not always the complete solution.

_

³⁶ In regard to the Biblical Hebrew words present in our text, we do not encounter this issue, as Biblical Hebrew does not contain \ddot{a} , \ddot{i} , \ddot{o} , and \ddot{u} .

³⁷ It is important to note that in some Arabic, Persian and Biblical Hebrew words, the variants k and k does not indicate the frontness or backness of the word unlike the Turkic words.

Table 10: Transcriptional System of Vowels

| Vowels | Letters |
|--------|---|
| a - | _(patach): Frequent; appears only in word-medial positions. |
| | (qamatz): Rare; appears only in word-medial positions, and usually in Biblical Hebrew words. |
| | رِ (chataf patach): Rare; appears only in word-medial positions, and only in Biblical Hebrew words. |
| | ¾ (aleph+patach): Frequent; appears only in word-initial positions. |
| | § (aleph+qamatz): Rare; appears only in word-initial positions, and usually in Biblical Hebrew words. |
| | ℵ ₋ (patach+aleph): Frequent; appears mostly in word-final positions and occasionally in word- |
| | medial positions. As will be mentioned (see 2.1.1.2.1.), only in Hebrew and Arabic words, and only |
| | in word-medial positions, aleph (x) is not a <i>mater lectionis</i> but denotes a Hebrew/Arabic glottal |
| | stop consonant, which is denoted by ['] in our transcription, but does not have a phonemic value in |
| | Karaim. Thus, such Arabic/Biblical Hebrew examples differ from this combination. |
| | እ ₋ (qamatz+aleph): Rare; appears only in word-final positions. |
| | ក_(patach+he): Frequent; appears only in word-final positions. |
| | ה,(qamatz+he): Rare; appears only in word-final positions. |
| ä | _(patach): Frequent; appears mostly in word-medial positions and rarely in word-final positions. |
| | ,(qamatz): Rare; appears only in word-medial positions. |
| | ¾ (aleph+patach): Rare; appears only in word-initial positions. |
| | <u>୪</u> (patach+aleph): Frequent; appears only in word-final positions. |
| | ה_(patach+he): Frequent; appears only in word-final positions. |
| e | (tzere): Frequent; appears mostly in word-medial positions and rarely in word-final positions. |
| | "(segol): Rare; appears in word-medial positions. |
| | (mobile shewa ³⁸): Rare; appears only in word-medial positions and always after the first letter of |
| | the word. Moreover, it is found exclusively in Biblical Hebrew words ³⁹ , with the exception of one |
| | Turkic word, e.g., [Lev 12:2] yedi. Note that, with the exception of a few instances, the Turkic word |
| | yedi was consistently written using mobile shewa. |
| | ⅓ (aleph+tzere): Frequent; appears only in word-initial positions. |
| | ¾ (aleph+segol): Rare; appears only in word-initial positions. |
| | |

_

 $^{^{38}}$ It should not be mistaken with the *shva nach* [], which appears frequently in our text and indicates that the following letter is a consonant.

³⁹ It should be noted that in Biblical Hebrew words, if mobile shewa [] appears before a guttural consonant such as *alef* [\aleph], *het* [π], *het* [π], and *ayin* [\aleph], its value is identical to the following vowel that appears after the guttural consonant (Harvianien 2013: 455).

| Vowels | Letters | | | | | |
|--------|--|--|--|--|--|--|
| | እ (tzere+aleph): Rare; appears only in word-final positions. | | | | | |
| | \aleph_{ν} (segol+aleph): Rare; appears only in word-final positions. | | | | | |
| | ה(tzere+he): Rare; appears only in word-final positions. | | | | | |
| | ה, (tzere+ segol): Rare; appears only in word final positions. | | | | | |
| | '(segol+yodh) Rare; appears in word-final positions. | | | | | |
| ï, i | אָי (aleph+hiriq+yodh): Frequent; only in word-initial positions. | | | | | |
| | ٠.(hiriq+yodh): Frequent; appears in word-medial and word-final positions. | | | | | |
| | (hiriq): Rare; appears only in word-medial positions. | | | | | |
| 0 | ነጻ (aleph+waw): Frequent; appears only in word-initial positions. | | | | | |
| | ำห (aleph+holam male) : Frequent; appears only in word-initial positions. | | | | | |
| | I (waw): Frequent; appears only in word-medial positions and word-final positions. | | | | | |
| | i (holam male): Frequent; appears only in word-medial positions and word-final positions | | | | | |
| | (holam haser): Rare; appears only in word-medial positions and only in Biblical Hebrew words. | | | | | |
| | (qamatz): Rare; only in word-medial positions, and exclusively in a few Biblical Hebrew words, | | | | | |
| | e.g., the word קרבָן is transcribed as <i>korban</i> and not as <i>karban</i> . In Modern Hebrew, this vowel point | | | | | |
| | is named as <i>qamatz qatan</i> , which does not differ in shape from the so-called <i>qamatz gadol</i> that represents the phoneme /a/. | | | | | |
| ö | א (aleph+waw): Frequent; appears only in word-initial positions. | | | | | |
| | אֹל (aleph+holam male) : Frequent; appears only in word-initial positions. | | | | | |
| | I (waw): Frequent; appears in word-medial and word-final positions. | | | | | |
| | i (waw with holam): Frequent; appears only in word-medial and word-final positions. | | | | | |
| u, ü | א (aleph+waw): Frequent; appears only in word-initial positions. | | | | | |
| | ৰ (aleph+shuruk): Frequent; appears only in word initial positions. | | | | | |
| | (aleph+kubutz): Rare; appears only in word-initial positions. | | | | | |
| | I (waw): Frequent; appears only in word-medial positions and word-final positions. | | | | | |
| | i (shuruk); Frequent; appears in word-medial and word-final positions. | | | | | |
| | (kubutz) = Rare; appears only in word-medial positions. | | | | | |
| | | | | | | |

There are certain orthographical features related to the notation of vowels in Hebrew script that warrant further clarification. Our text presents one such feature, known as *patach furtivum*. In specific Hebrew words, when a patach () – representing the vowel a (refer to Table 10) – appears after particular consonants such as cheth (π), he with mappiq (π), or ayin (ν), the vowel

a is pronounced before, rather than after, these consonants. It is noteworthy that the provided text contains only one instance of this phenomenon, occurring a total of 87 times. This word (for example, refer to Lev 1:5) is written as אַזְבָּה, which could potentially be misread as mizbeḥa 'altar'. However, due to the presence of the patach furtivum feature, the correct transcription should be mizbeaḥ.

2.1.1.2. Consonants

2.1.1.2.1. Consonant System

In our dataset, there are a total of 21 consonant phonemes: /b/, /3/, /c/, /d/, /f/, /g/, /h/, /h/

It should be emphasized that determining the precise phonetic values of the phonemes in our dataset is a challenging task⁴¹. However, clear phonemic variations can also be observed. For instance, the phoneme g demonstrates three distinct variants: a front variant, the voiced stop g; and two back variants, the voiced fricative g and the voiced stop g (Prik 1976: 29). Notably, the phonemic variant g is absent in Turkic lexemes, appearing only in certain loanwords (Prik 1976: 32). Following established orthographic conventions, this variant is represented in our transcription by g, which also denotes the front variant g. Crucially, the voiced fricative g does not manifest in Biblical Hebrew words located within a syllable consisting of back vowels, unlike in other words in our dataset. Similarly, the phoneme g also has two variants: a front stop g and a back stop g (Prik: 1976: 29). It is important to mention that in Arabic-Hebrew-Persian words, these variants can appear in reverse since these words are written based on their original orthographic form and does not follow Turkic characteristic.

-

⁴⁰ It should be noted that, given the fact that speakers of Trakai Karaim were also speaking a Slavonic language, the phonemes /ʒ/ and /c/ also existed in their sound inventory for non-Turkic words. As such, in the Trakai Karaim Bible translations, specific Biblical Hebrew or Slavonic words have these phonemes represented in transcription by scholars. However, Halich Karaim presents a different case, as there was a regular shift in its sound inventory from /ʒ/ to /ʒ/ and /č/ to /c/. For a comprehensive examination of the historical development of Western Karaim phonology, please refer to Németh's scholarly works (2014a; 2014b; 2015a).

⁴¹ It should be noted that Prik's description of Crimean Karaim consonants (1976: 29) may not be directly applicable to our dataset, as her description pertains to a different time period that includes Russian influences on the consonantal inventory not found in older sources.

It is also vital to recognize that certain orthographic characteristics of our text are inherent to the Hebrew script features for Biblical Hebrew and Arabic words, representing some consonant phonemes that lack equivalents in Turkic languages. Among these characteristics, the consonant aleph (x) signifies a glottal stop in certain Arabic/Hebrew words and lacks an equivalent in Karaim. As will be demonstrated, aleph (x) was also utilized to indicate vowels in specific combinations, given its role as a *mater lectionis* (see 2.1.1.1.2.). Nevertheless, aside from these functions, in medial positions within Biblical Hebrew and Arabic words in our text, aleph is represented by ['] in our orthography, which cannot be considered to possess phonemic value in Karaim and will therefore not be addressed in subsequent sections. Another notable aspect is the letter ayin (y), which was traditionally recognized as a voiced pharyngealized fricative but later evolved into a pharyngealized glottal stop in Hebrew, also lacking an equivalent in Karaim. However, in Arabic and Biblical Hebrew words within our translations, ayin (y) is consistently represented as ['] in every position of a word, which likewise does not possess phonemic value in our text and will therefore be excluded from subsequent section.

It is of paramount importance to recognize the orthographic discrepancy between the characters h and h, as identifying their exact phonemic distinctions presents a substantial challenge, an issue similarly underscored by Németh (2020: 29). Both characters are presumably representative of a voiceless fricative glottal. Interestingly, the transcriptional letter h cannot be employed in Turkic words within the analyzed text, unlike the letter h, which contrasts with Németh's observations in a Trakai Karaim Bible translation. Given the imperative to faithfully render Hebrew, Arabic, and Persian words in their (possible) authentic form utilizing the Hebrew script, it is plausible that h merely operates as an orthographic variant of h. A related phenomenon can be observed in Ottoman Turkish concerning the Arabic letters $hah(\tau)$, typically transcribed as h, and $hah(\tau)$, typically transcribed as h. It has been suggested that these characters did not represent distinct phonemes or phonemic variations in Ottoman Turkish (Kissling 1960: 4).

2.1.1.2.2. Orthography and Transcription of Consonant Sounds

Within the realm of Hebrew script, consonantal characters predominantly exhibit a single form consistently appearing in all word positions. However, certain circumstances necessitate the adoption of a distinct word-final form for specific consonants. It is crucial to highlight that, in contrast to the majority of Karaim Bible translations and prayer books examined in previous studies (e.g., Jankowski 1997; Sulimowicz 1972; Németh 2011, 2020; CrKB I), the current text

abstains from employing the so-called raphe () to differentiate the stops b, g, k, and p from their corresponding fricative counterparts v, y, h, and f. Rather, when particular letters, such as beth (2), gimel (3), kaf (5), and pe (5), lack a dagesh (), they signify both mentioned fricatives and their stop counterparts together. Conversely, in the presence of a dagesh with these letters, they exclusively denote the stops of these phonemes. Furthermore, this text lacks a specific combination to distinguish the fricative y from the stop g and the velar y.

Table 11: Transcriptional System of Consonants

| Consonants | Letters |
|------------|--|
| b | ☐ (beth): In all word positions. |
| | ☐ (beth with dagesh): In all word positions. |
| ž | 'λ (gimel with geresh): In all word positions. |
| č | ሄ (tsade): In word-initial and word-medial positions. |
| | γ (final tsade) In word-final positions. |
| d | 7 (dalet): In all word positions. |
| f | 5 (pe): In all word positions. |
| g, y | ۵ (gimel): In all word positions. |
| h | ก (he): In all word positions. |
| μ̈́ | ת (chet): Only in non-Turkic words (e.g., Biblical Hebrew, Arabic, Persian |
| | words), and in all positions. |
| ĥ | ⊃ (kaph): In word-initial and word-medial positions. |
| | ন (final kaph): In word-final positions. |
| k | ∃ (kaph with dagesh): In all word positions. |
| | ⊃ (kaph): In all word positions (Rare). |
| ķ | 구 (qoph): In all word positions. |
| 1 | ל (lamed): In all word positions. |
| m | ත (mim): In word-initial and word-medial positions. |
| | ☐ (final mim): In word final positions. |
| n | la (nun): In word-initial and word-medial position. |
| | ן (final nun): In word final position. |
| ŋ | λ (gimel): In word-medial and word-final positions. |
| p | 5 (pe with dagesh): In word-initial and word-medial positions. |
| | 5 (pe): In word-initial and word-medial positions (Rare). |
| | |

| Consonants | Letters |
|------------|--|
| | ሻ (final pe with dagesh): In word-final positions. |
| | ሻ (final pe): In word-final positions (Rare). |
| r | ¬ (resh): In all word positions. |
| S | D (samekh): In all word positions. |
| | $\dot{\overline{v}}$ (shin with sin dot): Only in some Biblical Hebrew words in all word |
| | positions. |
| š | ♥ (shin with shin dot): In all word positions. |
| | $\ensuremath{\mathfrak{V}}$ (shin): In all word positions. |
| t | ប (tet): In all word positions. |
| | A (tav with dagesh): In all word positions (Rare). |
| | π (tav): In all word positions. |
| V | າ (waw): In all word positions. |
| | 11 (double waw): In all word positions (Rare). |
| у | ' (yodh): In all word positions. |
| | " (double yodh): In all word positions (Rare). |
| Z | 7 (zayin): In all word positions. |

In our text, there are two notable orthographical tendencies regarding consonants. The first tendency is the avoidance of utilizing the waw (1) letter three times consecutively. As illustrated, waw can represent the vowels o, u, \ddot{o} , \ddot{u} , as well as the consonant v. Nonetheless, in specific words where there should be three characters for denoting vowel+consonant+vowel, only two instances of waw are used, as exemplified in [Lev 4:7] קוֹנשׁ kov(u)š. It is important to note that this phenomenon has also been observed in other Karaim texts (Németh 2011: 119–120). The second characteristic is the avoidance of using two kaph (כ) characters in succession in certain instances. Although this characteristic is not consistently observed, it is present in the majority of examples encountered, e.g., [Lev 15:33] אַרְכָּבֶה erkäk(k)e.

2.1.2. Phonological Variations and Adaptations

In this section, we will discuss the phonological variations and adaptations identified in our dataset, offering pertinent examples from the Crimean Karaim, Crimean Tatar, Ottoman Turkish, and Western Karaim languages, when applicable. For the majority of non-Turkic

lexemes, it may be more fitting to categorize the differences exclusively as adaptations, considering the disparities between the Turkic phonemic inventory and those of other languages. Furthermore, it is crucial to take into account the orthographical features of Hebrew and Arabic scripts when analyzing such lexical items.

2.1.2.1. Vowels

It is vital to stress that in the examination of phonological variations concerning vowels, Biblical Hebrew lexemes will not be investigated within this subsection's analysis. This is due to their primary representation of original Biblical Hebrew forms rather than providing accurate phonological values. Conversely, this characteristic is not regularly observed in Arabic-Persian lexemes, as the Hebrew script employed in our text markedly diverges from the classic Arabic script in terms of vowel representation. More precisely, the short vowels in Arabic and Persian words were not represented in classic Arabic orthography, unlike the Hebrew script that has been adapted in our text⁴². On the other hand, it becomes imperative to recognize that some attributes which will be presented pertaining to Arabic-Persian words may, in fact, reflect the complexities inherent in Arabic orthography, rather than phonological adaptations.

2.1.2.1.1. Vowel Assimilation

Concerning uninflected words⁴³, it is well-recognized that intersyllabic harmony, characterized by frontness versus backness, is a prevalent feature in most Turkic languages. Nonetheless, this rule encounters exceptions, notably in some polysyllabic native words and certain non-Turkic lexemes. This phonological principle is applicable to Ottoman Turkish and Karaim dialects, specifically within native words. While this harmony is generally attested, documented deviations do exist in the context of Crimean Turkish (Doerfer 1959a: 273) and Crimean Tatar (Berta and Csató 2022b: 322).

In the Lev of Göz. 1841, we also do not find any uninflected Turkic words that conspicuously lack intersyllabic front versus back harmony⁴⁴. Pertaining to non-Turkic words, in the majority

_

⁴² It is also important to note that the Arabic script features various diacritics in certain texts; however, the modified abjad script has been and continues to be more predominant.

⁴³ For further discussion on intrasyllabic vowel harmony in relation to inflected lexemes, refer to section 2.2.1.1.

⁴⁴ As previously mentioned (refer to section 2.1.1.1.2), the Hebrew script does not distinguish between the front-back counterparts of specific vowels, such as a- \ddot{a} , \ddot{i} -i, u- \ddot{u} , and o- \ddot{o} . The sole exception is the phoneme /k/, which demonstrates both back (k) and front (k) variations, represented by separate letters, and in Turkic-origin words, it reflects the frontness or backness of the word. Nevertheless, aside from this exception, there is no systematic method to determine the frontness or backness of a word. As a result, while some lexemes are unambiguous, others may be subject to debate. In such situations, Karaim dictionaries (CKED, KRPS) are consulted to establish the appropriate word.

of instances, their original forms, which lack fronting harmony, are maintained. Contrarily, in selected Arabic and Persian loanwords, occurrences arise where uninflected words, initially absent of front harmony, have adapted to incorporate fronting harmony by vowel assimilation. Predominantly in these instances, the assimilation is driven by the initial syllable. In several examples, it can be observed that, in addition to fronting, the degree of openness (see 2.1.2.1.2.) is also respected when the vowel in the initial syllable drives the assimilation of the vowel in the succeeding syllable.

Table 12: Vowel Assimilation in Arabic and Persian Loanwords 1

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|-------------------------------|-------------|--------------|-------------|-------------|----------|--------------|
| [hafte] هفته | ḥafta | ḥafta | afta | hafta | afta; | ḥafta; |
| 'week' <per></per> | [Lev 12:15] | (CKED: 174) | (KRUS: 78) | (ETD: 1120) | yafta | afta |
| (NS: 335) | | | | | (KRPS: | (KRPS: |
| | | | | | 85) | 607) |
| [<i>mīrā<u>s</u></i>] ميراث | meräs | meres | miras | miras | - | <i>meres</i> |
| 'heritage' <ar></ar> | [Lev 20:24] | (CKED: 239) | (KRUS: 358) | (ETD: 1070) | | (KRPS: |
| (NS: 596) | | | | | | 415) |
| [niyya(t)] نيّة | inyät | inyet; niyet | niet | niyet | niyet | niyeť |
| 'intention; aim' | [Lev 22:21] | (CKED: 197) | (KRUS: 393- | (ETD: 172) | (KRPS: | (KRPS: |
| <ar></ar> | | | 394) | | 419–420) | 420) |
| (NS: 648) | | | | | | |
| [<i>šāhid</i>] شاهد | šaḥad | šaḥad; šahat | šaat | šahid | - | - |
| 'witness' <ar></ar> | [Lev 5:1] | (CKED: 370) | (KRUS: 745) | (ETD: 754) | | |
| (NS: 828) | | | | | | |
| [šarīʿa(t)] شريعة | šaraʻat | šaraʻat | šeryat | šeriat | - | šarayat |
| 'law; religious | [Lev 18:4] | (CKED: 371) | (KRUS: 754) | (ETD: 758) | | (KRPS: |
| regulations' <ar></ar> | | | | | | 644) |
| (NS: 837) | | | | | | |

In contrast, one instance of assimilation exhibits a regressive pattern.

Table 13: Vowel Assimilation in Arabic and Persian Loanwords 2

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|-----------------------|-------------|---------------|-------------|------------|--------|--------|
| غريب [<i>yarīb</i>] | gerip | ķarib; gerip; | yarip | yarib | garip; | yarib |
| 'stranger' <ar></ar> | [Lev 17:15] | yarip; yerip | (KRUS: 146) | (ETD: 840) | yarib | (KRPS: |
| (NS: 305) | | (CKED: 294) | | | (KRPS: | 164) |
| | | | | | 158) | |

It should be noted that our dataset does not contain any exceptional examples that are not accounted for in Crimean Karaim dictionaries. Notably, in these instances, our text deviates from Ottoman Turkish forms while typically displaying Crimean Tatar characteristics. On the other hand, with one exception, it also exhibits similarities to Western Karaim. This could indicate that the adoption of such words into Crimean Karaim was not influenced by Ottoman Turkish. Nevertheless, in several instances, our text clearly exhibits the influence of Ottoman Turkish, as reflected in the use of Arabic or Persian words that originally featured fronting harmony but dissolved in Ottoman Turkish and are similarly represented in our text. Notably, this observation aligns with findings in the CKED dictionary, which suggests that the examples listed in the table below were adapted from Ottoman Turkish.

Table 14: Vowel Assimilation in Arabic and Persian Loanwords 3

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|--------------------------|--------------|-------------|-------------|------------|-------|-------|
| [ātaš] 'fire' | ateš | ateš; öteš | ateš | ateš | - | - |
| <per></per> | [Lev 18: 21] | (CKED: 55) | (KRUS: 75) | (ETD: 434) | | |
| (NS: 78–79) | | | | | | |
| [awlād] <ar> أو لاد</ar> | evlad | evlad | evlat | evlad | - | - |
| 'children' | [Lev 11:2] | (CKED: 158) | (KRUS: 767) | (ETD: 508) | | |
| (NS: 265) | | | | | | |
| بياض [bayāḍ] | beyaz | beyaz | beyaz | beyaz | - | - |
| 'white' <ar></ar> | [Lev 13:24] | (CKED: 81) | (KRUS: 109) | (ETD: 550) | | |
| (NS: 116) | | | | | | |

2.1.2.1.2. Degree of Openness

In the dataset under examination, various vowel transformations associated with the degree of openness are observed. These alternations manifest in both systematic and sporadic manners.

A-) $/\ddot{a}/ \sim /i/$: One such transformation is the fluctuation between i and \ddot{a} , which typically transpires in Turkic lexemes that previously contained a closed e. This fluctuation has its roots in Old Turkic (Róna-Tas 2022b: 122) and has been documented in various Turkic languages, including Crimean Turkish (Doerfer 1959a: 274), Crimean Tatar (Doerfer 1959b: 376), Ottoman Turkish (Kerslake 2022: 177–178), Halich Karaim (Musaev 1964: 57) and Trakai Karaim (Gülsevin 2016: 41). It should be noted that in our data, both variations are exhibited exclusively in specific lexical items.

Table 15: Fluctuations Between the Vowels /ä/ ~ /i/ in Turkic Words

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|----------------------|----------------|-------------|----------------|------------|--------|---------|
| ė:d- | it-/id- vs et- | et; ed- | et-; it-; id- | et-; it- | et- | eť- |
| 'to do' <tur></tur> | [Lev | (CKED: 157) | (KRUS: 791– | (ETD: 434; | (KRPS: | (KRPS: |
| (EDPT: 36-37) | 4:20/11:43; | | 792) | LET: 255- | 671) | 672) |
| | 20:25] | | | 256) | | |
| te:g- | tiy- vs deg- | tiy-; deg- | tiy-; deg- | deg- | tiy- | tiy- |
| 'to touch' | [Lev 5:2; | (CKED: 405) | (KRPS: 585) | (ETD: 690) | (KRPS: | (KRPS: |
| <tur></tur> | Lev 11:8] | | | | 526) | 526) |
| (EDPT: 475) | | | | | | |
| yėgirmi | yegirmi; | yigirmi; | yigirmi; | yigirmi | egirmi | egirmi; |
| 'twenty' <tur></tur> | yigirmi | žigirmi; | ǯïγïrmï; yirmi | (ETD: | (KRPS: | igirme |
| (EDPT: 915) | [Lev 27:3; | yigirme | (KRUS: 213) | 1139) | 654) | (KRPS: |
| | 27:5] | (CKED: 466) | | | | 654) |

In these examples, with the exception of te:g-, all instances exhibit etymological closed e. As observed, the languages presented display different characteristics for each lexeme, making it difficult to assert a systematic change. Nonetheless, it is important to emphasize that in certain instances, our data does not exhibit fluctuations and instead displays the phonemic variation e for the etymological closed e.

Table 16: Degree of Openness in Turkic Words

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|------------------------|-------------|-------------|-------------|------------|-------------|--------|
| ėkki 'two' <tur></tur> | eki | eki | eki | iki | eki; iki | eki |
| (EDPT:100-101) | [Lev 3:4] | (CKED: 148) | (KRUS: 773) | (ETD: 518) | (KRPS: 656) | (KRPS: |
| | | | | | | 656) |
| ėn- 'to go down' | en- | en- | en- | in- | en- | eń- |
| <tur></tur> | [Lev 9:22] | (CKED: 151) | (KRUS: 783) | (LET: 228) | (KRPS: 662) | (KRPS: |
| (EDPT: 168) | | | | | | 664) |
| ėšid- 'to hear' | ešit- | ešit- | ešit- | išit- | esit- | ešit- |
| <tur></tur> | [Lev 26:15] | (CKED: 157) | (KRUS: 795) | (ETD: 516) | (KRPS: 669) | (KRPS: |
| (EDPT: 257- | | | | | | 673) |
| 258) | | | | | | |

Moreover, three instances exemplify the variability in transmission, demonstrating the transition from high to low vowels in the initial syllables of non-Turkic words.

Table 17: Degree of Openness in Non-Turkic Words

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|-------------------------------|-------------|--------------|-------------|-------------|--------|--------------|
| [<i>mīrā<u>s</u></i>] ميراث | meräs | meres | miras | miras | - | meres |
| 'heritage' <ar></ar> | [Lev 20:24] | (CKED: 239) | (KRUS: 358) | (ETD: 1070) | | (KRPS: 415) |
| (NS: 596) | | | | | | |
| [vīrān] ويران | veran | veran; viran | viran | viran | veren | verań (KRPS: |
| 'ruined' <per></per> | [Lev 26:22] | (CKED: 437) | (KRUS: 135) | (ETD: 1114) | (KRPS: | 157) |
| (NS: 942) | | | | | 158) | |
| [mi <u>s</u> ķāl] مثقال | misķal vs | - | - | miskal | mitkal | mitkal |
| <ar> 'shekel'</ar> | mesķal | | | (LET: 680) | (KRPS: | (KRPS: 408) |
| (NS: 597) | [Lev 27:3; | | | | 408) | |
| | 27:4] | | | | | |

As observed, the rationale behind this transformation cannot simply be attributed to orthographical difficulties, as the vowels in the first syllables of the examples ويران $[v\bar{i}r\bar{a}n]$ and $[m\bar{i}r\bar{a}\underline{s}]$ should have been represented by the vowel i in the aforementioned Turkic languages. It is noteworthy that the characterization in these examples displays a common Karaim attribute, deviating from that of Crimean Tatar and Ottoman Turkish.

Finally, we encounter a single instance involving a Turkic lexeme where the vowel i has transitioned into the vowel \ddot{a} .

Table 18: Degree of Openness in Turkic Words 2

| Example | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|--------------|---------------|-------------|--------------|------------|-------------|--------|
| terin 'deep' | terän/derän | teren | deren; teren | derin | teren | ťeŕäń |
| <tur></tur> | [Lev 13:3/25] | (CKED: 396) | (KRUS: 159) | (ETD: 687) | (KRPS: 567) | (KRPS: |
| (EDPT: 551) | | | | | | 522) |

Ottoman Turkish once again demonstrates a distinction from the other examples under consideration.

B-) /a/ ~ /i/: As Musaev (1964: 57) states, the alternation between /a/ and /i/ predominantly takes place in Halich Karaim, with a particular focus on suffixes. Furthermore, a few instances can be observed in Trakai Karaim as well. Nonetheless, the juxtaposition of /a/ and /i/ is found in only a single non-Turkic example within our dataset and therefore it cannot be attributed to systematic characteristics.

Table 19: Transition between /a/ ~ /i/

| Example | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|------------------------|--------------------|-------|-------------|-------------|-------|-------|
| [makrūh] مکروہ | mïķruḥa vs mïķruḥ | - | mekruḫ | mekruh | - | - |
| 'abominable' <ar></ar> | [Lev 11:23; 11:11] | | (KRUS: 345) | (ETD: 1045) | | |
| (NS: 574) | | | | | | |

C-) $\mathbf{e} \sim \ddot{\mathbf{a}}$: As previously noted (see 2.1.1.1.1.), a certain opposition exists between e and \ddot{a} in various Turkic languages, encompassing Crimean Karaim (Prik 1976: 26) and Trakai Karaim (Berta and Csató 2022b: 320). In these languages, e consistently materializes in the initial syllables, while \ddot{a} appears in the subsequent syllables. Nevertheless, as corroborated by Jankowski (1997: 7; CrKB I: XXV) in Crimean Karaim Bible translations, variations between e and \ddot{a} can also be identified. Although the aforementioned pattern is predominantly observed in our text as well, certain examples exhibit fluctuations, with the vowel \ddot{a} manifesting in the initial syllables and e in the following syllables. It is worth noting that the fluctuation between e and \ddot{a} in the first syllables is also found in Crimean Turkish (Doerfer 1959a: 274).

Table 20: Fluctuations Between the Vowels $e \sim \ddot{a}$

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|-------------------|----------------|-------------|-------------|------------|-------------|--------|
| et 'flesh; | et vs ät | et | et | et | et | et |
| meat' <tur></tur> | [Lev 13:14; | (CKED: 157) | (KRUS: 790- | (ETD: 433) | (KRPS: 671) | (KRPS: |
| (EDPT: 33) | 10:19] | | 791) | | | 671) |
| temel | temel vs temäl | temel | temel | temel | - | - |
| 'foundation; | [Lev 25:47; | (CKED: 393) | (KRUS: 568) | (ETD: 602) | | |
| base' <gre></gre> | 4:7] | | | | | |
| (NS: 875) | | | | | | |

D-) $/o/ \sim /u/$; $/\ddot{o}/ \sim /\ddot{u}/$: In the given dataset, several instances reveal that there exists fluctuation between the vowels $o \sim u$ and $\ddot{o} \sim \ddot{u}$, in accordance with fronting harmony. These alternations have been documented in another Crimean Karaim Bible translation (Jankowski 1997:9) and identified in Crimean Tatar (Jankowski 2010: 79–80) as well as Trakai Karaim (Musaev 1964: 57).

Table 21: Fluctuations Between the Vowels $O \sim U$

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|--------------------------|--------------|---------------|--------------|-------------|----------|-----------|
| bol- 'to be' <tur></tur> | ol- vs ul- | bol-; ol- | bol-; ol- | ol- | bol- | bol- |
| (EDPT: 331-332) | [Lev 25:54 | (CKED: 89) | (KRUS: 119) | (ETD: 508) | (KRPS: | (KRPS: |
| | vs 25:49] | | | | 128) | 128) |
| buz- | boz- vs buz- | buz- | boz-; buz-; | boz- | buz- | buz- |
| 'to destroy' <tur></tur> | [Lev 11:35 | (CKED: 98) | bïz- | (ETD: 546) | (KRPS: | (KRPS: |
| (EDPT: 389-390) | vs 26:15] | | (KRUS: 117) | | 136) | 136) |
| kögürčgün | gögürčün vs | kögirčin; | gogeržin; | gügeržin | tigircin | kuģurčuń; |
| 'pigeon; dove' | gügürčin | kögürčün; | kogeržïn; | (ETD: 944) | (KRPS: | koģurčuń |
| <tur> (EDPT:</tur> | [Lev 15:14; | kögirčin | kogeržin; | | 525) | (KRPS: |
| 713) | 12:16] | (CKED: 216) | kogerčïn; | | | 396) |
| | | | kögeržin | | | |
| | | | (KRUS: 141) | | | |
| [murād] مراد | murad vs | mïrad; murad; | murat; mïrat | murad | - | - |
| 'wish; desire' | morad | murat | (KRUS: 367- | (ETD: 1002) | | |
| <ar></ar> | [Lev 22:29 | (CKED: 241) | 368) | | | |
| (NS: 611) | vs 19:5] | | | | | |
| öl- 'to die' <tur></tur> | öl- vs ül- | öl- | öl- | ül- | el- | öl- |
| (EDPT: 125-126) | [Lev 10:2 vs | (CKED: 267) | (KRUS: 413) | (ETD: 508) | (KRPS: | (KRPS: |
| | 11:31] | | | | 658) | 440) |
| uluš>ulus | ulus vs olus | ulus | - | - | ulus | ulus |
| 'country; nation' | [Lev 9:7; | (CKED: 425- | | | (KRPS: | (KRPS: |
| <tur></tur> | 7:20] | 426) | | | 578) | 578) |
| (EDPT: 152) | | | | | | |

As observed in our dataset, these fluctuations manifest in both Turkic and non-Turkic lexemes. There are instances in which the etymological low rounded vowels o and \ddot{o} transform into high rounded vowels u and \ddot{u} , as well as the reverse.

E-) $/a/\sim/\ddot{a}/:$ In our text, both a and \ddot{a} are represented by the same strategies (refer to section 2.1.1.1.2); nevertheless, the frontness or backness of a word can be determined due to the presence of suffixes containing the phoneme /k/, as the phonemic variants k and k are denoted by different letters (refer to section 2.1.1.2.1). Jankowski (CrKB I: XXVI) reports that alterations involving a and \ddot{a} are also evidenced in a Crimean Bible translation, potentially attributable to the preceding phoneme /y/. Conversely, the vowel-consonant sequence ay

undergoes a transformation to *ey* in Trakai Karaim, with these changes taking place prior to the phoneme /y/ (Musaev 1964: 55–56).

Table 22: Fluctuations Between the Vowels /a/ ~ /ä/ in Turkic Words

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|-----------------------|------------------|--------------|--------------|----------|--------|--------------|
| sač- 'to scatter; to | sač- vs säč- | sač-; seč- | sač- | sač- | cac- | čač- |
| sprinkle' <tur></tur> | [Lev 1:5; 26:5] | (CKED: 331) | (KRUS: | (ETD: | (KRPS: | (KRPS: 625) |
| (EDPT: 794) | | | 484) | 768) | 613) | |
| yaraštur- 'to set | yaraštïrmaķ vs | yaraštïr- | yaraštīr- | yaraš- | - | yaraštïr- |
| in order; make | yäräštirmäk | 'to fit; to | 'to fit; to | 'to | | 'to fit; to |
| ready' <tur></tur> | [Lev 24:7; 24:6] | arrange' | arrange' | beseem; | | arrange' |
| (EDPT: 973) | 'arrangement' | (CKED: 449) | (KRUS: | to suit' | | (KRPS: 232) |
| | | | 649) | (ETD: | | |
| | | | | 1135) | | |
| yaš 'fresh, moist; | yaš vs yäš | yaš | yaš; žaš | yaš | - | yaš |
| a year of one's | 'years; age' | 'years; age' | 'years; age' | 'years; | | 'years; age' |
| life' <tur></tur> | [Lev 27:7; 27:6] | (CKED: 454) | (KRUS: | age' | | (KRPS: 242) |
| (EDPT: 975–976) | | | 831) | (ETD: | | |
| | | | | 1128) | | |

Aside from the examples $sa\check{c}$ - 'to sprinkle' vs $se\check{c}$ - 'id', the remaining examples showing the a > \ddot{a} change are not present in Crimean Karaim dictionaries and the languages listed in the table above. It is also worth noting that these examples in our data might be attributed to scribal errors.

Finally, we also identify instances of non-Turkic words in our dataset where the phoneme /a/ has been adapted as /ä/. It is noteworthy that when examining the examples listed in the table below, the phonemic value of such adapted vowels in our text is not reflected in the original Arabic forms, and thus this adaptation might be dependent on orthographical difficulties. However, it should be noted that such alterations do not occur in Western Karaim.

Table 23: Transformation of /a/ into /ä/ in non-Turkic Words

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|------------------------|-----------|-------------|-------------|------------|-------------|--------|
| [pašmān] پشمان | pešman | pešman; | pešman | pišman | pasman | fašman |
| 'repentat' <per></per> | [Lev 5:5] | fešman | (KRUS: 539) | (ETD: 560) | (KRPS: 593) | (KRPS: |
| (NS: 706) | | (CKED: 277) | | | | 593) |

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|----------------------|-----------|-------------|-------------|------------|-------------|---------|
| [tarāzū] ترازو | täräzi vs | terezi | teraze | terazi | tarazuv | tarazu; |
| 'scales' <per></per> | teräzi | (CKED: 396) | (KRUS: 574) | (ETD: 579) | (KRPS: 514) | tarazuv |
| (NS: 879–880) | [Lev | | | | | (KRPS: |
| | 19:35; | | | | | 514) |
| | 19:36 | | | | | |

2.1.2.1.3. Degree of Roundness

Our text exhibits specific modifications pertaining to vowel roundness. The majority of these instances occur quite sporadically. As demonstrated in these examples, the degree of roundness varies bidirectionally, encompassing both the rounding of unrounded vowels and the unrounding of rounded vowels.

Table 24: Degree of Roundness in Non-Turkic Words

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|-------------------|--------------|----------------|---------------------|--------------|--------------|--------------|
| [ḫamīr] خمير | <i>ḥamur</i> | <i>ḥamur</i> | ḥamur; ḥamïr | <i>ḥamur</i> | <i>ḥamur</i> | <i>ḥamur</i> |
| 'leaven; dough' | [Lev 2:11] | (CKED: 175) | amur; <u>ķ</u> amir | (ETD: 675) | (KRPS: | (KRPS: |
| <ar></ar> | | | (KRUS: 672) | | 599-600) | 599-600) |
| (NS: 340) | | | | | | |
| [muṭlaḳ] مطلق | mïtlaķ | mïtlak; mutlaķ | mïtlak; mutlaķ | mutlaķ | - | - |
| 'absolute; surely | [Lev 2:11] | (CKED: 242) | (KRUS: 371) | (ETD: 1027) | | |
| <ar></ar> | | | | | | |
| (NS: 614) | | | | | | |

However, a notable pattern emerges where the consonant v, when present in certain contexts, influences mostly the subsequent unrounded vowel, causing it to shift to a rounded vowel. Note that this characteristic is also notable in some examples found in Crimean Tatar (Jankowski 2010: 83) and Western Karaim (Berta and Csató 2022b: 321).

Table 25: Influence of the Consonant phoneme /v/ on Unrounded Vowels

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|--------------------|------------|------------|--------------|-------------|------------|--------|
| ayïl 'an | avul | avul; avïl | avul; avïl | ayïl | avul | avul |
| enclosure for | [Lev 4:12] | (CKED: 58) | 'settlement; | 'a fold, or | 'camp' | 'camp' |
| livestock; cattle- | 'camp' | 'camp' | village' | pound for | (KRPS: 42) | (KRPS: |
| pen; sheep-fold; | | | (KRUS: 24) | sheep' | | 42) |

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|----------------------|----------------|---------------|----------------|----------------|----------------|-------|
| a settlement of | | | | (ETD: 472) | | |
| group of tents' | | | | | | |
| <tur> (EDPT:</tur> | | | | | | |
| 83) | | | | | | |
| [dawr] دور | dävür | devir | devir | devir | - | - |
| 'return; cycle; | [Lev 3:17] | (CKED: | (KRUS: 153) | (ETD: 694) | | |
| time; era' <ar></ar> | 'generation' | 137) | | | | |
| (NS: 205) | | | | | | |
| tavïšyan 'hare' | tavušan | davšan; | tavšan | tavšan | davsan; | - |
| <tur> (EDPT:</tur> | [Lev 11:6] | tavšan; | 'hare; rabbit' | (ETD: 796) | dafsan | |
| 447) | 'hare; rabbit' | tafšan 'hare; | (KRUS: 537) | 'hare; rabbit' | (KRPS: 168) | |
| | | rabbit' | | | 'hare; rabbit' | |
| | | (CKED: | | | | |
| | | 133) | | | | |

It is noteworthy that our dataset illustrates systematic transitions from the fricative Old Turkic consonant y to v, as is also evident in one of the examples shown above. However, in certain instances, we observe a vowel-consonant sequence shift ([iy]; $[ay] > [uv] \sim [ov]$:), where the influence of rounding tends to be solely regressive (refer to section 2.1.2.3.1).

2.1.2.1.4. Addition of a Vowel

Within the Lev of Göz. 1841, the incorporation of additional vowel sounds is predominantly observed in non-Turkic lexemes. Notably, this linguistic phenomenon manifests in three specific word positions.

Table 26: Addition of a Vowel in Word-Initial Positions

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|---------------------|--------------------|----------------|--------|--------|--------|--------|
| [murdār] مردار | umundar vs mundar | mïndar; mundar | murdar | murdar | murdar | murdar |
| 'unclean, impure, | [Lev 11:25; 5:2] | (CKED: 241) | (KRUS: | (ETD: | (KRPS: | (KRPS: |
| dirty' <per></per> | | | 368) | 1004) | 411) | 411) |
| (NS: 611–612) | | | | | | |
| [rang] 'colour' رنگ | irenk vs renk | renk; reng | renk | renk | reng | - |
| <per></per> | [Lev 14:52; 13:55] | (CKED: 328) | (KRUS: | (ETD: | (KRPS: | |
| (NS: 737–738) | | | 458) | 716) | 453) | |

As illustrated in the table above, the addition of vowels in word-initial positions is observed in some specific lexemes within our dataset. These examples coexist with their counterparts that do not exhibit the addition of vowels. Notably, this phenomenon is absent in the other languages demonstrated above for these lexemes, which have not been listed in Crimean Karaim dictionaries either.

Table 27: Addition of a Vowel in Word-Medial Positions

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|--|-------------------|----------------------|--------------|-----------|---------|--------|
| آبدان [$ar{a}bdar{a}n]$ + $l\ddot{\imath}k$ | avadanlïķ | - | - | avadanlïķ | - | - |
| <per> + <tur></tur></per> | [Lev 11:32] | | | (ETD: | | |
| 'tackle' | | | | 438) | | |
| (NS: 80) | | | | | | |
| عيب [ʾayb] 'fault; | ауїр | ауїв; ауїр | ауїр | ауїр | ауїр | ауїр; |
| error; shame' <ar></ar> | [Lev 15:3] | (CKED: 60) | (KRUS: 42) | (ETD: | (KRPS: | ayip |
| (NS: 84) | | | | 835) | 56) | (KRPS: |
| | | | | | | 56) |
| (yayb] <ar [yayb]<="" td="" غَیْب=""><td>ķayïp</td><td><u>к</u>аур; ұауір;</td><td>үауїр</td><td>ķayïp</td><td>-</td><td>-</td></ar> | ķayïp | <u>к</u> аур; ұауір; | үауїр | ķayïp | - | - |
| 'lost' (NS: 456) | [Lev 23:30] | yayb (CKED: | (KRUS: | (ÖTS 3: | | |
| | | 309; KRPS: | 145) | 2494) | | |
| | | 163) | | | | |
| tavišyan 'hare' | tavušan | davšan; | | tavšan | davsan; | - |
| <tur> (EDPT: 447)</tur> | [Lev 11:6] | tavšan; tafšan | | (ETD: | dafsan | |
| | 'hare; rabbit' | 'hare; rabbit' | | 796) | (KRPS: | |
| | | (CKED: 133) | | 'hare; | 168) | |
| | | | | rabbit' | 'hare; | |
| | | | | | rabbit' | |
| (waķt] <ar td="" وقت<=""><td>vaķit vs vaḫt</td><td>vaḫt; vahït;</td><td>vakit; vaķīt</td><td>vakit</td><td>vaḫt</td><td>vaḥt</td></ar> | vaķit vs vaḫt | vaḫt; vahït; | vakit; vaķīt | vakit | vaḫt | vaḥt |
| 'time' (NS: 932) | [Lev 14:56; 16:2] | vaķīt | (KRUS: | (ETD: | (KRPS: | (KRPS: |
| | | (CKED: 436) | 131) | 1112) | 156) | 156) |

The occurrence of additional vowels in the medial position of certain non-Turkic lexemes is a well-documented phenomenon in numerous Turkic languages, as illustrated by the languages detailed in the table above. This phenomenon can be ascribed to Turkic phonotactic constraints, which generally⁴⁵ cause consonant clusters to be separated in terminal positions.

_

⁴⁵ However, in Turkic languages, terminal clusters consisting of a single nasal, liquid, or sibilant are frequently observed (Johanson 2022a: 27).

Table 28: Addition of a Vowel in Word-Final Positions

| Example | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|------------------------|--------------------|-------|-------------|-------------|-------|-------|
| [makrūh] مکروہ | mïķruḥa vs mïķruḥ | - | mekruḫ | mekruh | - | - |
| 'abominable' <ar></ar> | [Lev 11:23; 11:11] | | (KRUS: 345) | (ETD: 1045) | | |
| (NS: 574) | | | | | | |

The final instance showcases a singular example where the additional vowel appears in a word-final position. This case may be attributable to an orthographical peculiarity, in which the word-final haa (*) in the Arabic script also represents e or a in Ottoman Turkish. Given that the word was not documented in Karaim dialects but in Ottoman Turkish and Crimean Tatar, it is possible that it was borrowed via Ottoman Turkish.

2.1.2.2. Consonants

2.1.2.2.1. Ottoman Turkish and Kipchak Consonantal Alternations

It is broadly acknowledged that numerous phonological distinctions exist between Oghuzic and Kipchak languages. The dataset at our disposal exhibits these contrasts, predominantly in the form of consonantal alternations. In light of the minor disparities between Ottoman Turkish and other Oghuzic languages⁴⁶, as well as the pervasive Ottoman Turkish influence throughout Crimea (refer to section 1.2.3.1.), we have opted to describe these features as 'Ottoman Turkish characteristics', even though the majority of them can also be classified as Oghuzic characteristics. As previously mentioned, the extensive Ottoman Turkish influence on the Crimean Karaim language serves to differentiate the varieties of Crimean Karaim (refer to section 1.2.3.2.). Consequently, the distribution of these linguistic features is essential for characterizing the language employed within the text.

Finally, It is worth noting that, given the absence of sections discussing detailed consonantal characteristics or alternations in Prik's description (1976) of Crimean Karaim, and Çulha's description (2019) being based solely on Crimean Karaim mejumas, selected portions of published Crimean Karaim Bible translations are utilized in the following section to illustrate the characterizations of Crimean Karaim with respect to these variations.

⁴⁶ For example, the voicing of initial *k*- is evident in languages such as Azeri and Turkmen, illustrated by the word *yara* 'black' (Johanson 2022b: 99). However, this characteristic is not observed in Ottoman Turkish (except for its eastern and southeastern dialects), e.g., *kara* 'black' (TLO II: 3618), and this absence of voicing is similarly reflected in our database, e.g., [Lev 13:31] *kara* 'black'.

A-) k-> g-: In Ottoman Turkish and other Oghuzic languages, the voicing of the initial unvoiced plosive k- in Turkic lexemes containing front vowels can be mostly observed in various instances (Johanson 2022b: 100). In contrast, the preservation of k- is a hallmark of such lexemes in the majority of Kipchak languages, encompassing Western Karaim (Çulha 2019: 68). With respect to Crimean Karaim Bible translations, the Kipchak characteristic is evident, e.g., $k\ddot{o}t\ddot{a}r$ - 'to lift', $k\ddot{o}k$ 'sky; heaven' (Jankowski 1997: 28–52), kel- 'to come', kelin 'bride; daughter-in-law' (Németh 2016: 178), kibik 'like; as', and $k\ddot{u}n$ 'day' (CrKB I: 166–217). Notably, the Ottoman Turkish feature pertaining to this trait has been observed in the Eastern dialect of Crimean Tatar (Kavitskaya 2010: 19) and Crimean Turkish (Doerfer 1959a: 274), concomitant with the Kipchak characteristic.

It is noteworthy that our dataset encompasses 26 unique instances of the aforementioned Ottoman Turkish feature, cumulatively appearing 387 times.

Table 29: Voicing of the Initial k-

| Lev | E | xamples | Occurrence |
|------------------------------------|-------------|--------------------------|------------|
| 25, 26, 27 | geč- | 'to pass' | 4 |
| 6, 8 | gečä | 'night' | 2 |
| 25 | gečin- | 'to live on' | 2 |
| 9, 10, 11, 12, 13, 14, 15, 16, 19, | gel- | 'to come' | 23 |
| 21, 23, 25 | | | |
| 11 | gelinčik | 'weasel' | 1 |
| 1, 9, 15, 16, 19, 23 | gendi | '-sef; the same' | 14 |
| 2, 4, 5, 6, 7, 10, 11, 12, 13, 14, | getir- | 'to bring' | 38 |
| 15, 17, 18, 20, 23, 24, 26 | | | |
| 11 | gevše-; | | 5+1+2 |
| | gevšemek ge | tir-/gevšemäk getir- | |
| | | 'to chew the cud' | |
| 11, 18 | gez- | 'to wander' | 8 |
| 4, 5, 6, 7, 9, 10, 12, 13, 14, 15, | gibi | 'like; as' | 32 |
| 18, 19, 20, 22, 25, 26, 27 | | | |
| 10, 16 | gir- | 'to enter' | 6 |
| 4, 6 ,10 11, 16 | girgiz- | 'to let in; to bring in' | 8 |
| 6, 8, 16, 21 | giy- | 'to wear' | 11 |
| 7, 18, 20 | gizli | 'hidden' | 4 |
| 26 | gök | 'sky, heaven' | 1 |
| 8, 10, 16 | gölmäk | 'shirt' | 4 |

| Lev |] | Examples | Occurrence |
|---------------------------------------|-------------|-----------------------|------------|
| 19, 26 | göŋül | 'heart' | 3 |
| 5, 9, 13, 14, 20, 25, 27 | gör- | 'to see' | 55 |
| 25, 27 | görä | 'according to; as to' | 7 |
| 5, 26 | gövdä | 'body' | 5 |
| 4, 10, 11, 13, 14, 20, 21, 24, 26 | göz | 'eye' | 12 |
| 5, 12, 15 | gügürčin/gö | igürčün 'pigeon' | 1+3 |
| 5, 25, 27 | gümüš | 'silver' | 9 |
| 1, 5, 6, 7, 8, 9, 10, 12, 13, 14, 15, | gün | 'day' | 111 |
| 16, 19, 22, 23, 24, 25, 26 | | | |
| 11 | güneš | 'sun' | 1 |

In the given dataset, the highest incidences of the specific Ottoman Turkish feature are found in Chapter 13 (63 words), Chapter 23 (43 words), and Chapter 14 (28 words). Conversely, the lowest frequencies of this attribute are observed in Chapter 2 (1 word), Chapter 1 (2 words), and Chapter 3 (2 words). Contrasting with this prominent Ottoman Turkish characteristic, the text includes a mere nine distinct lexical items, which appear 35 times (8%-92%) across 18 separate chapters, and retain the initial k-.

Table 30: *Preservation of the Initial k-*

| Lev | v Examples | | |
|---------------------|-------------------|---------------------------|-----|
| 14 | kel- | 'to come' | 2 |
| 18, 20 | kelin | 'bride; daughter in law' | 2 |
| 13 | ket- | 'to go' | 1 |
| 1, 3, 4, 7 | ketär- | 'to take away; to remove' | 11 |
| 2, 4, 5, 14, 17, 19 | ketir- | 'to bring' | 7 |
| 6 | kiy- | 'to wear' | 1 |
| 1, 14, 15 | kögürčin/kögürčün | 'pigeon' | 3+1 |
| 5, 7, 9, 10, 15, 16 | kötär- | 'to lift' | 6 |
| 27 | kümüš | 'silver' | 1 |

It is worth noting that among the examples, only five lexemes appear alongside their doublets in the text.

Table 31: Ottoman Turkish and Kipchak Doublets 1

| Glosses | Lev | Kip. Form | Occurrences | Lev | Ot.Tur Form |
|------------|------------|-------------------|-------------|------------------------|--------------------|
| 'to come' | 14 | kel- | 2 vs. 23 | 9, 11, 12, 13, 14, 15, | gel |
| | | | | 16, 19, 21, 23, 25 | |
| 'to bring' | 2, 4, 5, | ketir- | 8 vs. 38 | 2, 4, 5, 6, 7, 10, 11, | getir- |
| | 14, 17, 19 | | | 12, 13, 14, 15, 17, | |
| | | | | 18, 20, 23, 24, 26 | |
| 'to wear' | 6 | kiy- | 1 vs. 11 | 6, 8, 16, 21 | giy- |
| 'pigeon' | 1, 12, 14, | kögürčin/kögürčün | 4+1 vs. 1+3 | 5, 15 | gügürčin; gögürčün |
| | 15 | | | | |
| 'silver' | 27 | kümüš | 1 vs. 9 | 5, 25, 27 | gümüš |

This Ottoman Turkish feature also prevails over the Kipchak characteristic in the doublets, as there are 85 instances exhibiting the voicing of the initial k-, while only 17 examples display the opposite (83%-17%). Consequently, the Lev of Göz. 1841 presents a significant number of an Ottoman Turkish feature, which were uncommon in older canonical Crimean Karaim texts, and not listed in Prik's Crimean Karaim description.

B-) *t-* > *d-*: Another Ottoman Turkish characteristic involves the voicing of the initial plosive *t-*, which manifests as an initial *d-* in various Oghuzic languages for specific lexemes, while it is maintained in the majority of Kipchak languages (Johanson 2022b: 100), including Western Karaim (Çulha 2019: 78–79). Concerning Crimean Karaim Bible translations, the Kipchak characteristic prominently surfaces, e.g., *tiri* 'alive', *tiši* 'female' (Jankowski 1997: 35), *tigim* 'slice', *tüz* 'straight, plain,' (Németh 2016: 177), *töš* 'breast, bosom', *taya-* 'to prop up' (CrKB I: 178). Analogous to the previous feature, this attribute is also discernible in the eastern dialects of Crimean Tatar (Kavitskaya 2010: 19) and Crimean Turkish (Doerfer 1959a: 275), accompanied by contrasting Kipchak characteristics.

Our dataset demonstrates the presence of the Ottoman Turkish feature, encompassing a total of 161 occurrences across 33 unique lexemes. These instances are distributed throughout 24 distinct chapters within the text.

Table 32: *Voicing of the Initial t-*

| Lev | Ex | Occurrence | |
|------------|--------|----------------------------|----|
| 13, 25, 26 | dayïn | 'still; so far; yet; more' | 12 |
| 21 | daḫï | 'still; so far; yet more' | 1 |
| 1 | damla- | 'to drip' | 1 |

| Lev | Examp | bles | Occurrence | |
|----------------------------------|---------------|--------------------------|------------|--|
| 11 | deg- | 'to touch' | 5 | |
| 11, 13, 15, 27 | degil/dügül | 'not' | 1+8 | |
| 6, 7, 8, 11, 12, 13, 14, 15, 19, | degin/deginčä | 'up to; till' | 49+1 | |
| 22, 23, 24, 25, 26, 27 | | | | |
| 13 | degiš- | 'to exchange' | 9 | |
| 26 | demir | 'iron' | 1 | |
| 11 | deŋiz | 'sea' | 3 | |
| 11 | deri | 'skin' | 1 | |
| 13 | derän | 'deep' | 1 | |
| 11 | deve | 'camel' | 2 | |
| 6 | dilim | 'slice' | 1 | |
| 11, 16 | diri | 'alive' | 3 | |
| 16, 17 | diril- | 'to keep alive' | 2 | |
| 24 | diš | 'tooth' | 2 | |
| 3 | diši | 'female' | 1 | |
| 11 | diz | 'knee' | 1 | |
| 11, 12, 18, 22 | doy- | 'to be born' | 17 | |
| 19 | doyru | 'straight; right' | 4 | |
| 19 | doyrulïķ | 'rightness; correctness' | 1 | |
| 1, 16 | doyuš | 'birth' | 2 | |
| 25 | doķuzunǯï | 'nineth' | 1 | |
| 16, 21 | doldur- | 'to fill' | 2 | |
| 25 | dolyïnǯa | 'enormously; fully' | 1 | |
| 2, 5, 16 | dolu | 'full' | 4 | |
| 19, 23 | dördünži | 'fourth' | 2 | |
| 11, 20 | dört | 'four' | 6 | |
| 11 | duyan | 'falcon' | 1 | |
| 5, 19 | dut- | 'to hold' | 4 | |
| 7, 11, 15, 19 | dürli/dürlü | 'various' | 3+1 | |
| 11 | düš- | 'to fall' | 5 | |
| 14 | düz | 'open field' | 2 | |

The prominence of this Ottoman Turkish feature is most evident in Chapter 11 (44 words), Chapter 13 (18 words), and Chapter 15 (17 words). Conversely, Chapters 4, 9, and 10 lack this feature, while Chapters 2, 3, 5, 8, and 13 each exhibit only a single pertinent instance. Moreover, the initial *t*- is preserved in the text within 27 unique words, appearing a total of 233 times, and is found in every chapter with the exception of Chapter 23.

Table 33: Preservation of the Initial t-

| Lev | Exa | mples | Occurrence | |
|---------------------------------------|-----------|--------------------------|------------|--|
| 7, 27 | tay | 'mountain' | 2 | |
| 5 | tam- | 'to drip' | 1 | |
| 19 | tamya | 'brand-mark' | 1 | |
| 1, 3, 4, 8, 16, 24, 25 | taya- | 'to prop up' | 15 | |
| 21 | tayaķ | 'stick' | 1 | |
| 13 | terän | 'deep' | 7 | |
| 1, 4, 7, 8, 13, 15, 16 | teri/täri | 'skin' | 27+1 | |
| 21 | teš- | 'to pierce' | 1 | |
| 26 | tïnč | 'quiet' | 3 | |
| 19, 26 | tik | 'straight' | 2 | |
| 19 | tik- | 'to plant/set up' | 1 | |
| 2 | tilim | 'slice' | 1 | |
| 2 | tilimlä- | 'to portion out' | 2 | |
| 15, 27 | tip | 'bottom' | 2 | |
| 16 | tiri | 'living; alive' | 2 | |
| 17, 18, 19, 20, 25 | tiril- | 'to keep alive' | 10 | |
| 3, 4, 5, 14, 15, 27 | tiši | 'female' | 9 | |
| 5, 6, 7, 11, 12, 15, 21, 22 | tiy- | 'to touch' | 24 | |
| 23, 25 | toķuz | 'nine' | 2 | |
| 19 | tol- | 'to become full' | 1 | |
| 8,9 | toldur- | 'to fill-' | 2 | |
| 26, 25 | toyyunča | 'enough' | 2 | |
| 1, 2, 4, 8, 9, 14, 15, 17, 19, 22, 26 | tök- | 'to spill out' | 23 | |
| 7, 8, 9, 10 | töš | 'breast; bosom' | 9 | |
| 15 | töšäk | 'bed' | 7 | |
| 21, 22 | tul | 'widow' | 2 | |
| 9, 13, 18, 19, 20, 26, 27 | tur- | 'to get up; stand; stop' | 13 | |
| 14, 19, 25, 26, 27 | tut- | 'to hold' | 22 | |
| 1, 5, 7, 11, 18, 19, 20, 24, 25, 27 | tuvar | 'cattle; animal' | 30 | |
| 9, 21, 26 | tüš- | 'to fall down' | 5 | |
| 17, 26 | tüz | 'straight; plain' | 3 | |

Upon examination of the data, it is evident that the Kipchak feature exhibits a modest predominance over the Ottoman Turkish characteristic, constituting 59% as opposed to 41%. In relation to doublets, the analysis reveals the presence of 13 unique terms.

Table 34: Ottoman Turkish and Kipchak Doublets 2

| Glosses | Lev | Kip. Form | Occurrences | Lev | Ot.Tur Form |
|---------------------|-------------------------|------------|-------------|--------|-------------|
| 'to drip' | 5 | tam- | 1 vs 1 | 1 | damla- |
| 'deep' | 13 | terän | 6 vs 1 | 13 | derän |
| 'skin; leather' | 1, 4, 7, 9, 13, 15, 16 | teri; täri | 52+1 vs 1 | 11 | deri |
| 'slice' | 2 | tilim | 1 vs 1 | 6 | dilim |
| 'alive' | 16 | tiri | 2 vs 3 | 11, 16 | diri |
| 'to keep alive' | 17, 18, 19, 20, 25 | tiril- | 10 vs 2 | 16, 17 | diril- |
| 'female' | 3, 4, 5, 12, 14, 15, 27 | tiši | 11 vs 1 | 3 | diši |
| 'to touch' | 5, 6, 7, 11, 12, 15, | tiy- | 24 vs 6 | 11 | deg- |
| | 21, 22, 23 | | | | |
| 'nine' vs. 'nineth' | 23, 25 | toķuz | 2 vs 1 | 25 | doķuzïnǯï |
| 'to fill' | 8, 9 | toldur- | 2 vs 2 | 16, 21 | doldur- |
| 'to hold' | 14, 19, 25, 26, 27 | tut- | 22 vs 4 | 5, 19 | dut- |
| 'to fall' | 9, 21, 26 | tüš- | 5 vs 5 | 5 | düš- |
| ʻplain' | 17, 25, 26 | tüz | 4 vs 2 | 14 | düz |

Among the doublets, 143 instances display the Kipchak form, which prevails over the 30 documented Ottoman Turkish-type examples (83%-17%). While this Ottoman Turkish feature does not predominate as markedly as the preceding one, it still offers a substantial number of instances that were uncommon in Kipchak languages including Western Karaim (Çulha 2019: 79) and not enumerated in Prik's Crimean Karaim description (1976: 29–34).

C-) b- > \emptyset -: In a restricted group of lexemes, the omission of the initial b- is discernible in Ottoman Turkish and other West Oghuz languages, as opposed to Kipchak languages (Johanson 2022b: 101). This feature can also be found in the eastern dialect of Crimean Tatar (Doerfer 1959b: 379) and Crimean Turkish (Doerfer 1959b: 275), alongside contrasting examples. It should be noted that the Kipchak characteristic of preserving the initial b- in specific words is evident in Western Karaim (Çulha 2019: 48), and it frequently appears in Crimean Karaim Bible translations, e.g., $bil\ddot{a}n$ (Németh 2016: 170), bol- 'to become' (Jankowski 1997: 29; CrKB I: 165). Nevertheless, our data displays mixed features in two distinct words.

Table 35: *Opposition Regarding the Initial b-*

| Glosses | Lev | Kip. Form | Occurrence | Ot.Tur Form | Lev |
|---------|-----------------------|-----------|------------|-------------|-----|
| 'with' | All, except 11 and 13 | bilän | 188 vs 4+1 | ilän/ilen | 11 |

| Glosses | Lev | Kip. Form | Occurrence | Ot.Tur Form | Lev |
|--------------------|----------------|-----------|------------|-------------|------------------|
| 'to be; to become; | All, except 11 | bol- | 297 vs 37 | ol- | 1, 2, 4, 11, 12, |
| to happen' | | | | | 20, 22, 24 |

The occurrence of the initial b- in these two words is clearly predominant over the Ottoman Turkish characteristic (92% vs 8%). It is noteworthy that a significant proportion of this Ottoman Turkish feature (67%) is attested in Chapter 11, which does not exhibit any preservation of the initial b- for the demonstrated words.

D-) b- > v-: In West Oghuz languages, the spirantization of the initial b- is attested in some lexemes (Johanson 2022b: 101). Conversely, the preservation of the initial b- in such lexemes is evidenced in Western Karaim (Çulha 2019: 47) and frequently maintained in Crimean Karaim Bible translations, e.g., ber- 'to give' (Jankowski 1997: 47; CrKB I: 211), bar- 'to go' (Németh 2016: 169). On the other hand, both Ottoman Turkish and Kipchak Turkic characteristics can be observed in the eastern dialect of Crimean Tatar (Doerfer 1959b: 379) and Crimean Turkish (Doerfer 1959a: 275).

In the Lev of the Göz. 1841, merely two distinct items display this Ottoman Turkish feature, with these instances occurring a total of 88 times⁴⁷.

Table 36: *Spirantization of the Initial b-*

| Glosses | Lev | Occurrence | Ot.Tur Form |
|----------------|--------------------------|------------|-------------|
| 'there is/are' | 25 | 1 | var |
| 'to give' | all except 3, 12, 13, 21 | 87 | ver- |

It is essential to highlight that the Kipchak counterparts *ber*- 'to give' and *bar* 'there is/are', or any other Kipchak forms contrasting the Ottoman Turkish feature of spirantization of the initial *-b*, are not present in our dataset. In this regard, the Lev of Göz. 1841 unequivocally demonstrates an Ottoman Turkish characteristic.

E-) b- > m-: The nasalization of the initial b- to initial m- is typically observed in Turkic languages, encompassing both Kipchak and Oghuzic languages, with the exception of Ottoman Turkish (Johanson 2022b: 101). Notably, together with Ottoman Turkish, the retention of the

_

⁴⁷ Although the letter *beth* (2) can represent both v and b, all instances of the spirantization of the initial b- in the Lev of Göz. 1841 are written with the letter vav (1), which clearly indicates v.

initial *b*- in specific terms can also be observed in the context of the eastern dialect of Crimean Tatar (Doerfer 1959b: 383) and Crimean Turkish (Doerfer 1959a: 277), providing contrasting instances. Contrarily, the nasalization of the initial *b*- to initial *m*- is well documented in Western Karaim (Çulha 2019: 47) and frequently attested in Crimean Karaim Bible translations, e.g., *men* 'I' (Jankowski 1997: 33; Németh 2016: 185), *muni*' 'this+ACC' (CrKB I: 213).

In the Lev of the Göz. 1841, seven distinct lexemes manifest these opposing attributes. Of these, three words also appear alongside their respective doublets.

Table 37: *Nasalization of the Initial b-*

| Glosses | Lev | Kip. Form | Occurrences | Lev | Ot.Tur Form |
|----------------|-----------------------------|-----------|-------------|--------|-------------|
| 'brain' | - | - | 0 vs 1 | 24 | beyin |
| 'I' | 11, 14, 17, 18, 19, 20, 21, | men | 67 vs 2 | 11 | ben |
| | 23, 24, 25, 26 | | | | |
| 'I+GEN' | 25 | menim | 1 vs 0 | - | - |
| 'moustache' | 13 | mïyïķ | 1 vs 0 | - | - |
| 'saddle beast; | | - | 0 vs 1 | 15 | binek |
| saddle' | | | | | |
| 'this+DAT' | 10, 13, 14, 25 | типа/тиђа | 28+1 vs 1 | 23 | bиŋa |
| | | | | | |
| 'this+ACC' | 26 | типї | 1 vs 3 | 11, 26 | bunu/bunï |
| 'this+GEN' | 16, 26 | типїŋ | 2 vs 0 | - | - |

Nonetheless, the Kipchak forms are more prevalent than the Ottoman Turkish forms, with a ratio of 93% to 7%.

F-) **b-** > **p-**: In specific Turkic lexemes, the devoicing of an initial *b-* to a *p-* is observed in a limited number of attested instances (Johanson 2022b: 101). This type of devoicing is found in certain words within Ottoman Turkish, such as *parmaķ* (TLO I: 647). In Crimean Tatar, fluctuations are observed even for the same lexemes, e.g., *barmaķ* 'finger' vs *parmaķ* 'id' (KRUS: 97; 431). Conversely, in Western Karaim, the preservation of the initial *b-* is encountered (Çulha 2019: 48), e.g., T.Kar *barmaḥ* 'finger', *biš-* 'to cook' (KRPS: 103; 125); H.Kar *barmaḥ* 'finger', *bis-*'to cook' (KRPS: 103; 123).

In a number of examples within our dataset, these blended attributes are discernible, yet there are no occurrences of doublets. Notably, the Kipchak traits surpass the Ottoman Turkish traits, constituting 86%, compared to the latter's 14%.

Table 38: Ottoman Turkish and Kipchak Doublets 3

| Glosses | Lev | Kip. Form | Occurrences | Ot.Tur form | Lev |
|------------------|-----------------|-----------|-------------|-------------|--------------|
| 'finger' | 4, 8, 9, 14, 16 | barmaķ | 24 vs 0 | - | - |
| 'grape molasses' | 2 | bekmäz | 1 vs 0 | - | - |
| 'to cook' | - | - | 0 vs 4 | pišir- | 2, 8, 24, 26 |

G-) ol > o: The final characteristic manifests in the word-final position, where the Turkic third person singular personal and demonstrative pronoun ol typically appears without the terminal sonorant -l as o in West Oghuz languages, encompassing Ottoman Turkish. Conversely, in Western Karaim, the sole form is ol (Prik 1976: 98–100). Prik's depiction of Crimean Karaim reveals the presence of the Oghuzic form o (Prik 1976: 96–100), while in Crimean Karaim Bible translations, only the variant ol is encountered (Jankowski 1997: 31; Németh 2016: 174; CrKB I: 165). Notably, the o variant is also found in Crimean Tatar (Kavitskaya 2010: 45–46) and coexists with its counterpart ol in Crimean Turkish (Doerfer 1959a: 277).

In our dataset, there are 78 instances where the demonstrative and third person personal pronoun⁴⁸ *o* can be attested. With the exception of a single example in Chapter 7 (Lev 7:8), all pertinent occurrences are found in Chapter 11. Meanwhile, the Kipchak counterpart *ol* appears 1,377 times. As a result, the Kipchak feature significantly dominates the Ottoman Turkish trait (95% vs. 5%). However, Chapter 11 persistently exhibits the Ottoman Turkish characteristic, which is atypical in other Crimean Karaim Bible translations.

2.1.2.2.2. Other Consonantal Features

2.1.2.2.2.1. Aspiration

In the Lev of the Göz. 1841, the aspiration of the velar voiceless plosive -k- to velar fricative -h- can be attested in medial position in some Turkic and non-Turkic origin lexemes.

Table 39: Aspiration of -k->-h-

Göz. 1841 C.Tat Ot.Tur H.Kar T.Kar **Examples** C.Kar ahča akča; aķča aķča akča; ahca; ahča [Lev 22:11] (ETD: 477) (KRPS: 86) 'money' aḥča; aḥčï aķče; aķčï akca <Tur> (CC: 39) (CKED: (KRUS: (KRPS: 58; 35; 44) 46-47) 86)

⁴⁸ In the Lev of the Göz. 1841 and other Karaim Bible translations, due to its literal manner of translation, the pronoun o/ol usually replaces the Hebrew definite article $\lnot \lnot \lnot [ha]$ (see 2.2.2.5.2).

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|-----------------------|---------------|--------------|-----------|-------------|--------------|--------------|
| (waķt] 'time' وقت | vaḫt vs vaķit | vaķīt; vaḫt; | vaķīt; | vakt; vaķit | vaḫt | vaḫt |
| <ar></ar> | [Lev 16:2; | vahït | vakit | (ETD: 1112) | (KRPS: 156) | (KRPS: 156) |
| (NS: 932) | 14:57] | (CKED: | (KRUS: | | | |
| | | 436) | 131) | | | |
| adruķ 'different | ayruḫsï | ayruḫsï; | - | - | ayrïķsï | ayrïḫsï; |
| form; superior to | [Lev 6:9] | ayruķsu; | | | (KRPS: 54) | ayrïķsï |
| (others)' <tur></tur> | | ayruķsï | | | | (KRPS: 55) |
| (EDPT: 65) | | (CKED: | | | | |
| | | 62) | | | | |
| yaķīš- 'to | yaḫšï | yaḫšï; | yaḫšï | yaķīšīķ; | yaķsï | yaḫšï; yaḳšï |
| approach' <tur></tur> | [Lev 5:4] | yaķšï | (KRUS: | yaķīšīķlï | (KRPS: 219) | (KRPS: 219; |
| (EDPT: 908-909) | | (CKED: | 830) | (suitable; | | 241) |
| > yaḫši ʻgood; | | 441) | | handsome, | | |
| good looking'49 | | | | beautiful) | | |
| <tur></tur> | | | | (ETD: 1129) | | |
| (EDPT: 908) | | | | | | |
| taķī 'and; also; | daḫï | daḥï; daḥa; | taa; daa; | daha | dahïn; dayïn | dahïn |
| more' | [Lev 21:11] | daḥan; | taya | (ETD: 700) | (KRPS: 168) | (KRPS: 168) |
| <tur> (EDPT:</tur> | | daḥïn | (KRUS: | | | |
| 466) | | (CKED: | 534; 537) | | | |
| | | 130; 131) | | | | |

This feature is also present in other Crimean Karaim Bible translations, e.g., *valyt* 'time' (Sulimowicz 1972: 61; Németh 2016: 175; CrKB I: 213); *alyča* 'money' (Jankowski 1997: 41). It further manifests in Crimean Turkish (Doerfer 1959a: 272), appears sporadically in Crimean Tatar (Doerfer 1959b: 372), and is also found in Ottoman Turkish (Kerslake 2022: 178). Notably, this feature is prevalent in Western Karaim, particularly in Trakai Karaim (Musaev 1964: 71).

2.1.2.2.2.2. The Alternation Between $t \sim \check{c}$ and $t \sim k$

In certain lexical items of Crimean Karaim Bible translations, the initial consonant t- preceding front vowels occasionally undergoes a shift to the affricate \check{c} - or, at times, to the stop k-, which is a distinct feature of Western Karaim (Jankowski 1997: 9). This particular characteristic cannot be attested in Crimean Tatar, Crimean Turkish, or Ottoman Turkish. In this regard, the

_

⁴⁹ According to Clauson (EDPT: 908), the shift from $-k\tilde{s}$ to $h\tilde{s}$ is not documented prior to the 11th century.

Lev of the Göz. 1841 also presents a few instances of this phonological change, with one example appearing alongside its counterpart.

Table 40: *Consonant Shift Between t- > č- and t- > k-*

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|---------------|---------------------------------|----------------|--------|----------------|-------------|-------------|
| topla- | <i>čöplä</i> ⁵⁰ - vs | topla-; čöple- | topla- | topla- | topla-; | topla-; |
| 'to collect' | topla- | (CKED: 126- | (KRUS: | (ETD: 802-803) | copla- | ćopĺa; |
| <tur></tur> | [Lev 19:10; | 127; 408) | 600) | | (KRPS: 538; | (KRPS: 538; |
| (EDPT: 440) | 23:39] | | | | 614) | 628) |
| tile- | kilä- | kile-; dile- | tile- | dile- | kile- | kĺa- |
| 'to desire' | [Lev 26:21] | (CKED: 138; | (KRUS: | (ETD: 691–692) | (KRPS: 326- | (KRPS: 327) |
| <tur></tur> | | 211) | 587) | | 327) | |
| (EDPT: 492) | | | | | | |
| tilek | kiläk | dilek; kilek | tilek | dilek | klek | kĺak |
| 'wish desire' | [Lev 26:30] | (CKED: 138; | (KRUS: | (YTL: 343) | (KRPS: 327) | (KRPS: 327) |
| <tur></tur> | | 211) | 587) | | | |
| (EDPT: 498) | | | | | | |

2.1.2.2.3. Preservation and Devoicing of Final Lenes in Arabic and Persian Loanwords

It is widely acknowledged that the majority of Turkic languages contain a significant number of Arabic and Persian loanwords. The incorporation of these words leads to varying phonotactic patterns among the Turkic languages. For example, the fortition of final voiced stops in Arabic and Persian loanwords is predominantly observed in contemporary Turkish. Indeed, the subsequent table also demonstrates that Karaim dialects and Crimean Tatar exhibit devoicing of final lenes in most of the provided examples, which are uninflected forms.⁵¹ Notably, the Ottoman Turkish examples may not accurately reflect this phenomenon, as their phonological values were not consistently recorded in dictionaries.

In the majority of examples from our dataset, the preservation of final lenes in Arabic and Persian words is evident, as illustrated by the table below.

Table 41: Preservation of Final Lenes in Arabic and Persian Loanwords

⁵¹ For the situation regarding inflected forms, please refer to section 2.2.1.3.

⁵⁰ Presumably, this occurs through the form *töplä*- (Jankowski 1997: 9).

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|---------------------------|--------------|---------------|--------------|---------------|-------|--------|
| [awlād] 'children' أو لاد | evlad | evlad | evlat | evlad | - | - |
| <ar> (NS: 265)</ar> | 'child; son' | 'child; son' | 'child; son' | 'child; son' | | |
| | [Lev 11:2] | (CKED: 158) | (KRUS: 767) | (ETD: 508) | | |
| בر [dard] 'pain; sorrow; | derd | dert; derd | dert | derd | - | - |
| trouble' <per></per> | [Lev 13:57] | (CKED: 136) | (KRUS: 160) | (ETD: 685) | | |
| (NS: 202) | | | | | | |
| خراب [ḫarāb] | ḥarab | ḥarap; ḥarab | ḥarap | <i>ḥarab</i> | - | ђагар |
| 'devastated; destroyed' | [Lev 26:33] | (CKED: 177) | (KRUS: 673) | (ETD: 667) | | (KRPS: |
| <ar> (NS: 343)</ar> | | | | | | 600) |
| جساب [ḥisāb] | ḥesab | ḥesab; ḥesap; | esap | <u></u> ḥesab | - | - |
| 'calculation' <ar></ar> | [Lev 25:27] | esap | (KRUS: 788) | (ETD: 650) | | |
| (NS: 359) | | (CKED: 184) | | | | |
| [murād] مراد | murad vs | mïrad; murad; | murat; mïrat | murad | - | - |
| 'wish; desire' <ar></ar> | morad | murat | (KRUS: 367- | (ETD: 1002) | | |
| (NS: 611) | [Lev 22:29; | (CKED: 241) | 368) | | | |
| | 19:5] | | | | | |
| "išāhid] 'witness' شاهد | šaḥad | šaḥad; šahat | šaat | šahid | - | - |
| <ar> (NS: 828)</ar> | [Lev 5:1] | (CKED: 370) | (KRUS: 745) | (ETD: 754) | | |

However, despite the consistent presence of this feature, the Lev of the Göz. 1841 also exhibits a few inconsistent cases, which occur infrequently.

Table 42: Devoicing of Final Lenes in Arabic and Persian Loanwords

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|-----------------------|--------------|-------------|------------|------------|---------------|-------------|
| ['ayb] عيب | ауїр | ayïb; ayïp | ауїр | ауїр | ауїр | ayïp; ayip |
| 'fault; error; | [Lev 15:3] | (CKED: 60) | (KRUS: 42) | (ETD: 835) | (KRPS: | (KRPS: 56) |
| shame' <ar></ar> | | | | | 56) | |
| (NS: 84) | | | | | | |
| [āzād] 'free' | azat | azat | azat | azat | azat | azat |
| <per> (NS: 86)</per> | [Lev 19: 20] | (CKED: 64) | (KRUS: 34) | (ETD: 451) | (KRPS: | (KRPS: 48) |
| | | | | | 48) | |
| [žōmard] جومرد | žomart | žomart | žemert; | žumerd | <i>30mart</i> | žomart |
| 'generous' | [Lev 22:18] | (CKED: 106) | žumert | (ETD: 626) | (KRPS: | (KRPS: 176) |
| <per> (NS: 157)</per> | | | (KRUS: | | 173) | |
| | | | 734) | | | |

2.1.2.2.2.4. Consonant Deletion

In the Lev of the Göz. 1841, three examples exhibit consonant deletion. In two instances, the deletion occurs in Arabic loanwords that initially possessed double medial consonants.

Table 43: Consonant Deletion in Arabic Words

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|---------------------|-------------|-------------|---------------|------------|--------|--------|
| [sunna(t)] سنّة | sünät | sünet | sünnet | sünnet | sunet | - |
| 'tradition'; | [Lev 12:3] | (CrKB: 366) | (KRUS: 528) | (ETD: 743) | (KRPS: | |
| 'circumcision' | | | | | 484) | |
| <ar> (NS: 822)</ar> | | | | | | |
| [kuwwa(t)] قوّة | ķuvat | ķuvat | ķuvet; ķuvvet | ķuvvet | ķuvat | ķuvat |
| 'power; strength' | [Lev 26:26] | (CrKB: 323) | (KRUS: 297) | (ETD: 896) | (KRPS: | (KRPS: |
| <ar></ar> | | | | | 342) | 342) |
| (NS: 519) | | | | | | |

The examples clearly establish a differentiation between Karaim dialects and Crimean Tatar, along with Ottoman Turkish, as the double consonants have been omitted in all three Karaim dialects and the analyzed text.

In the final example, a Turkic lexeme displays the deletion of the initial *y*- in the text. It is widely recognized that the initial *y*- in Old Turkic has experienced particular shifts in modern Turkic languages. Although the initial *y*- is present in Karaim (Johanson 2022b: 96) and our dataset, we encounter a single instance exhibiting deletion.

Table 44: Deletion of Initial y-

| Example | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|--------------------------|------------|---------------|---------------|-------------|-------|-------|
| yüksek | öksäk | öksek; yüksek | yüksek; öksek | yüksek | - | - |
| 'tall; high' <tur></tur> | [Lev 8:13] | (CKED: 266) | (KRUS: 800) | (ETD: 1147) | | |
| (EDPT: 915–916) | | | | | | |
| | | | | | | |

2.1.2.2.5. Addition of a Consonant

In our corpus, we present a solitary instance that demonstrates the incorporation of a consonant within an Arabic lexical unit.

Table 45: Addition of a Consonant

| Example | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|----------------------|--------------|-------------|-------|------------|-------|-------|
| ميلة [ḥīla(t)] | <u>ķillä</u> | ḥillä | - | hile | - | - |
| 'trick; ploy; | [Lev 21:17] | (CKED: 185) | | (ETD: 661) | | |
| stratagem' <ar></ar> | • | | | | | |
| (NS: 364) | | | | | | |

Additionally, there is a feature in the Arabic script, the adoption of which exhibits variations in our text and several Turkic languages. In Arabic, the so-called *taa marbuta* (\tilde{s}) is transcribed as t and signifies the feminine gender of a noun or adjective. However, in its base form, the *taa marbuta* is not pronounced as a t. Yet, in specific grammatical contexts, such as when followed by a word beginning with a vowel or when the word is made definite with the definite article J [$\bar{a}l$], the taa marbuta may be pronounced as a t.⁵²

In our text, there are instances where the *t* is not pronounced in Arabic yet appears, whereas in other cases, it is not displayed.

Table 46: Variability in the Adoption of Arabic Taa Marbuta

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|---------------------------------------|-------------|----------------------|--------------|---------------|--------|---------|
| شريعة [$\check{s}arar{\iota}`a(t)$] | šaraʻat | šara [°] at | šeryat | šeriat | - | šarayat |
| 'law; religious | [Lev 18:4] | (CKED: 371) | (KRUS: 754) | (ETD: 758) | | (KRPS: |
| regulations' | | | | | | 644) |
| <ar></ar> | | | | | | |
| (NS: 837) | | | | | | |
| فائدة [$far{a}$ ' $ida(t)$] | fayda | fayda | fayda; payda | fa'ide; fayda | fayda | fayda |
| 'benefit; | [Lev 27:37] | (CKED: 160) | (KRUS: 651) | (TLO II: | (KRPS: | (KRPS: |
| advantage' | | | | 3460) | 592) | 592) |
| <ar></ar> | | | | | | |
| (NS: 275) | | | | | | |

As a result, the presence of the final -*t* in certain Arabic lexemes throughout our text and other Turkic languages displayed in the table above cannot be considered as the addition of a consonant, but rather as a fluctuation in the adoption of the Arabic phenomenon.

2.1.2.2.6. Replacement of Arabic Epiglottal Stop Ayin and Glottal Stop Aleph

In the preponderance of Arabic lexemes examined within our corpus, we do not discern any particular phonotactic adaptation concerning the representation of the Arabic epiglottal stop

⁵² For additional information, refer to Ryding (2005: 21–24).

ayin (ع) and glottal stop aleph (ا), even though such consonants are typically absent in a majority of Turkic languages, encompassing Karaim, Crimean Tatar, and Ottoman Turkish. Given that the Hebrew script incorporates the corresponding characters present in Arabic, specifically aleph [x] and ayin [y], these consonants frequently manifest in our transcriptions, despite their lack of phonological relevance in the dataset under investigation, e.g., [Lev 25:36] fa^iiz 'interest; usury' > فانض $[f\bar{a}^iid]$ 'surplus; excess; overflowing' (NS: 270). Nevertheless, in four instances, we observe that the adopted consonant in a medial position has been replaced by a consonant or a vowel, contingent upon the placement of the specified letters. For instance, in three cases, aleph is situated between two vowels and is substituted by the consonant y.

Table 47: Replacement of Arabic Glottal Stop Aleph

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|---|-----------|-------------|--------------|---------------|--------|--------|
| دائم [dāʾim] دائم | dayim | dayïm | daim; dayïm | da'im | - | - |
| 'continuing; lasting' | [Lev 6:6] | (CKED: 133) | (KRUS: 149) | (ETD: 242) | | |
| <ar> (NS: 184)</ar> | | | | | | |
| فائدة [$f\bar{a}$ ' $ida(t)$] 'benefit; | fayda | fayda | fayda; payda | faʾide; fayda | fayda | fayda |
| advantage' <ar></ar> | [Lev | (CKED: 160) | (KRUS: 651) | (TLO II: | (KRPS: | (KRPS: |
| (NS: 275) | 27:37] | | | 3460) | 592) | 592) |
| فيئات or [fiyāt] فيات | fiḥat | fiyat | fiyat; piyat | fì ʾat | - | - |
| $[f\bar{t}\bar{t}]$ <ar> 'in;</ar> | 'price' | 'price' | 'price' | 'price' | | |
| at+PL ⁵³ ' (NS: 288) | [Lev | (CKED: 163) | (KRUS: 661) | (ETD: 861) | | |
| | 19:20] | | | | | |

As evidenced, a similar strategy is also observed in other languages, as indicated in the table shown above.

In a particular instance involving a word of Arabic origin, the original letter ayin, situated between two consonants, is replaced by the vowel \ddot{i} in our text. The subsequent table reveals that this distinctive arrangement of the word has not been identified in the other languages illustrated.

Table 48: Replacement of Arabic Epiglottal Consonant Ayin

_

⁵³ In Arabic, the preposition $\stackrel{.}{=}$ [fi], meaning 'in' or 'at', conveys location, position, or a specific point in time. This preposition is followed by the plural suffix $\stackrel{.}{=}$ [$\bar{a}t$] in our example. However, the meaning of 'price' is explicitly defined in Ottoman Turkish (NS: 288) and does not appear in Arabic.

| Example | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|-----------------------------------|------------|-------------|-------------|--------|-------|-------|
| " (šamʾ] 'candle' (غمض [šamʾ) شمع | šamïdan | šamdan | šamdan | šamdan | - | - |
| دان [$d\bar{a}n$] 'holder' = | [Lev 24:4] | (CKED: 371) | (KRUS: 748) | (KRUS: | | |
| 'candlestick ' <ar> +</ar> | | | | 763) | | |
| <tur> (NS: 830)</tur> | | | | | | |

2.1.2.2.2.7. Metathesis

There are two types of metathesis in the text, with a total of five examples. The first occurs between the final consonant of the first syllable and the first consonant of the second syllable.

Table 49: *Metathesis*

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|---------------------|------------|-------------|----------------|------------|-------------|----------|
| yaymur | yamyur | yaymur | yaymur | yaymur | yamyur | yamyur; |
| 'rain' <tur></tur> | [Lev 26:4] | (CKED: 440) | (KRUS: 809) | (ETD: 273) | (KRPS: | yanyur; |
| (EDPT: 903- | | | | | 224–225) | yaŋɣur; |
| 904) | | | | | | (KRPS: |
| | | | | | | 224–225) |
| köŋlek | gölmäk | kölmek; | gölmek; kölmek | gömlek | kelmek | keĺṁak |
| 'shirt' <tur></tur> | [Lev 8:7] | gölmek | (KRUS: 141) | (ETD: 317) | (KRPS: 302) | (KRPS: |
| (EDPT: 732) | | (CKED: 216- | | | | 302) |
| | | 217; 167) | | | | |

It can be discerned that the phenomenon of metathesis in *köŋlek* is prevalent in Karaim dialects and Crimean Tatar. Conversely, metathesis in *yaymur* is exclusively observed in Western Karaim. Notably, the demonstrated examples do not showcase this feature in Ottoman Turkish sources.

Additional instances of metathesis transpire within the same syllable, specifically in a vowel-consonant sequence. It is worth noting that examples illustrating metathesis are found alongside their counterparts in Crimean Karaim dictionaries. Nevertheless, the other languages exhibited in the subsequent table do not display this form of metathesis.

Table 50: Metathesis 2

| Example | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|------------------|--------------|-------------------|-------------|------------|-------|-------|
| roba 'clothes; | urba vs ruba | ruba; urba; uruba | ruba; urba | ruba | - | - |
| dress' <it></it> | | (CKED: 328; 427) | (KRUS: 461) | (ETD: 717) | | |

| Example | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|-----------------------|-------------|------------------|-------------|------------|---------|---------|
| (NS: 743) | [Lev 11:40; | | | | | |
| | 11:28] | | | | | |
| [niyya(t)] نيّة | inyet | inyet; niyet | niet | niyyet | niyet | niyeť |
| 'intention' <ar></ar> | [Lev 7:16] | (CKED: 197) | (KRUS: 393) | (ETD: 172) | (KRPS: | (KRPS: |
| (NS: 648) | | | | | 419– | 420- |
| | | | | | 420) | 421) |
| [ķirmizī] قرمزی | ķïrïmzï | ķīrīmzī; ķīrmīzī | ḥïrmïzï; | ķïrmïzï | ķïrmïzï | ķïrmïzï |
| 'red' <per></per> | [Lev 14:49] | (CKED: 305) | ķïrmïzï | (ETD: 279) | (KRPS: | (KRPS: |
| (NS: 472) | | | (KRUS: 684) | | 387) | 387) |

2.1.2.3. Vowel-Consonant Sequences

2.1.2.3.1. Vowel-Consonantal Alternation

A-) $[\ddot{o}y] > [ev]$: In the dataset under analysis, we observe a vowel-consonant variation, specifically $[\ddot{o}y]$ transitioning to [ev]. This shift was previously noted by Jankowski, who posited that it is commonplace in Crimean Karaim and is likely rooted in the southern dialect of the Crimean Tatar language. Illustrative examples provided by Jankowski include evle, a variation of $\ddot{o}yle$ 'so; such; thus', and sevle-, a variation of $s\ddot{o}yle$ - 'to say' (Jankowski 2010: 88). Remarkably, our corpus also manifests a single occurrence of $sevl\ddot{a}$ -, which is likewise attested in Crimean Turkish (Doerfer 1959a: 277). In fact, this specific element exclusively emerges in Chapter 11, a section predominantly characterized by the linguistic traits of Crimean Turkish.

Table 51: *Shift from* $[\ddot{o}y] > [ev]$

| Example | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|-------------------|------------------|-------------|---------------|--------|--------|--------|
| sözle- 'to speak; | sevlä- vs sözlä- | sözle- | söyle-; söle- | söyle- | sezle- | śoźĺa |
| to say' | [Lev 11:2; 1:1] | (CKED: 363) | (KRUS: 489) | (ETD: | (KRPS: | (KRPS: |
| (EDPT: 863) | | | | 749) | 497) | 471) |

B-) [iy]; [ay] > [uv] ~ [ov]: It is widely recognized that the Old Turkic fricative y has diverse manifestations in the common Turkic languages, each adopting distinct strategies. It is of note that in numerous Kipchak languages, the phoneme presents itself as v. As noted in section 2.1.2.1.3, the consonant v has been observed to trigger a shift from unrounded to rounded vowels in select instances. Therefore a vowel-consonant sequence [Xy] in certain words in Old Turkic typically resonates as [O], [Uv], and [Iv] in a variety of Kipchak languages, commencing from the Middle Kipchak period (Berta and Csató 2022a: 154). In the context of Ottoman

Turkish, the fricative intervocalic and syllable-final γ generally experiences elision (Kerslake 2022: 178).

Within our dataset, a systematic transition from Old Turkic [iy] and [ay] into [uv] or [ov] is observed, attributable to the aforementioned factors.

Table 52: Shift from [iy] and [ay] to $[uv] \sim [ov]$

| Examples | Göz. 1841 | C.Kar | C.Tat | Ot.Tur | H.Kar | T.Kar |
|--|-------------|----------------|-------------|---------|--------|--------|
| arïy 'clean; pure' | aruv | aruv | aruv 'good' | arï | aruv | aruv |
| <tur> (EDPT: 213)</tur> | [Lev 6:4] | (CKED: 50) | (KRUS: 70) | (TLO I: | (KRPS: | (KRPS: |
| | | | | 152) | 76) | 76) |
| arkay 'the woof (or | arķuv | arķuv 'weft' | - | - | - | - |
| weft) of a woven | [Lev 13:48] | (CKED: 49) | | | | |
| material' <tur></tur> | | | | | | |
| (EDPT: 213) | | | | | | |
| buzayu 'calf' | bïzov | bïzuv; buzuv | buzav | buzayï | buzov | buzov |
| <tur>; <i>buzay</i>⁵⁴</tur> | [Lev 9:2] | (CKED: 83) | (KRUS: | (ETD: | (KRPS: | (KRPS: |
| 'calf' <tur></tur> | | | 124) | 545) | 137) | 137) |
| (EDPT: 391) | | | | | | |
| tarıylay 'cultivated | tarlov | tarla; tarlav; | tarla | tarla | tarlav | tarlav |
| field; field' <tur></tur> | [Lev 19:19] | tarlov; tarluv | (KRUS: | (ETD: | (KRPS: | (KRPS: |
| (EDPT: 541) | | (CKED: 385) | 551) | 799) | 515) | 515) |

As the table above demonstrates, our data, the three Karaim dialects, and Crimean Tatar exhibit a common characteristic regarding this feature in most cases, unlike Ottoman Turkish.

2.2. Morphology and Morphophonology

In this section, the analysis of morphology and morphophonology within the dataset will be carried out, with comparisons drawn between Crimean Tatar, Crimean Turkish, Ottoman Turkish, and all three dialects of Karaim. It is important to acknowledge that the availability of Crimean Turkish data is limited, which may lead to gaps in various categories. As specified in the relevant sections, the usage of most categories within sentences, along with their functions, will be analyzed in the syntax section (2.3.). As a result, this section's primary focus will be on

_

⁵⁴ This form has been observed in the Chagatai language.

presenting specific markers, their classification, morphological variants, and their corresponding equivalents in the Turkic languages under investigation.

2.2.1. Morphophonological Characteristics

2.2.1.1. Intrasyllabic Vowel Harmony

As previously stated (refer to section 2.1.2.1.1), the majority of Turkic languages demonstrate intersyllabic vowel harmony concerning backness and frontness in Turkic lexemes. Regarding inflected lexical items, we observe the same pattern of intrasyllabic vowel harmony in our text, and this is also evident in most Turkic languages, including Crimean Tatar, Turkish, Ottoman Turkish, and Karaim. However, exceptions also arise in specific cases, due to unvariable suffixes, such as {+Daš}, {+mač}, and {+Day} in Crimean Tatar, the {-yor} suffix in Ottoman Turkish, and Halich Karaim copula suffixes (Berta and Csató 2022b: 322). Moreover, the shifting of [ay] to [äy] in Trakai Karaim also leads to the dissolution of intersyllabic vowel harmony (Musaev 1964: 55–56). However, our data does not present any such anomalies.

In relation to rounding harmony in inflected words, our data generally does not demonstrate complete rounding harmony, mirroring the pattern seen in other Crimean Karaim Bible translations.

(1) [Lev 7:33]
$$bol$$
- sin to be-3SG.VOL 'it shall be'. [Lev 26:16] $urlu[k>y]+iniz$ offspring+2PL.POSS 'your offspring'

However, our data does contain contrasting examples. Similar fluctuations are also observed in the Eastern dialect of Crimean Tatar (Kavitskaya 2010: 26), Crimean Turkish (Doerfer 1959a: 274), and all three Karaim dialects (Musaev 1964: 50–51; Çulha 2019: 38–39). Regarding Ottoman Turkish, it is important to highlight that the development of rounded suffix harmony was a lengthy and intricate process, which did not culminate until the eighteenth century (Johanson 2022b: 106).

It warrants mentioning that the deverbal nominal formative marker {-(U)v} (refer to section 2.2.2.1.1.) demonstrates in certain instances that the terminal unrounded vowels of the verb stems undergo rounding prior to the application of this marker due to regressive assimilation.

(3) [Lev 8:35]
$$sakla-v > saklov$$
 to keep-DER 'holding'.

Finally, it is noteworthy that the rounding of suffixes, along with their distribution and occurrence, is an important and distinctive characteristic within Crimean Karaim Bible translations. Therefore, such variations of suffixes will be presented in detail in the relevant sections devoted to them.

2.2.1.2. Consonant Harmony

In several Turkic languages, a progressive assimilation process occurs whereby suffixes featuring initial voiced consonants are devoiced when they affix to words concluding with final voiceless consonants. This characteristic is observed in Crimean Tatar (Kavitskaya 2010: 34), Crimean Turkish (Doerfer 1959a: 275) Ottoman Turkish (Kissling 1960: 27–28; 44), and across all three dialects of Karaim (Çulha 2019: 89-91; Berta and Csató 2022b: 322). For instance, in our data, the past tense marker {+DI} undergoes morphophonological alterations influenced by both fronting harmony and consonant harmony when attaching to verb stems.

(4) [Lev 6:10] *ver-di-m* to give-PAST-1SG 'I have given'. [Lev 16:2] *ayt-ti*-Ø to say; to tell-PAST-3SG '|Lord| said'.

However, there exist certain instances where we encounter exceptions.

(5) [Lev 26:43] *hor it-di-\O* to despise-PAST-3PL 'they despised'.

It should be noted that all such exceptions will be discussed in the relevant section dedicated to the corresponding suffixes.

2.2.1.3. Intervocalic Lenition

In a range of Turkic languages, final stops, such as p, k, t, and \check{c} , can be observed in both monosyllabic and polysyllabic words, undergoing lenition when preceding a vowel in specific instances (Johanson 2022a: 28). This feature is also corroborated in Crimean Tatar (Kavitskaya 2010: 10), Crimean Turkish (Doerfer 1959a: 277), Ottoman Turkish (Kissling 1960: 22, 25–26), and across all three dialects of Karaim⁵⁵ (Musaev 1964: 82–83; Prik 1976: 43). Nonetheless, it is important to mention that this characteristic manifests sporadically, contingent upon the particular lexemes. In addition, it can be examined under two categories: those found within verb stems and those within nominal stems. In the context of nominal stems, this attribute is observed in numerous instances within our collected data.

⁵⁵ In the descriptions of Karaim dialects, the lenition of $\check{c} > \check{\jmath}$ was not demonstrated. According to Berta and Csató (2022b: 321), lenition of \check{c} sporadically occurs in West Kipchak languages.

(6) [Lev 1:11]
$$u\check{c}+u>u\check{z}u$$
 end+3SG.POSS
[Lev 1:3] $e\check{s}ik+i+n\ddot{a}>e\check{s}igin\ddot{a}$ entrance+3SG.POSS+DAT
[Lev 8:15] $barmak+\ddot{i}>barmay\ddot{i}$ finger+3SG.POSS
[Lev 18:18] $ay\ddot{i}p+\ddot{i}>ay\ddot{i}b\ddot{i}$ nakedness+3SG.POSS

surat+ïm > *suradïm*

However, there exist a few instances, albeit infrequent, that demonstrate inconsistencies.

face+1SG.POSS

(7) [Lev 2:2]
$$avu\check{c}+\ddot{\imath}>avu\check{c}\ddot{\imath}$$
 handful+3SG.POSS [Lev 11:33] $k\ddot{a}p+\ddot{\imath}>k\ddot{a}p\ddot{\imath}$ vessel+3SG.POSS [Lev 13:55] $art+\ddot{\imath}+n+da>art\ddot{\imath}nda$ back+3SG.POSS+PRNM+LOC

As for verb stems, According to Jankowski's description (2015a: 202), the lenition of the voiceless stops -k-, -k- and -p- between vowels in verbs is also present in Crimean Kipchak Karaim variant, with the exception of the case of -t-, e.g., $et\ddot{a}r$ '(she/he/it) does'. Examples showing this feature are evidenced in other Crimean Bible translations as well, e.g., $\breve{c}\ddot{r}yar$ - 'to take out; to extract; to bring out; to raise' (Jankowski 1997: 31; Németh 2016: 178; CrKB I: 191). In the Lev of the Göz. 1841, there exist two examples that show the lenition of the voiceless k > y and t > d in a verb stem. However, a counterpart of this example is also present.

Notably, in the majority of the examples, we do not observe the lenition of such plosive consonants in verb stems when they are affixed with certain suffixes.

(9) [Lev 7:4]
$$ket\ddot{a}r$$
 to take away; to remove [Lev 9:1] $\check{c}a\dot{k}\ddot{i}r$ to call [Lev 16:30] $et-\ddot{a}r-\emptyset$ to make-AOR-3SG

2.2.1.4. Vowel Deletion

[Lev 26:11]

In various Turkic languages, it is observed that when suffixes initiating with vowels applied to polysyllabic words bearing high vowels (i, i, u, \dot{u}) in their final syllables, frequently induce the elimination of these vowels. This phenomenon is further substantiated by instances found in

Crimean Turkish (Doerfer 1959a: 274), Crimean Tatar (Doerfer 1959b: 376–377), Ottoman Turkish (Kissling 1960: 30–36), and the three dialects of Karaim (Musaev 1964: 60; Çulha 2019: 40–41). Notably, this feature is primarily attested in our dataset.

(10) [Lev 8:7; 24:10]
$$oyul+i/u > oyli'/oylu$$
 son+3SG.POSS [Lev 19: 17] $g\ddot{o}\eta il+i > g\ddot{o}\eta li$ heart; mind+3SG.POSS

2.2.1.5. Consonant Deletion

Within our dataset, two types of consonantal deletion are evident. The first instance involves the deletion of the consonant in the agrist marker {-Ir} (refer to section 2.2.3.3.3). This occurs sporadically when the marker precedes the first-person singular and second-person plural markers.

The observed deletion could conceivably be ascribed to a scribal error. However, the fact that this particular form of consonantal deletion appears 21 times within our data suggests that it might extend beyond a mere isolated error. Thus, these instances could potentially be interpreted as evidence of a tendency towards consonantal deletion.

The second and the final example involves the combination of the relative pronouns $ne\check{c}ik$ 'how' and ki 'that; which', which appears as $ne\check{c}iki$ 'so as; so that' in our text (see for example Lev 4:10), whereas it appears as $ne\check{c}ik$ ki in other Crimean Karaim Bible translations (Sulimowicz 1972: 57; Jankowski 1997: 35; CrKB I: 178). It should be noted that the combination $ne\check{c}ik$ ki also appears in a Trakai Karaim Bible translation (Németh 2021: 622) and a Halich Karaim Bible translation (Olach 2013: 346), whereas it does not occur in Crimean Tatar⁵⁶ and Ottoman Turkish. In fact, this consonant deletion in our text might be attributed to an orthographical tendency where, in some instances, the text avoids using two kaph characters (5), representing the letter k, in a row (see 2.1.1.2.2.). Consequently, the status of this type of consonant deletion remains questionable.

-

⁵⁶ However, the word *nečik* alone appears in Crimean Tatar (KRUS: 392).

2.2.1.6. Metathesis

The phenomenon of metathesis in uninflected lexemes has been previously discussed (refer to section 2.1.2.2.2.7.). Notably our dataset presents only a single instance where this feature manifests due to the attachment of a marker.

(12) [Lev 19:27]
$$etraf + \ddot{i} + n > etarfin$$
 surroundings+3SG.POSS+ACC

2.2.1.7. The Alternation Between $y \sim n$

In specific instances, an alternation is observed between the dental n and velar y in various case, mood, and possessive markers. This fluctuation between these consonants has also been documented in Crimean Turkish by Doerfer (1959a: 272). Furthermore, the assimilation of the velar y by m or the dental n was noted to occur during the middle phases of Ottoman Turkish (Kerslake 2022: 178), which might have influenced similar phenomena in our dataset.

(13) [Lev 7:3 vs 25:24]
$$ver$$
-iniz vs ver -iniz to give-2PL.IMP
[Lev 26:21 vs 26:18] $yaz\ddot{\imath}klar+\ddot{\imath}\eta\ddot{\imath}z$ vs $yaz\ddot{\imath}klar+\ddot{\imath}n\ddot{\imath}z$ sins+2PL
[Lev 26:27 vs 26:23] $y\ddot{\imath}r\ddot{\imath}-s\ddot{a}-\eta iz$ vs $y\ddot{\imath}r\ddot{\imath}-s\ddot{a}-niz$ to walk-COND-2PL
[Lev 27:28 vs 27:22] $tutuvlu(k>y)+\ddot{\imath}+n\ddot{\imath}\eta$ vs $tutuvlu(k>y)+\ddot{\imath}+n\ddot{\imath}n$
 $possession+3SG.POSS+GEN$

It is important to highlight that the assimilation of velar η to dental n is infrequent in our data. One might presume that the letter gimel (\mathfrak{d}), representing velar η , and nun (\mathfrak{d}), signifying dental n, due to their visual similarity, could present a challenge to distinguish from each other, thereby attributing this fluctuation potentially to scribal error. However, the situation becomes clearer regarding the genitive case markers, which have a final $-\eta$ or -n. The Hebrew letter nun in the final position (\mathfrak{d}) differs from its initial and medial word form (\mathfrak{d}), thereby clearly distinguishing it from the letter gimel (\mathfrak{d}). This observation, in turn, suggests that the fluctuation cannot be explained solely due to orthographic difficulties.

2.2.1.8. The Alternation Between $z \sim s$

In our dataset, the negation in the aorist marker appears as $\{-\text{maz}\}$ and as $\{-\text{mas}\}$ (refer to section 2.2.3.1.1.2.), indicating a variation between -z and $-s^{57}$. It's worth noting that this

⁵⁷ Regarding the front counterpart, we solely encounter forms concluding in -z, e.g., [Lev 11:7] *gevšemez*; [Lev 26:20] *vermäz*.

alternation has been documented in both Crimean Turkish (Doerfer 1959a: 279) and Crimean Tatar (Doerfer 1959b: 385; Jankowski 2010: 78). In contrast, Crimean Karaim {-mAz} (Prik 1976: 138), Ottoman Turkish {-mAz} (Kissling 1960: 99–101), and Western Karaim {-mAs} (Musaev 1964: 285) each display only one of these forms.

(14) [Lev 7:18] sayïl-maz-Ø to be counted-NEG.AOR-3SG

[Lev 27:11] yuvuklaštïr-mas-lar to bring forward-NEG.AOR.-3PL

2.2.2. Non-Verbal Categories

2.2.2.1. Nominal Formation

In Turkic languages, nominal formation exhibits a pervasive characteristic that can be traced back to their early development stages. To demonstrate these shared characteristics and elucidate the morphophonological distinctions among the Turkic languages examined in this study, it would be beneficial to refer to the early stages of Turkic languages. Therefore, in this section, the situation of certain markers in earlier stages of Turkic will also be referred when applicable.

2.2.2.1.1. Suffixation

The derivation of nominal forms in Turkic languages involves the use of various morphological markers, which can be classified into two main categories: denominal markers and deverbal markers. Denominal markers derive nominal forms from existing nouns and adjectives, while deverbal markers derive nominals from verbs. Within each of these categories, there are many productive and unproductive suffixes that can be used to derive nominals. Our study will undertake an analysis of both productive and unproductive markers, including the investigation of already lexicalized items. The status of these markers will be determined based on the three categories listed in Zajączkowski (1932): productive, semi-productive, and unproductive⁵⁸.

2.2.2.1.1.1. Denominal Derivations of Nominals

This section presents the denominal markers that are used to create other nominals by attaching them to nominal bases within the corpus. Additionally, certain markers, such as the ordinal markers $\{+(I)n\check{J}\}$ and $\{+(U)n\check{J}\}$ (see 2.2.2.4.3), and the distributive marker $\{+(\check{S})Ar\}$ (see

-

⁵⁸ Zajączkowski's categorization is limited to Western Karaim. However, our investigation has not found any discrepancies in relation to his descriptions.

2.2.2.4.4), are discussed in separate sections, despite also functioning as denominal formative markers for nominals.

A-) {+(**A**)**K**}: This unproductive marker is found in a limited number of examples within our corpus, and its origins can be traced back to earlier periods of Turkic languages where it originally functioned as a kind of diminutive marker to indicate a smaller version of the base object in some examples (Erdal 1991a: 40–44).

(15) [Lev 8:7]
$$g\ddot{o}lmek > k\ddot{o}\eta\ddot{u}l + ek$$
 chest+DER 'shirt'.
[Lev 19:9] $ba\ddot{s} + ak$ head+DER 'ear of grain⁵⁹'.

The marker has been identified in Crimean Tatar (Jankowski 2010: 219) and all three dialects of Karaim, as reported in studies by Prik (1976: 56), Zajączkowski (1932: 17–18), and Musaev (1964: 177). Nouns containing this unproductive marker have also been observed in Ottoman Turkish, e.g., *bašak* 'ear' and *gömlek* 'shirt' (TLO I: 664; II: 4093).

B-) {+**An**}: Zajączkowski (1932: 18–19) posits that the marker under scrutiny originally served as a diminutive marker but later lost its diminutive function and no longer conveyed a distinct meaning from the base word. His examples that contain this unproductive marker with an etymological basis can be traced back to Crimean Tatar, Ottoman Turkish, all three dialects of Karaim, and our corpus, e.g., *oylan* 'son', *özen* 'river'. However, it is worth noting that this marker is frequently cited as indicating collectivity in such words (Róna-Tas 2022a: 65; Erdal 1991a: 91–92). Alongside the aforementioned lexemes, our corpus features one example where the marker is employed to create an adjective, potentially departing from Zajączkowski's description.

Notably, this particular instance has only been attested in Crimean Karaim (KRPS: 482; CKED: 365), and not in the other languages mentioned, such as Crimean Tatar, Ottoman Turkish, and Western Karaim dialects.

C-) {+(A)§}: In our study, this specific marker is identified in a single instance and appears a total of 13 times (see, for example, Lev 23:3) as *otura*§. It is crucial to emphasize that the word *otura*§ encompasses two separate meanings in Crimean Karaim, 'dwelling' and 'island'

⁵⁹ The initial recorded meaning of this word is 'iron head of an arrow or lance arrowhead' (Erdal 1991a: 40).

(CKED: 232), possibly stemming from the same etymological origins. Clauson (EDPT: 65) lists the word *otruy* 'island' and posits that it is more likely associated with the verb *otur*-, meaning 'to cut off', rather than the verb *otur*- meaning 'to sit'. This reasoning is grounded in the fact that *otur* 'to sit' is a relatively recent adaptation from the verb *olur*-. Notably, while the term 'island' is identified as *atay/aday* > *ada* in some Oghuz languages, it is represented as *otrač* in the Kuman language, potentially integrating the diminutive marker {+(X)č} (see Erdal 1991a: 44–47). It is plausible that the deverbal nominal formative marker was dissolved and integrated by the diminutive marker {+(X)č} after being succeeded by it. Upon scrutinizing the term *oturaš* in Karaim dialects, phonological disparities become evident, as demonstrated by C.Kar *oturaš/otraš* (CKED: 232), T.Kar *otrač* 'island' (KRPS: 434–435), and H.Kar *otrac*; *otlac* 'island' (KRPS: 434). As a result, this variation is likely a phonological manifestation of the marker {+(X)č} in Crimean Karaim. It is possible that the meaning 'dwelling' in Crimean Karaim may have been derived later due to the homonymous verb *otur*- meaning 'to sit'⁶⁰. Finally, it should be noted that neither the word *oturaš* or *otrač* nor the variant {+(A)š} marker appears in Crimean Tatar or Ottoman Turkish.

D-) $\{+\check{\mathbf{J}}\mathbf{A}\}$: The equative marker $\{+\check{\mathbf{J}}\mathbf{A}\}$ has been observed across various stages of Turkic languages, from early periods to modern forms (Johanson 2022a: 42). This marker predominantly attaches to nominals to produce adverbs (see 2.2.2.7.). Nevertheless, in a limited set of examples, it operates to derive adjectives from other adjectives.

(17) [Lev 13:19]
$$k\ddot{a}rm\ddot{a}\ddot{z}\ddot{z}+\ddot{z}a$$
 red+DER 'reddish'. [Lev 13:49] $y\ddot{a}\dot{s}il+\ddot{z}\ddot{a}$ green+DER 'greenish'.

The denominal derivational function of the {+ŠA} marker has been documented in all three Karaim dialects (Zajączkowski 1932: 43; Musaev 1964: 189–190; Çulha 2019: 141), as well as in Crimean Tatar (Kavitskaya 2010: 59) and Ottoman Turkish (Hagopian 1907: 76). Notably, Zajączkowski (1932: 54) characterizes this marker as productive.

E-) {+čA}: Erdal (2004: 177) emphasizes that the marker {+čA}, which serves as a diminutive marker, is not present in Old Turkic. Instead, it has been borrowed from Modern Persian into certain Turkic languages. Consequently, it should not be confused with the equative marker

_

⁶⁰ For an alternative explanation, see Zajączkowski 1932: 23.

 $\{+\check{\mathbf{J}}\mathbf{A}\}$. In the current dataset, there are instances where the marker functions to derive a noun from an adjective.

(18) [Lev 22:11]
$$ah+\check{c}a$$
 white+DER 'money'.

Furthermore, there are cases in which the aforementioned marker generates nouns from other nouns.

Notably, this unproductive marker has been identified in Crimean Tatar (Jankowski 2010: 236), Ottoman Turkish (Hagopian 1907: 76), and all three dialects of Karaim, as documented in studies by Zajączkowski (1932: 23–24) and Öztürk (2019: 56).

F-) {+dAš}: This derivational marker, denoting joint ownership or possession between two or more individuals, has been observed in Turkic languages since earlier periods (Erdal 1991a: 119–120). In all instances, it serves to generate nouns from other nouns.

This particular unproductive marker has been recorded in all three Karaim dialects (Zajączkowski 1932: 36–37; Musaev 1964: 108; Prik 1976: 55), as well as in Crimean Tatar (Kavitskaya 2010: 39) and Ottoman Turkish (Hagopian 1907: 78).

G-) {+II}; {+IU}: The markers {+II} and {+IU} are employed to derive adjectives from nominals and are prevalent in other Turkic languages, tracing back to the {+IXg} marker in earlier stages of Turkic languages (Erdal 1991a: 139–155). These quite productive markers generally convey possession or the material characteristics of an object.

(21) [Lev 1:9]
$$ot+lu$$
 fire+DER 'fiery'. [Lev 8:26] $yay+l\ddot{i}$ oil+DER 'oily'.

However, there are instances where the derived adjective signifies the origin or location of the subject or object.

(22) [Lev 16:29]
$$ver+li$$
 place+DER 'local'.

The note-worthy point to consider in our dataset is that adherence to rounding harmony was not uniform across all instances of adjective derivation using the {+II} and {+IU} markers.

Finally, it is important to mention that these markers have also been attested in all three dialects of Karaim, as well as in Crimean Tatar and Ottoman Turkish. Notably, except for Crimean Karaim (Prik 1976: 55; 81), the rest of the languages discussed in this study also present the rounded variant {+IU}⁶¹ (Hagopian 1907: 75; Zajączkowski 1932: 31–32; Musaev 1964: 172–174; Kavitskaya 2010: 55).

H-) {+IIK}; {+IUK}: These markers are frequently utilized and productive in Turkic languages, as highlighted by Erdal (1991a: 121–131). They are commonly employed to derive abstract nominal forms from other nominals. Within our dataset, we observe only two variations of these markers: {+IIK} and {+luk}.

In some examples, these markers are used to derive adjectives from nouns or vice versa.

As anticipated, this productive marker and its morphophonological variations have been attested in other Turkic languages, such as Crimean Tatar {+IIK}; {+IUK} (Kavitskaya 2010: 40; 55), Crimean Karaim {+IIK} (Prik 1976: 54; 81), Ottoman Turkish {+IIK}; {+IUK} (Hagopian 1907: 77–78), as well as in Western Karaim, including Trakai Karaim {+IIK}; {+IUK} and Halich Karaim {+IIK}; {+luk} (Zajączkowski 1932: 29–31; Musaev 1964: 104–107).

-

⁶¹ For obvious reasons (refer to 2.1.1.1.1.), Halich Karaim does not possess {+lü}.

I-) $\{+sA\}$: Zajączkowski (1932: 37–38) has listed⁶² the unproductive derivational marker $\{+sA\}$ for Western Karaim, with only one example provided. Similarly, our dataset features the same example where the marker appears as $\{+sa\}$.

It should be noted that this lexical item is listed in both Crimean Tatar (KRUS: 122) and Crimean Karaim (CKED: 94), but it does not appear in Ottoman Turkish⁶³.

J-) {+s**I**}: In our dataset, only two instances have been identified to contain the unproductive denominal marker {+sï}, which is observed to derive adjectives from nominals.

(27) [Lev 3:4]
$$artik+si$$
 surplus+DER 'extra'. [Lev 6:9] $ayruh+si$ holy+DER 'holy'.

The morpheme {+si} can be traced to its origins in Old Turkic, specifically {+sIg}, which functions to generate adjectives connoting similarity (Erdal 1991a: 70–73). This marker has been documented in both Ottoman Turkish and Western Karaim (Zajączkowski 1932: 39). Furthermore, the aforementioned lexemes incorporating this marker have also been identified in Crimean Karaim, e.g., *artiksi*, *ayruḥsi* (CKED: 49; 62). Nonetheless, sources pertaining to Crimean Tatar do not appear to reference this particular morpheme.

K-) {+s**Iz**}: The present marker has been identified in the earlier stages of Turkic languages (Erdal 1991a: 131–139) and serves to indicate a state of deficiency or absence. It is noteworthy that our corpus only includes a solitary instance of this productive marker.⁶⁴

(28) [Lev 12:2]
$$ar+siz$$
 shame+DER 'shameless'.

Lastly, the marker in question has been documented in all three Karaim dialects (Zajączkowski 1932: 32; Prik 1976: 81; Musaev 1977: 34), as well as in Crimean Tatar (Kavitskaya 2010: 55) and Ottoman Turkish (Hagopian 1907: 77).

_

⁶² Except for Karaim, he solely offers a few instances of this marker in Kazakh among the Turkic languages. Note that Erdal (1991a; 2004) does not list this marker for Old Turkic.

⁶³ The word appears as *boyunduruk* 'yoke' in Ottoman Turkish (TLO I: 951).

⁶⁴ Another example potentially containing the aforementioned marker has not been discussed due to its contested etymology. While most theories accept that the word *hirsiz* 'thief' incorporates the Turkish denominal marker {+sIz}, the noun stem's origin remains debated. Some scholars propose that it derives from the Arabic word *hayr* (NS: 361), while others argue that it stems from the Turkic *kir*, 'face' (Bozok 2018: 68–69).

L-) {+γï}; {+ki}: Within the current dataset, the denominal adjectival formative markers {+γï} and {+ki} can be observed. These productive elements have been documented in Crimean Tatar (Kavitskaya 2010: 55), Ottoman Turkish (Zajączkowski 1932: 35), and across all three dialects of Karaim (Zajączkowski 1932: 33–35; Prik 1976: 81; Musaev 1977: 176–177).

(29) [Lev 4:21]
$$burun+y\ddot{i}$$
 before; nose+DER 'first'.

In certain cases, the marker {+ki} occurs following the locative marker {+DA}, which signifies the location of the subject or object under description. It is noteworthy that the {+DAki} marker is prevalent in a multitude of Turkic languages, encompassing those previously mentioned.

2.2.2.1.1.2. Deverbal Derivations of Nominals

This section aims to elucidate the presence of derivational markers within our dataset, which facilitate the generation of nominal forms from verb roots. Upon examination, it becomes evident that certain non-finite verb modifiers, particularly some verbal noun markers (refer to Section 2.2.1.7.) and participle markers (refer to Section 2.2.1.6.), possess the ability to nominalize verbs. As a result, these markers have been incorporated into this section. Furthermore, it should be noted that a number of prominent verbal noun markers, such as $\{-(U)v\}$, $\{-(I)\check{s}\}$, and $\{-(U)\check{s}\}$, serve exclusively as derivational markers within the analyzed texts. Consequently, these markers will not be subject to further scrutiny as verbal nouns.

A-) {-(**A**)**K**}: The marker {-(A)K} is a widely used but an unproductive morpheme in Turkic languages, including Crimean Tatar (Jankowski 2010: 221), Ottoman Turkish (Hagopian 1907: 214) and all three dialects of Karaim (Zajączkowski 1932: 61–62; Musaev 1964: 111–112⁶⁵; Prik 1976: 56). Its origin can be traced back to Old Turkic {-(g)Ak} (Erdal 1991a: 74–76).

(31) [Lev 15:9] bin-ek to mount-DER 'saddle; saddle beast'.

[Lev 23:10] or-ak to harvest-DER 'sickle; harvest'.

⁶⁵ It is important to acknowledge that Prik and Musaev group the {-(A)K} marker together with {-(I)K} and {-(U)K} markers, which are analyzed separately in both Zajączkowski's exposition and our research (see entry G).

B-) **{-(A)t}:** Several instances of these unproductive markers have been observed in a limited number of cases within our dataset. Zajączkowski (1932: 84–85) identified these markers and their variations in Western Karaim and verified their existence in Ottoman Turkish. Previous stages of Turkic languages also exhibited this marker, which was documented as **{-(U)t}** (Erdal 1991a: 308–316). However, it has not been listed in studies exclusively focused on Crimean Karaim.

(32) [Lev 26:25] $\ddot{o}l-\ddot{a}t^{66}$ to die-DER 'plague; fatal disease'.

C-) $\{-\check{c}\}$; $\{-(I)\check{c}\}$: The $\{-\check{c}\}$ and $\{-(I)\check{c}\}$ derivational markers are employed to form nominals from verb stems as productive markers, but they exhibit subtle differences in their attachment to such stems. Specifically, the $\{-\check{c}\}$ marker is typically affixed to verb stems that end in $\{-n\}$ and were previously documented as $\{-(X)n\check{c}\}$ in Old Turkic. The final $\{-n\}$ in the verb stem and the verbal noun marker $\{-(X)\check{s}\}$ underwent subsequent development, leading to the evolution of $[n\check{s}]$ to $[n\check{c}]$, as evidenced by the appearance of the $[n\check{c}]$ form (Erdal 2004: 152). Our dataset indicates a case of the $\{-i\check{c}\}$ marker affixed to a verb stem that lacks the final $\{-n\}$.

(33) [Lev 11:3] $ay(i)r-i\check{c} > ayri\check{c}$ to separate-DER 'cleft'. [Lev 26:34] $t\ddot{i}n-\check{c}$ to become silent+DER 'silent'.

This marker was documented in all three Karaim dialects (Zajączkowski 1932: 87–89; Musaev 1964: 113; Prik 1976: 56–57) and Crimean Tatar (Kavitskaya 2010: 42). Furthermore, nouns marked by this derivational marker can also be observed in Ottoman Turkish, e.g., *sevinč* 'joy; rejoice' (Zajączkowski 1932: 89).

D-) {**-čI**}: Our dataset reveals the presence of a sole instance in which the productive marker {-či} is affixed to a verb stem in order to produce an adjective. This phenomenon can be traced back to earlier phases of Turkic languages, where it was expressed as {-(X)nčU} (Erdal 1991a: 285–290).

(34) [Lev 20:25] *irän-či* to hate+DER 'disgusting; hateful'.

Zajączkowski (1932: 89–90) notes that the marker {-čI} occurs sparingly, typically with verbs that end in -m or -n to create nominals or verbal adjectives. This phenomenon has also been

⁶⁶ This lexeme is also attested in Crimean Karaim, as evidenced by its inclusion in CKED: 268 and KRPS: 439.

observed in Crimean Tatar (Kavitskaya 2010: 42), Ottoman Turkish (Zajączkowski 1932: 90), and all three dialects of Karaim (Musaev 1964: 113; Öztürk 2019: 49–50).

E-) {-dAǯI}; {-dIǯI}: The text presents the use of an unproductive marker, {-däǯi}, and its variant, {-diǯi}. According to Erdal (2004: 282), the marker {-dAčI} has been attested in earlier phases of Turkic languages.

(35) [Lev 20:6]
$$bil$$
- $d\ddot{a}\ddot{z}i$ to know-DER 'sorcerer'. [Lev 20:27] bil - $di\ddot{z}i$ to know-DER 'sorcerer'.

This example has also been documented for Crimean Karaim (CKED: 83). Furthermore, Zajączkowski (1932: 89) includes this marker for Western Karaim and cites the same example as presented in our text. However, it has not been listed for Ottoman Turkish or Crimean Tatar.

F-) **{-I}; {-U}:** The unproductive deverbal derivational markers **{-U}** and **{-I}** are utilized to create nouns and adjectives and are commonly observed in various Turkic languages, including Crimean Tatar (Kavitskaya 2010: 42), Ottoman Turkish (Hagopian 1907: 213), and all three dialects of Karaim⁶⁷ (Zajączkowski 1932: 103–105; Musaev 1964: 177–178; Prik 1976: 57). Erdal (1991a: 172–223) notes that these markers, which can be traced back to **{-(X)g}** in earlier stages of Turkic languages, are used as subjects for intransitive verbs and objects for transitive verbs.

G-) {-(I)K}; {-(U)K}: Two markers, namely {-(I)K} and {-(U)K}, have been identified in our data as productive derivational suffixes that are attached to verb roots to form nominals. These markers typically signal the object or patient that is affected by the action expressed by the verb root. The origin of these suffixes can be traced back to Old Turkic, where they were documented as {-(O)k} (Erdal 1991a: 224–261).

| (37) | [Lev 13:10] | šiš-ik | to swell-DER | 'swelling'. |
|------|-------------|--------|--------------|-------------|
| | [Lev 13:25] | küy-ük | to burn-DER | 'burn'. |

⁶⁷ For obvious reasons (see section 2.1.1.1.1.) Halich Karaim do not possess the marker {-ü}.

| [Lev 17:14] | ķarīš-īķ | to mix-DER | 'mixed'. |
|-------------|----------|--------------|-----------|
| [Lev 21:7] | boz-uķ | to break-DER | 'broken'. |
| [Lev 22:19] | kilä-k | to wish-DER | 'wish'. |

This type of deverbal derivation is present in the grammar of all three Karaim dialects (Zajączkowski 1932: 63; Musaev 1964: 111–112; Prik 1976: 57), as well as in Crimean Tatar (Kavitskaya 2010: 42) and Ottoman Turkish (Hagopian 1907: 214). Among the aforementioned languages, it is noteworthy that the rounded variants of this suffix have not been documented in Crimean Tatar and Crimean Karaim. The outcome of our research yields yet another instance of rounded variants of a suffix that has not been accounted for in Prik's grammar of Crimean Karaim.

H-) {-(I)m}; {-(U)m}: By attaching to verb stems, this marker serves to generate nominals.

| (38) | [Lev 1:6] | buv-um | to choke-DER | 'joint'. |
|------|-------------------|----------------|---------------|---------------|
| | [Lev 8:11] | taķ-ïm | to attach-DER | 'tool'. |
| | [Lev 13:3; 13:34] | gör-üm; gör-im | to see-DER | 'appearance'. |

This unproductive particular suffix has been traced back to the early stages of Turkic languages, as documented by Erdal (1991a: 290–300). Its presence has also been attested in all three Karaim dialects (Zajączkowski 1932: 57–58; Musaev 1964: 113; Prik 1976: 57), as well as in Crimean Tatar (Kavitskaya 2010: 41) and Ottoman Turkish (Hagopian 1907: 213). It is worth noting that the rounded variant {-(U)m} has not been included in the aforementioned languages of Crimean Tatar and Crimean Karaim⁶⁹.

I-) **{-(I)n}:** Another unproductive marker, {-(I)n}, has been recorded in Crimean Tatar (Kavitskaya 2010: 43) as well as in all three dialects of Karaim (Zajączkowski 1932: 78–79; Musaev 1964: 178; Öztürk 2019: 50–51), along with Ottoman Turkish (Zajączkowski 1932: 78). It has also been documented in earlier stages of Turkic languages as {-(X)n} (Erdal 1991a: 300–308). This marker is used to derive nouns and adjectives from verbs.

(39) [Lev 4:13] *yašir-in* to hide-DER 'hidden'.

⁶⁸ The marker {-ük} is also absent in Halich Karaim for reasons that are evident (refer to 2.1.1.1.1.).

⁶⁹ For obvious reasons (see 2.1.1.1.1.), the Halich Karaim dialect does not feature the rounded variant {-üm}.

89

[Lev 11:37] ek-in to sow-DER 'crop'.

J-) {-(I)š}; {-(U)š}: In our dataset, the semi-productive markers {-(I)š} and {-(U)š} are added to verb stems to create nouns, and have been recorded as {-(X)š} since earlier stages of Turkic languages, as documented by Erdal (1991a: 262–275).

(40) [Lev 1:16]
$$doy$$
- $u\check{s}$ to be born+DER 'birth⁷⁰'. [Lev 23:17] $\ddot{u}le$ - $\check{s} > \ddot{u}l\ddot{u}\check{s}$ to share+DER 'share'.

This particular usage of the verbal noun marker has also been documented across all three Karaim dialects (Zajączkowski 1932: 58–59; Musaev 1964: 112–113; Prik 1976: 57), as well as in Crimean Tatar (Kavitskaya 2010: 42) and Ottoman Turkish (Hagopian 1907: 214).

K-) **{-(I)včI}; {-(u)vči}:** In our dataset, certain verbal adjectives formed with the productive markers {-(I)včI} and {-(u)vči} can also function as nominals. These markers, along with their variations, and their nominalizing functions are attested in Crimean Tatar, Ottoman Turkish, and all three dialects of Karaim (see 2.2.3.6.). In most cases, these markers can serve as both nominalizers and verbal adjectives simultaneously.

| (41) | [Lev 17:14] | aša-včï+lar | to eat-PART+PL |
|------|-------------|---------------|--|
| | | | 'the ones who eat; eaters'. |
| | [Lev 18:25] | otur-ïvčï+lar | to sit-PART+PL |
| | | | 'the ones who dwell; dwellers'. |
| | [Lev 24:18] | ur-uvčï | to strike-PART |
| | | | 'the one who strikes (kills); beater; killer'. |

L-) {-mA}: The verbal noun marker {-mA} (refer to 2.2.1.7.) additionally serves as a semi-productive derivational device, a characteristic frequently observed in Turkic languages (Erdal 1991a: 319–320) such as Crimean Tatar (Kavitskaya 2010: 41), Ottoman Turkish (Hagopian 1907: 213), and all three dialects of Karaim (Zajączkowski 1932: 92–93; Musaev 1964: 114; Prik 1976: 55–56). In our examples, it derives both nouns and adjectives.

(42) [Lev 2:2] tütid-mä to burn incense-DER 'incense'.

⁷⁰ The instance in question is part of a compound noun, as seen in Lev 1:16 gün doyušī 'east', where the term refers to 'day birth' in a literal sense.

[Lev 26:16] *kïzdir-ma*

to heat-DER

'fever'.

M-) **{-mAK}:** The infitive marker **{-mAK}** also generates nominals from verb stems, typically deriving entities associated with the base verb, a feature observable in the early phases of Turkic languages (Erdal 1991a: 110).

(43) [Lev 21:5] *čız-mak* to draw a line-DER 'line'.

[Lev 22:4] *tök-mäk* to spill-DER 'drop'.

This productive marker has been documented for all three Karaim dialects (Zajączkowski 1932: 95–96; Musaev 1964: 109–110; Prik 1976: 56), Crimean Tatar (Johanson 2010: 221) and Ottoman Turkish (Zajączkowski 1932: 96).

N-) {-mAn}: There exists only one example⁷¹ where the unproductive marker {-man} is attached to a verb stem and derives a noun in our data.

(44) [Lev 15:3] ak-man to flow-DER 'discharge'.

The derivational marker {-mAn} that forms nouns from verbs has been attested in earlier phases of Turkic languages (Erdal 1991a: 387–389), as well as in Ottoman Turkish (Zajączkowski 1932: 27) and all three Karaim dialects (Zajączkowski 1932: 26–27; Öztürk 2019: 53). In addition it is also possible to find words having this derivational marker in Crimean Tatar, e.g., *šiš-man*; to swell up+DER 'fat' (KRUS: 758).

O-) {-QAn}: In addition to its use as a participle marker (as discussed in section 2.2.3.6), {-QAn} is also a productive marker for deriving nominals from verbs. This marker is commonly used in many Turkic languages, including Crimean Tatar (Kavitskaya 2010: 76), Ottoman Turkish (Zajączkowski 1932: 80), and all three Karaim dialects (Zajączkowski 1932: 79–81; Öztürk 2019: 63).

(45) [Lev 2:14] *čirkin-gän* to scorch-PART 'scorched'.

[Lev 27:33] degiš-kän to change-PART 'substitute'.

P-) **{-KI}:** The available data only provides a single instance of the unproductive deverbal derivation marker **{-ki}** which is used to form nouns from verb stems.

-

⁷¹ Another word recorded in our data, [Lev 5:4] *yaman* 'wrong, bad, evil' (CKED: 444), may also contain the {-mAn} derivational marker. According to Erdal (1991a: 388), the word *yaman* appears to be related to the words *yavïz* and *yavlak*, suggesting that it may have been derived from *yav-man*.

(46) [Lev 11:34] $i\check{c}$ -ki to drink-DER 'drink'.

The derivational marker {-gI} can be traced back to earlier stages of Turkic languages and is used to form nouns and adjectives that denote the object of transitive verbs and the subject of intransitive verbs, as noted by Erdal (1991a: 320–323). This marker is listed in all three dialects of Karaim (Zajączkowski 1932: 66–68; Musaev 1964: 112; Çulha 2019: 151), Crimean Tatar (Jankowski 2010: 221) and Ottoman Turkish (Zajączkowski 1932: 68).

R-) {-(U)v}: The final deverbal derivational marker that converts verbs into nouns in our dataset is denoted as {-(u)v}. This productive marker has been documented in Crimean Tatar (Kavitskaya 2010: 43) as well as all three dialects of Karaim (Zajączkowski 1932: 73–75; Musaev 1964: 177–178; Prik 1976: 57–58). Originally, it functions as a verbal noun marker; however, in our text, it is used solely as a derivative device. It is noteworthy that this marker has not been included in the list for Ottoman Turkish (Kerslake 2022: 187).

(47) [Lev 25:14]
$$sat-uv$$
 to sell-DER 'sale'.
[Lev 8:35] $sakla-v > saklov^{72}$ to keep-DER 'holding'.

2.2.2.1.2. Compounding

In numerous Turkic languages, a prevalent similarity can be observed in the formation of compound nouns. The primary structure frequently encountered involves the possessive pattern, which consists of 'noun noun+3SG.POSS'. While this pattern exhibits a close association with genitive constructions, it is important to note that the attributed element is solely capable of accepting plural suffixes, and cannot accommodate any other suffixes including the genitive case markers.

| (48) | [Lev 2:13] | šart | tuz+i | |
|------|------------|-----------|------------------|-------------------------|
| | | covenant | salt+3SG.POSS | 'salt of the covenant'. |
| | [Lev 4:7] | otyam+lar | tütsü+si | |
| | | spice+PL | incense+3SG.POSS | 'spice incense'. |
| | [Lev 25:9] | kipurim | gün+ü | |
| | | atonement | day+3SG.POSS | 'atonement day'. |

_

⁷² Please note that only the form *saklav* is recorded for Crimean Karaim in the dictionaries (CKED: 337; KRPS: 461). Additionally, as the example also demonstrates, the distinctive characteristic of this marker, which induces rounding of the final unrounded vowels in verb stems, has been previously discussed (see Section 2.2.1.1.).

An alternative complement structure can be observed, characterized by the 'noun+noun' form. In this arrangement, the attribute often signifies the category, gender, variety, or material of the head. As a consequence, the attribute may assume the role of an adjective or indeed be an adjective, while the head invariably functions as a noun. Consequently, these constructions may be referred to as descriptive phrases.

Besides the given instances, the text displays a range of compound nouns copied from the Hebrew Bible as well, which can also be found in numerous Karaim Bible translations.

It should be noted that it is also possible to identify compound constructions that contain three elements, with one of the elements typically being an adjective.

Finally, it is worth noting that the attribute within such constructions can also manifest as a participle.

It is essential to highlight that all of the aforementioned strategies are prevalent across various Turkic languages (Johanson 2022a: 44), encompassing Crimean Tatar (Kavitskaya 2010: 44; Johanson 2010: 223–224), Ottoman Turkish (Kissling 1960: 41–42), and all three dialects of Karaim (Musaev 1964: 101–102; Prik 1976: 53).

2.2.2.2. Nominal Predicative Markers

Nominal predicative markers are affixed to nouns, pronouns, or adjectives to form non-verbal sentences. Our available data provides a limited number of examples for such suffixes. With regard to the 1SG.PRD, our dataset has revealed the use of three markers: {+im}, {+mIn}, and {+men}.

It is noteworthy that the marker {+min} is used exclusively with 1SG.PRO *men*. Moreover, only one example illustrates the use of the 1SG.PRD marker {+min}, whereas {+men} is exemplified in only two instances.

According to our corpus, the third person singular present tense lacks any markers; nevertheless, in most cases nominals are affixed to the copula {+DIr} and {+dur}⁷³. The {+DIr} marker manifests in different forms, which depend on the harmony of the front vowel and whether the word ends in a voiced/voiceless consonant or vowel.

(55) [Lev 1:9] '
$$ola+Ø+dir$$
 burnt offering+3SG.PRD+COP 'It is a burnt offering'.

[Lev 4:24] $hatat+Ø+dir$ sin offering+3SG.PRD+COP 'It is a sin offering'.

[Lev 25:12] $yovel+Ø+dir$ jubilee+3SG.PRD+COP 'It is a jubilee'.

The corpus contains a total of three examples (for the same lexical item) that display the {+dur} marker and, consequently, exhibit rounding harmony.

(56) [Lev 11:29]
$$bu+\emptyset+dur...$$
 this+3SG.PRD+COP 'This is...'.

⁷³ However, there are some examples where the copula $\{+DIr\}$ does not occur, e.g., [Lev 1:5] $e\check{s}i(k>g)+i+n+d\ddot{a}+\emptyset$ door+3SG.POSS+PRNM+LOC+3SG.PRD 'that is by the door of...'.

The marker {+tIr} only occurs after words that end in a voiceless consonant, and this is contingent on fronting harmony.

(57) [Lev 18:17]
$$ir\ddot{a}n\check{c}ilik+ \not O + tir$$
 abomination +3SG.PRD+COP
'It is an abomination'.

[Lev 18:22] $ikrahlik+ \not O + tir$ abomination+3SG.PRD+COP
'It is an abomination'.

Pertaining to the 3PL.PRD, the corpus unveils a multitude of suffixes. Among these, the primary one exhibits the identical combination observed in the 3SG.PRD, specifically $\{+\not O+DIr\}$. Customarily, in such circumstances, the third person plural pronoun is employed to delineate the personal ending.

(58) [Lev 8:28]
$$milu'im+Ø+dir\ alar$$
 ordination offering+3PL.PRD+COP they
'They are ordination offering'.

[Lev 18:17] $yuvuk+Ø+tir\ alar$ relative+3PL.PRD+COP they
'They are relatives'.

The subsequent suffix manifests as {+lAr}, which additionally precedes the copula marker {+dIr}.

Furthermore, there are instances where the order of the 3PL.PRD {+lAr} and the COP marker {+DIr} is reversed.

(60) [Lev 23:2]
$$bular+d\ddot{r}r+lar...$$
 these+COP+3PL.PRD 'these are...'.

In order to elucidate the circumstances in other Turkic languages under investigation within this study, Table 53, presented subsequently, compiles the nominal predicative markers for Crimean Tatar (Kavitskaya 2010: 61), Crimean Turkish (Doerfer 1959a: 277), Ottoman Turkish (Kissling 1960: 64–65), the corpus, Crimean Karaim (Prik 1976: 63–64), along with Halich and Trakai Karaim (Musaev 1977: 25).

Table 53: Comparision of Nominal Predicative Markers

| | Cr.Tat | Cr.Tur | Ot.Tur | Lev of | Cr.Kar | H.Kar | T.Kar |
|-----|---------------------------|---------------------------|----------------|---------------|------------|---------------|---------------------|
| | | | | Göz. 1841 | | | |
| 1SG | {+(I)m} | {+Im}, | {+(y)Im}, | {+Im}, | {+Im}, | {+men}, | {+mIn} |
| | | $\{+Um\},$ | $\{+(y)Um\}$ | $\{+mIn\},$ | $\{+m\}$ | $\{+m\}$ | |
| | | $\{+mIn\},$ | | {+men} | | | |
| | | $\{+mUn\},$ | | | | | |
| | | $\{+mAn\}$ | | | | | |
| 2SG | {+sIŋ} | $\{+_{S}I\mathfrak{y}\},$ | {+sIn}, | - | {+sIŋ} | {+sen}, | {+sIn}, |
| | | $\{+sIn\},$ | $\{+sUn\}$ | | | {+s} | {+s} |
| | | $\{+sU\eta\},$ | | | | | |
| | | $\{+sUn\},$ | | | | | |
| | | $\{+sAn\}$ | | | | | |
| 3SG | {+Ø}, | {+Ø}, | {+Ø}, | {+DIr}, | {+Ø} | {+DIr}, | {+DIr}, |
| | $\{+DIr\}$ | $\{+DIr\}$ | $\{+DIr\},$ | {+dur} | | {+Dur}, | {+DUr}, |
| | | | $\{+DUr\}$ | | | {+DI}, | $\{+DI\}, \{+DU\},$ |
| | | | | | | $\{+Du\},$ | $\{+D\}$ |
| | | | | | | $\{+D\}$ | |
| 1PL | {+mIz} | {+Iz}, | $\{+(y)Iz\},$ | - | {+mIz} | {+bIz}, | {+bIz}, |
| | | $\{+Uz\},$ | $\{+(y)Uz\}$ | | | {+buz}, | $\{+bUz\},$ |
| | | $\{+\ddot{i}s\},$ | | | | $\{+lArbIz\}$ | $\{+lArbIz\}$ |
| | | $\{+\ddot{i}\dot{h}\}$ | | | | | |
| 2PL | $\{+sI(\mathfrak{g}I)z\}$ | {+sIŋIz}, | {+sInIz}, | - | {+sIz} | {+sIz}, | {+sIz}, |
| | | $\{+sInIs\},$ | $\{+sUnUz\}$ | | | $\{+suz\},$ | $\{+sUz\},$ |
| | | $\{+sInIz\},$ | | | | $\{+lArsIz\}$ | {+lArsIz} |
| | | $\{+sIs\}$ | | | | | |
| 3PL | {+Ø}, | {+lAr}, | {+Ø}, | {+DIr}, | {+Ø}, | {+DIrlAr}, | {+DIrlAr}, |
| | $\{+DIr\}$ | $\{+DIrlAr\}$ | $\{+DIrlAr\},$ | {+DIrlAr}, | $\{+lAr\}$ | $\{+DUrlAr\}$ | $\{+DUrlAr\}$ |
| | | | $\{+DUrlAr\}$ | $\{+lArDIr\}$ | | | |
| | | | | | | | |

As observed, the examples from the corpus bear a striking resemblance to the description provided for Crimean Turkish. Additionally, with the exception of the 1SG.PRD variant {+mIn} and {+men}, it also shares a significant similarity to the descriptions for Crimean Tatar, Ottoman Turkish, and Prik's description for Crimean Karaim. Although there are some similar forms found in Western Karaim, it is evident that the corpus does not exhibit certain specific Western Karaim markers.

In our text, negation is realized through the employment of particles, such as *dügül*, found in eight instances. This particle appears as *tuvul* in the Northern dialect of Crimean Tatar and

as *degil* in Eastern dialects (Kavitskaya 2010: 112–113). The particle *degil* is also commonly used in Ottoman Turkish (Hagopian 1907: 35), while Prik (1976: 65) lists *dügül* as the negation particle for Crimean Karaim. Additionally, Western Karaim presents several other phonological variants, e.g., T.Kar *tuvul*, H.Kar *tivil* (KRPS: 568; 524). Only in one example mentioned in Chapter 11, the Ottoman Turkish variant *degil* is used together with the copula {+dir}. As a result, it is noteworthy that the particle *dügül* predominates in usage, comprising 89% of the instances, while *degil* accounts for a mere 11% within our dataset.

(61) [Lev 13:4] terän dügül 'it is not deep'.

[Lev 11:4] ayrī tirnaķli degildir 'it is is not cloven-hoofed'.

In several infrequent occurrences, the examined data exhibits the employment of the definite past tense, wherein the clitic *e*- is sequentially succeeded by the definite past tense marker {-dI}.

(62) [Lev 9:1] edi ol sekizinži gündä 'it was on the eighth day'.

Besides, it merits mention that instances arise where nominals are utilized in conjunction with the conditional mood, exhibiting a comparable strategy in which the clitic e- or i- precedes the conditional marker $\{-sA\}$.

(63) [Lev 13:3] terän isä 'if it is deep'.

Notably, the utilization of the clitic *e*-, succeeded by tense markers or conditional mood markers for nominal predicatives, is discernible in various other Turkic languages. This includes Ottoman Turkish, wherein the clitic manifests as *i*- (Hagopian 1907: 35), as well as Crimean Karaim, Crimean Tatar, and Western Karaim, where the clitic is also presented as *e*-(Kavitskaya 2010: 61; Musaev 1964: 120–130; Prik 1976: 66).

Finally, our data also shows the usage of existential predications, e.g., *var* 'existing' (occurring once), and *yok* 'non-existing' (occurring seven times), which were utilized under nominal predicatives to indicate whether the subject possessed or lacked the property described by the predicative.

(64) [Lev 25:30] ol šäḥärdä ki var aŋar kala

Lit. 'it is in the city which there is to it wall⁷⁴.

-

⁷⁴ Probably a calque of B.Heb *bā* '*îr* '*ăšer lō hōmāh* lit. 'in city which to him wall'.

'it is in the walled city'.

[Lev 26:37] *kuvuvči yoktir* 'there is no one who drives away'.

Notably, the mentioned existential predications are also present in other Turkic languages, e.g., Cr.Tat *bar*, *yok* (Kavitskaya 2010: 33); Ottoman Turkish *var*, *yok* (Hagopian 1907: 36); C.Kar *bar*, *yok* (Prik 1976: 66); T.Kar *bar*, *yoh/yoht* (KRPS: 102; 250); H.Kar *bar*, *yoht* (KRPS: 102; 251). It is noteworthy that the spirantization of the initial *b*- in the word *bar* reflects the influence of Ottoman Turkish in our text (see section 2.1.2.2.1.).

2.2.2.3. Nouns

2.2.2.3.1. Inflectional Categories

2.2.2.3.1.1. Number

The plural is shown by the marker {+lAr}, which is very common among Turkic languages (Johanson 2022a: 33), including Crimean Turkish (Doerfer 1959a: 276), Crimean Tatar (Kavitskaya 2010: 35), Ottoman Turkish (Kerslake 2022: 181) and all three dialects of Karaim (Musaev 1964: 124; Prik 1976: 62).

In examining the utilization of the plural suffix within the corpus, certain attributes exhibit non-Turkic features, particularly prevalent in Karaim Bible translations. For example, in our dataset, words succeeding numerals predominantly bear the plural suffix. This trait is present in other Karaim Bible translations across three dialects and is generally associated with Slavonic and Hebrew influence (Olach 2013: 122–124; Berta and Csató 2022b: 325).

Another non-Turkic characteristic regarding the plural suffix is its attachment to a noun when preceded by an adverb of quantity.

⁷⁵ See B.Heb *šêšet yāmîm* lit. 'six days'.

⁷⁶ See B.Heb *kāl- yəmê*; lit. 'all days'.

It is worth noting that this particular characteristic in Crimean Karaim should be attributed to Biblical Hebrew influence. This is supported by our analysis, which indicates that the usage of non-characteristic plural suffixes is based on the original Biblical Hebrew structure. For example, if the Hebrew expression lacks any numeral for 'two' but employs a dual form of the noun, the nouns modified by the numeral *eki* for 'two' appear in the singular form in our data.

On the other hand, it is worth noting that Biblical Hebrew expressions containing the numeral 'two', which are conveyed through the use of plural nouns, were also expressed using plural nouns in Lev of Göz. 1841.

Furthermore, the corpus showcases four instances where the employment of the plural suffix exhibits an atypical feature in Turkic languages. Intriguingly, these unconventional uses of the plural suffix also display a parallelism with Biblical Hebrew. In two examples, the plural suffix is affixed to adjectives, denoting similarity to the base item.

(70) [Lev 14:37]
$$y \ddot{a} \dot{s} i l + l \ddot{a} r$$
 green+PL 'greenish'⁷⁸. [Lev 14:37] $k \ddot{a} r m \ddot{a} \ddot{z} \dot{r} + l a r$ red+PL 'reddish'⁷⁹.

In the remaining two examples, adjectives that function as adverbs within the sentence are observed to incorporate the plural suffix.

-

⁷⁷ See B.Heb *rabbîm yāmîm*; lit. 'many days'.

⁷⁸ See B.Heb *yəraqraqqōt* 'greenish (plural adjective)'.

⁷⁹ See B.Heb '*ădamdammōt*; lit. 'reddish (plural adjective)'.

It is worth noting that some examples, certain uncountable nouns are also affixed with the plural suffix to indicate an abundance of types. It is important to note that unlike the aforementioned usages, this feature exists in some other Turkic languages as well.

2.2.2.3.1.2. Possessive Markers

Lev of Göz.1841 introduces possessive markers for all persons, and the examples provided show similarities to many other Turkic languages where such markers are typically manifested only as morphophonological differences.

In our data, examples for 1SG.POSS markers are exclusively observed in words that end with a consonant. Among these, the marker {+Im} is affixed only to words that have an unrounded vowel in their final syllable.

(73) [Lev 15:31]
$$miškan+im$$
 tabernacle+1SG.POSS 'my tabernacle'.
[Lev 26:30] $kil\ddot{a}k+im > kil\ddot{a}gim$ desire+1SG.POSS 'my desire'.

Overall, there are only two examples in which the 1SG.POSS marker is affixed to a word with a rounded vowel in its final syllable. In these examples, the marker undergoes rounding harmony and appears as {+um}.

100

⁸⁰ See B.Heb *yāmutū* '*ărîrîm*; lit. 'they shall die heirless ones'.

⁸¹ See B.Heb təmîmōt; 'complete (in plural form)'.

The 2SG.POSS markers are observed as $\{+(I)\eta\}$ and $\{+(U)\eta\}$. It should be noted that rounding harmony was not always respected in the usage of these markers.

(75) [Lev 18:7]
$$ana+\eta$$
 mother+2SG.POSS 'your mother'.

[Lev 18:15] $kelin+i\eta$ daughter-in-law+2SG.POSS 'your daughter-in-law'.

[Lev 18:21] $kul+u\eta$ servant+2SG.POSS 'your servant'.

[Lev 18:21] $urluk+i\eta > urluyi\eta$ offspring+2SG.POSS 'your offspring'.

[Lev 27:4] $kiyas+i\eta$ value+2SG.POSS 'your valuation'.

The 3SG.POSS markers, based on both rounding and fronting harmony, are observed as {+I} and {+U} after words that end in consonants. However, it should be noted that rounding harmony was not always respected in all examples.

(76) [Lev 1:4]
$$kol+u$$
 hand+3SG.POSS 'his hand'.

[Lev 1:16] $y\ddot{u}n+\ddot{u}$ feather+3SG.POSS 'its feathers'.

[Lev 5:11] $korban+\ddot{v}$ offering+3SG.POSS 'his offering'.

[Lev 5:17] $\ddot{z}\ddot{u}r\ddot{u}m+\dot{v}$ iniquity +3SG.POSS 'his iniquity'.

[Lev 6:3] $et+\dot{v}$ flesh+3SG.POSS 'his flesh'.

The 3SG.POSS marker appears as {+sI} after words that end in vowels, and therefore it does not undergo rounding harmony.

(77) [Lev 7:8]
$$teri+si$$
 skin+3SG.POSS 'its skin'.
[Lev 14:57] $tora+si$ law+3SG.POSS 'its law'.

A singular occurrence exists within the corpus where the 1PL.POSS marker is discerned as {+(i)miz}.

(78) [Lev 25:20]
$$tah\ddot{i}l+\ddot{i}m\ddot{i}z$$
 crop+1PL.POSS 'our crop'.

The second person plural markers manifest as $\{+(I)\eta Iz\}$ and $\{+(U)\eta Iz\}^{82}$. In some examples, the connecting vowel in the suffixes undergoes rounding harmony.

Lastly, the corpus comprises several possessive markers for the 3PL form. As anticipated, some of the 3PL.POSS markers are congruent with their 3SG.POSS counterparts.

(80) [Lev 14:42]
$$or(u)n+\ddot{\imath}+na$$
 ol $ta\check{s}+lar+n\ddot{\imath}\eta$ place+3PL.POSS+DAT the stone+PL+GEN 'to the place of those stones'.

In alternative instances, the 3PL.POSS marker manifests as {+lArI}.

In our text, if the possessed item is plural, then the word is attached with the plural marker {+lAr} before the POSS marker. It should be noted that if the possessed item is plural and the possessor is in the 3SG or 3PL form, then it is marked with {+I}, making it dependent on the context to determine the number. In other personal conjugations, the pertinent markers ensue after the plural suffix {+lAr} within the word.

⁸² Within the corpus, a solitary instance exists where the voiced velar nasal $-\eta$ - of the marker is realized as a voiced dental -n-, as exemplified in Lev 26:18 *yaziklariniz* 'your sins'. This phenomenon may be ascribed to a scribal error, considering the close proximity of the letter nun [1], representing n, and the letter gimel [λ], signifying η .

Table 54 presented below compiles the possessive markers for Crimean Tatar (Kavitskaya 2010: 61), Ottoman Turkish (Kissling 1960: 29–37), the corpus, and all three dialects of Karaim (Prik 1976: 67; Musaev 1977: 26).

Table 54: *Possessive Markers*

| | Cr.Tat | Ot.Tur | Lev of Göz. 1841 | Cr.Kar | H.Kar | T.Kar |
|-----|-------------------|-----------------------------|-------------------------|-------------------|----------------|------------------------|
| 1SG | {+(I)m} | $\{+(I)m\},$ | $\{+(I)m\}, \{+(u)m\}$ | $\{+(I)m\}$ | $\{+(I)m\},$ | {+(I)m}, |
| | | $\{+(U)m\}$ | | | $\{+(u)m\}$ | $\{+(U)m\}$ |
| 2SG | $\{+(I)n\}$ | $\{+(I)n\}, \{(I)n\},$ | $\{+(I)\mathfrak{y}\},$ | {+(I)ŋ} | $\{+(I)n\},$ | $\{+(I)y\},$ |
| | | $\{+(U)n\}, \{+(U)n\}$ | $\{+(U)\mathfrak{y}\}$ | | $\{+(u)n\}$ | $\{+(U)y\}$ |
| 3SG | {+(s)I} | $\{+(s)I\},$ | $\{+(s)I\},$ | $\{+(s)I\}$ | $\{+(s)I\},$ | $\{+(s)I\}, \{+(s)U\}$ |
| | | $\{+(s)U\}$ | {+U} | | $\{+(s)u\}$ | |
| 1PL | $\{+(I)mIz\}$ | $\{+(I)mIz\},$ | {+(ï)mïz} | $\{+(I)mIz\}$ | $\{+(I)mIz\},$ | $\{+(I)mIz\},$ |
| | | $\{+(U)mUz\}$ | | | $\{+(u)muz\}$ | $\{+(U)mUz\}$ |
| 2PL | $\{+(I)\eta Iz\}$ | $\{+(I)nIz; +(I)\eta Iz\},$ | $\{+(I)\eta Iz\},$ | $\{+(I)\eta Iz\}$ | $\{+(I)nIz\},$ | $\{+(I)yIz\},$ |
| | | $\{+(U)nUz\},$ | $\{+(U)\eta Uz\}$ | | $\{+(u)nuz\}$ | $\{+(U)yUz\}$ |
| | | $\{+(U)\eta Uz\}$ | | | | |
| 3PL | {+lAr}, | $\{+lArI\}, \{+(s)I\},$ | $\{+lArI\}, \{+I\}$ | $\{+(s)I\}$ | {+lArI} | {+lArI} |
| | $\{+(s)I\}$ | $\{+(s)U\}$ | | | | |

As can be seen, all the listed languages exhibit similar possessive markers with some phonological variations. The main difference between our text and Prik's description of Crimean Karaim is the use of rounded variants of the aforementioned markers.

2.2.2.3.1.3. Cases

Our dataset includes six distinct grammatical cases: nominative, accusative, genitive, dative, locative, and ablative. This section aims to demonstrate and compare the case markers of these categories with those found in Crimean Tatar, Crimean Turkish, Ottoman Turkish, and all three dialects of Karaim dialects. It is important to note that various functions of these cases (see 2.3.1.1.1.), their constructions with postpositions (see section 2.3.1.3.2.), and their conjugations with pronouns (see section 2.2.2.5.) will be thoroughly investigated in the relevant sections.

It is worth noting that our corpus presents some archaic and unproductive case markers, including the Old Turkic orientational formative marker {+DXn} (Erdal 2004: 376), attested in our data as {+tIn}, which appears in expressions such as [Lev 14:41] *ičkärtin* 'inside; from inside' and [Lev 24:3] *čiḫartin* 'outside; from outside'. The Old Turkic directive marker {+(g)ArU} (Erdal 2004: 370–371) is also present in the corpus, attested as {+(g)ArI}, and

appears in lexical items such as [Lev 4:12/10:5] *čihari/čiyari* 'outside', [Lev 7:34/27:7] *yokari/yoyari* 'above; upwards', and [Lev 11:2] *üzeri* 'on; upon; on the surface of'. Besides, the Old Turkic directive-locative marker {+rA}, found in expressions such as [Lev 18:3] *ora* 'there' and [Lev 6:13/11:32] *soŋra/sora* 'after', is also present in the corpus. In addition, our data attests to the presence of the equative marker {+ŠA} in several examples. In most cases, it functions as a derivational device (as explained in section 2.2.2.1.1.1) or appears in already lexicalized nominals, such as [Lev 25:23] *anša* 'only'. There are also some examples where its semantic function remains intact, such as in cases of abstract accordance or quantitative approximation, as in [Lev 5:10] *töräčä* 'according to the regulation' and [Lev 1:6] *buvumlarinža* 'as many as its pieces'. It merits mention that these unproductive markers are present in several lexical items across numerous Turkic languages. Nevertheless, this section does not aim to examine these unproductive case markers in depth.

2.2.2.3.1.3.1. Nominative Case

Consistent with the pattern observed in other Turkic languages (Johanson 2022a: 34), the nominative case in our text is unmarked by a case ending.

(83) [Lev 14:4]
$$kohen+\emptyset$$
 priest+NOM 'priest'.
[Lev 27:26] $\ddot{o}g\ddot{u}z+\emptyset$ ox+NOM 'ox'.

2.2.2.3.1.3.2. Accusative Case

Throughout the book, the accusative case is expressed through three primary variants (along with their respective morphophonological variations), which exhibit limited instances of rounding harmony. One of these variants is marked by the {+nI} suffix.

| (84) | [Lev 4:4] | buya+nï | bull+ACC |
|------|------------|---------|------------|
| | [Lev 14:5] | ķuš+nï | bird+ACC |
| | [Lev 14:6] | ezöv+ni | hyssop+ACC |

The marker {+nI} is attested in several Kipchak languages, including the central and northern dialects of Crimean Tatar (Jankowski 2010: 208) and all three Karaim dialects (Musaev 1964: 153–154; Prik 1976: 71). Additionally, this marker has been recorded in Crimean Turkish

(Doerfer 1959a: 276). While rounding harmony with this marker is relatively rare in our dataset, there are four instances⁸³ where the suffix adopts the {+nu} form.

(85) [Lev 9:20]
$$yay+lar+nu$$
 oil+PL+ACC
[Lev 11:3] $o+(n)u$ 3SG.PRO+PRNM.ACC
[Lev 11:9; 11:21] $bu+(n)u$ DEM.PRO+PRNM.ACC

Another manifestation of the accusative marker is {+n}, which is exclusively observed following the 3SG.POSS and 3PL.POSS markers. In the given corpus, it serves as an alternative to the {+nI} marker.

(86) [Lev 1:12]
$$ba\check{s}+\ddot{\imath}+n$$
 head+3SG.POSS+ACC [Lev 1:15] $ba\check{s}+\ddot{\imath}+n\ddot{\imath}$ head+3SG.POSS+ACC

In the dataset under analysis, 430 instances feature an accusative marker appended to a 3SG.POSS or 3PL.POSS marker. Among these, 138 cases (32%) employ the {+n} marker, while the {+nI} suffix is used in 292 instances (68%). The distribution of these markers is inconsistent, with occurrences of both markers sometimes appearing within the same sentence. It is noteworthy that the {+n} accusative marker is a pervasive feature across various Kipchak languages, including Crimean Tatar (Kavitskaya 2010: 37), typically manifesting after the 3SG.POSS or 3PL.POSS markers. Prik (1976: 74) highlights that in such structures Trakai Karaim exhibits the {+nI} marker, while Halich Karaim and Crimean Karaim employ {+n}. In contrast, Ottoman Turkish exclusively utilizes the {+nI} marker (Kissling 1960: 40–41).

The {+(y)I} accusative marker, identified as the final marker in our analysis, occurs merely 11 times in Chapter 11 of the dataset. This marker is frequently considered the Oghuzic equivalent of the Kipchak {+nI} and has been documented in Crimean Turkish (Doerfer 1959a: 276), Ottoman Turkish (Kissling 1960: 25–27), and Crimean Tatar (Doerfer 1959b: 381). While Prik's description (1976) does not mention it, Çulha's account of Crimean Karaim (2019: 87) notes the presence of this variant in mejumas, which predominantly display Crimean Turkish characteristics. Consequently, only a small portion of the analyzed text exhibits Crimean/Ottoman Turkish features regarding this marker.

⁸³ The examples *onu* and *bunu* contain the pronominal suffix, commonly referred to as $\{+n\}$.

-

| [Lev 11:4; 11:13] | $bu+lar+\ddot{\imath}$ | this+PL+ACC |
|-------------------|------------------------|----------------------|
| [Lev 11:6] | tavušan+ï | rabbit+ACC |
| [Lev 11:7] | <u>h</u> ïnzïr+ï | pig+ACC |
| [Lev 11:15] | ķuzyun+ï | raven+ACC |
| [Lev 11:16] | duyan+ï | hawk+ACC |
| [Lev 11:19] | balïķčin+ï | heron+ACC |
| [Lev 11:42] | olar+ï | they+ACC |
| [Lev 11:43] | ǯan+lar+ïŋïz+ï | soul+PL+2PL.POSS+ACC |
| [Lev 11:45] | siz+i | 2PL.PRO+ACC |

2.2.2.3.1.3.3. Genitive Case

The genitive marker {+nIn} is the primary form observed in the Lev of the Göz. 1841, although it appears as {+nIn} in only two instances. Moreover, the marker appears as {+nun} once throughout the book, demonstrating the presence of rounding harmony.

| (88) | [Lev 1:2] | Yišra'el+niŋ | Israel+GEN |
|------|-------------|--------------|----------------|
| | [Lev 1:5] | sïyïr+nïŋ | catlle+GEN |
| | [Lev 11:3] | tïrnaķ+nïn | fingernail+GEN |
| | [Lev 11:10] | suv+nun | water+GEN |

We observe the occurrence of another genitive marker, represented by $\{+(n)In\}$, exclusively in Chapter 11 of our dataset. This marker is observed only four times.

| (89) | [Lev 11:2] | Yisra 'el+in | Israel+GEN |
|------|-------------|--------------|-------------|
| | [Lev 11:45] | Mïsïr+ïn | Egypt+GEN |
| | [Lev 11:47] | ḥaram+ïn | unclean+GEN |
| | [Lev 11:47] | helal+in | clean+GEN |

The markers {+nXŋ} and {+(n)Xn} are present in many Turkic languages (Johanson 2022a: 34). Generally, the form {+(n)Xn} is found in Oghuzic languages⁸⁴, while the {+nXŋ} marker is observed in Kipchak languages. In Ottoman Turkish, it appears as {+(n)Xn} (Kissling 1960: 22–24). Crimean Turkish exhibits both variants (Doerfer 1959a: 276), while Crimean Tatar features Oghuzic variants in its eastern dialect and Kipchak forms in its central and northern dialects (Doerfer 1959b: 381). This variation is also observed in Çulha's (2019: 86) description of Crimean Karaim. However, Prik's description (1976: 71–74) and Musaev's study (1977: 41) only document the expected Kipchak variants, such as C.Kar {+nIn}; T.Kar {+nIn}, {+nUn}; H.Kar {+nIn}, and {+nun}.

2.2.2.3.1.3.4. Dative Case

Within the scope of our dataset, the dative marker primarily manifests as {+QA}. Its prevalence is dictated by a harmony that is evident both at the level of vowels and consonants.

| (90) | [Lev 2:14] | ot+ķa | fire+DAT |
|------|-------------|----------|--------------|
| | [Lev 14:41] | yer+ge | place+DAT |
| | [Lev 18:23] | tuvar+ya | cattle+DAT |
| | [Lev 25:47] | yarip+kä | stranger+DAT |

Additionally, the suffix appears as {+nA}⁸⁵ after the 3SG.POSS and 3PL.POSS markers.

(91) [Lev 4:7]
$$tem\ddot{a}l+i+n\ddot{a}$$
 base+3SG.POSS+DAT [Lev 21:6] $Te\eta ri+l\ddot{a}ri+n\ddot{a}$ God+3PL.POSS+DAT

Following the 1SG.POSS and 2SG.POSS markers, the dative marker emerges as {+A}, which is a prevalent feature in both Oghuzic and Kipchak languages.

| (92) | [Lev 10:3] | yuvuķ+lar+ïm+a | near+PL+1SG.POSS+DAT |
|------|-------------|----------------|----------------------|
| | [Lev 25:14] | dost+uη+a | friend+2SG.POSS+DAT |

0

⁸⁴ In Oghuz languages, the GEN markers typically appear as $\{+In\}$ or $\{+Un\}$ after final consonants, and $\{+nIn\}$ or $\{+nUn\}$ after final vowels. However, there are some exceptional cases that are only found in certain lexical items in Oghuz languages. For instance, the word su, meaning 'water', has an oblique stem suy+ before suffixes with initial vowels (Ragagnin 2022: 247), resulting in forms such as Turkish su(y)+un water+GEN.

⁸⁵ However, there is one exceptional example found in the book where the {+(y)A} DAT marker occurs after the 3SG.POSS suffix (a feature that cannot be found in Oghuzic languages or Crimean Tatar), e.g., [Lev 11:39] $le\breve{s}+i+y\ddot{a}$ 'carcass+3SG.POSS+DAT'.

Apart from the forms previously mentioned, certain chapters within the book exhibit the Oghuz-type {+(y)A} marker. However, except for Chapter 11, the expected Kipchak forms are predominantly utilized throughout the book. More specifically, this Oghuzic variant is only observed 32 times across six different items, present in two different chapters⁸⁶.

| (93) | [Lev 6:16] | ateš+ä | fire+DAT |
|------|------------------|--------------|---------------|
| | [Lev 11:4; 11:5] | siz+ä; siz+e | 2PL.PRO+DAT |
| | [Lev 11:24] | aḫšam+a | evening+DAT |
| | [Lev 11:31] | olar+a | they+DAT |
| | [Lev 11:39] | ye-me+ye | to eat+VN+DAT |

It is worth noting that 31 out of the 32 examples of this phenomenon appear in Chapter 11, while Chapter 6 only presents one example, resulting in a ratio of 97% to 3%.

These Oghuzic-Kipchak variants are found together in Crimean Turkish (Doerfer 1959a: 276), Crimean Tatar (Doerfer 1959b: 381), and Crimean Karaim (Prik 1976: 71–74), whereas Western Karaim only displays the Kipchak variants (Musaev 1977: 32). Ottoman Turkish, as expected, only exhibits the Oghuzic variants (Kissling 1960: 24–25). Thus, with the exception of a few limited examples, the text features the Crimean Kipchak Karaim characteristic for this marker.

2.2.2.3.1.3.5. Locative Case

Within our dataset, the locative marker emerges as {+DA}, which is a prevalent feature in several Turkic languages, including Crimean Tatar (Kavitskaya 2010: 37), Ottoman Turkish (Kissling 1960: 27–28), and all three dialects of Karaim (Prik 1976: 71–74; Musaev 1977: 29–33). As anticipated, this marker also emerges in accordance with both vowel and consonant harmonies⁸⁷.

⁸⁶ In four examples that have a final -k, the expected DAT marker $\{+ke\}/\{+k\ddot{a}\}$ appears as $\{+e\}/\{+\ddot{a}\}$. This is likely a scribal error that occurs to avoid noting two kaph (5) characters that denote 'k' in a row. This idea is supported by comparing other Karaim translations and similar examples from the corpus (see 2.1.1.2.2.). Therefore, such examples do not consist of the Oghuzic DAT marker. For instance, [Lev 15:33] $erk\ddot{a}k+[k]e$ 'male+DAT'; [Lev 20:4] $\ddot{o}ld\ddot{u}r+m\ddot{a}+m\ddot{a}k+[k]\ddot{a}$ 'to kill+NEG+VN+DAT; [Lev 25:7] $kiyik+[k]\ddot{a}$ 'beast+DAT'; and [Lev 26:5] $s\ddot{a}c+m\ddot{a}k+[k]\ddot{a}$ 'to sow+VN+DAT'.

⁸⁷ However, there exist a few conflicting instances where the marker $\{+dA\}$ appears following words that end in voiced consonants, such as in Lev 6:23 $kode\check{s}+d\ddot{a}$ holy+LOC. Additionally, owing to the restricted context, the expected LOC marker $\{+te\}/\{+t\ddot{a}\}$ is not detected in our dataset.

It merits mention that if the word contains 3SG.POSS or 3PL.POSS markers, the pronominal *-n*-follows the word stem before LOC markers, which is a common feature in Turkic languages.

(95) [Lev 14:11]
$$e\check{s}i[k>g]+i+n+d\ddot{a}$$
 door+3SG.POSS+PRNM+LOC
[Lev 21:1] $ulus+lar\ddot{i}+n+da$ nation+3PL.POSS+PRNM+LOC

2.2.2.3.1.3.6. Ablative Case

In our data, we encounter the ablative case marker as {+DAn}, which is the expected ablative suffix in Turkic languages (Johanson 2022a: 34). This exact form is found in Crimean Tatar (Kavitskaya 2010: 37), Crimean Karaim (Çulha 2019: 90), Ottoman Turkish (Kissling 1960: 28–29), and all three dialects of Karaim (Prik 1976: 71–74; Musaev 1977: 33).

Analogous to the prior instances, the marker undergoes both vowel and consonant harmony when affixed to words⁸⁸.

| (96) | [Lev 1:14] | ķuš+tan | bird+ABL |
|------|------------|-----------|------------|
| | [Lev 7:25] | tuvar+dan | cattle+ABL |
| | [Lev 8:32] | ötmäk+tän | bread+ABL |
| | [Lev 13:4] | teri+dän | skin+ABL |

Once again, it is worth noting that if the last vowel of a word is part of the 3SG.POSS or 3PL.POSS marker, the pronominal -*n*- is attached to the word stem before the ablative marker {+dAn}.

| (97) | [Lev 11:11] | et+leri+n+den | meat+3PL.POSS+PRNM+ABL |
|------|-------------|---------------|------------------------|
| | [Lev 14:8] | čadïr+ï+n+dan | tent+3SG.POSS+PRNM+ABL |

_

⁸⁸ However, in our dataset, certain counter-examples exist wherein the marker {+dAn} is observed following words that terminate in voiceless consonants, e.g., [Lev 8:32] $et+d\ddot{a}n$ 'flesh+ABL'; [Lev 21:12] $mikda\check{s}+dan$ 'sanctuary+ABL'.

2.2.2.4. Numerals

In this section, numerals from throughout the book will be analyzed. The data presents a few limited instances of cardinal, fractional, ordinal, and distributive numerals.

2.2.2.4.1. Cardinal Numbers

The numeral system is a fundamental feature shared by Turkic languages, exhibiting slight variations in phonology. In our data, we have identified a total of 20 different cardinal numbers. Cardinal numbers are essential to denote quantity, and play a crucial role in various aspects of the language, including counting, measuring, and expressing mathematical concepts.

Table 55: Cardinal Numerals

| Numerals | Lev of Göz. | Ot.Tur | Cr.Tat | C.Kar | T.Kar | H.Kar |
|--------------|-----------------|-------------|-------------|-------------|---------------|--------------|
| | 1841 | | | | | |
| one | bir | bir | bir | bir | bir | bir |
| two | eki | iki | eki | eki | ėki | eki |
| three | üč | üč | üč | üč | üč | ic |
| four | dört | dört | dört | dört | ďorť | dert |
| five | beš | beš | beš | beš | b́eš́ | bes |
| six | altï | altï | altï | altï | altï | altï |
| seven | yedi | yedi | yedi | yedi | yedi | yedi |
| eight | sekiz | sekiz | sekiz | sekiz | śegiź | segiz |
| nine | toķuz/doķuz | doķuz | doķuz | doķīz | toyuz | toyuz |
| ten | on | on | on | on | on | on |
| twelve | on eki | on iki | on eki | on eki | on ėki | on eki |
| fifteen | on beš | on üč | on üč | on beš | on üč | on ic |
| twenty | yegirmi/yigirmi | yirmi | yigirmi | yigirmi | ėgirme/ėgirmi | egirmi |
| thirty | otuz | otuz | otuz | otuz | otuz | otuz |
| thirty-three | otuz üč | otuz üč | otuz üč | otuz üč | otuz üč | otuz ic |
| fourty-nine | ķïrķ toķuz | ķïrķ doķuz | ķīrķ doķuz | ķïrķ toķuz | ķïrk toyuz | ķïrk toyuz |
| fifty | elli | elli | elli | elli | ėĺli | enli |
| sixty | altmïš | altmïš | altmïš | altmïš | altïmïš | altïmïs |
| sixty-six | altmïš altï | altmïš altï | altmïš altï | altmïš altï | altïmïš altï | altïmïs altï |
| hundred | уüz | уйг | уüz | уüz | yuź | iz |

The forms in our corpus sometimes present certain Oghuzic-Kipchak phonological oppositions, e.g., *tokuz* vs *dokuz* 'nine', as the initial *t*- becomes voiced in Oghuz languages in certain words while it is preserved in Kipchak languages (see section 2.1.2.2.1.).

In addition to the numerals listed above, our data features the lexeme [Lev 26:8] *tümen*, which refers to 'a division of ten thousand' and is typically employed as a military term. This term has been borrowed and is attested in numerous Turkic languages, ranging from early varieties to modern forms (EDPT: 507), e.g., Cr.Tat, *tümen* (KRUS: 622); Ot.Tur *tümen* (TLO I: 1480); T.Kar, *tuman* (KRPS: 570); H.Kar *kimen*, *timen* (KRPS: 320; 528).

2.2.2.4.2. Fractional Numerals

In Prik's Crimean Karaim description (1976: 87), there are examples of fractional numerals, such as 1/3 expressed as $\ddot{u} \Breve{c} + in \Breve{s} i pay + \Breve{i}$ (three+ORD portion+3SG.POSS) and 4/6 as $d\ddot{o}rt$ kere $alt\ddot{i} + n \Breve{s} \ddot{i} pay + \Breve{i}$ (four times six+ORD portion+3SG.POSS). However, in our data, fractions are constructed by adding a locative marker $\{+DA\}$ to the denominator of the fraction, followed by the numerator and the word pay, meaning 'portion', which is then followed by a third person singular possessive marker.

(98) [Lev 14:21]
$$on+da$$
 $bir pay+\ddot{i}$ $\ddot{o}z\ddot{a}k$ ten+LOC one portion+3SG.POSS wheat flour 'one tenth portion of wheat flour'.

The aforementioned structure is also observed in other Turkic languages, such as Crimean Tatar (Kavitskaya 2010: 50) and Ottoman Turkish (Hagopian 1907: 94). Similarly, in Western Karaim, a comparable structure to Prik's description of Crimean Karaim is encountered, where the denominator includes an ordinal marker preceding a word meaning 'portion', such as *helek* or *üles*, followed by 3SG.POSS, e.g., T.Kar *üčuńču helegi* 'one third'; H.Kar *besinci helegi* 'one fifth' (Musaev 1964: 211). Note that Western Karaim Bible translations demonstrate alternative structures for expressing fractions as well⁸⁹.

The present corpus features another structure that is similar to what is encountered in genitive constructions (see Section 2.3.1.1.). However, this specific structure is not mentioned

⁸⁹ Typically, the denominator in these structures is the Biblical Hebrew fractional number יְשִׁרוֹן ('iśśārōwn] 'tenth part; tithe' in the nominal case. For example, JSul.III.01 (H.Kar.B) [Lev 14:21] 'isaron ezek unun birni and ADub.III.73 (T.Kar.B) bir 'isaron öźak both mean 'a tenth of wheat flour'.

in the descriptions of Crimean Tatar or Ottoman Turkish, and it is not present in Western Karaim Bible translations.

(99) [Lev 14:10]
$$\ddot{u}\check{c}$$
 $pay+\ddot{i}$ $on+n\ddot{i}\eta$ $\ddot{o}z\ddot{a}k$ three portion+3SG.POSS ten+GEN wheat flour 'three tenths of wheat flour' [Lev 23:13] eki $pay+\ddot{i}$ $on+n\ddot{i}\eta$ $\ddot{o}zek$ two portion ten+GEN wheat flour 'two tenth portions of wheat flour'

Note that, fractions in the corpus are also indicated by certain lexemes, e.g., [Lev 27:30] *ondasï* 'tithe; one tenth', [Lev 6:13] *yarïmïsï* 'half', [Lev 25:30] *tamam* 'full'.

2.2.2.4.3. Ordinal Numerals

In our corpus, the ordinal numeral markers appear as $\{+(I)n\S I\}$ or $\{+(U)n\S I\}$ attached to the numeral stem. This feature can be traced back to early periods of Turkic languages (Johanson 2022c: 137). Similar patterns are also observed in Crimean Tatar with the marker $\{+(I)n\S U\}$ (Kavitskaya 2010: 51), Crimean Karaim with $\{+(I)n\S I\}$ (Prik 1976: 89), Ottoman Turkish with $\{+(I)n\S I\}$ and $\{+(U)n\S U\}$ (Kissling 1960: 61), Trakai Karaim with $\{+(I)n\check I\}$ and $\{+(U)n\S U\}$, and Halich Karaim with $\{+(I)ncI\}$ and $\{+(U)ncU\}$ (Musaev 1977: 39).

When the final syllable of a numeral in our corpus has a rounded vowel, the ordinal marker takes the form of {+UnžI}, with one exception in Lev 25:22.

| (100) | [Lev 5:11] | on+unǯï | ten+ORD | 'tenth'. |
|-------|-------------|---------------|-----------|-----------|
| | [Lev 7:17] | üč+ünži | three+ORD | 'third'. |
| | [Lev 19:24] | dör(t>d)+ünǯi | four+ORD | 'fourth'. |
| | [Lev 25:22] | doķuz+ïnžï | nine+ORD | 'ninth'. |

The marker, represented by {+inži}, appears after numerals that have a final consonant and a front, unrounded vowel in their last syllables.

(101) [Lev 5:24]
$$be\check{s}+in\check{\jmath}i$$
 five+ORD 'fifth'. [Lev 12:3] $sekiz+in\check{\jmath}i$ eight+ORD 'eighth'.

Nonetheless, when the numeral concludes with a vowel, the suffix {+nǯi} is employed:

| (102) | [Lev 8:22] | eki+nži | two+ORD | 'second'. |
|-------|-------------|----------|-----------|-------------|
| | [Lev 13:6] | yedi+nǯi | seven+ORD | 'seventh'. |
| | [Lev 25:11] | elli+nǯi | fifty+ORD | 'fiftieth'. |
| | [Lev 25:21] | altï+nǯï | six+ORD | 'sixth'. |

In our corpus, we have identified two adjectives that express the meaning of 'first'. One of these lexemes employs the $\{+\gamma\bar{\imath}\}$ marker (as discussed in section 2.2.1.1.1.) to form the adjective 'first' from the postposition *burun*, which means 'before'. For instance, [Lev 23:5] features the word *burunyi*, which appears five times in our data. Another adjective that conveys the meaning of 'initial' or 'first' is *ilk*, which has been attested in earlier stages of Turkic languages (Johanson 2022c: 137). It is important to note that the expected counterpart, *birinži*, meaning 'first', is not present in our data. The word *burunyi* and its similar variants are not found in Ottoman Turkish, but they are present in Crimean Tatar (KRUS: 127) and all three dialects of Karaim (KRPS: 141). Conversely, *ilk* is present in Ottoman Turkish (TLO I: 393), Crimean Tatar (KRUS: 192), and all three dialects of Karaim (KRPS: 199).

2.2.2.4.4. Distributive Numerals

The corpus provides an example of a lexeme attached with a distributive marker, and since this particular lexeme ends in a consonant, the suffix takes the form of {+er}. This suffix goes back to the typical Turkic distributive marker {+(š)Ar} (Johanson 2022a: 35), which is also attested in other Turkic languages, such as Crimean Tatar (Kavitskaya 2010: 51), Ottoman Turkish (Hagopian 1907: 96), Trakai Karaim (as {+(š)ArI}), and Halich Karaim (as {+(s)Ar}) (Musaev 1977: 39–40).

(103) [Lev 7:14]
$$bir+er$$
 one+DISTR 'one each'.

2.2.2.5. Pronouns

In this section, we will analyze the personal, demonstrative, reflexive, interrogative, relative and indefinite pronouns found in the Lev of the Göz. 1841.

2.2.2.5.1. Personal Pronouns

The corpus presents a limited number of personal pronouns examples, including those in all six cases, which are listed below.

Table 56: Personal Pronouns

| Cases | 1SG | 2SG | 3SG | 1PL | 2PL | 3PL |
|-------|---------|------|-----------|-----|------------|-----------------|
| NOM | ben/men | sen | ol/o | - | siz | olar/alar |
| GEN | menim | - | onïŋ/anïŋ | - | - | - |
| DAT | таŋа | saŋa | ona/aŋar | - | sizgä/size | alarya/olara |
| ACC | - | - | anï/onu | - | sizni/sizi | alarnï/olarï |
| LOC | - | - | anda | - | sizdä | alarda |
| ABL | - | - | andan | - | sizdän | alardan/olardan |

In our dataset, the personal pronouns in the nominative case have some phonological counterparts, including *ben* 'I' versus *men* 'id.', and *ol* 'she/he/it' versus *o* 'id.'. Nevertheless, the form *ben*, which is attested in Ottoman Turkish (Hagopian 1907: 47), appears only twice in the entire book (in Chapter 11). Conversely, the word *men*, which is common in Crimean Tatar (Kavitskaya 2010: 45) and all three dialects of Karaim (Prik 1976: 96; Musaev 1977: 41), is highly predominant, occurring 71 times (99% versus 1%). Furthermore, the third-person singular pronoun *ol*, documented in Western Karaim (Musaev 1977: 41), appears 127 times. In contrast, the form *o*, identified in Crimean Karaim (Prik 1976: 96), Crimean Tatar (Kavitskaya 2010: 45), and Ottoman Turkish (Hagopian 1907: 47), is observed a mere 12 times as a personal pronoun, accounting for 9% in comparison to the 91% prevalence of *ol*.

In addition to the previously mentioned counterparts, there exist further phonological alternations within personal pronouns, such as *olar* 'they' contrasted with *alar*. In our dataset, the third person plural pronoun *olar* appears solely 16 times, all in Chapter 11, while *alar* is encountered 133 times (11% as opposed to 89%). The form *olar* is documented for Crimean Tatar (Kavitskaya 2010: 45) and Crimean Karaim (Prik 1976: 96), whereas *alar* is recorded for Western Karaim (Musaev 1977: 41) and *onlar* for Ottoman Turkish (Hagopian 1907: 48).

In instances of the genitive case, the 1SG pronoun is accompanied by the suffix {+im}, as evidenced in Crimean Tatar (Kavitskaya 2010: 45), Crimean Karaim (Prik 1976: 96), and Ottoman Turkish (Hagopian 1907: 47). However, in Western Karaim (Musaev 1977: 41), it is found with the suffix {+in}. On the other hand, the 3SG pronoun marked with the genitive case suffix only appears once [Lev 19:17] as *oniŋ*, while it appears 34 times as *aniŋ* (1%-99%). Such

similar variants are also present in other Turkic languages, such as Crimean Tatar *oniŋ* (Kavitskaya 2010: 45), Crimean Karaim *aniŋ* (Prik 1976: 96), Ottoman Turkish *onuŋ* and *aniŋ* (Hagopian 1907: 47), and Western Karaim *aniŋ* (Musaev 1977: 41).

In instances of the dative case, our dataset presents various counterparts, such as 3SG.PRO+DAT as *aŋar* (64 times) and *ona* (5 times); 2PL.PRO+DAT as *sizgä* (43 times) and *sizä/size* (5+14 times); 3PL.PRO+DAT as *alarya* (33 times) and *olara* (once). Among these variations, the form *ona* is listed for both Crimean Tatar (Kavitskaya 2010: 45) and Ottoman Turkish (Hagopian 1907: 48) as *oŋa*, while *size* and *olara* are present in Ottoman Turkish (Hagopian 1907: 47–48; Kerslake 2022: 183). These Ottoman Turkish influences on personal pronouns are exclusively observed in Chapter 11. It is also worth mentioning that the non-Ottoman Turkish form *sizge* is attested in both Crimean Tatar (Kavitskaya 2010: 45) and all three dialects of Karaim, while *alarya* is documented solely for the three Karaim dialects (Prik 1976: 97; Musaev 1977: 41) among the languages cited. Lastly, the 3SG.PRO with dative case example *aŋar* is a specific form prevalent in Western Karaim (Musaev 1977: 41) as *anar* and can be traced back to earlier stages of Turkic languages (Johanson 2022c: 136).

In instances of the accusative case, the Kipchak forms *anï*, *sizni*, and *alarnï* are frequently observed (altogether 340 times). Among them, *sizni* is also present in Crimean Tatar (Kavitskaya 2010: 45), while Karaim dialects (Prik 1976: 97; Musaev 1977: 41) include all the mentioned examples. In the data at our disposal, we only find three instances (all in Chapter 11) where the accusative case forms *onu*, *sizi*, and *olarï* appear, which are typical of Ottoman Turkish (Hagopian 1907: 47–48; Kerslake 2022: 183).

Finally, another pair of counterparts can be observed between *olardan* '3PL+ABL' (3 times) and *alardan* 'id.' (9 times). As with the previous examples, an Ottoman Turkish form, *olardan* (Kerslake 2022: 183), appears only in Chapter 11 within our data.

2.2.2.5.2. Demonstrative Pronouns

The following are the demonstrative pronouns found in the Lev of the Göz. 1841. Many of the doublets are due to differences in the Kipchak and Oghuzic/Ottoman Turkish case suffixes.

Table 57: Demonstrative Pronouns

| Cases | Singular Demonstrative Pronouns | | | Plural Demonstrative Pronouns | |
|-------|---------------------------------|----------|------------|-------------------------------|-----------|
| _ | this | this one | that | these | those |
| NOM | bu | иšbи | ol/o | bular | alar/olar |
| GEN | типїŋ | - | onïŋ/ anïŋ | - | - |

| Cases | Singular Demonstrative Pronouns | | | Plural Demonstrative Pronouns | |
|-------|---------------------------------|----------|----------|-------------------------------|-----------------|
| | this | this one | that | these | those |
| DAT | buna | - | ona/anar | - | alarya/olara |
| | типа | | | | |
| ACC | bunï/bunu/munï | - | anï/onu | bularnï/bularï | alarnï/olarï |
| LOC | - | - | anda | - | alarda |
| ABL | - | - | andan | bulardan | alardan/olardan |

The provided examples demonstrate that when certain case markers are attached to the demonstrative pronoun bu, the initial b- becomes nasalized, as evident in $mun\ddot{v}$, $mu\eta a$, and $mun\ddot{v}$. It is significant to highlight that the forms featuring an initial b- are present in Ottoman Turkish (Hagopian 1907: 70–72), the eastern dialects of Crimean Tatar (Doerfer 1959b: 383), and documented for all three dialects of Karaim (Prik 1976: 99; Musaev 1977: 42). The nasalization of the initial b- to m- can be traced back to earlier periods of Turkic languages (Johanson 2022c: 136) and is also observed in some modern Turkic languages such as Turkmen, Kazakh, and Karakalpak (Karakoç 2022: 272; Abish 2022: 341).

It should be noted that the demonstrative pronoun ol/o is prevalently utilized in Karaim Bible translations as a result of the literal translation approach. This pronoun serves to represent the Biblical Hebrew definite article $\neg \neg \neg [ha-]$, notwithstanding the absence of a definite article in Turkic languages. This method is prevalent in the majority of Karaim Bible translations⁹⁰, as observed by Kowalski (1929: xxxix) and Pritsak (1959: 331).

According to our data, the demonstrative pronoun *ol* is used 1,256 times for this function, while the variant *o* is used only 55 times (96% vs 4%). This opposition is also observed in other Turkic languages, e.g., C.Tat, C.Kar, Ot.Tur, *o* 'that' (Kavitskaya 2010: 46; Prik 1976: 99–100; Hagopian 1907: 70–71) vs Western Karaim *ol* 'id.' (Musaev 1977: 41).

⁹⁰ According to Jankowski (2018: 55–56), the manuscript known as Evr. I 143 does not include the rendering of the Biblical Hebrew definite article. However, the language used in this manuscript is still a subject of debate.

2.2.2.5.3. Reflexive Pronouns

The data indicates that both $\ddot{o}z$ and gendi are used to express the concept of self, specifically reflexivity. These pronouns are typically accompanied by personal possessive markers to indicate the subject's reflexivity.

(105) [Lev 16:6]
$$gendi$$
 'self; herself; himself; itself; own'.
[Lev 9:7] $gendi+\eta$ self+2SG.POSS 'yourself'.
[Lev 23:14] $gendi+si$ self+3SG.POSS 'herself/himself/itself'.
[Lev 1:2] $gendi+\eta iz$ self+2PL.POSS 'yourselves'.
[Lev 16:22; Lev 9:2] $\ddot{o}z+i/\ddot{o}z+\ddot{u}$ self+3SG.POSS 'herself/himself/itself'.
[Lev 26:1] $\ddot{o}z\ddot{u}+\eta iz$ oneself+2PL.POSS 'yourselves'.

Kavitskaya (2010: 45) asserts that, within the Crimean Tatar, $\ddot{o}z$ represents a more neutral means of expressing reflexivity, while *kendi* is predominantly employed in the eastern dialect. In studies focused on Karaim, $\ddot{o}z$ is also frequently used as a reflexive pronoun (Prik 1976:105–106; Musaev 1977: 44), while *kendi* is more prevalent in Ottoman Turkish (Hagopian 1907: 48). As previously mentioned (see section 2.1.2.2.1.), the word *gendi* demonstrates an influence of Crimean Turkish, as it exhibits the voicing of the initial *k*- to *g*-. This phonetic change is not observed in Ottoman Turkish for this particular lexical item.

2.2.2.5.4. Interrogative Pronouns

The collected data exhibits the anticipated interrogative pronouns *ne*? which is widespread in many Turkic languages, including Crimean Tatar (Kavitskaya 2010: 48), Ottoman Turkish (Hagopian 1907: 82), and Karaim dialects (Prik 1976: 101; Musaev 1977: 42). It is noteworthy that interrogative pronouns in Turkic languages have the ability to take case and possessive markers, as stated by Johanson (2022c: 136).

Our dataset indicates that these pronouns are mainly used as relative pronouns, as evidenced by section 2.2.1.5.5. For instance, the interrogative pronouns *kim* 'who' and *nečik* 'how' only appear in a relative pronoun function within our dataset. It is worth noting that only two instances were identified in which the pronoun *ne* and its compound form with the postposition *učun* (i.e., *nučun*, signifying 'why') functioned as interrogative pronouns.

(106) [Lev 10:17] nučun aša-ma-dï-ŋïz

why eat-NEG-PAST-2PL 'why have you not eaten?'

[Lev 25:20] ne aša-lïm

what eat-1SG.VOL 'what shall we eat?'.

2.2.2.5.5. Relative Pronouns

As previously discussed, the interrogative pronouns *ne* 'what' and *kim* 'who', along with their inflected forms modified by case suffixes, can function as relative pronouns. In our dataset, these pronouns consistently precede another relative pronoun, *ki* 'which, that', which is of Persian origin and should not be confused with the homophonous Hebrew conjunction *ki* 'for, since, because' that frequently appears in our dataset. Moreover, the petrified form *nečiki*, meaning 'so as; so that', derived from the combination of the interrogative pronoun *nečik* 'how?' and the relative pronoun *ki*, also functions as a relative pronoun. It is essential to note that relative clauses in Turkic languages are typically based on participles, and structures incorporating relative pronouns are untypical (Johanson 2022a: 43). However this atypical Turkic characteristic can be seen in Crimean Tatar (Jankowski 2010: 225), Ottoman Turkish (Kerslake 2022: 192) and all three dialects of Karaim (Musaev 1964: 220–224; Jankowski 1997: 22–23) as well. For information on the structure of relative clauses introduced by these relative pronouns in our dataset, please refer to the relevant section (see section 2.3.2.2.2.2.).

| (107) | [Lev 4:20] | neči[k]ki | how that | 'so as; so that'. |
|-------|-------------|------------|---------------|-------------------|
| | [Lev 15:20] | ne ki | what that | 'what'. |
| | [Lev 14:30] | ne+dän ki | what+ABL that | 'from what'. |
| | [Lev 14:31] | ne+gä ki | what+DAT that | 'to what'. |
| | [Lev 10:17] | ne+ni ki | what+ACC that | 'what'. |
| | [Lev 4:23] | ki | which; that | 'which; that'. |
| | [Lev 15:32] | kim ki | who that | 'who'. |
| | [Lev 14:32] | kim+niŋ ki | who+GEN that | 'whose'. |

2.2.2.5.6. Indefinite Pronouns

The numeral one functions as an indefinite article⁹¹ in Turkic languages (Johanson 2022a: 33) which is utilized to establish indefinite phrases, e.g., [Lev 4:27] *bir ǯan* 'anyone' (lit. one soul'). Additionally, there are certain quantifiers that also create indefinite phrases in our data, such as *ǯümlä* 'all', [Lev 21:18] *ǯümlä kiši* 'every person'; *her/här* 'every', [Lev 7:6] *här erkäk* 'every man'; and *čoķ* 'many, much', [Lev 15:25] *čoķ günlär* 'many days'. Negative pronouns are formed by *hič* 'any', e.g., [Lev 17:12] *hič ǯan* 'nobody'. Equivalent quantifiers can also be found in Crimean Tatar (Kavitskaya 2010: 46–47), Ottoman Turkish (Hagopian 1907: 84–86), and all three dialects of Karaim (Öztürk 2019: 77–79).

Our data reveals the existence of various indefinite pronouns. In most cases, the numeral *bir* meaning 'one', and its various forms which, are followed by a 3SG.POSS marker once or twice, is used as an indefinite pronoun. This is also common in all three dialects of Karaim

In the majority of other cases, the formation of indefinite pronouns in our data is primarily based on interrogative pronouns, which may be followed by some enclitics.

The rest of the indefintie prouns as follows:

(110) [Lev 11:32] her šey 'everything'.

[Lev 19:11] karšīdaki 'the other person'; lit. 'the one opposite'.

-

⁹¹ In our data, it is worth noting that nouns often appear without the indefinite article bir, serving solely to refer to indefinite phrases. This phenomenon is particularly common in Karaim Bible translations, which follow the structure of Biblical Hebrew. For example, in Lev [22:14] da kiši ki ašasa (see same structure in B. Heb אָליל (ישׁכִּל), the word kiši 'person' refers to an indefinite person, resulting in the sentence translating to 'if a person eats'.

2.2.2.6. Adjectives

Adjectives can serve both attributive and predicative functions in Turkic languages as well as in our data.

(111) [Lev 16:24]
$$ayruhs\ddot{i} yer$$
 'holy place'. [Lev 19:2] $ayruhs\ddot{i}+d\ddot{i}r+men$ 'I am Holy'.

Clear differentiation between adjectives and nouns in our data is challenging due to the common practice in Turkic languages, as noted by Johanson (2022a: 34).

A similar ambiguity can exist between some adjectives and adverbs, as illustrated by the example below, where the adjective *koyu* meaning 'thick', functions as an adverb.

As previously indicated (see 2.2.2.1.1.1.), specific markers may be affixed nominals to create adjectives. In addition, participles, i.e. verbal adjectives are addressed separately (see 2.2.3.6.).

2.2.2.7. Adverbs

Our text presents a variety of adverbs, including those that indicate time, such as [Lev 11:24] ahšam 'evening' and [Lev 19:13] ertä 'morning'; manner, for instance, [Lev 24:19] alay 'in that way' and [Lev 7:21] gizli 'secretly'; spatial and directional references, exemplified by [Lev 6:4] čihari 'outside' and [Lev 18:3] anda 'there'; and degree, including [Lev 25:16] az 'little, few' and čoķ 'many; much'. The text also contains conjunctive adverbs that provide supplementary or complementary information to the previous clause or sentence, such as [Lev 25:45] dayin 'also'. Comparable categories and adverbs can be found in Crimean Tatar (Kavitskaya 2010: 57–58), Ottoman Turkish (Kissling 1960: 130–131), and all three dialects of Karaim (Musaev 1964: 188; Prik 1976: 92; Öztürk 2019: 80–86), as evidenced by the cited scholarly sources. As elucidated in the prior section (see section 2.2.2.6), most adjectives and other nominals can serve as adverbs without overt marking. However, certain suffixes are employed to generate adverbs from nominals. Among these, unproductive and productive case markers (see 2.2.2.1.1.1.) are found to be utilized in attaching nominals to form adverbs, as demonstrated in our data and commonly observed in Turkic languages (Johanson 2022a: 41).

(114) [Lev 10:18]
$$i\check{c}+k\ddot{a}ri$$
 inside+DIR 'into; inside'.
[Lev 20:4] $a\check{c}i\dot{k}+tan$ open+ABL 'openly'.
[Lev 25:30] $doly\ddot{i}n+\check{z}a$ full+DER 'fully'.

Notably in other examples nominals with postpositions also function as adverb.

In a specific adverb, (occurring a total of 15 times within our dataset), we encounter a common strategy of reduplication, a phenomenon frequently observed in many Turkic languages, which is prevalent in both adjectives and adverbs. The technique involves the preposition of a reduplicated first syllable. If such a syllable ends with a consonant, the final consonant is subsequently replaced with an m, n, r, or s. On the other hand, in instances where the syllable ends with a vowel, these aforementioned consonants are directly appended to the reduplicated segment that precedes the word stem (Johanson 2022a: 34).

Finally, with regard to verbal adverbs, otherwise known as converbs, a distinct analysis will be conducted, as discussed in Section 2.2.3.5. For a detailed examination of the diverse structures that shape adverbial clauses, please refer to section 2.3.2.2.2.2.

2.2.3. Verbal Categories

2.2.3.1. Verb Formation

Similar to the Nominal Formation section (2.1.1.1.), the current section aims to demonstrate and compare the derivational markers of our dataset with those found in Crimean Tatar, Ottoman Turkish, and all three dialects of Karaim. Moreover, we will refer to the early phases of Turkic languages to present the common shared characteristics and distinctions of the Turkic languages.

2.2.3.1.1. Suffixation

The process of verb derivation in Turkic languages involves the utilization of diverse morphological markers, which can be categorized into two principal groups: denominal markers and deverbal markers. Denominal markers are employed to create verbs from nominals, while deverbal markers produce verbs from existing verbs. Each of these categories comprises a multitude of productive and unproductive suffixes that can be utilized for deriving verbs. Our study will conduct an analysis of both productive and unproductive markers, encompassing the investigation of already lexicalized items. The status of these markers will be ascertained on the basis of the three categories outlined in Zajączkowski's (1932) framework: productive, semi-productive, and unproductive⁹².

2.2.3.1.1.1. Denominal Verb Formative Markers

A-) {+**A**}: The denominal verb-forming marker {+A} is frequently observed in Turkic languages, dating back to their early developmental stages (Erdal 1991b: 425–429). This semi-productive marker manifests in Crimean Tatar (Kavitskaya 2010: 78), Ottoman Turkish (Zajączkowski 1932: 130), and all three dialects of Karaim (Zajączkowski 1932: 130–132; Musaev 1964: 236; Prik 1976: 110). However, in the dataset under examination, only the marker {-a} appears to serve this specific function.

(117) [Lev 23:15]
$$san+a$$
 number+DER 'to count'.
[Lev 21:22] $a\check{s}+a$ food+DER 'to eat'.

B-) {+Ay}: Another example of a semi-productive marker is {+(A)y}, which in earlier stages takes the form of {+(A)d} (Erdal 1991b: 485–492), but is present in Crimean Tatar as {+(A)y} (Jankowski 2010: 119), Ottoman Turkish (Zajączkowski 1932: 138), and all three dialects of Karaim (Zajączkowski 1932: 136–137; Musaev 1964: 237–238; Prik 1976: 110). Our text presents only one instance of this marker.

(118) [Lev 26:9] $k\ddot{o}(p>b)+\ddot{a}y$ - many; much+DER 'to multiply'.

C-) $\{+dA\}$: This unproductive marker is attested as a variant of the highly productive denominal verb-forming marker $\{+lA\}$ (see entry G), most likely to avoid using the sonorant l- in the marker, occasionally after some nominals having final coronal consonants, such as -t, -d, -s, -z, -l, -r, or -n. Notably, our data provides only a single example for this marker.

(119) [Lev 19:11]
$$al+da$$
 cheat⁹³+DER 'to cheat'.

-

⁹² As previously noted in section 2.2.1.1.1, Zajączkowski's categorization pertains to Western Karaim alone. Nonetheless, our analysis has not revealed any divergences in connection with his descriptions.

⁹³ The noun *al* is not documented in Karaim; however, it has been recorded in earlier stages of Turkic languages, initially signifying 'device' or 'method of doing something'. Over time, it developed negative connotations, such

It is worth noting that this marker is attested in a limited number of examples in Crimean Tatar (Jankowski 2010: 119) and Western Karaim (Zajączkowski 1932: 148–149). Furthermore, the same verb originally featuring this marker can also be found in Ottoman Turkish and Crimean Karaim, e.g., *alda*- 'to cheat' (TLO I: 1176; CKED: 38).

E-) {+**KIr**}: This semi-productive marker typically appears in an onomatopoeic word stem to form a verb and is commonly attested from the early stages of Turkic languages (Erdal 2004: 228). In our text, the {+kir} variant of this marker is found in a limited number of examples.

(120) [Lev 9:1]
$$\check{c}a^{94}+\check{k}\ddot{i}r$$
 onom.+DER 'to call'. [Lev 9:24] $\check{k}\ddot{i}\check{c}+\check{k}\ddot{i}r$ onom.+DER 'to shout; to cry'.

In Western Karaim, this marker has been documented (Zajączkowski 1932: 139–140; Musaev, 1964: 239). It should be noted that petrified verbs originally featuring this marker can also be found in various other Turkic languages, such as C.Tat čaķīr- (KRUS: 689), C.Kar čayīr-/čaķīr- (CKED: 109; 112), and Ot.Tur čayīr- (TLO I: 1552), which all mean 'to call'.

F-) **{+lA}**: One of the most productive denominal verb formative markers is **{+lA}**, which is observed in diverse Turkic languages and has the same form in the early stages of Turkic languages (Erdal 1991b: 429–455).

(121) [Lev 1:1]
$$s\ddot{o}z+l\ddot{a}$$
 word; speech+DER 'to speak'.
[Lev 9:22] $alyi\ddot{s}+la$ blessing+DER 'to bless'.

It is important to mention that the marker $\{+lA\}$ can be combined with certain voice markers (see section 2.2.3.2.), including passive $\{-n\}$, causative $\{-t\}$, reflexive $\{-n\}$, and reciprocal $\{-\check{s}\}$. Given that there are instances where the marker $\{+lA\}$ cannot function independently without a voice marker in the verb stem⁹⁵, scholars also catalog such combinations separately. However, in this study, we present the markers $\{+lAn\}$, $\{+lA\check{s}\}$, and $\{+lAt\}$ under this entry and do not explore them individually.

(122) [Lev 13:51]
$$s\ddot{i}z+la-t$$
 onom.+DER-CAUS 'to cause pain'.

-

as 'deceit', 'guile' or 'dirty trick'. Note that, this noun in the meaning of 'trick' is attested in various Oghuzic and Kipchak languages as well (EDPT: 120).

⁹⁴ Cf. Zajączkowski 1932: 139–140; EDPT: 403–404; NS: 162.

⁹⁵ For instance, the verb *yuvukla*- to close+DER does not appear in Karaim by itself, but only in combination with the reciprocal marker {-š} as *yuvuk+la-š*, to close+DER-RECP meaning 'to approach' (CKED: 475).

| [Lev 5:8] | yuvuķ+laš | close+[DER-RECP] | 'to approach'. |
|------------|------------|----------------------|--------------------|
| [Lev 8:4] | top+la-n | full+DER-REFL | 'to gather'. |
| [Lev 8:35] | sïmar+la-n | commandment+DER-PASS | 'to be commanded'. |

It is important to note that the denominal verb-forming marker {+lA} and its combinations with the mentioned voice markers are also prevalent in Crimean Tatar (Jankowski 2010: 121), Ottoman Turkish (Kissling 1960: 235), and all three dialects of Karaim (Zajączkowski 1932: 147–150; Musaev 1964: 231–233; Prik 1976: 109).

G-) {+**rA**}: The presence of this particular unproductive morpheme in our dataset may be subject to debate. As per Erdal (1991b: 469–474), this morpheme can be observed in onomatopoeic nouns that are transformed into verbs, such as *čoķra* 'to bubble; to boil'; *maŋra*, 'to shout'; *kükre* 'to roar, to bellow, or to thunder'. Nevertheless, Erdal also asserts that the base forms of these verbs are not documented in Old Turkic (1991b: 473). In the texts we have examined, there are two instances⁹⁶ that could potentially feature these suffixes.

(123) [Lev 11:21]
$$s\ddot{i}c+ra$$
 $onom*+DER$ 'to jump; to leap'.
[Lev 13:51] $kalt+ra$ $onom*+DER$ 'to shiver; to tremble'.

Erdal (1991b: 471) also examined these two verbs in his study, wherein he categorizes *kaldra*-, meaning 'to rustle (of a garment)', under this marker, but expresses uncertainty regarding the verb *sičra*-, due to its potential relation to *sač*-, which signifies 'to sprinkle; to scatter'. In Zajączkowski's research (1932: 142), this marker is observed for Western Karaim. However, some of his examples, such as *tepre*- 'to move; to stir; to shake' and *titre*- 'to shiver; to shake' in the context of Western Karaim and Ottoman Turkish, may be considered questionable, given that their etymological origins are debated⁹⁷.

⁹⁶ In addition to examples mentioned in (123), there is another instance found in Lev 12:7 as *čoķraķ*, which likely traces back to the verb *čoķra-* (see Erdal 1991: 470), followed by the deverbal noun formative marker {-k}. However, it is important to note that this verb is not documented in the Karaim dictionaries.

⁹⁷ Erdal and Clauson did not investigate instances such as those found in Zajączkowski's study. Moreover, according to Nişanyan (NS: 200; 893), most of the examples provided for this marker can actually be traced back to two-syllable nominals with a final -z, to which the denominal verb formative {+A} is attached. For instance, *titiz*, meaning 'peevish', combined with {+e} results in *titre*-, and *tepiz*, meaning 'moving', combined with {+e} forms *tepre*-.

2.2.3.1.1.2. Deverbal Verb Formative Marker

It is important to emphasize that all voice markers, with the exception of the active voice which lacks a distinct marker, are encompassed in this category. A comprehensive analysis of these markers will be provided in Section 2.2.3.2. Apart from these voice markers, only one suffix in our dataset can be classified as a deverbal verb formative marker, specifically, the negation marker.

The negation marker {-mA}, along with its phonological variants, has been attested in basic and derived verb stems in Turkic languages since their early phases (Johanson 2022a: 37). This morpheme manifests in an identical manner across numerous instances within the dataset under examination.

Notwithstanding the consistent presence of the negative marker {-mA} prior to any mood or tense marker within a verb stem, the negative agrist displays a distinct form. Particularly, in combination with the agrist, the negative marker materializes as {-mAz} or {-mAs} before the 2SG, 2PL, 3SG, and 3PL personal endings. In contrast, it assumes the form {-mAm} for the 1SG conjugation, representing a petrified form that incorporates the 1SG personal marker⁹⁸.

| (125) | [Lev 26:11] | ḫor id-mäz-Ø | to despise-NEG.AOR-3SG |
|-------|-------------|----------------------|--------------------------------|
| | | | 'it will not despise'. |
| | [Lev 26:26] | toy-maz-sïz | to be satiated-NEG.AOR-2PL |
| | | | 'you do/will not be satiated'. |
| | [Lev 26:31] | ķabul et-mäm-dir | to accept-1SG.NEG.AOR-COP |
| | | | 'I do/will not accept'. |
| | [Lev 27:11] | yuvuk̞laštır-mas-lar | to offer-NEG.AOR-3PL |
| | | | 'they do/will not offer'. |

It is important to note that the negation marker {-mA} and its associated forms with the aorist have also been identified in Crimean Tatar (Kavitskaya 2010: 65, 111–112), Crimean Turkish

-

⁹⁸ It is important to note that the text does not offer any examples of first-person plural (1PL) usage.

(Doerfer 1959a: 279), Ottoman Turkish (Kissling 1960: 99–104), and across all three dialects of Karaim (Zajączkowski 1932: 127; Musaev 1964: 263–264, 285; Prik 1976: 125; 138).

2.2.3.1.2. Compounding

A highly productive strategy in Turkic languages for forming verbs from nominals involves the compounding of certain auxiliary verbs subsequent to the nominals. In our dataset, we observe the presence of *bol-/ol-* 'to become', *et-/it-* 'to do', *ver-* 'to give' *koy-* 'to put; to set', and *kil-* 'to do' verbs in such phraseological constructs. It is important to note that the nominals in these structures may have both Turkic and non-Turkic origins. Notably, these constructions are prevalent in Crimean Tatar (Kavitskaya 2010: 79–80), Ottoman Turkish (Kissling 1964: 152–153), and all three dialects of Karaim (Musaev 1964: 242–244; Prik 1976: 111).

| (126) | [Lev 6:23] | keparat | et- | |
|-------|-------------|----------------|-----------|---------------------------------|
| | | atonement | to make | 'make atonement'. |
| | [Lev 5:6] | yazïķlï | bol- | |
| | | sinful | to become | 'to be sinful'. |
| | [Lev 5:23] | amanat | ķoy- | |
| | | deposit | to put | 'to commit'. |
| | [Lev 11:24] | mundar | ol- | |
| | | unclean | to become | 'to become unclean'. |
| | [Lev 20:12] | ïķraḥlïķ | ķïl- | |
| | | disgustingness | to do | 'to do something disgusting'. |
| | [Lev 24:19] | ḥillä | ver- | |
| | | blemish; trick | to give | 'to cause a blemish'. |

2.2.3.2. Voice

2.2.3.2.1 Active Voice

In the active voice, the sentence's subject serves as the agent of the action, indicated by personal suffixes on the verb stem, a feature prevalent across Turkic languages. It is important to recognize that both transitive and intransitive verbs can be found within this category.

(127) [Lev 9:23]
$$gel$$
- di - \emptyset to come-PAST-3SG 'he came'.

[Lev 10:1] simarla-ma-di-Ø to order-NEG-PAST-3SG 'I did not ordered'.

2.2.3.2.2. Passive Voice

In opposition to the active voice, the passive voice denotes the target of the action as the subject. Both transitive and intransitive verbs may exhibit passive forms, signified by specific markers. In our dataset, suffixes representing passivity appear as {-(I)l}, {-(U)l}, {-(I)n}, and {-(U)n}, and are subject to rounding harmony in some examples. If the verb stem concludes with a vowel, the passive voice is typically denoted by the suffix {-n}. Nonetheless, there exists an instance of a verb terminating in a vowel with the suffix {-l} affixed.

Aside from this instance, the passive marker {-l} is not observed in verbs whose stems end with a vowel. Additionally, verbs containing the derivational suffix {-lA} (refer to 2.2.2.1.1.1.) within their stems consistently employ {-n} to signify the passive voice, rather than using {-l}.

Within the analyzed corpus, the affixes $\{-In\}$ or $\{-Un\}$ are observed when the verbal root concludes with the sonorant l.

(130) [Lev 1:4]
$$kabul\ ol\ un\ \ddot{r}$$
 to be accepted-PASS-AOR-3SG 'it is/will be accepted'. [Lev 4:28] $bil\ in\ s\ddot{a}$ to know-PASS-COND-3SG 'if it is known'.

The suffixes {-II} and {-UI} are discernible in all other verbs that lack a terminal sonorant {-I} in their stems.

(131) [Lev 10:14]
$$ver-il-di-l\ddot{a}r$$
 to give-PASS-PASS-3PL 'they were given'.
[Lev 10:16] $k\ddot{u}yd\ddot{u}r-\ddot{u}l-di- extstyle{ heta}$ to burn-PASS-PAST-3SG 'it was burnt'.

In Prik's grammar (1976: 112) the passive voice markers are listed as {-(I)l}, {-(I)n}, for Crimean Karaim. The similar markers are common in other Turkic languages as well, e.g., Cr.Tat {-(I)l}, {-(U)l}, {-(U)n} (Kavitskaya 2010: 74); Ot.Tur, {-(I)l}, {-(U)l}, {-(I)n} {-(U)n} (Hagopian 1907: 116); T.Kar {-(I)l}, {-(U)l}, {-(U)n} (Musaev 1977: 47); H.Kar {-(I)n, {-(u)n} (Musaev 1977: 48). As a result, the disparity between our dataset and Prik's account lies solely in the presence of rounded variants pertaining to the relevant affixes.

2.2.3.2.3. Causative Voice

In sentences displaying the causative voice, a causative agent is generally present alongside the primary agent. The causative agent bears responsibility for initiating or provoking the primary agent to execute the action.

In fact, the causative construction primarily exhibits ambiguity when dealing with impersonal verbs, complicating the process of distinguishing between two discrete agents. In these situations, the primary agent is an impersonal subject, which further hinders the differentiation between the agents involved. It is worth noting that all the impersonal verbs in Turkic languages are intransitive.

(133) [Lev 9:17]
$$tol\text{-}dur\text{-}d\ddot{v}$$
 $avu\check{c}+\ddot{i}+n\ddot{i}$ to become full-CAUS-PAST-3SG palm+3SG.POSS+ACC 'he filled his palm'; lit. 'he caused his palm to be full'. [Lev 6:4] $\check{c}i\dot{k}$ - ar - sin ol $k\ddot{u}lni$ to go out-CAUS-3SG.VOL the ash+ACC 'he shall take out the ashes'; lit. 'he shall cause the ash|es| to go out'.

The examples presented illustrate that the addition of causative markers to intransitive verbs results in the formation of transitive verbs, while the transitivity of already transitive verbs remains unaltered.

lit. 'he made the burnt offering to come near'; 'he brought the burnt offering'.

Our dataset encompasses numerous productive, semi-productive and unproductive markers for this distinct voice, including {-Ar}, {-Ur}, {-t}, {-DIr}, {-dUr}, {-tär}, {-yuz} and {-giz}. In other Turkic languages, the same voice category can be identified, accompanied by similar causative markers, such as Cr.Tat, {-(I)t}, {-(U)t}, {-Ir}, {-Ur}, {-DIr}, {-DUr} (Kavitskaya 72–74); Ot.Tur {-(I)t}, {-(U)t}, {-Ar}, {-Ir}, {-Ur}, {-DIr}, {-DUr}, {-GUr} (LET: 96–98; Kerslake 2022: 184); T.Kar, {-t}, {-Ir}, {-Ur}, {-DIr}, {-DUr}; and H.Kar {-t}, {-Ar}, {-Ir}, {-Ur}, {-DIr}, {-DUr}; Arrive (Musaev 1964: 248).

In the examined text, the morphemes {-Ar} and {-Ur} predominantly⁹⁹ manifest in monosyllabic stems characterized by a terminal consonant.

In addition, there is a single instance, as previously demonstrated in (132), wherein a verb stem ends with a terminal vowel and is followed by the causative suffix {-t}. In other instances where the verb stem concludes with voiced consonants, the causative suffix employed is {-dIr}.

In the remaining examples, with the exception of the verb *id*- 'to do', verbs concluding with voiceless consonants are accompanied by the suffix {-tIr}.

(137) [Lev 8:33]
$$al\ddot{s}-t\ddot{r}-\ddot{\varphi}$$
 $kol+u\eta\ddot{z}+n\ddot{z}$

⁹⁹ As for a contrasting example, see, [Lev 16:20] bit-tir-sin to end (int.)-CAUS-3SG.VOL 'he shall end |it|'.

'it makes your hand get accustomed'.

to disguise-CAUS-NEG-2PL.IMP soul+PL+2PL.POSS+ACC

'you shall not make your souls disgusted'.

Notably, a singular occurrence of an unproductive ¹⁰⁰ causative marker is attested as {-tär}, followed by an additional causative marker. It is worth noting that the same marker has been documented for Crimean Tatar as an unproductive causative marker (Kavitskaya 2010: 73) and also listed for Western Karaim in a limited number of examples (Zajączkowski 1932: 123).

'they shall make them bear'.

Lastly, it merits attention that our dataset showcases three distinct verbs featuring the semi-productive 101 causative markers {-yuz} and {-giz}.

It is important to highlight that these markers have also been documented for Crimean Tatar (Kavitskaya 2010: 73–74) and all three dialects of Karaim (Musaev 1964: 251; Prik 1976: 113). Indeed, the verb *turyuz* is present in all three dialects of Karaim (KRPS: 547) and attested in numerous Kipchak languages, such as C.Tat *turyuz* (KRUS: 608), as well as in Ottoman Turkish in a slightly different form, e.g., Ot.Tur *turğur*- 'to stop; to bring a halt; to raise' (not present in modern Turkish) (EDPT: 541). However, the verbs *oturyuz*- and *girgiz*- (listed as *kirgiz*-) were found exclusively in Crimean Karaim (KRPS: 322; 435). Upon observation, the final consonant -*r* in the East Old Turkic causative marker {-GUr} (Johanson 2022c: 137) undergoes a transformation to -*z* in the marker {-GUz}, a phenomenon which probably initially emerged during the Middle Kipchak period (Berta and Csató 2022a: 156).

¹⁰⁰ For information regarding this categorization, refer to Zajączkowski 1932: 129.

¹⁰¹ For further details, see Zajączkowski 1932: 129.

2.2.3.2.4. Reflexive Voice

In the reflexive voice, the agent carries out an action upon themselves. Our text features two verbs that exhibit reflexive voice through the attachment of the suffixes {-un} and {-in} to their stems. Prik's grammar (1976: 112) also lists reflexive markers {-(I)n} and {-(U)n} for Crimean Karaim. Notably, similar reflexive markers are prevalent in many Turkic languages, such as Cr.Tat {-(I)n}, {-(U)n} (Kavitskaya 2010: 74), Ot.Tur {-(U)n}, {-(I)n} (Hagopian 1907: 117), T.Kar {-(U)n}, {-(I)n}, and H.Kar {-(u)n}, {-(I)n} (Musaev 1977: 47).

Jankowski (1997: 15) notes that the reflexive pronoun *kendi*, 'herself/himself/itself,' is used as a lexical means of expressing reflexivity in a Crimean Karaim Bible translation. This method of conveying reflexivity is also observed in other Turkic languages, as pointed out by Johanson (2022a: 48), and within our dataset.

'you shall rejoice'.

(141) [Lev 15:11]
$$gendisi+n$$
 $yayka-ma-di-Ø$ himself+ACC to wash-NEG-PAST-3SG 'he has not wash himself'.

2.2.3.2.5. Reciprocal Voice

In our dataset, the reciprocal marker is identified as {-(ï)š}, a form that is prevalent in numerous Turkic languages with similar variations, such as Cr.Kar {-(I)š}, {-(U)š} (Prik 1976: 112), Cr.Tat {-(I)š}, {-(U)š} (Kavitskaya 2010: 74), Ot.Tur {-(I)š}, {-(U)š} (Hagopian 1907: 116–117), T.Kar {-(I)š}, {-(U)š}; H.Kar {-(I)s}, {-(u)s} (Musaev 1977: 47). This voice involves multiple subjects who partake in a shared action. Nevertheless, in some cases, a verb followed by a cooperative marker may express a distinct action from its base form and, consequently, not demonstrate a reciprocal function. Verbs of this nature require etymological analysis to ascertain whether these verbs originally had reciprocal functions and underwent semantic evolution to acquire new significance. For instance, in our data, [Lev 9:15] *al*- means 'to take' whereas [Lev 8:33] *ališ*- means 'to get accustomed'. According to Clauson (EDPT: 153), the

verb *ališ*- is derived from *al*-, possessing a cooperative function initially. However, in later stages, it acquired a new cognate meaning, 'to get accustomed', in some Turkic languages. Nevertheless, one example below clearly presents the reciprocal voice in our dataset.

2.2.3.3. Tense

2.2.3.3.1. Definite Past Tense

In the present text, the markers {-DI} and {-Du} are utilized to indicate the definite past tense, and they are subsequently followed by personal endings. The adoption of identical and similar forms is an extensively observed linguistic phenomenon across numerous Turkic languages, which includes Crimean Karaim (Prik 1976: 131), Crimean Tatar (Kavitskaya 2010: 66), Ottoman Turkish (Kissling 1960: 69), and Western Karaim (Musaev 1964: 272).

Table 58: Definite Past Tense Markers

| | Lev of Göz. | Cr. Kar | Cr. Tat | Ot. Tur | T. Kar | H. Kar |
|-----|-------------|----------|----------|----------------|--------------------|----------------|
| | 1841 | | | | | |
| 1SG | {-DIm} | {-DIm} | {-DIm} | {-DIm}, {-DUm} | {-DIm}, {-DUm} | {-DIm},{-Dum} |
| 2SG | - | {-DIŋ} | {-DIŋ} | {-DIŋ}, {-DIŋ} | {-DIy}, {-DUy} | {-DIn},{-Dun} |
| 3SG | {-DI}, | {-DI} | {-DI} | {-DI}, {-DU} | {-DI}, {-DU} | {-DI}, {-Du} |
| | {-Du} | | | | | |
| 1PL | - | {-DIK} | {-DIK} | {-DIK}, {-DUK} | {-Dïḥ, {-Dik}, | {-DIK}, {-Duķ} |
| | | | | | {-Duḥ}, {-Dük} | |
| 2PL | {-dïŋïz} | {-DIŋIz} | {-DIŋIz} | {-DIŋIz}, | {-DIyIz}, {-DUyIz} | {-DInIz}, |
| | | | | $\{-DU guz\}$ | | {-Dunïz} |
| 3PL | {-DIlAr} | {-DIlAr} | {-DIlAr} | {-DIIAr}, | {-DIlAr}, {-DUlAr} | {-DIlAr}, |
| | | | | {-DUlAr} | | {-Dular} |

The table above illustrates that the primary distinction between our data and that of Crimean Tatar, Crimean Karaim, and Ottoman Turkish is the use of the {-Du}¹⁰² suffix. It is noteworthy that this particular suffix occurs only four times in total, and across three distinct verb stems within the book.

_

¹⁰² Doerfer (1959a: 274) also observed the absence of the past tense marker $\{-DU\}$ in Crimean Turkish, e.g., $koy-di-\emptyset$ to put-PAST-3SG 'she/he/it put'.

(143) [Lev 10:3]
$$sus-tu-\emptyset$$
 to keep silent-PAST-3SG 'he kept silent'. [Lev 18:3] $otur-du-\eta \ddot{z}$ to dwell-PAST-2PL 'you dwelt'. [Lev 18:25/28] $kus-tu-\emptyset$ to vomit-PAST-3SG 'it vomited'.

With the exception of the instances that utilize the {-Du} suffix, the past tense marker for all other cases is {-DI}, irrespective of the roundness of vowels present within the verb stems.

(144) [Lev 5:1]
$$g\ddot{o}r-di-\cancel{\emptyset}$$
 to see-PAST-3SG 'he saw'. [Lev 8:15] $t\ddot{o}k-ti-\cancel{\emptyset}$ to pour out-PAST-3SG 'he poured out'. [Lev 8:17] $k\ddot{u}y-d\ddot{u}r-di-\cancel{\emptyset}$ to burn-CAUS-PAST-3SG 'he burned'.

In other instances, we observe that, contrary to rounding harmony, consonant harmony is generally adhered to when the marker is appended to words¹⁰³.

| (145) | [Lev 6:10] | ver-di-m | to give-PAST-1SG | 'I have given'. |
|-------|------------|------------|------------------|-----------------|
| | [Lev 8:5] | ayt-tï-Ø | to say-PAST-3SG | 'he said'. |
| | [Lev 10:5] | sözlä-di-Ø | to say-PAST-3SG | 'he had said'. |

2.2.3.3.2. Present Tense

١

Doerfer (1959a: 278) identified several markers in Crimean Turkish and classified them as praesens durativum, which include $\{-(A)yIr\}$, $\{-(I)yIr\}$, $\{-(U)yUr\}$, $\{-(A)yUr\}$, $\{-(I)yur\}$, $\{-(U)yur\}$, and $\{-(A)yor\}$. While these suffixes are not present in Prik's grammar (1976), Çulha (2019: 120) refers to them as present continuous tense markers and lists them for Crimean Karaim based on some Crimean Karaim mejumas that exhibit influences from Crimean Turkish features. It should be noted that this marker has been preserved since the early phases of Ottoman language in the 14th century. The marker was formed through the combination of a converb in $\{-(y)A\} \sim \{-(y)I\}$ with the aorist of the auxiliary verb $y\ddot{u}ri$ - 'to walk' (Kerslake 2022: 186). Subsequently, this form acquired a shortened version $\{-(A)y\ddot{u}r\}/\{-(I)y\ddot{u}r\}$ by the end of the fifteenth century and eventually developed into the form $\{-(I)yor\}$, which appears as a progressive tense marker in Ottoman Turkish (Kissling 1960: 80). However, it is worth noting that these markers are absent in Crimean Tatar (Kavitskaya 2010: 63–64), Western Karaim

¹⁰³ However, our data reveal three exceptions in which the past tense marker, commencing with the voiced consonant {-d}, is affixed to verb stems that end with an unvoiced final consonant, e.g., *degiš-di-Ø* to change-PAST-3SG 'it has changed'; [Lev 26:43] *hor it-di-Ø* to despise-PAST-3PL 'they despised'.

(Musaev 1977: 54–55), and texts written in Crimean Kipchak Karaim (Jankowski 2015a). This provides strong evidence for the influence of Ottoman Turkish on the analyzed data.

Table 59: Present Tense Markers

| | Cr.Kar (based on | Ot.Tur | Cr.Tat | T.Kar | H. Kar |
|-----|---------------------|-----------------|---------------|------------------|------------------------|
| | mejumas) | | | | |
| 1SG | {-(A)yIrIm} | {-(I)yorum}, | {-Am}, | {-AmIn}, {-Am}, | {-Amen}, |
| | | {-(U)yorum} | {-yIm} | {-ymIn}, {-ym} | {-Am}, |
| | | | | | {-ymen}, |
| | | | | | {-ym} |
| 2SG | {-(A)yIrsIn} | {-(I)yorsun}, | {-AsIŋ} | {-AsIn}, {-As}, | {-Asen}, |
| | | {-(U)yorsun} | | {-ysIn}, {-ys} | $\{-As\},$ |
| | | | | | {-ysen}, |
| | | | | | {-ys} |
| 3SG | {-(A)yIrlAr} | {-(I)yor}, | {-A}, {-y} | {-AdIr}, {-AdI}, | {-AdIr}, {-AdI}, |
| | | {-(U)yor} | | {-ydIr}, {-ydI} | ${-ydIr},{-ydI},$ |
| | | | | | $\{-At\}, \{-yt\}$ |
| 1PL | {-(A)yIrIz} | {-(I)yoruz}, | {-AmIz}, | {-AbIz}, {-ybIz} | {-AbIz}, {-ybIz} |
| | | {-(U)yoruz} | {-ymIz} | | |
| 2PL | {-(A)yIrsIz} | {-(I)yorsuŋuz}, | {-AsIŋIz} | {-AsIz}, {-ysIz} | {-AsIz}, {-ysIz} |
| | | {-(U)yorsuŋuz} | | | |
| 3PL | {-(A)yIr} | {-(I)yorlar}, | {-A(lAr), | {-AdIrlAr}, | {-AdIrlAr}, {-AdlAr}, |
| | | {-(U)yorlar} | $\{-y(lAr)\}$ | {-AdlAr}, | ${-ydIrlAr}, {-ydlAr}$ |
| | | | | {-ydlrlAr}, | |
| | | | | {-ydlAr} | |

In the Lev of the Göz. 1841, it is possible to find the markers {-(a)yïr}, {-yur} and {-ïyïr}, altogether 6 times. However, it is worth noting that except for one example in Chapter 25, all the examples appear in Chapter 11, which consists of the highest Crimean Turkish and/or Ottoman Turkish features among the other chapters. Hence, once again, the Lev of Göz. 1841 presents some Crimean Turkish features which are not present in Crimean Kipchak Karaim (and thus not in the most of the Crimean Karaim Bible translations), but Crimean Turkish Karaim.

(146) [Lev 11:26] ayïr-ma-yur-Ø to divide-NEG-PRES-3SG 'it does not divide'

| [Lev 11:26] | gevšemäk getir-me-yir-Ø | to chew cud-NEG-PRES-3SG |
|----------------|-------------------------|-----------------------------|
| | | 'it does not chew the cud'. |
| [Lev 11:41/42] | doɣ-ayïr-Ø | to be born-PRES-3SG |
| | | 'it swarms'. |
| [Lev 11:43] | ķozla-yïr-Ø | to lay eggs-PRES-3SG |
| | | 'it lays eggs'. |
| [Lev 25:16] | sat-ïyïr-Ø | to sell-PRES-3SG |
| | | 'he is selling'. |

Another present tense marker occurs in the book as {-AydIr}. Jankowski (1997: 15) also mentions this type of present tense marker which indicates the progressive actions in contrast to habitual ones. In our data, there are only examples of 1SG: {-eydirmin}, 3SG: {-AydIr}, and 3PL: {-aydïrlar}, which appears altogether 13 times.

| (147) | [Lev 14:34] | ver-äydir-min | to give-PRES-1SG 'I am giving'. |
|-------|--------------------|-----------------------|--|
| | [Lev 15:23] | otur-aydïr-Ø | to sit-PRES-3SG 'she is sitting'. |
| | [Lev 17:5, 21:9] | et-äydir-lär | to do-PRES-3PL 'they are doing'. |
| | [Lev 17:7] | az-aydïr-lar | to go astray-PRES-3PL 'they are going astray'. |
| | [Lev 18:3, 20:22] | getir-äydir-min | to bring-PRES-1SG 'I am bringing'. |
| | [Lev 18:24, 20:23] | sür-äydir-min | to drive away-PRES-1SG |
| | | | 'I am driving away'. |
| | [Lev 21:6] | yuvuklaštïr-aydïr-lar | to bring forward-PRES-3PL |
| | [Lev 21:8] | yuvuķlaštïr-aydïr | 'they are bringing forward'. to bring forward-PRES-3SG 'he is bringing forward'. |

It should be emphasized that while the aforementioned marker bears resemblance to those delineated in Table 59 for Western Karaim, the precise forms have not been documented for Crimean Tatar, Crimean Turkish, Ottoman Turkish, and Prik's (1976) Crimean Karaim grammar.

2.2.3.3.3. Aorist

The aorist marker {-(X)r} is a feature shared by Turkic languages (Johanson 2022a: 39), and as a result, we can find similar forms of this marker in Crimean Turkish¹⁰⁴ (Doerfer 1959a: 278), Crimean Tatar (Kavitskaya 2010: 64–65), Ottoman Turkish (Hagopian 1907: 152–153), and all three dialects of Karaim (Prik 1976: 138; Musaev 1977: 55–56).

Table 60: *Aorist Markers*

| | Cr.Kar | Ot. Tur | Cr.Tat | T.Kar | H. Kar |
|-----|---------------------------|-----------------------|-------------------------|-----------------|-----------------|
| 1SG | {-(A)rIm}, | {-(A)rIm}, | {-(A)rIm}, | {-(A)rmIn}, | {-(A)rmen}, |
| | $\{-(I)rIm\}$ | $\{-(I)rIm\},$ | $\{-(I)rIm\}$ | $\{-(I)rmIn\},$ | {-(I)rmen}, |
| | | $\{-(U)rUm\}$ | | $\{-(U)rmIn\}$ | {-(u)rmen} |
| 2SG | {-(A)rsIŋ}, | {-(A)rsIn}, | {-(A)rsIŋ}, | {-(A)rsIn}, | {-(A)rsen}, |
| | $\{-(I)rsI\mathfrak{y}\}$ | $\{-(I)rsIn\},$ | $\{-(I)rsIn\}$ | $\{-(I)rsIn\},$ | {-(I)rsen}, |
| | | {-(U)rsUn} | | $\{-(U)rsUn\}$ | {-(u)rsen} |
| 3SG | {-(A)r}, | $\{-(A)r\},$ | {-(A)r}, | {-(A)r}, | {-(A)r}, |
| | $\{-(I)r\}$ | $\{-(I)r\},$ | $\{-(I)r\}$ | $\{-(I)r\},$ | $\{-(I)r\},$ |
| | | $\{-(U)r\}$ | | $\{-(U)r\}$ | $\{-(u)r\}$ |
| 1PL | {-(A)rmIz}, | {-(A)rIz}, | {-(A)rmIz}, | {-(A)rbIz}, | {-(A)rbIz}, |
| | $\{-(I)rmIz\}$ | $\{-(I)rIz\},$ | $\{-(I)rmIz\}$ | $\{-(I)rbIz\},$ | $\{-(I)rbIz\},$ |
| | | $\{-(U)rUz\}$ | | $\{-(U)rbIz\}$ | $\{-(u)rbIz\}$ |
| 2PL | {-(A)rsIz}, | {-(A)rsIŋIz}, | $\{-(A)rsI(\eta I)z\},$ | {-(A)rsIz}, | {-(A)rsIz}, |
| | $\{-(I)rsIz\}$ | $\{-(I)rsI\eta Iz\},$ | $\{-(I)rsI(\eta I)z\}$ | $\{-(I)rsIz\},$ | $\{-(I)rsIz\},$ |
| | | $\{-(U)rsU\eta Uz\}$ | | $\{-(U)rsIz\}$ | $\{-(u)rsIz\}$ |
| 3PL | {-(A)rlAr}, | {-(A)rlAr}, | $\{-(A)r(lAr)\},$ | {-(A)rlAr}, | {-(A)rlAr}, |
| | $\{-(I)rlAr\}$ | $\{-(I)rlAr\},$ | $\{-(I)r(lAr)\}$ | $\{-(I)rlAr\},$ | $\{-(I)rlAr\},$ |
| | | {-(U)rlAr} | | $\{-(U)rlAr\}$ | $\{-(u)rlAr\}$ |

In the Lev of the Göz. 1841, aorist markers are used to form the future/present tense. These markers manifest as $\{-(\ddot{i})r\}$, $-\{(A)r\}$, and $\{-(U)r\}$ depending on the fronting and rounding harmony, and are followed by personal endings that differ slightly from those described in

136

_

The markers were listed as $\{-(I)r\}$, $\{-(U)r\}$ and $\{-(A)r\}$.

Prik's grammar (1976: 138), e.g., 1SG: {-ArmIn}; 2SG: -; 3SG: {-Ar}, {-ur}, {-ïr}; 1PL: -; 2PL: {-ArsIz}; 3PL: {-Ar(lAr)}. It can be observed that the 1SG personal marker {-mIn} is not present in the aforementioned languages, except for Trakai Karaim. However, the fluctuation in 1SG: {-Im}, {-mIn}, {-men}, 2SG: {-sIn}, {-sIn}, {-sen}, and 2PL: {-sIz}, {-sInIz} can be noticed in Crimean Turkish (1959a: 278–279) as well.

There exists a slight discrepancy between our data and Prik's description in terms of rounding harmony. Specifically, in the Lev of the Göz. 1841, only 13 occurrences were found for six distinct lexical items, where verb stems had rounded vowels, and the same rounding was reflected in the agrist suffix.

(148) [Lev 11:32]
$$yap-ul-ur-\emptyset$$
 to do-PASS-AOR-3SG 'it is done'. [Lev 11:39] $\ddot{u}l-\ddot{u}r-s\ddot{a}-\emptyset$ to die-AOR-COND-3SG 'if it dies'. [Lev 11:32] $ol-ur-\emptyset$ to become-AOR-3SG 'it will become'. [Lev 25:19; 26:5] $otur-ur-s\ddot{v}z$ to dwell-AOR-2PL 'you will dwell'.

However, in the majority of examples, the rounding harmony was not observed for this marker.

The variation in the use of the aorist marker in the text could be attributed to the specific characteristics of some chapters that display a strong influence of Oghuzic and/or Ottoman Turkish features. This could be seen in the presence of the variant of the aorist marker {-(U)r} in Crimean Turkish (Doerfer 1959a: 278) and Ottoman Turkish (as indicated in Table 60), which could have had an impact on the usage of the marker in the text.

2.2.3.3.4. Future Tense

The present text furnishes a solitary illustration for the categorical future tense marker, namely $\{-A\S Ak\}^{105}$, which conveys an event's anticipated occurrence in the future. Nevertheless, the aforementioned marker performs a distinctive function in the example expounded below, wherein it serves as a thematic marker that is succeeded by an auxiliary verb attached to the conditional mood marker $\{-sa\}$ (see section 2.2.3.4.4.).

 $^{^{105}}$ It is worth noting that the same marker also functions as a future participle marker in our data, as illustrated in Section 2.2.2.6.

Similar forms of the categorical future tense marker are found in several Turkic languages, e.g., C.Tat, {-AǯAk}, {-yǯAk} (Kavitskaya 2010: 66); C.Tur, {-eǯek}, {-aǯaḥ} (Doerfer 1959a: 279); C.Kar, {-AǯAk}; {-yǯAk}; Ot.Tur {-AǯAk} (Kissling 1960: 75–76). However, it is important to mention that in Western Karaim languages, the categorical future tense is marked by the use of the marker {-QAy}, which also serves as an optative mood marker (Musaev 1977: 54–55). Additionally, it should be acknowledged that the optative marker {-QAy} has been found to have prospective functions in other Turkic languages as well (Johanson 2022b: 113). Nevertheless, in our data, the marker {-QAy} only appears in the optative mood and not in any instance of the future tense, as indicated in section 2.2.2.4.3. Consequently, the use of this tense marker in our text differs from that of Western Karaim.

2.2.3.4. Mood

The present study's data encompasses four categories of mood, namely imperative, voluntative, optative, and conditional. It is worth noting that there exists a close connection between the imperative, voluntative, and optative moods in Turkic languages (Johanson 2022a: 39). As shown below, some scholars have made different categorizations for the same markers in certain Turkic languages. In the interest of clarity, this study has adopted the descriptions of other studies for the purpose of categorization.

2.2.3.4.1. Imperative

The imperative mood, customarily employed to issue directives, is exemplified extensively in the dataset, given that the Book of Leviticus predominantly comprises imperative sentences. This mood is formed with various markers in the Karaim dialects (Musaev 1964: 269; Prik 1976: 143), Crimean Tatar (Kavitskaya 2010: 70), Crimean Turkish (Doerfer 1959a: 278), and Ottoman Turkish (Kissling 1960: 88–89). It typically has forms for the second person singular and plural.

Table 61: *Imperative Mood Markers*

| | Cr.Kar | Cr.Tur | Ot. Tur | Cr.Tat | T.Kar | H. Kar | _ |
|-----|--------|--------|---------|--------|---------------|---------------|---|
| 2SG | {-Ø} | {-Ø} | {-Ø} | {-Ø} | {-Ø}, {-QIn}, | {-Ø}, {-QIn}, | |

| | Cr.Kar | Cr.Tur | Ot. Tur | Cr.Tat | T.Kar | H. Kar |
|-----|-------------------|-----------------|------------------------------------|-----------|---------------|------------------|
| | | | | | {-QUn} | {-k̞un}, {-y̞un} |
| 2PL | $\{-(I)\eta Iz\}$ | {-In}, {-Iŋ}, | $\{-(y)I(\eta I)z\},$ | {-(I)ŋIz} | {-(I)yIz}, | $\{-(I)nIz\},$ |
| | | ${-I\eta Iz},$ | $\{\text{-}(y)U(\mathfrak{y}U)z\}$ | | $\{-(U)yUz\}$ | $\{-(u)nuz\}$ |
| | | $\{-I\eta Iz\}$ | | | | |

As noted, Prik (1976: 144) does not include the 2SG.IMP markers {-QIn} and {-QUn} for Crimean Karaim in her description, instead considering them as features of Western Karaim. However, Jankowski's study (1997: 16) and Çulha's description (2019: 122–123) both include these forms for Crimean Karaim. Consistent with these sources, our data includes examples of the 2SG.IMP suffix as {-QIn}, while the 2PL.IMP suffix appears as {-(I)ŋIz}, {-(U)ŋIz}, or {-(U)ŋUz}.

As expected, in our dataset, the diverse forms of the imperative marker {-QIn} are determined in accordance with both the vowel and consonant harmony present within the verb stem.

In the matter of 2PL.IMP markers, it is significant to note that the variant {-(U)ŋUz} appears in a mere three instances within the text, where both complete and partial rounding harmony are adhered to in the context of the 2PL.IMP suffix.

Aside from these three instances identified in Chapter 11, rounding harmony is exclusively witnessed in the initial connective vowel of the 2PL.IMP as {-(U)ŋIz} when the verb stem concludes with a consonant in a rounded syllable, a scenario that only presents itself in two occurrences.

Finally, it is worth noting that the bare verb stem was not used to form the 2SG.IMP mood in the Lev of the Göz. 1841, which is a common practice in many Turkic languages, including Crimean Tatar, Crimean Turkish, and Ottoman Turkish (Johanson 2022a: 39). Although this type of 2SG.IMP is also listed for Western Karaim, the specific markers listed in Table 61 are mostly encountered in Western Bible Karaim translations. Therefore, this feature distinguishes the language of our data from Crimean Turkish Karaim.

2.2.3.4.2. Voluntative

In our text, the voluntative mood is expressed through forms for 1SG, 1PL, 3SG, and 3PL and is used to convey offers for actions by the subject or to represent the subject's indirect commands, wishes, and demands.

In our data, we encounter only two examples of the 1SG.VOL marker {-äyim} and the 1PL.VOL marker {-lim}.

The text features 3SG.VOL markers in the forms of {-sIn} and {-sUn}, while 3PL.VOL markers are represented by {-sInlAr} and {-sUnlAr}. Notably, the application of rounding harmony in the examples containing these markers in the text is generally inconsistent. In the majority of examples, the rounding harmony is not observed.

However, it should be noted that there are a total of 13 instances in Chapters 11 and 23 where the rounded variants {-sUn} and {-sUnlAr} are attested.

Similar and different markers are found in other Turkic languages as well. The table below illustrates the situation in Karaim dialects (Musaev 1964: 269, 286; Prik 1976: 143, 145),

Crimean Turkish (Doerfer 1959a: 278), Ottoman Turkish (Hagopian 1907: 171), and Crimean Tatar (Kavitskaya 2010: 70; Doerfer 1959b: 387).

Table 62: Voluntative Mood Markers

| | Cr.Kar | Cr.Tur | Ot. Tur | Cr.Tat | T.Kar | H. Kar |
|-----|-----------|-----------------|---------------|-------------|---------------|------------|
| 1SG | {-(A)yIm} | {-(A)yIm}, | {-(y)AyIm} | {-(A)yIm} | {-(A)yIm} | {-(A)yIm} |
| | | $\{-(A)ym\},$ | | | | |
| | | $\{-(A)yIn\}$ | | | | |
| 2SG | - | {-sAnA}, | {-sAnA} | {-sAnA}, | - | - |
| | | $\{-sA\eta A\}$ | | $\{-sAnA\}$ | | |
| 3SG | {-sIn} | {-sIn}, | {-sIn}, | {-sIn} | {-sIn}, | {-sIn}, |
| | | $\{-sUn\}$ | $\{-sUn\}$ | | $\{-sUn\}$ | {-sun} |
| 1PL | {-(A)yIK} | {-(A)yIK}, | {-(y)AlIm} | {-(A)yIK} | {-(A)yIK}, | {-(A)yIK}, |
| | | {-(a)yïḥ}, | | | {-(a)yïḥ} | {-(a)yïḥ} |
| | | $\{-(A)yK\},$ | | | | |
| | | $\{-(a)yh\},$ | | | | |
| | | $\{-(A)IIm\}$ | | | | |
| 2PL | - | - | {-sAŋIzA} | = | - | - |
| | | | | | | |
| 3PL | {-sInlAr} | {-sInlAr} | {-sInlAr}, | {-sInlAr} | {-sInlAr}, | {-sInlAr}, |
| | | | $\{-sUnlAr\}$ | | $\{-sUnlAr\}$ | {-sunlar} |

The 1SG.VOL marker {-(A)yIm} is attested in all of the languages listed above. However, the 1PL.VOL marker {-(A)IIm} is not listed in Crimean Tatar, Western Karaim, and Prik's Crimean Karaim grammar. Therefore, this marker appears to be a feature of Crimean Turkish and/or Ottoman Turkish, as demonstrated in the Table 62. On the other hand, the 3SG and 3PL voluntative mood markers {-sIn} and {-sInIAr} are present in all of the languages listed, including Karaim dialects (Musaev 1964: 269; Prik 1976: 143), Crimean Tatar (Kavitskaya 2010: 70), Crimean Turkish (Doerfer 1959a: 278), and Ottoman Turkish (Kissling 1960: 87). It is worth noting that the presence of rounded counterparts of these affixes in our dataset has not been documented in Crimean Tatar, Crimean Karaim, and Crimean Turkish. Their appearance might be attributed to Ottoman Turkish influence.

2.2.3.4.3. Optative

The optative mood is closely related to the voluntative mood as it also expresses the subject's desires, wishes, hope, and also polite commands. In specific structures, it is also used as the

subjunctive mood (Jankowski 1997: 16). Thus in our dataset, it is difficult to distinguish these two moods. In Prik's Crimean Karaim grammar (1976: 143–145) and Doerfer's description for Crimean Turkish (1959a: 278), there is no distinction between the optative and imperative mood markers. However, Ottoman Turkish (Kissling 1960: 88), Crimean Tatar (Kavitskaya 2010: 70–71), and Western Karaim (Musaev 1964: 286) present some different optative mood markers.

Table 63: Optative Mood Markers

| | Ot. Tur | Cr.Tat | T.Kar | H. Kar |
|-----|--------------------------------------|-----------------------------|------------|-----------|
| 1SG | {-(y)Am} | {-QAydIm}, | {-GeymIn}, | {-QAymen} |
| | | {-QAyIm} | {-ḫeymIn} | |
| 2SG | {-(y)AsIn}, | {-QAydIŋ}, | {-GeysIn}, | {-QAysIn} |
| | $\{-(y)AsUn\}$ | ${-QAysIn}$ | {-heysIn} | |
| 3SG | {-(y)A} | {-QAydI}, | {-Gey}, | {-QAy} |
| | | $\{-QAy\}$ | {-hey} | |
| 1PL | {-(y)AyIz}, | {-QAydIK}, | {-GeybIz}, | {-QAybIz} |
| | $\{-(y)AyUz\}$ | {-QAymIz} | {-ḫeybIz} | |
| 2PL | $\{-(y)AsI(\eta I)z\},$ | {-QAydInIz}, | {-GeysIz}, | {-QAysIz} |
| | $\{\text{-}(y)AsU(\mathfrak{y}U)z\}$ | $\{-QAyI(\mathfrak{g}I)z\}$ | {-ḫeysIz} | |
| 3PL | $\{-(y)A(lAr)\}$ | {-QAydIlAr}, | {-GeylAr}, | {-QAylAr} |
| | | | {-ḫeylAr} | |

Our data exhibits a mixed characteristic from the languages mentioned above, as it contains both $\{-(y)A\}$ and $\{-QAy\}$ markers that are attached to verb stems. Among them, the 3SG.OPT marker $\{-(y)a\}$ appears five times¹⁰⁶ in the text.

In other examples, we find the 2PL.OPT markers $\{-(y)\text{esiz}\}$ and $\{-(y)\text{äsiz}\}$ a total of 6 times. In these examples, we also find the negation marker $\{-\text{mA}\}$, which is attached to the verb stems.

1

¹⁰⁶ Among the pertinent examples, a distinct petrified conjunction can be identified, illustrated by *ol-ma-ya-Ø* (to be-NEG-VOL-3SG) in Lev 10:7, which conveys the meaning of 'or else'. It is essential to highlight that in Crimean Karaim, this conjunction more frequently appears as *bol-ma-yay-Ø* (to be-NEG-VOL-3SG), representing 'or else; lest' (CKED: 90). Consequently, this particular case accentuates both the phonological and morphological distinctions between Ottoman Turkish and Kipchak linguistic features.

[Lev 11:9] *ye-yesiz* to eat-2PL.OPT 'you shall eat'.

[Lev 11:44] *id-me-yäsiz* to do-NEG-2PL.OPT 'you shall not do'.

The dataset reveals that optative mood markers manifest as {-QAy} a total of 11 times, which have not been documented for Ottoman Turkish.

(159) [Lev 17:5] *korban et-käy-lär* to make sacrifice-OPT-3PL

'they shall make sacrifice'.

[Lev 19:17] ček-mä-gäy-sin to bear-NEG-OPT-2SG

'you shall not bear'.

[Lev 23:2] *čaķir-yay-siz* to call-OPT-2PL

'you shall call'.

According to Prik's grammar (1976: 145–146), the optative mood marker {-QAy} is characterized as a feature of Western Karaim. Similarly, Doerfer (1959a: 279) suggests that the markers {-yay} and {-gey} occasionally appear in Crimean Turkish due to influence from Northern Crimean Tatar. Consequently, our dataset exhibits a combination of both Ottoman Turkish and Kipchak linguistic characteristics for this mood.

2.2.3.4.4. Conditional

The conditional mood marker in many Turkic languages, including Crimean Tatar (Kavitskaya 2010: 70), Crimean Turkish (Doerfer 1959a: 278), Ottoman Turkish (Kissling 1960: 81, 83–85), and all three dialects of Karaim (Musaev 1964: 290; Prik 1976: 148), takes the form of {-sA} attached to the verb stem. This suffix undergoes fronting harmony and is followed by the personal endings.

Table 64: Conditinal Mood Markers

| | Cr.Kar | Cr.Tur | Ot. Tur | Cr.Tat | T.Kar | H. Kar |
|-----|----------|----------|----------|----------|----------------|----------|
| 1SG | {-sAm} | {-sAm} | {-sAm} | {-sAm} | {-sAm} | {-sAm} |
| 2SG | {-sAŋ} | {-sAŋ} | {-sAŋ} | {-sAŋ} | {-sAy} | {-sAn} |
| 3SG | {-sA} | {-sA} | {-sA} | {-sA} | {-sA} | {-sA} |
| 1PL | {-sAK} | {-sAK} | {-sAK} | {-sAK} | {-saḥ}, {-sek} | {-sAK} |
| 2PL | {-sAŋIz} | {-sAŋIz} | {-sAŋIz} | {-sAŋIz} | {-sAyIz} | {-sAnIz} |
| 3PL | {-sAlAr} | {-sAlAr} | {-sAlAr} | {-sAlAr} | {-sAlAr} | {-sAlAr} |

The conditional relation may be emphasized through the use of the conjunction *egär* which was borrowed from Persian. However, it is not obligatory to use this conjunction. The majority of examples in our data show that the conditional marker {-sA} is attached to the verb stem without any tense markers. This type of conditional structure refers to hypothetical actions or states in many Turkic languages (Johanson 2022a: 42). However, a detailed investigation of the context reveals that it does not necessarily imply a hypothetical or unlikely scenario, but rather potential actions or states that may happen in the future.

(160) [Lev 14:34]
$$ver$$
- $s\ddot{a}$ - m to give-COND-1SG 'if I give'. [Lev 17:3] soy - sa - \emptyset to slay-COND-3SG 'if he slays'. [Lev 25:14] al - sa - η to take-COND-2SG 'if you buy'.

In our dataset, we found a total of 13 examples in which the conditional suffix is attached to the verb stem following the aorist marker. In contrast, in two examples, the conditional suffix is attached to the e-/i- clitic, which comes after a thematic stem composed of an aorist marker. These structures are frequently encountered in other Turkic languages (Johanson 2022a: 42) and refer to potential actions or states of the subject. Therefore, our data does not reveal any distinction between these structures and the previously discussed one.

(161) [Lev 3:1]
$$yuvu \not k la \check{s} t \ddot{i} r - \ddot{i} r e - s \ddot{a} - \emptyset$$
 to bring forward-AOR CL-COND-3SG 'if he brings forward'.

[Lev 11:34] $gel - ir - se - \emptyset$ to come-AOR-COND-3SG 'if he comes'.

[Lev 11:39] $\ddot{u}l - \ddot{u}r - s \ddot{a} - \emptyset$ to die-AOR-COND-3SG 'if it dies'.

In the Lev of the Göz. 1841, a total of 11 examples can be found where the conditional mood is used to indicate actions that may have already occurred. It is noteworthy that all of these examples contain the clitic e-/i-. The past tense marker is attached to the thematic stem, and the clitic e-/i- is followed by the conditional marker.

(162) [Lev 5:21]
$$al-d\ddot{\imath}$$
 $i-s\ddot{a}-\cancel{\emptyset}$ to take-PAST CL-COND-3SG 'if he has taken'.

[Lev 13:4] degiširil-mä-di e-sä-Ø to change (int.)-NEG-PAST CL-COND-3SG 'if it has not changed'.

[Lev 13:37] $tur-d\ddot{i}$ $e-s\ddot{a}-\emptyset$ to stop-PAST CL-COND-3SG 'if it has stopped'.

In a single example found in the text, the conditional marker was attached to thematic stem, whereas the past tense marker followed the clitic e-. However, it is worth noting that the personal marker was attached to both the thematic stem and the clitic e- in this instance, which could possibly be attributed to a scribal error.

(163) [Lev 10:19] aša-sa-m e-y-dim to eat-COND-1SG CL-CONN-PAST-1SG 'if I had eaten'.

Finally, in our dataset, a solitary instance exhibits the future tense marker {-aǯak̞} succeeded by an auxiliary verb bearing the conditional marker. This conditional mood construction represents hypothetical actions or states that may occur any time in the future.

(164) [Lev 1:2] yuvuklaštir-ažak ol-sa-Ø
to bring forward-FUT AUX-COND-3SG
'if he ever brings |it| forward'.

2.2.3.5. Converbs

It is important to acknowledge that the data being examined encompasses a variety of converbs, which arise from the integration of thematic suffixes, adverbial cases, possessive suffixes, and postpositions. Combinations such as [{QAN}+POSS+CASE+ POST] and [{DIK}+POSS+CASE+ POST] will be explored in a separate section (see section 2.3.1.3.1.). This section is devoted exclusively to the analysis of individual markers. Prior to enumerating the relevant markers, it is pertinent to highlight that the verbal noun marker {-mA} operates as a converb marker in merely a single lexeme, which appears with considerable frequency (a total of 50 instances).

(165) [Lev 1:1] de-mä to say-VN 'saying'.

Musaev (1964: 296) likewise noted an analogous occurrence of this marker in Trakai Karaim, as demonstrated by comparable examples, e.g., *dema* 'saying'; *korma* 'seeing'. Nevertheless, this marker will not be categorized as a converb marker in the present discussion, as its employment is ascribed to the impact of Biblical Hebrew. This influence adheres to the

source text, rather than exemplifying a generative application, for which additional instances could be identified.

(166) [Lev 1:1] B.Heb: *lê.mōr* Preposition-l.to-say.V:QAL.INF.CONS

A-) {-**A**}; {-**y**}: The marker under investigation is manifested as {-A} in verbs with terminal consonants in Crimean Tatar (Kavitskaya 2010: 77), Ottoman Turkish (Kissling 1960: 95–96), and Karaim dialects (Prik 1976: 122; Musaev 1977: 61). Conversely, following verbs with final vowels, it adopts the shape of {-y} in Crimean Tatar and Karaim dialects, while it is presented as {-yA} in Ottoman Turkish.

Within the analyzed data, a sparse two lexicalized items were initially marked by the converb marker under investigation. The first example is the postposition *görä*, representing 'according to' and derived from the verb *gör* 'to see'. The postposition assimilated the converb marker {-ä}. Another instance discovered in the data is the solidified verbal adjective *ayayturyan*, which conveys the meaning of 'flowing'. This word is composed of the verb *ak* 'to flow', followed by the converb marker {-a}, subsequently {-y}, and the auxiliary verb *tur*- with the participle {-yan}. It is likely that this structure originated from an inadvertent error, as it encompasses both the converb marker {-a} and the {-y}. The participle *ayayturyan* is found in Crimean Karaim (CKED: 34; KRPS: 43), while in Western Karaim, the simultaneous use of two converb markers is not present in the equivalent form of the aforementioned term, e.g., [Lev 20:24] H.Kar (J.Sul.III.01) *aktiradoyan*; T.Kar (ADub.III.73): *ayadoyan* 'flowing'.

B-) {-**Ip**}; {-**Up**}: This morpheme is frequently observed in Turkic languages, typically represented as {-(X)p}, such as in Crimean Tatar {-(I)p} (Kavitskaya 2010: 77), Crimean Karaim {-(I)p} (Prik 1976: 121), Trakai Karaim {-(I)p}, {-(U)p}, and Halich Karaim {-(I)p}, {-(u)p} (Musaev 1977: 61). In Ottoman Turkish, however, the forms {-(y)Ip} and {-(y)Up} are encountered (Hagopian 1907: 208). Both of these variants are also present in Crimean Turkish, exemplified by *sävläyip*, meaning 'saying', and its synonymous term, *sävläp* (Doerfer 1959a: 278).

The dataset includes 20 instances of this particular converb marker. Nevertheless, since all verb stems in these examples conclude with a consonant, it remains unclear whether the marker would incorporate the linking -y-, which is solely identifiable in verbs with vowel-ending stems. In the majority of cases within the dataset, the markers manifest as {-Ip}.

(167) [Lev 1:3] al-ip to take-CONV 'taking'.

| [Lev 8:33] toldur- | | to fill-CONV | 'filling'. | |
|--------------------|--------|---------------|-------------|--|
| [Lev 11:24] | tiv-ip | to touch-CONV | 'touching'. | |

Notably, in a pair of verbs (appearing five times in total) within our dataset, it has been observed that the converb markers undergo vowel rounding harmony, a characteristic that has not been documented for Crimean Tatar, Crimean Turkish, and Crimean Karaim.

2.2.3.6. Participles

A-) {-(**A**)**r**}: Our dataset features a single lexical item that appears six times, in which the unproductive participle marker {-(A)r} is demonstrable. This marker is also found in Crimean Tatar (Jankowski 2010: 190–191), Ottoman Turkish (Kissling 1960: 81), and all three dialects of Karaim (Zajączkowski 1932: 99–101; Çulha 2019: 132).

B-) {-(I)včI}; {-(u)vči}; {-iǯi}: Participles formed with these markers typically convey properties related to the subject or object they refer to. Analogous variants of these markers have been documented in several other Turkic languages, e.g., C.Tat {-(U)vŠI}, {-IŠI} (Jankowski 2010: 190, 221); C.Kar {-(I)vŠI}, {-(I)vŠU} (Öztürk 2019: 36); Ot.Tur{-(y)UŠU}, {-(y)IŠI} (Zajączkowski 1932: 97–98); T.Kar {-(U)včU}, and H.Kar {-(u)vcu}, {-(i)vci} (Zajączkowski 1932: 61–62; Musaev 1964: 308). In the entire Lev of the Göz. 1841, two instances of the same lexical item were identified, featuring the marker {-(ï)ǯi}.

Nonetheless, the text also exhibits forms such as {-(I)včI} and {-(u)včï}, fulfilling the same function. In several instances, rounding harmony was solely observed in their connective vowels.

| [Lev 20:8] | ayruḫsï et-ivči | to sanctify-PART |
|-------------|-----------------|---------------------------|
| | | 'the one who sanctifies'. |
| [Lev 24:18] | ur-uvči | to kill-PART |
| | | 'the one who kills'. |

As previously stated (refer to Section 2.2.1.1.1.2), in a significant number of instances, this marker functions as a deverbal formative marker for nominals.

C-) {-(y)A¾AK}: It is employed five times throughout the entire text and can be found in Crimean Tatar (Kavitskaya 2010: 77) and Ottoman Turkish (Hagopian 1907: 187). However, it is not present in Western Karaim and Prik's Crimean Karaim Grammar¹⁰⁷.

| (179) | [Lev 11:47] | aša-l-aǯaķ | to eat-PASS-PART 'the thing that is going to be eaten'. |
|-------|-------------|----------------|---|
| | [Lev 14:4] | temiz bol-aǯaķ | to be clean-PART 'the one who will be cleansed'. |
| | [Lev 25:22] | gel-eǯäk | to come-PART 'the thing that is going to be come in'. |

In one example we find the negation marker {-ma}, that is attached to the verb stem, where the connective -y- in {-(y)AǯAK} also appears.

D-) {-QAn}; {-(y)An}: The participle {-QAn} is prevalent in a multitude of Turkic languages, encompassing the central and northern dialects of Crimean Tatar (Kavitskaya 2010: 76), Crimean Turkish (Doerfer 1959a: 279), and all three dialects of Karaim (Musaev 1964: 304; Prik 1976: 118). In fact, an Oghuzic variant of this morpheme exists, materializing as {-(y)An} in Ottoman Turkish (Kissling 1960: 90), the Southern dialect of Crimean Tatar (Kavitskaya 2010: 76), and Crimean Turkish (Doerfer 1959a: 279). Mirroring the situation in Crimean Turkish, our data showcases both variants, thereby highlighting the heterogeneous nature of

¹⁰⁷ Nonetheless, Culha (2019: 135-136) documents this marker, drawing upon Crimean Karaim mejumas as a source.

this participle. It is important to note that these markers signify entities engaged in or subjected to past or continuous actions¹⁰⁸.

In the dataset at hand, the participle {-QAn} manifests itself a cumulative total of 297 times throughout all chapters, with the sole exception of Chapter 1. Predictably, this form is created in accordance with both vowel and consonant harmony when appended to verb stems.

As previously elaborated, the participle {-(y)An} represents the Oghuzic equivalent of the Kipchak participle marker {-QAn}. This variant is identified a total of 27 times (involving 16 distinct verbs) within the text, rendering the Kipchak variant significantly more prevalent than this specific Oghuzic participle marker (92% to 8%). Furthermore, the Oghuzic feature appears in chapters 1, 2, 4, 11, 21, 23, and 27, with Chapter 11 exhibiting remarkable results, as nearly 75% of the examples (20 instances) are found in that particular chapter.

E-) {-mIš}: This postterminal participle is discernible in the Eastern dialect of Crimean Tatar (Jankowski 2010: 189), Crimean Karaim (Prik 1976: 118), and Ottoman Turkish (Kissling 1960: 89–90). Contrastingly, it is absent in Western Karaim as per Musaev's accounts (1964: 302–309; 1977: 62) ¹⁰⁹. Within our data, a sole example of the participle marker {-miš} is presented.

-

¹⁰⁸ Pertaining to the deverbal nominal formative functions of the marker {-QAn} within our dataset, see Section 2.2.1.1.1.2.

¹⁰⁹ Nevertheless, Gülsevin (2016: 99–100) records the presence of this marker within the Trakai Karaim.

(175) [Lev 11:37] *ek-il-miš ekin* to sow-PASS-PART crop

'the crop which has been sowed'.

F-) {-**DIK**}: In our dataset, this participle invariably follows possessive markers, during which the consonant k/k consistently undergoes voicing to y/g (with the exception of verbs containing the 3PL.POSS marker). It is documented in the Eastern dialect of Crimean Tatar (Jankowski 2010: 193) and Ottoman Turkish (Hagopian 1907: 187). Nonetheless, it is unattested in Prik's Crimean Karaim grammar and Western Karaim. Within our dataset, all occurrences are observed in the context of 3SG.

(176) [Lev 7:16] yuvuklaštir-diy-i gün+dä

to offer-PART-3SG.POSS day+LOC

'on the day that he offers'.

[Lev 7:36] sil-dig-i gün+dän

to anoint-PART-3SG.POSS day+ABL

'from the day that (the Lord) anointed'.

2.2.3.7. Verbal Nouns

A-) **{-mA}:** The verbal noun marker **{-mA}** is frequently observed in Turkic languages, with evidence present in Crimean Tatar (Jankowski 2010: 191–192), Ottoman Turkish (Kissling 1960: 90–91), and all three dialects of Karaim (Musaev 1964: 294–296; Prik 1976: 116).

(177) [Lev 11:21] sičra-ma to leap-VN 'leaping'.

[Lev 25:38] *ver-mä* to give-VN 'giving'.

B-) {-mAK}: The infinitive marker {-mAK} is quite common starting from early stages of Turkic languages (Johanson 2022c: 138), including Crimean Tatar (Kavitskaya 2010: 75), Ottoman Turkish (Kissling 1960: 68–69) and all three dialects of Karaim (Musaev 1964: 296; Prik 1976: 115).

(178) [Lev 8:15] et- $m\ddot{a}k$ + $k\ddot{a}$ to make-INF-DAT 'to make'.

[Lev 26:37] tur-mak to stand-INF 'to stand'.

2.2.4. Conjunctions

Our analysis has identified a variety of conjunctions present in the text. Notably, these conjunctions are predominantly of foreign origin, a characteristic shared among many Turkic languages, such as Crimean Tatar (Jankowski 2010: 268–272), Ottoman Turkish (Hagopian 1907: 230-233), and three dialects of Karaim (Musaev 1964: 321–327; Prik 1976: 157–158).

A significant feature observed in the use of conjunctions in Karaim Bible translations relates to the interpretation of the Biblical Hebrew conjunction - [[və]], which conveys meanings such as 'and', 'but', 'also', and 'then' (LVTL: 244–246). In Karaim translations, this Hebrew conjunction is represented by *da*, primarily defined as 'and', 'too', or 'also' (CKED: 130). However, due to the literal translation approach, *da* also functions in the mentioned meanings. The high frequency of this conjunction in the original text is reflected in our dataset, with a total of 1,761 instances recorded.

Below, we showcase the conjunctions identified in our dataset, categorized based on their respective functions.

A-) Connective

| (179) [Lev 1:1] | | da | 'and; then; also'. |
|-----------------|-------------------|--------------|---------------------|
| | [Lev 11:3; 11:7] | ve; vä | 'and'. |
| | [Lev 18:7; 19:23] | yaʻne; yaʻni | 'that is (to say)'. |

B-) Disjunctive

| (180) | [Lev 27:12] | geräk geräk | 'whether or'. |
|-------|-------------|-------------|----------------|
| | [Lev 7:26] | ne ne (da) | 'neither nor'. |
| | [Lev 10:7] | olmaya | 'or else'. |
| | [Lev 21:2] | ya (ya) | 'either or'. |

C-) Adversative

D-) Conditional

E-) Causal

[Lev 11:44] *ki* (of Hebrew origin) 'for; since; because'.

2.3. Syntax and Morphosyntax

This section will succinctly outline the syntactic and morphosyntactic characteristics of our dataset. It's important to note that all translations in the Karaim Bible consistently exhibit a common trait: they strictly adhere to the word and sentence order of Biblical Hebrew and show numerous instances of code copying (see Kowalski 1929: xxxviii-xxxix; Pritsak 1959: 338–339; Musaev 2004; Csató 2011; Olach 2013: 151–185;). It is worth noting that some of these properties do not appear in the spoken language (Musaev 1977: 69), while most of them can also be attributed to Slavic influence due to areal contacts (Csató 2011: 177–180). Given that a detailed analysis of the influence of Biblical Hebrew would not serve as a distinct feature to determine the language of our dataset, this study does not delve into an exhaustive examination of its impact¹¹⁰. Instead, it provides a brief overview of the general characteristics. Finally, in this section, we primarily provide examples from Biblical Hebrew with English glosses, except for the parts that require further examination.

2.3.1 Phrases

2.3.1.1. Nominal Phrases

The word order within nominal phrases partly exhibits Turkic traits. Among these traits, a commonly encountered Turkic order pattern in a nominal phrase can be seen: 'cardinal

¹¹⁰ For a comprehensive analysis of the influence of Biblical Hebrew on a Karaim Bible translation, refer to Olach (2013).

number+adjective attribute+head' (Johanson 2022a: 43), e.g., [Lev 8:26] bir kalin mača 'one unleavened cake'. Note that due to the meticulous rendering of the Hebrew Bible's definite article [ha] by the Karaim demonstrative pronoun ol (see section 2.2.2.5.2.), an atypical order also frequently appears in such phrases, e.g., [Lev 14:12] ol bir koy 'one lamb'; lit. 'the one lamb'. Among the typical Turkic characteristics, we observe the compound pattern of 'noun+noun+3SG.POSS' (Johanson 2022a: 44), e.g., [Lev 4:8] hatat buyasi' 'the bull of the sin offering'. However, a notable deviation in the genitive constructions within the text is readily apparent. In Turkic languages, the element order within a genitive construction is 'possessor+possessed item' (Johanson 2022a: 44). However, the examined corpus predominantly exhibits the reverse order. This distinctive feature can be ascribed to both Hebrew and Slavonic influence (Kowalski 1929: xxxviii-xxxix; Csató 2011: 177–178; Olach 2013: 153–154).

Table 65: Inverse Order of the Genitive Constructions

| Lev | | Hebrew Bible |
|-----|-------------------------|--------------|
| 9:6 | kə <u>b</u> ōw <u>d</u> | Yahweh |
| | glory-of | Jehovah |
| | | Göz. 1841 |
| | kavod+ï | YWY+nïŋ |
| | glory+3SG.POSS | YWY+GEN |
| | 'the glory of the Lord' | |

Upon analysis of various Bible translations in Turkic languages, we do not typically observe this untypical Turkic characteristic in the translation of Lev 9:6, e.g., Cr.Tat.B, *Rabbinin šureti* 'the glory of the Lord', Ot.Tur.B, *Rabbinj želali* 'the glory of the Lord'. However, in Western Karaim Bible translations, we see the influence of Biblical Hebrew word order, as demonstrated in H.Kar.B *šekinasi H-nin*, T.Kar.B *šeķinasi Adonay*{*nin*} 'the divine presence of the Lord'.

Of interest is the observation that our dataset contains a small number of infrequent instances where we observe the expected Turkic order of genitive constructions, as exemplified by [Lev 26:22] $t\ddot{u}z+ni\eta$ kiyi[k>g]+i; field+GEN animal+3SG.POSS 'animals of the field'. Beyond such isolated examples, our text notably exhibits the prevalent feature of Karaim Bible translation.

_

¹¹¹ Besides, noun phrases that function as appositives also present an inverse order, e.g., [Lev 1:5] *oylanlarï Aharonnïŋ ol kohenlär* 'the sons of Aaron the priests'.

Finally, a distinctive feature observed within our data is clearly showing the influence of Biblical Hebrew. For emphasis in Biblical Hebrew, the reduplication of certain nominals and nominal phrases¹¹² frequently creates universal quantification (see Waltke and O'Connor 2003: 116–117), which is typically indicated by specific quantifiers (see 2.2.2.5.6.) in Turkic languages. However, in our text, we encounter instances where this characteristic was replicated faithfully, adhering strictly to the original structure.

Table 66: Biblical Hebrew Reduplication 1

| Lev | | | | | Hebre | w Bible | | |
|------|-----------------|------------|-------------------|----------|-------|-------------------|---------|--|
| 17:3 | 7:3 'îš 'îš | | | | | | | |
| | man | | man | | | | | |
| | | | | | Göz. | 1841 | | |
| | kiši | | kiši | | | | | |
| | man | | man | | | | | |
| | 'any n | nan'. | | | | | | |
| 24:8 | | | | | Hebre | w Bible | | |
| | bəyōw | rm | haššabbā <u>t</u> | bəyōwn | ı | haššabbā <u>t</u> | | |
| | in day | of | the sabbath | in day o | of | the sabbath | | |
| | | | | | Göz | .1841 | | |
| | \overline{ol} | šabat | gün+de | ä | ol | šabat | gün+dä | |
| | the | sabbath | day+L0 | OC | the | sabbath | day+LOC | |
| | on ev | ery Sabbat | th day' | | | | | |

Upon examining the translation of Lev 17:3 and Lev 24:8 in some other Turkic Bible translations, it is notable that this particular feature of reduplication is not observed in either the Crimean Tatar or Ottoman Turkish versions, e.g., Cr.Tat.B *birev* 'someone'; *er raatliķ kunü* 'in every freeness day'; Ot.Tur.B *her adam* 'every man'; *her sebt gününde* 'in every Sabbath day'. Interestingly, this pattern is also absent in the Western Karaim Bible translations of Lev 17:3, e.g., H.Kar.B *har kisi* 'every man', T.Kar.B *nendiy kiši* 'any man'. However, an examination of Leviticus 24:8 reveals a similar pattern to our dataset, where reduplication of noun phrases occurs, interestingly in the Trakai Karaim Bible, it appears alongside the quantifier, e.g., H.Kar.B *ol šabat künde ol šabat künde*; T.Kar.B. *har šabat künündá har šabat künündá*. However, it is also noteworthy that these reduplications are also present in the so-called BSMS

¹¹² In one instance, we also observe this feature within a postpositional construction (refer to Section 2.3.1.2.2).

288 manuscript, another Crimean Karaim Bible translation, e.g., [Lev 17:3] *kiši kiši* 'any person'; [Lev 24:8] *ol šabat kündä ol šabat kündä* (CrKB I: 196, 209).

2.3.1.1.1. Case Functions

In this chapter, we aim to demonstrate the semantic functions of the case markers (see section 2.2.2.3.1.3) within our dataset. Note that petrified forms and specific verbs which require certain case governance will not be analyzed in depth. Instead, the general characteristics will be demonstrated. Given the scarcity of sources that offer an in-depth analysis of case functions in Karaim, Crimean Tatar, and Ottoman Turkish collectively, our exploration will be based on specific biblical verses from translations in these languages. However, it is critical to acknowledge that due to varied translational strategies, these verses might not perfectly encapsulate the true linguistic characteristics of the aforementioned languages. Consequently, our primary objective is to discern whether the functions observed in our data are unique to these translations, are a reflection of the structures found in Biblical Hebrew present exclusively in Karaim translations, or are widespread across these languages. Therefore, the descriptions provided herein do not constitute an exhaustive inventory of the case functions in Crimean Tatar, Ottoman Turkish, or Western Karaim. Instead, they aim to depict the characteristics of our data in relation to case functions. Consequently, during the comparisons, some biblical verses from the Crimean Tatar, Ottoman Turkish, and Western Karaim translations that are not suitable for specific contexts will not be included. However, this does not imply the absence of such functions in these languages, unless explicitly discussed in detail.

A-) Nominative Case: It precedes the following:

(i) subject

See also, [Lev 8:5] Cr.Tat.B. *Musa žemaatķa böyle dedi*' 'Moses said to the congregation as such'; Ot.Tur.B. *Musa cemaate dedi* 'Moses said to the congregation'; H.Kar.B. *ayttī Moše*; Tkar.B. *ayttī Moše* 'Moses said'.

(ii) subject complement:

sister+2PL.POSS+COP

she

'she is your sister'.

See also, [Lev 18:11] Cr.Tat.B; kiz kardašin; Ot.Tur.B senin kız karındašindir 'she is your sister'; H.Kar.B; atannin tuvduyundu senin ol; T.Kar.B ataynin tuvduyundu ol 'she is born of your father'.

(iii) indefinite direct object

(186) [Lev 10:1]

yuvuklaštir-dï-lar ald+ $\ddot{\imath}$ +na YWY+n $\ddot{\imath}\eta$ yat ot to bring forward-PAST-3PL before+3SG.POSS+DAT Lord+GEN strange fire

'they brought forward strange fire before the Lord'.

See also, [Lev 10:1] Cr. Tat.B kokulï otlar koydïlar 'they put fragrant fire'; [Lev 1:6] Ot. Tur.B ateš koyalar 'they shall put fire'; [Lev 10:1] H.Kar.B yuvuttular alnïna H-nïn yat ot 'they brought forward strange fire before the Lord'.

(iv) complement of certain postpositions

(187) [Lev 6:10] **ašam** gibi

guilt offering like 'like the guilt offering'

See also, [Lev 6:10] Cr. Tat.B *kabaat ičün kurban kibi*; Ot. Tur. B *suč ičin kurban gibi* 'like a sacrifice for **guilt**'; [Lev 10:19] H. Kar. B *bular kibik*; T. Kar. B *bular kibik* 'like these'.

(v) temporal expressions

(188) [Lev 8:35] otur-uŋïz gün da gečä

to sit-2PL.IMP day and night

'you shall sit day and night'.

See also, [Lev 8:35] Cr.Tat.B *geǯe-küŋdüz Rabbiniŋ bekčileri oluŋïz* 'you shall be God's watcher, **day** and **night**'; Ot.Tur.B *geǯe ve gündüz durasïz* 'you shall remain day and night'; H.Kar.B *olturunuz kün da kece*; T.Kar.B *olturuyuz kün da kece* 'you shall sit day and night'.

(vi) adressing indirective commands

(189) [Lev 19:29] *az-ma-sin* ol **yer** *halk+i*

to be mistaken-NEG-3SG.VOL the place people+3SG.POSS

'the people of the land shall not be mistaken'

See also, [Lev 19:29] Ot. Tur. B *vilayet zina etmeye* 'the county shall not commit adultery'; H. Kar. B *azmasin eli ol yernin*; T. Kar. B *azmasin eli ol yernin* 'the people of the land shall not not fall into prostitution'.

(vii) elements of compound nouns

(190) [Lev 25:9] *kipurim günü* atonement day+3SG.POSS 'day of atonement'.

See also, [Lev 25:9] Cr.Tat.B *ödeme künü*; Ot.Tur.B *kefaret günü* H.Kar.B *bosatliklar kün*; T.Kar.B *kipurim künü* 'day of atonement'.

(viii) possessed item of a genitive construction

(191) [Lev 17:5] oylan+lar+ï Yisra'el+nin
son+PL+3SG.POSS Israel+GEN 'sons of Israel'.

See also, [Lev 3:2] Cr.Tat.B kurbaninin bašī 'head of his sacrifice'; [Lev 5:9] Ot.Tur.B kurbaninin kanī 'the blood of his sacrifice'; [Lev 17:5] H.Kar.B uvullarī Yisra'elnin; T.Kar.B ulanlarī Yisra'elnin 'sons of Israel'.

(ix) quantifier-preceded relative pronouns introducing free relative clauses

(192) [Lev 27:9] her ne ki ver-sä-Ø andan every what that to give-COND-3SG 3SG.DEM.ABL

'any such that one gives from it'

See also, Ot.Tur.B *her ne ki Rabbe ondan verirse* 'any such that one gives from it'; H.Kar.B *barca ne ki niyet etse berme andan* 'any such that one wishes to give from it'; T.Kar.B *barča ne ki berśa andan* 'any such that one gives from it'.

(x) the interrogative pronoun *ne* in question sentences where the entity is unspecified

(193) [Lev 25:20] *ne aša-lim* **what** eat-1SG.VOL 'what shall we eat?'.

See also, [Gen 30:31] Cr.Tat.B *ne bereyim saŋa*? 'what shall I give you?'; [Lev 25:20] Ot.Tur.B *ne* yeyelim; H.Kar.B *ne* asarbiz; T.Kar.B *ne* ašarbiz 'what shall we eat?'.

(xi) aim or intention in adverbial clauses in verbals

(194) [Lev 14:29]

ver-sin $ba\check{s}+\ddot{\imath}$ $\ddot{u}st+\ddot{u}+n\ddot{a}$ olto give-3SG.VOLhead+3SG.POSSover+3SG.POSS+DATthe

temiz bol-yan+nïn kapara et-mä üst+ü+nä

to be clean-PN+GEN to make atonement over+3SG.POSS+DAT

'he shall put on the head of the one who is to be cleansed, to make atonement for him'

This structure is not common in Ottoman Turkish or Crimean Tatar, unlike in Karaim, e.g., [Lev 14:29] H.Kar.B *berin basy istine ol arïnuvcunun bosatlīķ kolma anïn ücün*; T.Kar.B *bersin bašī üstūńa ol arïnuvčunun bošatlīķ kolma anïn üčün* 'he shall put on the head of the one who is to be cleansed, to make atonement for him'.

B-) Accusative Case: It precedes the following:

(i) direct definite object

(195) [Lev 1:4] $taya-s\ddot{i}n$ $kol+u+n\ddot{i}$

to prop up hand+3SG.POSS+ACC 'he shall prop up his hand'

See also, [Lev 1:4] Cr.Tat.B kolunï koysun 'he shall put his hand'; [Lev 3:2] elini vaz' ede 'he shall put his hand'; H.Kar.B sunsun kolun; T.Kar.B sunsun ošol kolun 'he shall reach out his hand'.

(ii) relative pronouns which are followed by restrictive relative clauses

(196) [Lev 22:15] ne+ni ki $ay\"{i}r$ -sa-lar

what+ACC that to separate-COND-3PL

'which they separate...'

See also, [Lev 22:15], H.Kar.B *neni* ki ayirsalar 'which they separate'; [Lev 19:34] T.Kar.B *neni* ki siivasen 'what would you love'.

(iii) complement of certain postpositions

(197) [Lev 19:34] sev-gin anï gendin+ni gibi to love-2SG.IMP 3SG.PRO.ACC yourself+ACC as

'you shall love him as |you do| yourself'.

See also, [Lev 19:34] Ot.Tur.B *onu kendini gibi sevesin* 'you shall love him as |you do| yourself'.

C-) Genitive Case: It precedes the following:

(i) possessed item of a genitive construction

(198) [Lev 3:13]
$$oylan+lar+\ddot{\imath}$$
 Aharon+ $n\ddot{\imath}\eta$

See also, [Lev 3:13] Cr. Tat.B *Ḥarunniŋ oyullari*; Ot. Tur.B *Harunuŋ oğullari*; H.Kar.B *uvullari Aharonnun*; T.Kar.B *uvullari Aharonnun* 'sons of Aaron'.

(ii) the modifier in constructions indicating temporal expressions

(199) [Lev 23:24]
$$bir+i+n+d\ddot{a}$$
 ol $ay+n\ddot{i}\eta$

one+3SG.POSS+PRNM+LOC the month+GEN

'on the first |day| of the month'.

See also, [Lev 23:24] Cr.Tat.B; ay**ïŋ** birinǯi kününde; Ot.Tur.B ay**ïŋ** evvelkisinde; H.Kar.B burunyu küninde ay**nïn**; T.Kar.B burunyu kününda yanyay**nïn** 'the first day of the month'.

(iii) subject complement

(200) [Lev 25:23]
$$men+im+dir$$
 ol

1SG.PRO+**GEN**+PRD that 'it is mine'.

See also, [Lev 25:23] Cr.Tat.B *menimkidir*; Ot.Tur.B *benimdir*; H.Kar.B *menimdi* ol; T.Kar.B *menimdi* ol 'it is mine'.

(iv) complement of certain postpositions

(201) [Lev 22:5] a**nïŋ** bilän

DEM.PRO.GEN with 'with that'.

See also, [Lev 22:5] Ot. Tur. B onuŋla H. Kar. B anin bila ; T. Kar. B anin bila 'with it'.

D-) Dative Case: It precedes the following:

(i) physical terminus of a motion verb

(202) [Lev 14:34]

gel-sä- ηiz yer+i+näKena `an+niŋto come-COND-2PL.IMPplace+3SG.POSS+**DAT**Canaan+GEN

'If you come to the land of Canaan'.

See also, [Lev 14:34] Cr. Tat.B *Ḥanaan toprayina kirgeninizde*; Ot. Tur. B *Ken ʿan vilayetine girdiginiz zaman* 'when you enter the land of Canaan'; H. Kar. B *kelseniz yerine Kena ʿannin*; T. Kar. B *kelśayiz yerina Kena ʿannin* 'if you come to the land of Canaan'.

(ii) abstract/non-physical destination of a motion verb

(203) [Lev 3:7] $ald+\ddot{\imath}+na$ $YWY+n\ddot{\imath}\eta$

before+3SG.POSS+**DAT** Lord+GEN 'before the Lord'.

See also, [Lev 3:7] Cr.Tat.B; *Rabbinin önüne*; [Lev 4:4] Ot.Tur.B *Rabbin huzuruna*; [Lev 3:7] H.Kar.B *alnïna H-nïn*; T.Kar.B *alnïna Adonaynïn* 'before the Lord'.

(iii) direction of a motion verb

(204) [Lev 3:8]

 $ba\check{s}+\ddot{\imath}$ $\ddot{u}st+\ddot{u}+n\ddot{a}$ $\dot{k}orban+\ddot{\imath}+n\ddot{\imath}\eta$

head+3SG.POSS over+3SG.POSS+**DAT** sacrifice+3SG.POSS+GEN

'(prop up his hand) on the head of his offering'

See also, [Lev 3:8] Cr.Tat.B kurbaninin bašina; Ot.Tur.B kurbaninin baši üzerine; H.Kar.B basi istine karbaninin; T.Kar.B baši üstüńa karbaninin '(reach out/put his hand) on the head of his offering'.

(iv) reason, intended goal

(205) [Lev 1:3]

yuvukla \tilde{s} tir-sin ani kabullu[k>y]+i+na

to bring forward DEM.PRO.ACC acceptance+3SG.POSS+DAT

'he shall bring forward for his acceptance'.

See also, [Lev 16:17] Ot.Tur.B *kefaret etmeğe girdigi zaman* 'when he entered to make an atonement'; [Lev 1:3] H.Kar.B *yuvutsun anï kabulluk***ka**; T.Kar.B *yuvutsun anï kabulluk***ka** 'he shall bring forward for the acceptance'.

(v) adressee of a verbal communication

(206) [Lev 1:1]
$$\check{c}ak\ddot{i}r-d\ddot{i}-\emptyset$$
 $Mo\check{s}e+g\ddot{a}$

to call-PAST-3SG Moses+**DAT** 'And He called Moses'.

See also, [Lev 10:19] Cr.Tat.B; *Musaya böyle dedi* '(Lord) said that to Moses:'; [Lev 12:1] Ot.Tur.B *Musaya söyledi*; H.Kar.B *cayïrdï Mošege*; T.Kar.B *sözĺadi Adonay Mošega* '(Lord) spoke to Moses'.

(vi) the recepient/beneficiary of a material transfer

offering Lord+**DAT** 'an offering to the Lord'.

See also, [Exo 35:34] Cr.Tat.B *Rabbige baḥšiš* 'a gift to the Lord'; [Lev 1:2] Ot.Tur.B *Rabbe ķurban*; H.Kar.B *ķarban H-ya*; T.Kar.B *ķarban Adonayya* 'an offering to the Lord'.

(vii) temporal expressions

(208) [Lev 22:30]
$$ert\ddot{a}+g\ddot{a}$$
 $degin$

Morning+**DAT** until

See also, [Lev 22:30] Cr.Tat.B sabaya; Ot.Tur.B sabaha dek; [Lev 22:6] H.Kar.B ingirgedey; T.Kar.B tanyadejin 'until morning'.

(viii) complement of certain postpositions

DEM.PRO+**DAT** according to 'according to them'

See also, [Lev 18:4] Cr.Tat.B *olarya köre*; [Lev 25:51] Ot.Tur.B *onlara göre*; H.Kar.B *alarya köre*; T.Kar.B *alarya köra* 'according to them'.

(ix) end point of age range

(210) [Lev 27:3]

yegirmi yaš+tan da altmïs yaš+**ķa** degin twenty age+ABL and sixty age+**DAT** until

See also, [Lev 27:5] Ot.Tur.B *yigirmi yaša dek* 'until twenty years old'; H.Kar.B *altimis yas*kadejin; T.Kar.B *altmis yaškadejin* 'up to sixty years old'.

(x) exchange value

(211) [Lev 27:16]

da bol- $s\"{i}n...$ $urlu[k>y]+\ddot{i}$ $sab+u+n\ddot{i}\eta$

and to be-3SG.VOL seed+3SG.POSS stalk+3SG.POSS+GEN

arpa+lar+nïŋ elli miskal kümüš+**kä** barley+PL+GEN fifty shekel silver+**DAT**

See also, e.g., [Lev 27:16] Ot.Tur.B *elli miskal gümüše*; H.Kar.B *enli mitķal kimiske*; T.Kar.B *enli mitķal kümüška* '(it shall be) for fifty mithqals of silver.'.

(xi) reciprocal equivalence

(212) [Lev 24:20]

 $s\"{i}n\"{i}k$ $or(u)n+\ddot{i}+na$ $s\"{i}n\"{i}k$ $g\ddot{o}z$ $or(u)n+\ddot{i}+na$

Fracture place+3SG.POSS+**DAT** fracture eye place+3SG.POSS+**DAT**

 $g\ddot{o}z$ $di\check{s}$ $or(u)n+\ddot{\imath}+na$ $di\check{s}$

eye tooth place+3SG.POSS+DAT tooth

'_Lfracture for fracture¹¹¹³, _Leye for eye¹¹¹⁴, _Ltooth for tooth¹¹¹⁵'.

^{&#}x27;from twenty years old up to sixty years old'

^{&#}x27;A stalk of barley seed shall be fifty shekels of silver'.

¹¹³ Lit. 'fracture instead of fracture'. Probably a calque of B.Heb שֶׁבֶר תַּחָת שֶׁבֶר; lit. 'fracture instead of fracture'.

¹¹⁴ Lit. 'eye instead of eye'. Probably a calque of B.Heb עין תחת עין; lit. 'eye instead of eye'.

¹¹⁵ Lit. 'tooth instead of tooth'. Probably a calque of B.Heb שון תחת שון; lit. 'tooth instead of tooth'.

See also, [Lev 24:20] Ot.Tur.B; šikest yerine šikest göz yerine göz diš yerine diš; H.Kar.B süniķliķ süniķliķ ornuna köz köz ornuna tiš tiš ornuna; T.Kar.B süniķliķ süniķliķ ornuna köz köz ornuna tiš tiš ornuna 'ıfracture for fracture¹¹¹⁶, leye for eye¹¹¹⁷, looth for tooth¹¹¹⁸'.

(xii) Possessor

(213) [Lev 11:10]

her šey ki yok o+na kanat ve pul Every thing which there is no 3SG+DAT fin and scale

This function is not observed in Crimean Tatar and Ottoman Turkish. It is most likely a calque from Biblical Hebrew, and thus, it is present in Karaim translations, e.g., H.Kar.B *barca ne ki yoḥtu anar ḥanat da ḥabuḥ*; T.Kar.B *barča neki yoḥtur anar ḥanat da ḥabuḥ* "Leverything that has¹¹²⁰ no fins and scales'.

E-) Locative Case: It precedes the following:

(i) spatial adverbial

(214) [Lev 25:1] tay+i+n+da Sinay+niŋ Mountain+3SG.POSS+LOC Sinai+GEN 'on mount Sinai'.

See also, [Lev 25:1] Cr.Tat.B Sinay dayinda; Ot.Tur.B Sina dağinda; H.Kar.B tavinda Sinaynin; T.Kar.B tavinda Sinaynin 'on mount Sinai'.

(ii) temporal adverbial

(215) [Lev 6:13]

yarimisi+n ertä+dä da yarimisi+n aḥšam+da half+ACC morning+LOC and half+ACC evening+LOC

'a half of it in the morning, and a half of it at night...'.

^{&#}x27;leverything that **has**¹¹¹⁹ no fins and scales'.

¹¹⁶ Lit. 'fracture instead of fracture'. Probably a calque of B.Heb שֶבֶר תַּחָת שֶבֶר; lit. 'fracture instead of fracture'.

¹¹⁷ Lit. 'eye instead of eye'. Probably a calque of B.Heb עין תחת עין; lit. 'eye instead of eye'.

Lit. 'tooth instead of tooth'. Probably a calque of B.Heb שֶׁן תַּחָח שֶׁן; lit. 'tooth instead of tooth'.

¹¹⁹ Lit. 'everything which there is no to him'. Probably a calque of B.Heb כֹל אֲשֶׁר אֵין־לוֹ; lit. 'all which there is no to him'.

¹²⁰ Lit. 'everything which there is no to him'. Probably a calque of B.Heb כֹל אֲשֶׁר אֵין־לוֹ; lit. 'all which there is no to him'.

See also, [Lev 23:32] Ot.Tur.B ahšamda; H.Kar.B ingirde; T.Kar.B i{n}girdá 'at evening'.

(iii) positional/relational adverbial

(216) [Lev 6:22]

här erkäk kohen+lär+**dä** aša-sïn anï

every man priest+PL+LOC to eat-3SG.VOL DEM.PRO.ACC

'Every male among the priests shall eat it'.

See also, [Lev 6:22] H.Kar.B kohenlerde; T.Kar.B kohenlarda 'among the priests'.

(iv) possessor

(217) [Lev 13:29]

 $ki\check{s}i$ ya $hat\ddot{i}n$ ki bol-sa- \emptyset and $hat\ddot{i}k$ man or woman that to be-COND 3SG.PRO.LOC disease

See also, [Lev 22:20] Ot.Tur.B hič onda bir 'ayb olanı takrib etmeyesiz 'You shall not offer anything that has a blemish'; [Lev 13:29] H.Kar.B kisi yemese katin ki bolsa anda hastalik; T.Kar.B kisi yemesa katin ki bolsa anda hastalik 'If a man or woman has a disease'.

F-) Ablative Case

Ablative markers has many functions in our text. It signifies one of the following:

(i) departure locale in verbal movement

(218) [Lev 25:42]

čiy-ar-dï-m alar+nï Mïsïr yer+i+n+**dän**

to bring out-PAST-1SG 3PL.PRO+ACC Egypt place+3SG.POSS+**ABL**

See also, [Lev 25:42] Cr.Tat.B *olari Misir toprayindan alip čiyardim*; Ot.Tur.B *onlari Misir vilayetinden čiķardim*; H.Kar.B *ciyardim alarni yerinden Micrinin*; T.Kar.B *čiyardim alarni yerindan Micrinin* 'I brought them out from the land of Egypt'.

(ii) onset of temporal verbs

^{&#}x27;If a man or woman has a disease'.

^{&#}x27;I brought them out from the land of Egypt'.

See also, [Lev 24:3] Cr. Tat. B aḥšamdan sabayaǯe; Ot. Tur. B ahšamdan sabaha dek; H. Kar. B ingirden erten bïlayadeyin T. Kar. B ingirdan tanyadeyin 'from evening until morning'.

(iii) source of an action

(220) [Lev 5:21]

 $al-d\ddot{\imath}-\emptyset$ $i-s\ddot{a}-\emptyset$ $dost+\ddot{\imath}+n+dan$

to take-PAST-3SG CL-COND-3SG friend+3SG.POSS+PRNM+ABL

'if he has taken away |something| from his friend'.

See also, [Exo 11:2] Cr.Tat.B komšusindan... alsinlar 'they shall take from her/his neighbour' [Exo 11:2] konšusundan... isteye 'she/he shall ask from her/his neighbour'; [Lev 25:15] H.Kar.B satinalyin osol dostundan 'you shall buy from your friend'; [Exo 11:2] T.Kar.B ötkünč alsynlar kiši dostundan '|ever| person shall borrow from her/his friend'.

(iv) source material in composition

See also, [Lev 13:48] Cr.Tat.B *teriden yapülyan šeyde* 'in the thing that is made of leather' [Lev 13:52] Ot.Tur.B *eger yünden eger ketenden* 'whether is made of wool or linen'; [Lev 13:59] T.Kar.B *yundan yemeśa üskülüdan* '(a garment) of wool or linen'.

(v) portion/subset derived from a whole

(222) [Lev 14:14]

al-sïn ol kohen k̞an+ï+n+**dan**

to take-3SG.VOL the priest blood+3SG.POSS+PRNM+**ABL**

'the priest shall take **some of** the blood of...'.

See also, [Lev 1:2] Cr.Tat.B *tuvardan* '(you shall sacrifice) from the cattle; [Lev 14:14] Ot.Tur.B *kanïndan*; H.Kar.B *kanïndan*; T.Kar.B *kanïndan* '(he shall take some of) its blood'

(vi) subject matter of action

keparat et-sin
$$\ddot{u}st+\ddot{u}+n\ddot{a}$$
 ol kohen to make atonement-3SG.VOL over+3SG.POSS+DAT the priest

 $yaz\ddot{\imath}[\dot{k}>y]+\ddot{\imath}+n+dan$

sin+3SG.POSS+PRNM+ABL 'the priest shall make atonement for him concerning his sin'.

This particular feature is absent in both Crimean Tatar and Ottoman Bible translations, a contrast to its presence in Western Karaim Bible translations, e.g., H.Kar.B *bosatlik kolsun anin ücün ol kohen yaziyindan*; T.Kar.B *bošatlik kolsun anin ücün ol kohen yaziyindan* 'the priest shall make atonement for him concerning his sin'. Notably this function may represent a literal translation from the Hebrew Bible, e.g., [Lev 4:26] B.Heb: *mêḥaṭṭātōw* lit. 'from sin of him'.

(vii) object of comparison

See also, H.Kar.B *Da ol kohen ol ayarak karindaslarindan*; T.Kar.B *Da ol kohen ayarak karindaslarindan* 'the priest who is superior than his brothers'.

(viii) starting point of age range

(225) [Lev 27:6]

Da degin egär bir ay+danda beš väš+kä if month+ABL until And one and five age+DAT

'And if it is from a month old up to five years old'.

See also, [Lev 27:6] Cr.Tat.B bir ayliktan beš yašina kadar; Ot.Tur.B bir aydan beš yaša dek; H.Kar.B Da eger ay yastan da bes yaskadeyin; T.Kar.B Da eger ay yastan da beš yillaryadeyin 'And if it is from a month old up to five years old'.

(ix) complement of certain postpositions

(226) [Lev 14:8] andan sonra

DEM.PRO.ABL after 'according to them'

See also, [Lev 18:28] Cr.Tat.B *sizden evel* 'before you'; [Lev 14:8] Ot.Tur.B *ondan soŋra*; H.Kar.B *andan sortun*; T.Kar.B *andan sortun* 'after that'.

2.3.1.2. Verb Phrase

2.3.1.2.1. Biblical Hebrew Infinitival Paronomasia

In the context of Biblical Hebrew, a prevalent characteristic exists, characterized by the amalgamation of an infinitive absolute with a congruent finite verb form, a linguistic phenomenon recognized as infinivital paronomasia (Joosten 2009: 99; Waltke and O'Connor 2003: 585–586), highlighting the intensity of the verb in the sentence. Such a structure is similarly detected in Karaim Bible translations (Olach 2013: 146–147), featuring a non-finite verb inclusive of the verbal noun marker {-mA} preceding an identical verb in a conjugated form. This is typically ascribed to either Biblical Hebrew or Slavonic languages. However, since such structures exist¹²¹ in the oldest Karaim Bible translation (hitherto described), which is written in Crimean Karaim and dates back to the 17th century, the Slavonic influence solely might not be adequate to explain the phenomenon. Considering that the materials of spoken Karaim¹²² and secular texts (see Németh 2006: 18) also show the mentioned constructions, it is challenging to conclusively determine if it's solely a product of code copying from the Biblical source.

Notably, this linguistic phenomenon is also present in our dataset, occurring 31 times.

Table 67: Paronomastic Usage

| Lev | Hebrew Bible | | | |
|-------|-----------------------------------|--------------------------------------|--|--|
| 5:19 | āšōm āšam | | | |
| | to be guilty | he was guilty | | |
| | (to be punished.V:QAL.INF.ABS) | (to be punished.V:QAL.PERF.3SG.MASC) | | |
| | Göz. 1841 | | | |
| | günäḥli bol-ma | günäḥli bol-di-Ø | | |
| | to become guilty-VN | to become guilty-PAST-3SG | | |
| | 'he certainly has become guilty'. | | | |
| | | Hebrew Bible | | |
| 24:16 | mōw <u>t</u> | yūmā <u>t</u> | | |
| | to be put to death | he shall be put to death | | |

¹²¹ Personal communication, Assoc. Prof. Dr. Michał Németh.

1

¹²² Personal communication, Assoc. Prof. Dr. Michał Németh.

| (to die.V:QAL.INF.ABS) | (to die.V:HOP.IMPT.3SG.MASC) |
|------------------------------------|------------------------------|
| | Göz. 1841 |
| öl-mä | öl-dür-ül-sin |
| to die-VN | to die-CAUS-PASS-3SG.VOL |
| 'he shall surely be put to death'. | |

In Bible translations for languages that lack this specific structure, intensification is typically conveyed using an adverb, which often signifies 'certainly' or 'surely'. In fact, the opposition regarding this characteristic is observed between Karaim and other Turkic languages in Bible translations, e.g., [Lev 5:19], Cr.Tat.B kabaatli oldu 'he became guilty'; Ot.Tur.B elbette sučlu oldu 'he surely became guilty'; H.Kar.B. fasmanli bolma fasmanli boldu; T.Kar.B fašmanli bolma fašmanli boldu 'he surely became guilty; lit. 'becoming guilty guilty he became guilty'; [Lev 24:16] Cr.Tat.B mitlaka oldürilsin; Ot.Tur.B elbette öldürüle 'he surely be killed'; H.Kar.B elme eltirilsin; T.Kar.B ölma öltürülsün 'he surely be killed; lit. 'dying he should be killed'.

It is worth noting that our dataset contains two examples where the Hebrew Bible does not feature infinitival paronomasia, but our text does. Hereby, although rarely, our dataset does not show a strict parallel to Biblical Hebrew for this phenomenon¹²³.

Table 68: Paronomastic Usage 2

| Lev | | Hebrew Bible | | | |
|-------|-----------------------------|-------------------------------|--|--|--|
| 6:7 | haķrê <u>b</u> | | | | |
| | to bring near.V:HIP.INF.ABS | | | | |
| | | Göz. 1841 | | | |
| | yuvuklaštir-ma | yuvu <u>k</u> laštīr-sīn | | | |
| | to bring forwar | d-VN to bring forward-3SG.VOL | | | |
| | 'he shall define | tely bring forward' | | | |
| | | Hebrew Bible | | | |
| 25:14 | <u>t</u> imkərū | | | | |
| | to sell.V:QAL. | IMPT.2PL.MASC | | | |
| | | Göz. 1841 | | | |
| | satïn al-ma | satin al-saŋ | | | |
| | to buy-VN | to buy-COND-2SG | | | |
| | | | | | |

_

Remarkably, the presence of a similar structure in the Trabzon dialects of Turkey (Coşar 2015: 248–249) warrants further attention. This is especially significant considering the dialect's pronounced Kipchak influence, while it remains unaffected by the Hebrew Bible. Thus, further research might enlighten the remaining questions.

'if you certainly make a sale'.

However, due to deverbal nominal function of {-mA} (see 2.2.2.1.1.2.), it is uncertain whether these examples above truly demonstrate the phenomenon under discussion. This uncertainty persists despite the fact that such derived nominal forms (*satin alma*, *yuvuklaštirma*) are not listed in dictionaries.

2.3.1.3. Adjunct phrases

2.3.1.3.1. Converbial Constructions

Our text presents various converbial constructions, showing a pattern quite common in Turkic languages. These consist of action nominals, followed by possessive markers and case markers, preceding postpositions (see Johanson 2022a: 56). All the examples within our dataset function as time clauses. This type of construction forms the adverbial clauses (see 2.3.2.2.2.2.).

A-) {-dik}/{-dük}+POSS+LOC/ABL: In this structure, an action nominal suffix {-dik} or {-dük} is first attached to the verb stem, followed by personal possessive markers and then either by locative or ablative case markers respectively. Among these structures, those with ablative case markers are typically succeeded by postpositions denoting temporal relationships, such as *burun* 'before', or *soŋra/ sora* 'after'. Structures employing locative case markers generally signify 'when', while those with ablative case markers can denote 'before' or 'after', contingent upon the subsequent postposition. In our corpus, we have identified a total of 5 examples of this type of structure.

Table 69: Converbial Contructions

| Lev | Göz. 1841 | |
|-------|-----------------------------------|--------|
| 10:9 | gir-di[k>g]+iŋiz+dän | burun |
| | 'to enter-AN-2PL.POSS+ABL' | before |
| | 'before you go into' | |
| 11:31 | ül-dik+leri+n+dän | sora |
| | to die-AN-3PL.POSS-PRNM-ABL | after |
| | 'after they died' | |
| 11:32 | ül-dük+läri+n+dän | sora |
| | to die-AN-3PL.POSS-PRNM+ABL after | |
| | 'after they died' | |

| 16:17; | $gir-di[k>g]+i+n+d\ddot{a}$ |
|--------|-------------------------------|
| 16:23 | to enter-AN+3SG.POSS+PRNM+LOC |
| | 'when he goes in' |

It should be noted that this structure is commonly found in Ottoman Turkish (Kissling 1960: 186). However, in Karaim dialects and Crimean Tatar, we frequently encounter the following structure.

B-) {QAn}+POSS+DAT/LOC/ABL: These types of structures are identical to the previous ones, with one difference: the actional nominal is the Kipchak equivalent of Ottoman Turkish {-DUK}/{-DIK}, that is, {-QAN}. This feature is prevalent in Crimean Karaim (Kavitskaya 2010: 109) and the Karaim dialects (Musaev 1964: 306–307; Prik 1976: 120). Notably, examples within our data demonstrate that this structure can also include dative case markers, which are followed by specific postpositions that function as temporal boundary markers, indicating the duration or extent of an event up to a certain point in time, e.g., *kadar*; *degin* 'until'. As expected, this structure is much more common in our text, occurring 32 times and thus predominating against the structures that include the Ottoman Turkish characteristic (86% vs. 14%).

Table 70: Converbial Constructions 2

| Lev | Göz. 1841 | |
|-------|---------------------------------|-------|
| 15:31 | mundar et-kän+läri+n+dä | |
| | to defile-AN+3PL.POSS.PRNM.LOC | |
| | 'when they defile' | |
| 16:1 | öl-gän+i+n+dän | sonra |
| | to die-AN-3SG.POSS+PRNM+ABL | after |
| | 'after he died' | |
| 16:1 | yuvuklaš+kan+larï+n+da | |
| | to come forward-AN+3PL+PRNM+LOC | |
| | 'when they came forward' | |
| 23:14 | getir-gän+iŋiz+gä | degin |
| | to bring-AN+2PL.POSS+DAT | until |
| | 'until you (2PL) have brought' | |
| 23:43 | čïyar-yan+ïm+da | |
| | to bring out-ACT -LOC | |
| | 'when I brought (them) out' | |

C-) **{-mäs}+POSS+ABL:** In our dataset, we identify a structure that mirrors the construction of the previous ones, with the distinct difference that the action nominal manifests as **{-mäs}**.

Table 71: Converbial Constructions 3

| Lev | Göz. 1841 | |
|-------|------------------------------|--------|
| 15:31 | kel-mäs+i+n+dän | burun |
| | to come-AN+3SG.POSS+PRNM+ABL | before |
| | 'before he comes' | |

Notably in the translation of Lev 14:26 from other Turkic Bible versions, we observe a similar pattern, e.g., Cr.Tat.B, *kirmezden evel*; Ot.Tur.B, *girmezden evvel* 'before entering'; H.Kar.B *kelmesten burun*; T.Kar.B. *kelmastán burun* 'before coming'. Note that presenting the linear analysis of {-mäsi} as action nominal{-mäs}+3SG.POSS {+i} might be confusing due to its similar appearance to the possible combination of the verbal marker {-mä}+3SG.POSS {+si}. However, an example from another book of Göz.1841, clarifies the case, which clearly shows that the action nominal suffix is {-mAs} in such structures, e.g., [Gen 27: 4] *ölmäsimdän burun* 'before I die'.

2.3.1.3.2. Postpositional and Prepositional Constructions

In their native form, Turkic languages are known to be without prepositions. Nonetheless, in certain Turkic languages, one may observe fossilized lexical elements which include foreign prepositions. For example, despite the absence of prepositions in Crimean Tatar and Ottoman Turkish, the Arabic preposition $\[] [bi] \]$ meaning 'by' can be identified in specific words within these languages, such as Crimean Tatar $bilakis^{124}$ 'on the contrary' (KRUS: 111) and Ottoman Turkish $bizzat^{125}$ 'in person' (Hagopian 1907: 369). Conversely, the situation diverges in other Turkic languages like Karaim, Gagauz, and Khalaj, wherein the productive use of prepositions serves as an uncharacteristic Turkic syntactic feature due to external influences (Johanson

12

¹²⁵ The Arabic word includes several elements Arabic בוב [bi] 'by'+ לום [al] 'definite article (dat] 'person' = بالذات [bi'd-dat] 'in person'.

2022b: 114). It is important to emphasize that the presence of prepositions in Karaim is generally ascribed to both Slavonic and Biblical Hebrew influences ¹²⁶.

In our dataset, postpositions and prepositions can be primarily categorized into two main groups: those in their uninflected 127 form with case-marked complements, and those which are sets of auxiliary nouns marked by possessive markers and requiring case-marked complements¹²⁸.

A-) Uninflected Postpositions with Case Marked Complements

In our dataset, an assortment of postpositions is observed which are common in many languages with slight phonological changes in Crimean Tatar (Kavitskaya 2010: 81–84), Ottoman Turkish (Hagopian 1907: 104–110) and all three dialects of Karaim (Musaev 1964: 316–321; Prik 1976: 151–156). Notably, these postpositions appear in their basic form following complements that can be either case marked non-verbal forms or case marked non-finite verbs. The list below enumerates them:

Table 72: *Uninflected Postpositions with Case Marked Complements*

| Lev | Structure | Example | | |
|--------|--------------------------------|-------------------|-------------------|-----------------------------|
| 9:17 | COMP+ABL bašķa 'apart from; | 'ola+sï+n+dan | | bašķa |
| | aside from' | burn sacrifice+3 | SG.POSS+PRNM | beside |
| | | 'beside the burnt | sacrifice' | |
| 1:8; | COMP+NOM/GEN bilän; ilän; ilen | baš bilän | head with | 'with head'; |
| 11:43; | 'with' | olar ilän | them with | 'with them' |
| 11:21 | | | | |
| 18:2 | COMP+ABL burun 'before' | siz+dän | burun | |
| | | 2PL.PRO+ABL | before 'before | you (2PL)'. |
| 19:6 | COMP+DAT degin 'up to; till; | üč+ünži | gün+gä | degin |
| | until' | three+ORD d | ay+DAT | until 'until the third day' |
| 6:10 | COMP+NOM/ACC gibi 'like; as' | ḥatat gibi | sin offering like | 'like the sin offering' |
| 19:18 | | gendin+ni gibi | oneself+ACC lik | te 'like (you do) yourself' |

¹²⁶ However, it is also important to highlight that all the prepositions in our dataset are the result of inverse order of genitive constructions, which also appears in Armeno-Kipchak and the language of the Codex Cumanicus (see

Csató 2011: 171).

¹²⁷ It is important to recognize that some postpositions and prepositions may already exist in inflected forms. For example, göre (as previously mentioned in Section 2.2.2.5.), which means 'according to', is derived from the verb gör- 'to see' and is followed by the converb marker {-a}. In this context, such postpositions and prepositions are also considered as uninflected as they are in petrified forms.

¹²⁸ It is essential to highlight that, in the structures identified within our dataset, these case markers comprise dative, ablative, genitive markers, along with the nominative zero suffix.

| 25:51 | COMP+DAT görä 'according to; as | alar+ya | görä | |
|-------|---------------------------------|--------------|--------------|---------------------|
| | to' | they+DAT | according to | 'according to them' |
| 11:28 | COMP+DAT kadar | aḫšam+a | ķadar | |
| | 'up to; till; until' | evening+DAT | until | 'until the evening' |
| 25:15 | COMP+ABL sora; soŋra 'after; | yovel+dän | soŋra | |
| | following'. | 3SG.PRO+ABL | after | 'after the jubilee' |
| 4:3; | COMP+DAT/GEN učun; učin | günäḥ+i | učun | |
| 16:33 | 'for; for the sake of' | sin+3SG.POSS | for | 'for his sin' |

Interestingly, although prepositions are predominantly found in the following group (those marked by possessive markers), there is one exceptional instance in which a preposition in its petrified form appears before a relative pronoun, within the independent clause preceding the subordinate clause.

In fact, this preposition is classified as a postposition in dictionaries, exemplified by *deginče* 'up to, till' (CKED: 134). Taking into account the impact of Biblical Hebrew on Karaim Bible translations, this example might be ascribed to the structure of the original text¹²⁹.

B-) Auxiliary Nouns Marked by Possessive Markers with Case Marked Complements

However, many prepositions and postpositions can be traced back to sets of auxiliary nouns marked by possessive markers followed by case suffixes. Some of them exclusively appear as postpositions or as prepositions, while others can function as both. The following list offers an overview of these prepositions and postpositions that appear with case complements.

| (228) | * ald+POSS+CASE | 'before; in front of' | (POST; PREP) |
|-------|-------------------|--------------------------|--------------|
| | * ard+POSS+CASE | 'after; behind' | (PREP) |
| | * ara+POSS+CASE | 'between; among' | (POST; PREP) |
| | * karši+POSS+CASE | 'against; opposite side' | (PREP) |
| | * kat+POSS+CASE | 'side of' | (PREP) |

.

רביי The Hebrew relative pronoun אָשֶׁר ['ăšer], signifying 'which, who, that, because, when, or since', is often rendered by the Persian counterpart, ki, in our text (see 2.3.2.2.1.). The presence of a subordinate clause following the main clause and its postpositional nature within the sentence are atypical characteristics in Turkic languages. It is probably the slavish rendering of the original text, e.g., B.Heb 'ad 'ăšer yithār'; 'until which he is clean'.

The primary reason for the prepositional function of possessive-marked auxiliary nouns with case-marked complements lies in their fundamental role as elements of genitive constructions. These typically display an inverse order of the characteristic Turkic pattern (see section 2.3.1.1.) in our dataset. Consequently, as illustrated in the subsequent examples, the same auxiliary word might appear in either a prepositional or postpositional position, depending on the order within its genitive construction. Importantly, although quite rarely, our text does not consistently mirror the order of Biblical Hebrew genitive constructions.

(229) [Lev 18:18]
$$an+i\eta$$
 $ald+i+na$ 3SG.PRO+GEN before+3SG.POSS+DAT 'before her'. [Lev 18:23] $ald+i+na$ $tuvar+ni\eta$ before+3SG.POSS+DAT animal+GEN 'before animal'.

Finally, a phenomenon discussed in Section 2.3.1.1, which presents a reduplication of certain nominals for intensification, also appears in one example within our data. Here, a postpositional construction is reduplicated due to Biblical Hebrew influence.

Table 73: *Biblical Hebrew Reduplication* 2

| Lev | | | | Hebrew Bible | | |
|-----|--------------|-----------|---------|--------------|--|--|
| 6:5 | babbōķer | | babi | bōķer | | |
| | with morning | | with | n morning | | |
| | | Göz. 1841 | | | | |
| | ertä | bilän | ertä | bilän | | |
| | morning | with | morning | with | | |

As expected, such duplicates are not found in other Bible translations in Turkic languages, e.g., Cr.Tat.B *er saba*; Ot.Tur.B, *her sabah* 'every morning'. Interestingly, this duplication is also absent in Western Karaim Bible translations, e.g., H.Kar.B *har erten*, T.Kar.B *har ertenbülada*

'at every morning'. While once again in another Crimean Bible translation (manuscript BSMS 288) we find the same copy, e.g., *ertä bilän ertä bilän* 'every morning' (CrKB I: 173).

2.3.2. Sentences

2.3.2.1. Simple Sentences

Throughout the Book of Leviticus in Göz. 1841, the sentence syntax adheres strictly to the order of the original Hebrew Bible, an approach which is typically atypical in Turkic languages. The simple sentences in our dataset eschew the common Turkic word order (SOV) in favor of strictly following the original Biblical Hebrew order (VSO). This pattern is also frequently observed in other Karaim translations (Musayev 2002: 223; Olach 2013: 156).

Table 74: Word order in Simple sentences

| Lev | Hebrew Bible | | | | | | | |
|------|--------------|---------|------------|-----------------|-----|---------------|------------------|--|
| 8:10 | wayyiqqaḥ | | mōšeh | mōšeh 'eṯ-šemen | | hammišḥāh | | |
| | and took | | Moses | oil-of | | the anointing | | |
| | Göz. 1841 | | | | | | | |
| | da | al-dï | | Moše | ol | silmäk | yay+ï+nï | |
| | and | to take | e-3SG.PAST | Moses | the | anointing | oil+3SG.POSS+ACC | |

As anticipated, the analysis of the verse within Crimean Karaim and Ottoman Turkish Bible translations reveals the expected Turkish word order, e.g., Cr.Tat.B *Son Musa yaylav yayïnï aldï*, 'Then Moses took the anointing oil'; Ot.Tur.B *Hem Musa tedhin yağïnï alıp...* 'And Moses took the anointing oil...'. However, the Western Karaim translations exhibit the expected structure of Biblical Hebrew, e.g., H.Kar *Da aldy Moše osol ol sirtilmis javnï*; T.Kar *Da aldy Moše ošol jayïnmak javnï* 'And Moses took the anointing oil'. It should be noted that adherence to the structure of Biblical Hebrew is observed across all categories of simple sentences. The subsequent table illustrates this phenomenon specifically within the context of imperative sentences.

Table 75: Word order in Imperative Sentences

| Lev | | | Hebrew Bible |
|------|-------------|-------------|-----------------|
| 16:2 | dabbêr | 'el-'ahărōn | 'āḥî <u>k</u> ā |
| | speak (2SG) | to Aaron | brother of you |

| sözlägin | Aharon | ķardašiŋa |
|-----------|--------|-----------------|
| say (2SG) | Aaron | to your brother |

Similar to previous observations, we once again encounter the same method of translation in Karaim Bible translations, a style not seen in Crimean Tatar and Ottoman Turkish Bible translations, e.g., Cr.Tat.B harun ayana ayt; Ot.Tur.B karindašin Haruna söyle; H.Kar.B sözlegin Aharonya karindašina; T.Kar.B sözlagin Aharonya karindašiya.

Another noteworthy observation pertains to one of the most prevalent strategies employed in Karaim Bible translations, specifically the consistent rendering of the Biblical Hebrew [wa], meaning 'and, so, then, when, or, but, that' using the Karaim postpositive particle da 'and' (see Pritsak 1959: 338; Jankowski 1997: 22; Csató 2011: 179). Correspondingly, in our dataset, as well as in other Karaim Bible translations, the majority of sentences commence with da^{130} .

Table 76: Rendering of Biblical Henrew wa

| Lev | | | | | Hebrev | w Bible | |
|-----|-----------------------|----------|------------------------------|----------------|----------|---------------|-------------------|
| 8:9 | wayyā | iśem | hammişne <u>p</u> e <u>t</u> | ʻal- | rōšōw | | |
| | and he put the turban | | on | on head of him | | | |
| | | | | | Göz. | 1841 | |
| | da ķoy-dï-Ø ol sarīķ+ | | nï bašï | | üst+ü+nä | | |
| | and | to put-F | PAST-3SG the | turban | +ACC | head+3SG.POSS | over+3SG.POSS+DAT |
| | 'And l | | ne turban on his h | nead'. | | | |

As expected, a distinction between the Bible translations in other Turkic languages and Karaim Bible translations is apparent for this feature as well, e.g., Cr.Tat.B; bašīna čalmanī kiydirdi 'he put (lit. 'clothed) the turban on his head'; Ot.Tur.B ba 'dehu bašī üzerine 'imameyi kodu 'after that he put on his head'; H.Kar.B da koydu osol ol calmanī basī istine; T.Kar.B da koydu ošol ol micnefetni bašī üstüńa 'And he put the turban on his head'.

Finally, a significant characteristic is the rendering of the Biblical Hebrew conjunction ki for 'for; since; because' in its original form as ki (CKED: 210). This occurs in most sentences at the beginning of Karaim Bible translations (see Jankowski 1997: 22). It should be noted that

_

 $^{^{130}}$ As is demonstrated, only in Chapter 11 is this conjunction replaced with the Arabic ve, which was common in Ottoman Turkish and is still used in Turkish (see 2.2.4.).

this conjunction should not be confused with the homonymous Persian *ki*, which stands for 'which, that', and is common in many Turkic languages.

Table 77: Simple sentences using Biblical Hebrew word order

| Lev | | | | | Hebrew Bible |
|-------|--------|-----------|---------------|-----------------|--------------|
| 11:44 | kî | ʾănî | Yahweh | ʾĕlōhê <u>l</u> | <u>k</u> em |
| | for | I | Lord | your G | God |
| | | | | | Göz. 1841 |
| | ki | men+n | nin | YWY | Teŋri+ŋiz |
| | for | 1SG+1 | SG.PRD | lord | God+2SG.POSS |
| | 'For I | am the lo | rd your God'. | | |

As depicted in Table 77, the syntactical structure of Leviticus in Göz. 1841 once again reaffirms the adoption of the Biblical Hebrew word order, with the nominal predicate preceding the subject for emphasis. Contrarily, when compared to the Karaim Bible translations, Crimean Tatar and Ottoman Turkish translations present the expected Turkic order and also render the Hebrew conjunction *ki* either from their lexicons or omit it, e.g., Cr.Tat.B; *Men Rabbim* 'I am Lord'; Ot.Tur.B *zira Allahi'ŋi'z Rabb benim* 'Because I am the lord your God.'; H.Kar.B *Ki menmen H Tenriniz*; T.Kar.B *Ki menmen Adonay Tenriyiz* 'Because I am the lord your God.

2.3.2.2. Complex Sentences

2.3.2.2.1. Coordinate Complex Sentences

In such syntactic constructions, independent clauses are juxtaposed using certain conjunctions (see 2.2.4.). This practice is prevalent across various languages, including the Turkic language family. As mentioned earlier, due to the systematic method of rendering the Biblical Hebrew conjunction *wa* by Karaim da 'and', we encounter this type of sentence in the majority of cases.

(230) [Lev 15:13]

| Da | egär | temiz ķal-ïr-sa-Ø | | | ol | zav |
|---------|---------|----------------------------------|-------|--------|----|---------------------------|
| And | if | to remain clean-AOR-COND-3SG the | | | | the one who has discharge |
| aķma+ | -sï+n+a | lan | | da | | say-sïn |
| dischar | rge+3S | G.POSS+PRNM+ABL | | and | | to count-3SG.VOL |
| özünä | | | yedi | gün+la | är | temizli[k>g]+i |
| onesel | f+3SG. | POSS+DAT | seven | day+P | L | cleanness+3SG.POSS |

učun **da** yuv-sïn upraķ+lar+ï+nï

for **and** to wash-3SG.VOL dress+PL+3SG.POSS+ACC

da yuv-sïn et+i+ni tatlï suv+lar bilän

and to wash-3SG-VOL flesh+3SG.POSS+ACC sweet water+PL with

da temiz bol-sïn

and to be clean-3SG.VOL

Lit. 'And if the one who has a discharge remains cleansed of his discharge, and he shall count for himself seven days for his cleansing, and wash his clothes, and wash his flesh in sweet water, and he shall be clean.'

2.3.2.2.2. Subordinate Complex Sentences

2.3.2.2.1. Relative Clauses

The relative clauses used in our text can primarily be classified into two groups: those with Turkic characteristics and those with non-Turkic characteristics. In the first group, we encounter the use of participles, which is common in Turkic languages including Crimean Tatar, Ottoman Turkish, and Karaim variety (see 1.2.3.2.), in the dependent clauses (Johanson 2022a: 40).

(231) [Lev 14:11]

da turyuz-sïn ol temiz et-kän kohen kiši+ni ol and to set-3SG.VOL the to make clean-PN priest person+ACC the

temizlä-n-**gän**

to clean-PASS-PN

'And the priest who makes |him| clean shall set the man who is being cleaned'.

As for the other type, which is atypical for Turkic languages, we mostly encounter the systematic rendering of the Biblical Hebrew אָשֶׁר [ʾašer], meaning 'which, who, that, because, when, since', by the Persian conjunction ki. Note that in such a structure, the subordinate clause follows the main clause, which therefore presents untypical Turkic characteristics. It is also worth noting that this is common in Karaim, Krimchak, and Armeno-Kipchak (see Jankowski 2003b: 143–144).

Table 78: *Untypical Turkic Relative Clauses 1*

| 14 | wənōwdʻāh when becomes known | | haḥaṭṭā <u>t</u> | | 'ăšer | | ḥāṭə'ū |
|----|------------------------------|---------------|------------------|-------|-------|-------|--------------------------|
| | | | the sin | which | | | they have committed upon |
| | Göz. 1841 | | | | | | |
| | da | bil-in-sä-Ø | | ol | yazïķ | ki | yazïk̞lï bol-dï-lar |
| | and | to know-PASS- | -COND-3SG | the | sin | which | to be sinful-PAST-3PL |
| | anïŋ | | učun | | | | |
| | DEM. | POR.GEN | for | | | | |

The comparison of the translation of Lev 4:14 presents a contrast between Karaim and other Turkic languages discussed in this study, e.g., Cr.Tat.B *olarnin yapkan gunalari ačilyan vaķitta*; Ot.Tur.B *etdikleri günah ma 'lum olursa*; H.Kar.B *da bilinse ol yazik ki yazikli boldular*; T.Kar.B *da bilinsa ol yazik ki yazikli boldular* 'and if the sin which they have sinned is known'. As observed, both Ottoman Turkish and Crimean Tatar recurrently utilize participle markers in relative clauses, in contrast to the Karaim translation. It is noteworthy that such atypical structures often employ an interrogative pronoun as a relative pronoun (see Section 2.2.2.5.5), typically inflected with case suffixes and consistently followed by the Persian conjunction *ki*. Moreover, in verbal sentences that do not utilize the past tense, the verb embedded in the relative clause often bears the conditional marker {-sA}, which operates to signify hypothetical situations (refer to Section 2.2.3.4.4).

(232) [Lev 15:6]

da her ne ki otur-**sa**-Ø üst+ü+nä

and every what that to sit-COND-3SG over+3SG.POSS+DAT

mundar bol-sin

to be unclean 3SG.VOL 'And anything on which she sits shall be unclean'.

It should be highlighted that, along with other Karaim dialects, this unconventional structure was also prevalent in Ottoman Turkish, predominantly due to Persian influence. In fact, it was more prevalent in Old Ottoman (mostly in written sources) and began to be replaced by structures incorporating participle markers in Middle Ottoman (Kerslake 2022: 191), e.g., Ot.Tur.B; [Lev 5:16] *ne ki aldi ise* 'whatever he took'; [Lev 15:4] *her ne ki üzerinde oturursa* 'anything on which she sits'.

2.3.2.2.2 Adverbial Clauses

A-) Clause of Time

In such clauses, we frequently encounter converbial constructions as outlined in Section 2.3.1.3.1. These constructions are notably common in Turkic languages (Johanson 2022a: 40).

(233) [Lev 13:55]

da gör-sin ol kohen **yuv-ul-yan+ï+n+dan**

and to look-3SG.VOL the priest to wash-PASS-AN+3SG.POSS+PRNM+ABL

sonra ol hastalik+ni after the sickness+ACC

This common strategy is also apparent in Crimean Tatar (Kavitskaya 2010: 109–110), Ottoman Turkish (Kissling 1960: 181–182), and Karaim dialects (Musaev 1964: 306–307; Prik 1976: 120). In other examples, we simply encounter the use of certain postpositions appearing after the ablative case-marked complements, which is also common in the mentioned Turkic languages (see 2.3.1.3.2.).

(234) [Lev 23:40]

Da al- $i\eta iz$ $\ddot{o}z + \ddot{u}\eta iz + g\ddot{a}$ ol $burun \check{z}i$ $g\ddot{u}n + d\ddot{a}n$ And to take-2PL.IMP self+2PL.POSS+DAT the first day+ABL

burun yemiš+i+n sïylï ayač+nïŋ ḫurma+lar yaprak̞+lar+ï+nï before fruit+3SG.POSS+ACC choice tree+GEN date+PL leaf+PL+3PL.POSS

B-) Clause of Place

In the majority of examples of this type of clause, we encounter an atypical structure previously mentioned (see 2.3.2.2.2.1), where the Persian conjunction ki is used. In this structure, the relative pronoun ne appears before the word yer 'place', which is attached with a locative marker. This sequence is invariably followed by the Persian conjunction ki that introduce a clause describes the place.

^{&#}x27;And the priest shall look at the disease after it has been washed from it.'

^{&#}x27;And you shall take for yourselves before the first day the tree of choice fruits, the leaves of date palm'.

(235) [Lev 14:13]

Da soy-sin ol koy+ni ne yer+dä ki
And to slay-3SG.VOL the lamb+ACC what place+LOC that

soy-ar-Ø ol 'ola+nï

to slay-AOR-3SG the burnt offering+ACC

Note that this type is not attested in Crimean Tatar or Ottoman Turkish, as they use participle markers in the dependent clause, e.g., C.Tat.B kurban soyulyan erde 'in the place where the sacrifice is slayed'; Ot.Tur.B boğazlandiği yerde 'in the place where it is slaughtered'. However, Karaim Bible translations exhibit the mentioned structure, e.g., [Lev 14:13] H.Kar.B Da soysun osol ol kozunu ne orunda ki soysa osol ol hatatni; T.Kar.B Da soysun ošol ol kozunu ne orunda ki soysa ošol ol hatatni 'And he shall slay the lamb in the place where he slays the sin offering'.

In other examples, we usually find participle markers in verbs denoting the word *yer* 'place', which is common in Turkic languages.

(236) [Lev 1:16]

da tašla-sïn anï yan+ï+na ol

and to stone-3SG.VOL DEM.PRO.ACC side+3SG.POSS+DAT DEM.PRO

mizbeaḥ+nïŋ gün doyušï taraf+ḥa ol **kül tök-il-än yer+gä**altar+GEN east side+DAT the ash to pour out-PASS-PN place+DAT

'he shall cast it beside the altar on the east side, in the place where ashes are thrown away'.

The translation of this verse also employs this common strategy, e.g., [Lev 1:16] Cr.Tat.B *kul olyan erge* 'to the place where there are ashes'; Ot.Tur.B *mašriķa doyru olan külüŋ yerine* 'to the place of the ashes that is towards the east; H.Kar.B *ol kil tegiledoyac orunya* 'to the place where the ashes are thrown away'. Although the Trakai Bible translation (A.Dub.III.73) does not use the same structure for this translation, such characteristics can also be attested in Trakai Karaim (Gülsevin 2016: 117).

C-) Cause Clauses

^{&#}x27;And he shall slay the lamb in the place where he slays the sin offering'.

In this type of clauses, we encounter certain conjunctions (see 2.2.4.) which express the reason for an action in Turkic languages.

(237) [Lev 21:8]

Da ayruḥsï et-kin anï **zira**

And to sanctify-2SG.IMP DEM.PRO.ACC because

 $korban+\ddot{\imath}+n$ $Te\eta ri+\eta+ni\eta$ ol

offering+3SG.POSS+ACC God+3SG.POSS+GEN 3SG.PRO

yuvuklaštir-aydir-Ø

to bring forward-PRES-3SG

'And you shall sanctify him, **because** he is bringing forward the offerings of your God'.

As expected, such a structure can be attested in all Turkic languages investigated in our study, e.g., [Lev 21:8] Cr.Tat.B *čünki* o siziŋ Allaŋizniŋ ašini ketire; Ot.Tur.B zira Allahiŋiŋ ta 'amini takrib eder; H.Kar.B ki osol karbanin Tenrinnin ol yuvutadi T.Kar.B ki ošol karbanin Tenriynin ol yuvutadi 'And you shall sanctify him, because he is bringing forward the offerings of your God'.

D-) Conditional Clauses

As demonstrated earlier (see 2.2.3.4.4.), the conditional is indicated with the marker {-sA} in verb or clitic stems in Turkic languages. It should be noted that this structure appears only in the independent clause, which may be introduced by the Persian conjunction *egär* 'if', a common feature in most of the Turkic languages (Johanson 2022a: 57).

(238) [Lev 3:12]

Da egär ečki e-sä-Ø korban+ï da and if goat **CL-COND** sacrifice+3SG.POSS and

yuvuklaštir-sin ani ald+i+na YWY+niŋ

to birng forward-3SG DEM.PRO.ACC before+3SG.POSS+DAT Lord+GEN

'If his offering is a goat, then he shall offer it before the Lord'.

E-) Clause of purpose

In this type of clause, the expression of intention or aim of the action is found. In the majority of cases, the verb in the independent clause is accompanied by a verbal noun ending {-mA} or an infinitive ending {-mAK}, which may be followed by a dative, ablative marker, or can be in nominative form.

(239) [Lev 17:9]

Da $e \tilde{s} i \lceil k > g \rceil + i + n \ddot{a}$ ohel mo 'ed+niŋ ketir-mä-sä-Ø

And door+3SG.POSS+DAT tent of meeting+GEN to bring-NEG-COND-3SG

anï ķil-**ma**+**ya** anï YWY+**y**a

DEM.PRO.ACC to do-**VN**+**DAT** DEM.PRO.ACC Lord+DAT

The translation of biblical verses also shows this common strategy in Turkic languages, e.g., [Num 11:20] Cr.Tat.B nege kerek edi bizge Misirdan čiķip ketmege? lit. 'What was the necessity for us to leave Egypt'; [Lev 17:9] Ot.Tur.B; onu Rabbe takrib etmeğe ǯema ʿat meskeniniŋ kapisina getirmezse; H.Kar.B Da esigine 'ohel mo 'ednin keltirmese ani ķilma ani alninda H-nin; T.Kar.B Da ešigińa ohel mo 'ednin keltirmaśa ani ķilma ani Adonayya. 'And if he does not bring it to the door of the tent of meeting to make it to the Lord'.

In most other usages, the dependent clause is introduced by conjunction ki or da, and the verb in the dependent clause is followed by an optative or imperative marker.

(240) [Lev 19:17]

ügütlä-mäügütlä-gindost+uŋ+nïkito admonish-VNto admonish-2SG.IMPfriend+2SG.POSS+ACCthat

 $\check{c}ek$ - $m\ddot{a}$ - $g\ddot{a}y$ -sin $o+n\ddot{i}\eta$ $u\check{c}un$ $\check{g}\ddot{u}r\ddot{u}m$ to bear-NEG-OPT-2SG3SG.PRO+GENforiniquity

'you shall not regard your brother in your heart |as| an enemy, you thall surely admonish your friend, lest you bear iniquity because of him'.

_

^{&#}x27;And if he does not bring it to the door of the tent of meeting to make it to the Lord'.

¹³¹ Lit. 'admonishing, you shall admonish'. A calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), e.g., B.Heb הוֹכָחַ הוֹכְיהַ ; lit. 'to correct you shall correct'.

Similar strategies are found in Ottoman Turkish and Karaim translations as well, e.g., [Lev 10:6] Ot.Tur.B *esvabinjīzī yīrtman ki ölmeyesiz*; H.Kar.B *upraķlarīnīznī sekmeniz da elmeniz*; T.Kar.B *upraķlarīyīznī sökmayiz da ölmayiz* 'and do not rip your clothes, lest you die'.

F-) Clauses of Manner

In this variety of clauses, adverbs frequently surface, delineating the nature of the action—a characteristic prevalent in Turkic languages (refer to section 2.2.2.7. for further detail). In several instances, the postpositional clause is introduced with the term *nečiki* 'so as, so that'.

(241) [Lev 4:20]

da ķil-sin ol buya+ya **nečiki** ķil-dī-Ø

and to do-3SG.VOL bull+DAT how to do-PAST-3SG

buya+sï+na ol ḥatat+nïŋ

bull+3SG.POSS+DAR the sin offering+GEN

Bible translations demonstrate that the same structures were present in Ottoman Turkish and Karaim dialects, e.g., Ot.Tur.B bu genž boğaya böyle eyleye niže ki günah için takrib olunan genž boğaya etdi ise; H.Kar.B külsün tanaya ki nežik küldü tanasüna ol hatatnün alay külsün anar 'he shall do to the calf as he did to the calf of the sin offering'; T.Kar.B da külsün tanaya ki nečik qüldü tanasüna ol hatatnün ki özünün alay külsün anar 'he shall do to the calf as he did to the calf of the sin offering, which is his own'.

2.4. Lexicon

In the dataset under study, the total count of distinct lexical items amounts to 1,075¹³². The frequency distribution of these words is delineated in the following table.

Table 79: Distribution of Lexical Item in the Dataset

| Origin | Distribution |
|--------|--------------|
| Turkic | 726 (67.53%) |
| Arabic | 109 (10.14%) |

_

^{&#}x27;And he shall do to the bull as he did to the bull of the sin offering'.

¹³² Note that inflected forms are only counted when they refer to a new meaning. Besides, slight phonological differences are also considered as different examples.

| Origin | Distribution |
|------------------|--------------|
| Hebrew | 91 (8.47%) |
| Persian | 65 (6.05%) |
| Unknown | 24 (2.23%) |
| Arabic+Turkic | 22 (2.05%) |
| Persian+Turkic | 15 (1.39%) |
| Greek | 6 (0.56%) |
| Hebrew+Turkic | 5 (0.47%) |
| Italian | 2 (0.19%) |
| Arabic+Persian | 3 (0.28%) |
| Mongolian+Turkic | 2 (0.19%) |
| Turkic+Persian | 1 (0.09%) |
| Armenian+Turkic | 1 (0.09%) |
| Sogdian | 1 (0.09%) |
| Hungarian | 1 (0.09%) |
| Russian | 1 (0.09%) |

In the subsequent section, we will undertake the categorization and examination of these lexical items. The CKED dictionary will serve as our principal resource for our dataset during this analysis, offering insights into the etymology of some words and/or their recent donor language where applicable. However, it is worth mentioning that the CKED dictionary may not encompass the full complexity or provide clarification for some ambiguous cases.

2.4.1. Turkic Vocabulary

The primary vocabulary of the corpus prominently features Turkic lexical items, showcasing 726 distinct Turkic words. The majority of these words are traceable to Crimean Tatar, Crimean Turkish, Ottoman Turkish, and the three dialects of Karaim, often exhibiting minor phonological modifications. In specific instances, unique Oghuz and Kipchak lexical disparities can be identified, e.g., [Lev 11:30] güneš 'sun' [Lev 22:7] kuyaš 'id'. As expected, many of these Oghuz characteristics can be directly linked to Ottoman Turkish. Further examination reveals that animal names, predominantly featured in Chapter 11, appear to be direct borrowings from specific Ottoman Bible translations into the Lev of Göz. 1841, inclusive of some lexical copying errors (refer to section 2.4.6.). In conjunction with these words, the CKED dictionary indicates that, out of the 655 Turkic lexical items listed therein, 95 are identified as loanwords from Ottoman Turkish. Upon scrutinizing the lexical items credited to Ottoman Turkish, it is apparent that some of these are commonly found in Turkic vocabulary, exhibiting

Ottoman Turkish phonological characteristics (see section 2.1.2.2.1.), e.g., kel > gel- 'to come', which is also notable in Crimean Karaim. However, it is important to emphasize that no specific Crimean Tatar word was found in our dataset. As for the Turkic lexical items of Western Karaim, distinguishing them from Crimean Karaim poses a considerable challenge and thus only a few words can be attirbuted to Western Karaim as they show some Western Karaim phonological characteristics, e.g., [Lev 19:10] *čöplä*- 'to gather; to collect'; [Lev 26:21] *kile*- 'to wish'.

It is important to acknowledge that our dataset contains a total of 71 Turkic-origin words which are not referenced in either the CrKB or KRPS for Crimean Karaim. A subset of these words display only minor phonological alterations, while others do not appear in any recognizable form. Therefore, we believe it would be more appropriate to catalogue them separately. Subsequently, a table will present 26 Turkic lexical items that appear in dictionaries with minor modifications.

Table 80: Unlisted Variants of Turkic Lexical Items

| Göz. 1841 | | Dictionaries | | |
|-------------|----------|--------------|---------------------------|----------------------------|
| [Lev 19:18] | arķardaš | arķadaš | 'comrade; fellow' | (CrKB: 48; KRPS: 73) |
| [Lev 4:6] | barmaķ | parmaķ | 'finger' | (CKED: 275; KRPS: 446) |
| [Lev 20:27] | bildiǯi | bildeǯi | 'sorcerer; healer' | (CKED: 83; KRPS: 118–119) |
| [Lev 15:9] | binek | minek | 'saddle beast' 133 | (CKED: 243; KRPS: 407) |
| [Lev 21:10] | böyük | büyük | 'big; great' (via Ot.Tur) | (CKED: 100; KRPS: 144) |
| [Lev 11:22] | čekirtke | čegirtke | 'locust' | (CKED: 115; KRPS: 639) |
| [Lev 16:23] | češ- | teš-; čez- | 'to untie; to unfasten' | (CKED: 398; KRPS: 568) |
| [Lev 26:39] | čüri- | čiri-; čürü- | 'to rot, to decay' | (CKED: 123; KRPS: 629) |
| [Lev 1:15] | damla- | tamla- | 'to drip, to ooze' | (CKED 381; KRPS: 510) |
| [Lev 11:32] | deri | teri | 'skin; hide' | (CKED: 396; KRPS: 566) |
| [Lev 13:25] | derän | terän | 'deep' | (CKED: 396; KRPS: 567) |
| [Lev 16:10] | diri | tiri | 'living; alive' | (CKED: 404; KRPS: 529–530) |

_

 $^{^{133}}$ It is important to highlight that in our text, the word *binek* is used exclusively to signify 'saddle'. Therefore, this usage deviates from the definitions provided in standard dictionaries.

| Göz. 1841 | | Dictionaries | | |
|-------------|-------------|-----------------|--------------------------|----------------------------|
| [Lev 17:8] | diril- | tiril- | 'to keep alive' | (CKED: 404; KRPS: 539) |
| [Lev 2:2] | dolu | tolu | 'full; filled' | (CKED: 407; KRPS: 537) |
| [Lev 11:16] | duyan | yaduyan; yeduya | 'hawk' | (CKED: 440; KRPS 216) |
| [Lev 11:4] | degil | dügül | 'not; it is not' | (CKED: 143; KRPS: 182) |
| [Lev 15:22] | dürli | dürlü | 'various' (via Ot.Tur) | (CKED: 144) |
| [Lev 16:12] | girgiz- | kirgiz- | 'to let in; to bring in' | (CKED: 213; KRPS: 322) |
| [Lev 5:7] | gögürčün | kögürčün | 'pigeon' | (CKED: 216; KRPS: 336) |
| [Lev 26:36] | göŋül | köŋül | 'heart; mind' | (CKED: 217; KRPS: 338) |
| [Lev 13:3] | görüm | körüm | 'appearance; look' | (CKED: 219; KRPS: 339) |
| [Lev 26:16] | ķaltramaķ | ķaltïramaķ | 'to shiver; to tremble' | (CKED: 286; KRPS: 360) |
| [Lev 26:6] | ķaltratïvčï | ķaltïratïvčï | 'causing shivering' | (CKED: 286; KRPS: 360) |
| [Lev 11:6] | tavušan | tafšan | 'hare; rabbit' | (CKED: 379; KRPS: 518) |
| [Lev 23:36] | tiyilmäk | tïyïlmaķ | 'delay; constipation' | (CKED: 402; KRPS: 556) |
| [Lev 7:34] | yokarï | yuķarï | 'above; up' (via Ot.Tur) | (CKED: 467–468; KRPS: 246) |

As observable, some differences (a total of 15 examples) can be clearly attributed to Ottoman Turkish features (refer to 2.1.2.2.1.), while others exhibit sporadic changes, as previously illustrated (see sections 2.1.2.1 and 2.1.2.2.2.). On the other hand, the following table presents 45 lexical items that are not attested in either the CKED or KRPS for Crimean Karaim.

Table 81: Unlisted Turkic Lexical Items

| Göz. 1841 | Cr.Tat | Ot.Tur | H.Kar | T.Kar |
|----------------------|----------------|-------------|-------|-------|
| [Lev 11:5] ada | - | ada tavšanï | - | - |
| tavušan 'hare; | | (LET: 618) | | |
| European rabbit' | | | | |
| [Lev 11: 14] ak baba | aķbaba; aybaba | aķbaba | - | - |
| 'vulture' | (KRUS: 44) | (ETD: 475) | | |
| [Lev 15: 2] akkan | - | - | - | - |
| 'flow' | | | | |

| Göz. 1841 | Cr.Tat | Ot.Tur | H.Kar | T.Kar |
|---------------------------|-----------------|---------------------|------------------|------------------|
| [Lev 15:13] akma | - | - | - | - |
| 'flow' | | | | |
| [Lev 15:3] aķmaķ | - | - | aķmak (KRPS: 58) | aḥmaḥ (KRPS: 58) |
| 'flow' | | | | |
| [Lev 15:28] akman | - | - | - | - |
| 'flow' | | | | |
| [Lev 25:27] artkan | - | - | - | - |
| 'remainder' | | | | |
| [Lev 11:19] balïķčïn | baliķčil | baliķčil; baliķčin | - | - |
| 'heron; egret' | (KRUS: 95) | (LET: 397) | | |
| [Lev 2:11] bekmäz | bekmez; pekmez; | bekmez; petmez | - | - |
| 'grape molasses' | petmez | (ETD: 541) | | |
| | (KRUS: 105) | | | |
| [Lev 24:12] beyin | beyin | beyin | _134 | _135 |
| 'brain' | (KRUS: 104) | (LET: 92) | | |
| [Lev 21:20] <i>bodïr</i> | - | bodur 'dwarf; short | - | - |
| 'short; shortie' | | in stature | | |
| | | amounting to | | |
| | | deformity' | | |
| | | (ETD: 544) | | |
| [Lev 21:13] <i>boyliķ</i> | - | - | - | - |
| 'virginity' | | | | |
| [Lev 11:22] <i>žudžud</i> | - | <i>žiržir</i> | - | - |
| 'cicada' | | (YTL: 363) | | |
| [Lev 23:2] čaķirmaķ | - | - | - | - |
| 'convocation' | | | | |
| [Lev 11:14] <i>čaylaķ</i> | - | čaylaķ | - | - |
| 'kite' | | (ETD: 633) | | |
| [Lev 22:25] | - | - | - | čаурата <u></u> |
| čaypalmaķ | | | | (KRPS: 621) |
| 'corruption; defect' | | | | 'destruction; |
| | | | | extermination' |
| [Lev 21:5] čïzmaķ | - | - | - | - |
| 'line' | | | | |
| [Lev 11:13] <i>deŋiz</i> | - | deŋiz k̞artalï | - | - |
| ķartalï 'sea eagle' | | (Işık 2021: 350) | | |

 $^{^{134}}$ It appears as *miy* (KRPS: 406–407). 135 It appears as *miy* (KRPS: 406–407).

| Göz. 1841 | Cr.Tat | Ot.Tur | H.Kar | T.Kar |
|------------------------------|-------------|---------------------------|-------|-------|
| [Lev 11:36] | - | devširilme ¹³⁶ | - | - |
| devširilmä | | (ÖTS 2: 1192) | | |
| 'concentration' | | | | |
| [Lev 11:16] <i>deve ķušï</i> | deve ķušu | deve <u>k</u> ušu | - | - |
| 'ostrich' | (KRUS: 153) | (ETD: 228) | | |
| [Lev 18:11] | - | - | - | - |
| doyurtkan 'begotten' | | | | |
| [Lev 25:30] dolyïnža | tolyun | dolyun | - | - |
| 'fully' | (KRUS: 597) | (KRUS: 352) | | |
| [Lev 5:23] <i>dutup</i> | - | - | - | - |
| almaķ 'plundered | | | | |
| item' | | | | |
| [Lev 11:16] duyan | - | doyan | - | - |
| 'hawk' | | (ETD: 805) | | |
| [Lev 11:37] ekin | ekin | ekin | - | - |
| 'crop; cereal crops' | (KRUS: 774) | (ETD: 480) | | |
| [Lev 11:3] gevšemäk | kevšek | geviš | - | - |
| 'cud' | (KRUS: 222) | (LET: 200) | | |
| [Lev 11:3] gevšemäk | - | geviš getir- | - | - |
| getir- 'chew the cud' | | (LET: 200) | | |
| [Lev 11:3] <i>gevšä-</i> | kevše- | - | - | - |
| 'chew the cud' | (KRUS: 222) | | | |
| [Lev 11:30] göztöbä | - | köstebek | - | - |
| 'mole' | | (ETD: 942) | | |
| [Lev 11:30] güneš | - | güneš keleri | - | - |
| kelerisi lit. 'sun | | (Işık 2020a: 155) | | |
| lizard' | | | | |
| [Lev 11:18] ķašīķčī | ķašïķķuš | kašïkčï kušu | | |
| ķušī 'pelican' | (KRUS: 289) | (ÖTS 3: 2464). | | |
| [Lev 10:7] olmaya | - | - | - | - |
| 'or else; lest | | | | |
| [Lev 22:27] <i>öte</i> | _137 | öte | - | - |
| 'beyond; further' | | (ETD: 499) | | |

¹³⁶ Despite the fact that the precise form *devširilme* is not documented in our available Ottoman Turkish dictionaries, it does appear in ÖTS in the context of Turkish. However, it is not explicitly stated whether this form was indeed present in Ottoman Turkish, a possibility that remains highly plausible.

¹³⁷ In Crimean Tatar, the term $\ddot{o}te$ is only listed before the word $k\ddot{u}n$, denoting 'yesterday'. When it appears before $y\ddot{i}l$, it signifies 'last year' (KRUS: 423).

| Göz. 1841 | Cr.Tat | Ot.Tur | H.Kar | T.Kar |
|-----------------------------|--------|-------------------|-------|-------|
| [Lev 2:4] piširmä | _138 | piširmeklik | - | - |
| 'baking' | | (LET: 60) | | |
| [Lev 6:14] pišmäk | - | - | - | - |
| 'cooked' | | | | |
| [Lev 7:30] sallamaķ | - | - | - | - |
| 'wave offering' | | | | |
| [Lev 25:25] satķan | - | - | - | - |
| 'sold' | | | | |
| [Lev 8:2] silmäk yayï | - | - | - | - |
| 'anointed oil' | | | | |
| [Lev 11:29] sivri | - | sivri sïčan | - | - |
| sïčan 'shrew' | | (Işık 2021: 364) | | |
| [Lev 19:14] | - | - | - | - |
| sürünmäk 'obstacle' | | | | |
| [Lev 19:4] <i>tökmä</i> | - | - | - | - |
| 'molten' | | | | |
| [Lev 15:16] <i>tökmäk</i> | - | - | - | - |
| 'emission' | | | | |
| [Lev 11:17] <i>ügi ķušu</i> | - | ögü kušï | - | - |
| 'eagle owl' | | (TLO I: 535) | | |
| [Lev 16:32] yerine | - | yerine | | |
| 'instead' | | (ETD: 171) | | |
| [Lev 11:30] yïldïz | - | yïldïz keleri | - | - |
| kelerisi lit. 'star | | (Işık 2020a: 156) | | |
| lizard' | | | | |
| | | | | |

The analysis reveals that out of the 46 items assessed, 25 surface in Ottoman Turkish, 10 in Crimean Tatar, 2 in Trakai Karaim, and a solitary instance in Halich, while 19 items are conspicuously absent in these languages. A closer examination of these particular absent words reveals common derivational strategies, wherein the fundamental nominal or verb stem frequently appears in other languages.

2.4.2. Arabic Loanwords

Predominantly, the non-Turkic vocabulary is Arabic in origin, as demonstrated by 108 lexical items. The incorporation of these words presents distinct characteristics, as elucidated in section

 $^{^{138}}$ In Crimean Tatar, the word *piširme* stands for 'jam', which therefore differs from our data.

2.1.2. It is notable that, according to the CKED, out of the 85 Arabic loanword lexical items presented in the dictionary, 71 of them have been identified as being adopted into Crimean Karaim via Ottoman Turkish. Notably, 23 lexical items were not catalogued in either CKED or KRPS. In a manner parallel to the previous subchapter, we divide these unlisted items into two groups, with some appearing in dictionaries subsequent to minor phonological modifications while others are not preserved in any form. The subsequent table clarifies that 11 such items are present in dictionaries, albeit with certain alterations.

Table 82: Unlisted Variants of Arabic Loanwords

| Göz. 1841 | Dictionaries | | _ |
|---------------------------|--------------|----------------------------------|---------------------------|
| [Lev 5:21] amanat | emanet | 'trust; deposit' (via Ot.Tur) | (CKED: 150; KRPS: 660) |
| [Lev 17:7] dävür | devir | 'period; epoch' (via Ot.Tur) | (CKED: 137; KRPS: 183) |
| [Lev 19:20] fiḥat | fiyat | 'price' (via Ot.Tur) | (CKED: 163; KRPS: 594) |
| [Lev 10:13] <i>ḥaķķ</i> | haķ | 'truth; law; salary (via Ot.Tur) | (CKED: 176; KRPS: 606) |
| [Lev 11:37] <i>helal</i> | ḥelal | 'clean; pure' (via Ot.Tur) (| CKED: 182–183; KRPS: 608) |
| [Lev 26:28] <i>kaḥer</i> | ķaḥar | 'anger' | (CKED: 284) |
| [Lev 26:30] <i>kayïp</i> | үауїb | 'lost; missing' (via Ot.Tur) | (CKED: 171; KRPS: 163) |
| [Lev 12:6] <i>kumru</i> | hümri; hümrü | 'ring dove' (via Ot.Tur) | (CKED: 188; 374) |
| [Lev 11:19] legläg | leglek | 'stork' | (CKED: 229) |
| [Lev 19:5] morad | murad | 'aim; goal' (via Ot.Tur) | (CKED: 246; KRPS: 411) |
| [Lev 15:25] <i>zeyada</i> | ziyada | 'much; more' (via Ot.Tur) | (CKED: 482) |

Certain variations can once again be attributed to the adaptation of Ottoman Turkish forms or dedicated to sporadic changes. On the contrary, 12 lexical items were not listed in any similar form in the dictionaries.

Table 83: Unlisted Arabic Loanwords

| Göz. 1841 | Cr.Tat | Ot.Tur | T.Kar | H.Kar |
|-------------------------|--------------------------|-------------------------|-------|-------|
| [Lev 13:2] <i>ǯüzam</i> | <i>ǯüzam</i> (KRUS: 744) | <i>ğüzam</i> (LET: 477) | - | - |
| 'leprosy' | | | | |

| Göz. 1841 | Cr.Tat | Ot.Tur | T.Kar | H.Kar |
|--------------------------|---------------------------|---------------------------|-------|-------|
| [Lev 11:32] <i>espap</i> | - | esvap (LET: 357) | - | - |
| 'clothing; garment' | | | | |
| [Lev 26:26] fa 'iz | - | faiz (ETD: 846) | - | - |
| 'interest; usury' | | | | |
| [Lev 11:14] farzam | - | farzen 'supposing that' | - | - |
| 'although; though'' | | (ÖTS 2: 1548) | | |
| [Lev 25:24] <i>hïlas</i> | ḫalas; ḫales (KRUS: 670) | <i>ḫalas</i> | - | - |
| 'redemption' | | (LET: 629) | | |
| [Lev 11:19] hüdhüd | üdüd ķuš (KRUS: 628) | hüdhüd | - | - |
| 'hoopoe' | | (LET: 405) | | |
| [Lev 20:20] maḥrïm | marum; maḥrum | maḥrum | - | - |
| 'deprived' | (KRUS: 338; 340) | (ÖTS 3: 3020) | | |
| [Lev 7:18] makbul | makbul (KRUS: 331) | maķbul (YTL: 705) | - | - |
| 'accepted' | | | | |
| [Lev 25:21] mamur | mamur (KRUS: 334) | mamur (ETD: 1034) | - | - |
| 'cultivated' | | | | |
| [Lev 11:10; 11:23] | mekruh (KRUS: 345) | mekruh (LET: 2) | - | - |
| mïķruḥ; mïķruḥa | | | | |
| 'abominable' | | | | |
| [Lev 25:30] | maķam | maķam 'a place; a high | - | - |
| mukam 'location; | 'melody; an air of music' | office; an air (of music) | | |
| residence' | (KRUS: 330) | (ETD: 1040) | | |
| [Lev 26:16] tayin | tayin | ta 'yin | - | - |
| 'appointment' | (KRUS: 538) | (LET: 32) | | |

It can be observed that all the unlisted Arabic loanwords are present in Ottoman Turkish, with four of them also appearing in Crimean Tatar. It's noteworthy that none of these words are present in the Western Karaim languages.

It warrants mention that there exist two Arabic lexical items which were consistently utilized in lieu of their Turkic equivalents throughout Chapter 11 of the Göz. 1841. In other chapters, however, such Arabic words, which were prevalent in Ottoman Turkish cannot be attested.

Table 84: Usage of the Arabic ve and evlad in the Dataset

| Glosses | Lev of the Göz. 1841 | Lev of the Göz. 1841 (Only in Lev 11) |
|---------|----------------------|---------------------------------------|
| | | |
| 'and' | da | ve; vä |
| 'son' | oylan | evlad |

Regarding the derivative forms of Arabic loanwords, we identify some common strategies. Considering nominals, all loanwords are treated as nominals, even if they originate as verbal forms. Consequently, we discover numerous instances where expected denominal, highly productive formative markers—referenced in section 2.2.2.1.1.1—are appended to Arabic stems.

(242) [Lev 3:17]
$$\ddot{o}m\ddot{u}r+lik$$
 life+DER 'life long'. [Lev 12:2] $ar+s\ddot{i}z$ shame+DER 'shameless'.

It's significant to mention that we also come across instances where the derivation results from the compounding of Arabic and Persian elements.

In regard to verbal forms, we identify several predominant strategies. These primarily include the use of Turkic auxiliary verbs such as *et-/it-* 'to do; to make' and *ol-/bol-* 'to be; to become', among other verbs, which follow Arabic nominals. It is crucial to underscore that this structure is commonplace in numerous Turkic languages.

In certain instances, the Turkic denominal verb marker {+lA} is affixed to the word stem, a procedure that is commonplace in Turkic languages, as illustrated in section 2.2.3.1.1.1.

[Lev 26:32] maymun+la+n sad+DER+REFL 'to be upset'.

2.4.3. Persian Loanwords

Within our dataset, we identify a total of 65 words of Persian origin. Among these, seven were not indexed in Karaim dictionaries. Further analysis reveals that four among these seven exhibit slight discrepancies when compared with their dictionary entries.

Table 85: Unlisted Variants of Persian Loanwords

| Data | Dictionaries | | |
|----------------------------|------------------|-------------------------------------|------------------------|
| [Lev 11:27] <i>ženavar</i> | žanavar; žanïvar | 'animal; beast; butcher' | (CKED: 102; KRPS: 171) |
| [Lev 12:4] <i>hič</i> | <u>ḥeč</u> | 'nothing' | (CKED: 182; KRPS: 608) |
| [Lev 11:38] <i>toḥum</i> | toķum | 'seed; grain; lineage; family' (via | Ot.Tur) (CKED: 409) |
| [Lev 11:25] umundar | mundar | 'unclean; impure; dirty' | (CKED:245; KRPS: 410 |

However, three of them do not manifest in any recognizable form in Karaim dictionaries.

Table 86: Unlisted Persian Loanwords

| Göz. 1841 | Cr.Tat | Ot.Tur | H.Kar | T.Kar |
|-----------------------------------|--------|-----------------------------|-------|-------|
| [Lev 11:13] iron 'bearded vulture | - | iron; irun (Işık 2021: 350) | - | - |
| (gypaetus barbatus)' | | | | |
| [Lev 11:36] sarnič 'cistern' | - | sarnič (ETD: 52) | - | - |
| [Lev 11:19] <i>šepere</i> 'bat' | - | šepere (ETD: 66) | - | - |

It is significant to mention that, among the 58 Persian-origin words listed in the CKED, 39 of them are attributed to Ottoman Turkish.

Analogous to the previous subchapter, the derivation of Persian loanwords also exhibits similar characteristics to those found in Arabic loanwords. Pertaining to nominal derivations, we encounter expected markers (refer to section 2.2.2.1.1.1.) affixed to Persian loanwords.

(246) [Lev 4:13]
$$g\ddot{u}n\ddot{a}h+li$$
 sin+DER 'sinful'.
 [Lev 25:35] $din+da\check{s}$ religion+DER 'coreligionist'.

Regarding verb forms, we also observe the affixation of certain auxiliary verbs or the suffixation of the marker {+lA}.

However, there is a remarkable example where a non-Turkic word, a nominal is processed as a Turkic verb and subsequently conjugated as such, a phenomenon that is exceedingly rare in Turkic languages.

The verb is notable for its attestation in other Turkic languages as well, e.g., T.Kar *aškart*-(KRPS: 92) and H.Kar *askart*- (KRPS: 80). In Anatolian Turkish we also identify *ešker*- 'to be evident' and *eškert*- 'to make evident' (ÖTS 2: 1501). Furthermore, the verb form *eškerit*- is observed, deriving from *ešker it*- (ÖTS 2: 1501). As such, the form *aškärt*- might be explicated by vowel deletion in the auxiliary verb *it*-. However, the presence of forms such as *aškart*-; *askart*- in Western Karaim adds a layer of complexity to the discussion. It's noteworthy that although the word *ašķara* is present (CC: 44), no verb form of this word is observed in the Codex Cumanicus.

2.4.4. Biblical Hebrew Loanwords

In the analyzed corpus, there are 92 words of Biblical Hebrew origin. Out of these, 39¹³⁹ are not cataloged in Karaim dictionaries. Among these unlisted words, five bear significant resemblance to entries in the CKED dictionary.

Table 87: Unlisted Variants of Hebrew Loanwords

| Göz. 1841 | Dictionaries | | |
|----------------------------------|-----------------|--------------------------|-------------|
| [Lev 5:13; 6:23] kapara; keparat | kappara; kapara | a 'apology; forgiveness' | (CKED: 202) |
| [Lev 23:6] <i>mača</i> | mačča | 'unleavened bread | (CKED: 231) |
| [Lev 26:1] mačeva | maččeva | ʻpillar' | (CKED: 231) |
| [Lev 23:34] <i>suka</i> | sukka | 'booth' | (CKED: 363) |

¹³⁹ This group does not encompass any personal proper names.

_

| Göz. 1841 | Dictionaries | | |
|--------------------------|--------------|---------------------|-------------|
| [Leb 16:31] <i>šabat</i> | šabbat | 'saturday; Sabbath' | (CKED: 370) |

The primary divergence, as can be observed, resides in the medial gemination of specific consonants. This is related to a diacritical mark, known as *dagesh hazaq* (), found in Biblical Hebrew. When affixed to a letter representing a consonant, it signals gemination of that consonant, a feature not present in modern Hebrew. This phenomenon is attested in the lexical items illustrated in the above table. However, this characteristics is not indicated in our transcription, as we conform to the general transcriptional method employed in published Karaim Bible translations. As such, the differences displayed in the table primarily reflect transcription methodology, rather than linguistic variants. However, the subsequent table introduces 34 lexical items that are not cataloged in Karaim dictionaries.

Table 88: Unlisted Biblical Hebrew Lexical Items

| Biblical Hebrew | Göz. 1841 |
|---|---------------------------|
| ַם אָּשֶׁם (CEDHL: 58) | [Lev 5:15] ašam |
| בְּמָה [bāmāh] 'stage; pulpit; platform; high altar' (CEDHL: 76) | [Lev 26:30] bama |
| בהק [bōhak] 'a kind of disease (a hapax legomenon in the Bible); brightness' | [Lev 13:39] bohak |
| (CEDHL: 65) | |
| אַלֹּד [ʾêp̄ōd] 'ephod, a garment worn by the high priests' (CEDHL: 45) | [Lev 8:7] <i>efod</i> |
| [gibbêaḥ] 'bald on the forehead (a hapax legomenon in the Bible)' (CEDHL: | [Lev 13:41] gibeaḥ |
| 89) | |
| ֶּחֶגֶב [ḥāgāḇ] 'locust; grasshopper' (CEDHL: 207) | [Lev 11:22] <i>ḥagav</i> |
| ּחָמֵץ [ḥāmêṣ] 'hametz, leavened (bread)' (CEDHL: 222) | [Lev 2:11] <i>ḥameč</i> |
| ּתְרְגֹּל [ḥargōl] 'a kind of locust (a hapax legomenon in the Bible)' (CEDHL: 230) | [Lev 11:22] <i>ḥargol</i> |
| הַּשְּׁאָת [ḥaṭṭāt] 'sin; guilt; sin offering' (CEDHL: 213) | [Lev 4:3] hatat |
| חַזִיר [ḥăzîr] 'pig; swine' (CEDHL: 212) | [Lev 11:7] ḥazïr |
| תַּרֶם [hêrem] 'thing devoted; ban; devotion; destruction' (CEDHL 233) | [Lev 27:28] <i>ḥerem</i> |
| וֹשֶׁרְ [ḥōšen] 'breastplate of high priest' (CEDHL: 236) | [Lev 8:8] hošen |
| קרם [kêrêah] 'bald (a hapax legomenon in the Bible)'; place from which plants | [Lev 13:40] <i>ķereaḥ</i> |
| have been removed' (CEDHL: 593). | |
| קרְבָּן [kārbān] 'offering; sacrifice; oblation; victim' (CEDHL: 591) | [Lev 1:2] korban |
| קָּדְוֹשׁ [kִādַōš] 'holy; sacred; saint, saintly' (CEDHL: 562) | [Lev 11:44] <i>kuduš</i> |

| Biblical Hebrew | Göz. 1841 |
|---|--------------------------|
| קליקה [mêlîķā] 'nipping off (the neck of a bird)' (CEDHL: 350) | [Lev 5:8] meliķa |
| קלוּאִים [milluʾîm] 'filling in, setting; 'installation; supplement; addition' (CEDHL: | [Lev 7:37] milu 'im |
| 348) | |
| קישְׁקֵל [mišķāl] 'mithqal, a unit of mass; weight' (CEDHL: 678) | [Lev 19:35; 27:4] |
| | misķal; mesķal |
| מֹלֶדּ [mōlek] 'Molech' | [Lev 18:21] <i>Moleh</i> |
| נֶתֶק [netek] 'scab' (CEDHL: 431) | [Lev 13:30] neteķ |
| ְּבֵלֶה [nəḇêlāh] 'corpse, carcass; an animal that died a natural death' (CEDHL: 402) | [Lev 7:24] nevela |
| ן נָאַ [$n\bar{o}$ 'ê $p\bar{p}$] 'to commit adultery' (CEDHL: 400) | [Lev 20:10] no 'ef |
| פָסוּל [pāsul] 'unfit; defective; disqualified' (CEDHL: 515) | [Lev 7:18] <i>pasul</i> |
| לָּהֶתֶת [pəḥetet] 'sunken spot in leprosy (a hapax legomenon in the Bible); defect; | [Lev 13:55] peḥetet |
| fault' (CEDHL: 502) | |
| רָהָם [rāḥām] 'carrion vulture' (CEDHL: 613) | [Lev 11:18] <i>raḥam</i> |
| שַׁבְּחוֹן [šabbātōwn] 'Sabbath observance; complete rest' (CEDHL: 639) | [Lev 16:31] |
| | šabaton |
| אָחִיטָה [šəḥîṭa] 'ritual slaughtering' (CEDHL 649) | [Lev 17:5] šeḥita |
| מַשְׁ [šêm] 'name; as designation of God' (CEDHL: 664) | [Lev 24:11] <i>šem</i> |
| ְּטְרֵפָה [tərêp̄āh] 'an animal tom by wild beasts; an animal with organic defect; | [Lev 7:24] terefa |
| ritually forbidden food' (CEDHL: 252) | |
| חוֹר [$tar{o}r$] 'turtle-dove' (CEDHL: 695) | [Lev 14:30] tor |
| קָּמִים [tummîm] 'one of the two objects attached to the breastplate of the high | [Lev 8:8] tumim |
| priest' (CEDHL: 706) | |
| אּרִרים ($\tilde{u}r\hat{u}m$) 'one of the two objects attached to the breastplate of the high priest | [Lev 8:8] urim |
| (CEDHL: 14). | |
| אָרַ [$z\bar{o}w\underline{b}$] 'flux; issue' (CEDHL: 195) | [Lev 15:2] zav |
| זוֹנְה [zōnāh] 'harlot; prostitute' (CEDHL: 196) | [Lev 21:7] zona |

It is important to note that of the lexical items listed above, only the Arabic form of Biblical Hebrew *korban*, known as *kurban*, meaning 'sacrifice', is attested in Crimean Tatar (KRUS: 302), Ottoman Turkish (ETD: 299), and Western Karaim (KRPS: 347). The remaining lexical items are not attested in these languages.

Similar to the loanwords previously explored, all Biblical Hebrew loanwords in our dataset present as nominals. Their patterns of nominal derivation align with the strategies previously illustrated in this study.

On the other hand, the derivation of verbs from the nominals of these Hebrew words invariably involves the affixation of a Turkic auxiliary verb.

2.4.5. Loanwords from Other Languages

In our dataset, numerous lexical elements of diverse origins, including Turkic, Arabic, Persian, and Hebrew, are found. Notably, there are 14 distinct lexical elements that are considered loanwords, borrowed from various languages. It is essential to highlight that most of these loanwords are verifiable in Karaim dictionaries. In certain cases, the Crimean Karaim Dictionary (CKED) elucidates the donor language, and this information is subsequently incorporated into the table provided below. However, it is imperative to clarify that for other examples, while we identify their provenance, it does not necessarily imply that these are the donor languages.

Table 89: Loanwords from Other Languages

| Origin | Göz. 1841 |
|-----------------|---|
| Armenian+Turkic | [Lev 25:25] zügürt+lä- 'to become poor' |
| | (for the adjective zügürt see CKED: 483; NS: 985) (via Ottoman Turkish) |
| Greek | [Lev 2:1] timyan 'incense' (CKED: 400; KRPS: 556) |
| | [Lev 4:30;] temel 'foundation; base' (CKED: 393; NS: 875) |
| | (via Ottoman Turkish) |
| | [Lev 13:47] üskülü 'flax; linen' (CKED: 434; KRPS: 591) |
| | [Lev 19:13] "iryat 'servant, workman, hired hand' |
| | (CKED: 191; KRPS: 651; NS: 378) (via Ottoman Turkish) |
| | [Lev 21:20] kambur 'hump' (CKED: 287; KRPS: 361; NS: 434) |
| | [Lev 24:6] terapiz 'trapezium, a little table' (NS: 902) |
| Hungarian | [Lev 23:43] čalaš 'hut, cabin; shelter' (CKED: 109; KRPS: 622) |

| Origin | Göz. 1841 |
|------------------|---|
| Italian | [Lev 11:28; 11:25] ruba/urba 'clothes' (CKED: 328, 427; KRPS: 580; NS: 743) |
| | (via Ottoman Turkish) |
| Mongolian+Turkic | [Lev 16:16] tohta-yan 'one who stops; one who dwells (see for the verb tohta- |
| | CKED: 406; KRPS: 539–540) |
| Mongolian+Turkic | [Lev 19:24] <i>maḥt(a>o)-v</i> 'praising, praise' (CKED: 232–233; KRPS: 405) |
| Russian | [Lev 6:21] <i>čöräp</i> 'shell' (CKED: 127; KRPS: 602) |
| Sogdian | [Lev 14:46] aḥšam 'evening' (CKED: 35; KRPS: 86; NS: 40) |

2.4.6. Copies from Ottoman Bible Translations

Within the Book of Leviticus in the Torah, there exists a substantial inventory of animal nomenclature, each delineated as either clean or unclean pertaining to consumption and/or sacrificial purposes. However, the definitive equivalencies of certain animal appellations found within the Hebrew Bible remain ambiguous, thus presenting significant challenges to translators in identifying the precise counterparts to the Biblical Hebrew terminologies. Various words, absent in contemporary Hebrew, persist in a state of definitional uncertainty, potentially due to the folk taxonomic structures prevalent among the ancient Hebrews (refer Cansdale, 1970; Ferguson, 1974). In addition to this, numerous other words are categorized as *hapax legomena*, introducing further complexities in the structural analysis. Consequently, this gives rise to an extensive range of preferences concerning the ambiguous identification of certain animal names in Bible translations.

Within ourdataset, there exists a collection of 52 distinct animal names, representative of various species, including mammals (20), insects (4), reptiles (5), and birds (23). Upon juxtaposition of our data with other Karaim and Ottoman Turkish Bible translations, striking similarities become apparent between our text and the Ottoman Bible translations. Indeed, in previous publications (Işık 2020a; 2021), a series of articles have been published that meticulously analyze these lexical items in detail, contrasting them with the Ottoman Bible translations and another Crimean Karaim Bible translation (CrKB I). These analyses have culminated in the conclusion that the translator(s) or compiler(s) of the Göz. 1841 unambiguously replicated thirty-two animal names from Chapter 11 of the Lev from the Kieffer 1827 edition, which was itself premised on an earlier Ottoman Bible translation, Ali Bey 1665 (Işık 2021: 363)¹⁴⁰.

-

¹⁴⁰ For the descriptions of these two Ottoman Turkish Bible translations, please refer to the abbreviations section.

The forthcoming table will provide an exhaustive comparative assessment, this time including both the Halich and Trakai Karaim translations. It's important to note that specific examples, previously presented in dictionaries dedicated to Crimean Karaim, which do not necessarily signify direct replication, will be omitted from this table.

Table 90: Animal Names Appearing in Chapter 11 of Lev

| Lev | Biblical Hebrew | ADub.III.73 | JSul.III.01 | Kieffer 1827 | Göz. 1841 |
|-------|-------------------------------------|-------------|-------------|-------------------|---------------|
| | | (T.Kar.B) | (H.Kar.B) | (Ot.Tur.B) | |
| 11:5 | יָשָּכָּן [šāpān] | krolik | dafsan | ada tavšan | ada tavušan |
| | 'rock badger; cony | | | 'hare; European | 'hare; |
| | rabbit' | | | rabbit' | European |
| | | | | | rabbit' |
| 11:6 | אַרְנֶכֶת [ʾarneḇeṯ] | ķoyan | ķoyan | tavšan | tavušan |
| | 'hare; rabbit' | | | | |
| 11:7 | חָזִיר [ḥăzîr] | ḥazir | ḥazir | <u>h</u> ïnzïr | <i>ḫïnzïr</i> |
| | 'pig; swine' | | | | |
| 11:13 | נֶשֶׁר [nešer] | nešer | nešer | ķartal | ķartal |
| | 'eagle; vulture' | | | 'eagle' | 'eagle' |
| 11:13 | פֶּרֶס [peres] | peres | peres | iron/irun | ironï |
| | 'bearded vulture' | | | 'a kind of eagle' | 'a kind of |
| | | | | | eagle' |
| 11:13 | עְזְנִיָּה ['āzənîyāh] | ʿazniya | ʿazniya | deŋiz kartalï | deŋiz ķartalï |
| | 'black vulture' | | | 'sea-eagle' | 'sea-eagle' |
| 11:14 | דָאָה [$dar{a}$ $^{\circ}ar{a}h$] | da 'a | da ʾa | aķ baba | aķ baba |
| | 'eagle and/or | | | 'vulture' | 'vulture' |
| | vulture; kite; red | | | | |
| | kite' | | | | |
| 11:14 | אַיָּה [ʾayyāh] | aya | aya | čaylaķ | čaylaķ |
| | 'falcon; kite; hawk; | | | 'kite' | 'kite' |
| | black kite' | | | | |
| 11:16 | יַעֲנָה [yaʿănāh] | anķït | anķīt | deve ķušï | deve ķušu |
| | 'female ostrich; | | | | |
| | ostrich' | | | | |
| 11:16 | ๆกูซู [šāḥap̄] | šaḥaf | šaḥaf | ķuķu ķušï | ķuķu ķušï |
| | 'seagull' | | | 'cuckoo' | 'seagull'? |
| | | | | | (see CKED: |
| | | | | | 319; KRPS: |
| | | | | | 374)' |

| Lev | Biblical Hebrew | ADub.III.73 | JSul.III.01 | Kieffer 1827 | Göz. 1841 |
|-------|-----------------------------------|-------------------|--------------------|--------------------|----------------|
| | | (T.Kar.B) | (H.Kar.B) | (Ot.Tur.B) | |
| 11:16 | נץ [nêṣ] | ķïryïy | ķïrүïy | doyan | duyan |
| | 'hawk; falcon' | | | 'hawk; falcon' | 'hawk; |
| | | | | | falcon' |
| 11:17 | ּשֶׁלְנְ [šālā <u>k</u>] | šalaḫ | šalaķ | ķara bataķ | ķara bataķ |
| | 'cormorant; fish | | | 'cormorant' | 'tern?' (see |
| | owl' | | | | CKED: 290; |
| | | | | | KRPS: 363) |
| 11:17 | יַנְשׁוּף [yanšūp] | yanšuf | ķuzyun | to/tu | toyï |
| | 'long-eared owl; | | 'owl' | 'a species of owl' | 'ibis?' (see |
| | bee-eater' | | | | CKED: 410; |
| | | | | | KRPS: 535) |
| 11:18 | תִּנְשֶׁמֶת [tinšeme <u>t]</u> | tinšemet | tinšemet | ķиүи | ķоүи |
| | 'white owl; | | | 'swan' | 'swan' |
| | chameleon' | | | | |
| 11:18 | קאַת [<i>k̄āʾā<u>t</u></i>] | ķaʾat | puhac | ķašıkjï ķušï | ķašïkčï ķušï |
| | 'pelican; little owl; | | 'a species of owl' | 'pelican' | 'pelican' |
| | jackdaw' | | - | _ | _ |
| 11:18 | ַרָּחָם [<i>râḥâm</i>] | raḥam | гаḥат | | raḥamï |
| | 'carrion vulture' | | | | |
| 11:19 | חָסִידָה [ḥăsîḏāh] | <u></u> ḥasida | busyol | legleg | legläg |
| | 'stork | | | | |
| 11:19 | אָנָפָה [anafa] | anafa | anafa | balïkčïn | balïkčïn |
| | 'heron; egret' | | | 'heron' | 'heron' |
| 11:19 | דּוּכִיפַת [dūkַîp̄at] | duḫifat | duḫifat | hüdhüd | hüdhüdi |
| | 'hoopoe' | • • | | 'hoopoe' | 'hoopoe' |
| 11:19 | ַ יְעַטַלִּף [ʿ <i>ăṭallêp̄</i>] | yarï ķanat | yarï ķanat | šebpere | <u> </u> |
| | 'bat' | | | 'bat' | 'bat' |
| 11:22 | סָלְעָם [sālə ʿām] | salʿam | salʿam | judjud/jüdjüd | judjud/jüdjüd |
| | 'a kind of locust' | | | 'cicada' | 'cicada' |
| 11:22 | חָרְגֹל [ḥargōl] | ḥargol | <u> </u> | <u>ḥargol</u> | ḥargol |
| | 'a kind of locust' | | | | |
| 11:22 | קגָב [<u>ḥāḡāḇ]</u> | <u></u> ḥagav | <u> </u> | <u></u> ḥagab | <u> </u> hagav |
| | 'locust; | | | ~ | |
| | grasshopper' | | | | |
| 11:29 | | <u></u> holed | <u></u> ḥoled | gelinjik | gelinčik |
| | 'mole-rat, mole | - | • | 'weasel' | S |
| | weasel' | | | | |

| Lev | Biblical Hebrew | ADub.III.73 | JSul.III.01 | Kieffer 1827 | Göz. 1841 |
|-------|---------------------------------------|-------------|-------------|--------------------|-----------------|
| | | (T.Kar.B) | (H.Kar.B) | (Ot.Tur.B) | |
| | | | | | 'mole?' (see |
| | | | | | CKED: 164; |
| | | | | | KRPS: 166) |
| 11:29 | ַעַּכְּר [' $a\underline{k}bar{a}r$] | ʻaḫbar | scïcan | sïčan | sïčan |
| | 'mouse' | | | 'rat; mouse' | 'rat; mouse' |
| 11:29 | ַ צָב [ṣāḇ] | cav | scur | ķaplubuya 'turtle; | ķaplï baya |
| | 'turtle; turtoise; | | | turtoise' | 'turtle; |
| | a kind of lizard' | | | | turtoise' |
| 11:30 | אֲנָקָה [ʾănāķāh] | anaķa | anaķa | sivri sïčan 'shrew | sivri sïčan |
| | 'gecko; a kind of | | | mouse' | 'shrew |
| | lizard' | | | | mouse' |
| 11:30 | תַּם [kōaḥ] | kovaḥ | koaḥ | güneš keleri | güneš |
| | 'a kind of lizard' | | | lit. 'sun lizard' | kelerisi |
| | | | | | lit. 'sun |
| | | | | | lizard' |
| 11:30 | לְטָאָה [$l 	atilde{a} \hat{a} h$] | leta ʾa | leta ʾa | yïldïz keleri | yïldïz kelerisi |
| | 'a kind of lizard' | | | lit. 'star lizard' | lit. 'star |
| | | | | | lizard' |
| 11:30 | הֹמֶט [ḥōmeṭ] | ḥomet | ḥomet | kertenkele | kerten kelesi |
| | ʻlizard' | | | 'lizard' | 'lizard' |
| 11:30 | תַּנְשֶׁמֶת [tinšāme <u>t</u>] | tinšamet | tinšamet | köstebek | göz töbä |
| | 'white owl; | | | 'mole' | 'mole' |
| | chameleon' | | | | |

Upon inspection, all instances within our dataset align with the Ottoman Bible translation, while the Halich and Trakai Karaim versions exhibit variances, occasionally presenting untranslated Biblical Hebrew terms. The apparent reliance on Ottoman Bible translations clearly stems from challenges associated with the interpretation of ambiguous animal names. However, this replication process introduces intriguing errors, which are also erroneously reflected in dictionaries. An examination of these problematic items reveals several systematic errors. One of them is unnecessary duplication of accusative markers.

Table 91: Usage of Double Accusative Case Markers in the Dataset (Işık 2021: 366)

| Lev | Kieffer 1827 | Göz. 1841 |
|-------|--------------|-----------|
| 11:13 | iron+ï | iron+ï+nï |

| Lev | Kieffer 1827 | Göz. 1841 | |
|-------|----------------------------|--|--|
| | eagle+ ACC (Ogh.) | eagle+ACC (Ogh.)+ACC (Kip.) | |
| 11:16 | bayķuš+ï | bay kuš+ï+nï | |
| | owl+ACC (Ogh.) | owl+ACC (Ogh.)+ACC (Kip.) | |
| 11:17 | tu+yï | to+yï+nï | |
| | owl+ACC (Ogh.) | owl?+ACC (Ogh.)+ACC (Kip.) | |
| 11:18 | raḥam+ï | raḥam+ï+nï | |
| | carrion vulture+ACC (Ogh.) | carrion vulture+ACC (Ogh.)+ ACC (Kip.) | |
| 11:19 | hüdhüd+i | hüdhüd+i+ni | |
| | hoopoe+ACC (Ogh.) | hoopoe+ACC (Ogh.)+ACC (Kip.) | |
| 11:20 | ǯüdǯüd+i | ǯüdǯüd+i+ni | |
| | cicada+ACC (Ogh.) | cicada+ACC (Ogh.)+ACC (Kip.) | |

Another replication error entails the superfluous duplication of 3SG.POSS markers in three compound nouns¹⁴¹, along with the unnecessary addition of a 3SG.POSS marker in another compound noun.

Table 92: *Unusual Usage of 3SG.POSS Suffixes in the Dataset* (Işık 2021: 366)

| Lev | Kieffer 1827 | Göz. 1841 | |
|---------------------------------|----------------------|-------------------------------|--|
| 11:29 | güneš keler+i | güneš keler+i+si | |
| | sun lizard+3SG.POSS | sun lizard+3SG.POSS+3SG.POSS | |
| 11:29 | yïldïz keler+i | yïldïz keler+i+si | |
| | star lizard+3SG.POSS | star lizard+3SG.POSS+3SG.POSS | |
| 11:29 kertenkele kerten kele+si | | kerten kele+si | |
| | lizard | lizard+3SG.POSS | |

Lastly, it is noteworthy that a previous study's (Işık 2021) findings suggest that five words featured in the Crimean Karaim dictionaries may be inaccurately defined. The unusual forms in Göz. 1841 appear to have led dictionary compilers to draw comparisons with other Bible translations when formulating their descriptions. As previously highlighted, various Bible translations often reference differing animals within identical verses. It is crucial to note that all these problematic words originate from Shapsal's lexical corpus and are thus incorporated in the KRPS dictionary. This dictionary lists Karaim words in conjunction with Russian and Polish descriptions. As demonstrated earlier (Işık 2021: 367–368), the comparison between the

_

¹⁴¹ Regarding the structure of compound nouns, see Section 2.2.2.1.2.

Russian description of these five words in the KRPS and corresponding sections of a 19th-century Russian Bible translation (RSO 1876) reveals a clear parallel. Subsequently, the uncertain and erroneous animal nomenclature of Göz. 1841 in the dictionaries, is likely the result of comparisons with equivalent biblical verses from a Russian Bible. This approach, however, led to misinterpretations, as these words originally referred to distinct animals, corroborated by the evidence presented in Kieffer 1827.

Table 93: Erroneous Lexical Items in Karaim Dictionaries

| Lev | Göz. 1841 | Karaim Dictionaries | RSO 1876 | Kieffer 1827 |
|-------|-----------|--------------------------|-------------|-------------------------------|
| 11:17 | toyï | 'ibis' | ибис 'ibis' | to 'a kind of owl'+ACC > toyï |
| | | (CKED: 410; KRPS: 535) | | (Işık 2021: 367) |
| 11:16 | ķuķu kušï | (appears solely as kuku) | чайки | ķuķu kušї 'cuckoo' |
| | | 'seagull; mew' | 'seagulls' | |
| | | (CKED: 319; KRPS: 374) | | |
| 11:17 | ķarabataķ | 'tern; fisher (bird)' | рыболов | karabatak 'cormorant' |
| | | (CKED: 290; KRPS: 363) | 'fisher | |
| | | | (bird)' | |
| 11:17 | aķ-baba | 'kite' (KRPS: 377) | коршун | aķ baba 'vulture' |
| | ķušï | | 'kite' | |
| 11:30 | gelinčik | 'mole' | крот 'mole' | gelinjik 'weasel' |
| | | (CKED: 164; KRPS: 166) | | |

3. Conclusion

3.1. Phonology

Following a comprehensive analysis, it is evident that the phonemic inventory of the scrutinized text displays no marked disparities when contrasted with Crimean Tatar, Crimean Turkish, and Ottoman Turkish. In contrast, in line with expectations, the text does not incorporate the unique phonemes of Western Karaim, which demonstrate Slavonic influence. A deeper examination of the phonological attributes indicates that certain features, when juxtaposed with the aforementioned languages, do not exhibit a homogenous characteristic. These features predominantly hinge on chosen lexical items, complicating the identification of stable characteristic traits, e.g., fluctuations between the vowels /a/ ~ /ä/ in words [Lev 1:5 vs 26:5] sač- vs säč- 'to sprinkle' [Lev 24:7; 24:6] yaraštirmaķ vs yäräštirmäk 'arrangement' [Lev 27:7 vs 27:6] yaš vs yäš 'years; age', addition of a vowel in [Lev 11:25] umundar (see [Lev 5:2] mundar) 'unclean' irenk (see [Lev 13:55] renk) 'colour', [Lev 11:23] mikruḥa 'abominable'

(see [Lev 11:11] mikruh). Nonetheless, it is worthy of note that when comparisons are drawn, certain discrepancies become apparent between the studied languages. A number of features reveal instances of divergence from Ottoman Turkish characteristics while mirroring a recurring pattern discernible in Crimean Tatar, Crimean Turkish, and Western Karaim dialects in specific instances. These encompass the degree of openness in vowels, the adaptation of fronting harmony in some non-Turkic lexemes, the transformation of the velar voiceless plosive -k into the velar fricative -k, metathesis in some words, and the shifting of vowel-consonant pairs from [iy], [ay] to [uv] or [ov] (see sections 2.1.2.1. and 2.1.2.2.2.). In regard to the interrelationships between our dataset and the Western Karaim dialects, it can be posited that, barring a few characteristics such as the alternation from t->c- and t->k-, and the omission of consonants in certain Arabic words (see sections 2.1.2.2.2.2. and 2.1.2.2.2.4.), there is no significant association in features that diverge from Crimean Turkish, Crimean Tatar, and Ottoman Turkish. Notably, with regard to Crimean Tatar, we do not identify any distinctive phonological features that are exclusively present in both Crimean Tatar and our dataset.

In categorizing the phonological characteristics, the most salient features are those pertaining to Ottoman Turkish consonantal alternations (see section 2.1.2.2.1.), which fundamentally underpin the classification of Crimean Karaim variants. It is well-established that Bible translations penned in Crimean Karaim phonologically exhibit Kipchak traits, while secular texts, such as mejumas, frequently display Ottoman Turkish attributes. As alluded to previously, these distinctions are recognized in Jankowski's taxonomy (2015: 202–205), where the variant embodying Ottoman Turkish characteristics is designated as Crimean Turkish Karaim, and the one demonstrating Kipchak traits is labeled as Crimean Kipchak Karaim. Our text reveals some intriguing observations, as the influence of Ottoman Turkish phonological features is notably pronounced in a Crimean Karaim Bible translation. We classify such Ottoman Turkish consonantal features into seven subgroups. Among them certain Ottoman Turkish traits are dominant over the Kipchak counterparts, e.g., k- ~ g- (8% vs 92%); b- ~ v-(0% vs 100%), while others are not, e.g., $t \sim d$ - (59% vs 41%); $b \sim \emptyset$ (92% vs 8%); $b \sim m$ -(7% vs 93%), $b \sim p$ - (86% vs 14%), $ol \sim o$ (95% vs. 5%). It's important to underscore that the distribution of these features is characteristically heterogeneous. This can also be further corroborated by the occasional presence of both Ottoman Turkish and Kipchak equivalents within the same sentences in our dataset. However, in specific chapters, for instance, Chapter 11, such Ottoman Turkish features are highly predominant.

In conclusion, while it is arduous to categorize the variant of Crimean Karaim utilized in our text, given its amalgamation of both Crimean Kipchak Karaim and Crimean Turkish Karaim traits, it should be underscored that pronounced Crimean Turkish features are unusual for Crimean Karaim Bible translations. This aspect renders our text distinct among other Crimean Karaim Bible translations¹⁴². Accordingly, it can be convincingly proposed that, despite preserving fundamental Kipchak traits, the text also mirrors the phonological tendencies prevalent during the linguistic epoch of its publication.

3.2. Morphology and Morphophonology

The morphological and morphophonological attributes observed in our dataset once more showcase a fusion of characteristics. Primarily, in terms of general tendencies of morphophonological traits, numerous analogous patterns emerge. These are identifiable in our dataset and within the other Turkic languages explored in this current analysis, such as intersyllabic vowel harmony, consonant harmony, intervocalic lenition, and vowel deletion (refer to section 2.2.1.). Nonetheless, our text reveals certain unique characteristics specific to our dataset, which are absent in other languages. These include metathesis in a single inflected lexical item and a consonant deletion mostly in the agrist marker (refer to sections 2.2.1.5. and 2.2.1.6.). Yet, the most notable difference lies in the presence of rounded forms of particular suffixes which are absent in Crimean Tatar, Crimean Karaim Bible translations or as depicted in Prik's (1976) account for Crimean Karaim. Examples of these distinctions include the following: denominal nominal derivative markers {+IU}, {+IUK}; the deverbal nominal derivative markers $\{-U\}$, $\{-(U)K\}$, $\{-(U)m\}$, $\{-(U)\S\}$, $\{-(u)v\S\}$, $\{-(u)v\S\}$, the possessive markers $\{+(u)m\}$, $\{+(U)\eta\}$, $\{+(U)\eta Uz\}$; the accusative case marker $\{+nu\}$; the ordinal marker $\{+(U)n3I\}$; the passive voice marker $\{-UI\}$; the causative voice marker $\{-Ur\}$, $\{-dUr\}$, {-yuz}; the reflexive voice marker {-un}, the definite past tense marker {-Du}; the agrist marker $\{-(U)r\}$; the imperative mood markers $\{-(U)\eta Uz\}$, $\{-(U)\eta Iz\}$; the voluntative mood marker {-sUn}; the converb marker {-up}; and the participle marker {-(u)vči} (see sections 2.2.2. and 2.2.3.). It's crucial to acknowledge that these rounded variants also have unrounded counterparts, which are significantly more prevalent compared to the rounded versions. In addition, these characteristics should not be ascribed to the influence of Western Karaim. Instead, they seem to faintly echo an Ottoman Turkish morphophonological process which was

¹⁴² An exception exists with the so-called Ortaköy 1832-1835 edition, which was not published in Crimea, and its language is occasionally referred to solely as Turkish while it also presents some Karaim features (Jankowski 2018: 52–53).

also, albeit scarcely, prevalent in Crimean Turkish. Indeed, apart from the presence of the present tense marker {-AydIr} (see section 2.2.3.3.2.), our dataset does not exhibit any distinct Western Karaim features pertaining to this category.

In regard to the morphological dimension, the text uncovers a series of Ottoman Turkish markers that are not found in Prik's description of Crimean Karaim or Crimean Karaim Bible translations. Examples include the participle markers {-(y)An}, {-ïǯi}, {-DIk}; the first person voluntative mood marker {-(A)IIm}; the third person singular optative mood marker {-(y)A}; and the future tense marker {-AǯAK} (see sections 2.2.3.6., 2.2.3.4.2., and 2.2.3.3.4.). It's crucial to highlight that, except for the 1PL.VOL marker {-(A)IIm}, all the markers introduced have Kipchak equivalents within our text. Remarkably, these Kipchak variants predominate over their Ottoman Turkish counterparts (88% vs 12%). However, it is only in specific chapters, such as Chapter 11, that Ottoman Turkish characteristics are more prevalent. Nevertheless, the overarching conclusion is that while our text does display Ottoman Turkish morphophonological and morphological characteristics, they do not exhibit predominance. This stands in contrast to certain phonological characteristics discussed in the preceding subsection.

3.3. Syntax and Morphosyntax

The syntactic and morphosyntactic characteristics observed in the text yield anticipated results, as all Karaim Bible translations exhibit various influences from Biblical Hebrew features. These include the inverse order in genitive constructions, systematic renderings of Biblical Hebrew conjunctions such as ki and wa, the Hebrew definite article, repetition of certain nominals and postpositional structures, and the use of infinitival paronomasia by certain Karaim equivalents. Furthermore, these influences pervasively shape the structures and word order in both simple and complex sentences, elements that are atypical for Turkic languages. Perhaps the most common influence in complex sentences is the rendering of the Hebrew relative pronoun אַשֶּׁר ['ăšer] by certain pronouns. This presents an atypical relative clause structure that is common in Karaim and also existed in written Ottoman sources due to Persian influence. On the other hand, the text also exhibits some Turkic characteristics, as we observe the use of participles in relative clauses, the usage of specific conjunctions, and verbal noun markers in adverbial clauses, which bear similarities to aspects of Crimean Tatar, Ottoman Turkish, and other Karaim dialects. Furthermore, an analysis of case marker functions reveals that out of 43 different functions, only three of them demonstrate atypical Turkic usages, likely the result of Biblical Hebrew influence. Additionally, with respect to a specific Biblical Hebrew influence, such as the reduplication of specific nominals and postpositional constructions, e.g., [Lev 6:5] ertä bilän ertä bilän 'every morning'; [Lev 17:3] kiši kiši 'every man', our text diverges from the previously examined Western Karaim Bible translations. Instead, it exhibits similarities with another Crimean Karaim Bible translation (specifically, the manuscript BSMS 288). In conclusion, our text adheres closely to the parameters of Karaim Bible translation, reflecting an age-old canonical tradition along with certain Turkic characteristics which cannot be attributed to a specific Turkic language.

3.4. Lexicon

Within our dataset, we identify a total of 1075 unique lexical elements, 939 of which are documented in Karaim dictionaries. Upon evaluating the representation of these words in the languages exhibited in our text, it is frequently observed that a majority of the words are prevalent across all languages, albeit with differing phonological adaptations. In examining the lexical content of Western Karaim dialects, it is notable that, barring two verbs showing the Western Karaim phonological adaptations (see section 2.4.1.), our dataset displays no discernible influence from such Karaim dialects. Interestingly, our data also lacks any specific Crimean Tatar words. However, the situation is markedly different with respect to Ottoman Turkish. According to the CKED, 209 of the 939 listed examples originate from Ottoman Turkish as adopted in Crimean Karaim. In addition, our investigation reveals that this number increases to 286 out of 1075 total items when unlisted lexical items are taken into account. Thus, Ottoman Turkish served as a donor language for approximately 27% of the lexical elements in our dataset, contributing words of Turkic, Arabic, Persian, Greek, Italian and Armenian origin. The influence of Ottoman Turkish within the chapters is not uniformly distributed. Notably, Chapter 11 evidences a pronounced use of Ottoman Turkish lexicon and a systematic replacement of two Karaim words with their Arabic-origin Ottoman Turkish equivalents.

In relation to Biblical Hebrew, our dataset includes 38 words that, while absent in dictionaries, are present in Karaim Bible translations. Additionally, there exist twenty unique lexical items within our data that do not appear in any other sources. These items are primarily derived from common Turkic verb and nominal stems.

In concluding the examination of this category, we can revisit the insights derived from the previous subchapter. Although the text prominently features lexical elements intrinsic to Crimean Kipchak Karaim, a substantial presence of Crimean Turkish Karaim features is also

observed, attributed to the influence of Ottoman Turkish. Notably, such an intensive level of Ottoman Turkish lexical influence is yet to be observed in other Crimean Karaim Bible translations, except for the so-called Ortaköy 1835 edition, the language of which is sometimes referred to as Turkish.

3.5. Final Remarks

This study meticulously scrutinizes the linguistic properties of a section of the printed edition known as the Gözleve Bible, the language of which has sparked scholarly discourse. Our exhaustive investigation delineates these linguistic elements in detail, providing a comprehensive comparison with Crimean Tatar, Ottoman Turkish, all three Karaim dialects, and occasionally, Crimean Turkish. It is clear that the text exemplifies translation strategies typically found in all Karaim translations. This includes a literal translation approach, such as reflecting Biblical Hebrew syntax or replicating various linguistic characteristics of Biblical Hebrew that are atypical in Turkic languages. However, in a wide array of categories, the text appears to undergo Ottomanization across phonological, morphophonological, morphological, and lexical dimensions, sporadically incorporating direct borrowings from Ottoman Bible translations. Simultaneously, the text reveals pronounced Kipchak traits in these categories, which predominantly outweigh the Ottoman Turkish equivalents in general. Notably, the text intermittently showcases specific phonological features and lexical elements which, while not atypical for Turkic languages, are absent in other languages that have been compared with our dataset. In light of these findings, Shapira's assertion (2003: 696; 2013: 134) that this edition represents a hasty attempt to Tatarize or vulgarize older Łuck editions, resulting in superficial Tatarization, does not align with our results. Quite the contrary, the text does not display any features that are solely characteristic of Crimean Tatar. Rather, it manifests common Karaim traits alongside non-dominant Ottoman Turkish features, which were prevalent in both Crimean Turkish and Crimean Tatar. Therefore, based on Jankowski's descriptions (2015a: 202–205) of Crimean Karaim variants, the language of the Lev of Göz. 1841 encompasses a blend of Crimean Kipchak Karaim and Crimean Turkish Karaim. Undoubtedly, a more comprehensive analysis is required to provide an accurate portrayal of the entire edition.

References

- ABISH, Aynur 2022. 'Kazakh and Karakalpak'. In: Lars JOHANSON and Éva Á. CSATÓ (eds.) *The Turkic Languages*. London and New York: Routledge, 337–353.
- ANKORI, Zvi 1968. Karaites in Byzantium: The Formative Years, 970–1100. New York: AMS Press.
- AQTAY, Gülayhan 2009. Eliyahu ben Yosef Qılcı's Anthology of Crimean Karaim and Turkish Literature. İstanbul: Mehmet Ölmez Yayınları.
- ASTREN, Fred 2003. 'Karaite Historiography And Historical Consciousness'. In: Meira POLLIACK (ed.) *Karaite Judaism. A Guide to its History and Literary Sources*. Leiden, Boston: Brill, 25–69.
- ASTREN, Fred 2004. *Karaite Judaism and Historical Understanding*. Columbia: University of South Carolina Press.
- BARTHOLD, Vasily Vladimirovich 2004. *Orta Asya Türk Tarihi Dersleri* (Translated by DAĞ Hüseyin). Ankara: Çağlar Yayınları.
- BERTA Árpád and CSATÓ Éva Á. 2022a. 'Middle Kipchak'. In: Lars JOHANSON and Éva Á. CSATÓ (eds.) *The Turkic Languages*. London and New York: Routledge, 152–159.
- BERTA Árpád and CSATÓ Éva Á. 2022b. 'West Kipchak Languages'. In: Lars JOHANSON and Éva Á. CSATÓ (eds.) *The Turkic Languages*. London and New York: Routledge, 320–336.
- BESALEL, Yusuf 2001. Yahudilik Ansiklopedisi. Vol. III. İstanbul: Gözlem Yayıncılık.
- BOZOK, Emrah 2018. 'Hırsız Sözcüğünün Kökeni Üzerine'. *Uluslararası Beşeri Bilimler ve Eğitim Dergisi* 4 (9): 63–71.
- CANSDALE, George Soper 1970. Animals of Bible Lands. Exeter: Paternoster.
- CC = GRØNBECH, Kaare 1942. Komanisches Wörterbuch. Türkischer Wortindex zu Codex Cumanicus [Monumenta Linguarum Asiae Maioris Subsidia 1]. Kopenhagen: Munksgaard.
- CEDHL = Klein, Ernest 1987. A Comprehensive Etymological Dictionary of the Hebrew Language for Readers of English. Israel: Carta Jerusalem.
- CHAFUZ, Mark Ezrovič 1995. Russko-karaimskij slovaŕ. Krymskij dialekt. Moskva: Rossijskaja Akademija Nauk.
- CKED = AQTAY Gülayhan and JANKOWSKI Henryk 2015. *A Crimean Karaim-English Dictionary*. Poznań: Uniwersytet im. Adama Mickiewicza w Poznaniu. Katedra Studiów Azjatyckich. Wydział Neofilologii.

- CPED = STEINGASS, Francis 1963. *A Comprehensive Persian-English Dictionary*. London: Routledge & Kegan Paul Limited.
- CrKB = Jankowski Henryk, Aqtay Gülayhan, Cegiołka Dorota, Çulha Tülay and Németh Michał 2019. *The Crimean Karaim Bible*. Vol. I-II. Wiesbaden: Harrossowitz Verlag.
- COŞAR, Mevhibe A. 2015. 'Bir Kimlik İşaretleyicisi Olarak Dil ve Trabzon Ağızlarında Arkaik Hususiyetler'. *Karadeniz Araştırmaları Enstitüsü Dergisi* 1 (1), 241–253.
- CSATÓ, Éva Á. 2006. 'The Karaims: The Smallest Group of Turkic-Speaking People'. In: Ergun ÇAĞATAY and Doğan KUBAN (eds.) *The Turkic Speaking Peoples. 2,000 years of Art and Culture from Inner Asia to the Balkans.* Munich, Berlin, London, New York: Prestel, 384–403.
- CSATÓ, Éva Á. 2011. 'Word order properties in Trakai Karaim biblical translations'. In: Eser ERGUVANLI TAYLAN and Bengisu RONA (eds.) *Puzzles of Language: Essays in Honour of Karl Zimmer*. Wiesbaden: Harrossowitz Verlag, 169–186.
- ÇULHA, Tülay 2006. *Karaycanın Kısa Söz Varlığı*. *Karayca-Türkçe Kısa Sözlük* [Dil ve Edebiyat Dizisi 6]. İstanbul: Kebikeç Yayınları.
- ÇULHA, Tülay 2010. Kırım Karaycasının Katık Mecuması. Metin-Sözlük-Dizin. İstanbul: Mehmet Ölmez Yayınları.
- Çulha, Tülay 2019. *Karaycanın Karşılaştırmalı Grameri. Fonetik-Morfoloji (Kırım Ağzı Esasında*). Ankara: Türk Dil Kurumu Yayınları.
- DANON, Abraham 1925. 'The Karaites in European Turkey. Contributions to Their History Based Chiefly on Unpublished Documents'. In: *The Jewish Quarterly Review*. Vol. 15, No. 3. Philadelphia: University of Pennsylvania Press, 285–360.
- DOERFER, Gerhard 1959a. 'Das Krimosmanische'. In: Jean DENY et al. (eds.) *Philologiae Turcicae Fundamenta*. Wiesbaden: F. Steiner, 272–280.
- DOERFER, Gerhard 1959b. 'Das Krimtatarische'. In: Jean DENY et al. (eds.) *Philologiae Turcicae Fundamenta*. Wiesbaden: F. Steiner, 369–390.
- DUBIŃSKI, Aleksander 1991. 'Die Karaimen. Eine türkische Minderheit desalttestamentischen Glaubens'. *Acta Orientalia Belgica* 6: 213–225.
- DUBIŃSKI, Aleksander 1993. 'Lokalizacja języka karaimskiego w świetle jegorozwoju historycznego'. *Przegląd Orientalistyczny* nr 165–166: 37–42.
- EDPT = CLAUSON, Sir Gerard 1972. *An Etymological Dictionary of Pre-Thirteenth Century Turkish*. Oxford: The Clarendon Press.

- EFE, Kürşat 2017. 'Tarihî Türk Lehçelerinde Alıntı Hayvan İsimleri'. *Route Educational and Social Science Journal* 4 (5): 379–393.
- ERDAL, Marcel 1991a. *Old Turkic Word Formation*. Vol. I [Turcologica 7]. Wiesbaden: Otto Harrassowitz.
- ERDAL, Marcel 1991b. *Old Turkic Word Formation*. Vol. II [Turcologica 7]. Wiesbaden: Otto Harrassowitz.
- ERDAL, Marcel 2004. A Grammar of Old Turkic. Leiden: Bril.
- ETD = REDHOUSE, J. W. 1856. English and Turkish Dictionary. London: Bernard Quaritch.
- FERGUSON, Walter 1974. Living Animals of the Bible. New York: Scribner.
- FISHER, W. Alan 1978. Crimean Tatars. California: Hoover Institution Press.
- GOLDEN, Peter 1971. 'Hazar dili'. Türk Dili Araştırmaları Yıllığı Belleten. Vol. 19: 147–157.
- GOLDEN, Peter 2001. 'Khazars'. In: Mehmet TÜTÜNCÜ (ed.) *Turkish-Jewish Encounters*. Haarlem: Stichting SOTA, 29–49.
- GORDLEVSKIJ, Vladimir Aleksandrovič 1928. 'Leksika karaimskogo perevoda Biblii'. *Doklady Akademii Nauk SSSR* 5: 87–91.
- GÜLSEVİN, Selma 2013. 'Karay Türkçesinin Kendisine Has Özellikleri ve Türk Lehçeleri Arasındaki Yeri'. *Karadeniz Araştırmaları* Kış, 36: 207–222.
- GÜLSEVİN, Selma 2016. *Karay Türklerinin Dili (Troki Diyalekti*). Ankara: Türk Dil Kurumu Yayınları.
- HAGOPIAN, V. H. 1907. Ottoman-Turkish Conversation. London: Samson Loro Marston.
- HARKAVY, Abraham 1904. 'Karaites And Karaism'. In: Isidore SINGER et al. (eds.) *The Jewish Encyclopedia*. Vol. VII. New York, London: Funk and Wagnalls Company, 553–556.
- HARKAVY, Abraham and STRACK Hermann L. 1875. Catalog der hebräischen Bibliothek in St. Petersburg. Vol. 1–2. St. Petersburg: Ricker.
- HARVIAINEN, Tapani 2003. 'The Karaites in Eastern Europe and the Crimea: An Overview'. In: Meira Polliack (ed.) *Karaite Judaism. A Guide to its History and Literary Sources*. Leiden, Boston: Brill, 633–655.
- HARVIAINEN, Tapani 2013. 'Karaite Pronunciation Traditions: Modern'. In: Geoffrey Khan et al. (eds.) *Encyclopedia of Hebrew Language and Linguistics*. Vol. 2 'G-O'. Leiden, Boston: Brill, 453–457.
- IŞIK, Murat 2018. 'Oghuzic and Kipchak Characteristics in the Book of Leviticus, Gözleve Bible (1841)'. *Rocznik Orientalistyczny* LXXI/2: 66–76.

- IŞIK, Murat 2020a. 'The Animal Names in the Book of Leviticus of the Gözleve Bible (1841).

 Part I: Mammal, Insect and Reptile Species'. In: István ZIMONYI (ed.) *Ottomans Crimea Jochids. Studies in Honour of Mária Ivanics*. Szeged: University of Szeged,

 Department of Altaic Studies, 145–163.
- IŞIK, Murat 2020b. 'Some Biblical Hebrew Influence on the Karaim Bible Translations: The Book of Leviticus, Gözleve Bible (1841)'. In: Balázs BALATONI and Dorottya MIHÁLYI (eds.) *Mediterrán Tanulmányok XXX*. Szeged: University of Szeged, Department of Modern History and Mediterranean Studies, 115–130.
- IŞIK, Murat 2021. 'The Animal Names in the Book of Leviticus of the Gözleve Bible (1841).

 Part II: Bird Species'. *Acta Orientalia Academiae Scientiarum Hungaricae* 74 (2): 345–372.
- JANKOWSKI, Henryk 1994. 'Jak krymscy Karaimowie czytali pataḥ i co z tego wynika?'. In: Elżbieta Górska and Barbara Ostafin (eds.) *Studia Orientalia Thaddaeo Lewicki Oblata*. Kraków: The Enigma Press, 107–114.
- JANKOWSKI, Henryk 1997. 'A Bible Translation into the Northern Crimean Dialect of Karaim'. *Studia Orientalia* 82: 1–84.
- JANKOWSKI, Henryk 2003a. 'On the Language Varieties of Karaims in the Crimea'. *Studia Orientalia* 95: 109–130.
- JANKOWSKI, Henryk 2003b. 'Position of Karaim among the Turkic Languages'. *Studia Orientalia* 95: 131–153.
- JANKOWSKI, Henryk 2004. 'Karaims of the Crimea and Eastern Europe: Some Questions of Ethnicity and Identification'. *Studia Orientalia* 99: 83–92.
- Jankowski, Henryk 2008. 'The question of the existence of Crimean Karaim and its relation to Western Karaim'. In: Tamara Bairašauskaite, Halina Kobeckaite, and Galina Miškiniene (eds.) Orientas Lietuvos Didžiosios Kunigaikštijos visuomenes tradicijoje: totoriai irkaraimai. Orient in the Social Tradition of the Grand Duchy of Lithuania: Tatars and Karaims. Orient w tradycji społeczeństwa litewskiego: Tatarzy i Karaimi. Vilnius: Universiteto Leidykla, 161–168.
- JANKOWSKI, Henryk 2009. 'Translations of the Bible into Karaim'. *Religion Compass* 3/4: 502–523.
- JANKOWSKI, Henryk 2010. Język krymskotatarski. Warszawa: Dialog.
- JANKOWSKI, Henryk 2012. 'Literatura krymskokaraimska'. *Przegląd Orientalistyczny* 1–2: 50–68.

- JANKOWSKI, Henryk 2013. Karaim mejumas in Eupatoria. In: Tatiana PANG, Simone-Christiane RASCHMANN, and Gerd WINKELHANE (eds.) Unknown Treasures of the Altaic Worldin Libraries, Archives and Museums: 53rd Annual Meeting of the Permanent International Altaistic Conference, Institute of Oriental Manuscripts, RAS St. Petersburg, July 25–30, 2010. Berlin: Klaus Schwarz, 245–262.
- JANKOWSKI, Henryk 2015a. 'Crimean Turkish Karaim and the Old North-western Turkic Tradition of the Karaites'. *Acta Orientalia Academiae Scientiarum Hungaricae* 68/2: 199–214.
- JANKOWSKI, Henryk 2015b. 'Karaim and Krymchak'. In: Lily KAHN and Aaron RUBIN (eds.) *Handbook of Jewish Languages*. Leiden, Boston: 451–488.
- JANKOWSKI, Henryk 2018. 'Translation of the Tanakh into Crimean Karaim: History, Manuscripts, and Language'. In: Lily KAHN (ed.) Jewish Languages in Historical Perspective. [IJS Studies in Judaica. Vol. 17.] Leiden: Brill, 39–61.
- JOHANSON, Lars 2022a. 'The Structure of Turkic'. In: Lars JOHANSON and Éva Á. CSATÓ (eds.) *The Turkic Languages*. London and New York: Routledge, 26–59.
- JOHANSON, Lars 2022b. 'The history of Turkic'. In: Lars JOHANSON and Éva Á. CSATÓ (eds.)

 The Turkic Languages. London and New York: Routledge, 83–120.
- JOHANSON, Lars 2022c. 'East Old Turkic'. In: Lars JOHANSON and Éva Á. CSATÓ (eds.) *The Turkic Languages*. London and New York: Routledge, 132–144.
- JOOSTEN, Jan 2009. 'Three Remarks on Infinitival Paronomasia in Biblical Hebrew'. In: D. SIVAN, D. TALSHIR & C. COHEN (eds.) Zaphenath-Paneah. Linguistic Studies Presentedto Elisha Qimron on the Occasion of his Sixty-Fifth Birthday. Beer Sheva: Ben Gurion University of the Negev, 99–113.
- KARAKOÇ, Birsel 2022. 'Turkmen'. In: Lars JOHANSON and Éva Á. CSATÓ (eds.) *The Turkic Languages*. London and New York: Routledge, 262–286.
- KAVITSKAYA, Darya 2010. Crimean Tatar. Muenchen: Lincom Europa.
- KERSLAKE, Celia 2022. 'Ottoman Turkish'. In: Lars JOHANSON and Éva Á. CSATÓ (eds.) *The Turkic Languages*. London and New York: Routledge, 174–194.
- KISSLING, Hans Joachim 1960. Osmanisch-Türkische Grammatik. Wiesbaden: Otto Harrassowtiz.
- KIZILOV, Mikhail 2009. The Karaites of Galicia: An Ethnoreligious Minority Among the Ashkenazim, the Turks, and the Slavs, 1772–1945. Leiden: Brill.

- KIZILOV, Mikhail 2013. 'Critical Notes On Zsuzsanna Olach's study of The Halicz (Halich) Karaim Translation of Hebrew Biblical Texts'. In: T. T. ALLSEN, P. B. GOLDEN, R. K. KOVALEV, A. P. MARTINEZ (eds.) *Archivum Eurasiae Medii Aevi*. Vol. 20. Wiesbaden: Harrassowitz Verlag, 245–248.
- Kocaoğlu, Timur 2006. Karay: The Trakai Dialect. Muenchen: LINCOM GmbH.
- KOHLER Kaufmann and HARKAVY Abraham 1904. 'Karaites And Karaism'. In: Isidore SINGER et al. (eds.) *The Jewish Encyclopedia*. Vol. VII. New York, London: Funk and Wagnalls Company, 438–447.
- KOWALSKI, Tadeusz 1929. *Karaimische Texte im Dialekt von Troki*. Krakow: Polska Akademja Umiejętności.
- KRPS = BASKAKOV N. A., ZAJĄCZKOWSKI A., SHAPSHAL S. M. 1974. *Karaimsko-Russko-Pol'skij Slovar'*. Moskva: Russkij Jazyk.
- KRUS = Useinov, S. M. 2008. *Krymskotatarsko-Russko-Ukrainskiy Slovar'*. Aqmescit: Tezis Neshriyat Evi.
- KUZGUN, Şaban 2015. *Hazar ve Karay Türkleri*. İstanbul: Bilge Kültür Sanat.
- LET = REDHOUSE, J. W. 1884. A Lexicon, English and Turkish. Constantinople: A. H. Boyajian.
- LEWIS, Bernard 1956. 'A Karaite Itinerary Through Turkey in 1641–1642'. *Vakıflar Dergisi* 3: 315–327.
- LVTL = KOEHLER Ludwig and BAUMGARTNER Walter 1985. Lexicon in Veteris Testamenti Libros. Vol. 1–2. Leiden: E. J. Brill.
- MANN, Jacob 1935. *Texts and Studies in Jewish History and Literature*. Vol. II. Philadelphia: The Jewish Publication Society of America.
- MILLER, Philip 1993. Karaite Separatism in Nineteenth-Century Russia. Joseph Solomon Lutski's Epistle of Israeli's Deliverance. Cincinnati: Hebrew Union College Press.
- Musaevič 1964. *Grammatika karaimskogo jazyka. Fonetika i morfologija*. Moskva: Nauka.
- MUSAEV, Kenesbaj Musaevič 1977. Kratkiy grammaticheskiy ocherk karaimskogoyazyka. Moscow: Nauka.
- MUSAEV, Kenesbaj Musaevič 2002. 'Karaim Türkçesi' (translated by Mevlüt GÜLTEKİN). TÜBAR XI–Bahar, 197–230.
- MUSAEV, Kenesbaj Musaevič 2004. Sintaksis karaimskogo jazyka. Moskva: Nauka.

- NÉMETH, Michał 2006. 'Nieznane wiersze karaimskie Sergiusza Rudkowskiego(dialekt łuckohalicki). Edycja krytyczna' [Unpublished Master's Thesis]. Department of Turkish Philology, Jagiellonian University
- NÉMETH, Michał 2011a. Unknown Lutsk Karaim Letters in Hebrew Script (19th–20th Centuries): A Critical Edition. Krakow: Wydawnictwo Uniwersytetu Jagiellońskiego.
- NÉMETH, Michał 2011b. Zwięzła gramatyka języka zachodniokaraimskiego z ćwiczeniami. Poznań: Katedra Studiów Azjatyckich.
- NÉMETH, Michał 2014a. 'A Historical Phonology of Western Karaim. Alveolars And Front Labials in the South-Western Dialect'. *Studia Linguistica Universitatis Iagellonicae Cracoviensis* 131: 247–267.
- NÉMETH, Michał 2014b. 'A Historical Phonology of Western Karaim. The Evolution of Consonant Harmony in The North-Western Dialect'. *Studia Linguistica Universitatis Iagellonicae Cracoviensis* 131: 353–369.
- NÉMETH, Michał 2014c. 'An Early North-Western Karaim Bible Translation from 1720. Part 1. The Torah'. *Karaite Archives* 2: 109–141.
- NÉMETH, Michał 2015a. 'A historical phonology of Western Karaim. The process of its diversification into dialects'. *Studia Linguistica Universitatis Iagellonicae Cracoviensis* 132: 167–185.
- NÉMETH, Michał 2015b. 'An Early North-Western Karaim Bible Translation From 1720. Part 2. The Book Of Ruth'. *Karaite Archives* 3: 49–102.
- NÉMETH, Michał 2015c. 'An Early North-Western Karaim Bible Translation from 1720. Part 3. A contribution to the question of the stemma codicum of the Eupatorian print from 1841'. *Karaite Archives* 3: 103–118.
- NÉMETH, Michał 2015d. 'Nowy słownik języka krymskokaraimskiego Gulayhan Aqtay i Henryka Jankowskiego (2015) z dyskusją nad odrębnością tegojęzyka w klasyfikacji języków w tle'. *Almanach Karaimski* 4: 7–17.
- NÉMETH, Michał 2016. 'A Crimean Handwritten Translation of the Book of Ruth Dating from before 1678'. *Türk Dilleri Araştırmaları* 26 (2): 161–226.
- NEMOY, Leon 1950. 'Early Karaism (The Need for a New Approach)'. In: Abraham A. NEUMAN and Cyrus ADLER (eds.) *The Jewish Quarterly Review*. Vol. 40, No. 3. Philadelphia: University of Pennsylvania Press, 307–315.
- NEMOY, Leon 1952. *Karaite Anthology: Excerpts from the Early Literature*. New Haven: Yale University Press.

- NEMOY, Leon 1978. 'Karaites'. In: Emeri VAN DONZEL, Bernard LEWIS and Charles PELLAT (eds.) *The Encyclopaedia of Islam.* Vol. IV. Leiden–Boston: Brill, 603–608.
- OLACH, Zsuzsanna 2013. A Halich Karaim translation of Hebrew biblical texts. Wiesbaden: Harrassowitz Verlag.
- OLACH, Zsuzsanna 2016. 'A karaim nyelvű Gözlevei Biblia nyelvészeti tanulságai'. Keletkutatás ősz: 29–52.
- ÖTS = ÇAĞBAYIR, Yaşar 2007. Ötüken Türkçe Sözlük. Vol. 1–5. İstanbul: Ötüken Neşriyat.
- ÖZTÜRK, Abdulkadir 2019. Karay Türkçesinde İsim. Ankara: Fenomen Yayınları.
- PAWLINA, Agata 2016. 'The Pole Who Translated the Bible for the Turks'. *The Polish Journal of Biblical Research* 15/2: 31–37.
- POZNAŃSKI, Samuel 1909. 'Die karäische Literatur der letzten dreissig Jahre (1878–1908)'. Zeitschrift für hebräische Bibliographie 13: 110–118, 140–151, 180–181.
- POZNAŃSKI, Samuel 1910. 'Die karäische Literatur der letzten dreissig Jahre (1878–1908)'. Zeitschrift für hebräische Bibliographie 14: 57–61, 93–95, 112–115, 153–154.
- POZNAŃSKI, Samuel 1913. 'Karäisch-tatarische Literatur'. Keleti Szemle 13: 37-47.
- POZNAŃSKI, Samuel 1915. 'Karaites'. In: James HASTINGS (ed.) *The Encyclopaedia of Religion and Ethics*. Vol. VII. New York: Charles Scribner's Sons; Edinburgh: T&T Clark, 662–672.
- POZNAŃSKI, Samuel 1916. 'Karäische Kopisten und Besitzer von Handschriften'. Zeitschrift für Hebräische Bibliographie 19: 79–122.
- PRIK, Olga Ja 1976. *Očerk grammatiki karaimskogo jazyka (krymskij dialekt*). Makhachkala: Dagestanskoe Učebno-Pedagogičeskoe Izdatel'stvo.
- PRITSAK, Omeljan 1959. 'Das Karaimische'. In: Jean DENY et al. (eds.) *Philologiae Turcicae Fundamenta*. Wiesbaden: F. Steiner, 318–340.
- PRIVRATSKY, Bruce G. 2014. 'A History of Turkish Bible Translations'. [Online access: https://historyofturkishbible.wordpress.com; 05/10/2020].
- PROHOROV, Dmitriy Anatolyevich 2013. 'XIX. Yüzyılın İkinci Yarısı XX. Yüzyılın Başlarında Tavriya Vilayeti Karaim Türklerinin Etnik Ve Dil Durumu'. (Translated by Emine ATMACA and Reshide ADZHUMEROVA). *Gazi Türkiyat* Güz, 13: 209–214.
- RADLOFF, Wilhelm 1896. Proben der Volkslitteratur der Nördlichen Türkischen Stämme. Theil VII.Die Mundarten der Krym. St. Petersburg: Eggers et Co.
- RAGAGNIN, Elisabetta 2022. 'Azeri'. In: Lars JOHANSON and Éva Á. CSATÓ (eds.) *The Turkic Languages*. London and New York: Routledge, 242–261.

- RÓNA-TAS, András 2022a. 'The Reconstruction of Proto-Turkic and the Genealogical Question'. In: Lars JOHANSON and Éva Á. CSATÓ (eds.) *The Turkic Languages*. London and New York: Routledge, 60–74.
- RÓNA-TAS, András 2022b. 'Turkic Writing Systems'. In: Lars JOHANSON and Éva Á. CSATÓ (eds.) *The Turkic Languages*. London and New York: Routledge, 121–131.
- RYDING, C. Karin 2005. *A Reference Grammar of Modern Standard Arabic*. New York: Cambridge University Press.
- SAMOYLOVIČ, Aleksandr Nikolayevič 2000. 'O materialax Radlova po narodnoj slovesnosti krymskih tatar ikaraimov' (Re-edited from 1917). In: Aleksandr Nikolayevič SAMOYLOVIČ (ed.) *Izbrannye tudy o Kryme*. Simferopol: Dolja, 112–121.
- SCHÖNIG, Claus 2010. 'Osmanische Einflüsse auf das Krim-Areal'. In: Mancak-Wohlfeld, ELŻBİETA; Barbara PODOLAK (eds.) Studies on the Turkic world. A Festschrift for Professor Stanislaw Stachowski on the occasion of his 80th birthday. Krakow: Jagiellonian University Press, 107–119.
- SCHUR, Nathan 1992. History of the Karaites. Frankfurt and Main: Peter Lang.
- SHAPIRA, Dan 2001. A Karaite Poem in Crimean-Tatar from Mangup: A Source for Jewish-Turkish History (Judaeo-Turcica III)'. In: Mehmet TÜTÜNCÜ (ed.) Turkish-Jewish Encounters. Haarlem: Stichting SOTA, 79–98.
- SHAPIRA, Dan 2003. 'The Turkic Languages And Literatures of the east European Karaites'. In: Meira Polliack (ed.) *Karaite Judaism. A Guide to its History and Literary Sources*. Leiden, Boston: Brill, 657–707.
- SHAPIRA, Dan 2013. 'The Karaim translation of the Book of Nehemia copied in the 17th century's Crimea and printed in 1840/1841 at Gözleve, on the copyist of the manuscript, and some related issues'. *Karaite Archives* 1: 133–198.
- SHAPIRA, Dan. 2006. 'מסורות.' סיגול לשוני בתרגומי התנ"ך ללשונות תורכיות של הקראים. 13–14: 253–278.
- SKLARE David 2003. 'A Guide to Collections of Karaite Manuscripts'. In: Meira POLLIACK (ed.) *Karaite Judaism. A Guide to its History and Literary Sources*. Leiden, Boston: Brill, 893–924.
- SMĘTEK, Dorota 2012. 'Crimean Karaim Version Of Melukhat Sha'ul. Critical Edition and Linguistic Analysis'. [PhD Dissertation]. Poznań: Uniwersytet im. Adama Mickiewiczaw Poznaniu Wydział Neofilologii Katedra Studiów Azjatyckich Zakład Turkologii, Mongolistyki i Koreanistyki.

- SMĘTEK, Dorota 2015. 'Some lexical and phonological features of the language of a Crimean Karaim Bible translation'. In: *Türk Dilleri Araştırması Dergisi* 26 (2): 227–240.
- STEINSCHNEIDER, Moritz 1871. 'Karaitische Handschriften Anhang'. *Hebräische Bibliographie* 11: 37–46.
- SULIMOWICZ, Józef 1972. 'Materiał leksykalny krymskokaraimskiego zabytku językowego (druk z 1734 r.) I'. *Rocznik Orientalistyczny* 35, 1: 37–76.
- SZYSZMAN, Simon 1980. Le Karaïsme: Ses Doctrines et son Histoire. Lausanne: L'Age d'homme.
- TLO = MENINSKI, Franciscus à Mesgnien 1680. *Thesaurus Linguarum Orientalium Turcicae-Arabicae-Persicae: Lexicon*. Vol. I–IV. Viennæ.
- TOGAN, Zeki Velidi 1964. 'Hazarlar'. *İslam Ansiklopedisi*. Vol. 5. İstanbul: Milli Eğitim Bakanlığı Yayınları, 397–408.
- WAED = WORTABET William Thomson, PORTER Harvey, and WORTABET John 1984.

 Wortabet's Arabic-English Dictionary. Beirut: Librairie du Liban.
- WALFISH Barry D. and KIZILOV Mikhail 2011. *Bibliographia Karaitica: An Annotated Bibliography of Karaites and Karaism*. Leiden: Brill.
- WALFISH, Barry D. 2003. 'A Guide to Collections of Karaite Manuscripts'. In: Meira POLLIACK (ed.) *Karaite Judaism. A Guide to its History and Literary Sources*. Leiden, Boston: Brill, 925–959.
- WALTKE Bruce and O'CONNOR Micheal 1990. An Introduction to Biblical Hebrew Syntax.

 Winona Lake: Eisenbrauns.
- WENHAM, Gordon J. 1979. The Book of Leviticus. Michigan: Eerdmans.
- WEXLER, Paul 1983. 'Is Karaite a Jewish Language?'. *Mediterranean Language Review*. Vol 1: 27–54.
- YTL = BAHAEDDİN, M. 1927. Yeni Türkçe Lügat. İstanbul: Evkaf-ı İslamiye Matbaası.
- ZAJĄCZKOWSKI, Ananiasz 1926. 'Literatura Karaimska (Szkic bibliograficzny)'. *Myśl Karaimska*. Vol. 1, No. 3: 7–17.
- ZAJĄCZKOWSKI, Ananiasz 1932. Sufiksy imienne i czasownikowe w języku zachodniokaraimskim (przyczynek do morfologji języków tureckich). Les suffixes nominaux et verbaux dans la langue des Karaïms occidentaux (contribution à la morphologie des langues turques) [Prace Komisji Orjentalistycznej Polskiej Akademji Umiejętności 15]. Kraków: Nakładem Polskiej Akademii Umiejętności.

- ZAJĄCZKOWSKI, Ananiasz 1961. Karaims in Poland. History, Language, Folklore, Science. Warszawa: Państwowe Wydawnictwo Naukowe.
- ZAJĄCZKOWSKI, Ananiasz 1964. 'Die karaimische Literatur'. In: Luis BAZIN et al. (eds.) *Philologiae Turcicae Fundamenta*. Vol. 2. Wiesbaden: Franz Steiner, 793–801.
- ZAJĄCZKOWSKI, Ananiasz 1978. 'Karaites'. In: Emeri VAN DONZEL, Bernard Lewis and Charles Pellat (eds.) *The Encyclopaedia of Islam.* Vol. IV. Leiden–Boston: Brill, 608–609.

Appendix

Transcription

Preliminary Transcription Notes

The overarching strategies pertaining to our transcription have been extensively discussed earlier (refer to sections 1.1. and 2.1.1.). Consequently, this segment primarily enumerates supplementary notes that complement the previous explanations.

- * Given the text's use of Hebrew script, the exact value of vowels in non-Biblical words remains uncertain in certain vowel pairs *a-ä*, *o-ö*, *u-ü*, and *ï-i*. To navigate this ambiguity, we referred to the Crimean Karaim dictionary (CKED) for specific lexical items. Consequently, we did not apply certain Turkish phonotactics to Arabic and Persian words, adhering instead to the dictionary's illustrations, e.g., [Lev 19:17] *dušman* (CKED: 143) 'enemy' vs Trk *düšman* 'id'; [Lev 26:39] *günäh* (CKED: 169) 'sin' vs Trk *günah* 'id'.
- * Our transcription comprises only two punctuation marks. The first is a period (.) that demarcates the end of verses. The second is a short hyphen (-), which precedes the continuative syllables of a word that extends into the succeeding line. In select cases, certain case marking suffixes following a parenthesis also employ a short hyphen, e.g., *ada tavušan (kirpi) -nï* rabbit (hedgehog)+ACC. This notation also applies following the tetragrammaton YWY, e.g., [Lev 10:19] *YWY-nyŋ* Lord+GEN. We have faithfully replicated this approach in one instance where we found this sign in a folio, e.g., [Lev 26:30] *bama-lariŋizni*.
- * All proper nouns, instances of the tetragrammaton YWY and the term *Teŋri*; *Täŋri* denoting Jehovah, as well as all verse beginnings, are capitalized.
- * Although the text does not consistently link suffixes with the corresponding word stem attached, we have chosen not to represent when there is a separation, e.g., [Lev 9: 17] *ertä niŋ* > *ertäniŋ* 'morning+GEN'.

Editorial Notes

abc Karaim text

אבג text in Biblical Hebrew

[.] missing or illegible portion of text

[abc] reconstructed portion of text

(abc) deletion

[123] verse numbers added by the editor (abc) Karaim text written in parentheses

Labc¹¹²³ portion of text commented in a footnote

¹²³ן Hebrew portion of text commented in a footnote

Transcription of the Book of Leviticus from Gözleve Bible (1841)

93 rº

Chapter 1

- ייקרא 143 (1) עתרגום ספר וייקרא 143 (1) וַיִּקְרָא אֶל משֶׁה וַיִּדַדבֵּר יוָי אֵלָיו מֵאֹהֶל מוֹעֵד לַאמד 144 (2...)
- (...2) [1:1] Da čaķīrdī Mošegā ohel
- (3) mo'eddän da sözlädi YWY aŋar demä. דבר [1:2] Sözlägin oylanlarïna Yisra'elniŋ
- (4) da aytķīn alarya adam gendiņizdān egār yuvuķlaštīražaķ olsa ķorban YWY-ya
- (5) ol tuvardan ol sïyirdan ya ol koydan yuvuklaštïrïŋïz korbanïŋïznï. 🗅 ห
- (6) [1:3] Egär 'ola esä korbanï ol sïyïrdan saylam erkäkni alïp yuvuklaštïrsïn anï
- (7) ešiginä ohel moʻednin yuvuklaštirsin ani kabulluyina aldina YWY-nin. וסמך
- (8) [1:4] Da tayasın kolunı başı üstüna ol 'olanın da kabul olunı anar keparat etmägä
- (9) üstünä. ישׁהש [1:5] Da soysïn balasïn ol sïyïrnïn aldïna YWY-nïn
- (10) da yuvuklaštirsinlar oylanlari Aharonnin ol kohenlär ol kanni da sačsinlar ol
- (11) kannï ol mizbeah üstünä čüpčüvrä ki ešigindä ohel moʻednin. והפשיט
- (12) [1:6] Da terisin sïyïrsïn ol 'olanïŋ da buvumlasïn anï buvumlarïnǯa. ונתנו
- (13) [1:7] Da versinlär oylanlari Aharon[niŋ]¹⁴⁵ ol kohen ot ol mizbeaḥ üstünä da yaraštirsinlar
- (14) ayačlar ol ot üstünä. וערכו [1:8] Da yaraštirsinlar oylanlari Aharon[nin] ¹⁴⁶ ol
- (15) kohenlär ol buvumlarni ol baš bilän da ol ķarin yayını ol ayačlar üstünä

$93 v^{0}$

(1) ki ol ot üstünä ki ol mizbeah üstünä. וקרבו [1:9] Da karinini da ayaklarini

- (2) yuvsïn suvlar bilän da tütätsin ol kohen ol ǯümläni ol mizbeaḥta ʻoladïr ol
- (3) otlu korban ko[k]usï¹⁴⁷ kabulliknin YWY-ya. ואם [1:10] Da egär ol koydan esä korbani
- (4) ol koylardan ya ol ečkilärdän 'olaya saylam erkäkni alïp yuvuklaštïrsïn
- (5) anï. ושׁהט [1:11] Da soysïn anï uǯu yanïnda ol mizbeaḥnïn ki čafon tarafinda

¹⁴³ 'The translation of Book of Leviticus' in Biblical Hebrew.

¹⁴⁴ Lev 1:1 in Biblical Hebrew. It is also the beginning of a weekly Torah portion: Parashot Vayikra (Lev 1:1–5:26).

¹⁴⁵ JSul.IV.02A: *Aharon*; probably a scribal error. | BSMS 288: *Aharonnüŋ*. | JSul.III.01: *Aharonnun*. | A.Dub.III.73: *Aharonnun*.

¹⁴⁶ JSul.IV.02A: *Aharon*; probably a scribal error. | BSMS 288: *Aharonnüŋ*. | JSul.III.01: *Aharonnun*. | A.Dub.III.73: *Aharonnun*.

¹⁴⁷ JSul.IV.02A: kousi; a scribal error. | BSMS 288: isi. | JSul.III.01: iyisi. | A.Dub.III.73: iy.

- (6) aldına YWY-nın da sacsınlar oylanları Aharonnın ol kohenlar kanın ol mizbeah
- (7) üstünä čüpčüvrä. ונתח [1:12] Da buvumlasïn anï buvumlarïnǯa da bašïn da
- (8) ķarïn yayīnī yaraštīrsīn ol kohen alarnī ol ayačlar üstūnā ki ol ot
- (9) üstünä ki ol mizbeaḥ üstünä. והקרב [1:13] Da ol ķarïnnï da ol ayaķlarnï
- (10) yuvsin suvlar bilän da yuvuklaštirsin ol kohen ol jumläni da tütätsin ol
- (11) mizbeaḥta 'oladir ol otlu korban kokusi kabulliknin YWY-ya. ואם
- (12) אואם [1:14] Da egär ol kuštan esä 'ola korbanï YWY-ya da yuvuklaštïrsïn ol
- (13) kumrulardan ya o kögürčin balalarından korbanını. והקריבו [1:15] Da yuvuklaštırsın anı
- (14) ol kohen ol mizbeaḥķa da kessin bašīnī da tütātsin ol mizbeaḥta
- (15) da damlanïlsïn kanï duvarï üstünä ol mizbeaḥnïn. והסיר [1:16] Da ketärsin karnïnï
- (16) yünü bilän da tašlasın anı yanına ol mizbeahnın gün doyusı tarafka ol
- (17) kül tökilän yergä. וֹשׁׁסע [1:17] Da yïrtsïn anï k̞anatlarïndan (sïrtïndan)¹⁴⁸ ayïrmasïn
- (18) da tütätsin anï ol kohen ol mizbeaḥta ol ayačlar üstünä ki ol ot
- (19) üstünä 'oladir ol otlu korban kokusi kabulliknin YWY-ya.

- (20) (ביש (ב) ונפש [2:1] Da ǯan ki yuvuklaštirsa korban minha YWY-ya özek olsin
- (21) korbanï da töksin üstünä yay da versin üstünä tïmyan. והביאה
- (22) [2:2] Da getirsin anï oylanlarïna Aharonnïn ol kohenlärgä da avučlasïn andan dolu
- (23) avučīnī özāgindān da yayīndan ǯümlā tīmyanī bilān da tütātsin ol kohen tütidmāsini
- (24) ol mizbeaḥta otlu korban kokusi kabulliknin YWY-ya. והנותרת [2:3] Da ol
- (25) kalyannï ol minhadan Aharonya da oylanlarïnadïr kodeši kodešlärnin otlu
- (26) ķorbanlarindan YWY-nin. וכי [2:4] Da egär yuvuklaštirsan korban minha tandur piširmäsi

94 rº

- (1) özäk kalın mačalar karısılyanlar yay bilan da yufka mačalar silinganlar yay bilan. ואם
- (2) [2:5] Da egär minha ol tava ičindä esä korbanın özäk karısılyan yay bilän mača
- (3) bolsin. פתות [2:6] Tilimlämä tilimlägin ani tilimlär da koyyin üstünä yay
- (4) minḥadir ol. ואם [2:7] Da egär tänǯärädä pišen minḥasi esä korbanin özäk
- (5) yay bilän kilinsin. והבאת [2:8] Da ketirgin ol minhani bulardan YWY-ya

_

¹⁴⁸ JSul.IV.02A: Unvocalized text.

- (6) da yuvuklaštir[y]in¹⁴⁹ ani ol kohengä da yuvuklaštirsin ani ol mizbeahka. והרים
- (7) [2:9] Da ayirsin ol kohen ol minhadan tütsüsini da tütätsin ol mizbeahta otlu
- (8) korban kokusï kabullïknïŋ YWY-ya. והנותרת [2:10] Da ol kalyanï ol minhadan
- (9) Aharonya da oylanlarinadir kodeši kodešlärnin otlu korbanlarindan YWY-nin.
- (10) 55 [2:11] Här ol minha ki yuvuklaštirsaniz YWY-ya kilinmasin hameč mitlak
- (11) hič ačï ḥamur da hič bekmäz tütätmäniz andan otlu korban YWY-ya. קרבן
- (12) [2:12] Ilk korbanï yuvuklaštïrïŋïz alarnï YWY-ya da ol mizbeaḥka čïkmasïnlar
- (13) kokusïna kabullïknïŋ. וכל [2:13] Da ğümlä minha korbanïŋnï tuz bilän tuzlayïn
- (14) da eksitmegin šart tuzin Tenrinnin üstündän minhannin här korbanin üstünä
- (15) yuvuklaštiryin tuz. ואם [2:14] Da egär yuvuklaštirsan ilklär minhasini
- (16) YWY-ya arpa bašī čirkingān otķa taze arpa kürpāsi yuvuķlaštīryīn ilklārinnin
- (17) minḥasini. ונחתי [2:15] Da vergin üstünä yay da koyyin üstünä timyan minḥa-
- (18) dïr ol. והקטיר [2:16] Da tütätsin ol kohen tütsüsini kürpäsindän da yayından ğümlä
- (19) tïmyanï bilän otlu korban YWY-ya.

- (20) (גאם (ג) אם [3:1] Da egär šelamim korbanï esä korbanï egär ol sïyïrdan ol
- (21) yuvuklaštirir esä geräk erkäk geräk diši saylamni yuvuklaštirsin
- (22) anii aldina YWY-nin. וסמך [3:2] Da tayasin kolunii bašii üstünä korbaninin
- (23) da soysin ani ešiginda ohel moʻednin da sačsinlar oylanlari Aharonnin ol kohen-
- (24) lär ol kannï ol mizbeah üstünä čüpčüvrä. והקריב [3:3] Da yuvuklaštïrsïn
- (25) ol šelamim ķorbanindan otlu ķorban YWY-ya ol yayni ol yapķan ol ķarinni
- (26) da ǯümlä ol yaynï ki ol karïn üstünä. ואת [3:4] Da eki ol bögräklärni

94 v°

- (1) da ol yaynï ki üstlärindä ki ol emizlär üstünä da ol artïķsïnï ki
- (2) ol bayır üstünä ol bögräklär bilän ketärsin anı. והקטירו [3:5] Da tütätsinlär
- (3) anï oylanlarï Aharonnïŋ ol mizbeaḥta ol 'ola üstünä ki ol ayačlar üstünä
- (4) ki ol ot üstünä otlu korban kokusi kabulliknin YWY-ya. ואם [3:6] Da egär
- (5) ol koydan esä korbanï šelamim korbanïna YWY-ya erkäk ya tiši saylam-
- (6) nï yuvuklaštirsin anï. אַם [3:7] Egär kozu ol yuvuklaštirir esä korbanini

149 JSul.IV.02A: yuvuklaštirsin; a scribal error. | BSMS 288: yuvuklaštiryin. | JSul.III.01: yuvutsun. | A.Dub.III.73:

yuvutsun.

sin şin. | 35di.111.01. yavaısan. | 11.12d0.111.73.

- (7) da yuvuklaštirsin ani aldina YWY-niŋ. וסמך [3:8] Da tayasin koluni baši üstünä
- (8) ķorbaninin da soysin ani aldīna ohel moʻednin da sačsinlar oylanlari Aharonnin
- (9) kanïnï ol mizbeah üstünä čüpčüvrä. והקריב [3:9] Da yuvuklaštïrsïn ol šelamim
- (10) korbanından otlu korban YWY-ya tamam ol kuyruk yayını karsısına ol yumulyannın
- (11) ketärsin ani da ol yayni ol yapkan ol karinni da žümlä ol yayni ki ol
- (12) karın üstünä. ואת [3:10] Da eki ol bögräklärni da ol yaynı ki üstlärinä
- (13) ki ol emizlär üstünä da artïķsïnï ol bayïr üstünä ol bögräklär bilän
- (14) ketärsin anï. והקטירו [3:11] Da tütätsin anï ol kohen ol mizbeaḥta otlu
- (15) korban YWY-ya. ואס [3:12] Da egär ečki esä korbanï da yuvuklaštirsin ani aldina
- (16) YWY-nïŋ. וסמך [3:13] Da tayasïn kolunï bašï üstünä da soysïn anï aldïna ohel mo'ed-
- (17) niŋ da sačsïnlar oylanlari Aharonniŋ kanini ol mizbeah üstünä čüpčüvrä.
- (18) והקריב [3:14] Da yuvuklaštirsin andan korbanini otlu korban YWY-ya ol yayni ol
- (19) yapkan ol karïnnï da ǯümlä ol yaynï ki ol karïn üstünä. ואת [3:15] Da
- (20) eki ol bögräklärni da ol yayni ki üstlärinä ki ol emizlär üstünä
- (21) da artiķsīnī ki ol bayīr üstünā ol bögrāklār bilān ketārsin anī.
- (22) והקטירם [3:16] Da tütätsin alarni ol kohen ol mizbeahta otlu korban kokusina
- (23) kabullıknın ğümla yaynı YWY-ya. חקת [3:17] Ömürlik rasim davürlarinizga ğümla
- (24) oturašlarinizda hič yayni da hič ķanni ašamaniz.

- (25) (ז'דבר (ד) אידבר [4:1] Da sözlädi YWY Mošegä demä. דבר [4:2] Sözlägin oylanlarina
- (26) Yisra'elnin demä ǯan ki yazīķlī bolsa yanlīšlīķ bilān ǯümlā mičva-

95 ro

- (1) larından YWY-nın ki kılınmaslardır da kılsa birsini birdan alardan. אם [4:3] Egär
- (2) ol silingän kohen yazıklı bolsa günähi učun ol ulusnın da yuvuklastırsın
- (3) yazïyï učun ki yazïkli boldi buya balasin siyirnin saylamni YWY-ya hatatka.
- (4) והביא [4:4] Da ketirsin ol buyanï äšiginä ohel moʻednin aldïna YWY-nïn da tayasïn
- (5) kolunï bašï üstünä ol buyanïn da soysin ol buyanï aldïna YWY-nïn. ולקח
- (6) [4:5] Da alsïn ol silingän kohen kanïndan ol buyanïn da girgizsin anï ohel moʻedgä.
- (7) וטבל [4:6] Da mänčsin ol kohen barmayïnï ol kanya da sačsïn ol kandan yedi kerätlär
- (8) aldına YWY-nın aldına pardasının ol kodešnin. ונהן [4:7] Da versin ol kohen

- (9) ol kandan kovuš¹⁵⁰ köšeläri üstünä ol otyamlar tütsüsi mizbeahnin aldina
- (10) YWY-nin ki ohel mo'eddä da ğümlä kanin ol buyanin töksin temälinä mizbeah-
- (11) hïnïŋ ol 'olanïŋ ki ešigindä ohel mo'edniŋ. ואת [4:8] Da ǯümlä yayïn
- (12) ol ḥatat buyasinin ayirsin andan ol yayni ol yapkan ol karin
- (13) üstünä da ğümlä ol yayni ki ol karın üstünä. ואת [4:9] Da eki ol
- (14) bögräklärni da ol yaynï ki üstlärinä ki ol emizlär üstünä da ol
- (15) artiķsini ki ol bayir üstünä ol bögräklär bilän ketärsin ani. כאשר
- (16) [4:10] Nečiki ayïrïlïr ögüzindän ol šelamim korbanïnïn da tütätsin alarnï ol
- (17) kohen mizbeaḥï üstünä ol 'olanïŋ. ואת [4:11] Da terisin ol buyanïŋ da ǯümlä
- (18) etin bašin ayaklari bilan da karinin da tezagin. והוציא [4:12] Da čiyarsin takmil
- (19) ol buyanî avuldan čîḥarî temiz yergä ol kül tökän yergä da küydürsin anî
- (20) ayačlar üstünä otka ol kül töken yerdä küydürülsin. ואם [4:13] Da egär
- (21) ǯümlä ǯamaʻatï Yisra'elnin yanïlsalar da yašïrïn bolsa bir šey gözlärindän ol
- (22) kahalnın da kilsalar birni ğümlä mičvalarından YWY-nın ki kilinmaslardır
- (23) da günähli olsalar. ונודעה [4:14] Da bilinsä ol yazik ki yazik[li]¹⁵¹ boldilar anin učun
- (24) da yuvuklaštīrsīnlar ol kahal buya sīyīr balasīnī hatatka da getirsinlār
- (25) ani aldına ohel mo'ednin. וסמכו [4:15] Da tayasınlar kartları ol ğama'atnın kollarını
- (26) bašī üstünä ol buyanīŋ aldīna YWY-nīŋ da soysīn ol buyanī aldīna YWY-nīŋ.

$95 v^{o}$

- (1) והביאו [4:16] Da girgizsin ol silingän kohen kanindan ol buyanin ohel mo'edgä.
- (2) נטבל [4:17] Da mänčsin ol kohen barmayïnï ol kandan da sačsïn yedi kerätlär aldïna
- (3) YWY-nin aldina ol perdänin. ומן [4:18] Da ol kandan versin bučkaklari üstünä
- (4) ol mizbeahnin ki aldina YWY-nin ki ohel mo'eddä da ğümlä ol kanni
- (5) töksin binyatïna mizbeaḥïnïn ol 'olanïn ki ešigindä ohel mo'ednin. ואת
- (6) [4:19] Da ǯümlä yayïnï ayïrsïn andan da tütätsin ol mizbeaḥta. ועשה [4:20] Da ķilsïn
- (7) ol buyaya nečiki ķildi buyasina ol hatatnin alay ķilsin anar da keparat
- (8) itsin üstlärinä ol kohen da bošatïlïr alarya. והוציא [4:21] Da čïyarsïn ol buyanï
- (9) avuldan čī[h]arī¹⁵² da küydürsin anī nečiki küydürdi ol burunyī buyanī ḥatatī ol ķahal-
- (10) nı̈ndı̈r ol. אשׁר [4:22] Egär aya yazı̈klı̈ bolsa da kı̈lsa birni ǯümlä mičvaları̈ndan YWY
- (11) Täŋrisinin ki kilinmaslardir yanlišlik bilan da günähli bolsa. או [4:23] Ya

¹⁵⁰ JSul.IV.02A: kovš; an orthographical tendency (see 2.1.1.2.2.).

¹⁵¹ JSul.IV.02A: yazik; a scribal error. | BSMS 288: yazikli. | JSul.III.01: yazikli. | A.Dub.III.73: yazikli.

¹⁵² JSul.IV.02A: čikari/čikäri; a scribal error. | BSMS 288: čihari. | JSul.III.01: tiskartin. | A.Dub.III.73: tiškartin.

- (12) bilinsä anar yazïyï ki yazïklï boldï anïn učun da getirsin korbanïnï ulayïn
- (13) ečkilärnin saylam erkäk. וסמך [4:24] Da tayasïn kolunï bašï üstünä ol ulaknïn
- (14) da soysin ani ne yerdä ki soysa ol 'olani aldina YWY-nin hatatdir ol.
- (15) ולקח [4:25] Da alsïn ol kohen kanından ol hatatnın barmayı bilan da versin kovuš¹⁵³
- (16) köšeläri üstünä mizbeahinin ol 'olanin da kanini töksin temälinä mizbeahinin
- (17) ol 'olanın. ואת [4:26] Da ğümlä yayını tütätsin ol mizbeahta yayı gibi
- (18) korbanïnïn ol šelamimlärnin keparat etsin üstünä ol kohen yazïyïndan
- (19) da bošatïlïr aŋar. ואם [4:27] Da egär bir ǯan yazïklï bolsa yaŋlïšlïk bilän ulusïndan
- (20) ol yernin külyanında birni mičvalarından YWY-nın ki külinmaslardır da günühli
- (21) bolsa. או [4:28] Ya bilinsä anar yazïyï ki yazïklï boldï da getirsin korbanïnï
- (22) ulayın ečkilarnin saylamnı yazıyı učun ki yazıklı boldı. וסמך [4:29] Da tayasın
- (23) ķolunī bašī üstünā ol ḥatatnīŋ da soysīn ol ḥatatnī yerindā ol
- (24) 'olanın. ולקח [4:30] Da alsın ol kohen kanından barmayı bilan da versin kovuš¹⁵⁴ köseläri
- (25) üstünä mizbeaḥīnīŋ ol 'olanīŋ da ǯümlä kanīnī töksin temelinä ol
- (26) mizbeaḥnïŋ. ואת [4:31] Da ǯümlä yayïnï ketärsin nečiki ketärildi yay üstündän

96 ro

- (1) korbanïnïn ol šelamimlärnin da tütätsin ol kohen mizbeahta kokusïna
- (2) kabullïknïn da keparat etsin üstünä ol kohen da bošatïlïr anar. ואם
- (3) [4:32] Da egär kozu getirsä korbanïnï hatatka saylam tišini getirsin anï. זסמך
- (4) [4:33] Da tayasın kolu[n]i¹⁵⁵ bašı üstünä ol hatatnın da soysın anı hatatka ne
- (5) yerdä ki soyar ol 'olanï. ולקח [4:34] Da alsïn ol kohen kanïndan ol hatatnïn
- (6) barmayı bilan da versin bučkakları üstünä mizbeahının ol 'olanın da ğümlä
- (7) kanını töksin binyatına ol mizbeahnın. ואת [4:35] Da ğümla yayını ketarsin
- (8) nečiki ketärilir yayï ol koynïn korbanïndan ol šelamimlärnin da tütätsin ol
- (9) kohen alarni ol mizbeaḥta otlu korbanlari üstünä YWY-nin da kapara etsin
- (10) üstünä ol kohen yaziyi učun ki yazikli boldi da bošatilir aŋar.

Chapter 5

(11) (הופשׁ (ה) ונפשׁ [5:1] Da ǯan ki yazïklï bolsa da ešitsä avazïn antnïn da ol šaḥad

¹⁵⁴ JSul.IV.02A: kovš; an orthographical tendency (see 2.1.1.2.2.).

¹⁵³ JSul.IV.02A: *kovš*; an orthographical tendency (see 2.1.1.2.2.).

¹⁵⁵ JSul.IV.02A: koluni; a scribal error. | BSMS 288: kolun. | JSul.III.01: kolun. | ADub.III.73: kolun.

- ya gördi ya bildi egär anlatmasa da kötärsin günähin. או
- (13) [5:2] Ya ǯan ki tiysä ǯümlä mundar nemägä ya gövdäsinä mundar kiyiknin ya
- (14) gövdäsinä mundar tuvarnin ya gövdäsi[n]ä¹⁵⁶ mundar kozlayannin da yaširin bolsa
- (15) andan da ol mundar boldi da günähli boldi. או [5:3] Ya ki tiysä mundarliyina
- (16) adamnın ğümla mundarlışın ki mundar bolir anın bilan da yasırın
- (17) bolsa andan da ol bildi da günähli boldi. או [5:4] Ya ǯan ki ant etsä
- (18) sözlämä erinlär bilän yaman etmä ya yaḥšï etmä ǯümlägä ki sözlär ol adam
- (19) ant bilän da yašïrïn bolsa andan da ol bildi da günähli boldï birgä bulardan.
- (20) והיה [5:5] Da bolyay ki pešman bolsa birgä bulardan da iķrar etsä ki yaziķlï
- (21) boldï anïŋ učun. והכיא [5:6] Da getirsin pešmanlïyïnï YWY-ya yazïyï učun ki
- (22) yazıklı boldı tišini ol koydan kozu ya ulayın ečkilarnin hatatka
- (23) da keparat etsin üstünä ol kohen yaziyindan. ואם [5:7] Da egär yetmäsä kuvati koyya
- (24) da getirsin pešmanlīyīnī ki yazīķlī boldī eki ķumrular ya eki balalarīn
- (25) gögürčünnin YWY-ya birni hatatka da birni 'olaya. והביא [5:8] Da ketirsin
- (26) alarni ol kohenga da yuvuklaštirsin birni ki hatatnin burun da melika

$96 v^{o}$

- (1) etsin bašini karšisindan ensäsinin da ayirmasin. הזה [5:9] Da sačsin kanindan ol
- (2) ḥatatnin duvari üstünä ol mizbeaḥnin ol kalyan ol kandan tamsin binyatina
- (3) ol mizbeahnïn hatatdïr ol. ואת [5:10] Da ol ekinǯini kïlsïn 'ola töräčä
- (4) da kapara etsin üstünä ol kohen yazïyïndan ki yazïklï boldï da bošatïlïr
- (5) aŋar. וֹאם [5:11] Da egär kuvatï yetmäsä eki kum[ru]larɣa¹⁵⁷ ya eki balalarïna gögürčünniŋ
- (6) da getirsin korbanını ki yazıklı boldı onunğı payın ol kebičnin özäk hatat-
- (7) ka koymasin üstünä yay da vermäsin üstgä timyan zira hatatdir ol. והביאה
- (8) [5:12] Da getirsin anï ol kohengä da avučlasïn ol kohen andan dolu avuču¹⁵⁸ bilän tütsüsi-
- (9) ni da tütätsin ol mizbeaḥta utlu korbanlari üstünä YWY-niŋ ḥatatdir
- (10) ol. וכפר [5:13] Da kapara etsin üstünä ol kohen yaziyi učun ki yazikli boldi

_

¹⁵⁶ JSul.IV.02A: *gövdäsindä*; probably a scribal error. | BSMS 288: *gövdäsinä*. | JSul.III.01: *gevdesine*. | ADub.III.73: *gövdasińa*.

¹⁵⁷ JSul.IV.02A: kumlarya; a scribal error. | BSMS 288: torlarya. | JSul.III.01: bedenelerge. | ADub.III.73: bedenálarga.

¹⁵⁸ JSul.IV.02A: Spelled אבוצו; a scribal error.

- (11) birsinden bulardan da bošatīlīr aņar da bolsīn ol kohengā minḥa gibi. וידבר
- (12) [5:14] Da sözlädi YWY Mošegä demä. נפֿש [5:15] Šan ki tansa tanmak da yazıklı bolsa
- (13) yanlığlık bilan kodeşlarından YWY-nın da getirsin peşmanlığını YWY-ya saylam
- (14) kočķar ol koydan ķīymātin bilān gümüš miskallar miskalī bilān ol kodešnin
- (15) ašamya. ואת [5:16] Da neni ki yazikli boldi ol kodešdän ödesin da bešinži
- (16) payïnï arttïrsïn üstünä da versin anï kohengä da ol kohen kapara etsin
- (17) üstünä kočkari bilan ol ašamnin da bošatilir anar. ואם [5:17] Da egär ǯan ki
- (18) yazıklı bolsa da kilsa birni ğümlä mičvalarından YWY-nın ki kilinmaslardır
- (19) da bilmädi günähli boldï da čeksin žürümin. והביא [5:18] Da getirsin kočkar saylam
- (20) ol koydan kiyasin bilan ašamya ol kohenga da kapara etsin üstünä ol
- (21) kohen yanlıslıyı učun ki yanıldı da ol bilmadi da bosatılır anar. אשׁם
- (22) [5:19] Ašamdīr ol günäḥli bolma günäḥli boldī YWY-ya. וידבר [5:20] Da sözlädi YWY
- (23) Mošegä demä. נפשׁ [5:21] Šan ki yazïklï bolsa da tansa tanmak YWY-ya da tansa
- (24) dostïna amanatķa ya ķol bilän ķoyyan[y]a¹⁵⁹ ya zülüm bilän ya dutup aldï isä
- (25) dostindan. או [5:22] Ya tapsa tas bolyanni da tansa anar da ant etsä yalyanya
- (26) bir učun ǯümlädän ki ķïlar ol adam yaziķlï bolma alar bilän. והיה

97 ro

- (1) [5:23] Da bolyay ki yazıklı bolsa da pešman bolsa da kaytarsın ol zülümni ki zülümlädi
- (2) ya ol dutup almakni ki dutup aldi ya ol amanatni ki amanat koyuldi
- (3) birgäsinä ya ol tas bolyannï ki taptï. או [5:24] Ya ǯümlädän ki ant etsä
- (4) anın učun yalyanya da ödäsin anı başı bilan da beşin isin arttırsın
- (5) üstünä aŋar ki ol anïŋ versin anï pešman bolyan günündä. ואת [5:25] Da
- (6) pešmanliyini getirsin YWY-ya saylam kočkar ol koydan kiyasin bilan ašam-
- (7) ya ol kohengä. וכפר [5:26] Da kapara etsin üstünä ol kohen aldïna
- (8) YWY-nın da boşatılır anar birsi učun ğümladan ki kılar günahli
- (9) bolsa anïη bilän.

Chapter 6

-

¹⁵⁹ JSul.IV.02A: koyyanna; a scribal error. | BSMS 288: koyyanya. | JSul.III.01: koymayinda. | ADub.III.73: koymayinda.

- פרשת צו¹⁶⁰ן (10)
- 161 וידבּר ינַי אֱל משֵׁה לַאמר (11)
- (12) [6:1] Da sözlädi YWY Mošegä demä.
- (13) צו [6:2] Sïmarlayïn Aharonya da oylanlarïna demä budïr torasï
- (14) ol 'olanın oldır ol 'ola očayı üstünä ki ol mizbeah üstünä ğümlä
- (15) ol gečä ol ertäyä degin da otu ol mizbeahnin yansin anda. ולבשׁ
- (16) [6:3] Da giysin ol kohen kisäyi uprak da kisäyi könčäk kiysin eti üstünä da ayïrsïn
- (17) ol külni ki örtär ol ot ol 'olanï ol mizbeaḥ üstünä da koysïn anï
- (18) yanına ol mizbeahnın. ופּשׁט [6:4] Da češsin upraklarını da giysin öŋgä upraklar
- (19) da čīķarsīn ol külni avuldan čīḥarī aruv yergä. והאש [6:5] Da ol ot ol mizbeaḥ
- (20) üstünä yansin anda sönmäsin da yandirsin üstünä ol kohen ayačlar ertä
- (21) bilän ertä bilän da yaraštïrsïn üstünä ol 'olanï da tütätsin üstünä
- (22) yayların ol šelamimlarnin. אט [6:6] Dayım ateš yansın ol mizbeah üstünä
- (23) sönmäsin. וזאת [6:7] Da budir torasi ol minhanin yuvuklaštirma
- (24) yuvuklaštirsin ani birsi oylanlarinin Aharonnin aldina YWY-nin aldina
- (25) ol mizbeaḥnin. והרים [6:8] Da ayirsin andan avuči bilan özägindän ol minhanin

$97 v^{0}$

- (1) da yayından da ğümlä ol timyanni ki ol minha üstünä da tütätsin ol
- (2) mizbeahta kokusï kabullïknïn tütsüsi YWY-ya. והנותרת [6:9] Da ol kalyannï andan
- (3) ašasïnlar Aharon da oylanlari mačalar ašalsin ayruhsi yerdä azbarinda ohel moʻed-
- (4) nin ašasinlar ani. אל [6:10] Pišmäsin ḥameč paylarini verdim ani otlu
- (5) korbanlarımdan kodeši kodešlärnindir ol hatat gibi da ašam gibi. כל
- (6) [6:11] Šümlä erkäk ki oγlanlarinda Aharonnin ašasin ani räsimi dunyanin dävürlärinizgä
- (7) otlu korbanlarından YWY-nın har kim ki tiysa alarya ayruhsı bolsın. וידבר
- (8) [6:12] Da sözlädi YWY Mošegä demä. זה [6:13] Budïr korbanï Aharonnïŋ da oylanlarïnïŋ
- (9) ki yavuklaštīryaylar YWY-ya silingān gündān soņra ki silsā anī onunžī
- (10) payï ol kebičnin özek minha dayïm yarïmïsïn ertädä da yarïmïsïn ahšamda.
- על [6:14] Tava üčünä yay bilän kilinsin kavurulyanni getirtgin ani (pišmäklärin)

¹⁶⁰ The weekly Torah portion: Parashot Tzav (Lev 6:1–8:36).

¹⁶¹ Lev 6:1 in Biblical Hebrew.

- (12) yanaštiryin dilimlar minhasini yuvuklaštiryin kokusina kabulliknin YWY-ya.
- (13) קו [6:15] Da ol silingän kohen ornina oylanlarindan kilsin ani räsimi dunyanin YWY-
- (14) ya täkmil tütätilsin. וכל [6:16] Da här minhasï kohennin täkmil atešä
- (15) bolsïn ašalmasïn. וידבר [6:17] Da sözlädi YWY Mošegä demä. דבר [6:18] Sözlägin
- (16) Aharonya da oylanlarina demä budir torasi ol hatatnin ne yerdä ki
- (17) soyulsa ol 'ola soyulsin ol hatat aldina YWY-nin kodeši kodešlärnin-
- (18) dir ol. הכהן [6:19] Ol kohen ol ḥatat etkän anï ašasïn anï ayruḥsï yerdä
- (19) ašalsïn azbarïnda ohel moʻednin. כל [6:20] Här ne ki tiysä etinä ayruhsï
- (20) bolsïn da egär sačilsa kanindan ol uprak üstünä ki sačilsa üstünä
- (21) yuvyïn ayruḥsï yerdä. וכל [6:21] Da čöräp sayït ki pišsä ičindä sïndïrïlsïn
- (22) da egär bakir sayitda pišsä da kirilsin da yaykalsin suvlar bilän. כל [6:22] Här
- (23) erkäk kohenlärdä ašasïn anï kodeši kodešlärnindir ol. וכל [6:23] Da her hatat
- (24) ki girgizilsä kanindan ohel mo'edgä keparat etmägä kodešdä
- (25) ašalmasïn otka küydürülsin.

(26) (ז) וואת (7:1] Da budir torasi ol ašamnin kodeši kodešlärnindir ol.

98 rº

- (1) במקום [7:2] Ne yerdä ki soysalar ol 'olanï soysïnlar ol ašamnï da
- (2) kanïnï sačsïn ol mizbeah üstünä čüpčüvrä. ואת [7:3] Da ǯümlä yayïnï
- (3) yuvuklaštirsin andan ol kuyrukni da ol yayni ol yapkan ol karinni. ואת [7:4] Da
- (4) eki ol bögräklärni da ol yaynï ki üstlärinä ki ol emizlär üstünä
- (5) da ol artīķsini ol bayir üstünä ol bögräklär bilän ketärsin ani. והקטיר
- (6) [7:5] Da tütätsin alarni ol kohen ol mizbeahta otlu korban YWY-ya ašamdir
- (7) ol. כל [7:6] Här erkäk kohenlärdä ašasïn anï ayruhsï yerdä ašalsïn kodeši
- (8) kodešlärnindir ol. כחטאת [7:7] Nečiki hatat alay ašam bir tora alarya
- (9) ol kohen ki kapara etsä anïŋ bilän aŋar bolsïn. והכהן [7:8] Da ol kohen ol
- (10) yuvuklaštiryan kišinin 'olasini terisi ol 'olanin o kohengä ki
- (11) yuvuklaštirdi ani anar bolsin. וכל [7:9] Da här minha ki pišsä tandurda da här kilinyan
- (12) tenţärädä ya tava ičindä ol kohengä ol yuvuklaštiryan ani anar bolsin.
- (13) וכל [7:10] Da här minha karïšïlyan yay bilän ya kuru ğümlä oylanlarïna Aharon-
- (14) nïn bolsïn här kišigä kardašï gibi. וזאת [7:11] Da budïr torasï ol šelamim

- (15) korbaninin ki yuvuklaštirsa YWY-ya. אם [7:12] Egär šükürlik učun yuvuklaštirsa
- (16) anï da yuvuklaštirsin ol šükürlik korbani bilan kalin mačalar karišilyanlar
- (17) yay bilän da yufka mačalar silingänlär yay bilän da kavurulyan özägi kalïnlar
- (18) karïšïlyanlar yay bilän. על [7:13] Ḥameč kalïn ötmäklär bilän yuvuklaštïrsïn korbanïnï
- (19) šükürlik šelamim ķorbanïnïŋ yanïna. והקריב [7:14] Da yuvuklaštïrsïn andan birer
- (20) ekmäk här bir soy korban ekmägindän ayïrmak YWY-ya ol kohengä ol sačkan
- (21) ol šelamim kanini anar bolsin. ובשר [7:15] Da eti šükürlik korbaninin šelamimlärinin
- (22) korban etkän günü[n]dä¹⁶² ašalsin konmasin andan ertägä degin. ואם [7:16] Da egär
- (23) inyät ya ǯomartlïķ esä šelamim ķorbanï yuvuķlaštïrdïyï gündä ķorbanïnï
- (24) ašalsin da sabaḥinda da ol kalyan andan ašalsin. והנותר [7:17] Da ol kalyan ol
- (25) šelamim etindän ol üčünži gündä otka küydürilsin. ואם [7:18] Da egär
- (26) ašalma ašalsa etindän šelamim ķorbanīnīn ol üčünži gündā maķbul bolmastīr

98 vo

- (1) ol yuvuklaštīryan anī ķorban sayīlmaz anar bayat (pasul¹⁶³) bolīr da ol ǯan
- (2) ol ašayan andan günähini čekär. והבשׁר [7:19] Da ol et ki tiysä birsinä här dürli
- (3) mundar šeynin ašalmasin otķa küydürilsin da ol ķodeš eti här temiz
- (4) kiši ašasin kodeš etini. והנפש [7:20] Da ol žan ki ašasa et ol šelamim
- (5) korbanından ki YWY-ya da mundarlıyı olsa üzürindü da kesilir ol ğan oluslarından.
- (6) נופּשׁ [7:21] Da ǯan ki tiysä ǯümlä mundarya mundarlïyïna adamnïŋ ya mundar
- (7) tuvarya ya ğümlä mundar iränčigä da gizli ašasa etindän ol šelamim
- (8) korbanının ki YWY-ya kesilir ol ğan oluslarından. וירבר [7:22] Da sözlädi YWY
- (9) Mošegä demä. דבר [7:23] Sözlägin oylanlarina Yisra'elnin demä hič yayin ögüznin
- (10) ne koynin ne ečkinin ašamaniz. והלב [7:24] Da yayi nevelanin da yayi terefanin
- (11) ķīlīnsīn hār iškā anča[k̞] ¹⁶⁴ ašama ašamaŋīz anī. כֹי [7:25] Zira hār ašayan kimsä
- (12) yay ol tuvardan ki yuvuklaštirir andan otlu korban YWY-ya kesilir ol
- (13) gizli ašayan ǯan uluslarindan. וכל [7:26] Da hič kanni ašamaniz ǯümlä
- (14) oturašlarinizda ne ķuštan ne da tuvardan. כל [7:27] Žümlä žan ki ašasa hič
- (15) kan da kesilir ol ğan uluslarından. וידבר [7:28] Da sözlädi YWY Mošegä demä.
- (16) דבר [7:29] Sözlägin oylanlarïna Yisra'elniŋ ol yuvuklaštïryan šelamim korbanïnï YWY-

-

¹⁶² JSul.IV.02A: günündä; a scribal error. | BSMS 288: künindä. | JSul.III.01: küninde. | ADub.III.73: kününda.

¹⁶³ JSul.IV.02A: A partially unvocalized text.

¹⁶⁴ JSul.IV.02A: anča; a scribal error. | BSMS 288: da. | JSul.III.01: vale. | ADub.III.73: da.

- (17) ya getirsin korbanïnï YWY-ya korbanïndan šelamimlärnin. ידיו [7:30] Kollarï getirsinlär
- (18) otlu ķorbanların YWY-nın ol yaynı ol toš bilan getirsin anı ol
- (19) töšni sallamaya anï önünä sallamak aldına YWY-nın. והקטיר [7:31] Da tütätsin
- (20) ol kohen ol yaynï ol mizbeaḥta da bolsïn ol töš Aharonya da oylanlarïna. ואת
- (21) [7:32] Da ol oŋ butnï veriŋiz kötärip sallamak ol kohengä korbanlarïndan šelamimleriŋizniŋ. המקריב
- (22) [7:33] Ol yuvuklaštiryan ol šelamim kanini da ol yayni oylanlarindan Aharonnin
- (23) aŋar bolsïn ol oŋ but payya. כי [7:34] Zira ol öŋünä sallamak töšini da
- (24) ol yokarı sallamak butunı aldım yanından oylanlarının Yısra'elnin šelamim
- (25) korbanlarından da verdim alarnı Aharon ol kohenga da oylanlarına rasimina dunyanın
- (26) yanından oylanlarının Yisra'elnin. זאת [7:35] Budır büyüklik payı Aharonnın da büyüklik

99 rº

- (1) payï oylanlarïnin otlu korbanlarindan YWY-nin yuvuklaštiryan gündän alarni
- (2) kohenlik etmägä YWY-ya. אשׁר [7:36] Ki sïmarladï YWY vermä alarya sildigi
- (3) gündän alarni yanindan oylanlarinin Yisra'elnin ömirlik räsimi dävürlärinä.
- (4) זאת [7:37] Budïr ol tora 'olaya minḥaya da ḥatatka da ašamya
- (5) da milu'imgä da ol šelamim korbanïna. אשׁר [7:38] Ki sïmarladï YWY Mošegä
- (6) tayında Sinaynın sımarladıyı günda oylanlarına Yısra'elnin yuvuklastırmaya
- (7) ķorbanlarini YWY-ya yabaninda Sinaynin.

Chapter 8

- (8) (הידבר (ה) פוידבר [8:1] Da sözlädi YWY Mošegä demä. קה [8:2] Alyïn Aharonnï da
- (9) oylanlarını birgasina da ol upraklarnı da ol silmak yayını
- (10) da ol ḥatat buyasïnï da eki ol ķočķarlarnï da ol mačalar selläsini.
- (11) ואת [8:3] Da ǯümlä ol ǯamaʻatnï toplayïn ešiginä ohel moʻednin. ויעש
- (12) [8:4] Da ķīldī Moše nečiki sīmarladī YWY aŋar da toplandī ol ǯamaʿat ešiginā ohel
- (13) mo'ednin. ויאמר [8:5] Da ayttï Moše ol žama'atka budir ol söz ki sïmarladï
- (14) YWY kïlmaya. ויקרב [8:6] Da yuvuklaštïrdï Moše Aharonnï da oylanlarïnï da yuvdï
- (15) alarnı suvlar bilan. ויתן [8:7] Da verdi üstünä ol gölmäkni da kušadı anı
- (16) ïnčķīr bilān da giydirdi aŋar ol šušmannī da verdi üstünā ol efodnī

- (17)da kušadī anī bayī bilān ol efodnīŋ da bayladī anī anīŋ bilān. רישם [8:8] Da koydī
- (18)üstünä ol hošenni da verdi ol hošengä ol Urimni da ol Tumimni.
- וישם [8:9] Da koydï ol sarïknï bašï üstünä da koydï ol sarïk üstünä (19)
- (20)karšīsīna yüzlārinin ol altīn čičekni ol kodeš tažīnī nečiki sīmarladī YWY
- Mošegä. ויקח [8:10] Da aldï Moše ol silmäk yayïnï da sildi ol miškannï (21)
- da ǯümläni ki anda da ayruhsï etti alarnï. ריז [8:11] Da sačtï andan ol (22)
- mizbeah üstünä yedi kerätlär da sildi ol mizbeahnï da ǯümlä ta[k]ïmlarïnï¹⁶⁵ (23)
- (24)da ol kumyanni da tasläkänini ayruhsi etmägä alarni. ויצק [8:12] Da tökti yayindan
- (25)ol silmäknin bašī üstünä Aharonnin da sildi ani ayruhsi etmägä ani.
- (26)זיקרב [8:13] Da yuvuklaštirdi Moše oylanlarin Aharonnin da giydirdi alarya gölmäklär

99 vo

- (1) kušadī alarnī inčķir bilan da giydirdi alarya öksak kavuklar nečiki simarladī
- (2) YWY Mošegä. ויגש [8:14] Da yuvuklaštirdi ol hatat buyasini da tayadi
- (3) Aharon da oylanlari kollarini baši üstünä ol hatat buyasinin. וישׁחּט [8:15] Da soydi
- (4) da aldī Moše ol kannī da verdi bučķaklarī üstünā ol mizbeahnīn čüpčüvrā
- (5) barmayî bilân da sačtî ol mizbeahnî da ol kannî tökti temâlinâ ol mizbeahnîn
- da ayruhsï etti anï keparat etmäkkä üstünä. ויקה [8:16] Da aldï ğümlä ol yaynï (6)
- (7) ki ol karın üstünä da ol bayırnın artiksisini da eki ol bögräklarnı
- da yaylarını da tütätti Moše ol mizbeahta. ואת [8:17] Da ol buyanı da terisini (8)
- (9) da etini da tezägini küydürdi otka avuldan čihari nečiki simarladi YWY
- (10)Mošegä. ויקרב [8:18] Da yuvuklaštirdi ol 'ola kočkarini da tayadilar Aharon
- (11)da oylanlari kollarini baši üstünä ol kočkarnin. וישחט [8:19] Da soydi da sačti
- (12)Moše ol kanni ol mizbeah üstünä čüpčüvrä. ואת [8:20] Da ol kočkarni
- (13)buvumladî buvumlarîn a da tüt atti Moše ol basnî da ol buvumlarnî da ol
- (14)karın yayını. ואת [8:21] Da ol karınını da ol ayaklarını yuvdı suvlar bilan da tütätti
- Moše tekmil ol kočkarni ol mizbeahta 'oladir ol kokusina kabulliknin (15)
- (16)otlu korbandir ol YWY-ya nečiki simarladi YWY Mošegä. ויקרב
- (17)[8:22] Da yuvuklaštirdi ol ekinži kočkarni ol milu im kočkarini da tayadilar Aharon
- (18)da oylanlari kollarini baši üstünä ol kočkarnin. וישׁחּט [8:23] Da soydi da aldi
- (19)Moše kanindan da verdi yimšayi üstünä ol on kulayinin Aharonnin da baš barmayi

¹⁶⁵ JSul.IV.02A: ta[.]imlarini; partially illegible text. | BSMS 288: sayitlarin. | JSul.III.01: savutlarin. |

ADub.III.73: savutlarin.

- (20) üstünä ol on kolunin da baš barmayi üstünä ol on ayayinin. ויקרב
- (21) [8:24] Da yuvuklaštīrdī oylanlarīn Aharonnīŋ da verdi Moše ol kandan yīmšayī üstünä ol
- (22) oŋ kulaklarının da baş barmayı üstünä ol oŋ kollarının da baş barmayı
- (23) üstünä ol on ayaklarinin da sačti Moše ol kanni ol mizbeah üstünä
- (24) čüpčüvrä. ויקח [8:25] Da aldï ol yaynï da ol kuyruknï da žümlä ol yaynï ki
- (25) ol karın üstünä da ol bayırnın artıksısını da eki ol bögraklarnı
- (26) da yaylarini da ol on butni. ומסל [8:26] Da ol mačalar sellesindän ki aldina YWY-

100 ro

- (1) nïŋ aldï bir kalïn mača da bir kalïn yaylï ötmäk da bir yufka da koydï
- (2) ol yaylar üstünä da ol oŋ but üstünä. ויתן [8:27] Da verdi ol ǯümläni
- (3) avučlarī üstünā Aharonnīŋ da avučlarī üstünā oylanlarīnīŋ da salladī alarnī
- (4) önünä sallamak aldına YWY-nın. ויקה [8:28] Da aldı Moše alarnı avučları üstündän
- (5) da tütätti ol mizbeaḥta ol 'ola üstünä milu'imdir alar kokusina
- (6) kabulliknin otlu korbandir ol YWY-ya. ויקח [8:29] Da aldi Moše ol töšni
- (7) da salladī anī sallamaķ aldīna YWY-nīn ķočķarīndan ol milu'imnin Mošegā
- (8) boldï payya nečiki sïmarladï YWY Mošegä. ויקה [8:30] Da aldï Moše yayından ol
- (9) silmäknin da ol kandan ki ol mizbeah üstünä da sačtī Aharon üstünä upraklarī
- (10) üstünä da oylanlari üstünä da upraklari üstünä oylanlarinin birgäsinä
- (11) da ayruhsi etti Aharonni upraklarini da oylanlarini da oylanlarinin upraklarini
- (12) birgäsinä. ויאמר [8:31] Da ayttï Moše Aharonya da oylanlarïna piširiniz ol
- (13) etni ešigindä ohel mo'ednin da anda ašaniz ani da ol ötmäkni ki ol
- (14) milu'im sellesindä nečiki sïmarladïm demä Aharon da oylanlarï ašasïnlar
- (15) anï. והנותר [8:32] Da ol kalyannï ol etdän da ol ötmäktän otka küydürüniz. ומפתח
- (16) [8:33] Ešigindän ohel moʻedniŋ čïkmaŋïz yedi günlär tamam bolyan gününä degin milu'im
- (17) günläriniz zira yedi günlär toldurip alištirir kolunizni. כאשר [8:34] Nečiki
- (18) kildi ušbu gündä simarladi YWY kilmaya kapara etmä üstünizgä. ופתח
- (19) [8:35] Da ešigindä ohel moʻedniŋ oturuŋïz gün da gečä yedi günlär da saklaŋïz saklovïn
- (20) YWY-nïŋ da ölmäŋiz zira alay sïmarlandïm. ויעש [8:36] Da kildï Aharon da oylanlarï
- (21) ǯümlä ol sözlärni ki sïmarladï YWY kolu bilän Mošenin.

פרשת שמיני¹⁶⁶ן (22)

נט) וַנְיָהִי בַּיוֹם הַשְּׁמִינִי קָּדָא מֹשֶׁה לְאַהְרֵן וּלְנִקְנִי יִשְׂרָא¹⁶⁷ (ני) (23...)

(...23) [9:1] Da edi ol

(24) sekizinži gündä čaķīrdī Moše Aharonnī da oylanlarīnī da ķartlarīn

$100 v^{0}$

- (1) Yisra'elnin. ויאמר [9:2] Da aytti Aharonya alyin üzünä bizov balasin siyirnin
- (2) hatatka da kočkar 'olaya saylamlarn[ï] da yuvuklaštiryin aldina YWY-nin. ואל
- (3) [9:3] Da oylanlarına Yisra'elnin sözlägin demä alınız ulayın eckilarnin hatatka
- (4) da bizov da koy birar yašarlar saylamlarni 'olaya. ושור [9:4] Da ögüz da kočkar šelamim-
- (5) lärgä korban etmä aldina YWY-nin da minha karišilyan yay bilän ki sahi
- (6) bügün kavodi YWY-nin aškara bolunir sizgä. ויקחו [9:5] Da aldilar neni ki simarladi
- (7) Moše aldīna ohel moʻednin da yuvuklaštīlar žümlä ol žamaʻat da turdīlar
- (8) aldına YWY-nın. ויאמר [9:6] Da ayttı Moše budır ol söz ki sımarladı YWY
- (9) ķiliŋiz da aškara bolsin sizgā kavodi YWY-niŋ. ויאמר [9:7] Da aytti Moše
- (10) Aharonya yuvuklašķīn ol mizbeaḥķa da ķīlyīn ḥatatīŋnī da 'olannī da kapara etkin
- (11) gendin učun da ol ulus učun da kilyin ol ulusnin korbanini da kapara
- (12) etkin alar učun nečiki sïmarladî YWY. ויקרב [9:8] Da yuvuklaštî Aharon ol
- (13) mizbeaḥķa da soydī ol ḥatat bīzovunī ki anīŋ. ויקריבו [9:9] Da yuvuķlaštīrdīlar
- (14) oylanları Aharonnın ol kannı anar da mäncti barmayını ol kanya da verdi buckakları
- (15) üstünä ol mizbeahnin da ol kanni tökti binyatina ol mizbeahnin. ואת
- (16) [9:10] Da ol yaynï da ol bögräklärni da ol artiksïnï ol bayïrdan ol hatatdan
- (17) tütätti ol mizbeahta nečiki sïmarladī YWY Mošegä. ואת [9:11] Da ol etni
- (18) da ol terini küydürdi otka avuldan čiḥari. וישׁחט [9:12] Da soydi ol 'olani
- (19) da yerištirdilär oylanlarï Aharonnïn anar ol kannï da sačtï anï ol mizbeah üstünä
- (20) čüpčüvrä. ואת [9:13] Da ol 'olanï yerištirdilär anar buvumlarïnža da ol
- (21) bašnï da tütätti ol mizbeah üstünä. וירחץ [9:14] Da yuvdï ol karïnnï da ol
- (22) ayaklarnï da tütätti ol 'ola üstünä ol mizbeaḥta. ויקרב [9:15] Da yuvuklaštīrdï

¹⁶⁶ The weekly Torah portion: Parashot Shemini (Lev 9:1–11:47).

¹⁶⁷ Lev 9:1 in Biblical Hebrew.

¹⁶⁸ JSul.IV.02A: saylamlarne; a scribal error. | BSMS 288: tügällär. | JSul.III.01: tigellerni. | ADub.III.73: tüģallarni.

- (23) ķorbanin ol ulusnin da aldī ol ḥatat ulayīnī ki ulusnin da soydī anī
- (24) da ḥatat etti anï burunyïnï gibi. ייקרב [9:16] Da yuvuklaštirdï ol 'olanï da kildï
- (25) anï šaraʿatča. ויקרב [9:17] Da yuvuklaštirdï ol minhanï da toldurdï avučïnï andan
- (26) da tütätti ol mizbeah üstünä 'olasından baška ol ertänin. וישׁחט

101 r^o

- (1) [9:18] Da soydï ol ögüzni da ol kočkarnï ol šelamim korbanïnï ki ulusnïn da yerištirdilär
- (2) oylanları Aharonnın ol kannı anar da sactı anı ol mizbeah üstünä cüpcüvrä
- (3) ואת [9:19] Da ol yaylarnï ol ögüzdän da ol kočkardan ol kuyruknï da ol yapkannï
- (4) da ol bögräklärni da artiksisin ol bayırnın (ayırdılar) (169. וישׁימו [9:20] Da koydılar ol yaylarnu
- (5) ol töšlär üstünä da tütätti ol yaylarni ol mizbeaḥta. ואת [9:21] Da ol töšlärni
- (6) da ol on butnî salladî Aharon sallamak önünä YWY-nîn nečiki sîmarladî YWY
- (7) Mošegä. וישׂא [9:22] Da kötärdi Aharon kollarini ol uluska da alyišladi alarni
- (8) da evvel endi kilmaktan ol hatatni da ol 'olani da ol šelamimlärni. ויבא
- (9) [9:23] Da geldi Moše da Aharon ohel moʻedgä da čiķtilar da alyišladilar ol ulusni
- (10) da aškara boldī kavodī YWY-nīŋ ǯümlä ol uluska. והצא [9:24] Da čīķtī ot
- (11) aldından YWY-nın da yaktı ol mizbeah üstünä ol 'olanı da ol yaylarnı da gördi
- (12...) ğümlä ol ulus da kičkirdilar da tüştilär yüzläri üstünä.

Chapter 10

- (...12) ויקחו [10:1] Da aldïlar
- (13) oylanlari Aharonnin Nadav da Avihu här kiši maḥtasini da verdilär alarda ot
- (14) da koydilar üstünä tütsü da yuvuklaštirdilar aldina YWY-nin yat ot ki
- (15) sïmarlamadï alarya. ותצא [10:2] Da čïķtï ot aldïndan YWY-nïŋ da yaķtï alarnï
- (16) da öldilär aldina YWY-nin. ויאמר [10:3] Da aytti Moše Aharonya oldir ne
- (17) ki sözlädi YWY demä yuvuklarima ayruhsi bolirmin da aldina ğümlä ol
- (18) ulusnın hörmätlänirmin da sustu Aharon. ייקרא [10:4] Da čaķirdi Moše Miša'elni
- (19) da Elčafanni oylanlarin 'Uzi'el ayačasinin Aharonnin da aytti alarya yuvuklašiniz
- (20) kaldırınız kardaşlarınızını yanından yüzlürinin ol kodeşnin avuldan ciharı.
- (21) ויקרבו [10:5] Da yuvuklaštīlar da kaldīrdīlar alarnī gölmäkläri bilän avuldan čīyarī

-

¹⁶⁹ JSul.IV.02A: Unvocalized word.

- (22) nečiki sözlädi Moše. ויאמר [10:6] Da ayttï Moše Aharonya da El'azarya da Itamar
- (23) oylanlarına başlarınıznı ağmanız da upraklarınıznı yirtmanız da ölmüniz
- (24) da ǯümlä ol ǯama ʿat üstünä ačuvlanmasïn da kardašlarïŋïz ǯümlä ǯama ʿat Yisra ʾel-
- (25) niŋ yïlasïnlar ol küymäkni ki küydürdi YWY. ומפתח [10:7] Da ešigindän ohel moʻed-
- (26) nin čikmaniz olmaya ölärsiz zira silmäk yayi YWY-nin üstinizdä da kildilar

101 v^o

- (1) sözinčä Mošenin. וידבר [10:8] Da sözlädi YWY Aharonya demä. יין [10:9] Yanıï šarap
- (2) eski šarap ičmägin nä sen ne da oylanlarin birgänä girdiginizdän
- (3) burun ohel moʻedgä da ölmäniz ömürlik räsim dävürlärinizgä. ולהבדיל [10:10] Da alay
- (4) ayïrmaya gelip arasïna ol kodešnin da arasïna ol yenilnin da arasïna ol
- (5) mundarnin da arasına ol aruvnin. ולהורות [10:11] Da ögrätmägä gelip oylanlarına
- (6) Yisra'elnin ğümlä ol räsimlärni ki sözlädi YWY alarya kolu bilän Moše-
- (7) niŋ. וידבר [10:12] Da sözlädi Moše Aharonya da El'azarya da Itamar oylanlarina
- (8) ol kalyanlarya aliŋiz ol kalyan minhani otlu korbanlarindan YWY-niŋ da ašaŋiz
- (9) anï mačalar yanïna ol mizbeaḥnïn zira kodeši kodešlärnindir ol. ואכלתם
- (10) [10:13] Da ašaniz ani ayruhsi yerdä zira hakkin da hakki oylanlarinnindir ol otlu
- (11) korbanlarından YWY-nın ki alay sımarlandım. ואת [10:14] Da ol sallamak tösini da
- (12) ol ayırmak butunı asanız temiz yerdä sen da oylanların da kızların birgana ki
- (13) hakkin da hakki oylanlarinnin verildilär šelamim korbanlarindan oylanlarinin Yisra'el-
- (14) nin. שוק [10:15] Ol ayırmak butunı da ol sallamak töšini otlu korbanları bilan ol
- (15) yaylarnın getirsinlar sallamaya sallamak aldına YWY-nın da bolsın sana da oylanlarına
- (16) birgānā ömürlik rāsimgā nečiki sīmarladī YWY. ואת [10:16] Da ol ḥatat
- (17) ulayını arama aradı Moše da muna küydürüldi da ačuvlandı El'azar üstünä
- (18) da Itamar üstünä ol kalyan oylanlari Aharonnin demä. מדוע [10:17] Nučun ašamadiniz
- (19) ol ḥatatnï ol kodeš ornïnda ki kodeši kodešlärnindir ol da anï verdi
- (20) sizgä kötärmägä günähin ol ǯamaʻatnïn keparat etmägä alar učun aldïna
- (21) YWY-nïn, און [10:18] Muna kanï girgizilmädi ol kodešgä ičkäri ašama ašanïz
- (22) anï ayruhsï yerdä nečiki sïmarladïm. וידבר [10:19] Da sözlädi Aharon Mošegä
- (23) muna bügün yuvuklaštirdilar hatatlarini da 'olalarini aldina YWY-nin učradilar
- (24) mana bular gibi da ašasam eydim hatat ätini bügün yahsi bolirmi

(25) edi 'enayatlarïna YWY-nïŋ. וישמע [10:20] Da ešitti Moše da yaḥšï boldï gözlärinä.

Chapter 11

(26) (אי) וידבר [11:1] Da sözlädi YWY Mošegä da Aharonya demä alarya.

102 ro

- (1) דברו [11:2] Sevläniz evladlarïna Yisra'elin demä budur o ḥayvanat ki yeyäsiz ǯümlä
- (2) o tuvardan ki o yer üzerinä. כל [11:3] Här čatal tïrnaklïyï ve ayrīǯïnï ayïrmayïn
- (3) tirnaknin gevšemäk getirenni tuvarda onu ašaniz. אך [11:4] Amma gevšemäk
- (4) getirenlerden ve _Lčatal tïrnaķlïlardan¹¹⁷⁰ bularī yemeyesiz o deveni farzam ki gevšer amma
- (5) tamam ayrî tîrnaklî degildir murdardîr o sizä. ואת [11:5] Ve o ada tavušan
- (6) (kirpi)¹⁷¹ -nï¹⁷² farzam ki o gevšer amma tamam tïrnak ayïrmaz murdardïr o size.
- (7) ואת [11:6] Ve o tavušanï farzam ki gevšer amma tamam tïrnak ayïrmadï murdardïr
- (8) o size. ואת [11:7] Ve o hinziri (hazirni) farzam ki čatal tirnaklidir vä tamam ayirir
- (9) ayırmasın tırnaknın amma o gevšeme gevšemez murdardır o sizä. מבשרם
- (10) [11:8] Etlärindän yemeniz ve lešlerine degmeniz ḥaramdir olar size. את [11:9] Bunu yeyesiz
- (11) ǯümlädän ki suvda her šey ki ona kanat ve pul suvda denizlerde da özänlärdä
- (12) olari yeyesiz. וכל [11:10] Ve her šey ki yok ona kanat ve pul denizlerde özänlärdä
- (13) ğümlä kozlamasından suvnun ve ğümlä o diri ğandan ki suvda mikruhdir
- (14) olar size. נְשׁקִץ [11:11] Ve mikruh olsunlar size etlerinden yemey[e]siz¹⁷³ ve lešlerini
- (15) ikrah idäsiz. כל [11:12] Her šey ki yok ona kanat ve pul suvlarda mikruhdir
- (16) o size. ואת [11:13] Ve bulari ikrah idiniz o kuštan ašalmasinlar mikruhdir
- (17) olar o kartalnî (kara kušnî) ve iron[n]î¹⁷⁴ ve deŋiz kartalînî. ואת [11:14] Ve ak babanî
- (18) ve čaylakni žinsinžä. את [11:15] Här kuzyuni žinsinčä. ואת [11:16] Ve deve kušuni

¹⁷⁰ JSul.IV.02A: *türnaķ čatallīlardan*; probably a scribal error. | BSMS 288: *tuyaķ etivčilārdān*. | JSul.III.01: *tuyaklavculardan*. | ADub.III.73: *tuyaklavčulardan*.

¹⁷¹ Although it is written in a parenthesis, the word differs from *ada tavušan* 'rabbit' and stands for 'hedhegog'. This different interpretation is also present in BSMS 288, e.g., *kirpi*. | JSul.III.01: *krolik*. | ADub.III.73: *krolik*. ¹⁷² The accusative suffix belongs to the word *tavušan*.

¹⁷³ JSul.IV.02A: *yemeysiz*; probably a scribal error. | BSMS 288: *ašamaŋïz*. | JSul.III.01: *asamanïz*. | ADub.III.73: *ašamayïz*.

¹⁷⁴ JSul.IV.02A: *ironïnï*; erroneously copied from some Ottoman Bible translations, e.g., Cod. Or. 1101a-f: *eronï/erunï*; 4B.Or.131-1: *ironï/irunï* (see Işık 2021: 350). | BSMS 288: *peresni*. | JSul.III.01: *peresni*. | ADub.III.73: *peresni*.

- ve bay¹⁷⁵ ķušnī ve ķuķu ķušīnī ve duyanī žinsinčā. ואת [11:17] Ve ügi ķušunī ve ķara bataknī
- (20) ve to[n]ï 176 . ואת [11:18] Ve koyunï ve kašïkčï kušïnï ve rahamnï 177 . ואת [11:19] Ve leglägni ve balïkčïnï
- (21) ǯinsinčä ve thüd hüdni¹¹⁷⁸ ve šepereni. כל [11:20] Ve her učan kušu dört ayak üzerinä gezen
- (22) mïkruḥdïr o size. אך [11:21] Salt bunu yeyesiz ǯümlä doɣurduɣundan o kušnïŋ
- (23) o gezen dört ayak üzärine ki ona dizlär yokarïdan ayaklarïna sïčrama
- (24) olar ilen o yer üzerine. את [11:22] Bularnï olardan ašaŋïz o čekirtkeni
- (25) žinsinžä ve žudžudī žinsinčä ve ḥargolnī žinsinžä ve ḥagavnī žinsinžä.
- (26) וכל [11:23] Ve ğümlä doyurduyu o kušnin¹⁷⁹ ki ona dört ayaklar mikruhadir o

102 vo

(1) size. ולאלה [11:24] Ve bularya tiyip mundar olursïz här deggen lešlerinä mundar

- (2) olsïn o aḥšama degin. וכל [11:25] Ve o (tašïyan) ķaldïran läšlärindän yïķasïn urbalarïnï
- (3) umundar olsun ol ahšama degin. לכל [11:26] Här hayvan ki o čatal tïrnaklidir
- (4) vä tamam ayirmak ayirmayur ve gevšemek getirmeyir murdarlardir olar sizä här
- (5) deggen lešlerinä mundar olsïn o ahšama degin. וכל [11:27] Žümlä gezen tabanlarï
- (6) üzerine ğümlä ğenavarda o gezen dört ayak üzerine haramlar olar size ğümlä
- (7) o tiygän lešlerinä mundar olsun o ahšama degin. והנשא [11:28] Ve o kaldïran
- (8) lešlärini yïkasïn rubalarïnï da mundar olsun o ahšama kadar haramlardïr olar
- (9) size. אוד [11:29] Ve budur size o ḥaram doyyan da o kozlayan o yer üzerinä
- (10) o gelinčik ve o sičan ve o kapli baya žinsinžä. והאנקה [11:30] Ve sivri sičan

_

¹⁷⁵ JSul.IV.02A: Spelled בֵּ; a scribal error.

¹⁷⁶ JSul.IV.02A: *toyïnï*; a scribal error. Although the Karaim dictionaries have listed the form *toyï* for Crimean Karaim denoting 'ibis' (CKED: 410, KRPS: 535), it was erroneously copied from some Ottoman Bible translations, e.g., Cod. Or. 1101a-f: *to/tu*; 4B.Or.131-1: *to/tu* 'a species of owl' (see Işık 2021: 356–357). | BSMS 288: *yanšufnï*. | JSul.III.01: *kuzyunnï*. | ADub.III.73: *yanšufnï*.

¹⁷⁷ JSul.IV.02A: *raḥamini*; probably a scribal error. | BSMS 288: *raḥamani*. | JSul.III.01: *raḥamni*. | ADub.III.73: *raḥamni*.

¹⁷⁸ JSul.IV.02A: *hüdhüdini*; erroneously copied from some Ottoman Bible translations e.g., Cod. Or. 1101a-f: *hüdhüdi*; 4B.Or.131-1: *hüdhüdi* (see Işık 2021: 359). | BSMS 288: *duḫifatni*. | JSul.III.01: *duḫifatni*. | ADub.III.73: *duhifatni*.

¹⁷⁹ JSul.IV.02A: kušunnin; a scribal error. | BSMS 288: kušnin. | JSul.III.01: kusnun. | ADub.III.73: kušnun.

- ve Lgüneš keleri 180 ve Lyïldïz keleri 181 ve Lkerten kele 182 ve Lgöz töbä[k] 183 . אלה [11:31] Bular o ḥaramlar
- (12) size ǯümlä o doyyan da ǯümlä o deggen olara üldiklerindän sora
- (13) ḥaram olsun o aḥšama degin. וכל [11:32] Ve her šey ki düšerse üzerinä olardan
- (14) üldüklärindän sora haram olsun ğümlä ayač avadanlıkdan (sayıttan) yahud
- (15) espap ya deri ya čuval här avadanlik ki yapulur iš olar ilän suvlara
- (16) girgizilsin ve ḥaram olsun o aḫšama kadar ve pak olur. וכל [11:33] Ve ǯümlä saksï
- (17) kapï ki düšerse birisi olardan ičerisinä här ne ki ičerisindä haram
- (18) olsun da o kapnï kirinjiz. מכל [11:34] Žümlä o yemekdän ki ašalir ki gelirse üzärinä suv
- (19) ḥaram olur ve ǯümlä ički ki ičilir här kap ičindä ḥaram olur.
- (20) וכל [11:35] Ve her šey ki düšerse üzerine lešlerindän ḥaram olur furun vä očaķ bozulsun
- (21) ḥaramlardir olar vä ḥaramlar olsunlar sizä. וכל [11:36] Tek češmä ve sarnič
- (22) devširilmesi suvnun olur helal ve deggen lešlerine haram olur. וכל [11:37] Da egär
- (23) düšerse lešlerindän här dürlü ekilmiš ekin üzerine ki ekilse helaldir
- (24) o. וכל [11:38] Ve egär verilsä suv tohum üzerinä ve düšersä lešlerindän üzerinä
- (25) ḥaramdir o size. וכי [11:39] Da egär ülürsä o tuvardan ki ḥelaldir o sizä
- (26) yemeye o deggen leši[n]ä¹⁸⁴ ḥaram olsun o aḥšama kadar. והאכל [11:40] Ve o yeyen

$103 r^{0}$

- (1) lešindän yïkasïn urbalarïnï da mundar olsun o ahšama kadar ve o kaldïran lešini
- (2) yïkasïn rubalarïnï da mundar olsun o aḥšama kadar. זכל [11:41] Ve ǯümlä o doyyan
- (3) šey ki doyayîr o yer üzerinä mîkruh šeydir yelmesin. כל [11:42] Här gezän
- (4) kursak üzerinä ve här gezän dört ayak üzerinä žümlä o doyyandan ki doyayïr o
- (5) yer üzerinä yemeyäsiz olari zira mikruh šeydir olar. אל [11:43] İkrah
- (6) idtirmäniz ǯanlarinizi biri ilän o doyyannin ki kozlayir da mundar olunmaniz o-

¹⁸⁰ JSul.IV.02A: güneš kelerisi; erroneously copied from 4B.Or.131-1: güneš keleri (see Işık 2021: 366). | BSMS 288: kovaḥ. | JSul.III.01: koaḥ. | ADub.III.73: kovaḥ.

¹⁸¹ JSul.IV.02A: *yildiz kelerisi*; erroneously copied from 4B.Or.131-1: *yildiz keleri* (see Işık 2021: 366–367).| BSMS 288: *leta'a*. | JSul.III.01: *leta'a*. | ADub.III.73: *leta'a*.

¹⁸² JSul.IV.02A: *kerten kelesi*; erroneously copied from some Ottoman Bible translations, e.g., Cod. Or. 1101a-f; *kerten kele*; 4B.Or.131-1: *kerten kele* (see Işık 2021: 366). |BSMS 288: *homet*. | JSul.III.01: *homet*. | ADub.III.73: *homet*.

¹⁸³ JSul.IV.02A: *göz töbä*; erroneously copied from the Ottoman Bible translations, e.g., Cod. Or. 1101a-f: *köstebek*; 4B.Or.131-1: *köstebek* (see Işık 2021: 363). | BSMS 288: *tinšemet*. | JSul.III.01: *tinšamet*. | ADub.III.73: *tinšamet*.

¹⁸⁴JSul.IV.02A: *lešiyä*; probably a scribal error. | BSMS 288: *gövdäsinä*. | JSul.III.01: *gevdesine*. | ADub.III.73: *gövdasińa*.

- (7) lar ilän zira kapanir akiliniz olar ilän. כי [11:44] Ki menmin YWY Tänriniz ayruhsi
- (8) olunuz ve olunuz ķudušlar zira ķudušim ben ve ḥaram idmeyäsiz ǯanlarinizi ǯümlä
- (9) o doyyan ilän o ķïmïldayan o yer üzerinä. כי [11:45] Ki menmin YWY o čïķaran
- (10) sizi yerindän Mïsïrïn olma size Täŋrigä d[a]¹⁸⁵ oluŋuz kudušlar zira kudušïm
- (11) ben. ואת [11:46] Budur torasï o tuvarnïn ve o kušnïn ve žümlä o diri
- (12) ǯan[nïŋ]¹⁸⁶ o ķïmïldayan suvlarda ve ǯümlä ǯan[nïŋ]¹⁸⁷ o doyuran o yer üzerinä.
- (13) להכדיל [11:47] Farķ idmegä arasïna o ḥaramïn ve arasïna o helalin ve arasïna
- (14) ašalažaķ ḥayvanatnin ve arasina o ḥayvanatnin ki ašalmayažaķ.

- ¹⁸⁸ר תזריע (15)
- ווָדַבר יוָי אֵל משֶׁה לֵאמר¹⁸⁹ן, (16...)
- (...16) [12:1] Da sözlädi YWY Mošegä demä. דבר
- (17) [12:2] Sözlägin oylanlarina Yisra'elnin demä hatin ki urluk etsä
- (18) da doyursa erkäk da mundar bolsin yedi günlär günläri gibi arsizliyinin nida-
- (19) lïyïnïŋ mundar bolsïn. וביום [12:3] Da ol sekizinǯi gündä sünätlänsin eti
- (20) aklafinin. ושׁלשים [12:4] Da otuz üč günlär otursin kanlari bilän temiz-
- (21) liknin hič kodešgä tiymäsin da ol mikdaška gelmäsin tamam bolyanïna degin
- (22) günläri temizliginin. ואם [12:5] Da egär tišini doyursa da mundar bolsïn eki hafta
- (23) nidalïyï učun da altmïš altï günlär otursïn kanlarï učun temizliginin.
- (24) ובמלאת [12:6] Da tamam bolyanında günları temizliginin oyulya ya kızya getirsin
- (25) bir yašar koy 'olaya da balasın gügürčinnin ya kumru hatatka ešiginä ohel

$103 v^{0}$

- (1) mo'ednin ol kohengä. והקריבו [12:7] Da yuvuklaštirsin ani aldina YWY-nin da kapara
- (2) ätsin üstünä da aruv bolsïn čokrayından kanlarının budir torası ol
- (3) doyuryannin erkäkni ya tišini. ואם [12:8] Da egär yetmäsä kuvati koyya da alsin
- (4) eki kumrular ya eki balalarin kögürčinnin birni 'olaya da birni hatatka da kapara
- (5) etsin üstünä ol kohen da temiz olur.

_

¹⁸⁵ JSul.IV.02A: *d*; a scribal error. | BSMS 288: *da*. | JSul.III.01: *da*. | ADub.III.73: *da*.

¹⁸⁶ JSul.IV.02A: *ǯannŋïn*; a scribal error. | BSMS 288: *ǯannïŋ*. | JSul.III.01: *ʒannïn*. | ADub.III.73: *ǯannïn*.

¹⁸⁷ JSul.IV.02A: *ǯannŋïn*; a scribal error. | BSMS 288: *ǯannïŋ*. | JSul.III.01: *ʒannïn*. | ADub.III.73: *ǯannïn*.

¹⁸⁸ The weekly Torah portion: Parashot Tazria (Lev 12:1–13:59).

¹⁸⁹ Lev 12:1 in Biblical Hebrew.

- (6) (גי) וידבר [13:1] Da sözlädi YWY Mošegä da Aharonya demä. אדם [13:2] Adam ki
- (7) bolsa terisindä etinin šiši[k]¹⁹⁰ ya oyuz ya ak lekä
- (8) da bolsa terisindä etinin hastaliyi žüzam derdliknin da getirilsin Aharon ol
- (9) kohengä ya birsinä oylanlarindan ol kohenlärgä. וראה [13:3] Da görsin ol kohen
- (10) ol hastaliķni terisindā ol etnin da sač hastalikta degiširildi isā ak
- (11) da görümi ol hastaliknin terän isä terisindän etinin hastaliyi ǯüzam derdliknin
- ol da görsä anï ol kohen da mundar etsin anï. ואם [13:4] Da egär ak leke
- (13) esä ol terisindä etinin da terän dügül esä görümï ol teridän da sačï
- (14) degiširilmädi esä ak da bekläsin ol kohen ol hastaliyi yedi günlär. וראהו
- (15) [13:5] Da görsin anï ol kohen ol yedinži gündä da muna ol hastalïk turdï renkindä
- (16) yayılmadı ol hastalık terida da beklasin anı ol kohen yedi günlar ekinği. וראה
- (17) [13:6] Da görsin ol kohen anï ol yedinǯi gündä ekinǯi kerät da muna sönükkän
- (18) ol hast[a]lik¹⁹¹ da yayilmadi ol hastalik teridä temiz etsin ani ol kohen
- (19) oyuzdir ol da yuvsin upraklarin da temiz bolsin. ואם [13:7] Da egär yayilma
- (20) yayilsa ol oyuz teridä görüngänindän sonra ol kohengä temiz bolyanindan
- (21) soņra da görünsin ekinži kerät ol kohengä. וראה [13:8] Da görsä ol kohen
- (22) da muna yayıldı ol oyuz teridä da mundar etsin anı ol kohen ğüzam derdlikdir
- (23) ol. נגע [13:9] Hastalïyï ǯüzam derdlik ki bolsa adamda da getirilsin ol kohen-
- (24) gä. וראה [13:10] Da görsin ol kohen da muna ak šišik teridä da ol degišti ak
- (25) sač da saylïyï čiy etnin šiši[k]dä¹⁹². צרעת [13:11] Eskirgän ǯüzam derdlikdir
- (26) ol terisindä etinin da mundar etsin ani ol kohen beklämäsin ani zira mundar-

104 ro

- (1) dïr ol. ואם [13:12] Da egär yayïlma yayïlsa ol ǯüzam derdlik täridä da yapsa ol
- (2) juzam derdlik jumla terisin ol hastaliknin başından da ayaklarına degin jumla
- (3) görüminä gözlärinin ol kohennin. וראה [13:13] Da görsin ol kohen da muna yapti ol
- (4) jūzam derdlik jūmlä etini temiz etsin ol hastalīķnī jūmläsi
- (5) degišdi ak [t]emi[z]dir¹⁹³ ol. וביום [13:14] Da görüngän gündä anda čiy et mundar

¹⁹⁰ JSul.IV.02A: šišik; a scribal error. | BSMS 288: šišik. | JSul.III.01: sisik. | ADub.III.73: šišik.

¹⁹¹ JSul.IV.02A: hastlik; a scribal error. | BSMS 288: hastalik. | JSul.III.01: hastalik. | ADub.III.73: hastalik.

¹⁹² JSul.IV.02A: šišikdä; a scribal error. | BSMS 288: šišikdä. | JSul.III.01: sisikte. | ADub.III.73: küvmaktá.

¹⁹³ JSul.IV.02A: [.]*em*i[.]*dir*; a partially illegible text. | BSMS 288: *aruvdir*. | JSul.III.01: *aruvdu*. | ADub.III.73: *aruvdur*.

- (6) bolsin. וראה [13:15] Da görsin ol kohen ol čiy etni da mundar etsin ani ol
- (7) čiy et munda[r]dïr¹⁹⁴ ol ǯüzam derdlikdir ol. אר [13:16] Ya ki kaytsa ol čiy et
- (8) da degiširildi isä akka da gelsin ol kohengä. וראה [13:17] Da görsin anï ol
- (9) kohen da muna degiširildi ol hastalik akka da temiz etsin ol kohen ol
- (10) hastaliķni temizdir ol. וכשר [13:18] Da et ki bolsa anda terisindä čiban
- (11) da oŋalsa. והיה [13:19] Da bolsa yerindä ol čïbannïŋ ak šišik ya ak lekä
- (12) ķīrmīzīža da görünsin ol kohengä. וראה [13:20] Da görsin ol kohen da muna görümi
- (13) alčaķ ol teridän da sačī degišdi aķ da mundar etsin anī ol kohen
- (14) hastaliyi ğüzam derdliknindir ol čibanda yayildi. ואם [13:21] Da egär görsä ani
- (15) ol kohen da muna yoktir anda ak sač da alčak dügüldir ol teridän da ol
- (16) sönükkän da bekläsin anï ol kohen yedi günlär. ואם [13:22] Da egär yayïlma yayïlsa
- (17) teridä da mundar etsin ol kohen anï hastalïkdïr ol. ואם [13:23] Da egär yerindä
- (18) tursa ol ak lekä yayilmasa küyügi ol čibannindir ol da temiz etsin ani
- ol kohen. או [13:24] Ya et ki bolsa terisindä küygäni otnïŋ da bolsa čiy
- (20) sayï ol küygännin beyazča ķïrmïzïǯa lekä ya ak. וראה [13:25] Da görsin anï ol
- (21) kohen da muna degiširildi ak sač lekädä da görümi derän ol teridän ǯüzam
- (22) derdlikdir ol küyükdä yayildi da mundar etsin ani ol kohen hastaliyi ğüzam
- (23) derdliknindir ol. ואם [13:26] Da egär görsä anï ol kohen da muna yoktïr lekä-
- (24) dä ak sač da alčak dügüldir ol teridän da ol sönükkän da bekläsin anï ol
- (25) kohen yedi günlär. וראהו [13:27] Da görsin anï ol kohen ol yedinǯi gündä egär
- (26) yayılma yayılsa teridä da mundar etsin ol kohen anı hastalıyı ğüzam derdlikinindir

104 vo

- (1) ol. ואם [13:28] Da egär yerindä tursa ol lekä yayilmasa teridä da ol
- (2) sönükkän šišigi ol küyüknindir ol da temiz etsin anï ol kohen zira
- (3) küyügi ol küyüknindir ol. ואיש [13:29] Da kiši ya hatin ki bolsa anda hastalik
- (4) bašta ya sakalda. וראה [13:30] Da görsin ol kohen ol hastalikni da muna görümi
- (5) terän ol teridän da anda sarï sač inǯä da mundar etsin anï ol kohen neteķ-
- (6) dir ol ǯüzam derdligi¹⁹⁵ ol bašnïŋ ya ol sakalnïŋdïr ol. וכי [13:31] Da egär
- (7) görsä ol kohen hastaliyin ol neteknin da muna dügüldir görümi terän ol

¹⁹⁴ JSul.IV.02A: *mundandir*; probably a scribal error. | BSMS 288: *mundardir*. | JSul.III.01: *murdardi*. | ADub.III.73: *murdardi*.

¹⁹⁵ JSul.IV.02A: derdirligi; a scribal error. | BSMS 288: mečora liyi. | JSul.III.01: cara atidi. | ADub.III.73: cara ati.

- (8) teridän da kara sač yoktir anda da bekläsin ol kohen hastaliyin ol neteknin
- (9) yedi günlär. וראה [13:32] Da görsin ol kohen ol hastalïknï ol yedinži gündä da muna
- (10) yayılmadı ol netek da bolmadı anda sarı sac da görümi ol neteknin terän
- (11) dügül ol teridän. והתגלה [13:33] Da tïraš olsïn da ol netekni tïraš idmäsin
- (12) da bekläsin ol kohen ol netekni yedi günlär ekinži kerät. וראה [13:34] Da görsin
- (13) ol kohen ol neteķni ol yedinži gündā da muna yayīlmadī ol neteķ teridā
- (14) da görimi terän dügül ol teridän da temiz etsin anï ol kohen da yuvsïn upraklarïn
- (15) da temiz bolïr. ואם [13:35] Da egär yayïlma yayïlsa ol netek teridä
- (16) temiz bolyanından sonra. וראהו [13:36] Da görsin anı ol kohen da muna yayıldı ol
- (17) neteķ teridā aramasīn ol kohen ol sarī sačķa mundardīr ol. ואם
- (18) [13:37] Da egär renkindä turdï esä ol netek da kara sač bitti esä anda oŋaldï ol
- (19) netek temizdir ol da aruv etsin ani ol kohen. ואיש [13:38] Da kiši ya hatin ki
- (20) bolsa terisindä etlärinin lekälär ak lekälär. וראה [13:39] Da görsin ol kohen
- (21) da muna terisindä etlärinin ak lekälär sönükkänlär bohakdir ol yayildi
- (22) teridä temizdir ol. ואיש [13:40] Da kiši ki yurkulsa sačī bašīnīŋ ķereaḥdīr
- (23) ol temizdir ol. ואם [13:41] Da egär yüzlärinin tarafından yurkulsa sači bašının
- (24) gibeaḥdir temizdir ol. וכי [13:42] Da egär bolsa artta ya aldda ḥastaliķ aķ
- (25) ķīrmīzīža yayīlī turyan žūzam derdlik ol ardīnda ya aldīnda. וראה [13:43] Da görsin
- (26) anï ol kohen da muna šišigi ol hastaliķnin ak ķirmiziža ardindan ya

105 ro

- (1) aldında görümi gibi ğüzam derdlik terisinin etnin. איש [13:44] Žüzam derdlikdir
- (2) ol mundardir ol mundar etmä mundar etsin ani ol kohen bašinda-
- (3) dîr hastaliyî. והצרוע [13:45] Da ol žüzam derdli ki anda ol hastalik upraklarî
- (4) bolsïnlar yirtilyanlar da baši bolsin ačiķ da miyiķni sarsin da mundardir
- (5) mundardir čaķirsin. כל [13:46] Žümlä günlär ki ol hastalik anda mundar bolsin
- (6) mundardir ol yalyiz otursin avuldan čihari oturaši. והבגד [13:47] Da ol
- (7) upraķ ki bolsa anda hastalīyī žūzam derdliknin yūn upraķta ya üskūlū upraķda.
- (8) אל [13:48] Ya boyda ya arkuvda üskülüdän da yündän ya teridä ya ǯümlä teri
- (9) išindä. והִיה [13:49] Da bolsa ol hastalïk yäšilǯä ya kirmïzïǯa uprakta ya
- (10) teridä ya boyda ya arkuvda ya ğümlä teri takimda hastaliyi ğüzam derdliknin-
- (11) dir ol da görünsin ol kohengä. וראה [13:50] Da görsin ol kohen ol hastalïknï
- (12) da bekläsin ol hastalikni yedi günlär. וראה [13:51] Da görsin ol hastalikni ol yedinži

- (13) gündä ki yayildi ol hastalik uprakta ya boyda ya arkuvda ya teridä
- (14) ǯümlägä ki ķïlïnïr ol teri iškä sïzlatïǯï ǯüzam derdilikdir
- (15) ol hastalik mundardir ol. ושׂרף [13:52] Da küydürsin ol uprakni ya ol boyni
- (16) ya ol arķuvnī ki yündān ya üskülüdān ya ǯümlā ol teri taķīmnī ki
- (17) bolsa anda ol hastalik zira sizlati[3]i¹⁹⁶ 3üzam derdlikdir ol otka
- (18) küydürülsin¹⁹⁷. ואם [13:53] Da egär görsä ol kohen da muna yayilmadi ol hastalik
- (19) upraķta ya boyda ya arķuvda ya ǯümlä teri taķïmda. וצוה [13:54] Da sïmarlasïn
- (20) ol kohen da yuvsïnlar neni ki anda ol hastalïk da bekläsin anï yedi günlär
- (21) ekinži kerät. וראה [13:55] Da görsin ol kohen yuvulyanindan sonra ol hastalikni
- (22) da muna degišmādi ol ḫastalīķ renkini da ol ḫastalīķ yayīlmadī mundardīr ol
- (23) otka küydürgin ani pehetetdir ol artinda ya aldında. ואם [13:56] Da egär
- (24) görsä ol kohen da muna sönükkän ol hastalik yuvulyanindan sonra ol
- (25) da yirtsin ani ol upraktan ya ol teridan ya ol boydan ya ol arkuvdan.
- (26) ואם [13:57] Da egär görünsä dayin uprakta ya boyda ya arkuvda ya ğümlä teri

$105 v^{o}$

- (1) taķīmda yayīlyan derddir ol otķa küydürgin anī neni anda ol hastalīķ.
- (2) והבגד [13:58] Da ol uprak ya ol boy ya ol arkuv ya ğümlä ol teri takim
- (3) ki yuvsan da ke[ts]ä¹⁹⁸ alardan ol hastalïk da yuvulsïn ekinži kerät da temiz
- (4) bolsïn. זאת [13:59] Budïr torasï hastalïyïnïn ǯüzam derdliknin ol yün upraknïn
- (5) ya ol üskülünin ya ol boy[n]ïn¹⁹⁹ ya ol arkuvnïn ya ǯümlä teri takïmnïn
- (6) temiz etmägä anï ya mundar etmägä anï.

Chapter 14

 200 ן תזריע (7) פרשת (7)

- (8) (יד) נְיְדַבֵּר [14:1] Da sözlädi YWY Mošegä demä. זאת [14:2] Bu bolsïn torasï
- (9) ol ǯüzam derdliknin temiz bolaǯak günündä da ketirilsin
- (10) ol kohengä. ויצא [14:3] Da čiķsin ol kohen avuldan čiḥari da görsin ol kohen
- (11) da muna oŋaldï hastalïyï ol ǯüzam derdlikniŋ ol ǯüzam derdli bolyandan.

247

¹⁹⁶ JSul.IV.02A: sïzlatïyï; a scribal error. | BSMS 288: eksitivči. | JSul.III.01: sïzlatadoyandï. | ADub.III.73: tavusadoyan.

¹⁹⁷ JSul.IV.02A: *küydürüläsin*; a scribal error. | BSMS 288: *küydürülsin*. | JSul.III.01: *kivdirilsin*. | ADub.III.73: *küvdürülsün*.

¹⁹⁸ JSul.IV.02A: kestä; a scribal error. | BSMS 288: ketsä. | JSul.III.01: ketse. | ADub.III.73: ketśa.

¹⁹⁹ JSul.IV.02A: boyyin; a scribal error. | BSMS 288: boynin. | JSul.III.01: osnovanin. | ADub.III.73: osnovanin.

²⁰⁰ The weekly Torah portion: Parashot Metzora (Lev 14:1–15:33).

- (12) וצוה [14:4] Da sïmarlasïn ol kohen da alsïn temiz bolažaķ učun eki yabani helal ķušlarï
- (13) da sälvi ayači da kirmizi²⁰¹ iränkli yipek da ezöv. וצוה [14:5] Da simarlasin ol kohen
- (14) da soysin ol bir kušni čöräp sayitda tatli suvlar üstünä. את [14:6] Ol
- (15) yaban kušnī alsīn anī da ol sālvi ayačnī da ol ķīrmīzī irānkli yipekni
- (16) da ol ezövni da mänčsin alarnï da ol yaban kušnï kanïna ol soyulyan kušnïn
- (17) ol tatli suvlar üstünä. והזה [14:7] Da sačsin ol temizlängän üstünä
- (18) ol ǯüzam derdliliktän yedi kerätlär da temiz etsin anï da yibirsin ol yaban
- (19) kušnī yüzü üstünä ol düznin. וכבס [14:8] Da yuvsïn ol temiz bolyan upraklarīnī
- (20) da tïraš idsin ǯümlä sačï[n]ï²⁰² da yuvunsïn suvlar bilän da temiz bolsïn da andan
- (21) soŋra gelsin ol avulya da otursïn čadïrïndan čïḥarï yedi günlär. והיה
- (22) [14:9] Da bolγay ol yedinži gündä tïraš idsin žümlä sačï[n]ï²⁰³ bašïnï da sakalï[n]ï²⁰⁴ da
- (23) kirpi[k]lärin²⁰⁵ gözlärinin da ğümlä sačini tiraš idsin da yuvsin upraklari[n]i²⁰⁶
- (24) da yuvsïn etini suv bilän da temiz bolsïn. וביום [14:10] Da ol sekizinži gündä alsïn

106 ro

- (1) eki saylam koylar da bir tiši koy bir yašar saylam da üč payï onnïn
- (2) özäk karïšïlyan yay bilän da bir seyik yay. והעמיד [14:11] Da turyuzsïn ol temiz
- (3) etkän kohen ol temizlängän kišini da alarnï aldïna YWY-nïŋ ešigindä ohel
- (4) mo'ednin. ולקה [14:12] Da alsïn ol kohen ol bir koynï da yuvuklaštirsïn anï
- (5) ašamya da ol seyik yaynï da sallasïn alarnï sallamak önünä YWY-nïη.
- (6) ושחט [14:13] Da soysin ol koyni ne yerdä ki soyar ol hatatni da ol 'olani yerindä
- (7) ol kodešnin ki hatat gibidir ol ašam ol kohengä kodeši kodeš-
- (8) lärnindir ol. ולקח [14:14] Da alsïn ol kohen kanïndan ol ašamnïn da versin ol
- (9) kohen yimšayï üstünä on kulayïnïn ol temiz bolyanïn da baš barmayï üstünä
- (10) oŋ kolunïŋ da baš barmayï üstünä oŋ ayayïnïŋ. ולקה [14:15] Da alsïn ol kohen
- (11) ol seyik yaydan da koysin ol sol avuču üstünä (özünin) ol kohennin.
- (12) וטבל [14:16] Da mänčsin ol kohen on barmayïnï ol yaydan ki ol sol avuču üstünä

²⁰¹ JSul.IV.02A: kirimizi; probably a scribal error. | BSMS 288: kirmizi. | JSul.III.01: kirmizi. | ADub.III.73: kirmizi.

²⁰² JSul.IV.02A: sačiņi; a scribal error. | BSMS 288: sačin. | JSul.III.01: cacin. | ADub.III.73: čači.

²⁰³ JSul.IV.02A: *sačiηi*; a scribal error. | BSMS 288: *sačin*. | JSul.III.01: *cacin*. | ADub.III.73: *čačin*.

²⁰⁴ JSul.IV.02A: sakaliņi; a scribal error. | BSMS 288: sakalin. | JSul.III.01: sahalin. | ADub.III.73: sayalin.

²⁰⁵ JSul.IV.02A: *kirpiķlärin*; a scribal error. | BSMS 288: *kirpiklärin*. | JSul.III.01: *kaslarin*. | ADub.III.73: *kašlarin*.

²⁰⁶ JSul.IV.02A: *upraklariŋi*; a scribal error. | BSMS 288: *upraklarin*. | JSul.III.01: *upraklarin*. | ADub.III.73: *upraklarin*.

- (13)da sačsïn ol yaydan barmayî bilän yedi kerätlär aldına YWY-nın. ומיחד [14:17] Da kalyan
- (14)ol yaydan ki avuču üstünä versin ol kohen yimšayi üstünä on kulayinin
- (15)ol temiz bolyannın da baş barmayı üstünä on kolunın da baş barmayı
- (16)üstünä on ayayinin ol ašam kaninin yeri üstünä. והנותר [14:18] Da ol
- kalyannï yaydan ki avuču üstünä ol kohennin versin bašï üstünä ol temiz (17)
- (18)bolyannin da kapara etsin üstünä ol kohen aldina YWY-nin. ועשה
- (19)[14:19] Da ķilsin ol kohen ol ḥatatni da kapara etsin ol temiz bolyan učun
- (20)mundarliyindan da andan sonra soysin ol 'olani. והעלה [14:20] Da čiyarsin ol kohen
- (21)ol 'olanï da ol minhanï ol mizbeahka da kapara etsin üstünä ol kohen
- (22)da temiz bolsın. ואם [14:21] Da egär yarlı esä ol da kuvatı yetmäz esä da alsın bir
- (23)koy ašamya sallamakka kapara etmä üstünä da onda bir payī özäk karīšīlyan
- (24)yay bilan minhaya da seyik yay. ושׁתוֹי [14:22] Da eki kumrular ya eki balaların kögürčinnin
- (25)negä ki yetsä kuvatï da bolsïn birsi hatat da ol birsi 'ola. והביא
- (26)[14:23] Da getirsin alarni ol sekizinži gündä temiz bolyanina ol kohengä ešiginä

$106 v^{0}$

- ohel mo'ednin aldına YWY-nın. ולקח [14:24] Da alsın ol kohen ol asam koyunı da ol (1)
- (2) seyik yayni da sallasin alarni ol kohen sallamak aldina YWY-nin. ושׁהט
- (3) [14:25] Da soysin ol ašam koyuni da alsin ol kohen kanindan ol ašamnin da versin
- yimšayī üstünä on kulayīnīn ol temiz bolyannīn da baš barmayī (4)
- üstünä oŋ kolunïŋ da baš b[a]rmayï²⁰⁷ üstünä oŋ ayayïnïŋ. (5)
- ומן [14:26] Da ol yaydan koysin ol kohen ol sol avuču üstünä (özünin) ol kohennin. (6)
- (7) והזה [14:27] Da sačsin ol kohen ol on kolunin barmayi bilan ol yaydan ki ol sol avuču
- (8) üstünä yedi kerätlär aldına YWY-nın, ונתן [14:28] Da versin ol kohen ol yaydan
- (9) ki avuču üstünä yimšayi üstünä on kulayinin ol temiz bolyannin da baš
- (10)barmayî üstünä on kolunin da baš barmayî üstünä on ayayinin orni üstünä
- (11)kanının ol asamnın. והנותר [14:29] Da ol kalyanı ol yaydan ki avuču üstünä
- (12)ol kohennin versin bašī üstünä ol temiz bolyannin kapara etmä üstünä
- (13)aldına YWY-nın. ועשה [14:30] Da kılsın birni ol torlardan ya balalarından ol

²⁰⁷ JSul.IV.02A: bermayï; probably a scribal error. | BSMS 288: barmayï. | JSul.III.01: barmayï. | ADub.III.73: barmayï.

- (14) kögürčünnin nedän ki yetsä kuvati. את [14:31] Negä ki yetsä kuvati ol birni hatat
- (15) da ol birni 'ola ol minha üstünä da kapara etsin ol kohen ol temiz bolyan
- (16) üstünä aldına YWY-nın. ואת [14:32] Budır torası kimnin ki anda hastalıyı ğüzam
- (17) derdliknin da kimnin ki yetmäsä kuvatï temiz bolyanïnda. וידבר [14:33] Da sözlädi
- (18) YWY Mošegä da Aharonya demä. כי [14:34] Ki gelsäŋiz yerinä Kenaʿannïŋ ki men veräydirmin
- (19) sizgä tutuvlikka da versäm hastalivin žuzam derdliknin evindä yerinin tutuvluvinizni[n]²⁰⁸.
- (20) ובא [14:35] Da gelsin kim ki anın ol ev da anlatsın ol kohenga dema hastalık
- (21) gibi göründi mana evdä. וצוה [14:36] Da sïmarlasïn ol kohen da bošatsïnlar ol evni
- (22) kelmäsindän burun ol kohen görmägä ol hastalikni ki mundar bolmayay ğümlä ki
- (23) ol evdä da andan sonra gelsin ol kohen görmägä ol evni. וראה [14:37] Da görsin
- (24) ol hastalïknï da muna ol hastalïk duvarlarïnda ol evnin yol yol yäšillär
- (25) ya kirmizilar da görümläri alčak ol duvardan. ויצא [14:38] Da čiksin ol kohen ol
- (26) evdän ešiginä ol evnin da bekläsin ol evni yedi günlär. ושׁב [14:39] Da kaytsin

107 ro

- (1) ol kohen ol yedinži gündä da görsä da muna yayildi ol hastalik duvarlarinda
- (2) ol evnin. וצוה [14:40] Da sïmarlasïn ol kohen da suvursïnlar ol tašlarnï ki alarda
- ol hastalik da tašlasinlar alarni šahardan čihari mundar yergä. ואת [14:41] Da ol
- (4) evni ķīrsīn ičkārtin čūpčūvrā da töksinlār ol topraķnī ki ķīrdīlar
- (5) šaḥardan čiḥari mundar yerge. ולקחו [14:42] Da alsinlar öŋgä tašlar da ketirsinlär
- (6) ornîna ol tašlarnîn da öngä toprak alsîn da sîlasîn ol evni. ואם [14:43] Da egär
- (7) kaytsa ol hastalik da bitsä ol evdä suvuryanindan sonra ol tašlarni
- (8) da ķiryanindan sonra ol evni da silayanindan sonra. ובא [14:44] Da gelsä ol kohen
- (9) da görsä da muna yayildi ol hastalik ol evdä sizlatiji juzam derdlikdir
- (10) ol (ol) evdä mundardir ol. ונתץ [14:45] Da yiksin ol evni tašlarini da ol
- (11) ayačlarini da žümlä toprayin ol evnin da čiyarsin šahardan čihari mundar
- (12) yergä. והבא [14:46] Da ol kelgän ol evgä ǯümlä beklägän günlärdä anï
- (13) mundar bolsin ol ahšamya degin. והשוכב [14:47] Da ol yatkan ol evdä yuvsin
- (14) upraklarini da ol ašayan ol evdä yuvsin upraklarini. ואם [14:48] Da egär gelmä

²⁰⁸ JSul.IV.02A: tutuvluyiŋizniz; a scribal error. | BSMS 288: tutuvliyiŋizniŋ. | JSul.III.01: tutuvluyunuznun. | ADub.III.73: tutuvluk yeriyiznin.

250

- (15) gelsä ol kohen da görsä da muna yayilmadi ol hastalik ol evdä silayanindan
- (16) sonra ol evni da temiz etsin ol kohen ol evni zira onaldi ol hastalik.
- (17) ולקח [14:49] Da alsïn sačmaya ol evni eki kušlar da sälvi ayačï da kirmizï²⁰⁹ iräŋkli
- (18) yipek da ezöv. ושׁהט [14:50] Da soysïn ol bir kušnï čöräp sayïtka tatlï suvlar
- (19) üstünä. ולקח [14:51] Da alsïn ol sälvi ayačïnï da ol äzövni da ol kïrmïzï
- (20) irankli yipekni da ol yaban kušni da mančsin alarni kanina ol soyulyan kušnin
- (21) da ol tatlī suvlarya da sačsīn ol evgā yedi kerātlār. ווזטא [14:52] Da sačsīn
- (22) ol evni ķanī bilān ol ķušnīŋ da ol tatlī suvlar bilān da ol yaban ķušu
- (23) bilän da ol sälvi ayači bilän da ezöv bilän da ol ķirmizi iränkli yipek bilän.
- (24) ושלח [14:53] Da yibirsin ol yaban kušnï šaḥardan čïḥarï yüzü üstünä ol düznin
- (25) da kapara etsin ol ev učun da temiz bolsïn. זאת [14:54] Budïr ol tora här
- (26) hastaliyina ol žüzam derdliknin da netekkä. ולצרעת [14:55] Da žüzam derdliginä ol

107 vo

- upraķnīŋ da evgä. ולשאת [14:56] Da šiši[k]kä²¹⁰ da oyuzya da lekägä. להורות [14:57] Ögrätmägä
- (2) ol mundar vaķītnī da ol temiz vaķītnī budīr torasī ol ǯüzam derdliknin.

Chapter 15

- (3) (טו [15:1] Da sözlädi YWY Mošegä da Aharonya demä. דברו [15:2] Sözlägin
- (4) oylanlarına Yısra'elnin da aytınız alarya kiši kiši ki bolsa akkan
- (5) etindän zavliyi mundardir ol. וזאת [15:3] Da bu ola mundarliyi zavliyindan
- (6) suvan akti isä ayipli etindän akmayi ya koyu akip kapadi isä ayip
- (7) etini aķmanindan mundardir ol. כל [15:4] Žümlä ol töšäk ki yatsa üstünä
- (8) ol zav mundar bolsin da ğümlä ol sayit ki otursa üstünä mundar
- (9) bolsïn. ואיש [15:5] Da bašï ki tiysä töšäginä yuvsïn upraklarïnï da yuvunsïn
- (10) suvlar bilän da mundar bolsïn ol aḥšamya degin. והיושב [15:6] Da ol oturyan
- (11) ol sayıt üstünä ki otursa üstünä ol zav yuvsın uprakları [n] i²¹¹
- (12) da yuvunsin suvlar bilän da mundar bolsin ol ahšamya degin. והנוגע [15:7] Da ol
- (13) tiygän etinä ol zavnin yuvsin upraklarini da yuvunsin suvlar bilän da mundar

²⁰⁹ JSul.IV.02A: kirimzi; probably a scribal error. | BSMS 288: kirmizi. | JSul.III.01: kirmizi. | ADub.III.73: kirmizi.

20

²¹⁰ JSul.IV.02A: *šišikkä*; a scribal error. | BSMS 288: *šišikkä*. | JSul.III.01: *sisikke*. | ADub.III.73: *šišiknin*.

²¹¹ JSul.IV.02A: *upraklarini*; a scribal error. | BSMS 288: *upraklarin*. | JSul.III.01: *upraklarin*. | ADub.III.73: *upraklarin*.

- (14) bolsïn ol aḥšamya degin. וכי [15:8] Da ki tükürsä ol zav temizgä da yuvsïn
- (15) upraķlarīnī da yuvunsīn suvlar bilān da mundar bolsīn ol aḥšamya degin.
- (16) וכל [15:9] Da ǯümlä ol binek ki atlansa üstünä ol zav mundar bolsïn. וכל
- (17) [15:10] Da ǯümlä ol tiygän ǯümlägä ki bolsa tibinä mundar bolsïn ol aḫšam-
- (18) ya degin da ol kötärgän alarni yuvsin upraklarini da yuvunsin suvlar bilän
- (19) da mundar bolsin ol aḥšamya degin. וכל [15:11] Da ǯümlä ki tiysä aŋar ol zav
- (20) da gendisin yaykam[a]dï²¹² suvlar bilän da yuvsïn upraklarïn da yuvunsïn suvlar bilän
- (21) da mundar bolsïn ol aḥšamya degin. וכלי [15:12] Da čöräp sayït ki tiysä aŋar ol
- (22) zav sïndïrïlsïn da ǯümlä ayač sayït yaykalsïn suvlar bilän. וכי [15:13] Da egär
- (23) temiz kalirsa ol zav akmasından da saysın özünä yedi günlär temizligi učun
- (24) da yuvsïn upraklarini da yuvsïn etini tatli suvlar bilan da temiz bolsïn. וביום
- (25) [15:14] Da ol sekizinži gündä alsïn özünä eki kumrular ya eki balalarïn gögürčünnin
- (26) da gelsin aldïna YWY-nïn ešiginä ohel moʻednin da versin alarnï ol kohengä.

- (1) רעשה [15:15] Da kilsin alarni ol kohen birni hatat da ol birni 'ola da kapara
- (2) etsin üstünä ol kohen aldïna YWY-nïŋ zavlïyïndan. ואיש [15:16] Da kiši ki
- (3) čīķsa andan tökmägi urluķnīŋ da yuvsīn suvlar bilān ǯümlä eti[n]i²¹³ da mundar
- (4) bolsïn ol ahšamya degin. וכל [15:17] Da žümlä uprak da žümlä teri ki bolsa
- (5) üstünä tökmägi urluknin da yuvulsin suvlar bilän da mundar bolsin ol ahšam-
- (6) ya degin. ואשה [15:18] Da hatin ki yatsa kiši ani tökmägin urluknin da yuvunsinlar
- (7) suvlar bilän da mundar bolsïnlar ol ahšamya degin. ואשה [15:19] Da hatin ki
- (8) bolsa zava kan bolsa akkanî etindân yedi günlâr bolsîn yîraklîyî učun
- (9) da ǯümlä ol tiygän anar mundar bolsïn ol ahšamya degin. וכל [15:20] Da her
- (10) ne ki yatsa üstünä nidaliyinda mundar bolsin da her ne ki otursa
- (11) üstünä mundar bolsin. וכל [15:21] Da ǯümlä ol tiygän töšäginä yuvsin
- (12) upraķlarīnī da yuvunsīn suvlar bilān da mundar bolsīn ol aḥšamya degin.
- (13) וכל [15:22] Da ǯümlä ol tiygän här dürli sayı̈tka ki otursa üstünä yuvsı̈n
- (14) upraklarini da yuvunsin suvlar bilan da mundar bolsin ol ahšamya degin. ואם
- (15) [15:23] Da egär ol töšäk üstünä esä ol ya ol sayit üstünä ki ol
- (16) oturaydir üstünä tiygänindä anar mundar bolsin ol ahšamya degin. ואם

-

²¹² JSul.IV.02A: *yaykamdi*; a scribal error. | BSMS 288: *yaykamadi*. | JSul.III.01: *tastirmasa*. | ADub.III.73: *yuymasa*.

²¹³ JSul.IV.02A: *etini*; a scribal error. | BSMS 288: *etin*. | JSul.III.01: *gufun*. | ADub.III.73: *gufun*.

- (17) [15:24] Da egär yatma yatsa kiši anï da bolsa nidalïyï üstünä da mundar bolsïn yedi
- (18) günlär da ǯümlä ol töšäk ki yatsa üstünä mundar bolsïn. ואשה
- (19) [15:25] Da hatin ki aksa akkani kaninin čok günlär dügül vahti bilän nidaliyinin
- (20) ya ki aksa zeyada nidalïyï vakïtï üstünä ğümlä gü[n]lärindä²¹⁴ akkanïnïn
- (21) mundarlïyïnïŋ günläri gibi nidalïyïnïŋ bolsïn mundardïr ol. כל [15:26] Šümlä
- (22) ol töšäk ki yatsa üstünä ğümlä günlärindä zavlïyïnïn töšägi gibi nida-
- (23) lïyïnïŋ bolsïn aŋar da ǯümlä ol sayït ki otursa üstünä mundar bolsïn
- (24) mundarlïyï gibi nidalïyïnïŋ. ופל [15:27] Da ǯümlä ol tiygän alarya mundar bolsïn
- (25) da yuvsin upraķlarini da yuvunsin suvlar bilan da mundar bolsin ol aḫšamya
- (26) degin. ואם [15:28] Da egär temiz kalïrsa akmanından da saysın özünä yedi günlär da andan

$108 v^0$

- (1) soŋra temiz bolsïn. וביום [15:29] Da ol sekizinǯi gündä alsïn özünä eki
- (2) kumrular ya eki balalarin kögürčinnin da getirsin alarni ol kohengä ešiginä
- (3) ohel moʻednin. ועשה [15:30] Da kilsin ol kohen ol birni hatat da ol birni
- (4) 'ola da kapara etsin üstünä ol kohen aldına YWY-nın zavlıyından mundarlıyının.
- (5) והזרתם [15:31] Da ayïrïnïz oylanlarïn Yisra'elnin mundarlïklarïndan da ölmäsinlär
- (6) mundarlıkları bilan mundar etkanlarında miskanımnı ki ortalarında. זאת
- (7) [15:32] Budir torasi ol zavnin da kim ki čiksa andan tökmägi urluknin mundar
- (8) bolma anın bilan. והרוח [15:33] Da ol nida yı̈raklı̈yı̈nda da ol zav zavlı̈yı̈nda erkak[k]e²¹⁵
- (9) da tišigā da kišigā ki yatsa mundar hatīn bilān.

Chapter 16

פרשת אחרי מות²¹⁶ן (10)

יוי נְיָדַבֵּר יוי אֶל מֹשֶׁה אַחֲרֵי מות שְׁנֵי בְּנֵי אַחֲרן בְּקַרְבָתָם לְפְנֵי יוי נִיִדַבֵר יוי אָל מֹשֶׁה אַחֲרֵי מות שְׁנֵי בִּנִי אַחָרן בְּקַרְבָתָם לְפְנֵי יוי

נימְתָּוּ²¹⁷ן נימְתָוּ (12...)

(...12) (יי) אַבֶּר (יוֹ [16:1] Da sözlädi YWY Mošegä ölgänindän sonra eki oylanlarï

(13) Aharonnin yuvuklašķanlarinda aldina YWY-nin da öldilär. ויאמר [16:2] Da ayttī

²¹⁴ JSul.IV.02A: güzlärindä; a scribal error. | BSMS 288: künlärindä. | JSul.III.01: kinlerinde. | ADub.III.73: künlarinda.

²¹⁵ JSul.IV.02A: *erkäke*; an orthographical tendency (see 2.1.1.2.2.).

²¹⁶ The weekly Torah portion: Parashot Acharei Mot (Lev 16:1–18:30).

²¹⁷ Lev 16:1 in Biblical Hebrew.

- (14) YWY Mošegä sözlägin Aharon ķardašīņa da girmäsin här vahtta ol ķodešgä
- (15) ičkärtin ol perdägä aldïna ol kapaknïn ki ol sanduk üstünä da ölmäsin
- (16) ki bulut bilän aškara bolïrmïn ol kapak üstünä. בזאת [16:3] Munïŋ bilän
- (17) girsin Aharon ol kodešgä buya kani bilan balasi siyirnin hatatka da kočkar
- (18) 'olaya. כתפת [16:4] Kisäyi kodeš gölmägi giysin da kisäyi könčäklär bolsïnlar eti
- (19) üstünä da kisäyi ïnčkir bilän baylansin da kisäyi sarik bilän sarilsin kodeš
- (20) upraklaridir alar da yuvsin suvlar bilan etini da giysin alarni. ומאת [16:5] Da katindan
- (21) ǯama ʿatīnīŋ oylanlarīnīŋ Yisra elniŋ alsīn eki ulaķlarīn ečkilärniŋ ḥatatķa
- (22) da bir kočķar 'olaya. והקריב [16:6] Da yuvuklaštīrsīn Aharon ol ḥatat buyasīnī ki
- (23) anïŋ da kapara etsin gendi učun da evi učun. ולקח [16:7] Da alsïn eki ol ulaklarnï
- (24) da turyuzsin alarni aldina YWY-nin ešiginä ohel mo'ednin. ונתן [16:8] Da versin

- (1) Aharon eki ol ulaklar üstünä čeklär bir ček YWY-ya da bir ček 'Aza' zel-
- (2) gä. והקריב [16:9] Da yuvuklaštīrsīn Aharon ol ulaķnī ki čīķtī üstūnā ol ček
- (3) YWY-ya da kilsin ani hatat. והשעיר [16:10] Da ol ulak ki čikti üstünä ol ček 'Aza'zel-
- (4) gä turyuzulsin diri aldina YWY-nin kapara etmä üstünä yibirmägä ani
- (5) 'Aza'zelgä ol yabanya. והקריב [16:11] Da yuvuklaštirsin Aharon ol hatat buyasini
- (6) ki anın da kapara etsin gendi učun da evi učun da soysın ol hatat buyasını
- (7) ki anı̈n, ולקח [16:12] Da alsı̈n dolu ol mahtanı̈ köz otlar üstündän ol mizbeah-
- (8) nïn aldïndan YWY-nïn da dolu avučlarïnï tütsüsin otyamlarnïn inǯä
- (9) da girgizsin ičkärtin ol perdägä. ונהן [16:13] Da versin ol tütsini ol ot üstünä
- (10) aldına YWY-nın da yapsın bulutı ol tütsünin ol kapaknı ki ol sahadatlık
- (11) üstünä da ölmäsin. ולקה [16:14] Da alsın kanından ol buyanın da sačsın barmayı
- (12) bilän aldina ol kapaknin gün doyušina da aldina ol kapaknin sačsin yedi
- (13) kerätlär ol kandan barmayi bilän. ישׁחט [16:15] Da soysin ulayin ol hatatnin ki
- (14) ulusnın da girgizsin kanı[n]i²¹⁸ ičkärtin ol pärdägä da kılsın kanı[n]i²¹⁹ nečiki kıldı
- (15) ķanīna ol buyanīn da sačsīn anī ol ķapaķ üstūnā da aldīna ol ķapaķnīn.
- (16) וכבר [16:16] Da kapara etsin ol kodeš učun mundarliklarindan oylanlarinin Yisra'el-
- (17) nin da tanmaklardan ğümlä yazıklarına da alay kilsin ohel mo'edgä ol
- (18) tohtayan birgälärinä ortasinda mundarliklarinin. וכל [16:17] Da hič adam bolmasin

²¹⁹ JSul.IV.02A: kanini; a scribal error. | BSMS 288: kanin. | JSul.III.01: kanina. | ADub.III.73: kanina.

²¹⁸ JSul.IV.02A: kanini; a scribal error. | BSMS 288: kanin. | JSul.III.01: kanin. | ADub.III.73: kanin.

- (19) ohel mo'eddä girdigindä kapara etmägä kodešdä čikkanina degin da kapara
- (20) etsin gendi učun da evi učun da žümlä kahali učun Yisra'elnin. ויצא
- (21) [16:18] Da čiķsin ol mizbeaḥķa ki aldīna YWY-niŋ da kapara etsin aniŋ učun
- (22) da alsïn kanindan ol buyanin da kanindan ol ulaknin da versin bučkaklari üstünä
- (23) ol mizbeaḥnïŋ čüpčüvrä. והזה [16:19] Da sačsïn üstünä ol kandan barmayï
- (24) bilän yedi kerätlär da temiz etsin anï da ayruḥsï etsin anï mundarlïklarindan
- (25) oylanlarinin Yisra'elnin. וכלה [16:20] Da bittirsin kapara etmäktän ol kodešni
- (26) da ohel mo'edni da ol mizbeaḥnï da yuvuklaštīrsīn ol tiri ulaķnï. וסמך

109 vo

- (1) [16:21] Da tayasın Aharon eki kolları [n] z²²⁰ bası üstünä ol tiri ulaknın da ikrar etsin
- (2) üstünä ğümlä yazıklarını da versin alarnı bası üstünä ol ulaknın
- (3) da yibirsin kolu bilän hadir kišinin ol yabanya. נושא [16:22] Da kötärsin ol ulak
- (4) özi üstünä ğümlä günählärini veran yergä da yibirsin ol ulakni yabanya. וכא
- (5) [16:23] Da gelsin Aharon ohel moʻedgä da češsin ol kisäyi upraklarnï ki giydi girdigindä
- (6) ol kodešgä da koysin alarni anda. ורחץ [16:24] Da yuvsin etini suvlar bilän
- (7) ayruhsï yerdä da giysin upraklarïnï da čiksïn da kilsin 'olasini da 'olasin
- (8) ol ulusnïn da kapara etsin gendi učun da ol ulus učun. ואת [16:25] Da yayïn
- (9) ol hatatnin tütätsin ol mizbeahta. והמשלה [16:26] Da ol yibirgän ol ulakni
- (10) 'Aza'zelgä yuvsin upraklarini da yuvsin etini suvlar bilan da andan sonra
- (11) girsin ol avulya. ואת [16:27] Da ol ḥatat buyasini da ol ḥatat
- (12) ulayını ki girgizildi kanları kapara etmägä kodešdä čiyarsın avuldan čiharı
- (13) da küydürsinlär otka terilärini da etlärini da tezäklerini. והשרף [16:28] Da ol
- (14) küydürgän alarni yuvsin upraklarini da yuvsin etini suvlar bilän da andan sonra
- (15) gelsin ol avulya. והיחה [16:29] Da bolsïn sizgä räsiminä dunyanïn ol yedinǯi
- (16) ayda onunda ol aynın kıynanız ğanlarınıznı da hič iš kılmanız ol yerli
- (17) da ol yarip ol dirilgän ortanïzda. כי [16:30] Ki ušbu gündä kapara etär
- (18) üstünizgä temiz etmägä sizni ǯümlä yazïklarïnizdan aldina YWY-nin temiz
- (19) bolunız. שבת [16:31] Šabat šabatondir ol sizgä da kiynanız ǯanları̈nıznı räsimi
- (20) dunyanïn. וכפר [16:32] Da kapara etsin ol kohen ki silsä anï da ki doldursa kolunï

-

²²⁰ JSul.IV.02A: kollarin; a scribal error. | BSMS 288: kollarin. | JSul.III.01: kollarin. | ADub.III.73: kollarin.

- (21) kohenlik etmägä atasïnïn yerine da giysin ol kisäyi upraklarnï²²¹ ol kodeš
- (22) upraķlarini. וכפר [16:33] Da kapara etsin miķdašin ol ķodešnin da ohel mo'ed-
- (23) ni da ol mizbeaḥnï kapara etsin da ol kohenlär učun da ǯümlä ulusï
- (24) učin ol ķahalniņ kapara etsin. והיתה [16:34] Da bolsin bu sizgā räsiminä
- (25) dunyanın kapara etmä oylanları učin Yisra'elnin ğümlä yazıklarından bir
- (26) kerät yilda da kildi nečiki simarladi YWY Mošegä.

Chapter 17

- (1) (יידבר (יז) בר [17:1] Da sözlädi YWY Mošegä demä. דבר [17:2] Sözlägin Aharonya
- (2) da oylanlarina da ğümlä oylanlarina Yisra'elnin da aytkin alarya
- (3) budir ol söz ki simarladi YWY demä. איש [17:3] Kiši kiši žama 'atindan Yisra 'el-
- (4) niŋ egär soysa ögüz ya koy ya ečki avulda ya ki soysa avuldan čïḥarī.
- (5) און [17:4] Da ešiginä ohel moʻednin getirmäsä ani yuvuklaštirmaya korban YWY-
- (6) ya aldına miškanının YWY-nın kan sayılır ol kisigä kan tökti
- (7) da kesilsin ol kiši ortasïndan uluslarïnïŋ. למען [17:5] Anïŋ učun ki ketirgäylär
- (8) oylanlari Yisra'elnin korbanlarini (šehitalarini²²²) ki alar korban (šehita²²³) etäydirlär
- (9) yüzü üstünä ol tüznin da getirsinlär alarnï YWY-ya ešiginä ohel mo'ed-
- (10) nin ol kohengä da korban etkäylär alarnï korbanlarïn šelamimlärnin YWY-ya.
- (11) וזרק [17:6] Da sačsin ol kohen ol kanni mizbeah üstünä YWY-nin ešigindä ohel
- (12) mo'ednin da tütätsin ol yayni kokusina kabulliknin YWY-ya. ולא [17:7] Da korban
- (13) etmäsinlär artik korbanlarini šaytanlarya (₁ulak sifatlarina¹²²⁴) ki alar azaydirlar
- (14) ardlarından ömürlik rasim bolsın bu alarya davürlarına. ואלהם [17:8] Da alarya
- (15) aytkïn kiši kiši ǯama ʿatïndan Yisra ʾelnin ya ol yariptän ki dirilsä
- (16) ort[a]ları̈nda²²⁵ egär čı̈yarsa 'ola ya korban. ואל [17:9] Da ešiginä ohel mo'ed-
- (17) nin ketirmäsä anï kilmaya ani YWY-ya da kesilsin ol kiši uluslarindan.
- (18) איש [17:10] Da kiši kiši žama atindan Yisra elnin ya ol yariptän ol tirilgän

²²³ JSul.IV.02A: Unvocalized text.

²²¹ JSul.IV.02A: *upraķlaran*ï; a scribal error. | BSMS 288: *upraķlarn*ï. | JSul.III.01: *upraķlarn*ï. | ADub.III.73: *upraķlarn*ï.

²²² JSul.IV.02A: Unvocalized text.

²²⁴ JSul.IV.02A: Unvocalized text.

²²⁵ JSul.IV.02A: *ortlarinda*; a scribal error. | BSMS 288: *ortalarinda*. | JSul.III.01: *ortalarinda*. | ADub.III.73: *ortalarinda*.

- (19) ortalarında egär asasa hic kan da veri[r]min²²⁶ hisimimni ol kannı asayan ğanya
- (20) da kesärmin anï ortasïndan ulusïnïn. כי [17:11] Zira ǯanï ol etnin ol kan ičindä-
- (21) dir ol da men verdim anï sizgä ol mizbeaḥ üstünä kapara etmägä ǯanlarïŋïz
- učun ki ol kan ol žan učun kapara etär. על [17:12] Anïŋ učun ayttïm
- (23) oylanlarïna Yisra'elnin hič žan sizdän ašamasïn kan da ol yarip ol tirilgän
- (24) ortanızda asamasın kan. ואיש [17:13] Da kisi kisi oylanlarından Yısra'elnin da ol
- (25) yariptän ol tirilgän ortalarinda ki avlasa avun kiyiknin ya kušnin ki ašalir
- (26) da töksä kanini yapsin ani toprak bilän. כי [17:14] Zira ǯani ǯümlä tännin

110 v°

- (1) kanî ğanî bilân (karîšîk) -dir²²⁷ ol da ayttîm oylanlarîna Yisra'elnin hič
- (2) tenniŋ kanïnï ašamaŋïz zira ǯanï här tenniŋ kanïdïr ol ǯümlä ašavčïlarï
- (3) kesilsin. וכל [17:15] Da här ǯan ki (yaŋïlïp²28) ašasa nevelanï ya terefanï geräk yerlidän
- (4) geräk geriptän da yuvsïn upraklarïnï da yuvsïn etini suvlar bilän da mundar
- (5) bolsïn ol aḥšamya degin da temiz bolsïn. ואם [17:16] Da egär upraklarï[n]ï²²⁹ yuvmasa
- (6) da etin yuvmasa günähini čekär.

Chapter 18

- (7) (יודבר (יחי) וידבר [18:1] Da sözlädi YWY Mošegä demä. דבר [18:2] Sözlägin oylanlarina
- (8) Yisra'elnin da aytkin alarya menmin YWY Tenriniz. כמעשה [18:3] İši
- (9) gibi Mïsïr yerinin ki oturdunïz anda kilmanïz da iši gibi Kena'an yerinin
- (10) ki men getiräydirmin sizni oraya kilmaniz da kanunlari bilan gezmäniz. את
- (11) [18:4] Šara atlarımnı kılınız da rasımlarımnı saklanız gezmaga alar bilan menmin
- (12) YWY Tenriniz. ושמרתם [18:5] Da saklanız räsimlärimni da šara atlarımnı ki kilsa
- (13) alarnï ol adam da tirilir alar bilän menmin YWY. איש [18:6] Kiši kiši hič
- עעעעיina täninin yuvuklašmaniz aškärtmägä 'ayip menmin YWY. ערות
- (15) [18:7] 'Ayïbïn atannın ya'ne 'ayıbın anannın ačmayın anandır ol ačmayın 'ayıbını.
- ערות (16) ערות [18:8] 'Ayïbïn hatïnïnïn atannïn ačmayïn 'ayïbï atannïndïr ol. ערות

-

²²⁶ JSul.IV.02A: *verimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: *berirmin*. | JSul.III.01: *berimen*. | ADub.III.73: *berirmen*.

²²⁷ JSul.IV.02A: The copula -dir belongs to the postposition bilän although it appears after the parenthesis.

²²⁸ JSul.IV.02A: Unvocalized text.

²²⁹ JSul.IV.02A: *upraklarini*; a scribal error. | JSul.III.01: *upraklarin*. | ADub.III.73: *upraklarni*.

- (17) [18:9] 'Ayïbïn kiz kardašinnin kizi atannin ya kizi anannin geräk doγγani evnin (atandan²³⁰)
- (18) geräk doyyanï čïḥarïnïŋ (yayrïsïndan²³¹) ačmayïn ʿayïplarï[n]ï²³². ערוּת [18:10] ʿAyïbïn
- (19) ķīzīnīŋ oylunnīn ya ķīzīnīn ķīzīnnīn ačmayīn 'ayīplarīnī zira 'ayībīndīr
- (20) alar. ערות [18:11] 'Ayı̈bı̈n k̞ı̈zı̈nı̈n hatı̈nı̈nı̈n atannı̈n doyurtk̞anı̈ atannı̈n sayı̈lı̈r
- (21) ķīz ķardašīŋdīr ol ačmayīn 'ayībīnī. ערות [18:12] 'Ayībīn ķīz ķardašīnīŋ atannīŋ
- (22) ačmayın zira yuvuyı atannındır ol. ערות [18:13] 'Ayıbın kız kardasının anannın
- (23) ačmayın zira yuvuyı anannındır ol. ערות [18:14] 'Ayıbın kardasının atannın
- (24) ačmayin (ya'ni²³³) hatinina yuvuklašmayin yengäčändir ol. ערות [18:15] 'Ayibin
- (25) kelininnin ačmayin hatini oylunnindir ol ačmaniz 'ayibin. ערות [18:16] 'Ayibin
- (26) hatininin kardašinnin ačmayin 'ayibi kardašinnindir ol. ערות [18:17] 'Ayibin hatinnin
- (27) da ķīzīnīŋ ačmayīn ne ķīzīn oylunīŋ ne ķīzīn ķīzīnīŋ almayīn ačmaya 'ayībīnī yuvuķtīr

111 rº

- (1) alar iränčiliktir ol (almaķ)²³⁴ ואשה [18:18] Da hatīn ķīz ķardašī üstūnā
- (2) almayin kündäš idmägä anar sayliyinda ačmaya ayipi[n]i²³⁵ anin aldina.
- (3) אוֹן [18:19] Da hatinya nidaliyinda mundarliyinin yuvuklašmayin ačmaya 'ayibini.
- (4) אוֹ [18:20] Da ḫatïnïna dostuŋnïŋ (k̞aršïdakiŋniŋ²³⁶ saɣlïɣïnda) vermägin yatuvuŋnï urluk
- (5) vermägä mundar bolmaya anin bilän. ומזרעך [18:21] Da urluyindan vermägin atešdän
- (6) kečirmägä Molehkä da yänil etmägin Tenrinnin adini menmin YWY.
- (7) את [18:22] Da erkäk bilän yatmayïn yatuvun hatïnnïn ïkrahliktïr ol. ובכל
- (8) [18:23] Da hič tuvarya vermägin yatuvuŋnï mundar bolmaya anïŋ bilän da ḫatïn turmasïn
- (9) aldïna tuvarnin košulmaya anar ikrahliktir ol. אל [18:24] Mundar bolmaniz
- (10) ǯümlä bular bilän zira ǯümlä bular bilän mundar boldïlar ol uluslar
- (11) ki men süräydirmin aldığızdan. ותטמא [18:25] Da mundar boldığı ol yer da sayındım
- (12) günähini üstünä da kustu ol yer oturïvčïlarïnï. ושמרתם [18:26] Da saklanïz siz

²³¹ JSul.IV.02A: Unvocalized text.

²³⁴ JSul.IV.02A: Unvocalized text.

²³⁰ JSul.IV.02A: Unvocalized text.

²³² JSul.IV.02A: 'ayiplariŋi; a scribal error. | BSMS 288: ayiplarin. | JSul.III.01: ayiplerin. | ADub.III.73: ayiplarin.

²³³ JSul.IV.02A: Unvocalized text.

²³⁵ JSul.IV.02A: 'ayiplariŋi', a scribal error. | BSMS 288: ayibin. | JSul.III.01: ayibin. | ADub.III.73: ayipin.

²³⁶ JSul.IV.02A: Unvocalized text.

- (13) räsimlärimni da šara atlarimni da ķilmaniz žümlä ušbu iķrahliklardan ol
- (14) yerli da ol yarip ol tirilgän ortanizda. כי [18:27] Zira ǯümlä ušbu ïķraḥlïķnï
- (15) ķīldīlar kišilāri ol yernin ki sizdān burun da mundar boldī ol yer. ולא [18:28]
- (16) Da kusmasin ol yer sizni da mundar etkäninizdä ani nečiki kustu ol
- (17) ulusnï ki sizdän burun. ²³⁷כי [18:29] Ki saḥi här kimki (gizli²³⁸) k̞ïlma ǯümlä ušbu
- (18) ikrahliklardan kesilirlär ol kilyan žanlar ortasindan uluslarinin. ושמרתם [18:30]
- (19) Da saklanız saklovumnı kilmamaya kanunlarından ol ikrahlıklarnın ki kilindilar
- (20) aldığızya da mundar bolmanız alar bilan menmin YWY Tenriniz.

- 239רשת קדושים (21) פרשת קדושים (21) (21) (22...) (22...)
- (...22) [19:1] Da sözlädi YWY Mošegä demä.
- (23) דבר [19:2] Sözlägin ǯümlä ǯama ʿatīna oylanlarīnnī Yisra ʾelnin
- (24) da aytkin alarya ayruhsilar boluniz ki ayruhsidirmen YWY Tenriniz. איש

111 vo

- (1) [19:3] Här kiši anasından da atasından korkunız da sabatlarımnı saklanız menmin YWY
- (2) Teŋriŋiz. אל [19:4] Kayïrïlmanïz ol yoklïklarya da tökmä abak kïlmaŋïz özüŋizgä
- (3) menmin YWY Teŋriŋiz. וכל [19:5] Da egär korban etsäŋiz šelamim korbanïnï YWY-ya
- (4) moradiniz učun korban etiniz ani. ביום [19:6] Korban etkän gününizdä ašalsin
- (5) da sabaḥindan da ol kalyan ol üčünži güngä degin otka küydürülsin. ואם
- (6) [19:7] Da egär ašalma ašalsa ol üčünži gündä bayatdir (pasuldir²⁴¹) ol šelamim
- (7) kabul bolmastir. ואוכליו [19:8] Da (bilip²⁴²) ašavčilari günähini čekär zira
- (8) kodešin YWY-nïŋ yäŋil etti da kesilir ol ǯan uluslarïndan. ובקצרכם
- (9) [19:9] Da oryanınızda orayın yeriniznin tavusmayın küyürin tarluvunnun ormaya da başayın

²³⁹ The weekly Torah portion: Parashot Kedoshim (Lev 19:1–20:27).

²⁴¹ JSul.IV.02A: Unvocalized text.

²³⁷ JSul.IV.02A: deest. | BSMS 288: כי . | JSul.III.01: כי . | ADub.III.73: כי

²³⁸ JSul.IV.02A: Unvocalized text.

²⁴⁰ Lev 19:1 in Biblical Hebrew.

²⁴² JSul.IV.02A: Unvocalized text.

- (10) orayınının basaklamayın. ופרמך [19:10] Da borlalıyınınının četimlämägin da danäsin borlalıyınının
- (11) čöplämägin yarlïya da yaripkä kemiškin alarnï menmin YWY Teŋriŋiz. אֹ [19:11] Hïrsïzlïk
- (12) etmäniz da tanmaniz da aldamaniz kimsä karšidakisini. ולא [19:12] Da ant etmäniz
- (13) adim bilän yalyanya da yenil etmägin adin Tenrinnin menmin YWY. לא [19:13] Zülümlämägin
- (14) dostunni da dutup almayin konmasin yanina ḥaki iryatinnin ertägä degin. אל
- (15) [19:14] Ķaryamayīn sayīrnī da aldīna soķurnīŋ vermāgin sürünmāk da ķorķķīn Tenrindān menmin
- (16) YWY. אָל [19:15] Ķīlmaŋïz ķīŋɣïrlīķ šara atta yüz etmägin yüzlärin yarlïnïŋ
- (17) da siylamayin yüzlärin zenginnin doyrulik bilän šara'at etkin dostuna. אל
- (18) [19:16] Yürümäniz kambaz ulusinda turmayin kani üstünä dostunnin menmin YWY.
- (19) אָל [19:17] Dušman tutmayin kardašinni gönlindä ügütlämä ügütlägin dostunni
- (20) ki čekmägäysin onïn učun ǯürüm. אֹ [19:18] Öč almayïn da kin kovmayïn oylanlarïna
- ulusïŋnïŋ da sevgin arkardašïŋnï gendiŋni gibi menmin YWY. אדק [19:19] Resimlärimni
- (22) saklanız tuvarını košmayın eki žins tarlovunnı sačmayın eki žins da uprak
- (23) eki žins ša atnez čikmasin üstü[ŋ]ä²⁴³. ואיש [19:20] Da kiši ki yatsa hatinni
- (24) tökmägin urluknin da ol karavaš olup keläšingän bašina da yulunma yulunmadi
- (25) ya azatlik [f]ihati²⁴⁴ verilmedi anar täftiš bolsin ölmäsinlär egär azat
- (26) bolmadī isä. והכיא [19:21] Da ketirsin pešmanlīyīnī YWY-ya ešiginā ohel mo'ed-

- (1) nin kočķar ašamya. וכפר [19:22] Da kapara etsin anin učun ol kohen kočķari
- (2) bilän ol ašamnin aldina YWY-nin yaziyi učun ki yazikli boldi da bošatilir
- (3) aŋar yazïyïndan ki yazïklï boldï. וכי [19:23] Egär gelsäŋiz ol yergä da tiksäŋiz här
- (4) dürli yemiš ayači da aklafli sayiniz aklafini ya'ni yemišini üč yillar bolsin
- (5) sizgä 'arellär gibi ašalmasïn. ובישנה [19:24] Da ol dördünži yïlda bolsïn
- (6) žümlä yemiši kodeš mahtovlar YWY-ya. ובשנה [19:25] Da ol bešinži yilda

260

²⁴³ JSul.IV.02A: *üstünä*; probably a scribal error. | BSMS 288: *üstüŋä*. | JSul.III.01: *istine*. | ADub.III.73: *üstüya*. ²⁴⁴ JSul.IV.02A: *kihatï*; a scribal error.

- (7) ašanjiz yemišini arttirmaya sizgä ma[ḫ]sulunii²⁴⁵ menmin YWY Tenriniiz. אל [19:26] Ašamanjiz
- (8) ol kan bilän kušnašlīk etmäniz da bulutka bakmanīz. אֹ [19:27] Kuršalamanīz
- (9) et[raf]in²⁴⁶ bašiŋizniŋ da čaypamaŋiz kenarindaki sakaliŋniŋ. ושרט [19:28] Da yirtmak ölü
- (10) učun vermäniz etinizdä da tamyali yazi vermäniz özünizdä menmin YWY. אל
- (11) [19:29] Yeŋil etmägin kiziŋni azdirmaya ani da azmasin ol yer ḥalki da tolmasin ol
- yer zinaliķ. את [19:30] Šabatlarimni saklaniz da mikdašimdan korkuniz menmin YWY.
- (13) אל [19:31] Kayïrïlmanïz ol kamčïlarya da ol bildiǯilärgä izlämäniz mundar bolmaya
- (14) alar bilän menmin YWY Tenriniz. מכני [19:32] Aldïndan pirnin turyïn da sïylayïn
- (15) yüzlärin kartnın da korkkın Tenrindan menmin YWY. וכי [19:33] Da egär tirilsä
- (16) birgänä yarip yerinizdä kïynamanïz anï. כאזרה [19:34] Yerli gibi sizdän bolsïn
- (17) sizgä ol yarip ol tirilgän birgänizgä da sevgin anï gendinni gibi zira
- (18) yariplär ediniz Mïsïr yerindä menmin YWY Tenriniz. אֹ [19:35] Kïlmanïz kïnyïrlïk
- (19) šara atda ölčädä miskalda täräzidä. מאזני [19:36] Doyru teräzi doyru taš
- (20) doyru kebič da doyru seyik bolsîn sizgä menmin YWY Tenriniz ki čîyardîm
- (21) sizni Mïsir yerindän. ושמרתם [19:37] Da saklanız ğümlä räsimlärimni da ğümlä
- (22) šara atlarimni da ķiliņiz alarni menmin YWY.

- (23) (בר (ב) [20:1] Da sözlädi YWY Mošegä demä. וידבר [20:2] Da oylanlarïna Yisra'el-
- (24) niŋ aytkin kiši kiši oylanlarından Yısra elnin da ol yarıptan ol
- (25) tirilgän Yisra'el da ki versä urluyindan Molehke ölmä öldürülsin
- (26) ulusï ol yernin töpäläsinlär anï taš bilän. ואני [20:3] Da men veri[r]min²⁴⁷ hišimimnï

112 vo

- (1) ol (gizli vergän¹²⁴⁸) kišidä da kesärmin anï ortasindan ulusinin ki urluyindan
- (2) verdi Molehke mundar etmäk učun mikdašimni da yenil etmägä ayruhsi adimni.

²⁴⁵ JSul.IV.02A: *maksuluni*; a scribal error. | BSMS 288: *taḥīlīn*. | JSul.III.01: *bitisin anīn*. | ADub.III.73: *bitišin anīn*

²⁴⁶ JSul.IV.02A: *etarfin*; a scribal error. | BSMS 288: *kïyïrïn*. | JSul.III.01: *kïrïyïn*. | ADub.III.73: *kïrïyïn*.

²⁴⁷ JSul.IV.02A: *verimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: *berirmin*. | JSul.III.01: *berimen*. | ADub.III.73: *berirmen*.

²⁴⁸ JSul.IV.02A: Unvocalized text.

- (3) ואם [20:4] Da egär örtmä örtsälär ulusï ol yernin gözlärini ol kišidän ačïktan
- vergänindä urluyïndan Moleḫkä öldürmämäk[k]ä²⁴⁹ anï. ושמתי [20:5] Da koyarmïn men
- (5) hı̈sımımnı ol kisiga da mispahasına da kesarmin anı da ğumla ol azyanlarnı
- (6) artindan azmaya ardindan ol Molehnin ortasindan uluslarinin. והנפש
- (7) [20:6] Da ol ǯan ki kayïrïlsa ol kamčïlarya da ol bildäǯilärgä azma artlarïndan
- (8) da veri[r]min²⁵⁰ hı̈sımı̈mnı̈ ol ǯanya da kesarmin anı̈ ortası̈ndan ulusı̈nı̈n. והתקדשׁתם
- (9) [20:7] Da ayruhsï boluŋïz da boluŋïz ayruhsïlar ki menmin YWY Teŋriŋiz. ושמרתם
- (10) [20:8] Da saklanız räsimlärimni da kılınız alarnı menmin YWY ayruhsı etivci sizni.
- (11) כי [20:9] Ki kiši kiši ki ķaryasa atasïnï ya anasïnï ölmä öldürülsin
- (12) atasın ya anasın karyadı kanları basına ola. ואיש [20:10] Da kisi ki no'eflik
- (13) etsä hatini bilan kisinin ki no eflik etsä hatini bilan arkadasinin
- (14) ölmä üldürülsin²⁵¹ ol no'eflik etkän kiši da ol no'eflik etkän hatin.
- (15) ואיש [20:11] Da kiši ki yatsa hatini bilan atasinin 'ayibin atasinin ačti ölmä
- (16) öldürülsinlär ekisi kanlari bašlarina ola. ואישׁ [20:12] Da kiši ki yatsa
- (17) kelini bilän ölmä öldürülsinlär ekisi ikrahlik kildilar kanlari bašlarina
- (18) ola. ואוש [20:13] Da kiši ki yatsa erkäk bilän yatuvun hatinnin ikrahlik
- (19) ķīldīlar ekisi ölmä öldürülsinlär ķanlarī bašlarīna. ואיש [20:14] Da kiši
- (20) ki alsa hatinni da anasini irancilikdir ol otka küydürsinlar ani da birni
- (21) alardan da bolmasin iränčilik ortanizda. ואיש [20:15] Da kiši ki versä yatuvun
- (22) tuvarya ölmä öldürülsin da ol tuvarnï da öldürüniz. ואשה [20:16] Da hatin ki
- (23) yuvuklašsa žümlä tuvarnin birinä dört ayakla[r]²⁵² yatmaya ani da öldürgin ol
- (24) hatïnnï da ol tuvarnï ölmä öldürülsinlär kanlarï bašlarïna. ואישׁ [20:17] Da kiši
- (25) ki alsa ķīz ķardašīnī ķīzīn atasīnīŋ ya ķīzīn anasīnīŋ da görsā 'ayībīnī da ol
- (26) görsä anin 'ayibini iränčiliktir ol da kesilsinlär gözlärinčä oylanlarinin

113 rº

- (1) uluslarinin 'ayibin kiz kardašinin ačti günähini čeksin. ואיש [20:18] Da kiši
- (2) ki yatsa nida hatinni da ačsa 'ayibini čoyrayini ačti da ol ačti kanlarinin

²⁴⁹ JSul.IV.02A: *öldürmämäkä*; an orthographical tendency (see 2.1.1.2.2.).

²⁵⁰ JSul.IV.02A: *verimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: *berirmin*. | JSul.III.01: *berimen*. | ADub.III.73: *berirmen*.

²⁵¹ JSul.IV.02A: *üldürüläsin*; a scribal error. | BSMS 288: *öldürülsin*. | JSul.III.01: *eltirilsin*. | ADub.III.73: *öltürülsün*.

²⁵² JSul.IV.02A: *ayaklap*; probably a scribal error.

- (3) čoyrayını da kesilsinlar ekisi ortasından uluslarının. וערות [20:19] Da 'ayıbın kız
- (4) ķardašīnīŋ [anaŋnīŋ da ķīz ķardašīnīŋ]²⁵³ ataŋnīŋ ačmaɣīn kim ki yuvuɣīnīŋ ayībīnī ačtī günäḥlärin čeksinlär.
- (5) ואיש [20:20] Da kiši ki yatsa yengäčäsini ayačasïnïŋ 'ayïbïnï ačtï yazïklarïn čeksinlär
- (6) maḥrimlar ölsinlär. ואיש [20:21] Da kiši ki alsa kardašinin hatinini nida gibi-
- (7) dir ol kardašinin 'ayibini ačti mahrimlar bolsinlar. ושמרתם [20:22] Da saklaniz
- (8) ğümlä räsimlärimni da ğümlä šara atlarımnı da kılınız alarnı da kusmasın
- (9) sizni ol yer ki men getiräydirmin sizni or[a]ya²⁵⁴ oturmaya anda. ולא [20:23]
- (10) Da yürümäniz räsimläri bilän ol ulusnin ki men süräydirmin aldinizdan zira
- (11) jümlä bularnï kildilar da bezdim alardan. ואמר [20:24] Da ayttim sizgä siz
- (12) meräsläniz yerlärini da men veräyim anï sizgä meräslämägä anï yer ayayturyan süt da bal
- (13) menmin YWY Tenriniz ki ayirdim sizni ol uluslardan. והבדלתם [20:25] Da ayiriniz
- (14) arasina ol helal tuvarnin mundar tuvarya da arasina ol
- (15) mundar kušnin helalya da iranči etmaniz žanlarinizni tuvar bilan da kuš bilan
- (16) ya ğümlä bilän ki kï[y]mïldar²⁵⁵ ol yer üstünä ki ayïrdïm sizgä mundar
- (17) etmägä. והייתם [20:26] Da bolunïz mana ayruhsïlar ki ayruhsïdïrmïn YWY
- (18) da ayirdim sizni ol uluslardan bolmaya menim. ואיש [20:27] Da kiši ya hatin ki
- (19) bolsa alarda ķamčī ya bildiži ölmä öldürülsinlär taš bilän tašlasïnlar
- (20) alarnï kanlarï bašlarïna.

- 256 רשת קדושים (21)
- (22) ניאמֶר יוָי אֶל מֹשֶׁה אֱמוֹר אֶל הַכֹּהָנִים בְּנֵי אהָרן וְאָמַרְתָּ אֲלֵהֶם (22)
 - 257 לְנָפָשׁ לֹא יָטְמא כָּעְמיו (23)
- (23) [21:1] Da ayttï YWY Mošegä aytkin
- (24) ol kohenlärgä oylanlarina Aharonnin da aytkin alarya ölügä mundar bolmasin

113 vo

263

²⁵³ JSul.IV.02A: deest, added on the basis of BSMS 288: anaŋnïŋ da kiz kardašinïŋ. | JSul.III.01: anannïn da tuvduyunun. | ADub.III.73: anaynïn da tuvduyunun.

²⁵⁴ JSul.IV.02A: *orya*; a scribal error. | BSMS 288: *anda*. | JSul.III.01: *ari*. | ADub.III.73: *arī*.

²⁵⁵ JSul.IV.02A: kürmüldar; probably a scribal error. | BSMS 288: küymüldar. | JSul.III.01: tebrenedi. | ADub.III.73: tebranadi

²⁵⁶ The weekly Torah portion: Parashot Emor (Lev 21:1–24:23).

²⁵⁷ Lev 21:1 in Biblical Hebrew.

- (1) uluslarinda. כי [21:2] Ki ančak yakinina ol yuvuk anar anasina ya da atasina
- ya oylina ya ķizina ya ķardašina. ולאחוחו [21:3] Da ķiz ķardašina ol boy ķiz
- ol yuvuk aŋar ki bolmadï koǯaya aŋar tiyip mundar bolsïn. לא [21:4]
- (4) Mundar bolmasin koža (tiyip hatinina) uluslarinda yenil bolmaya ol. אל
- (5) [21:5] Yurķmasïnlar yur[k]maķ²⁵⁸ bašlarïnda da kenarïndaki saķallarïnïŋ tïraš idmäsinlär
- (6) da etlärindä čizmasinlar čizmaķ. קדושים [21:6] Ayruḥsilar bolsinlar Tenrilärinä
- (7) da yeŋil etmäsinlär adin Teŋriläriniŋ zira otlu korbanlarin YWY-niŋ korbanin
- (8) Tenrilärinin alar yuvuklaštiraydirlar da bolsinlar ayruhsi. אשה [21:7] Zona hatinni
- (9) ne bozuk yolluni almasinlar ne sürülgän hatinni erindän almasinlar zira
- (10) ayruhsidir ol Tenrisinä. וקדשתו [21:8] Da ayruhsi etkin ani zira korbanin
- (11) Tenrinnin ol yuvuklaštīraydīr ayruhsī bolsīn sana ki ayruhsīdīrmen YWY
- (12) ayruhsï etivči sizni. ובת [21:9] Da kizi kohen kišinin egär bašlasa azmaya
- (13) atasinin kohenligini ol pasul etäydir otka küydürülsin. והכהן [21:10] Da ol
- (14) böyük kohen kardašlarından ki kuyulsa bası üstünä ol silmak yayı ve doldursa
- (15) koluni giymägä ol kodeš upraklarini bašini ačmasin da upraklarini yirtmasin.
- (16) ועל [21:11] Da hič öli žanya gelmäsin dahi atasina ne anasina mundar bolmasin.
- (17) ומן [21:12] Da ol mikdašdan čikmasin da yenil etmäsin mikdašin Tenrisinin zira
- (18) tažī silmāk yayīnīŋ Teŋrisiniŋ üstünādir menmin YWY. והוא [21:13] Da ol ḥatīnnī
- (19) boylıkları bilan alsın. אלמנה [21:14] Tulnı ne sürülgannı ne bozuk yollunı ne zonanı
- (20) bularnî almasîn ki ančak boy kîznî uluslarîndan alsîn özünä hatînya. ולא [21:15]
- (21) Da yenil etmäsin urluyini uluslarinda ki menmin YWY ayruhsi etivči ani.
- (22) דידבר [21:16] Da sözlädi YWY Mošegä demä. דבר [21:17] Sözlägin Aharonya demä kiši
- (23) urluyindan dävürlärinä ki bolsa anda hillä yuvuklašmasin yuvuklaštirma korbanin
- (24) Tenrisinin. כי [21:18] Ki ğümlä kiši ki anda hillä yuvuklašmasïn sokur kiši
- (25) ne topal ne eksik buvumli ne artik buvumli. או [21:19] Ya kiši ki bolsa
- (26) anda sınık ayak ya čolak kol. או [21:20] Ya kambur ya inặä (bodir) ya ak

(1) tüškän gözinä ya koturli ya egri buvumli ya tešen taša[k̩]li¹²⁵⁹. כל [21:21] **Ž**ümlä

²⁵⁸ JSul.IV.02A: *yurmak*; a scribal error. | BSMS 288: *yulkmak*. | JSul.III.01: *yulkuv*. | ADub.III.73: *yulkuv*. | ²⁵⁹ JSul.IV.02A: Illegible text; reconstructed on the basis of another printed copy of the edition: *täšäkli*. | BSMS 288: *tešan*. | JSul.III.01: *sisken ayipli*. | ADub.III.73: *šiškan ayipli*.

- (2) kiši ki anda ḥillä urluyïndan Aharon ol kohennin yuvuklašmasïn yuvuklaštïrmaya
- (3) otlu ķorbanlarin YWY-nin hillä isä anda ķorbanin Tenrisinin yuvuklašmasin
- (4) yuvuklaštīrmaya. לחם [21:22] Ķorbanīn Teŋrisiniŋ kodešlärindän ol kodeslärniŋ
- (5) lakin ol kodešlärdän ašasïn. אַן [21:23] Tek ol pärdägä gelmäsin da ol mizbeaḥ-
- (6) ķa yuvuķlašmasīn zira hillā v[a]r anda¹²⁶⁰ da yenil etmäsin miķdašlarīmnī ki
- (7) menmin YWY ayruḥsï etivči alarnï. וידבר [21:24] Da sözlädi Moše Aharonya
- (8) da oylanlarina da ğümlä oylanlarina Yisra'elnin.

- (9) (בב [22:1] Da sözlädi YWY Mošegä demä. דבר [22:2] Sözlägin Aharonya
- (10) da oylanlarina da ayirilsinlar kodešlärindän oylanlarinin Yisra'elnin da yenil
- (11) etmäsinlär ayruhsi adimni ki alar kodeš etäydirlär mana menmin YWY. אמר
- (12) [22:3] Aytķīn alarya dävürlärinizgā žümlā kiši ki yuvuķlašsa žümlā urluyunīzdan
- (13) ol kodešlärgä ki kodeš etärlär oylanlarï Yisra'elnin YWY-ya da mundarlïyï
- (14) bolsa üstünä da kesilir ol ǯan aldïmdan menmin YWY. איש [22:4] Kiši
- (15) kiši urluyindan Aharonnin da ol žüzam derdli ya zav kodešlärdän ašamasin
- (16) deginčä ki temiz bolïr da ol tiygän ǯümlä mundar ǯanlïya ya kišigä ki
- (17) čiķsa andan tökmägi urluķniņ. אל [22:5] Ya kišigä ki tiysä ǯümlä
- (18) kozlayanya ki mundar bolir anin bilan ya adamya ki mundar bolir anin bilan
- (19) ǯümlä mundarlïyïna. נפשׁ [22:6] ǯan ki tiysä anar da mundar bolsïn ol ahšam-
- (20) ya degin da ašamasin ol kodešlärdän ki ančak yuvsa etini suvlar bilän.
- (21) ובא [22:7] Da batsa ol kuyaš da temiz olir da andan soŋra ašasin ol kodešlärdän
- (22) zira ašīdīr ol. נבלה [22:8] Nevelanī ne terefa ašamasīn mundar bolmaya
- (23) anin bilän menmin YWY. ושמרו [22:9] Da saklasinlar saklovumni da čekmäsinlär anin
- (24) učun günäh ğürümüni da ölärlär anın bilän egär yenil etsälär anı menmin YWY
- (25) ayruhsï etivči alarnï. וכל [22:10] Da hič yabanžï ašamasïn kodešni müsäfiri kohennin
- (26) ne ïryatï ašamasïn kodešni. וכהן [22:11] Da kohen egär satïn alsa ǯan satïn

114 v^o

- (1) alyanın ahcasının ol asasın andan da doyyanı evinin alar asasınlar asından.
- (2) ובת [22:12] Da kizi kohennin ki bolsa yat kišigä ol ayirmayindan ol kodeš-

²⁶⁰ JSul.IV.02A: *hillä ver anda*; a scribal error. | BSMS 288: *ayïp anda*. | JSul.III.01: *ayip anda*. | ADub.III.73: *ayip anda*.

- (3) lärnin ašamasin. ובת [22:13] Da ķizi kohennin ki bolsa tul ya sürülgän da urluķ
- (4) bolmasa anar da kaytsa evinä atasinin yašliklari gibi ötmägindän atasinin
- (5) ašasïn da hič yabanži ašamasïn andan. ואישׁ [22:14] Da kiši ki ašasa kodešni yanlišlik
- (6) bilän da arttirsin bešinžisin üstünä da versin kohengä ol kodešni.
- (7) און [22:15] Da yenil etmäsinlär kodešlärin oylanlarinin Yisra'elnin neni ki
- (8) ayırsalar YWY-ya. והשיאו [22:16] Da čektärtsinlär alarya pešmanlık ğürümüni
- (9) ašayanlarinda kodešlärindän ki menmin YWY ayruhsi etivči alarni.
- (10) דבר [22:17] Da sözlädi YWY Mošegä demä. דבר [22:18] Sözlägin Aharonya da oylanlarïna
- (11) da ğümlä oylanlarina Yisra'elnin da aytkin alarya kiši kiši ğama'atından
- (12) Yisra'elniŋ da ol γariptän Yisra'eldä ki yuvuklaštïrsa korbanïnï [ǯ]ümlä²⁶¹ inyätlärini da ǯümlä
- (13) jomartlïklarini ki yuvuklaštirsalar YWY-ya ʻolaya. לרצונכם [22:19] Kiläginiz
- (14) bilän saylam erkäkni sïyïrdan koylardan ya da ečkilärdän. כל [22:20] Šümlä ki
- (15) anda hille yuvuklaštirmaniz zira kabullikka bolmaz sizgä. ואיש
- (16) [22:21] Da kiši ki yuvuklaštīrsa šelamim ķorbanīnī YWY-ya ayīrmaya inyāt ya
- (17) jomartliķķa siyirdan ya ķoydan saylam bolsin ķabulliķķa hič hille
- (18) bolmasın anda. עורת [22:22] Sokurnı ne kiriknı ne sakatnı ne sinirlini ne
- (19) oyuzluni ne egri buvumlini yuvuklaštirmaniz bularni YWY-ya da otlu korban
- (20) vermäniz alardan ol mizbeah üstünä YWY-ya. ושור [22:23] Da ögüz ya koy artikli ya
- (21) eksikli žomartlik kilyin ani da inyätkä kabul bolmastir. ומעוך
- (22) [22:24] Da ezilgänni ne yančilyanni nä üzülgänni ne bičilgänni yuvuklaštirmaniz YWY-ya
- (23) da yerinizdä ķilmaniz. ומיד [22:25] Da ķolundan yat ulusnin yuvuklaštirmaniz ķorbanin
- (24) Tenriniznin ğümlä bulardan zira ki čaypalmakları alarda hille alarda kabul
- (25) bolmaslardir sizgä. וידבר [22:26] Da sözlädi YWY Mošegä demä. שור [22:27] Ögüz
- (26) ya koy ya ečki ki doysa da bolsin yedi günlär anasinin yaninda da ol
- (27) sekizinži gündän da ötegä kabul bolir korbanya otlu korban YWY-ya. ושור

115 rº

(1) [22:28] Da ögüz ya koy anï da balasïnï soymanïz bir gündä. וכי [22:29] Da egär korban

(2) etsäŋiz šükürlik korbanïnï YWY-ya muradïŋïz učun korban etiŋiz. ביום [22:30] Ol

-

²⁶¹ JSul.IV.02A: gümlä; a scribal error. | BSMS 288: barča. | JSul.III.01: bar. | ADub.III.73: bar.

- (3) gündä ašalsin kaldirmaniz andan ertägä degin menmin YWY. ושמרתם
- (4) [22:31] Da saklanız mičvalarımnı da kiliniz alarnı menmin YWY. ולא [22:32] Da yenil etmäniz
- (5) ayruhsi adimni da ayruhsi bolirmin ortasinda oylanlarinin Yisra elnin
- (6) menmin YWY ayruhsï etivči sizni. המוציא [22:33] Ol čïyaryan sizni Mïsïr yerindän
- (7) bolma sizgä Tenrigä menmin YWY.

- (8) (בג) וידבר [23:1] Da sözlädi YWY Mošegä demä. דבר [23:2] Sözlägin oylanlarïna
- (9) Yisra'elnin da aytkin alarya va'däläri YWY-nin ki čaķiryaysiz
- (10) alarnı čakırmakları kodešnin bulardırlar alar va dälärim. ששׁת [23:3] Altı günlär
- (11) ķīlīnsīn iš da ol yedinži gündā šabat šabaton čaķīrmaķ ķodeš hič iš
- (12) ķilmaŋiz šabatdir ol YWY-ya ğümlä oturašlariŋizda. אלה [23:4] Bulardirlar
- va däläri YWY-nin čaķirmaklari kodešnin ki čaķiryaysiz alarni va dälärindä. בחדש
- (14) [23:5] Ol burunyï ayda on dördünǯi günündä ol aynïŋ ol eki aḥšam arasïna
- (15) pesaḥ YWY-ya. ובחמשה [23:6] Da on bešinži günündä ušbu aynïn ḥagï ol
- (16) mačalarnin YWY-ya yedi günlär mačalar ašanız. ביום [23:7] Ol burunyı gündä čakırmak
- (17) kodeš bolsin sizgā hič ķulluķ išini ķilmaņiz. והקרבתם [23:8] Da yuvuķlaštīriņiz
- (18) otlu korban YWY-ya yedi günlär ol yedinži gündä čakirmak kodeš hič
- (19) kulluk išini kilmaniz. וידבר [23:9] Da sözlädi YWY Mošegä demä. דבר [23:10] Sözlägin
- (20) oylanlarına Yisra'elnin da aytkın alarya ki gelsaniz ol yerge ki men veraydirmin
- (21) sizgā da orsaņīz orayīnī da getiriņiz ilk orayīŋīznīŋ kültāsini ol kohengā.
- (22) הניף [23:11] Da sallası̈n ol kültäni önünä YWY-nı̈ŋ moradı̈ŋı̈z učun sabaḥı̈ndan ol šabat-
- (23) nïŋ sallasïn anï ol kohen. ועשׁיהם [23:12] Da kiliŋiz sallayan günüŋizdä ol kültäni
- (24) saylam koy bir yašar 'olaya YWY-ya. ומנחתו [23:13] Da minḥasï eki payï onnïŋ
- (25) özek karišilyan yay bilän otlu korban YWY-ya kokusi kabulliknin da kuymayi
- (26) yanı šarap dördünği payı ol siyiknin. ולחם [23:14] Da ekmäk ne kavurulan arısı ne arpa

$115 v^{o}$

(1) bašī ašamaņīz gendisinā degin ušbu günnin getirgāninizgā degin korbanīn

- (2) Tenriniznin räsimi dunyanin dävürlärinizgä žümlä oturašlarinizda. וספרתם
- (3) [23:15] Da sananız özünizgä sabahından ol šabatnın getirgän gününizdän ol sallamak
- (4) kültäsini yedi šabatlar tamamlar bolsinlar. עד [23:16] Sabaḥina degin ol yedinži
- (5) šabatnin sananiz elli gün da yuvuklaštiriniz yani minha YWY-ya. ממושבותיכם
- (6) [23:17] Oturašlarinizdan getiriniz sallamaķ ötmägini eki eki ülüši onnin özek
- (7) bolsïnlar ḥameč pišsinlär ilklär²⁶² ķorbanï YWY-ya. והקרבתם [23:18] Da yuvuķlaštīrīŋïz
- (8) ol ötmäk bilän yedi saylam koylar birär yašarlar da buna sïyir balasi bir
- (9) da kočķarlar eki bolsinlar 'ola YWY-ya da minhalari da kuymaklari otlu korban
- (10) kokusï kabullïknïŋ YWY-ya. ועשיתם [23:19] Da kiliŋiz ulayin ečkilärniŋ birni hatat-
- (11) ka da eki koylar birär yašarlar šelamim korbanïna. והניף [23:20] Da sallasïn ol kohen
- (12) alarnı ol ilklar ötmagi bilan sallamak önü[n]ä²⁶³ YWY-nın eki koylar bilan
- (13) kodeš bolsïnlar YWY-ya ol kohengä. וקראתם [23:21] Da čaķïrïŋïz gendisindä ušbu
- (14) günnin čaķīrmaķ ķodeš bolsīn sizgā hič ķulluķ išini ķīlmanīz rāsimi
- (15) dunyanın ğümla oturaşlarınızda davurlarınızga. ובקצרכם [23:22] Da oryanınızda
- (16) yäriniznin orayini tavusmaniz kiyirin tarlovunnin oryaninda da bašayin orayinnin
- (17) bašaklamayin yarliya da yaripkä kemiškin alarni menmin YWY Tenriniz. וידבר
- (18) [23:23] Da sözlädi YWY Mošegä demä. דבר [23:24] Sözlägin oylanlarïna Yisra'elnin demä
- (19) ol yedinži ayda birindä ol aynïn bolsïn sizgä šabaton sayïnmaklïyï
- (20) ķičķirmaķnin čaķirmaķ ķodeš. כל [23:25] Hič ķulluķ išini ķilmaniz da yuvuķlaštiriniz
- (21) otlu korban YWY-ya. וידבר [23:26] Da sözlädi YWY Mošegä demä. אד
- (22) [23:27] Tek onunda ušbu yedinži aynin ol yulmaklar günüdir ol čaķirmak kodeš
- (23) bolsïn sizgä da kïynanïz ǯanlarïnïznï da yuvuklaštïrïnïz otlu korban YWY-ya.
- (24) וכל [23:28] Da hič iš ķīlmanīz gendisindā ušbu günnin zira günü yulmaklarnīndīr
- (25) ol kapara etmägä üstünizgä aldïna YWY Tenriniznin. כי [23:29] Ki ǯümlä
- (26) ol ǯan egär kïynalmasa gendisindä ušbu günnin kesilir uluslarindan. וכל

(1) [23:30] Da ğümlä ol ğan ki kïlsa hič iš gendisindä ušbu günnin da kayıp etärmin

(2) ol ǯannï ortasïndan ulusïnïn. כל [23:31] Hič iš ķilmaniz räsimi dunyanin

²⁶² JSul.IV.02A: Spelled אָילִיכָּלָר; a scribal error.

²⁶³ JSul.IV.02A: önüŋä; a scribal error. | BSMS 288: aldina. | JSul.III.01: alninda. | ADub.III.73: alninda.

- (3) dävürlärinizgä ǯümlä oturašlarinizda. שבת [23:32] Šabat šabatondir ol sizgä
- (4) da ķīynanīz ǯanlarīnīznī toķuzīnda ol aynīn aḥšamda aḥšamdan aḥšamya
- (5) degin šabat idiniz šabatinjizni. וירבר [23:33] Da sözlädi YWY Mošegä demä.
- (6) דבר [23:34] Sözlägin oylanlarïna Yisra'elnin demä on bešinži günindä ušbu yedinži
- (7) aynın hagı ol sukalarnın yedi günlar YWY-ya. ביום [23:35] Ol burun gündä
- (8) čaķīrmaķ ķodeš hič ķulluķ išini ķīlmaŋīz. שבעת [23:36] Yedi günlär
- (9) yuvuķlaštīrīŋīz otlu ķorban YWY-ya ol sekizinži gündā čaķīrmaķ ķodeš bolsīn
- (10) sizgä da yuvuklaštïrïŋïz otlu korban YWY-ya tiyilmäktir ol hič kulluk išini
- (11) ķilmaŋiz. אלה [23:37] Bulardirlar vadālāri YWY-niŋ ki čaķiryaysiz alarni čaķirmaķlari
- (12) kodešnin yuvuklaštīrmaya otlu korban YWY-ya 'ola da minha šelamim da kuymaklar
- (13) kemin günnin günündä. מלבד [23:38] Šabatları̈ndan baška YWY-nı̈n bahsı̈sısları̈nı̈zdan
- (14) bašķa ki versāniz YWY-ya. אך [23:39] Tek on bešinži günündā ol yedinži
- (15) aynın toplayanınızda ol yernin mahsulunı hag etiniz hagin YWY-nın yedi
- (16) günlär ol burunži gündä šabaton da ol sekizinži gündä šabaton. ולקחתם
- (17) [23:40] Da aliŋiz özüŋizgä ol burunǯi gündän burun (sukalik²²⁶⁴) yemišin siyli ayačniŋ
- (18) hurmalar yapraklarını da butayın kalın yapraklı teräknin da talların özännin da sevininiz
- (19) aldına YWY-nın yedi günlar. וחגתם [23:41] Da hag etiniz anı hag YWY-ya yedi günlar
- (20) yilda räsimi dunyanin dävürlärinizgä ol yedinji ayda hag etiniz ani.
- (21) בסכות [23:42] Sukalarda oturunïz yedi günlär ǯümlä ol yerli Yisra'eldä
- (22) otursunlar sukalarda. למען [23:43] Anïn učun bilgäylär dävürläriniz ki čalašlarda
- (23) oturyuzdim Yisra'el oylanlarini čiyaryanimda alarni Misir yerindin menmin YWY
- (24) Tenriniz. וידבר [23:44] Da sözlädi Moše vadälärin YWY-nïn Yisra'el oylanlarina.

- (25) (בד) וידבר [24:1] Da sözlädi YWY Mošegä demä. צו [24:2] Sïmarlayïn Yisra'el
- (26) oylanlarina da alsinlar sana temiz zeytün yayi iskän jadan čikma

116 vo

- (1) y[a]rīķķa²⁶⁵ yandīrma čīraķ dayīm. מחוץ [24:3] Čīḥartīn pārdāsinā ol šāḥadātlikniņ
- (2) ohel mo'eddä yaraštirsin ani Aharon ahšamdan ertägä degin aldina YWY-nin

_

²⁶⁴ JSul.IV.02A: Unvocalized text.

²⁶⁵ JSul.IV.02A: yeriķķa; a scribal error. | BSMS 288: yariķķa. | JSul.III.01: yariķliķķa. | ADub.III.73: yariķliķķa.

- (3) dayim räsimi dunyanin dävürlärinizgä. על [24:4] Ol temiz altin šamidan üstünä
- (4) yaraštīrsīn ol čīraķlarnī aldīna YWY-nīn dayīm. ולקחת [24:5] Da alyīn özek
- (5) da piširgin anī on eki ķalīn ekmäklär eki ülüši onnīŋ bolsīn ol bir
- (6) kalin ekmäk. ושמת [24:6] Da koyyin alarni eki yäräštirmäklär alti ol yäräštirmäkni
- (7) ol temiz terapiz üstünä aldïna YWY-nïŋ. ונתח [24:7] Da vergin ol yaraštïrmaķ
- (8) üstünä temiz timyan da bolsin korbanya tütsügä otlu korban YWY-ya. ביום
- (9) [24:8] Ol šabat gündä ol šabat gündä yaraštīrsīn anī aldīna YWY-nīŋ dayīm
- (10) katından oylanlarının Yısra'elnin sartı dunyanın. והיתה [24:9] Da bolsın Aharonya
- (11) da oylanlarina da ašasinlar ani ayruhsi yerdä ki kodeši kodeslärnindir ol
- (12) aŋar otlu korbanlarindan YWY-niŋ resimi dunyaniŋ. איצא [24:10] Da čikti oyli
- (13) Yisra'el hatinnin da ol oyli Misirli kišinin Yisra'el oylanlarinin arasından
- (14) da talaštīlar avulda oylu ol Yisra'el hatīnnīŋ da ol Yisra'el kiši.
- (15) ניקב [24:11] Da belgili etti oylï ol Yisra'el hatïnnïŋ ol šemni da karyadï da getirdilär
- (16) anï Mošegä da atï anasïnïŋ Šelomit kizï Dibriniŋ ševetindän Dannïŋ. ויניחוהו
- (17) [24:12] Da koydilar ani saklovda beyin olmaya alarya buyruyi bilan YWY-nin. וידבר
- (18) [24:13] Da sözlädi YWY Mošegä demä. הוצא [24:14] Čïyaryïn ol karyavčïnï avuldan čïharï
- (19) da tayasınlar ğümlä ol eğitkanlar kollarını bağı üstüna da töpalasınlar anı
- (20) ǯümlä ol ǯamaʿat. אל [24:15] Da oylanlarïna Yisraʾelnin sözlägin demä kiši
- (21) kiši ki karyasa Tenrisini da čeksin günähini. ונקב [24:16] Da karyavčī adīn YWY-nīn
- (22) ölmä öldürülsin taš atma taš atsïnlar üstünä ǯümlä ol ǯama ʿat nečiki yarip
- (23) alay yerli karyayanında šemni öldürülsin. ואיש [24:17] Da kiši ki ursa
- (24) jümlä janin adamnin ölmä öldürülsin. ומכה [24:18] Da uruvči janin tuvarnin
- (25) ödesin anï ǯan ornïna ǯan. ואישׁ [24:19] Da kiši ki versä ḥillä
- (26) dostina nečiki kildi alay kilinsin anar. שבר [24:20] Sinik ornina sinik

- (1) göz ornïna göz diš ornïna diš nečiki versä hillä adamya alay verilsin
- (2) aŋar. ומכה [24:21] Da uruvčī tuvarnī ödäsin anī da uruvčī adamnī öldürülsin.
- (3) משׁפּט [24:22] Bir šara at bolsın sizgä nečiki yarip alay yerli bolsın ki menmin
- (4) YWY Tenriniz. וידבר [24:23] Da sözlädi Moše oylanlarina Yisra'elnin da čiyardilar ol
- (5) karyavčini avuldan čihari da töpelädilär ani taš bilän da oylanlari Yisra'elnin
- (6) ķīldīlar nečiki sīmarladī YWY Mošegā.

- פרשת בהר²⁶⁶ן (7) פרשת בהר²⁶⁷ן (26...) פרשת בהר²⁶⁷ן אַל משָה בָּהַר סִינַי לָאמר
- (...8) [25:1] Da sözlädi YWY Mošegä
- (9) tayında Sinaynın demä. דבר [25:2] Sözlägin oylanlarına Yisra'el-
- (10) niŋ da aytkin alarva ki gelsäŋiz ol yergä ki men ver[ä]ydirmin²⁶⁸ sizgä da šemita
- (11) bolsïn ol yer šemita YWY-ya. พัพ [25:3] Altï yïllar sačķīn tarluvuŋnï da altï
- (12) yïllar butayïn borlalïyïnnï da toplayïn mahsulïnï. ובשנה [25:4] Da ol yedinǯi yïlda
- (13) šabat šabaton bolsïn yergä šemita YWY-ya tarlovunnî sačmayîn da borlaliyinnî
- (14) butamayın. את [25:5] Orayının kiyigini ormayın da daneların borlalarınının
- (15) čöplämägin šemita yïlï bolsïn yergä. והיתה [25:6] Da bolsïn šemitasï ol
- (16) yernin sizgä ašamaya sana da kuluna da karavašina da iryatya da müsäfirgä
- (17) ol tirilgänlär birgänä. ולבהמתך [25:7] Da tuvarïna da kiyik[k]ä²⁶⁹ ki yerindä bolsïn
- (18) jümlä tahili ašamaya. וספרת [25:8] Da sanayin özünä yedi šemitalarin yillarnin
- (19) yedi yillar yedi kerätlär da bolarlar sana günläri yedi šemitalarinin ol yillarnin
- (20) kirk tokuz yilda. והעברת [25:9] Da gečirgin kičkirmak šofar sesini ol yedinži
- (21) ayda onunda ol aynın ol kipurim günündä gečiriniz šofar sesini
- (22) jumla yerinizda. וקדשתם [25:10] Da ayruhsi etiniz ol ellinji yilnin senesini
- (23) da čaķīrīŋīz maʿaflīķ yerde ǯümlä oturīvčīlarīna yoveldir ol bolsīn sizgā

117 vo

- (1) da kaytīnīz här kiši tutuvlīyīna da här kiši mišpahasīna kaytīnīz. יובל [25:11] Yovel-
- (2) dir ol yïlï ol ellinži yïlnïŋ bolsïn sizgä sačmaŋïz da ormaŋïz kiyiklerini
- (3) da čöplämäniz danälärini (ambarlarya²⁷⁰). כי [25:12] Ki yoveldir ol kodeš bolsïn
- (4) sizgä ol tarlovdan alïp ašanïz kiyik mahsulïnï. בשנת [25:13] Yïlïnda
- (5) ušbu yovelnin kaytiniz här kiši tutuvliyina. וכי [25:14] Da egär satsaniz satuv
- (6) dostuna ya satin alma satin alsan kolundan dostunnin kiynamaniz kiši
- (7) kardašīnī. במספר [25:15] Sanī bilān yīllarnīn ol yoveldān sonra satīn alyīn

²⁶⁸ JSul.IV.02A: *veriydirmin*; probably a scribal error. | BSMS 288: *beräydirmen*. | JSul.III.01: *beremen*. | ADub.III.73: *beŕamen*.

²⁶⁶ The weekly Torah portion: Parashot Behar (Lev 25:1–26:2).

²⁶⁷ Lev 25:1 in Biblical Hebrew.

²⁶⁹ JSul.IV.02A: *kiyikä*; an orthographical tendency (see 2.1.1.2.2.).

²⁷⁰ JSul.IV.02A: Unvocalized text.

- (8) (₁yovelgä degin¹²⁷¹) ķatīndan dostuņnīņ sanī bilān maḥsul yïllarīnīņ satsīn saņa.
- (9) לפי [25:16] Čokliyina görä ol yillarnin čok etkin satuvuni da azliyina görä
- (10) ol yïllarnïŋ az etkin satuvunï zira satïn maḥsullarnïŋ ol satïyïr saŋa.
- (11) אוֹ [25:17] Da kiynamaŋiz kiši karšidakini²⁷² da korkkin Teŋriŋden zira menmin YWY
- (12) Teŋriŋiz. ועשיתם [25:18] Da ķīlīŋīz rāsimlārimni da šaraʿatlarīmnī saklaŋīz da ķīlīŋīz
- (13) alarnï da oturïrsïz ol yer üstünä eminlik bilän. ונתנה [25:19] Da verir ol yer
- (14) maḥsulunï da ašarsïz toyyunča da oturursïz eminlik bilän üstünä. וכי
- (15) [25:20] Da egär aytsaŋïz ne ašalïm ol yedinǯi yïlda muna sačmasaķ da toplamasaķ
- (16) taḥi'limi'zni. וצויתי [25:21] Da simarlarmi'n alyı'simni sizgä ol altin'yi yilda
- (17) da ķīlar ol taḥīlnī (mamur²⁷³) üč ol yīllarya. וזרעתם [25:22] Da sačarsīn ol sekizinǯi
- (18) yïlda da ašarsïz ol tahildan eskini ol dokuzinži yïlya degin yeriškinžä
- (19) geleǯäk maḥsulï ašarsïz eskini. והארץ [25:23] Da ol yer satïlmasïn bašbatra
- (20) (ömürlik²⁷⁴) zira menimdir ol yer anğa yariplär gibi da oturašlar gibi (oturïrsïz²⁷⁵)
- (21) sizlär birgämä. ובכל [25:24] Da ğümlä yerindä tutuvluyïŋïznïŋ hilaslik veriniz
- (22) ol yergä. כי [25:25] Ki zügürtläsä kardašïŋ da satsa tutuvluyïndan
- (23) da gelsä hilas etivčisi ol yuvuk anar da hilas idsin satkanin kardašinin. ואיש
- (24) [25:26] Da kiši ki bolmasa anar hilas etivči da yetsä kuvati da yerišsä hilasliyinža.
- (25) וחשב [25:27] Da ḥesab etsin yïllarïn satuvunïn da kaytarsïn ol artkannï ol kišigä ki

- (1) sattï aŋar da kaytsïn tutuvluyïna. ואם [25:28] Da egär yetmäsä kuvatï kaytarmaya aŋar
- (2) da bolsin satuvu kolunda ol satin alivčinin ani ol yovel yili[n]a²⁷⁶ degin
- (3) da čĭksïn yoveldä da kaytsïn tutuvluyïna. מואיני [25:29] Da kiši ki satsa oturaš ev
- (4) kalalī šāḥārdā da bolsīn va dāsi hīlaslīyīnīn tamam bolyanīna degin yīlī satuvunīn
- (5) bir yïl bolsïn va däsi hïlaslïyïnïn ואם [25:30] Da egär hïlas olunmasa dolyïnǯa anar
- (6) tamam yil da mukam bolsin ol ev ki ol šähärdä ki var anar kala bašbatra
- (7) (ömürlik) satın alı̈vëıya anı davurlarına hılas erikmasın yovelda. ובתי [25:31] Da evları
- (8) ol salalarnın ki yoktır alarya kala čüpčüvrä tüzü bilan ol yernin sayılır

²⁷² JSul.IV.02A: kardašīdakini; a scribal error. | BSMS 288: dostun. | JSul.III.01: dostun. | ADub.III.73: dostun.

²⁷¹ JSul.IV.02A: Unvocalized text.

²⁷³ JSul.IV.02A: Unvocalized text.

²⁷⁴ JSul.IV.02A: A partially unvocalized text.

²⁷⁵ JSul.IV.02A: A partially unvocalized text.

²⁷⁶ JSul.IV.02A: yilina; probably a scribal error. | BSMS 288: yilina. | JSul.III.01: yilina. | ADub.III.73: yilina.

- (9) hïlaslïk bolsïn anar da yoveldä hïlas čïksïn. וערי [25:32] Da šähärläri ol Levi-
- (10) lilärnin evläri šähärlärinin tutuvliklarinin ömürlik [h]ilasliyi²⁷⁷ bolsin Levi-
- (11) lilärgä. ואשר [25:33] Da egär hilas idsä birsi ol Levilärdän da hilas čiksin
- (12) satuvu evnin da šäḥäri tutuvluyïnïn yoveldä zira evläri šäḥärlärinin ol Levi-
- (13) lärnin oldir tutuvliklari ortasinda oylanlarinin Yisra'elnin. ושׁדה
- (14) [25:34] Da tarlovu salasïnïŋ šäḥärläriniŋ satïlmasïn ki tutuvluyï dunyanïŋdïr ol alarya.
- (15) וכי [25:35] Da egär zügürtläsä dindaš ķardašīŋ da taysa ķolu birgäŋä da kuvat[l]andïryïn²⁷⁸ anï
- (16) gerek yaripni gerek müsäfirni da gečinsin birgänä. אל [25:36] Almayïn yanïndan fa'iz ne
- (17) mamele fa izni da korkkin Teŋriŋizdan da gečinsin kardašin birgana. את [25:37] Ahčanni
- (18) vermägin anar fa'izgä da faydaya vermägin ašinni. אני [25:38] Menmin YWY Tenriniz
- (19) ki čiyardim sizni Misir yerindan verma sizga Kena an yerini bolma sizga
- (20) Teŋrigä. וכי [25:39] Da egär zügürtläsä dindaš kardašïŋ ki birgäŋä da satïlsa saŋa kullanmayïn
- (21) anï kulluyin kulnïn. כשׁכיר [25:40] Ïryat gibi müsäfir gibi bolsïn birgänä yïlïna
- (22) degin ol yovelnin kulluk etsin birgänä. ויצא [25:41] Da čiksin yanindan ol da oylanlari
- (23) birgäsinä da kaytsïn mišpahasïna da tutuvluyïna atalarïnïn kaytsïn. כי [25:42] Zira
- (24) kullarımdır alar ki ciyardım alarnı Misir yerindan satılmasınlar satılmayın kulnın.
- (25) איז [25:43] Erklänmägin aŋar zor bilän da k̞ork̞k̞[i]n²⁷⁹ Teŋriŋizdän. ועבדך [25:44] Da k̞uluŋ da k̞aravašiŋ

118 vo

- (1) ki bolsalar saŋa yanından ol uluslarnın ki cupcuvralarınızga alardan satın
- (2) aliniz kul da karavaš. וגם [25:45] Da dayin oylanlarindan ol oturašlarnin ol tirilgänlär
- (3) birgänizgä alardan satīn alīnīz da mišpaḥalarīndan ki birgānizgā ki doyursalar
- (4) yerinizdä da bolsinlar sizgä tutuvlikka. והתנחלתם [25:46] Da üliš alip üläširsiz alarni
- (5) oylanlarinizya artinizdan meräslämägä tutuvlik dunyaya degin alarni kullaniniz

²⁷⁷ JSul.IV.02A: *kiläsligi*; probably a scribal error. | BSMS 288: *yulovu*. | JSul.III.01: *yulunmak*. | ADub.III.73: *yulumak*.

²⁷⁸ JSul.IV.02A: *kuvatnandïryïn*; a scribal error. | BSMS 288: *kip tutkïn*. | JSul.III.01: *kiplegin*. | ADub.III.73: *kiplagin*.

²⁷⁹ JSul.IV.02A: korkkan; a scribal error. | BSMS 288: korkkin. | JSul.III.01: korkkun. | ADub.III.73: korkkun.

- (6) da kardašlarinizya oylanlari Yisra'elnin kiši kardašina erklänmägin anar zor bilän.
- (7) וכי [25:47] Da anar yetsä kuvati yaripnin ya oturašnin birgänä da zügürtläsä kardašin ki
- (8) birgäsinä da satilsa yaripkä oturašķa ki birgänä ya goy temelinä mišpaḥasina
- (9) yaripnin. אחרי [25:48] Satïlyanïndan sonra hïlaslïk bolsïn anar birsi kardašlarïndan
- (10) hilas idsin ani. אר [25:49] Ya ayačasi ya oyli ayačasinin hilas idsin ya yuvuyindan
- (11) teninin mišpaḥasindan hilas idsin ani ya yetsä kuvati da özi hilas ulinsin. והשב
- (12) [25:50] Da ḥesap etsin satīn alīvčīsī bilān satīldīyī yīlīndan aŋar yīlīna degin ol yovelniņ
- (13) da bolsïn gümüši satuvunïn sanï bilän yïllarnïn günläri gibi ïryatnïn bolsïn birgäsinä.
- (14) אַם [25:51] Egär dayïn čok kaldï esä yïllardan alarya görä kaytarsïn hilaslïyïnï gümüšindän
- (15) satuvunïŋ. ואם [25:52] Da egär az kaldï esä yïllardan yïlïna degin ol yovelniŋ da hesab
- (16) etsin aŋar yïllarya görä da kaytarsin hïlaslïyïnï. כֹשׁכִיר [25:53] Ïryat gibi yïldan
- (17) yïlya bolsïn birgäsinä erklänmäsin anar zor bilän gözlärinčä. ואם [25:54] Da egär
- (18) hilas olmasa bular bilan da čiķsin yilinda ol yovelnin ol da oylanlari
- (19) birgäsinä. כי [25:55] Zira mana oylanlari Yisra'elnin kullar kullarimdir alar
- (20) ki čïyardïm alarnï Mïsïr yerindän menmin YWY Tänriniz.

- (21) או [26:1] Ķīlmaŋïz özüŋizgä abaklar da pesel da mačeva turɣuzmaŋïz özüŋizgä da bušamak
- (22) taš vermāniz yerinizdā baš urmaya üstūnā ki menmin YWY Tānriniz.
- את [26:2] Šabatlarïmnï saklanïz da mikdašïmdan korkunïz menmin YWY.

119 ro

אם בָּחָקוֹתַי תֶלֶכוּ וָאֶת מְצְוֹתַּי תִּשְׁמִרוּ וַעֲשִׂיתֶם אֹתַם 280 וּ אָת בָּחָקוֹתַי תָלֶכוּ וָאֶת מְצְוֹתַי תִּשְׁמִרוּ וַעֲשִׂיתֶם אֹתַם. (1...)

- (...1) אם [26:3] Egär räsimlärim
- (2) bilän yürüsäŋiz da mičvalarïmnï saklasaŋïz da kilsaŋïz alarnï. ונחתי [26:4] Da veri[r]min²⁸¹
- (3) yamyurlarinizni vahtlarinda da verir ol yer mahsulini da ayači ol tüznin verir yemišini.

²⁸⁰ Lev 26:3 in Biblical Hebrew. It is the beginning of a weekly Torah portion: Parashot Bechukotai (Lev 26:3–27:34)

²⁸¹ JSul.IV.02A: *verimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: *berirmin*. | JSul.III.01: *berimen*. | ADub.III.73: *berirmen*.

- והשיג [26:5] Da yetär sizgä indir basmak bay bozumu[n]a¹²⁸² kadar da bay bozumu (4) yetär urluk
- säčmäk[k]ä²⁸³ kadar da ašarsiz ašinizni tovyunča da oturursiz eminlik bilän yerinizdä. (5)
- ונחתי [26:6] Da veri[r]min²⁸⁴ esänlik ol yerdä da yatarsïz da bolmaz kaltratïvčï da (6) kesärmin
- (7) yaman kiyikni ol yerdän da kilič gečmäz yerinizdä. ורדפתם [26:7] Da kuvarsiz dušmanlarinizni
- da tüšärlär aldınızya kılığ bilan. 1571 [26:8] Da kuvarlar sizdan beš kiši yüzni (8)
- (9) da yüz kiši sizdän tümänni kuvarlar da tüšärlär dušmanlariniz aldinizya kilič
- bilän. ופניתי [26:9] Da kayrïlïrmïn sizgä da yayïldïrï[r]mïn²⁸⁵ sizni da köbäytirmin sizni (10)
- (11)da turyuzurmin šartimni birganizga. ואכלתם [26:10] Da ašarsiz eskini eskirganni
- da eskini aldından yanının cıyarıp tökarsiz. ונחתי [26:11] Da veri[r]min²⁸⁶ miškanımnı (12)ortaŋïzda
- (13)da hor idmäz moradim sizni. והתהלכתי [26:12] Da yürürmin ortanizda da bolurmin
- (14)sizgä Tenrigä da siz bolursïz mana uluska. אני [26:13] Menmin YWY Tenriniz ki
- čiyardim sizni Misir yerindan bolmaktan alarya kullar da sindirdim čomačalarin (15)
- boyunsanïznïn da yürüttim sizni tik boy bilän. ואם [26:14] Da egär ešitmäsäniz mana (16)
- (17)da kïlmasanïz ğümlä ušbu mičvalarnï. ואם [26:15] Da egär räsimlärimni hor etsäniz
- (18)da egär šara atlarimni hor idsä žaniniz kilmamaka žumlä mičvalarimni buzmaya
- (19)siz šartïmnï. หรุ [26:16] Dayin men kilarmin muni sizgä da tayin etärmin üstünizgä
- ķaltramaķ ol suvuķ ḫast[a]liķni²⁸⁷ da ol ķizdīrma ḫast[a]liķni²⁸⁸ tavusīvčilar gözlārni (20)
- da sïzlatïvčïlar ǯannï da sačarsïz boška urluyïnïznï da ašarlar anï dušmanlarïnïz. (21)
- ונחתי [26:17] Da veri[r]min²⁸⁹ hı̈sımı̈mnı̈ sizda da kı̈rılı̈rsı̈z aldı̈na dusmanları̈nı̈znı̈n (22)

²⁸² JSul.IV.02A: bay bozumuna; a scribal error. | BSMS 288: borla čöplämäkkä. | JSul.III.01: ïstiryan birtikke. | ADub.III.73: yïštïrmakka.

²⁸³ JSul.IV.02A: *säčmäkä*; an orthographical tendency (see 2.1.1.2.2.).

²⁸⁴ JSul.IV.02A: *verimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: berirmin. | JSul.III.01: berimen. | ADub.III.73: berirmen.

²⁸⁵ JSul.IV.02A: yayildirimin; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: yayildirirmin. | JSul.III.01: yaydirirmen. | ADub.III.73: yayarmen.

²⁸⁶ JSul.IV.02A: *verimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: berirmin. | JSul.III.01: berimen. | ADub.III.73: berirmen.

²⁸⁷ JSul.IV.02A: *hastliķni*; a scribal error. | BSMS 288: *hastaliķni*. | JSul.III.01: *hastaliķni*. | ADub.III.73: ḥastalïknï.

²⁸⁸ JSul.IV.02A: hastliknii; a scribal error. | BSMS 288: hastaliknii. | JSul.III.01: hastaliknii. | ADub.III.73: ḥastalïknï.

²⁸⁹ JSul.IV.02A: *verimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: berirmin. | JSul.III.01: berirmen. | ADub.III.73: berirmen.

- (23) da erklänirlär sizgä dušmanlarïnïz da kačarsïz da bolmaz kuvuvčï sizni. ואם
- (24) [26:18] Da egär bularva degin ešitmäsäniz mana da arttïrï[r]mïn²⁹⁰ ügütlämä sizni yedi kazev
- yazı̈kları̈nı̈z učun. וֹשׁברתי [26:19] Da sı̈ndı̈rı̈[r]mı̈n²9¹ öktamlığın kuvatı̈nı̈znı̈n da veri[r]min²9²

119 vo

- (1) göklärinizni demir gibi da yerinizni bakir gibi. ותם [26:20] Da bitär bošķa kuvatiniz
- (2) da vermäz yeriniz mahsulini da ayači ol yernin vermäz yemišini. ואם [26:21] Da egär yürüsäniz
- (3) birgämä učur bilän da kilämäsäniz ešitmä mana da arttïrï[r]mïn²⁹³ üstünizgä
- (4) urmak yedi kazev yazıklarınız učun. והשלחתי [26:22] Da [yibi]ri[r]min²⁹⁴ sizgä ol tüznin kiyigini
- (5) da tuv etär sizni da kesär tuvarinizni da az etär sizni da veran bolirlar yollariniz.
- (6) ואם [26:23] Da egär bular bilän ügütlänmäsäŋiz maŋa da yürüsäniz birgämä učur bilän.
- (7) והלכתי [26:24] Da yürürmin dayïn men birgäŋizgä učur bilän da urarmïn sizni dayïn men
- (8) yedi kazev yazıklarınız učun. והבאתי [26:25] Da getiri[r]min²⁹⁵ üstünizgä kılıc öc alıvcı
- (9) öčün šartnin da topla[n]irsiz²⁹⁶ šaḥarlarinizya da yibiri[r]min²⁹⁷ ölät ortanizda da verili[r]siz²⁹⁸
- (10) kolu[n]a²⁹⁹ dušmannin. בישברי [26:26] Sindiryanimda sizgä ötmäk kuvatini da piširirlär
- (11) on hatinlar ekmäginizni bir tandurda da kaytarirlar ašinizni čeki bilan da ašarsiz

²⁹⁰ JSul.IV.02A: *arttïrïmïn*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: *arttïrïrmïn*. | JSul.III.01: *arttïrïrmen*. | ADub.III.73: *arttïrïrmen*.

²⁹¹ JSul.IV.02A: *sindirimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: *sindirirmin*. | JSul.III.01: *sindirimen*. | ADub.III.73: *sindirimen*.

²⁹² JSul.IV.02A: *verimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: *berirmin*. | JSul.III.01: *berimen*. | ADub.III.73: *berirmen*.

²⁹³ JSul.IV.02A: *arttirimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: *arttirirmin*. | JSul.III.01: *arttirirmen*. | ADub.III.73: *arttirirmen*.

²⁹⁴ JSul.IV.02A: *biyirimin*; probably a scribal error. | BSMS 288: *yibirmin*. | JSul.III.01: *iydirirmen*. | ADub.III.73: *iyarmen*.

²⁹⁵ JSul.IV.02A: *getirimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: *ketirirmin*. | JSul.III.01: *keltirimen*. | ADub.III.73: *keltirirmen*.

²⁹⁶ JSul.IV.02A: *toplaŋïrsïz*; probably a scribal error. | BSMS 288: *čöplänirsiz*. | JSul.III.01: *ïstïrïnïsiz*. | ADub.III.73: *yïštïrïlsayïz*.

²⁹⁷ JSul.IV.02A: *yibirimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: *yibirirmin*. | JSul.III.01: *iydirimen*. | ADub.III.73: *iyarmen*.

²⁹⁸ JSul.IV.02A: *verilisiz*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: *berilirsiz*. | JSul.III.01: *berilisiz*. | ADub.III.73: *berilirsiz*.

²⁹⁹ JSul.IV.02A: koluna; a scribal error. | BSMS 288: koluna. | JSul.III.01: koluna. | ADub.III.73: koluna.

- (12) da toymazsïz. ואם [26:27] Da egär munïŋ bilän ešitmäsäŋiz maŋa da yürüsäŋiz birgämä učur
- (13) bilän. והלכתי [26:28] Da yürürmin birgänizgä učur kaheri bilän da ügütlärmin sizni
- (14) dayın men yedi kazev yazıklarınız učun. ואכלתם [26:29] Da ašarsız oylanlarınıznın
- (15) etini da ķizlariŋizniŋ etini ašarsiz. והשמדתי [26:30] Da kayip idarmin bama-lariŋizniï
- (16) da kesärmin kuyaš sifatlarinizni da veri[r]min³⁰⁰ gövdälärinizni gövdäleri üstünä
- (17) iränčilikläriniznin da hor idär kilägim sizni. ונחתי [26:31] Da veri[r]min³⁰¹ šaḥarlarïnïznï veran
- (18) da veran etärmin mikdašlarinizni da kabul etmämdir korbanlariniznin kokusini. והשמתי
- (19) [26:32] Da veran etärmin men ol yerni da maşmunlanırlar üstünä dusmanlarııız ol oturyanlar
- (20) anda. ואתכם [26:33] Da sizni sačarmin uluslarda da suvuri[r]min³⁰² ardiŋizdan kilič
- (21) da bolïr yeriniz veran da šaharlarïnïz bolurlar harab. אז [26:34] Ol vahtta tamam etär
- (22) ol yer šemitalarini ğümlä günlärindä veranliyinin siz yerindä olip dušmanlariniznin
- (23) ol vahtta tïnč bolïr ol yer da tamam etär šemitalarïnï. כל [26:35] Žümlä günlärindä
- (24) veranliyinin tinč bolir neni ki tinč bolmadi šemitalarinizda oturyaninizda
- (25) üstünä. והנשארים [26:36] Da ol kalyanlar sizdä da getiri[r]min³⁰³ yïmšaklïk gönüllärindä

$120 r^{0}$

- (1) yerlärindä dušmanlarinin da kuvar alarni avazi šuvuldayan yapraknin da kačarlar kiličtan
- (2) kačmasī gibi da tüšärlär da bolmaz kuvuvčī. וכשלו [26:37] Da sürünürlär här kiši kardašī
- (3) bilän tutki aldından kiliğnin da kuvuvçi yoktır da bolmaz sizgä turmak aldına

³⁰⁰ JSul.IV.02A: *verimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: *berirmin*. | JSul.III.01: *berimen*. | ADub.III.73: *berirmen*.

³⁰¹ JSul.IV.02A: *verimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: *berirmin*. | JSul.III.01: *berimen*. | ADub.III.73: *berirmen*.

³⁰² JSul.IV.02A: *suvurimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: *suvurimin*. | JSul.III.01: *suvurumen*. | ADub.III.73: *suvurumen*.

³⁰³ JSul.IV.02A: *getirimin*; probably a tendency towards consonant deletion (refer to section 2.2.1.5). | BSMS 288: *ketirirmin*. | JSul.III.01: *keltirimen*. | ADub.III.73: *keltirirmen*.

- (4) dušmanlariniznin. ואבדתם [26:38] Da tas bolirsiz ortasında hanlıklarnın da tavusur sizni yeri dušmanlarınıznın. [26:39] Da ol kalyanlar sizdä čürirlar günähläri bilän
- (5) yerlärindä dušmanlariniznin da dayin günähläri bilän atalarinin birgälärinä čürirlär.
- (6) והתודו [26:40] Da ikrar etirlär günählärini da günähin atalarinin tanmaklari bilän ki
- (7) tandïlar maŋa da dayïn neni ki yürüdilär birgämä učur bilän. אף [26:41] Dayïn men yürürmin
- (8) birgälärinä učur bilän da getiri[r]min³⁰⁵ alarnï yerinä dušmanlarïnïn da ol vahtta sïnïyïr
- (9) ol yabuk gönülläri da ol vahtta bittirirlär günählärinin ğürümüni. וזכרתי
- (10) [26:42] Da sayınırmın sartımnı ki Yaʻakov bilan da dayın sartımnı ki Yichak bilan da dayın
- (11) šartïmnï ki Avraḥam bilän sayïnïrmïn da ol yerni da sayïnïrmïn. והארץ [26:43] Da ol yer
- (12) kemišilir alardan da tamam etär šemitalarini veran bolyaninda alardan da alar
- (13) bittirirlär günählärinin ğürümüni anın učun ki šara atlarımnı hor ettilär
- (14) da anın učun ki räsimlärimni hor itdi ğanları. ואף [26:44] Da dayın bunı kılarmın
- (15) bolyanlarında yerinda duşmanlarının ki hor etmamdir alarnı da iranmamdir alarnı
- (16) tavusmaya alarni buzmaya šartimni birgalarina zira menmin YWY Tenrilari. ווכרתי
- (17) [26:45] Da sayînirmîn alarya šartîn burunyîlarnîn ki čîyardîm alarnî Mîsîr yerindân gözlärinčä
- ol uluslarnın bolmaya alarya Tenrigä menmin YWY. אלה [26:46] Bulardırlar ol räsimlär
- (19) da ol šara atlar da ol toralar ki verdi YWY özünin arasına da arasına oylanlarının
- (20) Yisra'elnin tayında Sinaynın kolu bilan Mošenin.

(21) (בר (בז) אובר [27:1] Da sözlädi YWY Mošegä demä. דבר [27:2] Sözlägin oylanlarïna Yisra'el-

(22) nin da aytķīn alarya kiši ki ayīrsa inyāt ķīyasīn bilān ǯanlar YWY-

³⁰⁴ JSul.IV.02A: deest. Reconstructed on the basis of ADub.III.73: ואבדתם Da taspolursiz ortasünda ḥanliklarnün da tavusur sizni yeri dušmanlariyiznün. | BSMS 288: deest. | JSul.III.01: Da taspolusiz ḥanliklar arasüna da tavusur sizni yeri dusmanlariniznün.

³⁰⁵ JSul.IV.02A: *getirimen*; probably a tendency towards consonant deletion (refer to section 2.2.1.5) or a scribal error. | BSMS 288: *ketirirmin*. | JSul.III.01: *keltirimen*. | ADub.III.73: *keltirirmen*.

- (23) ya. והיה [27:3] Da bolsin kiyasin ol erkäk yegirmi yaštan da altmiš yaška degin
- (24) da bolsin ķiyasin elli miskal gümüš miskali bilan ol kodešnin. ואם [27:4] Da egär
- (25) tiši esä ol da bolsïn ķïyasïŋ otuz meskal. ואם [27:5] Da egär beš yaštan da yigirmi

120 v°

- (1) yäškä degin da bolsïn ķïyasïŋ ol erkäk yigirmi miskal da tišigä on miskal.
- (2) ואם [27:6] Da egär bir aydan da beš yäškä degin da bolsïn ķïyasïŋ ol erkäk beš miskal
- (3) gümüš da tišigä ķïyasïŋ üč miskal gümüš. ואם [27:7] Da egär altmïš yaštan da yoyarï
- egär erkäk isä da bolsïn kïyasïn on beš miskal da tišigä on miskal. ואם [27:8] Da egär
- (5) yarlı esä ol kıyasından da turyuzsın anı aldına ol kohennin da kıyaslasın anı
- (6) ol kohen negä görä ki yetsä kuvati ol inyät etkännin kiyaslasin ani ol kohen.
- (7) ואם [27:9] Da egär tuvar isä ki yuvuklaštirirlar andan korban YWY-ya her ne ki versä
- (8) andan YWY-ya bolsïn kodeš. אל [27:10] Täbdil idmäsin anï da degišmäsin anï
- (9) yaḥšini yamanya ya yamanni yaḥšiya da egär degištirmä degištirsä tuvarni tuvarya
- (10) da bolsin ol da degištirmäsi bolsin kodeš. ואם [27:11] Da egär ǯümlä mundar tuvar isä
- (11) ki yuvuklaštirmaslar andan korban YWY-ya da turyuzsin ol tuvarni aldina ol kohen-
- (12) niŋ. והעריך [27:12] Da kiyaslasin ol kohen ani geräk yaḥšini geräk yamanni kiyasiŋča ey kohen
- (13) alay bolsïn. ואם [27:13] Da egär hilas idmä hilas idsä ani da arttirsin bešinžisini
- (14) ķïyasïŋ üstünä. איש [27:14] Da kiši ki kodeš etsä evini kodeš YWY-ya da kïyaslasïn anï
- ol kohen geräk yaḥšīnī da geräk yamannī ne kadar ki kïyaslasa anī ol kohen alay tursīn. ואמ
- (16) [27:15] Da egär ol kodeš etkän hilas idersä evini da arttirsin bešinži gümüš kiyasin üstünä
- (17) da bolsin anin. ואם [27:16] Da egär tarlovindan tutuvluyinin kodeš etsä kiši YWY-ya
- (18) da bolsin kiyasin urluyina görä urluyi sabunin arpalarnin³⁰⁶ elli miskal kümüškä.
- (19) אם [27:17] Egär yilindan ol yovelnin kodeš etsä tarlovini kiyasinča tursin. ואם
- (20) [27:18] Da egär ol yoveldän sonra kodeš etsä tarlovini da hesap etsin anar ol kohen ol
- (21) gümüšni ol kalyan yïllarya görä yïlïna degin ol yovelnin da eksilsin kïyasïndan.
- (22) ואם [27:19] Da egär hilas idmä hilas idsä ol tarlovni ol kodeš etkän ani da arttirsin

³⁰⁶ JSul.IV.02A: *arparlarniŋ*; probably a scribal error. | BSMS 288: *arpalarniŋ*. | JSul.III.01: *arpalar*. | ADub.III.73: *arpalar*.

- (23) bešinži gümüš ķīyasīŋ üstünä da muķam bolsīn aŋar. ואם [27:20] Da egär ḫīlas idmäsä ol
- (24) tarlovnï da egär satsa ol tarlovnï öŋgä kišigä ḫïlas olmasïn artīķ. והיה
- (25) [27:21] Da bolsïn ol tarlov vaʻdäsi čïkkaninda yoveldä kodeš YWY-ya tarlovu gibi ol herem-

121 r^o

- (1) niŋ kohengä bolsïn tutuvluyï. ואם [27:22] Da egär satïn alan tarlovïnï ki dügül tarluvïndan
- (2) tutuvluyïnïn kodeš etsä YWY-ya. וחשׁב [27:23] Da hesap etsin anar ol
- (3) kohen ķararīn ol ķīyasīŋnīŋ yīlī[n]a³⁰⁷ degin ol yovelniŋ da versin ķīyasīŋnī ol
- (4) gündä kodeš YWY-ya. בשנת [27:24] Yilinda ol yovelnin kaytsin ol tarlov
- (5) aŋar ki satïn aldï anï yanïndan aŋar ki anïŋ tutuvluyï ol yerniŋ.
- (6) בוכל [27:25] Da ğümlä kiyasin bolsin ol kodeš miskali bilän yigirmi
- (7) čegirdäk bolsïn ol miskal. אך [27:26] Tek tuŋyuč ki tuŋyuč bolsa YWY-ya ol
- (8) tuvarda ķodeš etmäsin kiši anī gerāk ögüz gerāk ķoy YWY-nīŋdīr ol.
- (9) ואם [27:27] Da egär ol mundar tuvardan isä da hilas idsin kiyasin bilän
- (10) da arttirsin bešin[ʒ]isini³⁰⁸ üstünä da egär hilas ulunmasa da satilsin kiyasin
- (11) bilän. אך [27:28] Tek ğümlä herem ki herem etsä kiši YWY-ya ğümlädän
- (12) ki anin adamdan ya tuvardan ya tarlovundan tutuvluyinin satilmasin da hilas
- (13) ulunmasın her herem kodeši kodešlärnindir ol YWY-ya. כל [27:29] Här
- (14) herem ki herem bolsa ol adamdan hilas ulunmasin ölmä öldürülsin.
- (15) וכל [27:30] Da ğümlä ondasï ol yernin urluyïndan ol yernin yemišindän ol
- (16) ayačnin YWY-nindir ol kodeš YWY-ya. ואם [27:31] Da egär hilas idmä hilas
- idsä kiši ondasïndan bešinžisin arttïrsïn üstünä. וכל [27:32] Da žümlä
- (18) ondasï sïyïrnïn ya koynïn här ne ki gečsä tibinä ol tayaknïn ol
- (19) onunži bolsin kodeš YWY-ya. אֹז [27:33] Aramasin arasina yahšinin yamanya
- (20) da degištirmäsin anī da egār degištirmā degištirsā anī da bolsīn ol
- (21) da degiškä[n]i³⁰⁹ bolsïn kodeš hilas olunmasin אלה [27:34] Bulardïrlar ol mičvalar

³⁰⁷ JSul.IV.02A: yiliŋa; probably a scribal error. | BSMS 288: yilina. | JSul.III.01: yilina. | ADub.III.73: yilina.

³⁰⁸ JSul.IV.02A: *bešingisini*; a scribal error. | BSMS 288: *bešinžisini*. | JSul.III.01: *besinci ilisini*. | ADub.III.73: *bešinči ülüšün*.

³⁰⁹ JSul.IV.02A: degiškäŋi; a scribal error. | BSMS 288: degištirmägi. | JSul.III.01: alïskanï. | ADub.III.73: alïškanï.

(22...) ki sïmarladï YWY Mošegä oylanlarïna Yisra'elniŋ tayïnda Sinaynïŋ.

וְאֵלֶה (...22)

 310 קניז אַשֶּר צָנָה יוָי אֶת משֶׁה אֶל בְּנֵי יִשׁראָאֵל בְּהַר סִינָי (23)

 311 נָּרוּדְ ינִי לְעוֹלָם אָמֵן וְאָמֵן ינָי לְעוֹלָם בָּרוּדְ ינִי לְעוֹלָם אָמֵן (24)

 $^{^{310}}$ Lev 27:34 in Biblical Hebrew. 311 Psalm 89:52 in Biblical Hebrew: 'Blessed be the LORD forever! Amen and amen'.

Translation

Preliminary Remarks

The core strategies concerning our translation into English of the collected data have been previously elucidated (see section 1.1.). Subsequently, in this section, we delve into the intricate details of our translation process for further clarity.

- * Although the line numbers of the folios are clearly indicated in the transcription, they have been intentionally omitted from the English translation. This is due to inherent discrepancies in the syntactic arrangement between the Karaim text and English, resulting in instances where our translation might not precisely mirror the original line-by-line structure.
- * In the footnotes, an intentional approach has been applied to not differentiate the grammatical gender of inanimate subjects and objects in the literal translations from Biblical Hebrew. This strategic decision has been made to ensure a logical parallelism in the English language rendition, thereby bypassing potential ambiguities.
- * In Turkic languages, grammatical suffixes independently denote a variety of grammatical categories, yet they can also coexist with certain non-suffixical elements. These non-suffixical components, primarily used for emphasis, do not alter the essential meaning of the phrase. For instance, consider *eger ol gel-se-Ø* 'if he comes', *ol gel-se-Ø* 'if he comes', and *gel-se-Ø* 'if he comes'. It should be noted that such structural differences are not reflected in our English translation.
- * Given that the functions of diverse grammatical categories have been elucidated in our study, the literal connotations of these categories will not be subjected to further analysis unless particular circumstances necessitate their notation in the footnotes. For instance, the phrase [Lev 6:18] ne yerdä ki soyulsa ol 'ola translates as 'in the place where the burnt offering is slaughtered', but a more literal rendering would be 'in what place if the burnt offering is slaughtered,' among others. However, should a grammatical category be employed in a manner that deviates from its conventional usage due to various translation strategies, such instances will invariably be highlighted in the footnotes. An example is the translation of the phrase [Lev 5:3] ol bildi da günäḥli boldi' lit. 'he knew and he became sinful'. Due to the context and structure of the English language, this sentence was translated as '|if| he knows |of it|, then he will be guilty', despite the original sentence's past tense markers. In this scenario, since the past

tense markers in Karaim do not serve the function of the present tense, this discrepancy has been duly indicated in the footnotes unlike the previously mentioned example.

- * The utilization of the '|' symbol within the text signifies our necessary editorial insertions to enhance the structural coherence of the English translation. However, such marks have not been employed when the relevant structures have been comprehensively explained in the footnotes.
- * The application of certain prepositions and case governance, diverging from typical English characteristics, has not been explicitly indicated, as these elements have been adapted to align with English standards, e.g., [Lev 6:23] *otka küydürülsin* 'it shall be burnt **in** fire'; lit. 'it shall be burnt **to** fire'.
- * Variations in verb selection, inclusive of light verbs, have consistently been highlighted, e.g., [Lev 16:13] *versin* ol tütsini ol ot üstünä 'he shall **put** the incense on the fire', lit. 'he shall **give** the incense on the fire'.
- * The adaptation of plurality or singularity to English norms has been unmarked in the footnotes, e.g., [Lev 16:24] *yuvsin etini suvlar bilän* 'he shall wash his flesh with **water**', lit. 'he shall wash his flesh with **waters**'.

Editorial Notes

abc the translation of the Karaim text

abc a term that cannot be translated into English

|abc| editorial addition due to the linguistic constraints of English

(abc) the translation of the Karaim written in parentheses in the original text

tabc def¹¹²³ a passage commented in a footnote verse numbers added by the editor

text in Biblical Hebrew (only found in comparisons in footnotes)

Translation of the Book of Leviticus from the Gözleve Bible (1841)

93 ro

Chapter 1

[1:1] And the Lord called Moses, and spoke to him from the tent of meeting, saying, [1:2] 'Speak to the sons of Israel, and say to them, "If Lany man from you¹³¹² ever brings forward an offering to the Lord, you shall bring your offering from the livestock, from the cattle, or from the flock³¹³. [1:3] If his offering is a burnt offering from the cattle, the shall take a healthy male and bring it forward¹³¹⁴ to the door of the tent of meeting for his acceptance before the Lord. [1:4] And he shall prop up his hand on the head of the burnt offering and it will be accepted for him to make atonement for him. [1:5] And he shall slaughter the young bull before the Lord, and the sons of Aaron the priests shall bring the blood, and sprinkle the blood round about on the altar that is by the door of the tent of meeting. [1:6] And he shall skin the burnt offering, and shall cut it Linto as many as pieces possible¹³¹⁵. [1:7] And the sons of Aaron the priests Lishall put¹³¹⁶ fire on the altar, and arrange woods on the fire. [1:8] And the sons of Aaron the priests shall arrange the pieces, |together| with the head, and the stomach fat on the woods

93 v°

that are on the fire which is on the altar. [1:9] And he shall wash its stomach and its legs with water. And the priest shall burn all on the altar. It is a burnt offering, a sacrifice by fire, an aroma of acceptance to the Lord. [1:10] And if his sacrifice is from the flock³¹⁷ of the sheeps or of the goats, the shall take a healthy male and bring forward it¹³¹⁸. [1:11] And he shall slaughter it by the edge of the altar that is on the north side before the Lord, and the sons of Aaron the priests shall sprinkle its blood round about on the altar. [1:12] And he shall cut it tinto as many as pieces possible¹³¹⁹, and its head and its stomach fat and the priest shall arrange them on the woods that are on the fire which is on the altar. [1:13] And he shall wash the stomach and the legs with water, and the priest shall bring all of it and burn it on the altar. It is a burnt offering, an offering made by fire, an aroma of acceptance to the Lord. [1:14] And if

³¹² Lit. 'man from yourselves'.

³¹³ Lit. 'sheep'; a partially mistranslation, see B.Heb צאן; 'flock; sheep'.

³¹⁴ Lit. 'by taking a healthy male he shall bring it forward'.

³¹⁵ Lit. 'according to its joints'.

³¹⁶ Lit. 'they shall give'; see B.Heb יָתְנוּן; lit. 'they give'.

Lit. 'sheep'; a partially mistranslation, see B.Heb צאן; 'flock; sheep'.

³¹⁸ Lit. 'by taking a healthy male he shall bring it forward'.

³¹⁹ Lit. 'according to its joints'.

his burnt sacrifice to the Lord is of birds, then he shall bring his offering of turtledoves or of young pigeons. [1:15] And the priest shall bring it forward to the altar, and cut its head, and burn it on the altar, and its blood shall be tdrained out on the wall of the altar. [1:16] And he shall remove its stomach with its feathers, and cast it beside the altar on the east side, in the place where ashes are thrown away. [1:17] And he shall tear it open by its wings (by its back), but shall not divide it, and the priest shall burn it on the altar, on the woods that are on the fire. It is a burnt offering, sacrifice made by fire, an aroma of acceptance to the Lord.

Chapter 2

[2:1] And if a soul brings a grain offering to the Lord, his offering shall be of wheat flour and he shall pour oil on it and put³²¹ incense on it. [2:2] And he shall bring it to sons of Aaron the priests. And he shall take out a handful³²² of its wheat flour, and of its oil, with all its incense. And the priest shall burn its incense on the altar, |it is| an offering made by fire, of an aroma of acceptance to the Lord. [2:3] And the remnant of the grain offering is Aaron's and his sons'. It is the holy of holies of the burnt offerings of the Lord. [2:4] And if you bring an offering of a grain offering baked in the oven,

94 ro

lit shall be unleavened cakes 1323 of wheat flour mixed with oil; or unleavened thin bread anointed with oil. [2:5] And if your offering is a grain offering |baked| on a pan, it shall be of wheat flour unleavened bread, mixed with oil. [2:6] You shall break it into pieces 1324, and pour 325 oil on it. It is a grain offering. [2:7] And if your offering is a grain offering baked in a saucepan, it shall be made of wheat flour with oil. [2:8] And you shall bring the grain offering |that is made| of these |things| to the Lord, and bring it forward to the priest, and he shall bring it to the altar. [2:9] And the priest shall separate from the grain offering its incense, and shall burn it on the altar. It is an offering made by fire, an aroma of acceptance to the Lord. [2:10] And the remnant of the grain offering is Aaron's and his sons'. It is the holy of holies of the burnt offerings of the Lord. [2:11] If you bring any grain offering to the Lord, it shall surely not

³²⁰ Lit. 'dripped down'.

³²¹ Lit. 'he shall give', see B.Heb נָתַן; 'he gives'.

³²² Lit. 'grasp the full palm of his hand of it', see B.Heb קמץ מְשֶׁם מָלֹא, lit. 'grasp of it fullness of his fistful'.

³²³ Lit. 'thick unleavened bread'.

³²⁴ Lit. 'slicing you shall slice it |into| pieces'. A construction which has an intensifying function (see 2.3.1.2.1.) and is usually attributed to Biblical Hebrew influence. However, the Hebrew Bible does not present the exact structure for this verse, see B.Heb פַּתְּהֹים; lit. 'to morsel it |into| morsels'.

³²⁵ Lit. 'put'.

be made leavened, |because| you shall never burn sour leaven or honey³²⁶ |as| an offering made by fire to the Lord. [2:12] |As for| the offering of the firstfruits¹³²⁷, bring them to the Lord, but they shall not go up to the altar for an aroma of acceptance. [2:13] And tyou shall salt¹³²⁸ all your grain offerings and you shall not let the salt of the covenant of your God cease from your grain offering; you shall bring salt on all your offerings. [2:14] And if you bring forward a grain offering of the Lord, you shall bring forward ears of barley scorched by the fire, fresh barley groats |for| a grain offering of your tirstfruits¹³³⁰. [2:15] And tyou shall put¹³³¹ oil on it, and lay on it incense. It is a grain offering. [2:16] And the priest shall burn its incense, of the groats, and of the oil with all the incense, |as| an offering by fire to the Lord.

Chapter 3

[3:1] And if his offering is a sacrifice of a peace offering, if he brings it forward of the cattle, either a male or female, he shall bring forward a healthy one before the Lord. [3:2] And he shall prop up his hand on the head of his offering, and slaughter it at the door of the tent of meeting; and Aaron's sons the priests shall sprinkle the blood on the altar round about. [3:3] And he shall bring it forward from the sacrifice of the peace offering |as| an offering by fire to the Lord, the fat that covers the stomach, and all the fat that is on the stomach. [3:4] And the two kidneys,

$94 v^{o}$

and the fat that is on them, that is on the entrails, and the extra above the liver: he shall take them³³² away along with the kidneys. [3:5] And Aaron's sons shall burn it on the altar on the burnt sacrifice which is on the woods that are on the fire, |it is| an offering by fire, an aroma of acceptance to the Lord. [3:6] And if his offering for a sacrifice of peace offering to the Lord is of the sheep, either male or female, he shall bring forward a healthy one. [3:7] If he brings forward a lamb for his offering, then he shall brings it forward before the Lord. [3:8] And he shall prop up his hand on the head of his offering, and slaughter it before the tent of meeting; and Aaron's sons shall sprinkle the blood on the altar round about. [3:9] And he shall bring it forward from the sacrifice of the peace offering |as| an offering made by fire to the Lord, he shall take away the whole fat of the tail from opposite the backbone, and the fat that covers the

³²⁶ Lit. 'grape molasses'.

³²⁷ Lit. 'first offering'.

³²⁸ Lit. 'you shall salt with salt'. Probably a calque of B.Heb בַּמֵלָח הַמְלָח; lit. 'with the salt you shall salt it'.

³²⁹ Lit. 'firsts'.

³³⁰ Lit. 'firsts'.

³³¹ Lit. 'you shall give', see B.Heb וָתַה; lit. 'you give'.

³³² Lit. 'it'.

stomach, and all the fat that is upon the stomach. [3:10] And the two kidneys, and the fat that is on them, which is at the entrails, and the extra above the liver: he shall take them³³³ away along with the kidneys. [3:11] And the priest shall burn it on the altar, |it is| an offering by fire to the Lord. [3:12] If his offering is a goat, then he shall offer it before the Lord. [3:13] And he shall prop up his hand over its head and slaughter before the tent of meeting, and the sons of Aaron shall sprinkle its blood on the altar round about. [3:14] And he shall bring it forward from his offering, |as| an offering made by fire to the Lord: the fat that covers the stomach, and all the fat that is on the stomach. [3:15] And the two kidneys, and the fat that is on them, which is at the entrails, and the extra above the liver: he shall take them³³⁴ away along with the kidneys. [3:16] And the priest shall burn them on the altar; |it is| an offering made by fire, an aroma of acceptance. All fat is the Lord's. [3:17] |As| an eternal statute throughout your generations, in all your dwellings: you shall not eat any fat and any blood.

Chapter 4

[4:1] And the Lord spoke to Moses, saying, [4:2] 'Speak to the sons of Israel, saying, "If a soul becomes sinful through a mistake against any of the commandments

95 ro

of the Lord that are not to be done, and does any one of them. [4:3] If the anointed priest becomes sinful, for the iniquities of the people, then he shall bring forward |an offering| for his sin which he has sinned, a healthy young bull of cattle, to the Lord for a sin offering. [4:4] And he shall bring the bull to the door of the tent of meeting before the Lord, and shall prop up his hand on the bull's head, and slaughter the bull before the Lord. [4:5] And the anointed priest shall take of the bull's blood, and bring it to the tent of meeting. [4:6] And the priest shall dip his finger in the blood, and sprinkle some of the blood seven times before the Lord, before the veil of the sanctuary³³⁵. [4:7] And the priest thall put 1336 some of the blood on the horns 337 of the altar of the sanctuary before the Lord in the tent of meeting, and shall pour all the blood of the bull at the base of the altar of the burnt offering which is at the door of the tent of meeting. [4:8] And he shall remove all the fat of the bull of the sin offering, the fat that covers the

³³³ Lit. 'it'.

³³⁴ Lit. 'it'.

³³⁵ Lit. 'holy'; a B.Heb lexical copy, e.g., שָׁדָשׁ; lit. 'holiness'.

³³⁶ Lit. 'he shall give', see B.Heb נתן; 'he gives'.

³³⁷ Lit. 'hollow corners'.

³³⁸ Lit. 'incense of the spices'. Probably a calque of B.Heb, e.g., קטרת הסמים; lit. 'incense of the spices'.

stomach, and all the fat that is on the stomach. [4:9] And the two kidneys, and the fat that is on them, which is on the entrails, and the extra above the liver: he shall take them³³⁹ away along with the kidneys, [4:10] As it is separated from the ox of the sacrifice of peace offerings, the priest shall burn them on the altar of the burnt offering. [4:11] And the skin of the bull, and all its flesh, along with its head, its legs, and its stomach, and its dung. [4:12] And he shall carry the whole bull outside the camp to a clean place, to the place where the ashes are poured out out on the woods with fire: it shall be burned at the place where the ashes are poured out out of the eyes of the assembly, and if they do any of the commandments of the Lord that are not to be done, and if they become guilty, [4:14] And if the sin which they have sinned is known, the congregation shall bring forward a bull, the young of cattle for a sin offering, and bring it before the tent of meeting. [4:15] And the elders of the congregation shall prop up their hands on the head of the bull before the Lord, and they shall slaughter the bull before the Lord.

$95 v^{o}$

[4:16] And the anointed priest shall bring some of the bull's blood to the tent of meeting. [4:17] And the priest shall dip his finger in the blood, and sprinkle it seven times before the Lord, before the veil. [4:18] And the shall put1342 some of the blood on the horns343 of the altar which is before the Lord, that is in the tent of meeting, and shall pour all the blood at the base of the altar of the burnt offering, which is at the door of the tent of meeting. [4:19] And he shall separate all its fat from it, and burn it on the altar. [4:20] And he shall do to the bull as he did to the bull of the sin offering, so he shall do it in that way and the priest shall make atonement for them, then it will be forgiven them. [4:21] And he shall carry the bull outside the camp, and burn it as he burned the first bull. It is a sin offering of the assembly. [4:22] If a leader344 becomes sinful and does one of any commandments of the Lord his God that are not to be done, by mistake and if he becomes guilty, [4:23] Or if his sin which he has sinned is known to him, then he shall bring his offering, a kid of the goats, a healthy male. [4:24] And he shall prop up his hand on the head of the kid, and slaughter it in the place where he slaughters the burnt

³³⁹ Lit. 'it'.

³⁴⁰ Lit. 'to a clean place, to the place, which pours out the ash', see B.Heb אָל־שֶּׁכֶּהְ הַדְּשֶׁבָּן, lit. 'to a clean place, the place pouring out the greasy ash'.

³⁴¹ Lit. 'on the place, which pours out the ash', see B.Heb צֵל־שֶׁפֶּךְ הַדְּשֶׁן; lit. 'on the place pouring out the greasy ash'.

³⁴² Lit. 'he shall give', see B.Heb יָהַן; lit. 'he shall give'.

³⁴³ Lit. 'hollow corners'.

³⁴⁴ Elder brother; a form of address to an older man (CKED: 33).

offering before the Lord: it is a sin offering. [4:25] And the priest shall take some of the blood of the sin offering with his finger, and put³⁴⁵ it on the horns³⁴⁶ of the altar of burnt offering, and shall pour out its blood at the base of the altar of burnt offering. [4:26] And he shall burn all its fat on the altar, as the fat of the sacrifice of peace offerings, and the priest shall make atonement for him concerning his sin, then it will be forgiven him. [4:27] And if a soul of his nation of the land becomes sinful through a mistake, when he did one of the commandments of the Lord which are not to be done, and if he becomes guilty, [4:28] Or if his sin which he has sinned is known to him, then he shall bring his offering, a kid of the goats, a healthy one, for his sin which he has sinned. [4:29] And he shall prop up his hand on the head of the sin offering, and slaughter the sin offering in the place of the burnt offering. [4:30] And the priest shall take some of the blood with his finger, and put³⁴⁷ it on the horns³⁴⁸ of the altar of burnt offering, and shall pour all of its blood at the base of the altar. [4:31] And he shall take away all its fat, just as the fat was taken away from

96 ro

the sacrifice of peace offerings, and the priest shall burn it on the altar for an aroma of acceptance to the Lord; and the priest shall make atonement for him, then it will be forgiven him. [4:32] And if he brings a lamb |as| his sacrifice for a sin offering, he shall bring a healthy female. [4:33] And he shall prop up his hand on the head of the sin offering, and slaughter it for a sin offering in the place where he slaughters the burnt offering. [4:34] And the priest shall take some of the blood of the sin offering with his finger, and put it on the horns of the altar of burnt offering, and shall pour all of its blood at the base of the altar. [4:35] And he shall take away all its fat, as the fat of the lamb is taken away from the sacrifice of the peace offerings, and the priest shall burn them on the altar, on the offerings made by fire to the Lord; and the priest shall make atonement for his sin that he has sinned, then it will be forgiven him.

Chapter 5

[5:1] And if a soul becomes sinful, and hears the voice of an oath |when| he is a witness, whether he has seen or known |of it|, if he does not tell |of it|, then he shall bear his iniquity. [5:2] Or if a soul touches any unclean thing, whether it is a carcass of an unclean wild animal, or it is a

³⁴⁵ Lit. 'he shall give', see B.Heb נתן; lit. 'he gives'.

³⁴⁶ Lit. 'hollow corners'.

³⁴⁷ Lit. 'he shall give', see B.Heb <u>וַתוּ;</u> lit. 'he gives'.

³⁴⁸ Lit. 'hollow corners'.

carcass of an unclean cattle, or it is a carcass of an unclean twarming thing 1349, and if it becomes hidden from him, then the will be unclean and guilty 1350, [5:3] Or if he touches the uncleanness of a man, of any kind of uncleanness that he becomes unclean with it, and if it becomes hidden from him, and the knows 1351 |of it|, then the will be 1352 guilty. [5:4] Or if a soul swears speaking with lips to do evil, or to do good, twhatsoever it be 1353 that a man says with an oath, and if it becomes hidden from him, |when| the knows 1354 of it the will be 1355 guilty in |one| of these |things|. [5:5] And it shall be, if he becomes repentant in one of these |things|, and if he admits that he has sinned tin that thing 1356, [5:6] Then he shall bring his repentance to the Lord for his sin which he has sinned, a female from the flock, a lamb or a kid of the goats, for a sin offering; and the priest shall make atonement for him concerning his sin. [5:7] And if the cannot afford 1357 a lamb, then he shall bring his repentance |for the sin| which he has sinned: two turtledoves, or two young pigeons, to the Lord, one for a sin offering and one for a burnt offering. [5:8] And he shall bring them to the priest, and he shall offer one of them which is for the sin offering first, and nip off

96 vo

its head from³⁵⁸ its neck, but shall not separate it. [5:9] And he shall sprinkle some of the blood of the sin offering on the wall of the altar, and drip out the rest of the blood at the base of the altar. It is a sin offering. [5:10] And he shall do the second for a burnt offering according to the regulation, and the priest shall make atonement for him concerning his sin which he has sinned, and it will be forgiven him. [5:11] But if the cannot afford³⁵⁹ two turtledoves or two young pigeons, then he shall bring his offering for the sin which he has sinned as the tenth part of a *kebič*³⁶⁰ of wheat flour. He shall not put oil on it, and shall not put incense on it, because it is a sin offering. [5:12] And he shall bring it to the priest, and the priest shall take out his handful³⁶¹ of the incense and burn it on the altar, on the offerings made by fire to the Lord. It is a sin

³⁴⁹ Lit. 'the one which lays eggs'.

³⁵⁰ Lit. 'he became unclean and became guilty'.

³⁵¹ Lit. 'he knew'.

³⁵² Lit. 'he became'.

³⁵³ Lit. 'to all'.

³⁵⁴ Lit. 'he knew'.

³⁵⁵ Lit. 'he became'.

³⁵⁶ Lit. 'for it'.

³⁵⁷ Lit. 'his power does not suffice'.

³⁵⁸ Lit. 'from the opposite of'.

³⁵⁹ Lit. 'his power does not suffice'.

³⁶⁰ A measure; an amount of wheat which is given to miller in exchange for grinding (ÖTS 3: 2513).

³⁶¹ Lit. 'grasp of it with his full palm', see B.Heb מלוא קמצו; lit. 'fullness of his fistful'.

offering. [5:13] And the priest shall make atonement for him for his sin that he has sinned in one of these |things|, then it will be forgiven him. And the remainder shall be the priest's, like the grain offering. [5:14] And the Lord spoke to Moses, saying, [5:15] 'If a soul misleads¹³⁶³ and becomes sinful through a mistake in the holy things of the Lord, then he shall bring his repentance to the Lord, a healthy ram out of the flock, with your valuation by mithgals of silver, by the mithgal of the holy, for a guilt offering. [5:16] And he shall pay for what he has become sinful from the holy thing, and shall add fifth part to it and give it to the priest, and the priest shall make atonement for him with the ram of the guilt offering, then it will be forgiven him. [5:17] And if a soul becomes sinful and does one of the commandments of the Lord which are not be done, and did not know |it|, and became guilty, then he shall bear his iniquity. [5:18] And he shall bring a healthy ram out of the flock with your valuation for a guilt offering to the priest, and the priest shall make atonement for him for the mistake he erred and did not know |it|, then it will be forgiven him. [5:19] It is a guilt offering, he certainly has become guilty¹³⁶⁴ |against| the Lord. [5:20] And the Lord spoke to Moses, saying, [5:21] 'If a soul becomes sinful and misleads the Lord, and deceives his friend in a matter of deposit or |in a matter of something| which has been pledged, or |in a matter of something| by oppression or if he has taken away |something| from his friend, [5:22] Or if he finds something lost and deceives about it, and if he swears to a falsehood for one of any of these that man does being sinful by them:

97 ro

[5:23] Then it shall be, if he becomes sinful and repented, then he shall restore what he got by oppression 1366 or the thing that he plundered 1367, or the deposit that was committed to him 1368, or the lost thing that he found, [5:24] Or anything about which he swears to a falsehood for it; then he shall pay it in full 369 and shall add a fifth to it, and give it to him to whom it belongs, on the day of his repentance. [5:25] And he shall bring his guilt offering to the Lord, a healthy

³⁶² Lit. 'it'.

³⁶³ Lit. 'misleads |a| misleading |act|'. Probably a calque of B.Heb תְּמְעֵל מַעֵל; lit. 'misleads |a| misleading |act|'.

³⁶⁴ Lit. 'becoming guilty he became guilty'. Probably a calque of a Biblical Hebrew construction, which has an intensifying function (see 2.3.1.2.1.), see B.Heb אָשֹׁם אָשֶׁם; lit. 'to be guilty, he was guilty'.

³⁶⁵ Lit. 'misleads |a| misleading |act|'. Probably a calque of B.Heb תְּמֶעֹל מְעֵל; lit. 'misleads |a| misleading |act|'.

³⁶⁶ Lit. 'the oppression which he oppressed'. Probably a calque of B.Heb הָּוְלָה אֲשֶׁר נְּוָל; lit. 'the pillage which he pillaged'.

³⁶⁷ Lit. '|the thing| which is taken by holding which he took it by holding'. Probably a calque of B.Heb הָעֹשֶׁק אֲשֶׁר י, lit. 'the extortion which he extorted'.

³⁶⁹ Lit. 'with his head'.

ram out of the flock with your valuation, for a guilt offering, to the priest. [5:26] And the priest shall make atonement for him before the Lord, then it will be forgiven him for any thing out of all that he does and if |thus| becomes guilty through it'.

Chapter 6

[6:1] And the Lord spoke to Moses, saying, [6:2] 'Command Aaron and his sons, saying, "This is the law of the burnt offering: it is the burnt offering on the hearth on the altar all night until the morning, and the fire of the altar shall burn there. [6:3] And the priest shall put on his linen cloths, and he shall put his linen trousers on his flesh, and separate the ashes which the fire covers with the burnt offering on the altar, and he shall put them beside the altar. [6:4] And he shall put off his garments, and put on other garments, and take the ashes outside the camp to a clean place. [6:5] And the fire on the altar shall burn on it, and it shall not go out, and the priest shall burn woods on it Levery morning 1370, and arrange the burnt offering on it, and he shall burn on it the fat of the peace offerings. [6:6] The fire shall burn on the altar continually, it shall not go out. [6:7] And this is the law of the grain offering: a son of Aaron Lshall certainly bring it forward 1371 before the Lord, before the altar. [6:8] And he shall separate from it a handful of the wheat flour of the grain offering,

97 v°

and its oil, and all the incense which is on the grain offering, and shall burn it on the altar, |as| an aroma of acceptance to the Lord. [6:9] And Aaron and his sons shall eat the remainder. Unleavened bread shall be eaten in the holy place, they shall eat it in the courtyard of the tent of meeting. [6:10] It shall not be baked |with| leaven, I have given it |as| their portions of my offerings by fire, it is the holy of holies, like the sin offering and like the guilt offering. [6:11] All the males among the sons of Aaron shall eat it, |as| an Leternal statute 1373 for your generations, from the offerings of the Lord made by fire; every one who touches them shall be holy." [6:12] And the Lord spoke to Moses, saying, [6:13] 'This is the offering of Aaron and of his sons which they shall offer to the Lord after the day of anointment if |the Lord| anoints

³⁷⁰ Lit. 'with morning with morning'. Probably calque of B.Heb בַּבַּקֶר בַּבַּקֶר; lit. 'with the morning with the morning' (see 2.3.1.3.2.).

³⁷¹ Lit. 'bringing forward he shall bring forward'. A construction which has an intensifying function (see 2.3.1.2.1.) and is usually attributed to Biblical Hebrew influence. However, the original text does not present the exact structure for this verse: B.Heb הַקְרֶב ; lit. 'to bring forward'.

³⁷² Lit. 'with his palm', see B.Heb בָּקמִצוֹ; lit. 'with his fistful'.

¹³⁷³ Lit. 'statue of world'. Probably a calque of B.Heb הק־עוֹלָם; lit. 'statue of world'.

Aaron³⁷⁴; a tenth of a *kebič*³⁷⁵ of wheat flour |for| a grain offering continually, a half of it in the morning, and a half of it at night. [6:14] It shall be made with oil into a pan, you shall bring³⁷⁶ the thing 1377 which is fried, and you shall bring it (the cooked ones) forward, you shall bring forward the portions of the grain offering for an aroma of acceptance to the Lord. [6:15] And the anointed priest |who will be| in his place from among his sons, shall do it to the Lord |as| an eternal statute¹³⁷⁸, it shall be a wholly burnt. [6:16] And every grain offering for the priest shall be wholly burnt, it shall not be eaten'. [6:17] And the Lord spoke to Moses, saying, [6:18] 'Speak to Aaron and to his sons, saying, "This is the law of the sin offering: in the place where the burnt offering is slaughtered, the burnt offering shall be slaughtered before the Lord, it is the holy of holies. [6:19] The priest who makes the sin offering shall eat it. It shall be eaten in the holy place, in the courtyard of the tent of meeting, [6:20] If anything touches its flesh, it shall be holy, and if some of its blood is sprinkled on a garment, if it is sprinkled on it, you shall wash in the holy place. [6:21] And the earthen³⁷⁹ vessel in which it is boiled³⁸⁰ shall be broken, and if it is boiled³⁸¹ in a copper vessel, then it shall be broken and washed with water. [6:22] Evey male among the priests shall eat it. It is the holy of holies. [6:23] And every sin offering from which |any of | the blood is brought to the tent of meeting to make atonement in the holy |place|: it shall not be eaten, it shall be burnt in fire.

Chapter 7

[7:1] And this is the law of the guilt offering. It is the holy of holies.

98 ro

[7:2] In the place where they slaughter the burnt offering, they shall slaughter the guilt offering, and he shall sprinkle its blood on the altar round about. [7:3] And he shall bring forward all its fat: the tail fat and the fat that covers the stomach. [7:4] And the two kidneys, and the fat that is on them, that is on the entrails, and the extra above the liver: he shall take them³⁸² away along with the kidneys. [7:5] And the priest shall burn them on the altar for an offering made by fire

³⁷⁴ Lit. 'him', see B.Heb אֹתוֹ; lit. 'him'.

³⁷⁵ 'A measure; an amount of wheat which was given to miller in exchange for grinding' (ÖTS 3: 2513).

³⁷⁶ Lit. 'make |someone| bring'.

³⁷⁷ Lit. 'it'.

³⁷⁸ Lit. 'statue of world'. Probably a calque of B.Heb הָק־עוֹלָם; lit. 'statue of world'.

³⁷⁹ Lit. 'tile'.

³⁸⁰ Lit. 'cooked'.

³⁸¹ Lit. 'cooked'.

³⁸² Lit. 'it'.

to the Lord. It is a guilt offering. [7:6] Every male among the priests shall eat it. It shall be eaten in the holy place. It is the holy of holies. [7:7] As the sin offering is, so is the guilt offering, there is one law for them: if the priest makes atonement with it, shall have it 1383. [7:8] And the priest who brings forward any person's burnt offering, shall have 1384 the skin of the burnt offering which he has brought forward. [7:9] And all the grain offering if it is baked in the oven, and all that is prepared in the saucepan or in the frying pan, the priest who brings it forward, shall have it 1385. [7:10] And all the sons of Aaron shall have 1386 every grain offering mixed with oil or dry, it shall be for every person fraternally³⁸⁷. [7:11] And this is the law of the sacrifice of peace offerings if |someone| brings it forward to the Lord. [7:12] If he brings it forward for thanksgiving, he shall bring it forward with the sacrifice of thanksgiving unleavened cakes mixed with oil and unleavened thin breads anointed with oil, and cooked cakes of wheat flour mixed with oil. [7:13] He shall bring forward his offering with the leavened thick bread, beside the sacrifice of thanksgiving of his peace offerings. [7:14] And from it, he shall brings forward one bread from each type³⁸⁸ of offering breads as an offering to the Lord, and it shall be the priest's who sprinkles the blood of the peace offerings. [7:15] And the flesh of the sacrifice of his peace offerings for thanksgiving shall be eaten on the day that it is offered, it shall not be left out of it until the morning. [7:16] But if the sacrifice of his offering is a vow³⁸⁹ or a generosity offering, it shall be eaten the day that he offers his sacrifice, and the next morning¹³⁹⁰ the remainder of it shall be eaten. [7:17] And the remainder of the flesh of the sacrifice shall be burnt with fire on the third day. [7:18] And if from the flesh of the sacrifice of his peace offerings, is eaten at all¹³⁹¹ on the third day, it will not be accepted,

98 vo

and it shall not be counted as an offer for the one who offered it¹³⁹². It will be a profanation (unfit) and the soul who eats of it shall bear his iniquity. [7:19] And if the flesh touches any kind of unclean thing, it shall not be eaten; it shall be burnt with fire. And the holy flesh: every

³⁸³ Lit. 'it shall be to him', see B.Heb לוֹ יָהָיָה; lit. 'it shall be to him'.

³⁸⁴ Lit. 'it shall be to him', see B.Heb לוֹ יָהְיֵה; lit. 'it shall be to him'.

³⁸⁵ Lit. 'it shall be to him', see B.Heb לוֹ תַהָּיָה; lit. 'it shall be to him'.

³⁸⁶ Lit. 'it shall be to all the sons of Aaron', see B.Heb הָּהֶה לְכַלֹּבְנֵי אָהָרן; lit. 'it shall be to all the sons of Aaron'.

³⁸⁷ Lit. 'like his brother', see B.Heb בָּאָחִין; lit. 'as brother to him'.

³⁸⁸ Lit. 'family; lineage; race'.

³⁸⁹ Lit. 'intention'.

³⁹⁰ Lit. 'its morning', see B.Heb מְּמֶחֶרָת; lit. 'from morrow'.

³⁹¹ Lit. 'being eaten if it is eaten'. Probably a calque of a Biblical Hebrew construction, which has an intensifying function (see 2.3.1.2.1.), see B.Heb אָם הַאָּכל יֵאָכל יֵאָכל : if to be eaten it is being eaten'.

³⁹² Lit. 'him'.

clean person shall eat the holy flesh. [7:20] But if the soul eats of the flesh of the sacrifice of peace offerings, that is the Lord's, and if he has his uncleanness on him, then the soul will be cut off from his people. [7:21] And if a soul touches any unclean thing, the uncleanness of man, or unclean cattle, or any unclean abominable thing, and secretly eats the flesh of the sacrifice of peace offerings that is the Lord's, the soul will be cut off from his people". [7:22] And the Lord spoke to Moses, saying, [7:23] 'Speak to the sons of Israel, saying, "You shall not eat any of the fat of ox, or of sheep, or of goat. [7:24] And the fat of nevala³⁹³, and the fat of terefa³⁹⁴, may be used in lany other use 1395, but you shall definitely not eat it 1396. [7:25] Because anybody who eats the fat from the animal from which one would brings forward an offering made by fire to the Lord; the soul that eats secretly will be cut off from his people. [7:26] And you shall not eat any blood in all of your dwelling places, neither of a bird or either of a cattle. [7:27] If any soul eats any blood, then the soul will be cut off from his people". [7:28] And the Lord spoke to Moses, saying, [7:29] 'Speak to the sons of Israel, saying, "The one who brings forward the sacrifice from his peace offerings to the Lord shall bring his offering to the Lord of the sacrifice of his peace offerings. [7:30] |With| his |own| hands |he| shall bring the offerings of the Lord made by fire, he shall bring the fat with the breast, the breast to wave as³⁹⁷ a wave offering before the Lord. [7:31] And the priest shall burn the fat on the altar, and the breast shall be for Aaron and his sons. [7:32] And you shall give the right thigh to the priest for a heave offering of the sacrifices of your peace offerings. [7:33] The one who brings forward the blood of the peace offerings and the fat from among the sons of Aaron, shall have 1398 the right thigh as his part. [7:34] For I have taken the breast of the wave offering and the thigh of the heave offering of the children of Israel from the sacrifices of their peace offerings, and I have given them to Aaron, the priest, and to his sons as an reternal statute 1399 from the children of Israel." [7:35] This is the greatness portion of Aaron and greatness

99 rº

portion of his sons, from the offerings of the Lord made by fire, from the day he brought forward them to serve as priests to the Lord, [7:36] Which the Lord commanded to be given them from

³⁹³ An animal that died a natural death (CEDHL: 402).

³⁹⁴ An animal torn by wild beasts (CEDHL: 252).

³⁹⁵ Lit. 'to every work'; see, B.Heb לְּכָל־מְלָאכָה; lit. 'for any of work'.

³⁹⁶ Lit. 'eating you shall not eat it'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb אָכל לא תאכלהוי; lit. 'to eat you shall not eat it'.

³⁹⁷ Lit. 'before'.

³⁹⁸ Lit. 'it shall be to him', see B.Heb לוֹ חָהָיֵה; lit. 'it shall be to him'.

²⁹⁹ Lit. 'to statue of world'. Probably a calque of B.Heb לְחָקָת עוֹלָם; lit. 'to statue of world'.

the day that he anointed them from among the children of Israel, as an eternal statute throughout their generations." [7:37] This is the law of the burnt offering, for the grain offering, and for the sin offering, and for the guilt offering, and for the ordination offering, and for the sacrifice of the peace offerings, [7:38] Which the Lord commanded Moses on mount Sinai on the day that he commanded the sons of Israel to bring forward their offerings to the Lord in the wilderness of Sinai.

Chapter 8

[8:1] And the Lord spoke to Moses, saying, [8:2] 'You shall take Aaron and his sons with him, and the clothes, and the anointing oil, and the bull of the sin offering, and two rams, and the basket of unleavened bread. [8:3] And gather all the congregation together to the door of the tent of meeting'. [8:4] And Moses did as the Lord commanded him, and the congregation was gathered to the door of the tent of meeting. [8:5] And Moses said to the congregation, 'This is the word which the Lord commanded to do'. [8:6] And Moses brought Aaron and his sons, and washed them with water. [8:7] And the put 1400 the shirt on him, and girded him with the belt, and clothed him with the mantle, and put⁴⁰¹ the ephod on him. Then he girded him with the belt of the ephod, and tied with it. [8:8] And he put the breastplate on him; and he put in the breastplate the Urim and the Thummim. [8:9] And he put the turban on his head and on the turban, on opposing its front, he put the golden plate⁴⁰², the holy crown, as the Lord commanded Moses. [8:10] And Moses took the anointing oil, and anointed the tabernacle and all that was in it, and sanctified them. [8:11] And he sprinkled some of it on the altar seven times, and anointed the altar and all its tools, the pitcher and its bowl, to sanctify them. [8:12] And he poured of the anointing oil upon Aaron's head, and anointed him to sanctify him. [8:13] And Moses brought forward Aaron's sons, and dressed them in shirts

99 vo

girded them with belts, and dressed them in skullcaps⁴⁰³, as the Lord commanded Moses. [8:14] And he brought forward the bull of the sin offering, and Aaron and his sons propped up their hands on the head of the bull of the sin offering. [8:15] And he slaughtered it, and Moses took the blood, and put⁴⁰⁴ it on the corners of the altar round about with his finger, and sprinkled the

⁴⁰⁰ Lit. 'he gave', see B.Heb יָּהֵן; lit. 'he is giving'.

⁴⁰¹ Lit. 'he gave', see B.Heb יָתוּ; lit. 'he is giving'.

⁴⁰² Lit. 'flower'. Probably a calque of B.Heb צִיץ; lit. 'flower'.

^{403 &#}x27;Tall sheepskin hat'.

⁴⁰⁴ Lit. 'he gave', see B.Heb יהן; lit. 'he is giving'.

altar, and poured the blood at the base of the altar, and sanctified it to make atonement for it. [8:16] And he took all the fat that was on the stomach, and the extra of the liver, and the two kidneys, and their fat, and Moses burned it on the altar. [8:17] And he burned the bull, and its skin, and its flesh, and its dung in fire outside the camp, as the Lord commanded Moses. [8:18] And he offered the ram of the burnt offering, and Aaron and his sons propped up their hands on the head of the ram. [8:19] And he slaughtered it, and Moses sprinkled the blood on the altar round about. [8:20] And he portioned out the ram into as many as pieces possible 1405, and Moses burned the head, and the pieces⁴⁰⁶, and the fat of the stomach. [8:21] And he washed the stomach and the legs with water, then Moses burned the whole ram on the altar. It is a burnt sacrifice, an aroma of acceptance, an offering made by fire to the Lord, as the Lord commanded Moses. [8:22] And he offered the second ram, the ram of ordination, and Aaron and his sons propped up their hands on the head of the ram. [8:23] And he slaughtered it, and Moses took some of the blood of it and put it on the lobe 407 of Aaron's right ear, and on the thumb of his right hand, and on the big toe of his right foot. [8:24] And he brought forward the sons of Aaron, and Moses put some of the blood on the lobe 408 of their right ears, and on the thumbs of their right hands, and on the big toes of their right feet, and Moses sprinkled the blood on the altar round about. [8:25] And he took the fat, and the tail, and all the fat that was on the stomach, and the extra of the liver, and the two kidneys, and their fat, and the right thigh. [8:26] And from the basket of unleavened breads that was before the Lord,

$100 r^{o}$

he took a thick unleavened bread, and a thick oiled bread, and a thin bread, and put them on the fat and on the right thigh. [8:27] And he put all upon the palms of Aaron, and upon the palms of his sons, and waved them as 409 a wave offering before the Lord. [8:28] And Moses took them from their palms, and burned them on the altar on the burnt offering. They are ordination offering, an aroma, an offering made by fire to the Lord. [8:29] And Moses took the breast, and waved it for an wave offering before the Lord. It became Moses's portion of the ram of ordination, as the Lord commanded Moses. [8:30] And Moses took of the anointing oil, and of the blood which was on the altar, and sprinkled it on Aaron, and on his garments, and on his

⁴⁰⁵ Lit. 'according to its joints'.

⁴⁰⁶ Lit. 'joints'.

⁴⁰⁷ Lit. 'its soft |place|'.

⁴⁰⁸ Lit. 'its soft |place|'.

⁴⁰⁹ Lit. 'before'.

sons, and on the garments of his sons with him, and sanctified Aaron, and his garments, and his sons, and the garments of his sons with him. [8:31] And Moses said to Aaron and to his sons, 'Boil the flesh at the door of the tent of meeting, and eat it there, and |eat| the bread that is in the basket of ordination, as I commanded, saying, "Aaron and his sons shall eat it." [8:32] And you shall burn the remainder of the flesh and of the bread in fire. [8:33] And you shall not go out of the door of the tent of meeting for seven days, until the days of your ordination end. For the will consecrate you for seven days. [8:34] As he has done this day, so the Lord has commanded to do, to make atonement for you. [8:35] And you shall sit at the door of the tent of meeting day and night seven days, and you shall keep the testament of the Lord so you do not die, for I have been commanded so'. [8:36] And Aaron and his sons did all the words that the Lord commanded by the hand of Moses.

Chapter 9

[9:1] And it was on the eighth day, |that| Moses called Aaron and his sons, and the elders of

$100 v^0$

Israel. [9:2] And he said to Aaron, 'Take for yourself a young calf of the cattle for a sin offering, and a ram for burnt offering, the healthy ones, and offer them before the Lord. [9:3] And speak to the sons of Israel, saying, "You shall take a kid of the goats for a sin offering, and a calf and a lamb, healthy one-year-olds, for a burnt offering. [9:4] And an ox and a ram for peace offerings to sacrifice before the Lord, and a grain offering mixed with oil. Because, truly, today the divine presence of the Lord will appear to you. [9:5] And they took what Moses commanded before the tent of meeting, and all the congregation came near and stood before the Lord. [9:6] And Moses said, 'This is the word which the Lord commanded that you shall do, and the divine presence of the Lord shall be evident to you'. [9:7] And Moses said to Aaron, 'Come forward to the altar and make your sin offering, and your burnt offering, and make atonement for yourself, and for the people, and make the offering of the people, and make atonement for them, as the Lord commanded. [9:8] And Aaron came near to the altar, and slaughtered the calf of the sin offering which was for himself. [9:9] And the sons of Aaron brought forward the blood to him, and he dipped his finger in the blood, and put and the fat, and the kidneys, and the

298

_

⁴¹⁰ Lit. 'he will make your hand accustomed seven days by filling |it|'. A calque of B.Heb שָׁבְעַת יָמִים יְמֵלָּא אֶת־יָדְכֶּם; iit. 'seven of days he shall fill your hand'.

⁴¹¹ Lit. 'he gave', see B.Heb יָּתֵן; lit. 'he is giving'.

extra above the liver of the sin offering, on the altar, as the Lord commanded Moses. [9:11] And he burned the flesh and the skin in fire outside the camp. [9:12] And he slaughtered the burnt offering, and Aaron's sons brought the blood to him and he sprinkled on the altar round about. [9:13] And they brought the burnt offering to him tin as many as pieces possible 1412, and the head, and he burned them on the altar. [9:14] And he washed the stomach and the legs, and burned them on the altar. [9:15] And he brought the people's offering, and took the kid of the sin offering that was for the people, and slaughtered it, and made a sin offering like the first. [9:16] And he brought forward the burnt offering, tand offered 1413 it according to the regulation. [9:17] And he brought the grain offering, tand take out a handful 1414 of it, and burned it on the altar beside the burnt sacrifice of the morning.

101 ro

[9:18] And he slaughtered the ox and the ram for a sacrifice of peace offerings which was for the people; and Aaron's sons brought him the blood which he sprinkled on the altar round about. [9:19] And the fat of the ox and of the ram, the |fat| tail, and that which covers |the entrails|, and the kidneys, and the extra of the liver (they divided). [9:20] They put the fat on the breasts and he burned the fat on the altar. [9:21] And Aaron waved the breasts and the right thigh as 415 a wave offering before the Lord, as the Lord commanded Moses. [9:22] And Aaron lifted up his hands toward the people, and blessed them, and after 416 he had come down from making the sin offering, and the burnt offering, and peace offerings. [9:23] And Moses and Aaron came to the tent of meeting, and came out, and blessed the people, and the divine presence of the Lord became evident to all the people. [9:24] And then fire came out from before the Lord, and consumed the burnt offering and the fat on the altar, and all the people saw this and they shouted and fell on their faces.

Chapter 10

[10:1] And Nadab and Abihu, the sons of Aaron, Leach of them 1417 took his censer, and put 418 fire in them, and put incense on them, and they offered strange fire before the Lord, which he had not commanded them. [10:2] And fire came out from before the Lord and consumed them,

⁴¹² Lit. 'according to its joints'.

⁴¹³ Lit. 'and he did', see B.Heb נֵיצְשֶׂהָ; 'and he did'.

Lit. 'and he filled his palm', see B.Heb יִימַלָּא כָפּוֹ; lit. 'and he is filling his palm'.

⁴¹⁵ Lit. 'before'.

⁴¹⁶ Lit. 'first; beforehand; initially'.

⁴¹⁷ Lit. 'every person'.

⁴¹⁸ Lit. 'they gave', see B.Heb יְּחְנוּ; lit. 'they are giving'.

and they died before the Lord. [10:3] And Moses said to Aaron, 'This is it that the Lord spoke, saying, "I will be sanctified to those who are near to me, and I will be glorified before all the people", and Aaron kept silent. [10:4] And Moses called Mishael and Elzaphan, the sons of Uzziel, the uncle of Aaron, and said to them, 'Come near, carry⁴¹⁹ your brothers from before the surfaces of sanctuary out of the camp'. [10:5] And they came near and carried⁴²⁰ them, with their shirts, out of the camp, as Moses had said. [10:6] And Moses said to Aaron, and to Eleazar and to Ithamar, his sons, 'Do not uncover your heads and do not rip your clothes, lest you die, and lest he be angry at all the congregation. But your brothers, the whole congregation of Israel, shall weep for the burning which the Lord has kindled. [10:7] And you shall not go out from the door of the tent of meeting, or you will die, because the anointing oil of the Lord is on you'. And they did

$101 v^{0}$

according to the word of Moses. [10:8] And the Lord spoke to Aaron, saying, [10:9] 'Do not drink new wine nor old wine, neither you nor your sons with you, before you go into the tent of meeting, lest you die. It shall be a statute forever throughout your generations. [10:10] In the same way¹⁴²¹ |when| it comes to distinguish between holy and worthless, and between unclean and clean, [10:11] And |when| it comes to teach the sons of Israel all the statutes which the Lord has spoken to them by the hand of Moses'. [10:12] And Moses spoke to Aaron, and to Eleazar and to Ithamar, his sons who were left, 'Take the grain offering, the remainder of the offerings of the Lord made by fire and eat it unleavened beside the altar, because it is the holy of holies. [10:13] And you shall eat it in the holy place, because it is your share, and the share of your sons, of the sacrifices of the Lord made by fire. Because I have been commanded so. [10:14] And you shall eat the breast of the wave offering and the thigh of the partial offering in a clean place, you and your sons, and your daughters with you. Because it is your share, and the share of your sons which were given out of the sacrifices of peace offerings of the sons of Israel. [10:15] They shall bring the thigh of the partial offering and the breast of the wave offering with the offerings made by fire of the fat to offer it for a wave offering before the Lord, and it shall be yours and your sons' with you, by a statute forever, as the Lord has commanded'.

419 Lit. 'lift'.

⁴²⁰ Lit. 'lift'.

⁴²¹ In that way.

[10:16] And Moses Ldiligently sought 1422 the goat of the sin offering, and, lo, it was burnt, and he was angry at Eleazar and Ithamar, the sons of Aaron who remained, saying, [10:17] 'Why have you not eaten the sin offering in the holy place? Because it is the holy of the holies, and |God| has given it to you to bear the iniquity of the congregation, to make atonement for them before the Lord. [10:18] Lo, the blood of it was not brought into the holy place, Lyou shall indeed have eaten it 1423 in the holy place, as I commanded'. [10:19] And Aaron said to Moses, 'Lo, today they offered their sin offerings and their burnt offerings before the Lord, and |things| like these have befallen me, and if I had eaten the sin offering today, would it been good in the Lord's eyes?' [10:20] And Moses heard |that| and it appeared 424 good in his eyes.

Chapter 11

[11:1] And the Lord spoke to Moses and to Aaron, saying to them,

102 ro

[11:2] 'Speak to the sons of Israel, saying, "These are the animals that you shall eat among all the livestock that are on the earth. [11:3] All that have divided hoofs, and are cloven-hoofed and that chew the cud among the livestock, you shall eat." [11:4] But among those that chew the cud or have divided hoofs, you shall not eat these: the camel, because 425 it chews the cud, but is not a completely cloven-hoofed, it is unclean to you. [11:5] And the rabbit 426 (hedgehog); although it chews the cud, but it does not part the complete hoof, it is unclean to you. [11:6] And the hare; although it chews the cud, but it does 427 not part the complete |hoof|, it is unclean to you. [11:7] And the pig (swine); although it does have a divided hoof and is cloven-hoofed 428, but it does not chew the cud, it is unclean to you. [11:8] You shall not eat of their flesh and do not touch their carcasses, they are forbidden to you. [11:9] You shall eat these of all that are in the water: 1 everything that has 1429 fins 430 and scales in the water, in the seas, and in the rivers,

⁴²³ Lit. 'eating you shall eat it'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb אָכוֹל תַּאַכְלוּ; lit. 'to eat you shall eat it'.

⁴²⁴ Lit. 'became'.

⁴²⁵ Lit. 'although'.

⁴²⁶ Lit. 'island rabbit'.

⁴²⁷ Lit. 'did'.

⁴²⁸ Lit. 'it completely parts its hoof's dividing |part|'.

Lit. 'everything which to him'. Probably a calque of B.Heb כל אָשֶר־לו; lit. 'all which to him'.

⁴³⁰ Lit. 'wing'.

you shall eat them. [11:10] And reverything that has 1431 no fins and scales, in the sea, in the rivers, of all creatures of the water that swarm⁴³² and of all the living creatures⁴³³ that are in the waters: they are detestable to you. [11:11] And they shall be detestable to you. You shall not eat of their flesh and abominate their carcasses. [11:12] Everything that has 1434 no fins and scales in the water shall be detestable to you. [11:13] And you shall abominate these among the birds, they shall not be eaten, they are detestable: the eagle (black bird), and the bearded vulture, and the sea eagle, [11:14] And the vulture, and the kite by its kind. [11:15] Every raven by its kind, [11:16] And the ostrich, and the owl and the cuckoo, and the falcon by its kind. [11:17] And the little owl¹⁴³⁵, and the cormorant, and the screech owl¹⁴³⁶, [11:18] And the swan, and the pelican, and the raham⁴³⁷, [11:19] And the stork, and the heron by its kind, and the hoopoe, and the bat. [11:20] And every sliping insect 1438, that goes on four legs shall be detestable to you. [11:21] Among all creatures that have been born of the flying insects¹⁴³⁹, you shall eat only these those go on four legs, which thave knees above their feet, to leap by them on the earth¹⁴⁴⁰. [11:22] You shall eat these of them: the locust of by its kind, and the cicada by its kind, and the $hargol^{441}$ by its kind and $hagav^{442}$ by its kind. [11:23] And all the |creatures| that were born, of those insects⁴⁴³ which have four feet, it shall be detestable

102 vo

to you. [11:24] And you will be unclean by touching these: anyone who touches their carcasses shall be unclean until the evening. [11:25] And the one who (the one who carries) lifts their carcasses shall wash his clothes, and be unclean until the evening. [11:26] Every animal that has divided hoof and Lis not cloven-hoofed¹⁴⁴⁴, and does not chew the cud, they are unclean to you. Everyone who touches their carcasses shall be unclean until the evening. [11:27] And all

 $^{^{431}}$ Lit. 'everything which there is no to him'. Probably a calque of B.Heb לֵלְ אֲשֶׁר אֵין־לֹו; lit. 'all which there is no to him'.

⁴³² Lit. 'lay eggs'.

⁴³³ Lit. 'soul', see B.Heb נָלָּשׁ; lit. 'soul'.

⁴³⁴ Lit. 'everything which there is no to him'. Probably a calque of B.Heb כֹל אֲשֶׁר אֵין־לוֹ; lit. 'all which there is no to him'.

⁴³⁵ Eagle owl.

⁴³⁶ Uncertain translation; 'a species of owl; an owl' (WAED: 372, CPED: 804).

⁴³⁷ Probably 'carrion vulture' (vultur percnopterus) (CEDHL:613).

Lit. 'flying bird', see B.Heb הַעוֹף; lit. 'the flyer'.

⁴³⁹ Lit. 'flying bird', see B.Heb העוף; lit. 'the flyer'.

⁴⁴⁰ Lit. 'to him knees from above to its feet to leap with them on the land'. Probably a calque of B.Heb לְנַתֵּר בָּהֵן ; lit. 'to him shanks from above to its feet to hop with them on the land'.

⁴⁴¹ Probably a kind of locust (CEDHL: 230).

⁴⁴² Probably 'locust' or 'grasshopper' (CEDHL: 207).

⁴⁴³ Lit. 'bird', see B.Heb הַּעֹוֹף; lit. 'the flyer'.

⁴⁴⁴ Lit. 'a complete division it does not divide'.

that walks upon its paws 1445 of all beasts that go on four legs, those are forbidden to you; whoever touches their carcasses shall be unclean until the evening. [11:28] And the one who lifts the carcasses of them shall wash his clothes, and be unclean until the evening. They are forbidden to you. [11:29] And this is forbidden to you: the creatures that were born and that swarm⁴⁴⁶ on the earth: the weasel, and the rat, and the turtoise by its kind, [11:30] And the shrew mouse, and the sun lizard, and the star lizard, and the lizard, and the mole. [11:31] These are the forbidden |things| to you |among| all that were born. Whoever touches them, after they died, shall be unclean until the evening. [11:32] And anything on which anyl of them falls, after they have died, shall be forbidden | whether | it is any equipment (vessel) of wood, or clothing, or skin, or sack, whatever equipment that could be use for work with them, shall be put into water, and it shall be unclean until the evening, then it becomes clean. [11:33] And every pot⁴⁴⁷ into which any of them falls, whatever that is inside of it, it shall be unclean, and you shall break the pot. [11:34] Of all food which is eaten on which water comes, it will be unclean, and all drink that is drunk in every vessel will be unclean. [11:35] And anything that any part of their carcass falls on will be unclean. The oven and the hearth: they shall be destroyed, they are unclean, and shall be unclean to you. [11:36] Only: a spring or a cistern, a concentration of water will be clean, and whatever touches their carcasses will be unclean. [11:37] But if |any part of their carcasses falls on any sowing seed which is sown, it is clean. [11:38] And if any water is put 1448 on the seed and any part of their carcass falls on it, it is unclean to you. [11:39] And if any of livestock which is clean to eat for you dies, the one who touches its carcass shall be unclean until the evening. [11:40] And the one who eats

103 ro

of its carcass shall wash his clothes, and will be unclean until the evening, and the one who carries⁴⁴⁹ its carcass shall wash his clothes, and will be unclean until the evening. [11:41] And everything that was born, that was born on the earth is detestable. It shall not be eaten. [11:42] And every|thing| which walks on a belly and every|thing| that walks on four legs, among all creatures that were born, that were born on the earth: you shall not eat them, because they are detestable things. [11:43] You shall not make your souls disgusted with any of the creatures

⁴⁴⁵ Lit. 'on its soles', see B.Heb עַל־כַּפֵּין; lit. 'on soles of him'.

⁴⁴⁶ Lit. 'lay eggs'.

⁴⁴⁷ Flowerpot vessel.

⁴⁴⁸ Lit. 'it was given', see B.Heb יָהַן; lit. 'it was given'.

⁴⁴⁹ Lit. 'lifts'.

that were born, that lay eggs and you shall not be unclean with them, because your mind will be closed by them. [11:44] Because I am the Lord your God, you shall consecrate yourselves¹⁴⁵⁰ and be holy, because I am holy, and you shall not defile your souls with any creature that was born and that moves on the earth. [11:45] Because I am the Lord who brought you up out of the land of Egypt to be your God, therefore be holy, because I am holy". [11:46] This is the law of the livestock, and of the bird, and of every living soul that moves in the water, and of every soul that gives birth on the earth, [11:47] To distinguish between the forbidden and the clean holy, and between the animal that will be eaten and the animal hat will not be eaten.

Chapter 12

[12:1] And the Lord spoke to Moses, saying, [12:2] 'Speak to the sons of Israel, saying, "If a woman conceives, and bears a male |child|, then she shall be unclean seven days, as in the days of her menstruation illness, she shall be unclean. [12:3] And on the eighth day the flesh of his foreskin shall be circumcised. [12:4] And thirty-three days she shall sit with the blood of her purification, she shall not touch |anything| holy, nor come into the sanctuary until the days of her purification be fulfilled. [12:5] But if she bears a female |child|, then she will be unclean two weeks, tas in her menstruation 1453, and sixty-six days she shall sit for the blood of her purification. [12:6] And when the days of her purification are fulfilled for a son, or for a daughter, she shall bring a lamb of the first year for a burnt offering, and a young pigeon or a turtledove for a sin offering to the door of the tabernacle

103 vo

of meeting, to the priest. [12:7] And he shall bring it forward before the Lord and make atonement for her and she shall be clean from the flow⁴⁵⁴ of her blood. This is the law for the one who bears either a male or a female. [12:8] And if ₁she cannot afford¹⁴⁵⁵ a lamb, then she shall take two turtledoves, or two young pigeons, one for the burnt offering, and one for the sin offering, and the priest shall make atonement for her, then she will be clean."

Chapter 13

⁴⁵⁰ Lit. 'you shall be holy', see B.Heb נְהָיִיתֶם קְדֹשִׁים; lit. 'you become holy-ones'.

⁴⁵¹ Lit. 'between the clean', see B.Heb בִין הַטָּהֹר; lit. 'between the clean'.

⁴⁵² Lit. 'between the animal', see B.Heb בֵין הַחַיָּה; lit. 'between the animal'.

⁴⁵³ Lit. 'for her menstruation'. Probably a mistranslation of Hebrew preposition ; 'as, like' with 'כָּי; 'for; since; because', see B.Heb בָּנְדְּתָה ; lit. 'as isolation of her'.

⁴⁵⁴ Lit. 'fountain; spring'. Probably a calque of B.Heb וֹלָקר lit. 'fountain'.

⁴⁵⁵ Lit. 'her power does not suffice'.

[13:1] And the Lord spoke to Moses and Aaron, saying, [13:2] 'If a man has in the skin of his flesh a swelling, a lichen, or a white spot, and if it becomes in the skin of his flesh a disease of leprosy, then he shall be brought to Aaron the priest or to one of his sons the priests. [13:3] And the priest shall look on the disease in the skin of his flesh, and if the hair in the disease is changed into white, and the appearance of the disease is deeper than the skin of his flesh, it is the disease of leprosy and if the priest looks on him, then he shall pronounce⁴⁵⁶ him unclean. [13:4] And if it is the white spot which is on the skin of his flesh, and if its appearance is not deeper than the skin, and the hair has not changed into white, then the priest shall shut up the disease seven days. [13:5] And the priest shall look on him on the seventh day, and, lo, if the disease in its colour has stopped, and the disease has not spread on the skin, then the priest shall shut him up seven days a second |time|. [13:6] And the priest shall look on him on the seventh day for the second time, and, lo, if the disease is dim and if the disease has not spread on the skin, the priest shall pronounce⁴⁵⁷ him clean. It is a lichen, and he shall wash his clothes, and shall be clean. [13:7] But if the lichen spreads farther on the skin, after he has shown himself to the priest, after his cleansing, he shall show himself to the priest for a second time. [13:8] And if the priest looks on, and, lo, if the lichen has spread on the skin, than the priest shall pronounce⁴⁵⁹ him unclean. It is a leprosy. [13:9] If the disease of leprosy is in a man, then he shall be brought to the priest. [13:10] And the priest shall look, and, lo, |if| there is a white swelling on the skin, and it has changed the hair into white, and there is a living 460 raw flesh in the swelling, [13:11] It is an aging leprosy in the skin of his flesh, and the priest shall pronounce⁴⁶¹ him unclean, and shall not shut him up, because he is

104 ro

unclean. [13:12] And if the leprosy Lapreads farther 1462 on the skin, and if the leprosy covers all the skin of |the one with| the disease from his head to his foot, to wherever it is within sight of the priest's eyes 1463, [13:13] And the priest shall look: and, lo, |if| the leprosy has covered all

⁴⁵⁶ Lit. 'make'.

⁴⁵⁷ Lit. 'make'.

⁴⁵⁸ Lit. 'spreading if it spreads'. Probably a calque of a Biblical Hebrew construction, which has an intensifying function (see 2.3.1.2.1.), see B.Heb אַם־פָּשֹה תִפְּשֶׂה; lit. 'if to diffuse it is diffusing'.

⁴⁵⁹ Lit. 'make'.

⁴⁶⁰ Lit. 'the health of...'.

⁴⁶¹ Lit. 'make'.

⁴⁶² Lit. 'spreading if it spreads'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb אָם מְסנֵא הָסנֵי lit. 'if to bud it is budding'.

⁴⁶³ Lit. 'to all appearance of eyes of the priest'. Probably a calque of B.Heb לָכֶל־מַרְאֵה עֵינֵי הַכּהַן; lit. 'to all of appearance of eyes of the priest'.

his flesh, he shall pronounce⁴⁶⁴ |the one with| the disease clean. It has all changed and turned white. He is clean. [13:14] And on the day the raw flesh appears on him, he shall be unclean. [13:15] And the priest shall look on the raw flesh, and pronounce⁴⁶⁵ him unclean, |because| the raw flesh is unclean, it is the disease of leprosy. [13:16] Or if the raw flesh turns |again| and is changed to white, he shall come to the priest. [13:17] And the priest shall look him: and, lo, if the disease is turned to white, the priest shall pronounce⁴⁶⁶ | the one with | the disease clean. He is clean. [13:18] And if there is a flesh in the skin of which there is an abscess, and it heals, [13:19] And if there is a white swelling or a white-reddish spot in the place of the abscess, it shall be seen by the priest. [13:20] And the priest shall look: and lo, if its appearance is lower than the skin, and its hair has changed into white, the priest shall pronounce⁴⁶⁷ him unclean. It is the disease of leprosy, it spread in the boil. [13:21] But if the priest looks at it, and, lo, there is no white hairs in it, and if it is not lower than the skin, and is dim, then the priest shall shut him up seven days. [13:22] And if it spreads⁴⁶⁸ on the skin, then the priest shall pronounce⁴⁶⁹ him unclean. It is a disease. [13:23] And if the bright spot stays in its place, and does not spread, it is the burn of the abscess, and the priest shall pronounce⁴⁷⁰ him clean. [13:24] Or if there is a flesh, in the skin of which there is a burn of a fire, and the living⁴⁷¹ raw |flesh| of the burn has a white-reddish or white spot, [13:25] Then the priest shall look on it: and, lo, if the hair in the spot has turned white and its appearance is deeper than the skin, it is a leprosy spread in the burn, and the priest shall pronounce⁴⁷² him unclean. It is the disease of leprosy. [13:26] But if the priest looks at it, and, lo, there is no white hair in the spot, and it is no lower than the skin, and is dim, then the priest shall shut him up seven days. [13:27] And the priest shall look on him on the seventh day, and if it spreads farther 1473 on the skin, then the priest shall pronounce⁴⁷⁴ him unclean. It is the disease of

104 v°

⁴⁶⁴ Lit. 'make'.

⁴⁶⁵ Lit. 'make'.

⁴⁶⁶ Lit. 'make'.

⁴⁶⁷ Lit. 'make'.

⁴⁶⁸ Lit. 'spreading |if| it spreads'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb אָם מְסנֵא הָסנֵי, lit. 'if to bud it is budding'.

⁴⁶⁹ Lit. 'make'.

⁴⁷⁰ Lit. 'make'.

⁴⁷¹ Lit. 'its health'.

⁴⁷² Lit. 'make'.

Lit. 'spreading if it spreads'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb אָם מְסנְא תְסנֵי ; lit. 'if to bud it is budding'.

⁴⁷⁴ Lit. 'make'.

leprosy. [13:28] And if the spot stays in its place, and does not spread on the skin, and is dim, it is a swelling of the burn, then the priest shall pronounce⁴⁷⁵ him clean. Because it is a scar⁴⁷⁶ of the burn. [13:29] If a man or woman has a disease on the head or the beard, [13:30] Then the priest shall look on the disease: and, lo, if its appearance is deeper than the skin and there is in it yellow thin hair, then the priest shall pronounce⁴⁷⁷ him unclean. It is a scab, a leprosy of the head or the beard. [13:31] And if the priest looks at the disease of the scab, and, lo, if its appearance is deeper than the skin and there is no black hair in it, then the priest shall shut up the one with the disease of the scab seven days. [13:32] And the priest shall look at the disease on the seventh day: and, lo, if the scab has not spread, and there is no yellow hair in it, and the appearance of the scab is not deeper than the skin, [13:33] Then he shall shave himself, but he shall not shave the scab, and the priest shall shut up him |the one with| the scab seven days a second time. [13:34] And the priest shall look at the scab on the seventh day: and, lo, if the scab has not spread on the skin, and its appearance is not deeper than the skin, then the priest shall pronounce⁴⁷⁸ him clean, and he shall wash his clothes, and he will be clean, [13:35] And if the scab spreads farther 1479 on the skin after his cleansing, [13:36] Then the priest shall look at him, and, lo, if the scab has spread on the skin, |then| the priest shall not inspect for yellow hair, he is unclean. [13:37] And if the scab stays in its colour and black hair has grown in it, |then| the scab is healed, |then| he is clean, and the priest shall pronounce⁴⁸⁰ him clean. [13:38] And if a man or a woman has spots, white spots in the skin of their flesh. [13:39] Then the priest shall see, and, lo, |if| the spots on the skin of their flesh are dim white, |then| it is a bohak⁴⁸¹ that has spread on the body. He is clean. [13:40] And if a man's hair is fallen of, he is bald. He is clean. [13:41] And if his hair has fallen off from the part of his head |towards| his face, |then| he is gibeah⁴⁸², he is clean. [13:42] And if there is a white-reddish disease on the back of his head or on the front of |his head|, it is a leprosy spreading on the back of |his head| or on the front of his head. [13:43] And the priest shall look on it, and, lo, if the swelling of the disease appears white-reddish, on the back of his head

_

⁴⁷⁵ Lit. 'make'.

⁴⁷⁶ Lit. 'burn'.

⁴⁷⁷ Lit. 'make'.

⁴⁷⁸ Lit. 'make'.

⁴⁷⁹ Lit. 'spreading |if| it spreads'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb אָם מְסנָא חָסנֵי, lit. 'if to bud it is budding'.

⁴⁸⁰ Lit. 'make'.

⁴⁸¹ A kind of disease (CEDHL: 65); harmless eruption of the skin, skin-disease' (LVTL: 111).

⁴⁸² Bald on the forehead (CEDHL: 89).

or on the front of his head, like the appearance of leprosy in his skin of the flesh, [13:44] He is a leprous man, he is unclean, the priest shall pronounce⁴⁸³ him unclean. His disease is in his head. [13:45] And the leprous |man| in whom the disease is: his clothes shall be ripped, and his head shall be bare, and he shall cover over his moustache, and shall cry out, "Unclean, unclean!", [13:46] All the days in which the disease shall be in him he is unclean. He shall dwell alone, his dwelling shall be 1484 outside the camp. [13:47] And the garment in which the disease of leprosy is, |whether| in garment of wool, or in garment of linen, [13:48] Either in the warp or in the weft of the linen or of the wool or in leather or in lany item that is made of leather ¹⁴⁸⁵. [13:49] And if the disease is greenish or reddish in the garment, or in the skin, or in the warp, or in the weft, or in any item of skin: it is the disease of leprosy, and it shall be shown to the priest. [13:50] And the priest shall look on the disease, and shut up the disease seven days. [13:51] And he shall see the disease on the seventh day. If the disease has spread in the garment, either in warp or in weft, or in the leather, or in any item that is made of leather 1486: the disease is a pain causing leprosy, it is unclean. [13:52] And he shall burn that garment, or the weft or the woof, that is of woollen or linen, or any item that is made of leather 1487 if there is disease in them. Because it is a pain causing leprosy, it shall be burned in the fire. [13:53] And if the priest sees it, and, lo, the disease has not spread in the garment, or in the warp, or in the weft, or in any item that is made of leather [1488], [13:54] Then the priest shall command, and they wash the thing in which there is the disease, and he shall shut it in seven days a second time. [13:55] And the priest shall look at the disease after it has been washed from it, and, lo, the disease has not changed its colour, and the disease has not spread: it is unclean. You shall burn it in the fire: it is a sunken spot, |whether| on the back or on the front. [13:56] And if the priest sees it, and, lo, the disease is dim after it was washed, then he shall tear it out of the garment, or out of the leather, or out of the weft, or out of the woof. [13:57] And if it still appears in the garment, or in the weft, or in the woof, or in any item of

$105\ v^o$

⁴⁸³ Lit. 'make'.

⁴⁸⁴ Lit. 'is'.

⁴⁸⁵ Lit. 'in all leather work'. Probably a calque of B.Heb בָּכִל־מֶלְאַכֶת עוֹר; lit. 'in any of work of leather'.

⁴⁸⁶ Lit. 'in all leather work'. Probably a calque of B.Heb בָּבֶל־מְלַאכֶת עוֹר; lit. 'in any of work of leather'.

⁴⁸⁷ Lit. 'in all leather work'. Probably a calque of B.Heb בָּכֶל־מֶלְאֶרֶת עוֹר; lit. 'in any of work of leather'.

⁴⁸⁸ Lit. 'in all leather work'. Probably a calque of B.Heb בָּכִל־מֶּלְאַכָּת עוֹר; lit. 'in any of work of leather'.

skin: it is a spreading disease, you shall burn in fire that in which the disease is. [13:58] And the garment, or the weft, or the woof, or any leather item: if you wash |them| and if the disease departs from them,then it shall be washed a second time, and shall be clean'. [13:59] This is the law of the disease of leprosy in a garment of wool or linen, or weft, or woof, or in any leather item, to pronounce⁴⁸⁹ it clean or to pronounce⁴⁹⁰ it unclean.

Chapter 14

[14:1] And the Lord spoke to Moses, saying, [14:2] 'This shall be the law of the disease of leprosy on the day of his cleansing: And he shall be brought to the priest. [14:3] And the priest shall go out of the camp, and the priest shall look on, and, lo, if the disease of leprosy has been healed from the one who has leprosy, [14:4] Then the priest shall command, and |one| shall take for the one who is going to be cleansed: the two clean living⁴⁹¹ birds, and cypress wood, and scarlet silk, and hyssop. [14:5] And the priest shall command and he shall slay tone of the birds¹⁴⁹² in an earthen vessel over sweet water. [14:6] The living⁴⁹³ bird: he shall take it and the cypress wood and the scarlet silk and the hyssop, and shall dip them and the living⁴⁹⁴ bird in the blood of the bird that was slayed over the sweet water. [14:7] And he shall sprinkle on the one who is to be cleansed from the leprosy seven times, and shall pronounce⁴⁹⁵ him clean, and shall let the living⁴⁹⁶ bird go over the surface of the field. [14:8] And the one being cleansed shall wash his clothes, and shave off all his hair, and wash himself with water, then shall be clean. And after that he shall come into the camp, and shall live outside of his tent seven days. [14:9] And it will be on the seventh day that he shall shave all his hair, his head and his beard, and his eyelashes of his eyes. He shall shave off all his hair, and he shall wash his clothes, and he shall wash his flesh with water, and he shall be clean. [14:10] And on the eighth day he shall take

106 ro

⁴⁸⁹ Lit. 'make'.

⁴⁹⁰ Lit. 'make'.

⁴⁹¹ Lit. 'wild'.

⁴⁹² Lit. 'the one bird', see B.Heb הַאָּחָת; lit. 'the bird the one'.

⁴⁹³ Lit. 'wild'.

⁴⁹⁴ Lit. 'wild'.

⁴⁹⁵ Lit. 'make'.

⁴⁹⁶ Lit. 'wild'.

two healthy male lambs, and one healthy one-year-old female lamb, and three tenths of wheat flour, mixed with oil, and one sevik⁴⁹⁷ oil. [14:11] And the priest who makes him clean shall set the man who is being cleaned, and those things 1498 before the Lord, at the door of the tent of meeting. [14:12] And the priest shall take one lamb, and offer it for a guilt offering, and the seyik⁴⁹⁹ oil, and wave them for a wave offering before the Lord. [14:13] And he shall slay the lamb in the place where he slays the sin offering and the burnt offering in the holy |place|, because the guilt offering is like the sin offering, it is the priest's, it is the holy of holies. [14:14] And the priest shall take some of the blood of the guilt offering, and the priest shall put 1500 it on the lobe⁵⁰¹ of the right ear of the one who is to be cleansed, and on the thumb of his right hand, and on the big toe of his right foot. [14:15] And the priest shall take some from the $seyik^{502}$ oil, and pour it into the palm of the priest's (his own) left hand. [14:16] And the priest shall dip the finger of his right hand in the oil that is on his left palm, and shall sprinkle some of the oil with his finger seven times before the Lord. [14:17] And the rest of the oil that is on his palm, the priest Lshall put¹⁵⁰³ it on the lobe⁵⁰⁴ of the right ear of the one who is to be cleansed, and on the thumb of his right hand, and on the big toe of his right foot, over the blood of the guilt offering. [14:18] And he shall put⁵⁰⁵ the remnant of the oil that is on the priest's palm on the head of the one being cleansed, and the priest shall make atonement for him before the Lord. [14:19] And the priest shall make the offering, and make atonement for the one who is to be cleansed from his uncleanness, and afterward he shall slay the burnt offering. [14:20] And the priest shall bring out the burnt offering and the grain offering to the altar, and the priest shall make atonement for him, and he shall be clean. [14:21] And if he is poor and lif he cannot afford¹⁵⁰⁶, then he shall take one lamb for the guilt offering to be a wave offering to make atonement for himself, and one tenth of wheat flour mixed with oil for a grain offering, and a seyik⁵⁰⁷ oil, [14:22] And two turtledoves or two young pigeons, whichever he can afford¹⁵⁰⁸;

_

⁴⁹⁷ The capacity of a bucket (CKED: 348).

⁴⁹⁸ Lit. 'them', see B.Heb אָתָם; lit. 'them'.

⁴⁹⁹ The capacity of a bucket (CKED: 348).

בתן Lit. 'he shall give', see B.Heb נתן; lit. 'he gives'.

⁵⁰¹ Lit. 'its soft |place|'.

⁵⁰² The capacity of a bucket (CKED: 348).

בימן Lit. 'he shall give', see B.Heb יָמֵן; lit. 'he shall give'.

⁵⁰⁴ Lit. 'its soft |place|'.

⁵⁰⁵ Lit. 'he shall give', see B.Heb יָתוּ; lit. 'he shall give'.

⁵⁰⁶ Lit. 'if his power does not suffice'.

⁵⁰⁷ The capacity of a bucket (CKED: 348).

⁵⁰⁸ Lit. 'to what if his power suffices'.

and one shall be a sin offering and one shall be a burnt offering. [14:23] And he shall bring them on the eighth day for his cleansing to the priest to the door

106 vo

of the tent of meeting before the Lord. [14:24] And the priest shall take the lamb of the guilt offering and the $seyik^{509}$ oil, and the priest shall wave them as a wave offering before the Lord. [14:25] And he shall slay the lamb of the guilt offering, and the priest shall take some of the blood of the guilt offering, and the priest shall put 1510 it on the lobe 511 of the right ear of the one who is to be cleansed, and on the thumb of his right hand, and on the big toe of his right foot. [14:26] And the priest shall pour some of the oil into the palm of his (own) left hand. [14:27] And the priest shall sprinkle with his right finger some of the oil that is on his left palm seven times before the Lord. [14:28] And the priest shall put⁵¹² some of the oil that is on his palm, on the lobe⁵¹³ of the right ear of the one who is to be cleansed, and on the thumb of his right hand, and on the big toe of his right foot, over the place of the blood of the guilt offering. [14:29] And the shall put 1514 the remnant of the oil that is on the priest's palm on the head of the one who is to be cleansed, to make atonement for him before the Lord. [14:30] And he shall offer⁵¹⁵ one of the turtledoves or of the young pigeons, whichever he can afford¹⁵¹⁶. [14:31] Whichever he can afford¹⁵¹⁷: one for a sin offering and the other for a burnt offering with the grain offering. Then the priest shall make atonement for the one who is to be cleansed before the Lord. [14:32] This is the law of him in whom is the disease of leprosy, and who cannot afford¹⁵¹⁸ for his cleansing'. [14:33] And the Lord spoke to Moses and to Aaron, saying, [14:34] 'If you come to the land of Canaan which I am giving to you for a possession, and if I give the disease of leprosy to a house of the land of your possession, [14:35] Then the one who owns the house shall come and explain the priest, saying, "It appeared like a disease to me in the house." [14:36] Then the priest shall command and they shall empty the house before the priest comes to see the disease, so that all that is in the house shall not become unclean, and, after that, the priest shall come to see the house. [14:37] And he shall look on the disease, and, lo, if

⁵⁰⁹ The capacity of a bucket (CKED: 348).

בתן Lit. 'he shall give', see B.Heb נתן; lit. 'he gives'.

⁵¹¹ Lit. 'its soft |place|'.

בתן Lit. 'he gives', see B.Heb נָתַן; lit. 'he gives'.

⁵¹³ Lit. 'its soft |place|'.

⁵¹⁴ Lit. 'he shall give', see B.Heb יָתוֹ; lit. 'he shall give'.

⁵¹⁵ Lit. 'do', see B.Heb צְשֶׁה; 'do'

⁵¹⁶ Lit. 'from what if his power suffices'.

⁵¹⁷ Lit. 'to what if his power suffices'.

⁵¹⁸ Lit. 'if his power does not suffice'.

the disease is on the walls of the house |with| strips⁵¹⁹ |appearing| greenish⁵²⁰ or reddish⁵²¹ and their appearances are deeper⁵²² than the wall. [14:38] Then the priest shall go out of the house to the door of the house and shut the house up seven days. [14:39] And the priest shall come back

107 ro

on the seventh day, and if he sees, and, lo, the disease was spread in the walls of the house. [14:40] Then the priest shall command and they shall pull out the stones in which the disease is and they shall cast them into an unclean place outside the city. [14:41] And he shall scrape⁵²³ the house inside round about, and they shall pour out the earth that they scraped⁵²⁴ outside the city into an unclean place. [14:42] And they shall take other stones, and bring them to the place of those stones, and he shall take other earth and shall smear the house. [14:43] And if the disease comes back and appears⁵²⁵ in the house after that he has taken away¹⁵²⁶ the stones and after he has scraped⁵²⁷ the house, and after he has smeared it, [14:44] Then if the priest comes and sees, and, lo, the disease has spread in the house: it is a pain causing leprosy in the house, it is unclean. [14:45] And he shall destroy the house, its stones and its woods and all the earth of the house, and he shall take them out of the camp into an unclean place. [14:46] And the one who comes into the house during all the days when it is shut up, shall be unclean until the evening. [14:47] And the one who lies in the house shall wash his clothes, and the one who eats in the house shall wash his clothes. [14:48] And if the priest comes⁵²⁸ in and looks, and, lo, the disease has not spread in the house after the house was smeared, then the priest shall pronounce⁵²⁹ the house clean, because the disease was healed. [14:49] And to cleanse⁵³⁰ the house, he shall take two birds, and cypress wood, and scarlet silk, and hyssop. [14:50] And he shall slay one of the birds over an earthen vessel over sweet water. [14:51] And he shall take

⁵¹⁹ Lit. 'road road'.

⁵²⁰ Lit. 'greens'. Probably a calque of B.Heb יָרקרקֹת; lit. 'greenish ones'.

⁵²¹ Lit. 'reds'. Probably a calque of B.Heb אָדְמְדָּמִת; lit. 'reddish ones'.

⁵²² Lit. 'low'. Probably a calque of B.Heb שָׁפָל; lit. 'low'.

⁵²³ Lit. 'break'.

⁵²⁴ Lit. 'broke'.

ברה Lit. 'buds'. Probably a calque of B.Heb פָּרַה; lit. 'buds'.

⁵²⁶ Lit. 'pulled out'. Probably a calque of B.Heb הלץ; lit. 'pulled out'.

⁵²⁷ Lit. 'broke'.

⁵²⁸ Lit. 'coming if he comes'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb אָבּר בּאֹ יָבאׁ lit. 'if to come he is coming'.

⁵²⁹ Lit. 'make'.

⁵³⁰ Lit. 'to sprinkle'.

the cypress wood, and the hyssop, and the scarlet silk, and the living⁵³¹ bird, and dip them in the blood of the slaughtered bird, and in the sweet waters, and sprinkle it at the house seven times. [14:52] And he shall cleanse⁵³² the house with the blood of the bird, and with the sweet water, and with the living⁵³³ bird, and with the cypress wood, and with the hyssop, and with the scarlet silk. [14:53] But he shall let go the living⁵³⁴ bird out of the city over the surface of the field, and make atonement for the house, and it shall be clean'. [14:54] This is the law for all kind of disease of leprosy, and for a scab, [14:55] And for the leprosy

107 vo

of a garment, and for a house, [14:56] And for a swelling, and for a lichen, and for a spot, [14:57] To teach Lwhen it is unclean 1535, and Lwhen it is clean 1536; this is the law of the disease of leprosy.

Chapter 15

[15:1] And the Lord spoke to Moses and to Aaron, saying, [15:2] 'Speak to the sons of Israel and say to them, "If a person⁵³⁷ has a discharge out of his flesh, |because of| his discharge he is unclean. [15:3] And this shall be his uncleanness if the liquid has flowed from his discharge, the discharge from his lexual organ¹⁵³⁸ whether it has flowed thickly and covered his lexual organ¹⁵³⁹ from his discharge, it is unclean. [15:4] If he lies on any bed which has the discharge, it shall be unclean, and if he sits on any item⁵⁴⁰ |which has the discharge|, it shall be unclean. [15:5] And if his head touches his bed, he shall wash his clothes and himself with water, and he shall be unclean until the evening, [15:6] And if the one who sits on |any| item on which the one who has the discharge sits, he shall wash his clothes and shall wash himself with water, and shall be unclean until the evening. [15:7] And the one who touches on the flesh of the one who has the discharge, he shall wash his clothes and shall wash himself with water, and shall be unclean until the evening. [15:8] If the one who has the discharge spits on him who is clean, he shall wash his clothes and wash himself with water, and shall be unclean until the evening.

⁵³¹ Lit. 'wild'.

⁵³² Lit. 'to sprinkle'.

⁵³³ Lit. 'wild'.

⁵³⁴ Lit. 'wild'.

⁵³⁵ Lit. 'the unclean time'.

⁵³⁶ Lit. 'the clean time'.

⁵³⁷ Lit. 'person person'. Probably a calque of B.Heb אָישׁ אִישׁ, lit. 'person person' (see 2.3.1.1.).

⁵³⁸ Lit. 'his shameful flesh'.

⁵³⁹ Lit. 'his shameful flesh'.

⁵⁴⁰ Lit. 'vessel'.

[15:9] And any saddle which was used⁵⁴¹ by the one who has the discharge, it shall be unclean. [15:10] And anybody who touches anything that was under him, he shall be unclean until the evening, and the one who carries those |things| shall wash his clothes and wash himself with water, he shall be unclean until the evening. [15:11] And anyone whom the one who has the discharge touches, and has not washed himself with water, he shall wash his clothes and wash himself with water, and he shall be unclean until the evening. [15:12] And if the one who has the discharge touches an earthen vessel, it shall be broken, and every wooden vessel shall be washed with water. [15:13] And if the one who has a discharge tis cleansed of his discharge, then he shall count for himself seven days for his cleansing, and wash his clothes, and wash his flesh in sweet water, and he shall be clean. [15:14] And on the eighth day he shall take for himself two turtledoves or two young pigeons and come before the Lord to the door of the tent of meeting, and give them to the priest.

108 ro

[15:15] And the priest shall offer⁵⁴³ them: one for a sin offering, and the other for a burnt offering. And the priest shall make atonement for him before the Lord for his discharge. [15:16] And if a drop of a man's semen comes out from him, he shall wash all his flesh with water and be unclean until the evening, [15:17] And if there exists a drop of semen on any garment and any leather, it shall be washed with water, and it shall be unclean until the evening. [15:18] And if a man who has a drop of semen lies with a woman, they shall wash themselves with water and be unclean until the evening. [15:19] And if a woman has a discharge, if her discharge from her flesh is blood, there shall be seven days tfor her menstruation¹⁵⁴⁴, and anyone who touches her shall be unclean until the evening. [15:20] And anything on which she lies during her menstrual impurity shall be unclean and anything on which she sits shall be unclean. [15:21] And anyone who touches her bed shall wash his clothes and wash himself with water, and he shall be unclean until the evening. [15:22] And anyone who touches any kind of vessel on which she sits shall wash his clothes and wash himself with water, and he shall be unclean until the evening. [15:23] And if it is on her bed, or on any vessel on which she is sitting, when he touches it, he shall be unclean until the evening. [15:24] And if any man lies⁵⁴⁵ with her and her

⁵⁴¹ Lit. 'be jumped'; 'be leaped'.

⁵⁴² Lit. 'remains clean'.

⁵⁴³ Lit. 'do', see B.Heb צְּשֶׂה; 'do'.

⁵⁴⁴ Lit. 'for her being away', see B.Heb בַּנְדָּתָה; lit. 'in isolation of her'.

⁵⁴⁵ Lit. 'lying if he lies'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb אָם שֶׁכֹב יִשְׁכַב , lit. 'if to lie he is lying'.

menstruation is on him, he shall be unclean seven days. And any bed on which he lies shall be unclean. [15:25] And if a woman whose Lblood flows 1546 for many days, not at the time of her menstruation, or if it flows beyond 547 the time of her menstruation, all the days of the discharge of her uncleanness shall be as the days of her menstruation, she shall be unclean. [15:26] If she lies on any bed during all the days of her discharge, it shall be to her as the bed of her menstruation, and any item 548 that she sits on shall be unclean, like the uncleanness of her menstruation. [15:27] And everyone who touches them shall be unclean and shall wash his clothes, and wash himself with water, and he shall be unclean until the evening. [15:28] And if she is cleansed 549 of her discharge, she shall count for herself seven days and after

$108 v^{o}$

that she shall be clean. [15:29] And on the eighth day she shall take for herself two turtledoves or two young pigeons, and bring them to the priest, to the door of the tent of meeting, [15:30] And the priest shall offer⁵⁵⁰: one for a sin offering, and one for a burnt offering, and the priest shall make atonement for her before the Lord for the discharge of her uncleanness. [15:31] And you shall separate the children of Israel from their uncleanness, lest they die with their uncleanness when they defile my tabernacle that is among them." [15:32] This is the law of him who has a discharge and of him from whom a drop of semen goes out, |so as| to become unclean by it, [15:33] And of her who is away in her menstruation, and of the one who has discharge, a discharge of a man, and of a woman, and of the person who lies with an unclean woman.

Chapter 16

[16:1] And the Lord spoke to Moses after two sons of Aaron has died, when they came forward before the Lord and died. [16:2] And the Lord said to Moses, 'Speak to Aaron, your brother, that⁵⁵¹ he shall not come at all times to the holy |place| within the veil before the cover which is over the ark, lest he die, because I will appear in the cloud over the cover. [16:3] Aaron shall come into the holy |place| with a young bull from the herd for a sin offering and a ram for a burnt offering. [16:4] He shall put on the holy linen shirt, and the linen trousers shall be upon

⁵⁴⁷ Lit. 'more'.

⁵⁴⁸ Lit. 'vessel'.

⁵⁴⁹ Lit. 'remains clean'.

⁵⁵⁰ Lit. 'do', see B.Heb צְשָׂה; 'do'.

⁵⁵¹ Lit. 'and', see B.Heb 1; lit. 'and'.

his flesh, and he shall be girded with a linen belt, and he shall be wrapped with the linen turban. These are holy garments and he shall wash his flesh with water, and put them on. [16:5] And he shall take two kids of the goats from the congregation of the sons of Israel for a sin offering, and one ram for a burnt offering. [16:6] And Aaron shall offer his bull of the sin offering which is for himself, and make atonement for himself and for his house. [16:7] And he shall take the two goats, and set them before the Lord at the door of the tent of meeting. [16:8] And Aaron shall cast 552

109 ro

lots over the two goats, one lot for the Lord and one lot for Azazel. [16:9] And Aaron shall offer the goat on which the lot for Lord arose and he shall make it a sin offering. [16:10] And the goat, on which the lot for Azazel arose, shall be stood alive before the Lord to make for atonement over it, to let him go to Azazel, to the wilderness. [16:11] And Aaron shall bring the bull of the sin offering which is for himself, and make atonement for himself, and for his house, and shall slay the bull of offering which is for himself. [16:12] And he shall take a censer⁵⁵³ full of embers of fire from off the altar before the Lord, and a handful⁵⁵⁴ of fragrant incense, and he shall bring it within the veil. [16:13] And he shall put⁵⁵⁵ the incense on the fire before the Lord, |so that|556 the cloud of the incense may cover the cover that is on the testimony, lest he die. [16:14] And he shall take of the blood of the calf and sprinkle it with his finger before the cover eastward, and he shall sprinkle of the blood with his finger before the cover seven times. [16:15] Then he shall slay the goat of the sin offering that is for the people, and bring its blood within the veil, and do |with| its blood as he did with the blood of the calf, and he shall sprinkle it on the cover and before the cover. [16:16] And he shall make atonement for the holy |place|, for the uncleanness of the children of Israel and for their deceits in all their sins, and he shall do so for the tent of meeting that dwells with them in the midst of their uncleanness. [16:17] And there shall be no man in the tent of meeting when he goes in to make atonement in the holy place until he comes out. He shall make atonement for himself, and for his house, and for all the congregation of Israel. [16:18] And he shall go out to the altar that is before the Lord, and make atonement for it and he shall take of the blood of the bull, and of the blood of the goat,

⁵⁵² Lit. 'he shall give', see B.Heb נָתַן; lit. 'he gives'.

⁵⁵³ Lit. 'firepan'.

⁵⁵⁴ Lit. 'his full palms'.

⁵⁵⁵ Lit. 'give'.

⁵⁵⁶ Lit. 'and'.

and put⁵⁵⁷ it on the corners of the altar round about. [16:19] And he shall sprinkle of the blood on it with his finger seven times, and cleanse it, and make it holy from the uncleanness of the sons of Israel. [16:20] And he shall stop atoning for the holy |place|, and the tent of meeting, and the altar. He shall bring the live goat.

$109 v^{0}$

[16:21] And Aaron shall lay both his hands on the head of the live goat, and confess over him all their iniquities. He shall put⁵⁵⁸ them on the head of the goat, and he shall send him away by the hand of a ready man into the wilderness. [16:22] And the goat shall carry on itself all their iniquities to a ruined place, and he shall let go the goat into the wilderness. [16:23] And Aaron shall come into the tent of meeting, and shall take off the linen garments which he had put on when he came into the holy place, and shall put them there. [16:24] And he shall wash his flesh with water in the holy |place|, and put on his garments, and come out, and make his burnt offering, and the burnt offering of the people, and make atonement for himself and for the people. [16:25] And the fat of the sin offering: he shall burn on the altar. [16:26] And the one who let go the goat to Azazel shall wash his clothes, and wash his flesh with water, and after that come into the camp. [16:27] And the bull of the sin offering, and the goat of the sin offering whose blood was brought in to make atonement in the holy |place|: one shall carry |them| out outside the camp, and with fire they shall burn their skins, and their flesh, and their dung. [16:28] And the one who burns them shall wash his clothes, and wash his flesh with water, and after that he shall come into the camp. [16:29] And this shall be a statute forever to you: in the seventh month, on the tenth day of the month, you shall afflict your souls and shall not work at all, either a local or a stranger that lives among you. [16:30] Because on that day the priest makes atonement for you to cleanse you from all your sins. You shall be clean before the Lord. [16:31] It is a Sabbath of solemn rest to you, and you shall afflict your souls by a statute forever¹⁵⁶⁰. [16:32] And the priest whom |one| shall anoint and who will fill his hand to serve as a priest in his father's place shall make atonement and put on the linen clothes, the holy clothes. [16:33] And he shall make atonement for the holy sanctuary, and for the tent of meeting and he shall make atonement for the altar, and he shall make atonement for the priests and for

בון 'he shall give', see B.Heb נָתוֹ; lit. 'he gives'.

ביל Lit. 'he shall give', see B.Heb נָתוֹ; lit. 'he gives'.

היים Lit. 'to statue of world'. Probably a calque of B.Heb לחשת עוֹלָם; lit. 'to statue of world'.

[.] Statue of world'. Probably a calque of B.Heb הקת עולם; lit. 'statue of world'.

all the people of the congregation. [16:34] And this shall be, a _Lstatute forever¹⁵⁶¹ to you to make atonement for the sons of Israel for all their sins once a year'. And he did as the Lord commanded Moses.

110 ro

Chapter 17

[17:1] And the Lord spoke to Moses, saying, [17:2] 'Speak to Aaron, and to his sons, and to all the sons of Israel, and say to them, "This is the word which the Lord has commanded, saying, [17:3] 'If any man¹⁵⁶² of the congregation of Israel slays cattle or a lamb or a goat in the camp, or slays outside the camp, [17:4] And if he does not bring it to the door of the tent of meeting to offer an offering to the Lord before the tabernacle of the Lord, bloodshed⁵⁶³ will be counted to that person, as if he shed the blood of someone, and that person shall be cut off from among his people, [17:5] Therefore the children of Israel shall bring their sacrifices (their shechitas) which they offer (shechita) over the surface of the fields, and they shall bring them to the Lord, to the door of the tent of meeting, to the priest, and sacrifice them for peace offerings to the Lord. [17:6] And the priest shall sprinkle the blood on the altar of the Lord at the door of the tent of meeting, and burn the fat for an aroma of acceptance to the Lord. [17:7] And they shall no more offer their sacrifices to devils (to their goat faces¹⁵⁶⁴) after whom they are going astray. This shall be a statute forever 1565 to them throughout their generations. [17:8] And you shall say them, "If lany person¹⁵⁶⁶ from the congregation of Israel or of the strangers lives among them brings out a burn offering or a sacrifice, [17:9] And if he does not bring it to the door of the tent of meeting to make it to the Lord, then that man shall be cut off from among his people. [17:10] If any person 1567 from the congregation of Israel or of the strangers who live among them eats any blood, then I will set 1568 my wrath against that soul who eats blood and will cut him off from among his people. [17:11] Because the soul of the flesh is |covered| in blood and I have given it to you on the altar to make atonement for your souls. Because it is the blood that makes atonement for the soul." [17:12] Therefore I have said to the children of Israel: "No soul

⁵⁶¹ Lit. 'to statue of world'. Probably a calque of B.Heb לְּחָשָּת עוֹלָם; lit. 'statue of world'.

⁵⁶² Lit. 'person person'. Probably a calque of B.Heb אָלישׁ אָיל lit. 'person person' (see 2.3.1.1.).

⁵⁶³ Lit. 'blood', see B.Heb [7]; lit. 'blood'.

⁵⁶⁴ Lit. 'to their kid faces' (here the 'kid' refers the animal), see B.Heb לְשִׁעִירָם; lit. 'to he-male goats'.

⁵⁶⁵ Lit. 'statue for life', see B.Heb וחקת עולם; lit. 'statue of world'.

⁵⁶⁶ Lit. 'person person'. Probably a calque of B.Heb איש איש (lit. 'person person' (see 2.3.1.1.).

⁵⁶⁷ Lit. 'person person'. Probably a calque of B.Heb אָישׁ אָישׁ; lit. 'person person' (see 2.3.1.1.).

⁵⁶⁸ Lit. 'and I give', see B.Heb ונתתי; lit. 'and I give'.

among you shall eat blood, nor shall the stranger that lives among you eat blood." [17:13] And if Lany person¹⁵⁶⁹ from the children of Israel or from the strangers who live among them hunts game, either an animal or a bird that may be eaten and if he pours out its blood, he shall cover it with earth. [17:14] Because the life of all living creatures

$110 v^0$

is its blood along with its soul (mixed), and I said to the children of Israel: "You shall not eat the blood of any living creatures, because the life of all living creatures is its blood; every one of those who eat it shall be cut off." [17:15] And every soul who (by mistake) eats *nevala*⁵⁷⁰ or *terefa*⁵⁷¹, whether it is a local or a stranger, he shall wash his clothes and shall wash his flesh with water and shall be unclean until the evening, then he shall be clean. [17:16] But if he does not wash his clothes and his flesh, then he will bear his iniquity".

Chapter 18

[18:1] And the Lord spoke to Moses, saying, [18:2] 'Speak to the sons of Israel and say to them, "I am the Lord your God. [18:3] You shall not do like what is done in the land of Egypt¹⁵⁷², wherein you dwelt, and you shall not do like what is done in the land of Canaan¹⁵⁷³, to where I am bringing you, and you shall not walk by their statutes. [18:4] You shall do my judgments and keep my statutes to walk by them. I am the Lord your God. [18:5] And you shall keep my statutes and my judgments. If a man does them, then he will live by them. I am the Lord. [18:6] Lany man¹⁵⁷⁴ to any close |relative| of his flesh: you shall not come close to uncover nakedness⁵⁷⁵. I am the Lord. [18:7] You shall not uncover the nakedness⁵⁷⁶ of your father, that is to say, the nakedness⁵⁷⁷ of your mother. She is your mother. You shall not uncover her nakedness⁵⁷⁸. [18:8] You shall not uncover the nakedness⁵⁷⁹ of your father's wife. It is your

⁵⁶⁹ Lit. 'person person'. Probably a calque of B.Heb אָישׁ אָישׁ; lit. 'person person' (see 2.3.1.1.).

⁵⁷⁰ An animal that died a natural death (CEDHL: 402).

⁵⁷¹ An animal torn by wild beasts (CEDHL: 252).

⁵⁷² Lit. 'as deed of the place of Egypt'. Probably a calque of B.Heb, e.g., בָּמֶעֲשֵׂה אֶרֶץ־מָצְרֵיִם; lit. 'as deed of land of Egypt'.

⁵⁷³ Lit. 'as deed of the place of Canaan'. Probably a calque of B.Heb, e.g., כְּמַעֲשֵׂה אֶרֶץ־כְּנַעַן; lit. 'as deed of land of Egypt'.

בני 1 ביט איש איש איים 1. Frobably a calque of B.Heb אָישׁ אָישׁ; lit. 'person person' (see 2.3.1.1.).

⁵⁷⁵ Lit. 'shame'; 'defect'.

⁵⁷⁶ Lit. 'shame'; 'defect'.

⁵⁷⁷ Lit. 'shame'; 'defect'.

⁵⁷⁸ Lit. 'shame'; 'defect'.

⁵⁷⁹ Lit. 'shame'; 'defect'.

father's nakedness⁵⁸⁰. [18:9] The nakedness⁵⁸¹ of your sister, the daughter of your father or the daughter of your mother, whether she was born at home (from your father) or born outside (from somebody else): you shall not uncover her nakedness⁵⁸². [18:10] The nakedness⁵⁸³ of your son's daughter or of your daughter's daughter: you shall not uncover their nakedness⁵⁸⁴. Because they are your nakedness⁵⁸⁵. [18:11] The nakedness⁵⁸⁶ of your father's wife's daughter: she is regarded as begotten of your father. She is your sister, you shall not uncover her nakedness⁵⁸⁷. [18:12] You shall not uncover the nakedness⁵⁸⁸ of your father's sister, because she is a relative of your father. [18:13] You shall not uncover the nakedness⁵⁸⁹ of your mother's sister, because she is your mother's relative. [18:14] You shall not uncover the nakedness⁵⁹⁰ of your father's brother, (that is to say) you shall not approach to his wife, she is your aunt-in-law. [18:15] You shall not uncover the nakedness⁵⁹¹ of your daughter-in-law. She is your son's wife. You shall not uncover her nakedness⁵⁹². [18:16] You shall not uncover the nakedness⁵⁹³ of your brother's wife. It is your brother's nakedness⁵⁹⁴. [18:17] You shall not uncover the nakedness⁵⁹⁵ of a woman and her daughter. You shall not take either her son's daughter or her daughter's daughter. You shall not uncover her nakedness⁵⁹⁶; they are relatives.

111 rº

It is an abominable |thing| (to take). [18:18] And you shall not take a woman to her sister to make her a co-wife to uncover her⁵⁹⁷ nakedness⁵⁹⁸ before her⁵⁹⁹ when she⁶⁰⁰ is alive⁶⁰¹. [18:19] And you shall not come close to a woman in the time of her menstrual uncleanness to uncover

⁵⁸⁰ Lit. 'shame'; 'defect'.

⁵⁸¹ Lit. 'shame'; 'defect'.

⁵⁸² Lit. 'shames'; 'defects'.

⁵⁸³ Lit. 'shame'; 'defect'.
584 Lit. 'shame'; 'defect'.

⁵⁸⁵ Lit. 'shame'; 'defect'. 586 Lit. 'shame'; 'defect'.

⁵⁸⁷ Lit. 'shame'; 'defect'.

⁵⁸⁸ Lit. 'shame'; 'defect'.

⁵⁸⁹ Lit. 'shame'; 'defect'.

⁵⁹⁰ Lit. 'shame'; 'defect'.

⁵⁹¹ Lit. 'shame'; 'defect'.

⁵⁹² Lit. 'shame'; 'defect'.

⁵⁹³ Lit. 'shame'; 'defect'.

⁵⁹⁴ Lit. 'shame'; 'defect'.

⁵⁹⁵ Lit. 'shame'; 'defect'.

⁵⁹⁶ Lit. 'shame'; 'defect'.

⁵⁹⁷ The co-wife's.

⁵⁹⁸ Lit. 'shame'; 'defect'.

⁵⁹⁹ The first wife.

⁶⁰⁰ The first wife

⁶⁰¹ Lit. 'healthy'.

her nakedness⁶⁰². [18:20] And you shall not give your bed to your friend's (when that person is healthy¹⁶⁰³) wife for offspring, to become unclean by it. [18:21] And you shall not let¹⁶⁰⁴ any of your offspring pass through the fire to Molech, neither disrespect the name of your God. I am the Lord. [18:22] And you shall not lie with a man like lying with a woman. It is an abomination. [18:23] And you shall not give your bed to any animal to become unclean by it, neither shall |any| woman stand before an animal to join with it. It is an abomination. [18:24] You shall not become unclean by any of these, because by all these the nations that I am driving out before you have become unclean. [18:25] And the land has become unclean, therefore I thought about its iniquity and the land vomited out its inhabitants. [18:26] And you shall keep my statutes and my judgments, and shall not do any of these abominations, neither the local or the stranger that lives among you. [18:27] Because all the people of the land who were before you did all these abominable things, and the land became unclean. [18:28] So the land shall not vomit you out when you make it unclean, as it vomited out the nations that were before you. [18:29] Because, trully, anyone who does (secretly) any of these abominable things: the souls that did those will be cut off from among their nations. [18:30] And you shall keep my testament not to do any one of these abominable laws which were done before you, and you shall not become unclean by them. I am the Lord your God.

Chapter 19

[19:1] And the Lord spoke to Moses, saying, [19:2] 'Speak to all the congregation of the sons of Israel and say to them, "Be holy, because I, the Lord your God, am holy.

$111 v^{0}$

[19:3] Every person shall fear his mother and his father, and you shall keep my Sabbaths. I am the Lord your God. [19:4] Do not turn to idols⁶⁰⁵ and you shall not make to yourselves molten idols. I am the Lord your God. [19:5] And if you do a sacrifice of peace offerings before the Lord, you shall offer it at your will. [19:6] It shall be eaten on the day that you offer it, and the Lord tay¹⁶⁰⁶, and the remainder until the third day, it shall be burned in the fire. [19:7] And Lif it is eaten at all¹⁶⁰⁷ on the third day, it is a profanation (it is unfit), that peace offering shall not

⁶⁰² Lit. 'shame'; 'defect'.

⁶⁰³ Lit. 'in the health of your opposite |person|'.

⁶⁰⁴ Lit. 'you shall not give', see B.Heb לא־תְּהָן; lit. 'you shall not give'.

⁶⁰⁵ Lit. 'nonexistences'.

⁶⁰⁶ Lit. 'from its morning', see B.Heb ממחרת; lit. 'from morrow'.

be accepted. [19:8] And its eaters (those who have been eaten it by choice)¹⁶⁰⁸ bear their iniquity, because he has despised the Holy of the Lord, and that soul will be cut off from among his nation. [19:9] And when you reap the harvest of your land, you shall not finish harvesting the edge of your field, and you shall not collect the leftover ears of your harvest. [19:10] And you shall not glean your vineyard and you shall not collect the seed of your vineyard, but you shall leave them to the poor and the stranger. I am the Lord your God. [19:11] You shall not steal, and not deceive, and nobody shall cheat the other one. [19:12] And you shall not swear by my name falsely, and shall not disrespect the name of your God. I am the Lord. [19:13] You shall not oppress your friend and you shall not rob, a hired worker's wages shall not stay with you¹⁶⁰⁹ until the morning. [19:14] You shall not curse the deaf, and you shall not put⁶¹⁰ an obstacle before the blind, and you shall fear your God. I am the Lord. [19:15] You shall not do injustice in judgement, you shall not contempt the poor¹⁶¹¹, and you shall not favor the rich¹⁶¹². You shall judge your fellow with justice. [19:16] You shall not walk |as| a slanderer among your people, you shall not stand by at your fellow's blood. I am the Lord. [19:17] You shall not regard your brother in your heart as an enemy, you shall surely admonish your friend, lest you bear iniquity because of him. [19:18] You shall not take revenge, and you shall not hate⁶¹⁴ the sons of your people, and you shall love your friend as |you do| yourself. I am the Lord. [19:19] You shall keep my statutes. You shall not mate your cattle two kinds. You shall not sow your field two kinds. And a garment of two kinds, shaatnez⁶¹⁵: it shall not go on you. [19:20] And if a person lies sexually 616 with a woman, who is slave and engaged to a man 1617 and she has not been redeemed at all 1618, or the price of her freedom has not been given to her: there shall be an investigation, they shall not be killed if she was not free. [19:21] And he shall bring his guilt offering to the Lord, to the door of the tent of meeting:

-

⁶⁰⁸ Lit. 'knowingly'.

⁶⁰⁹ Lit. 'be put to your side'.

Lit. 'you shall not give', see B.Heb לא תַּהָן; lit. 'you shall not give'.

⁶¹¹ Lit. 'you shall not do face the faces of the poor', see B.Heb לֹא־תַשָּׂא כְּנֵי־דָל; lit. 'you shall not lift up the faces of poor one'.

⁶¹² Lit. 'you shall not respect the faces of the rich', see B.Heb לֹא מֶהְדֵּר פְּנֵי גָדוֹל; lit. 'you shall not honor the faces of great one'.

⁶¹³ Lit. 'admonishing, you shall admonish'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), e.g., B.Heb הוֹכָח תּוֹכִים, tit. 'to correct you shall correct'.

⁶¹⁴ Lit. 'drive away hatred'.

⁶¹⁵ Mixed stuff; mixed linen (CKED: 370).

⁶¹⁶ Lit. 'spill of semen', see B.Heb שֶׁכְבַת־זֶרֵע; lit. 'emission of semen'.

⁶¹⁷ Lit. 'to his head'.

⁶¹⁸ Lit. 'redeeming, she has not been redeemed'. Probably a calque of B.Heb הָפְּדָה לֹא נִפְּדָּחָה; lit. 'to be ransomed she was not ransomed'.

a ram for a guilt offering. [19:22] And the priest shall make atonement for him with the ram of the guilt offering before the Lord for his sin which he has sinned, then it will be forgiven for him concerning his sin that he has sinned. [19:23] And if you come to the land, and plant any kind of fruit trees, you shall regard its uncircumcision, that is its fruit, as uncircumcised. It shall be |unclean for | three years to you, it shall not be eaten like the ones uncircumcised. [19:24] But in the fourth year all the fruit shall be holy, and offered as praises for the Lord. [19:25] And in the fifth year you shall eat of its fruit to increase its product for you. I am the Lord your God. [19:26] You shall not eat with the blood, you shall not do fortune-telling, and you shall not do soothsaying¹⁶¹⁹. [19:27] You shall not shave around¹⁶²⁰ the sides of your head and not weaken the edges of your beard. [19:28] You shall not make 1621 a tear in your flesh for the dead and not make a tattoo⁶²² on you. I am the Lord. [19:29] You shall not despise your daughter to make her a prostitute, lest the people of the land be mistaken and the land become full of adultery. [19:30] You shall keep my Sabbaths, and fear my sanctuary. I am the Lord. [19:31] You shall not turn to soothsayers and not seek the sorcerers to become unclean by them. I am the Lord your God. [19:32] You shall stand before an aged person, and you shall respect the face of an old man, and fear your God. I am the Lord. [19:33] And if a stranger lives with you in your land, you shall not oppress him. [19:34] The stranger that lives among you shall be to you like one of you and and you shall love him as |you do| yourself, because you |too| were strangers in the land of Egypt. I am the Lord your God. [19:35] You shall not do injustice in judgment, in measurements, in weights, in scales. [19:36] There shall be true scale, true stone, a true kebič⁶²³, and a true seyik⁶²⁴ for you. I am the Lord your God who brought you out of the land of Egypt. [19:37] And you shall keep all my statutes, and all my judgments, and do them. I am the Lord."

Chapter 20

[20:1] And the Lord spoke to Moses, saying, [20:2] 'And say to the sons of Israel, "LAny person¹⁶²⁵ of the sons of Israel, or of the strangers who live in Israel: if he gives of his offspring

⁶¹⁹ Lit. 'you shall not look at cloud'. Probably an influence of B.Heb לא חָעננו; lit. 'you shall not bring cloud'.

¹⁶²⁰ Lit. 'you shall not encircle', see B.Heb לא חקפו, lit. 'you shall not round'.

⁶²¹ Lit. 'you shall not give', see B.Heb לא תחנו; lit. 'you shall not give'.

בתבת קעקע brand-marked writing '. Probably a calque of B.Heb בתבת קעקע; lit. 'writing of brand-mark'.

⁶²³ An amount of wheat which was given to miller in exchange for grinding (ÖTS 3: 2513).

⁶²⁴ The capacity of a bucket (CKED: 348).

⁶²⁵ Lit. 'person person'. Probably a calque of B.Heb איש איש; lit. 'person person' (see 2.3.1.1.).

to Molech, the shall surely be killed¹⁶²⁶, the people of the land shall stone him with stones. [20:3] And I will set¹⁶²⁷ my wrath

112 vo

against that (the one who gives secretly) man, and I will cut him off from among his people, because he has given of his offspring to Molech, to make my sanctuary unclean and to despise my holy name. [20:4] And if the nation of the land close⁶²⁸ their eyes to that person when he openly gives of his offspring to Molech, not to kill him, [20:5] Then I will set 1629 my wrath against that man and against his family and will cut off him and all who go astray after him, to go astray after Molech, from among their people. [20:6] And if the soul turns to the soothsayers and the sorcerers, going astray after them, then I will set 1630 my wrath against that soul and will cut him off from among his people. [20:7] And you shall be consecrated⁶³¹, and |therefore| you shall be holy because I am the Lord your God. [20:8] And you shall keep my statutes, and do them, because I am the Lord who makes you holy. [20:9] For everyone⁶³² who curses his father or his mother, shall surely be killed 1633; he has cursed his father or his mother, his blood shall be on his head. [20:10] And if a person commits adultery with another man's wife, commits adultery with his friend's wife, the person who commits adultery and the women who commits adultery shall surely be killed 1634. [20:11] And if a person lies with his father's wife, |then| he has uncovered his father's nakedness⁶³⁵, both of them shall surely be killed¹⁶³⁶. Their blood shall be on their heads. [20:12] And if a person lies with his daughter-in-law, both of them shall surely be killed 1637, they have done a disgusting |thing|, their blood shall be on their

_

⁶²⁶ Lit. 'dying, he shall be killed'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb מות יוּמָת; lit. 'to be put to death he shall be put to death'.

⁶²⁷ Lit. 'And I will give', see B.Heb ואָני אָתַן; lit. 'And I shall give'.

⁶²⁸ Lit. 'covering, if they cover'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb אָם דָּעָלָם יַעָלִימָּוּ ; lit. 'if to obscure they are obscuring'.

⁶²⁹ Lit. 'And I will give', see B.Heb נְאֲנִי אֶתַן; lit. 'And I shall give'.

⁶³⁰ Lit. 'And I will give', see B.Heb נְאָנִי אָהַן; lit. 'And I shall give'.

⁶³¹ Lit. 'holy'.

⁶³² Lit. 'person person'. Probably a calque of B.Heb איש איש; lit. 'person person' (see 2.3.1.1.).

⁶³³ Lit. 'dying, he shall be killed'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb מות יומת; lit. 'to be put to death he shall be put to death'.

⁶³⁴ Lit. 'dying, he shall be killed'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb מוֹת יוּמָת; lit. 'to be put to death he shall be put to death'.
635 Lit. 'shame'; 'defect'.

⁶³⁶ Lit. 'dying, they shall be killed two of them'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb ממוֹת־יוּמְתוּ שְׁנֵיהֶם; lit. 'to be put to death they shall be put to death two of them'.

⁶³⁷ Lit. 'dying, they shall be killed two of them'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb ממות־יוּמְתוּ שְׁנֵיהֶם; lit. 'to be put to death they shall be put to death two of them'.

heads. [20:13] And if a person lies with a man like lying |with| a woman, Loth of them shall surely be killed¹⁶³⁸. They have done a disgusting |thing|, their blood |shall be| on their heads. [20:14] And if a person takes a wife and her mother: it is an abomination. They shall be burned with fire, he and one of them, so there shall be no abomination among you. [20:15] And if a person Lies with¹⁶³⁹ an animal, Lhe shall surely be killed¹⁶⁴⁰, and you shall kill the animal. [20:16] And if a woman approaches any four-footed animal to sleep with it then kill the woman and the animal. Lathey shall surely be killed¹⁶⁴¹, their blood |shall be| on their heads. [20:17] And if a person takes his sister, his father's daughter, or his mother's daughter, and sees her nakedness⁶⁴² and she sees his nakedness⁶⁴³, it is an abomination and they shall be cut off in the sight of the sons of

113 rº

their people. He has uncovered his sister's nakedness⁶⁴⁴, he shall bear his iniquity. [20:18] And if a man lies with a menstruating woman, and uncovers her nakedness⁶⁴⁵, he has exposed her flow⁶⁴⁶ and she has uncovered the flow⁶⁴⁷ of her blood. And both of them shall be cut off from among their people. [20:19] And you shall not uncover the nakedness⁶⁴⁸ of your mother's sister and of your father's sister. Whoever uncovered the nakedness⁶⁴⁹ of his relative: they shall bear their iniquity. [20:20] And if a man lies with the wife of his uncle, |then| he has uncovered his uncle's nakedness. They shall bear their iniquity, they shall die childless¹⁶⁵⁰. [20:21] And if a man takes the wife of his brother, it is like the menstrual impurity. He has uncovered his brother's nakedness, they shall be childless⁶⁵¹. [20:22] And you shall keep all my statutes, and all my judgments, and do them to that the place where I am bringing you to dwell therein

⁶

⁶³⁸ Lit. 'dying, they shall be killed'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb ממוֹת־יוּמְתוּ, lit. 'to be put to death they shall be put to death'.

⁶³⁹ Lit. 'gives his bed to'.

⁶⁴⁰ Lit. 'dying, he shall be killed'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb מוֹת יוּמָת; lit. 'to be put to death he shall be put to death'.

⁶⁴¹ Lit. 'dying, they shall be killed'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb ממות־יוּמְתוּ, lit. 'to be put to death they shall be put to death'.

⁶⁴² Lit. 'shame'; 'defect'.

⁶⁴³ Lit. 'shame'; 'defect'.

⁶⁴⁴ Lit. 'shame'; 'defect'.

⁶⁴⁵ Lit. 'shame'; 'defect'.

⁶⁴⁶ Lit. 'fountain; spring'. Probably a calque of B.Heb מָלֶר lit. 'fountain'.

⁶⁴⁷ Lit. 'fountain; spring'. Probably a calque of B.Heb וֹנָּלֶר lit. 'fountain'.

⁶⁴⁸ Lit. 'shame'; 'defect'.

⁶⁴⁹ Lit. 'shame'; 'defect'.

⁶⁵⁰ Lit. 'they shall die |being | depraved |of children|', see B.Heb צֶרִירִים יָמְתוּ; lit. 'heirless-ones they shall die'.

⁶⁵¹ Lit. 'deprived (plural) |of children|'. see B.Heb צֶרירִים יָהִיי; lit. 'they shall be heirless-ones'.

⁶⁵² Lit. 'and', see B.Heb 1; lit. 'and'.

may not vomit you out. [20:23] And you shall not walk by the statutes of the nation which I am driving out before you, because they have done all these |things|, and I have been disgusted by them, [20:24] But I have said to you: 'You shall inherit their land, and I shall give it to you to inherit it, a land flowing |with| milk and honey. I am the Lord your God, who has separated you from the nations. [20:25] You shall make a distinction between the clean animal and the unclean animal, and between the unclean bird and the clean |bird|, and you shall not make your souls abominable by animal or by bird or by all that moves on the ground which I have separated for you to pronounce⁶⁵³ unclean. [20:26] And you shall be holy to me, because I, the Lord, am holy and have separated you from other people to be mine. [20:27] And if a man or a woman is a sorcerer or a healer, they shall surely be killed¹⁶⁵⁴, they shall stone them with stones¹⁶⁵⁵, their blood |shall be| on their heads.

Chapter 21

[21:1] And the Lord said to Moses, 'Speak to the priests, the sons of Aaron, and say to them that "One shall not be unclean

113 vo

for a dead |person| among his people, [21:2] But for his relative, that is near unto him, for his father, or for his mother, or for his son, or for his daughter, or for his brother. [21:3] And for his unmarried sister who is near unto him who has not been a husband's, he may become unclean by touching her. [21:4] A husband shall not be unclean (by touching his wife) among his people to disrespect |himself|. [21:5] [They shall not pluck the hair on their heads 1656, and they shall not shave off the edge of their beards, nor shall they draw |any| lines in their flesh. [21:6] They shall be holies to their God and not disrespect the name of their God, because they are bringing forward the offerings of the Lord made by fire, the offerings of their God, and they shall be holy. [21:7] They shall not take a prostitute or a [depraved one 1657] and they shall not take a woman [divorced from 1658] her husband, because he is holy to his God. [21:8] And you shall sanctify him, because he is bringing forward the offerings of your God, he shall be holy

⁶⁵³ Lit. 'make'.

⁶⁵⁴ Lit. 'dying, they shall be killed'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb ממוֹת־יוּנְתְּה ; lit. 'to be put to death they shall be put to death'.

⁶⁵⁵ Lit. 'they shall stone with stone'. Probably a calque of B.Heb בָּאֶבֶן; lit. 'with stone they shall stone'.

⁶⁵⁷ Lit. 'lone with a vicious path'.

⁶⁵⁸ Lit. 'driven out by'. Probably a calque of B.Heb גרושה; lit. 'being driven out'.

to you, because I, the Lord, who makes you holy, am holy. [21:9] And if the daughter of a priest begins to go astray, she is defiling her father's priesthood. She shall be burned in fire. [21:10] And if the anointing oil is poured on the one who is the superior priest among his brothers and he fills his hand to wear the holy garments, he shall not uncover his head nor tear his clothes. [21:11] He shall not come to any dead soul, not even for his father and for his mother. He shall not be unclean. [21:12] He shall not go out of the sanctuary, and defile the sanctuary of his God, because the crown of the anointing oil of his God is on him. I am the Lord. [21:13] And he shall take a wife in her virginity. [21:14] A widow, or a divorced⁶⁵⁹ |woman|, or a depraved⁶⁶⁰ |woman|, or a prostitute: he shall not take these but he shall only take a virgin girl of his own people as a wife for himself. [21:15] And he shall not defile his offspring among his people, because I am the Lord who makes him holy." [21:16] And the Lord spoke to Moses, saying, [21:17] 'Speak to Aaron, saying, "If a person from your offspring throughout their generations has any defect¹⁶⁶¹, he shall not approach to offer the offerings of his God. [21:18] Because any person who has any defect 1662 shall not come near: |neither| a blind man, nor lame, nor the one who has any missing joints, nor the one who has any additional joints, [21:19] Or if a person has a broken foot or a missing hand, [21:20] Or he is humped or slender (shortie), or has a defect

114 rº

in his eyes¹⁶⁶³, or scabies, or a curved joint, or pierced testicles. [21:21] Any person Lwho has any defect¹⁶⁶⁴ from the offspring of Aaron the priest, shall not come near to bring forward the offerings of the Lord made by fire, if Lhe has any defect¹⁶⁶⁵, he shall not come near to bring forward the offering of his God, [21:22] But he may eat of the offerings of his God, from the holies of holies. [21:23] Only: he shall not come to the veil and approach the altar, because Lhe has a defect¹⁶⁶⁶, and he shall not disrespect my sanctuary, because I am the Lord who makes them holy." [21:24] And Moses spoke to Aaron and to his sons and to all the sons of Israel.

Chapter 22

Lit. 'expelled'. Probably a calque of B.Heb גָרוּשָׁה; lit. 'one being driven out'.

⁶⁶⁰ Lit. 'the one with a vicious path'.

¹⁶⁶¹ Lit. 'if becomes trick in him'. Probably a calque of B.Heb אֵשֶר יָהְיָה בוֹ מַנִּם; lit. 'who in him |there| is blemish'.

⁶⁶² Lit. 'in him trick'. Probably a calque of B.Heb בוֹ מוֹם; lit. 'in him blemish'.

⁶⁶³ Lit. 'one in whose eye white has fallen'.

⁶⁶⁴ Lit. 'in him trick'. Probably a calque of B.Heb בוֹ מַנּם; lit. 'in him blemish'.

⁶⁶⁵ Lit. 'trick in him'. Probably a calque of B.Heb בוֹ מוּם; lit. 'in him blemish'.

⁶⁶⁶ Lit. 'there is trick in him'. Probably a calque of B.Heb בוֹ מוֹם; lit. 'in him blemish'.

[22:1] And the Lord spoke to Moses, saying, [22:2] 'Speak to Aaron and his sons and they shall be separated from the holy things of the sons of Israel, that⁶⁶⁷ they shall not disrespect my holy name, because they are making |them| holy for me. I am the Lord. [22:3] Say to them, to your generations, "If any person from your offspring approaches to the holy |things| which the sons of Israel make holy for the Lord, and his uncleanness appears on them, he will be cut off from before me. I am the Lord. [22:4] Any person¹⁶⁶⁸ from the offspring of Aaron who is leprous or has a discharge shall not eat of the holy |things| until he is clean. And the one who touches any unclean creatures or a person from whom a drop of semen has come out 1669, [22:5] Or a soul, who touches any creature that swarms⁶⁷⁰ and becomes unclean by them, or touches a man and becomes unclean by them, whatever his uncleanness [22:6] If a soul touches such a thing¹⁶⁷², he shall be unclean until the evening and shall not eat of the holy |things| unless he washes his body with water. [22:7] And if the sun goes down, then he becomes clean, and afterward he shall eat of the holy |things|, because those are his food. [22:8] He shall not eat nevala⁶⁷³ nor terefa⁶⁷⁴ to become unclean by it. I am the Lord. [22:9] And they shall keep my testament, and they shall not bear the punishment of the sin for it and they die therefore⁶⁷⁵ if they disrespect it. I am the Lord who makes them holy. [22:10] And any stranger shall not eat the holy |thing|: a guest of the priest or a hired servant shall not eat the holy |thing|. [22:11] But if a priest buys a soul, the one who is bought

$114 v^0$

with his money may eat of it. And those who are born in his house: they may eat of his food. [22:12] And if a priest's daughter Lis married to a foreigner 1676, she may not eat of the offering part of the holy things. [22:13] And if a priest's daughter becomes a widow or divorced 677, and has no offspring and returns to her father's house, as in her youth, she shall eat of her father's bread. But no foreigner shall eat of it. [22:14] And if a person eats of a holy |thing| by mistake, he shall add its fifth part of it to it and give the holy |thing| to the priest. [22:15] And they shall

⁶⁶⁷ Lit. 'and', see B.Heb 3; lit. 'and'.

⁶⁶⁸ Lit. 'person person'. Probably a calque of B.Heb אָישׁ אָישׁ; lit. 'person person' (see 2.3.1.1.).

⁶⁶⁹ Lit. 'comes out'.

⁶⁷⁰ Lit. 'lays eggs'.

⁶⁷¹ Lit. 'to all his uncleanness', see B.Heb לָלל טָמָאָתו; lit. 'to any of his uncleanness'

⁶⁷² I it 'it'

⁶⁷³ An animal that died a natural death (CEDHL: 402).

⁶⁷⁴ An animal torn by wild beasts (CEDHL: 252).

⁶⁷⁵ Lit. 'with them', see B.Heb 12; lit. 'with it'.

⁶⁷⁶ Lit. 'becomes to |a| stranger', see B.Heb הָהָיָה לְאִישׁ; lit. 'she is becoming to |a| man'.

ה 'Eit. 'expelled', see B.Heb גרושה; lit. 'one being driven out'.

not disrespect the holy |things| of the sons of Israel which they separate to the Lord. [22:16] And they shall make them bear the iniquity for the guilt⁶⁷⁸ when they eat their holy |things|. Because I am the Lord who makes them holy." [22:17] And the Lord spoke to Moses, saying, [22:18] 'Speak to Aaron and his sons and all the sons of Israel and say to them, "If a person¹⁶⁷⁹ of the congregation of Israel or of the strangers in Israel offers his offering, for all his vows⁶⁸⁰ and for all his generosity offerings which he offers to the Lord for a burnt offering, [22:19] Together with your |own| will, |you shall bring forward| a healthy male of the cattle, of the lambs, or of the goats. [22:20] You shall not offer anything that that a blemish 1681. Because it will not be acceptable for you. [22:21] And if a person offers a peace offering to the Lord to dedicate a vow⁶⁸² or as generosity |offering| from the cattle or from the sheep, it shall be healthy to be accepted; there shall be no blemish⁶⁸³ in it. [22:22] Blind or broken or defective or disabled, or one with a curved joint, or one with a lichen: you shall not bring these as an offer to the Lord and you shall not give as an offering made by fire of them on the altar to the Lord. [22:23] A bull or a sheep that has any additional joints or any missing joints is you may do it for a generosity offering, but for a vow⁶⁸⁶, it will not be accepted. [22:24] And that which is crushed, or broken, or suffering, or cut: you shall not offer to the Lord, nor do it in your land. [22:25] And from a stranger's hand, you shall not offer an offering of your God of any of these, because their defect is in them, a blemish⁶⁸⁷ is in them; they will not be accepted for you." [22:6] And the Lord spoke to Moses, saying, [22:27] "If a bull or a lamb or a goat is born, then it shall be next to its mother seven days, and from the eighth day and thereafter it will be accepted |for you| as an offering by fire to the Lord.

115 rº

[22:28] And a bull or a sheep: you shall not slay it and its young Lon the same day¹⁶⁸⁸ [22:29] And if you offer a sacrifice of thanksgiving to the Lord, sacrifice it for your |own| will. [22:30] On that day it shall be eaten, you shall not leave |any| of it until morning. I am the Lord. [22:31]

_

⁶⁷⁸ Lit. 'regret',

Lit. 'person person'. Probably a calque of B.Heb אָישׁ אִישׁ (lit. 'person person' (see 2.3.1.1.).

⁶⁸⁰ Lit. 'intentions'.

⁶⁸¹ Lit. 'in him trick'. Probably a calque of B.Heb בוֹ מוּם; lit. 'in him blemish'.

⁶⁸² Lit. 'intention'.

⁶⁸³ Lit. 'trick', see B.Heb מום; lit. 'blemish'.

⁶⁸⁴ Lit. 'with extra'.

⁶⁸⁵ Lit. 'with absent'.

⁶⁸⁶ Lit. 'intention'.

⁶⁸⁷ Lit. 'trick', see B.Heb מום; lit. 'blemish'.

⁶⁸⁸ Lit. 'in one day'. Probably a calque of B.Heb בָּיוֹם אָחָד; lit. 'in day one'.

And you shall keep my commandments and do them. I am the Lord. [22:32] And you shall not disrespect my holy name, and I will be holy among the children of Israel; I am the Lord who sanctifies you, [22:33] Who brought you out of the land of Egypt to be your God, I am the Lord'.

Chapter 23

[23:1] And the Lord spoke to Moses, saying, [23:2] 'Speak to the sons of Israel and say to them, "The appointed times of the Lord, which you shall call holy convocations: these are my appointed times: [23:3] Six days work shall be done, and on the seventh day is a Sabbath of solemn rest, a holy convocation, do not do any work. It is the Sabbath to the Lord in all your dwelling places. [23:4] These are the appointed times of the Lord, the holy convocations which you shall call in their appointed times: [23:5] In the first month, on the fourteenth day of the month at twilight 1689, is the Passover to Lord. [23:6] And on the fifteenth day of this month is the feast of unleavened bread to the Lord. You shall eat unleavened bread seven days. [23:7] On the first day it shall be a holy convocation to you: you shall not do any work of service. [23:8] And you shall offer an offering by fire to the Lord seven days. In the seventh day is a holy convocation: you shall not do any work of service." [23:9] And the Lord spoke to Moses, saying, [23:10] 'Speak to the sons of Israel and say to them, "If you come to the land that I am giving you and reap its harvest, bring the sheaf of the first of your harvest to the priest. [23:11] And he shall wave the sheaf before the Lord, for your wish, on the next day after the Sabbath 1690 the priest shall offer it. [23:12] And that day when you wave the sheaf you shall do a healthy one-year-old lamb for a burnt offering to the Lord. [23:13] And its grain offering |shall be two parts of ten of wheat flour mixed with oil, an offering by fire to the Lord with an aroma of acceptance. And its drink offering |shall be| a wine offering, a fourth part of a sivik⁶⁹¹". [23:14] And you shall not eat bread or groats, or heads of barley

115 vo

until the selfsame day, until you have brought a offering of your God; it is a Lstatute forever¹⁶⁹² throughout your generations in all your dwellings. [23:15] And you shall count for yourselves

Lit. 'to between two evenings'. Probably a calque of B.Heb בֵּין הַעַרבַּיִם; lit. 'between the evenings'.

⁶⁹⁰ Lit. 'from the morning of the Sabbath'. Probably a calque of B.Heb מְמָּחֶרֵת הַשֶּׁבָּׁת; lit. 'from morrow of the Sabbath'

⁶⁹¹ A measure of liquid (CKED: 358).

⁶⁹² Lit. 'statue of world'. Probably a calque of B.Heb הָקָת עוֹלָם; lit. 'statue of world'.

1 from the next day after the Sabbath 1693, from the day that you brought the sheaf of wave offering: seven Sabbaths shall be complete. [23:16] You shall count until the day after the seventh Sabbath, fifty days and you shall bring forward a new grain offering to the Lord. [23:17] You shall bring from your dwelling places two breads for the wave offering, two parts of ten. They shall be of wheat flour, they shall be baked leavened, they are the firstfruits⁶⁹⁵ to the Lord. [23:18] And you shall offer with the bread seven healthy lambs one-year-old, and to these |you shall add a calf and two rams: they shall be a burnt offering to the Lord, with their grain offerings and their drink offerings, an offering by fire, |with| an aroma of acceptance to the Lord. [23:19] And you shall offer⁶⁹⁶ one kid of the goats for a sin offering and two one-yearold lambs for a sacrifice of peace offerings. [23:20] And the priest shall wave them with the bread of the firstfruits⁶⁹⁷ for a wave offering before the Lord with the two lambs. They shall be holy to the Lord, for the priest. [23:21] And you shall proclaim on the selfsame day, it may be a holy convocation to you: you shall not do any work of service. |It is| a statute forever for you in all your dwelling places throughout your generations. [23:22] And when you reap the harvest of your land, you shall not reap¹⁶⁹⁹ the edge of your field when you reap and you shall not collect the remains of your harvest: you shall leave them for a poor and for a stranger: I am the Lord your God." [23:23] And the Lord spoke to Moses, saying, [23:24] 'Speak to the children of Israel, saying, "In the seventh month, on the first day of the month, it shall be Sabbath, a memorial of praise⁷⁰⁰, a holy convocation for you. [23:25] You shall not do any work of service, and offer an offering by fire to the Lord." [23:26] And the Lord spoke to Moses, saying, [23:27] 'Only on the tenth day of this seventh month is the day of atonement: it shall be a holy convocation for you and you shall trouble your souls and offer an offering to the Lord. [23:28] And you shall not do any work on that very day, because it is a day of atonement, to make atonement for you before the Lord your God. [23:29] Because if any soul does not trouble himself on that very day, |then| he will be cut off from his people.

116 r°

-

⁶⁹⁴ Lit. 'the morning of '. Probably a calque of B.Heb מָמֶהֶרת הַשֶּׁבָּת, lit. 'from morrow of'.

⁶⁹⁵ Lit. 'the offering of firsts'.

⁶⁹⁶ Lit. 'do', see B.Heb עַשָּׁה; 'do'

⁶⁹⁷ Lit. 'firsts'.

⁶⁹⁸ Lit. 'statue of world'. Probably a calque of B.Heb הְקֵת עוֹלָם; lit. 'statue of world'.

⁶⁹⁹ Lit. 'you shall not finish', see B.Heb לא־תַכְלָה; lit. 'you shall not finish'.

^{700 &#}x27;Shout; cry', see B.Heb הרועה; 'shout; blast of war; the sound of trumpet; alarm'.

[23:30] And whoever does any work on that very day, I will destroy that soul from among his people. [23:31] You shall not do any work, it is a statute forever throughout your generations in all your dwelling places. [23:32] It shall be a Sabbath of solemn rest for you, and you shall trouble your souls. On the ninth day of the month at evening, from evening to evening you shall keep on your Sabbath⁷⁰¹. [23:33] And the Lord spoke to Moses, saying, [23:34] 'Speak to the sons of Israel, saying, "On the fifteenth day of this seventh month is the feast of Sukkoth, seven days to the Lord. [23:35] On the first day |it shall be| a holy convocation: you shall not do any work of service. [23:36] Seven days you shall bring forward an offering by fire to the Lord, on the eighth day there shall be a holy convocation to you and you shall bring forward an offering by fire to the Lord. It is a break 702; you shall not do any work of service. [23:37] These are the appointed times of the Lord which you shall call the holy convocations to offer an offering by fire to the Lord, a burnt offering, and a grain offering, a sacrifice offering, and drink offerings, on its appointed day, [23:38] Besides the Lord's Sabbaths, and besides your gifts, which you give to the Lord. [23:39] Only: on the fifteenth day of the seventh month, when you have gathered the produce of the land, celebrate the feast of the Lord seven days. On the first day is a Sabbath, and on the eighth day is a Sabbath. [23:40] And you shall take for yourselves before the first day the tree of choice fruits (for a booth), the leaves of date palm and the branches of thick leafy tree and willows of the river, and you shall rejoice before the Lord seven days. [23:41] And celebrate it as a feast to the Lord seven days in the year. It is a statute forever¹⁷⁰³ throughout your generations, celebrate it in the seventh month. [23:42] You shall dwell in booths seven days. All locals in Israel shall dwell in booths, [23:43] So that your generations may know that I made the sons of Israel dwell in shelters when I brought them out of the land of Egypt. I am the Lord your God". [23:44] And Moses spoke the appointed times of the Lord to the sons of Israel.

Chapter 24

[24:1] And the Lord spoke to Moses, saying, [24:2] 'Command the sons of Israel and they shall bring pure oil from beaten olives

116 vo

⁷⁰¹ Lit. 'do Sabbath your Sabbath'. Probably a calque of B.Heb תְּשֶׁבְּתוּ שֶׁבַּהְּכֶם lit. 'you shall sabbath (cease) your Sabbath'.

⁷⁰² Lit. 'delay; constipation'.

^{. &#}x27;statue of world'. Probably a calque of B.Heb הקת עולם; lit. 'statue of world',

for the lamp to you to burn the candle constantly. [24:3] Outside the veil of the testimony, in the tabernacle of meeting, Aaron shall arrange it from evening to morning before the Lord constantly to be a statute forever¹⁷⁰⁴ throughout your generations. [24:4] He shall arrange the candles on a clean candlestick before the Lord continually, [24:5] And you shall take wheat flour and bake twelve thick bread: each thick bread shall be two parts of ten. [24:6] And you shall set them in two rows, six in a row, on the clean table before the Lord. [24:7] And you shall put pure incense on |each| row, and it shall be for an offering, for the incense, an offering by fire to the Lord. [24:8] On every Sabbath day¹⁷⁰⁵ he shall arrange it before the Lord always, from the sons¹⁷⁰⁶ of Israel |by| an teverlasting covenant¹⁷⁰⁷. [24:9] And it shall be for Aaron and his sons, and they shall eat it in a holy place, because it is the holy of holies to him from the Lord's offerings by fire, a statute forever⁷⁰⁸. [24:10] And a son of an Israelite woman, |who was also| a son of an Egyptian man went out among the sons of Israel, and the son of the Israelite woman and an Israelite man fought in the camp. [24:11] And the Israelite woman's son specified the Name and cursed. And they brought him to Moses. And his mother's name was Shelomith, the daughter of Dibri, of the tribe of Dan. [24:12] And they put him under guard, to determine 1709 it for them, by the Lord's command. [24:13] And the Lord spoke to Moses, saying, [24:14] 'Bring the curser out of the camp and all who heard |him| shall lay their hands on his head and all the congregation shall stone him. [24:15] And you shall speak to the sons of Israel, saying, if a person⁷¹⁰ curses his God, then he shall bear his sin. [24:16] And the one who curses the Name, shall surely be put to death 1711. All the congregation shall certainly stone him¹⁷¹², as well the stranger, as the native: when he curses the Name, he shall be put to death. [24:17] And if a person kills any man's life, the shall surely be put to death [24:18] And the killer of an animal's life¹⁷¹⁴ shall pay for it: a life for a life. [24:19] And if a person

⁷⁰⁴ Lit. 'statue of world'. Probably a calque of B.Heb הַקְּח עוֹלָם; lit. 'statue of world'.

⁷⁰⁵ Lit. 'on the Sabbath day on the Sabbath day'. Probably a calque of B.Heb בְּיוֹם הַשֶּׁבָּת בְּיוֹם הַשֶּׁבָּת; lit. 'in day of the Sabbath in day of the Sabbath'.

⁷⁰⁶ Lit. 'the side of the sons'.

⁷⁰⁷ Lit. 'covenant of world'. Probably a calque of B.Heb בָּרִית עוֹלָם; lit. 'covenant of world'.

⁷⁰⁸ Lit. 'statue of world'. Probably a calque of B.Heb חָקַת עוֹלָם; lit. 'statue of world'.

⁷⁰⁹ Lit. 'to be brain'.

⁷¹⁰ Lit. 'person person'. Probably a calque of B.Heb אָישׁ אָיָל lit. 'person person' (see 2.3.1.1.).

The Lit. 'dying, he shall be killed'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb מוֹת יוֹמָת ; lit. 'to be put to death he shall be put to death'.

The Lit. 'stoning they shall stone'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb וְנוֹם יְרָגִוֹם יְרָגִוֹם יִרְנִמוֹם יִנְיִמוֹם וֹנוֹיִי יִרְנִמוֹם בּוֹיִם יִרְנִמוֹם בּוֹיִם יִרְנִמוֹם בּוֹיִם יִרְנִמוֹם בּוֹיִם יִרְנִמוֹם בּוֹיִם יִרְנִמוֹם בּוֹיִם יִרְנִמוֹם בּוֹיִם יִרְנִמוֹם בּוֹיִם יִרְנִמוֹם בּוֹיִם יִרְנִמוֹם בּוֹיִם יִרְנִמוֹם בּוֹיִם יִרְנִמוֹם בּוֹיִם יִרְנִמוֹם בּוֹיִם יִרְנִמוֹם בּוֹיִם בּוֹיִם בּוֹיִם בּוֹיִם בּוֹיִם בּוֹיִם בּוֹיִם בּוֹיִם בּוֹיִם בּוֹיִם בּוֹיִם בּוֹיִם בּוֹיִם בּיִּבְּיִם בּוֹיִם בּיִּבְּיִם בּיִּבְיִם בּיִּבְיִם בּיִּבְּיִם בּיִּבְּים בּיִּבְּים בּיִּבְים בּיִּבְּים בּיִּבְּים בּיִּבְים בּיִּבְים בּיִּבְים בּיִּבְים בּיִּבְים בּיִּבְים בּיִּבְים בּיִּבְים בּיִּבְים בּיִּבְים בּיִּבְים בּיִּבְים בּיִּבְים בּיִּבְים בּיִּבְּים בּיִּבְים בּיִּבְים בּיִּבְים בּיִּבְים בּיִּבְים בּיִבְּים בּיִּבְים בּיִּבְּים בּיִּבְּים בּיִּבְים בּיִּבְּים בּיִּבְים בּיִּבְים בּיִּבְּים בּיִּבְים בּיִּבְּים בּיִּבְּים בּים בּיִּבְּים בּיִּבְּים בּיִבְּים בּיִּבְּים בּיִּבְּים בּיִּבְּים בּיִבְּים בּיִבְּים בּיִבְּים בּיִבְּים בּיִבְּים בְּיִבְּים בְּיבְּים בּיִבְּים בּיִבְּים בּיִבְּים בְּיבְּים בְּיבְּים בְּיבְּים בּיִּבְּים בּיִּם בְּיבְּים בּים בּיבּים בּיבְּים בּיבְּים בּיבְּים בּיבְּים בְּיבְּים בּים בּיבּים בּיבּים בּים בּיבּים בְּיבְּים בְּיבְּים בְּיבְּים בְּיבְּים בְּיבְּים בְּיבְּים בְּים בְּיבְּים בְּיבְּים בְּיבְּים בְּיבִּים בְּיבְּים בְּיבְּים בּים בּיבּים בּיבּים בּיבּים בּיבּים בּיבּים בּיבּים בּיבּים בּיבּים בּיבְים בּיבְים בְּיבְים בְּיבִים בְּיבְים בְּיבְים בְּיבְים בְּיבְּים בְּיבְים בְּיבְים בְּיבְים בְּיבְים בְּיבְים בּיבְים בּיבּים בּיבּים בּיבּים בּיבְים בְּיבְים בְּיבְים בְּיבְּים בְּיבְים בְּיבְיבְים בְּיבְּים בְּיבְים בְּיבְּים בְּיבְּים בְּיבְים בְּיבְּים בְּיבְים בְּיבְים בּיבְּים בְּיבְים בּיבְים בְּיבּים בּיבְים בּיבְּים בּיבּים בּיבּים בּיבּים בּיבּים בּיבּים בּיבּים בּיבּים בּיבּים בּיבּים בּיבּים בּיבּים בּיבּים בּיבּים בּיבְיבּים בּיבְים בּיבְיבְים בּיבְיבּים בּיבְים בּיבְים בּיבְים בּיבְים

¹¹³ Lit. 'dying, he shall be killed'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb מות יומת; lit. 'to be put to death he shall be put to death'.

¹¹⁴ Lit. 'and |the| beater of animal's life', see B.Heb וְמַבָּה נָפָשׁ־בְּהַמָּה; lit. 'and one smithing of soul of beast'.

_Lcauses a blemish⁷⁷¹⁵ in his fellow: as he has done, so shall it be done to him. [24:20] _LFracture for fracture¹⁷¹⁶,

117 ro

Leye for eye¹⁷¹⁷, Ltooth for tooth¹⁷¹⁸: as Lhe has caused a blemish¹⁷¹⁹ in a man, so Lshall it be done¹⁷²⁰ to him. [24:21] LAnd the killer of an animal¹⁷²¹ shall pay for it, Land the killer of a man¹⁷²² shall be put to death. [24:22] There shall be one law for you, it shall be as well for the stranger, as for the native. Because I am the Lord your God'''. [24:23] And Moses spoke to the children of Israel and they brought the curser to the outside of the camp and stoned him with stones and the children of Israel did as the Lord commanded Moses.

Chapter 25

[25:1] And the Lord spoke to Moses on Mount Sinai, saying, [25:2] 'Speak to the sons of Israel and say to them, "If you come to the land that I am giving you, then the land shall have a remitting, a remitting to the Lord. [25:3] Six years you shall sow your field and six years you shall prune your vineyard and gather its produce. [25:4] And in the seventh year there shall be the Sabbath of solemn rest for the land, a remitting to the Lord. You shall not sow your field and prune your vineyard. [25:5] You shall not reap your wild harvest, and you shall not gather the grapes⁷²³ of your vineyard. There shall be a year of a remitting to the land. [25:6] And the remitting of the land shall be food⁷²⁴ for you, and for your servants and for maidservants, and for your hired hand, and for |those| guests who live with you. [25:7] And for your cattle, and for the beasts that are in your land: all its produce |shall be| to eat. [25:8] And you shall count for yourself seven |weeks of| remitting of years: seven years seven times. And the days of the seven |weeks of| remitting of years will be forty-nine years for you. [25:9] And |you shall blow the trumpet of crying¹⁷²⁵ in the seventh month, on the tenth day of the month: on the day of

⁷¹⁵ Lit. 'if he gives trick', see B.Heb יהן מום; lit. 'he is giving blemish'.

⁷¹⁶ Lit. 'fracture instead of fracture'. Probably a calque of B.Heb שֶבֶר תַּחָת שֶבֶר; lit. 'fracture instead of fracture'.

⁷¹⁷ Lit. 'eye instead of eye'. Probably a calque of B.Heb עין חחת עין; lit. 'eye instead of eye'.

⁷¹⁸ Lit. 'tooth instead of tooth'. Probably a calque of B.Heb שֵׁן תַּחַת שֵׁן; lit. 'tooth instead of tooth'.

Lit. 'if he gives trick', see B.Heb יָתוּן מוּם; lit. 'he is giving blemish'.

⁷²⁰ Lit. 'it shall be given'. Probably a calque of B.Heb ינתן; lit. 'it shall be given'.

¹⁷²¹ Lit. 'and |the| beater of animal', see B.Heb וְּמֶבֶה בְהָמֶה; lit. 'and one smithing of beast'.

T22 Lit. 'and |the| beater of man', see B.Heb וְמְכָּה אָדָם; lit. 'and one smithing of man'.

⁷²³ Lit. 'seeds'.

⁷²⁴ Lit. 'to eat'.

⁷²⁵ Lit. 'you shall pass the voice of the trumpet of crying'. Probably a calque of B.Heb הַעֲבַרְהָּ שׁוֹפַר הְרוּעָה; lit. 'you make pass the trumpet of blast'.

atonement Lyou shall blow the trumpet of crying ¹⁷²⁶ through all your land. [25:10] And you shall sanctify the year of the fiftieth year and proclaim liberty in the land to all its inhabitants. It is a jubilee, it shall be for you.

$117 v^{0}$

And you shall return every person to his possession, and every person to his family. [25:11] That year of the fiftieth year is a jubilee for you: you shall not sow, and not reap the wild ones, and not gather the grapes⁷²⁷ (to storehouses). [25:12] Because it is a jubilee, it shall be holy to you. You shall take the wild produce from the field |and| eat it. [25:13] In the year of this jubilee you shall return every person to his possession. [25:14] And if you make a sale 1728 to your friend or make a purchase from your friend 730, you shall not harass one his brother. [25:15] You shall buy from your fellow by the number of years after the jubilee (until the jubilee), by the number of years of crops he shall sell to you: [25:16] According to the multitude of years you shall increase its selling price 1731, and according to the fewness of years you shall decrease its selling price¹⁷³². Because he is selling you the crops for sale. [25:17] A person shall not torment one another, and you shall fear your God. Because I am the Lord your God. [25:18] And you shall do my statutes and keep my judgments and do them, then you will dwell in the land securely. [25:19] And the land will give its produce, and you will eat your fill, and dwell in it securely." [25:20] And if you say, "What shall we eat in the seventh year, behold, if we do not sow or gather in our produce?" [25:21] Then I will command my blessing on you in the sixth year, and it will produce⁷³³ a crop (ripe) for three years. [25:22] And you will sow in the eighth year and eat the old produce until the ninth year: you will eat the old one until its produce arrives. [25:23] And the land shall not be sold completely (for a lifetime), because the land is mine, you are only like strangers and like dwellers (you dwell) with me. [25:24] And in all the land of your possession you shall give a redemption for the land. [25:25] Because if your brother becomes poor and sells a part of his property, and his redeemer who is close to him

⁷²⁶ Lit. 'you shall pass the voice of the trumpet of crying'. Probably a calque of B.Heb הַּעֲבַרְתָּ שׁוֹפַר מְּרוּצָה; lit. 'you make pass the trumpet of blast'.

⁷²⁷ Lit. 'seeds'.

⁷²⁸ Lit. 'if you sale sale'. Probably a calque of B.Heb הְמְּכָּרוּ מְמְכָּרוּ, lit. 'you are selling sale'.

⁷²⁹ Lit. 'buying if you buy'. A construction which is usually attributed to a Biblical Hebrew influence (see 2.3.1.2.1.). However, the original text does not present the mentioned structure for this verse, see B.Heb קנה; lit. 'to buy'

^{. (}ניה the hand of your friend'. Probably a calque of B.Heb מָבֶּד עַמִיתָּד, lit. 'from hand of your companion'.

⁷³¹ Lit. 'sale'.

⁷³² Lit. 'sale'.

⁷³³ Lit. 'do'.

comes, then he shall redeem what his brother has sold. [25:26] And if a person has no redeemer, but Lis able to redeem it by himself¹⁷³⁴ and Lfinds enough¹⁷³⁵ Lfor his redemption¹⁷³⁶: [25:27] Then he shall count the years of his sale and restore the remainder to the person to whom

118 ro

he sold it, and he shall return to his possession. [25:28] And if the cannot afford to return it to him, his sale shall be in the hand of the one who bought it until the year of jubilee, and it shall come out in the jubilee, and he shall return to his possession. [25:29] And if a person sells a dwelling house in a walled city, he shall have a redemption period⁷³⁸ until the year of its sale ends. His redemption period⁷³⁹ shall be for a year. [25:30] And if it is not redeemed by the completion of a full year, then the house that is in the walled city¹⁷⁴⁰ shall completely (for a life time) be a residence to the one who bought it throughout his generations. It shall not be released in the jubilee. [25:31] And the village houses that have no wall around them shall be counted as the fields of the country. It shall have redemption, and the redemption shall come out in the jubilee. [25:32] And the cities of the Levites, the houses of the cities of their possession: the Levites shall have eternal redemption. [25:33] And if one of the Levites redeems, then the redemption shall come out, a sale of a house or a city of his possession, in the jubilee, because the houses of the cities of the Levites are their possession among the sons of Israel. [25:34] And the fields of the villages of their cities shall not be sold, because it is their possession forever¹⁷⁴¹. [25:35] And if your co-religionist brother becomes poor and his hand slips with you, then you shall strengthen him whether he is a stranger or a guest and he shall live with you. [25:36] You shall not take interest or usury from him, and you shall fear your God, and your brother shall live with you. [25:37] You shall not lend him your money for interest, and you shall not give him your food for profit. [25:38] I am the Lord your God, who brought you out of the land of Egypt to give you the land of Canaan, to be God to you. [25:39] And if your co-religionist brother becomes poor beside you and is sold to you, you shall not use him as a servant's work. [25:40] He shall be with you like a hired hand, like a guest, he shall

⁷³⁴ Lit. 'his power does suffice'.

⁷³⁵ Lit. 'if attains'.

⁷³⁶ Lit. 'according to his redemption'.

⁷³⁷ Lit. 'his power does not suffice'.

⁷³⁸ Lit. 'deadline'.

⁷³⁹ Lit. 'deadline'.

⁷⁴⁰ Lit. 'in the city which there is not to it wall'. Probably a calque of B.Heb בָּעִיר אֲשֶׁר־לֹא לוֹ חֹמֶה; lit. 'in city which not to him wall'.

⁷⁴¹ Lit. 'possession of world'. Probably a calque of B.Heb אַהוַּת עוֹלָם; lit. 'possession of word'.

serve you until the year of the jubilee. [25:41] And then he shall go out from you, he and his children with him, and shall return to his own family and to the possession of his fathers shall he return. [25:42] Because they are my servants, whom I brought out from the land of Egypt. They shall not be sold like the sale of a servant. [25:43] You shall not rule over him Lby force¹⁷⁴² and fear your God. [25:44] And your servants and maidservants:

118 vo

if they become yours, |they shall be| from the nations that are around you, you shall buy servants and maidservants from them. [25:45] And also from the children of the settlers who live with you: you shall buy from them and of their families that are with you, whom they begat in your land. And they shall become a posession for you. [25:46] And you shall take them as a heritage and share them for your children after you, to inherit as a possession. You shall use them forever⁷⁴³. But your brothers, the children of Israel, a man toward his brother: you shall not rule over them oppressively⁷⁴⁴. [25:47] And if a stranger or a settler with you can afford it¹⁷⁴⁵, and your brother with him becomes poor and is sold to the stranger |and| settler with you or to a member⁷⁴⁶ of a gentile of a stranger's family: [25:48] After he is sold the shall have redemption¹⁷⁴⁷. One of his brothers shall redeem him. [25:49] Either his uncle or his uncle's son shall redeem him, or a close relative 748 of his family shall redeem him, or if the can afford it¹⁷⁴⁹, he shall be redeemed |by himself|. [25:50] And he shall calculate |himself| with his buyer from the year that he was sold to him until the year of jubilee and the silver of his sale shall be according to 1750 the number of years. It shall be with him like the days of a hiredworker. [25:51] If there are still many years left, he shall return |money| according to them for his redemption from the silver of his sale. [25:52] And if there remain a few years until the year of jubilee, he shall calculate it for him according to |his| years to return money for his redemption. [25:53] And like a hired servant he shall be with him year by year, and he shall not rule over him oppressively⁷⁵¹ in your sight. [25:54] And if he is not redeemed by these |years|, he shall go out

⁷⁴² Lit. 'by hard', see also B.Heb בְּכֵּרֶדְ; lit. 'in rigor'.

⁷⁴³ Lit. 'until word'.

⁷⁴⁴ Lit. 'by hard', see also B.Heb בְּפֶרֶך; lit. 'in rigor'.

⁷⁴⁵ Lit. 'his strength suffices'.

⁷⁴⁶ Lit. 'base'.

⁷⁴⁷ Lit. 'redemption shall become to him'. It might be a calque of B.Heb גְּאָלָה תְּהָיֶה־לּוֹ; lit. 'redemption it shall become to him'.

⁷⁴⁸ Lit. 'from his relative of his body'. Probably a calque of B.Heb מָשָאָר בָּשֶׂרוֹ; lit. 'from kin of flesh of him'.

⁷⁴⁹ Lit. 'his strength suffices'.

⁷⁵⁰ Lit. 'by', see also B.Heb 3; lit. 'by'.

⁷⁵¹ lit. 'by hard', see also B.Heb בפרך; lit. 'in rigor'.

in the year of jubilee, he and his sons with him. [25:55] Because the sons of Israel are servants to me. They are my servants, whom I brought out from the land of Egypt. I am the Lord your God.

Chapter 26

[26:1] You shall not make idols for yourselves, and you shall not set up an idol or a pillar for yourselves and Lyou shall not set up¹⁷⁵² a figured stone in your land to bow down over it, because I am the Lord your God. [26:2] You shall keep my Sabbaths and fear my sanctuary. I am the Lord.

119 ro

[26:3] If you walk by my statutes, and if you keep my commandments and do them: [26:4] Then I will give you your rains in their time and the land will give its crop, and the tree of the field will give its fruit. [26:5] And the threshing will be enough for you until the grape gathering |time|, and the grape gathering will be enough for you to the sowing⁷⁵³ |time|; and you will eat your food to the full, and you will dwell in your land safely. [26:6] And I will give peace in the land, and you will lie down, and there will be no one threatening you, and I will cut off evil wild animals out of the land, and a sword will not pass through your land. [26:7] And you will drive away your enemies, and they will fall before you by the sword. [26:8] And five of you will drive away a hundred, and a hundred of you will drive away ten thousand, and your enemies will fall before you by the sword. [26:9] And I will turn to you, and I will make you spread, and I will multiply you, and I will establish my covenant with you. [26:10] And you will eat the old getting older, and you will take out the old before the new. [26:11] And I will set 1754 my tabernacle among you, and my will will not despise you. [26:12] And I will walk among you, and I will be a God to you, and you will be a people to me. [26:13] I am the Lord your God, who brought you out from the land of Egypt, from being servants to them, and I have broken the wedges of your yoke, and I made you walk by keeping your head up high¹⁷⁵⁵. [26:14] And if you will not listen to me and will not do all these commandments, [26:15] And if you despise my statutes, or if your soul despise my judgments not to do all my commandments, to break my covenant: [26:16] I also will do this to you: I will even appoint over you trembling, cold disease

The Tit. 'you shall not give'. Probably a calque of B.Heb אָ מְתְּנוּ, lit. 'you shall not give'.

⁷⁵³ Lit. 'seed sowing'.

⁷⁵⁴ Lit. 'I give', see also B.Heb נתתי; lit. 'I give'.

⁷⁵⁵ Lit. 'with upright stature'; see also B.Heb קוֹמְמִיוּת: פֿ, lit. 'upraised'.

and fever⁷⁵⁶ that consume eyes and make the soul ache, and you will sow your seed in vain, and your enemies will eat it. [26:17] And I will set my wrath against you, and you will die before your enemies, and your enemies will reign over you, and you will flee Leven if ¹⁷⁵⁷ there is none who drives away you. [26:18] And if until Lthis time ¹⁷⁵⁸ you do not listen to me, then I will discipline you seven times |more| for your sins. [26:19] And I will break the pride of your power, and I will make ¹⁷⁵⁹

119 vo

your heavens as iron, and your earth as copper. [26:20] And your strength will come to an end in vain, and your land will not give its crop, and the tree of the land will not give its fruits. [26:21] And if you walk with me with resistance, and you will not want to listen to me, then I will increase the striking on you seven times |more| for your sins. [26:22] And I will send the wild animals of the fields on you, and they will slash you, and they will slay your cattle, and they will make you few in number, and your roads will be ruined. [26:23] And if you are not disciplined to me by these |things|, and if you walk with me with resistance, [26:24] Then I will also walk with you with resistance, and also will strike you, seven times for your sins. [26:25] And I will bring a sword over you that takes vengeance, the vengeance of my covenant. And you will be gathered to your cities, I will send the plague among you, and you will be given to the hand of the enemy. [26:26] And when I break the strength of bread, then ten women will bake your bread in one oven, and they shall give back your bread by weight, and you will eat but not be full. [26:27] And if, despite all this 760, you do not listen to me, and you walk with me with resistance. [26:28] Then I will walk with you in anger of resistance, and I will also discipline you, seven times for your sins. [26:29] And you will eat the flesh of your sons, and you will eat the flesh of your daughters. [26:30] And I will destroy your hills ⁷⁶¹, and I will cut off your sun-images¹⁷⁶², and I will cast¹⁷⁶³ your carcasses on the carcasses of your idols⁷⁶⁴, and my soul will despise you. [26:31] And I will ruin your cities, and I will ruin your sanctuaries, and I will not accept the aroma of your offerings. [26:32] And I will ruin the land,

⁷⁵⁶ Lit. 'heat |causing| disease'.

⁷⁵⁷ Lit. 'and', see also B.Heb ;; lit. 'and'.

⁷⁵⁸ Lit. 'these', see B.Heb אֶלָה; 'these'.

⁷⁵⁹ Lit. 'I give'. Probably a calque of B.Heb נְתָּהִי; lit. 'I give'.

⁷⁶⁰ Lit. 'with this', see also B.Heb דָּוֹאָת; lit. 'by this'.

⁷⁶¹ Lit. 'I make your high places disappear'. Probably a calque of B.Heb הָשְׁמַדְהֵּי אֶת־בָּמֹתֵיכֶם; lit. 'I exterminate your high places'.

⁷⁶² Probably a calque of B.Heb אַת־פָּגרִיכ; lit. 'your solar images'.

⁷⁶³ Lit. 'I give', see also B.Heb יַתַּתִּי 'I give'.

⁷⁶⁴ Lit. 'disgustingness'.

and your enemies who dwell there will be frightened. [26:33] I will scatter you among the nations, and I will draw out a sword after you, and your land will be ruined, and your cities will be destroyed. [26:34] Then⁷⁶⁵ the land will complete its remittings in all the days of its desolation, and you will be in the land of your enemies, then⁷⁶⁶ the land will rest and complete its remittings. [26:35] In all the days of its desolation it will rest, the rest that it did not have on your remittings when you dwelt in it. [26:36] And those of you who are left, I will send softness into their hearts

120 rº

in the lands of their enemies. And the sound of a rustling leaf will drive away them, and they will run away as running away from a sword, and they will fall and there shall be no one who drives away you. [26:37] And they will stumble, one other 1767, as if to escape before a sword, when⁷⁶⁸ there is no one who drives away, and you will have no |strength| to stop before your enemies. [26:38] And you will perish among the kingdoms, and the land of your enemies will destroy you. [26:39] And those of you who remain will rot for their iniquity in your enemies' lands, and also for the iniquities of their fathers, they will rot with them. [26:40] And they will confess their iniquity and the iniquity of their fathers for their trespass which they trespassed against me, and also that they have walked me with resistance, [26:41] And I also will walk with them with resistance, and I will bring them into the land of their enemies. And then their closed hearts will bow down, and then they will fulfil the punishment of their iniquity. [26:42] And I will remember my covenant that |I made| with Jacob, and also my covenant that |I made| with Isaac and also I will remember my covenant that |I made| with Abraham, and I will remember this land. [26:43] And the land will be abandoned by them and it will complete its remittings when it is ruined without them, and they will fulfill the punishments of their sins because they despised my laws and because their souls despised my commandments. [26:44] And I will do to them this: when they are in the land of their enemies, I will not despise them, and I will not hate them to destroy them, to break my covenant with them, because I am the Lord their God. [26:45] And I will remember for them the covenant of the first ones, whom I brought out from the land of Egypt in the sight of the nations, to be God to them. I am the

⁷⁶⁵ Lit. 'at that time', see B.Heb אָן; lit. 'at that time'.

⁷⁶⁶ Lit. 'at that time', see B.Heb 's; lit. 'at that time'.

⁷⁶⁷ Lit. 'every person with his brother', see also B.Heb אָישׁ־בָּאָחִין; lit. 'person with brother of him'.

⁷⁶⁸ Lit. 'and', see also B.Heb 1; lit. 'and'.

Lord." [26:46] These are the statutes and rights and laws that the Lord gave between him and the sons of Israel on Mount Sinai by the hand of Moses.

Chapter 27

[27:1] And the Lord spoke to Moses, saying, [27:2] 'Speak to the children of Israel, and say to them, "If a person makes⁷⁶⁹ a special vow⁷⁷⁰ by your valuation of souls for the Lord. [27:3] Then your valuation shall be this: |If it is| a male from twenty years old up to sixty years old, then your valuation shall be fifty mithqals of silver, by the mithqal of the Holy. [27:4] And if it is a female, then your valuation shall be thirty mithqals. [27:5] And if it is from five years old up to twenty

120 vo

years old, then your valuation shall be this: |for| a male twenty mithqals, and for a female ten mithqals. [27:6] And if it is from a month old up to five years old, then your valuation shall be this: |for| a male five mithgals of silver and for a female three mithgals of silver. [27:7] And if it is from sixty years old or above: if it is a male, then your valuation shall be fifteen mithqals, and for a female ten mithgals. [27:8] But if he is poorer than your valuation, then one shall stand him before the priest, and the priest shall value him. According to what the vower can afford¹⁷⁷¹, shall the priest value him. [27:9] And if it is from a cattle that they bring forward as an offering to the Lord, all of it that he gives of it to the Lord shall be holy. [27:10] He shall not exchange it or change it, good for bad or bad for good. And if he exchanges cattle for cattle, then it and the one exchanged shall be holy. [27:11] And if it is any unclean cattle from which they do not offer as an offering to the Lord, then one shall stand the animal before the priest. [27:12] And the priest shall value it, whether it is good or bad, according to your valuation, oh priest, so shall it be. [27:13] And if he redeems it, then he shall add a fifth to your valuation. [27:14] And if a person makes his house holy to the Lord, then the priest shall value it as either good or bad: as the priest values it, so shall it stand. [27:15] And if the one who makes it holy redeems his house, then he shall add the fifth of the silver to your valuation, and it shall be his. [27:16] And if a person makes a part of his field that is his possession holy to the Lord, then your valuation shall be according to its seed. A stalk of barley seed shall be fifty shekels of silver. [27:17] If he makes his field holy from the year of jubilee, he shall stand according to

⁷⁶⁹ Lit. 'separates'.

⁷⁷⁰ Lit. 'intention'.

⁷⁷¹ Lit. 'if his power suffices'.

your valuation. [27:18] And if he makes his field holy after the jubilee, then the priest shall count for him the silver according to the years that remain until the year of jubilee, and it shall be deducted from the valuation. [27:19] And if the one who makes the field holy redeems⁷⁷² it, then he shall add a fifth of silver to your valuation, and it shall be a residence to him. [27:20] And if he does not redeem the field, and if he sells the field to another person, it shall not be redeemed anymore. [27:21] And the field, when it is released in the jubilee, shall be holy to the Lord like a devoted field:

121 rº

its possession shall be the priest's. [27:22] And if a buyer of a field, that is not a field of his possession, makes it holy to the Lord. [27:23] Then the priest shall count the amount of your valuation until the year of jubilee, and he shall give your valuation on that day as a holy |thing| to the Lord. [27:24] In the year of jubilee the field shall return to him who bought it from him, to him who has a possession of the land. [27:25] And all your valuations shall be by the mithqal of the Holy: twenty stone of a fruit shall make a shekel. [27:26] Just: a firstborn, which as a firstborn from the cattle to the Lord: a person shall not make it holy whether it is an ox, or a sheep, it is the Lord's. [27:27] And if it is an unclean animal of the cattle, then he shall redeem it at your valuation and add a fifth to it, and if it is not redeemed, then it shall be sold at your valuation. [27:28] Just: if a person devotes any devoted |thing| to the Lord of all that he has from human or cattle or from a field of his possession, it shall not be sold nor redeemed: every devoted |thing| is a holy of holies to the Lord. [27:29] Anyone devoted, who has been devoted from a human shall not be redeemed, the shall be killed¹⁷⁷³. [27:30] And all the tithe of the land, from of the seed of the land, from the fruit of the tree is the Lord's. It is holy to the Lord. [27:31] And if a person redeems⁷⁷⁴ some of his tithe, he shall add a fifth to it. [27:32] And all the tithe of the cattle or of the sheep, everything that passes under the rod, the tenth shall be holy to the Lord. [27:33] One shall not search between good and bad, and change it; and if he does change it¹⁷⁷⁵, then it and its substitute shall be holy. It shall not be redeemed." [27:34] These are the commands which the Lord commanded Moses for the sons of Israel on Mount Sinai.

772 Lit. 'redeeming if he redeems'. Probably a calque of a Biblical Hebrew construction which has an intensifying

¹¹² Lit. 'redeeming if he redeems'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb אָם־גָּאל יְגָאַל; iti. 'if to redeem he is redeeming'.

T73 Lit. 'dying, he shall be killed'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb מות יומָת; lit. 'to be put to death he shall be put to death'.

The Lit. 'redeeming if he redeems'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb אָם־גָּאל יִגְאָל , iti. 'if to redeem he is redeeming'

The changing if he changes'. Probably a calque of a Biblical Hebrew construction which has an intensifying function (see 2.3.1.2.1.), see B.Heb אַם־הָמֵר יְמִירְנּוּל; lit. 'if to redeem he is redeeming'.



נְיּלְרָא אֶל משֶׁה וַיִּדְבֵּר יְּנָי אָלִיו מֵאֹהֶל מוּעֵד לֵאמר: דַצַקּוְרְיִי משֶׁה גַא אֹהֶל מוֹעֵד לֵאמר: דַצַקּוְרְיִי משֶׁה גַא אֹהֶל מוֹעַר דַן צַלּיְרְנָה יִשְׁרָאֵל נִיגוּ בּאַמְפְּיִן אַלְרְנָה יִ אָדָם גֵּנְדִינִיזְיֵן אָנָר יוּבּיְקְלַשְּטִירִינִי קְרָבּּן יִיָּר גַּה וּ אוֹל מּוּנְרָדִן אוֹל סִיגִיְרְדַן יֵא אוֹל מִיְרַן וּ יוּבּיּקְלַשְּטִירִינִי קְרָבּּנְיְיִיְיִי יִ אַם אוֹל מּוּנְרָבן אוֹל סִיגִיְרְדַן יֵא אוֹל מוֹיָרו וְ בַּנְלֹם אֶרְבַּבְּנִי אַלִּינָ יִיִּבְּקְלְשְׁמִירְכִין אַנִי וּ בְּשִׁר אוֹלְנִיר אַנֵּר בְּשְׁרִי אוֹל מוֹעֵד נִיג יוּבּיִקְלְשְׁמִירְכִין אַנִי וּ בַּבְּיִי אוֹל מוֹעָד נִיג אוֹל עוֹלְנִיג וּ בַּבְּיכִין אוֹל מִינִינִי אוֹל מִוְנִי בּיִשׁי אוֹסְמוּנָא אוֹל עוֹלְנִיג וּ בַּבְּכִין אוֹל סִיגִיְנִי אַלְּרִינָה יְּיִי נִינִ וּ בּיִבְּיִיִּלְשְׁמִירְכִינְלְּר אוֹלְנִיר אַנִּר וּ בְּעִייִיכִין קוֹלִינִי אוֹל מִנְבְּה אוֹלְנִיר אוֹלְנִיר אוֹל מִינְרָה וּ בְּבּיבְּרָה וְ נִיג אוֹל בֹּהֵן לֵירִ אוֹל מִנְבָּה אֹלְרִינְבְּה אוֹלְנִיר אוֹל מִוֹבְר אוֹלְנִירִי אַבְּרוֹ אוֹל עוֹלְנִיג וּ בִּנִבוּהמְלְחִינְנִי אוֹל מִוֹבְי אוֹל מוֹלְנִי אוֹל מוֹעִד נִיג אוֹל מוֹעד נִיג אוֹל בּהֵן עִינִי אוֹל מִוֹבְי אוֹל מוֹעד נִיג אוֹל בּהֵן נִיג אוֹל מִוֹעד בִּה אַנְיִיבְיוֹ אוֹל מוֹלְנִינִי אַבְּבְּיוֹנְנִי וּ אוֹל בּוֹן מִינִים אוֹל מוֹעָד בִּי בְּיבּייִלְייִר אוֹל אוֹם אוֹסְמוּנְא : וערכו דִיבְשְׁמִירְסִינְלָר אוֹל אוֹם אוּסְוּנִין וּ אוֹל בִּיב וּ בִּילְרְינִין וְאוֹל בָּיִב וּ אוֹל מִינִין וּ אוֹל בִּיב וּ אוֹל מִינִין וּ אוֹל בּיבוּ מְלַרְנִינִי וּ אוֹל בִּיבוּ וְ אוֹל מִינִין וּ אוֹל בִּיבְיוֹנְי וּ אוֹל בִּיבִין וּ אוֹל בִּיב וּ וּערנוּ בִי בִּילְר אוֹל מִינִין וּ אוֹל מִינִין וּ אוֹל מִינִין וּ אוֹל אַנְעְלֵר אִים מִּבּינִי וּ אוֹל אִינְעלְר אִים מִּבּינִין וּ אוֹל אִינְעלְר אוֹב בּיבּנִים וְּבִי וּ אוֹל אִינְייִין וּ אוֹל אִינְייִי וְ אוֹל אִינְייִין וּ אוֹל אִינְייִי בּוּ בִּילְר אִי וּיִבּין בְּיוֹי בְּיוֹי בְּיוֹי בְּיִייְיִייְיִייְ בְּיוֹייְנִייְ מִינְיוֹייִין מִינְייִייִייְייְייִייְיִיוּ וְּיִינְייְיְיִייִייְיוּ וְיִיּבְיוֹייְייִיְייִייִייְייִייְיוּ בְּיוּייִייִין וּינִייִייְייִיּיוֹי

פי אול אוט אוסטונה ו פי אול מופח אוסטונא : וקרבו דקריניני דאיקלריני יוּכְכִין כּוֹכְלַר בִּילוֹ ו דַשִּישַמְסִין אוֹל בּהֵן אוֹל ג׳וּמְלָא נִי אוֹל מִוְבַּחַשַהועוֹכְה דִיר אוֹלו אוֹמְלוּ קָרָבֵּן קוֹסִי קבּולְלִיקְנִיג ינָי גַא : ואם דַאָּגַר אוֹל קוֹיָדַן אָכָה קְרְבָּנִי ו אול קוילַרְבוּן יא אול אַצְבּילַרְבוּן עוֹלְנַה ו בַּנְבַם אָרְבַּב נִי אַלִיפּ יובוּקְלַשְּאִירְבִין אַני : ושחם דַסוֹיִסִין אַנִי אוג׳וּ וַנִינְדֵא אול מִוֹבֶּחַ נִיג ו כִּי צְפוּן מַרַפִּינְרַא אַלְרִינַה ינָי נִיג ו דַפַּצְּסִינְלָר אוֹגְלַנְלַרֵי אַהֶּרן נִיג אוֹל כַּהֵוֹ לַרַ קַנִינֵי ו אוֹל כִוְבָּחַ אָסְפוּנַה צָפָצוּבְרַה: ונתח דַפּוּבוּמְלַמִין אַנִי פּוּבוּמְלַרִינְגַ׳ה ו דַא בַּשׁין דַא אוט אוט אול פּהָן אַלרְנִי ו אול אוּכְטוּנָא ו בִּי אוֹל אוט אוֹל אַרְנִי ו אוֹל אַנְצְלְרַ אוּכְטוּנָא ו בִּי אוֹל אוט אוּכְטוּנָא ו בִּי אוֹל מִוְבָּחַ איִכְטוּנָא : והקרב דאול קריְנְנִי דַאוֹל אַיְקְלַרְנִי יוּבְסִין סוּבְלַר בִּילֵן וַ דַיובוּקְלַשְׁמִירְסִין אוֹל בּהֵן אוֹל גו׳מְלַאנִי ו דַמוּמַמְסִין אוֹל מוֹבֶת טַה ו עולְרִיר אוֹל ו אוֹמְלוּ קָרְבֵּן קוֹקוֹפִי קבּוּלְלִיק נִיג יני נַה : ואם ואם דַאָנַר אול קוּשְׁשֵּן אָסַה ו עולָה קרְבְּנִי יֹנָי נַה ו דַיוּבְוּקרַשְׁשִירְסִין אול קוְמְרוּלַרְרֵן ו יַא אוֹ כּוֹגְוּרְצִין בַּלַלַרִינְדַן קָרְכּנִינִי : והקריבו דיובוְקַלְשְׁמִירְכִין אַנִי אול בּהַן אוֹל מִוְבֵּחַ קה ו בא בּּכְכִין בַּשִׁינִי ו דַמוּטְמִכִין אוֹל מִוְבַּחַטַה ו דַדַמְלַנִילְסִין ַקְנִי וּ דוּבַרִי אוֹסְשוּנָה אוֹל מִוְבֵּחַ נִיג : והסיר דַבִּשַּׂרְסִין קּרְנִינִי יונו בילו ז רַשַשְׁלַסִין אַנִי יַנִינַה אוֹל מִוְבָּחַ נִיג גוּן דוגושִׁי שַרַאף לאו אול פול טוּבְלַן וִרְגַה : ושםע : בַוּרְמָּסִין אֵנִי בַנְמְלַרִינְדַן (סירטינדן) אַיוִרְמַסִין ו דטושַמְּכִין אַנִי אוֹל כּהָן אוֹל מִוְבָּחַשַה ו אוֹל אַנְצְלַר אוּסְטוּנָה בִּי אוֹל אוֹט אוקטונא ו עולה דיר אולו אוטלו קרבן קוקוםי קבוללקניג יוני גה:

לְרָבֶנְלְרִינְדְן יִינְי נִיג : וכי דַאָּרֶן גַּה דָאוּגְלַנְלְרִינָא דִיר ו קוֹדְשִׁי קוֹדְשׁ לְרָנִיגָ אוֹלְסִין אוֹמָלּי לְרָנִיגָ אוֹלְבָּן מִנְחָדְ מִנְחָדְ מִנְחָדְ דִּאוֹבְלְּלְרִינָה אַבְּרן נִיג אוֹל כּהָן לַרְגַה ו דַאבוּצְלַסִין אַנְדְן דוֹלוּ בָּהַנְינִי אוֹגִינְדְן דִינִינְּדְן ג'וִמְלָא מִמְינִי בִּיֵלְו דְשׁישִׁסְין אוֹל כּהָן שוֹמָאִרְמַסִינִי אוֹל מִוְבָּן דִינִינְדְן ג'וִמְלָא מִמְינִי בִּיַלְו דְשׁישִׁסְין אוֹל כּהָן שוּמָאִרְמַסִינִי אוֹל מִוְחָה דֵן אוֹשְלּוּ קְּרָנִיג יוֹבוּקּלְשְׁמִירְסֵג קְרָבָּן מִנְחָה תַנְּדִוּר פִּישִׁירְמַסִי קְבּוּלְלִיכְנִיג ייִר ו קוֹדְשִׁי קוֹדְשׁ לַרְנִיג אוֹמְלוּ קּרְינִא דִיר ו קוֹדְשִׁי קוֹדְשׁ לַרְנִיג אוֹמְלוּ קּרְנִיג אוֹמְלוּ בִּילְרִינָא דִיר ו קוֹדְשִׁי קוֹדְשׁ לַרְנִיג אוֹמְלוּ בִּישִׁירְמַסִי בִּאֹנְרְינִבְּן יִינִי נִיג : וכי דַאָּנֵר יוֹבוּקְלְשְׁמִירְסֵּג קְרָבְּן מִנְחָה תַנְּדִּר פִּישִׁירְמַסִי בּוּאַרְנִיבְּן מִנְּתָר בִּי בּיִבּוּ בְּוֹבְּיִי יִינִי נִיג : וכי דַאָּנְרְינִא יִיר וּ קוֹדְשִׁי קוֹרְה בִּישִׁירְמַסִי בּירְנִינְ אוֹל מִנְחָה תַנְרָּוּר בּיִשִּירְמַסִי בּיר וּבִּיּעְרִינְיבוּן ייִנִי נִיג : וכי דַאּנְרְינִבּן יוֹבוּילְנִילְרִינָּא דִיר ו קוֹבִיקּלְייִנְיבוּ מִנְּיִם מְנִינְן יִנְינִי נִיג : וכּי דִּאָּרְיִבְּי יוֹבוּיְלְבְּעִייִרְסֵּג קְרָבְּלְיִינָּא דִיר ו קוֹנְישִי קוֹנִין עִנִינִי בּי בּיִינִים בּיִּים בְּיִנְם מְנִים מְּנִים מְנִים בּיִים בּיּיִים בּיִים בּיִים בּיּיִּים בּיּיִים בּיוֹים בּיּים בּיִים בּיּבְּים בְּיִים בּיּיִים בּיִים בְּיִבּים בְּיִינִים בְּיִים בּיּיִים בְּיִים בּיּבּים בְּיִבּים בְּיִבּים בְּיִים בְּיִים בְּיִינִים בּיּיִים בּיּיִים בּייִים בּיּיִים בּינְנִיבּים בּיים בּיִים בּיּיִים בְּיִיים בּיוּים בּיוֹנְיים בּיוֹים בּיוֹים בּיוֹים בּיוֹים בּיוֹים בּיוֹים בּייִים בּוּייִים בּייים בּיוֹבּים בּייִים בּיים בּיוֹבּיים בּיוֹים בּיים בּייִים בּייִים בּייִיבּים בְּיִיבּים בְּיִים בְּיִים בְּיִים בְּיִבּים בְּייִים בְּיִים בְּיִים בְּיִּבְּים בְּיִים בְּיִים בְּיִים בְּיִים בְּיוֹים בְּיִים בְּיִים בְּיִים בּיוּבְים בְּיבְיים בְּיבְּיוּים בְּיִיבְים בְּיִים בְּיִים בְּיִּיבְים בְּיִבְייִים בְּיִיבְים בְּיבְּי

ארוב.

אונב קלין מצלר קרישלנילר יג בילן ו דיופקא מצלר סילייננילר יג בילן: ואם דאנר מוְתָה אוֹל שַנַה אִיצִינְרָא אֵסַה לְקְבְּנִיג ו אווַבַ קַרִישִׁילְנוֹ יַג בִּילֵן מַצְה בּוֹלְסִין : פתות שִילִימְלַמַה שִילִימְלַנִין אַנִי שִילִימְלַרוֹדַקוּיְנִין אוּסְמוּנַה יַג וּ מנחה דיר אול : ואם באנר שנניבדא פישו מנחסי אסה קרבענו אייב יג בּיַלוֹ קילִינְסִין : והכאת דַבָּטִירְגִין אוֹל מִנְחָה נִי בּוּלַרְדֵן יְנִי נַה ו דיובוּקלשְטִירְסִין אַנִי אוֹל כּהוֹ נַה וֹ דִיוּבוּקלשְטִירְסִין אַנִי אוֹל כּהוֹ נַה וֹ דִיוּבוּקלשְטִירְסִין אַנִי אוֹל כּהוֹ נַה וֹ דִיוּבוּקלשְטִירְסִין אַנִי אוֹל דַאַייִרְסִין אול כּהָן אוֹל מִנְחָה דַן שוּמְסוּסִינִי ו דַשוּמַמְסִין אוֹל מִוְבָּחַטַה ו אוֹמְלוּ קרבן קוקוםי בפולליקניג יני גה : והנותרת דאול בלנני אול בונה דן ו אַהְרן גַא דַאוֹגְלְנִלְנִינַה דִיר ו קוֹרָשִׁי קוֹרָשׁ לַרְנִיג אוֹמְלוּ קּרְבְּנְלַרִינְדֵן יִנָּי נִיג : כל הַר אול מִנְחָה בָּי יוּבָקְלַשְּׁמִירְסַגִּיז יָנִי גַה ו מִילִינְמַסִין חָמֵץ ו מִימְלַקּ היץ אַצי כַמור דַהִּיץ בִּבְּמֵוֹ ו שוַשְשְׁמֵנִיוֹ אַנְדֵן אוֹשְלוּ כְּרְבָּן ינְי נַה : קרבן אילָב קרְבָנִי יובוקְלַשְׁמִירִינִיז אַלְרָנִי ינִי נֵא ו באוֹר מִוֹבֶּחַ קַה צִיקְמַבִינְלַר קוקוסינא קבוּרְלִיקְנִיג : וכל דַג׳וּמְלֵא מִנְחָה ְקרְבְּנִיגנִי טוּז בִּילֵן טוּיְלַגִּין וּ דאבּסִימְמֵגין שַרְט מוּזִין מִנְרִינְנִיג ו אוּסְמוּנְדוֹ מִנְּחַנְנִיג ו הַר קּרְבְּנִיג אוּסְמוּנֵא יוּבוְקלשְׁמִירְגִין מוּז : ואם דָאָגרַ יוּבוּקלשְׁמִירְסֵג איִלְבְּלֵר מִנְחָה סִינִי יני גה ואַרְפַּה בּשִׁי ציְרְבִּינְגן אוּשְקה שַוּן אַרְפַּא בּוּרְפַּסיויוּבוְּקלשְׁשִירְגין אִלְבְּלֹרִינְנִיג מנְחָה סִינִי : ונתתי דַוֹנִגִין אוּסְשׁוּנֹא יַג ו דַקּוֹיְגִין אוּסְשׁוּנֵא מִמְיַיֵן ו מִנְחָה דיר אול: והקטיר דטישַאָסין אול כּהַן טוּטְכּוּסִינִי כּוּרְפַסִינְדֵן דיַגִּינְדַן ו ג'וּמְלֵא מִמְיַנִי בִּילַן אוֹמְלוּ ְלְרָבֵן יֹנְי נַה :

ואם דאגר שְלְמִים קְרְבְּנִי אִמַה קְרְבְּנִי וּ אִנְר אוֹל מִיגִר שְלְמִים קְרְבְּנִי אִמַה קְרְבְּנִי וּ אִגר אוֹל מִיגִּרְרָבְּן אוֹל מִוּעָר נִיג וּ וֹממך דַמַיְמִין קוֹלוּנִי בַּשִּׁי אוּסְמוּנָה קְרְבְּנִינִיג וּ אַלְרִינֵא ייִנִי נִיג : וֹממך דַמַיְמִין קוֹלוּנִי בַּשִּׁי אוּסְמוּנָה קְרְבְּנִינִיג וּ דַמַיְמִין קוֹלוּנִי בַּשִּׁי אוּסְמוּנָה קְרְבְּנִינִיג וּ בְּמִי אִנְיִנְא אַרֶּל מוֹעֵר נִיג וּ דַמַצְמִינְלַר אוֹגְלַנְּלְרֵי אַהְרוֹ נִיג אוֹל בּהָּן לֵר וּ אוֹל שִׁנְנִי אוֹל מִוְבָּן אוֹסְמוּנָא צוּבְּצוֹכְרַה : וֹהקריב דֵיוּבוּקְלַשְּׁמִירְסִין אוֹל שְׁלְמִים קּרְבְּנִינְין וּ אוֹמְלוּ קְרְבְּן יִיְי נֵה וּ אוֹל יַנְנִי אוֹל יַבְּנִי אוֹל בּוּנְרַבְּלְרָנִי וּ בִּי אוֹל בּוּנְרַבְּלְרָנִי אוֹל בּוּנְרַבְּלְרָנִי וּ בִּי אוֹל בּוּנְרַבְּלְרָנִי וּ אוֹל בּוּנְרַבְּלְרָנִי אוֹל יִנִנִי וּ בִּי אוֹל בּוּנְרַבְּלְרְנִי אוֹל יִנְנִי וּ בִּי אוֹל בּוּנְרַבְּן אוֹסְמוּנִא : וֹאת דָּא אָבִי אוֹל בּוּנְרַבְּלְרָנִי אוֹל בּוּנְרַבְּלְרָנִי וּ אוֹל יִנְנִי אוֹל יִנְנִי וּ בִּי אוֹל בּוּנְרַבְּלְרָנִי אוֹל בּוּנְרַבְּלְרָנִי אוֹל מִינִם לְנִינִי וּ בִּי אוֹל בִּנִין אוֹסְמוּנְא : וֹאת בּיוֹי אוֹל יִנִנִי וּ בִּינִי וּ בִּי אוֹל קּרִינְנִי וּ בִּי אוֹל בִּינִים אוֹל בּוּרְבְּלְרְנִייִי

מיקהא

דַאול יַנְנִי כִּי אוּסְטְלָרִינְרַאוֹ כִּי אוֹל אָסִיוְלַר אוּסְטוּנַא ו דַאוֹל אַרְטִיְקְסִינִי כִּי אול בַּגִיר אוּסְשִינַא ו אוֹל בּוֹגְרַבְּלֵר בִּילַן בִּשְרְסִין אַנִי : והקשירו דשושַאְסִינְלַה אַני אוּגְלַנְלַרֵי אַהַרן נִיג אוֹל מִוְבִּחַטַה אוֹל עוֹלָה אִנְסמונא וֹבִּי אוֹל אַנְצְלַר אִנְסְמוּנֵאוֹ בִּי אוֹל אוֹט אוּסְטוּנַא ו אוֹטְלוּ כְּרְבָּן קוֹקוּסִי קבּוּלְלִיְקנִיג יָנִי נַה : ואם דאַנַרְ אול קוידו אסד בּרְבָּנִיו שְׁלָמִים בְּרְבָּנִינה יני נַה ו אַרְבַבּ יַא מִישׁי ו סַנְּלַם נִי יוֹבוּקְלַשְׁמִירְסִין אַנִי: אם אָגַר קוזוּ אוֹל יוּבוּקְלַשְׁמִירִיר אָסַה קְרְבָּנִינִי וּ דיובוּקְלַשְׁטִירְסִין אַנִי אַלְרִינֵה ינָי נִיג : וסמך דַשַיַסִין סולוני בַּשִׁי אוּסְשוּנֵא קָרְבָּנִינִיג ו דַסוּיְסִין אַנִּי אַלְרִינַה אהֶל מוֹעֵר נִיג ו דַסַצְּסִינְלַר אוֹגְרַנְּלְרֵי אַהְרן נִיג קניני אוֹל מִוָבֶחַ אוּסְמוּנַה צוּפְצוּבְרַה : והקריב _ דיוּבוּקְלְשְׁמִיְרִסִין אוֹל שְׁלְמִים קָרְבָּנִינְרָן אוֹמְלוּ קָרְבַּן יְיָי גַה ו הָמָם אוֹל קויְרוּק ינִינִי ו קַרְשִׁיסִינַה אוֹל יוּסוּלְגַנְנִיג בּמַרְסִין אַנִי ו דַאוֹל יַנְנִי אוֹל יַפְּקוֹ אוֹל סַרִינְנִי ו דַא ג׳וּסְלֵא אוֹל יַנְנִי בִּי אוֹל קַרִין אוּסְשוּנַה : ואת דא אָבִי אוֹל בּוֹנְרַכְּלַרְנִי ו דַאוֹל יַגְנִי כִּי אוּסְטְלַרִינַאו בי אול אָמִיוְלַר אוּסְמוּנַאוֹ דָא אַרְמִיּקְמִינִי אוֹל בַּגִיר אוּסְמוּנַאוּ אוֹל בּוֹגְרַבְּּלַר בִּילַן בַּטַרְסִין אַנִי : והקשירו דַטוּשַאָסִין אַני אוֹל כּהֵן אוֹל מִוְבַחַשַה ו אוֹשְלוּ לַרְבָּן אָיָ נַא : ואם דאַנַר אָצְבִּי אָסַה לָרְבְּנִי וֹ דַיוררוּקְלְשְׁטִיְרְטִין אַנִּי אַלְרִינַא פּיָר נִיג : וסמך דַמַיִּסִין קולוני בַשִּׁי אוּסְמונה ו דַסוּיְסִין אַנִי אַלְרִינַה אהָל מוִער ניג ו דַסַצְסִינְלַר אוֹנְלַנְלְרִי אַהַרן נִיג קנִינִי אוֹל מִוֹבֵּחַ אוּסְטוּנַא צוּפְצוּבְרַא : והקריב דיובוְקלשְׁמִירְסִין אָנְרַן קּרְבְּנִינִי ו אוֹמְלוֹ קִרְבְּן אַיְּ בַחו אול יַנְנִי אוֹל יַפְקוֹ אוֹל קַרִינְנִי דַא ג׳וִמְלַא אוֹל יַנְנִי וֹ בִּי אוֹל קַרִין אוּסְטוּנַה : ואת דַא אַבִּי אוֹל בּוֹנְרַבְּלַרְנִי ו רַאוֹל יַנְנִי בִּי אוּסְטְלֵרינַה ו בִּי אוֹל אָמִיוְלֵר אוּסְטוֹנַא ו בא אַרְטִיקְּסִינִי כִּי אוֹל בַּנִיר אוּסְטוּנֵא ו אוֹל בּוּנְרָכְּלַר בִּילַן בִּטַרְסִין אַנִי : והקטירם דַמּוּטַמְסִין אַלַרְנִי אוֹל כּהָן אוֹל מִוְבַּחַטֵּה ו אוֹמְלוּ קְּרְבַּן קוֹקוסִינֵא בַּרֹּילְיִקְנִיג ו ג׳וּמְלָא יָנְנִי יִנִי בה : חקת אוֹמוְרֹלִיבּ רַסִים דווּרְלַריגִיוְגַא ו ג׳וּמְלַא

אוטנרשְלַרִיגִיזָרא ו הָץ יַגְנִי דַהָּץ קוֹ נִי אַשְׁמִגִיז :
וידבר דַּטוֹלְרִי יֹנִי מֹשָׁה נָא דִמַה : דבר טוֹלְגִין אוּגְלַנְלַרִינָא
יִשְׁרָאל נִיג דַמַה ו גַ׳ן כִּי יִזִיקְלִי בּוּלְסַה יִגְלִישְׁלִיק בּיּלַן ו ג׳וִמְלא מִצְוָהְ

לַרִינְרָן ייני נִיג כִּי קִילִינְמַסְלַרְדִיר ו דַקִילְסֵא בִּירְסִנִי בִּיְרַדֵן אַלַרְדָן ו אם אָנֵר אול סִירִינְנַן כהו יַזִיקְלִי בּוֹלְכַה גוּנַחִי אוּצוּן אוֹל אוּלוּסְנִיג וַ דִיוּבַקְלַשְּׁמִירְסִין יַוֹינִי אוִצוֹן כִּי יַזִיקְלִי בּוּלְרִי ו בּוֹנָה בַּלְכִין סִינִירְנִיג סַגְּלַם נִיוּ יֹּתְ נַאוּחַטְאת קה: והביא דַכּטירָסין אול בּוָגני אַשיגינַא אהָל מוער נִיג אַלְרִינַה יְיָּה נִיג ו דַפַיִּסִין קולוני בשי אוכשונא אול בונניג ו דַסויִסין אול בונני אַרְדִינָה יְנִי נִיג : ולקח באַלְסִין אול סִילִינְנַן כּהַוֹ קנִינְדַן אול בּוּגנִיג ו דַגִירְנִיזְסִין אַנִי אהֶל סוֹער נַא : ומבל דַמַנְצְסִין אוֹל כּהֵן בַּרְמַנִינִי אוֹל קַנְנֵא ו דַסַצְסִין אוֹל בַנְרַן יִרִי בּרְטְלֵר שַּלְרִינַה יְיֹי נִיג ו אַלְרִינַא פַּרְדַסִינִיג אוֹל קוֹרֶשׁ נִיג : ונתן דֵוֹרְסִין אוֹל כּהָן אול בַנְבַן קווש בּוֹשִׁרֵרִי איִכְטוּנָא אול אוֹמְיַמְלֵר פוּמְסוּסִי מוֹבְּחַיִנִיג אַלְרִינָא ינף ניג ו כי אהל סוער דה ו דא ג'יִסְלַא קנין אול בוּגַנִיג ו פוּפְסִין מַסַלִינַא מִוּבְחַ הִיניִג אוֹל עוֹלְנִיג כִּי אָשִׁינִיְנַדא אֹהֶל מוִעֵר נִיג : וֹאת דַא ג׳וִמְלֵא יַנִין אול חַשָאת בּוּנַסִינִיג אַיִירְסִין אַנְרַן ו אוֹל יַנְנִי אוֹל יַפַּקן אוֹל קרין אוסטינא דא ג׳וִסְלָא אול יַגני כִּי אוֹל קריון אוסְטוּנא : ואת דא אָבי אוֹל בּוֹנְרַבְּלַרְנִי וַ דַאוֹל יַנְנִי בִּי אוֹסְאַלַרִינֵא ו בִּי אוֹל אִמִיְזֹלֵר אוּסְמוּנֵא ו רַאוּל אַרְמִיקְּסִינִי בִּי אוֹל בּנִיר אוּסְמוּנַא ו אוֹל בּוֹנְרַבְּלַר בִּיַלוֹ בִּמַרְסִין אַנִי : כאשר נָצִיכִּי אַיִרִילִיר אוגוּזִינְרַן אוֹל שְׁלְמִים קְּרְבְנִינִיג ו דַמוּמַאְסִין אַלַּרְנִי אוֹל בהן מוְבַּחַי אוּסְטוּנַה אוֹל עוֹלְנִיג : ואת דא מִרִיסִין אוֹל בּוּגַנִיג דא ג׳וּסְלֵא אִמִין וּ בַּשִׁין אַיַקְלַרִי כִּילַן דַקַרִינִין דַמָּוָנִין : והוציא דִצִינַרְסִין הַּבְּמִיל אול בוגני אַכוּלְרַן צִיכָרִי ו מָמִיז וַרְגַה אול פוּל מוּפּן יַרְגַה ו דַפּוּיְרוּרְסִין אַנִי אַנְאַלַר איִסְטוּנַה אוֹטְקָאו אוֹל כּוֹל טוֹכּן יִהְרָא כּוּיְרוּרוֹלְסִין: ואם דָאָנָר ג׳וְמְלֵא ג׳מַעמִי יִשְׁרָאֵל נִיג יִגיּלְסַלֶרוּ דַיַשִּׁירִין בּוּלְסַא בִּיר שִׁיי גוּוְלַרִינְדֵן אוֹל קְחַל נִיג ו רָקִילְסַלֵר בִּיר נִי גִיּמְלֵא מִצְנָה לַרִינְרַן יְיֹי נִיג פּי קִילִינְמַסְלֹרְרִיר ו דַגוּנַחְלִי אוּלְסַלַר: ונורעה דַבּילִינִסָא אוֹל יַזִיק בּי יַזִיק בּוּלְרִילֵר אַנִיג אוצון ו דיובוקלשְמִירְסִינְכֵר אוֹל בְּהָל בּוֹנֵה סִינִיר בּלַסִינִי חַאָאת כַּה ו דַנִּמִירְסִינְלֵר אָנִי אַלְרִינָה אֹהָל מוֹעֵר נִיג : וסמכוּ דַשַּיִסיְנַלַר קַרְטְלֵרי אוֹל גַ׳ מַעְטִנִיג קּוּלְלַרִינָי בַשׁי אָסְאַנָא אוֹל בּוּגנִיג אַלְרִינַה יֹנָי נִיג ו דְסוֹיִסִין אוֹל בּוּגנִי אַלְרִינַה יֹנָי נִיג וּ והביאו

והביאר דַנִירְנִיוְסִין אוֹל סִילִינְנַן כּהֵן קַנִינְרַן אוֹל פּוּגַנִיגו אֹהֶל מוֹער נַא ומבל דַמַנְצְּמִין אוּל כּהָן בַּרְמַנִינִי אוֹל קַנְדֵן וֹ דַסַצְּמִין יְדִי כּּרַמְלַרִי אוּסְשוּנֵא אול מוְבַּחַ נִיגוּ בִּי אַלְרִינֵא יֹנִי נִיג בִּי אֹהֶל מוֹער דַה וֹרָא ג׳וֹמְלָא אוֹל קּנְנִי ו מוֹכְּמִין בּנְיַיִמִינָא מִוְבֵּחִינִיג אוֹל עוֹלָה נִיג ו בִּי אַשִׁיגִינְרָה אֹהֶל מוֹער נִיג: ואת דַא ג׳וּמְלֵא יַגִינִי אַיִירְסִין אַנְבַן ו דַשוּשִאַסִין אוֹל מִזְבַּחַשַה: ועשה דַקִילְסִין אוֹל בּוּגְנָא נִצִיבִּי קִילְרִי בּוּגַסִינָא אוֹל חַשְאת נִיג וַעלֵי קִילְסִין אַנֵר ו דַבְּבְּבְּת אָמְסִין אוסְמְלַרִינאַ אול בּהָן ו דַבּוֹשַׁמִילִיר אַלַרְגַה : והוציא דַצִיגַרְסִין אול בּוּגַנִי אַבוּלְרַן צִיפָּרִי ו דַפּוּיְרוּרְסִין אַנִּי ו נִצִיפִּי פּוּיְרוּרְרִי אוֹל פוּרוּנְיִי פּוּגְנִי ו חַאָאתִי אוֹל קהָל נִיג דִיר אול: אשר אָנֶר אַנָה יַזִיּקְלִי בּוֹלְסַא וֹ דַקּיְלְסַה בִּירְנִי ג׳וּמְלֵא מִצְּיֹהְ לַרִינְדֵן יְיִיּ בַּיְּרִיִּסִינִיג ו בִּי קִירִיְנַמַסְלַרְרִיר ו יַגְלִישְׁלִיק בִּילַן ודַגוּנַחְלִי בּוּלְסֵה : או יֵא בּילִינְסַה אַנֵר יַזִינִי ו בּי יַזִיקְלִי בּוּלְדִי אַנִיג אוּצוּן ו דַגִּמִירְסִין קּרְבְּנִינִי אוּלְנִין אַצְבּילַרְנִיג סַגְּלָם אַרְבַּב : וסמך דַטַיַסין קולוני בַשִׁי אוּסְשוּנֵא אול אוּלַקְנִיג ו בסויְסִין אַנִינָא יָרֶדַה כִּי סויְסַה אוֹל עוֹלָנִי אַלְדִינַה יֹנְי נִיג) חַמְאת דִיר אוֹל : ולקח בַאַלְסִין אוֹל בּהֵוֹ קנִינְרֵן אוֹל חַמְאת נִיג בַּרְמַנִי בִּילַן ו דַוֹרְסִין קוּוֹשׁ בושברי אוסמונא מובחיניג אול עולניג ו בא קניני מובסין ממלי נא מובחיניג אול עולניג : ואת דא ג'ומלא יגיני מוממסין אול מוֹבֶּחַ מַה ו יַגי גיבּי אָרְבְּנִינִיג אוֹל שְׁלְמִים לַרְנִיגו דַבְּפְּרַת אָמְסִין אוּסְמוּנֵא אוֹל כּהֵן וּ יַזְינִינְרַן דַבשַטִילִיר אַגַר : ואם דַאָנַר בִּיר גַ׳ן יַזִיקָלִי בּוֹלְסַה יַגְלִישְׁרִיק בִּילַן אוּלוּסִינְדַן אול יִרְנִיג ו קּילְגַניְנְדָה בִּיר נִי מִצְנָה לַרִינְרַן ינִי נִיבּי קִילְינְמַקְלַרְרִיר ו בַּנִינְחַלִּי בּוֹלְפָה : אוֹ יֵא בִּילִינִפָה אַנֵר וַזִינִי כִּי יַזִיּקלִי בּוֹלְדִי ו דַגְמִירְסִין קְרְבָּנִינִי אולנין אאבילרניג סנלמני ו יווגי אוצון כי יויקלי בולדי : וסמך דטיסיו קולוני בַשִׁי אוּסְפוּנאַ אול חַמָּאת נִיג ו דַסוּיִסִין אול חַמְאת נִי יִרִינְדַה אול עולְנִיג : ולקח דַאַלְטִין אול פַהָּן קַנִינְרַן בַּרְמַנִי בִּילו ו הַוְרָסִן קווש בּוֹשֵׁלֵרי אוּסְטוּנֵא מִוּבַחִינִיג אוּל עוּלְנִיג ו דַא ג׳וּחְרָא קנִינִי טוְבַּסִין מִמְלִינַא אוּל מוֹבֶּח נִיג : ואת דא ג'ומְלא יִגִינִי בִּטַרְסִין ו גַּצִיבִּי בִּטַרִילְרִי יַג אוּסְטוּנְרַן קרבניניג

לְרְבָּנִינִיג אוֹל שְׁלְמִים לַרְנִיג ו דַמוּמַמְסִין אוֹל כּהוֹן אוֹל מִוְבַּחַמָּה לְּנִיג ו דַבּבְּּנָינִי הַשְּׁמִים לַרְנִיג ו דַבּוּמַמְיִיר אַנִּר : ואם בּבְּּנִינִי הַשְּׁמִים לַרְנִיג ו בְּבִּבְּנִינִי הַשְּׁמִים לַרְנִיג וּ בְּיִבְּנִינִי הַטְּאת לַה וֹ סַגְּלָם מִישִׁינִי נִּמִירְסִין אַנִּי : וֹכֹּהְ בַּיְבִּינִין קוֹלוּנְי בִּשִׁי אוֹל מוֹלְנִי בִישׁ אוֹל חַמְאת לָה וֹ סַגְּלָם מִישִׁינִי נִמִירְסִין אַנִּי הַמְּאת לָה ו נִּג וּ בְּכוּיְכִין אוֹל פֹהוֹ לִנִינְ הַאוֹל חַמְאת לָה וּ נִג וּ בְּבּינִינִי הַוֹלְנִי זוֹלְם בִּיְצְקְלְרִי אוֹל מוֹבָּחָינִי אוֹל מוֹלְנִי בִיּא ג׳וּמְלָא מִיבְּבוֹ אוֹל מוֹבְּחָי בִּיְנִי בּוֹלְייִ אוֹל מִוְבָּחוֹ נִיג זְרְבְּנִי אוֹל מִוְבָּח נִיג : ואת דַא ג׳וִמְלא יְנִינִי בָּמִרְסִין וּ בִּנְיִמְינָה אוֹל מִיְבָּח נִיג : ואת דַא ג׳וּמְלֵא יְנִינִי בָּמִרְסִין וּ אוֹל מִיְבָּח בִּי מוֹבְּחִי בִּיִּימִיה אוֹל מִיְבָּחְ נִיג זְרְבִּי אוֹל מִיבְּח נִיג : ואת דַא ג׳וּמְלֵא יְנִינִי בָּשִּרְסִין וּ אוֹל מִוְבָּח בִיּוֹ בְּוֹיִי בְּנִי מוֹבְּחְ מִיֹי בְּנִי מִוֹבְּח מוֹל מִוֹבְחַשְׁה וּ אוֹל מִיבְּחָשֵה וּ אוֹל מִיְנִיג בְּרְנִי אוֹל מִיבְּחָשֵה וּ אוֹל מִיְבָּחְשָׁה וּ אוֹל מִיְבָּחְ בִּי בּוֹינִי בְּיִי מִינִי בְּי אוֹל מִיְבָּחְ בִּוֹ בְּי אוֹל מִיְבָּם בְּוֹינִי בְּי אוֹל מִיְבָּם בְּוֹבְיוֹ אוֹל מִיְבְּי אוֹל מִיְבְּים בּוֹי בְּיוֹי אוֹל מִיְבָּחְשֵׁה וּ אוֹל מִיְבָּחְשָׁה וּ אוֹלוֹי מְרְבְּנִינִי אוֹל מִיְבָּחְשִׁה וּ אוֹל מִינְבִי אוֹל מִיְבָּחְימִיה וּ אוֹל מִיְבָּחְימִיה וּ אוֹל מִינְבִּי בְּי אוֹל מִיבְּי בְּיִי מִינִי בְּי בְּייִי בְּיוֹב בְּיִי בְּיִבְּי בְּיִי בְּיִי בְּי בְּיִי בְּינִי בְּיוֹי בְּינִי בְּיִים בְּיוֹבְיוֹ מִיבְּי בְּיוֹים בְּינִייִם בְּיוֹים בְּינִייִּ בְּיִי בְּיוֹם בִּינִי בְּי בְּיוֹם בְּינִים בְּינִייִּ בְּבְּינִיין בּיִּינִייִים בְּינִייִי בְּיִים בְּינִיים בְּיִים בְּייִים בְּייִים בְּייִים בְּינִיים בְּייִים בְּיִים בְּינִיים בְּיִים בְּיִינִייִּ בְּיים בְּיִים בְּיִיבְיי בְּיוֹים בְּיִינִי בְּיִים בְּינִייִּבְיי בְּיוֹים בְּינִיים בְּיִינִיים בְּיִינִייִים בְּיוֹבְייִים בְּיוֹייִיים בְּיִינִיים בְּיִינִייִי בְּיִים בְּיִיים בְּיִינִייים ב

אוּכְטוּנֵא אוֹל כּהַן יַזִינִי אוּצוּן כִּי יַזִיקְלִי בּוֹלְדִין דַבּוֹשַׁטִילְר אַגֵּר :

(ה) ונפש דַגֵּ׳ן כִּי יַזִיּקְלִי בּוֹלְטַה ו דַאִשִׁיטְטַה אַוַזִין אַנְטִנִיג ו דַאוֹל שַחַר

יא גורְדִי יַא בִּילְדִי וּ אָגַר אַנְלְטְמַמַה דַבּוּמַרְסִין גוּנְדִין וּ אָבָר אַנְלְטְמַמַה דַבּוּמַרְסִין גוּנָדִין וּ אָבָר אַנְלְטְמַמַה דַבּוּמַרְסִין גוּנָדִין וּ אָבָר בּיִיבְּנִיג וּ דִאָּעָר בּיִלְנִיג וּ יַא גוּבְדַסִינָרה מוּנְדֵר בּוֹלְנִיג וּ דִישִּירִין בּוֹלְנִיג וּ בִישִּירִין בּוֹלְנִיג בּילן וּ דִישִּירִין בּוֹלְנִי בְּגוּנְחָלִי בּוֹלְרִי : אוֹ יַא בִּי מִייְּטָה מוּנְדַרְלְּנִגה בּילוּן וּ דָאוֹל בִּילְרִי דַגוּנְחָלִי בּוֹלְרִי : אוֹ יַא בִּין בּי אַנִם אַטְּסָה בּיֹלְנוֹ בִּילְנִי בּילְן וּ דִישִּירִין בּוֹלְטֵה אַנְּדֵן וּ דַאוֹל בִּילְרִי בּוּלְנֵי בּיּלְרִי דְגוּנְחָלִי בּוֹלְרֵין : אוֹ בִילְנִי בּיּלְנִי בְּילְנִי בְּילְנִי בְּילְנִי בְּילְנִי בְּילְרִין : אוֹ בִילְיִי בְּילְנִי בְּילְנִי בְּילְרִין : אוֹ בִּילְרִי בִּילְנִינְ בִּילְנִי בְּילְרֵן : בּוֹלְנִי בִּילְנִינִי בִּי מִשְׁמַן בּוֹלְכָה בִּילְנִין וּ בִּישְׁמַן בּוֹלְכֵה בִּילְנִין וּ בְּעִבְּילְינִינִי יִיִּי נַאִּמְרִין בּוּלְרֵן : בּוֹלְנִי בִּי אַנְיִבְּי בִּילְנִינִי בִּי מִיּעְרִין בּוּלְרֵין : אוֹלְבְּיִבְּיְנִינִי בִּי מִישְׁנִין בּוֹלְירִי בְּוֹלְיִרְין : וּאִם בּיִּבְּיוֹין בּילְרִי בִּיּנְינִינִי בִּי מִיּשְׁתִּן בּילְנִינִי בִּי זִייְנִינִי בִּי זִייְנִינִי בִּי זִייִנְינִי בִּי זִייִנְיִבְן : וּאם דְּבִּירְנִינִי בִּי זִייְלְלִי בְּוֹלְייִינִי בִּי זִייְלְלִי בּוֹן : וּאם בּילִין : וּאם בּבִּין בִּייִנִי בִּי זִייְלְלִי בְּיוֹיִין : וּאם בּיִרְנִי מִּילְבְי בִּיוֹיְנִיי בִּי זִייְלְלִישְׁמִייְסִין בּּיבְּיִין וּוּנִיי בְּיִילְנִייִי בִּי זִייְלְלִישְׁמִירְסִין בִּילְיִי בִּיוֹן וּ בְּיִלְיִיִי בִּי זִיִיְלְלִי בְּיוֹי מִילְּבִי בְּיִינִי בְּיִינִי בְּיִי בִּיִיְיְיִי בְּי מִייְלְּיִים בּיוֹן וּי בְּעְלְייִבְייִין : וּאִבּין בּייִבְייים בּיּיִייִים בּיוֹן וּי בְּילְבְּיִים בּיוֹיְייִי בְּיוֹי בִייִיְיבִּי בְּיוּ בְּיִילְיִייִּבְי בִּיוֹי בִּייְיִיי בִּיוֹי בִייִילְּיִיים בּוּיוֹין בּיוֹ בְיִילְייִבְּי בְּיוֹי בִּייִיְיְם בּּילְיי בִּיוֹין בְּיוֹין בְּיוֹי בְּיוֹייְבְּיי בִּייִים בְּיוֹים בְּיוֹין וּיוּבְיוּבְילְייִים בּיוֹיוּים בְּיוֹים בְּיוֹי בִּיבְייִים בְּיוֹים בּיוֹין בְּיוּם בְּיִילְים בּיוֹים בְּיוֹים בְי

ווכרא

אַמְסִין בּשִׁינִי קַרְשִׁיסִינְרַן אַנְסַסִינִיג דַאַיִירְמַסִין : והוה דַסַצְסִין קנִינְרַן אול חַמְאת נִיג דוּבַנִי אוּסְמוּנֵא אוֹל מוֹפֹחַ נִיג ו באוֹל צֹלְגוֹ אוֹל צֹנְבוֹ ומֹסְמוֹ בּנְיְיִמִינִא אוֹל מְוֹבֵח נִיגוֹחַטְאת דִיר אוֹל : ואת דאוֹל אַבִּינְגְ׳ינִי קִילְסִין עוֹלָה מוֹרַצַא וֹ דַבּבְּבָרה אַמְסִין אִיּסְטוּנַה אוֹל בּהֵן ו יִויגּיְנַרן בִּי יַוּיְקְלִי בּוּלְרִי _דבּוּשְׁמִילִיר אַנֵר : ואם דאָנר קוּבָטִי וְשְמַסַה אָבִי קוּמְלַרְנָא יַא אָבִּי בַּלְלְרִינַה גוֹגוּרְצוּנְנִיג וּ דַגִּמִירְמִין בְּרְבָּנִינִי ו בִּי יַזִיקְלִי בּוֹלְרִיו אוֹנוּנְגִיי בַּיִין אוֹל בֵּבִּיצְנִיג ו אוֹב חַמְאת קהו קיימסין איסטונה נג דורמסן אסטנה מסיין ווירא המאת דיר אול: והביאה דגשירְסִין אַנִי אוֹל כּהָן גַא דַאַבוּצְלַסִן אוֹל כַהָן אַנְדֵן דוֹלוּ אבוצוּ בִּילַןשוּטְסוּסי ניו במישׁמְסון אול מִוְבּחַמַה ו אוֹמְלוּ קּרְבְּוְלֵרִי אוּסְמוּנֵא יְיָּי נִיג ו חַמְאת בִיר אוֹל : וכפר דַכַּפַּרָה אַשְׁסִין אוּסְטוּנָה אוֹל כּהֵן ו יַוִיגִי אוצוּן כִּי יַזִיְּקְלִי בּוְּלְדִיוּ בּוֹרְסִינְרֵן בּוּלַרְרָן דַבּוּשׁמִילִיר אַנֵר ו בַבּוּלְסִין אוֹל בּנִן נֵא מִנְחָה נִיבִּי : וירבר דַכווְלַדֵי יני משָה נַא דַמַה : נפש נֵין כִּי מַנְסַא מַנְסַק ו דַיִּזִיקְלִי בּוּלְסַא יַנְלִישְׁלִיִק בִּירַוּ קורשׁ צַרִינְדָן זֹנָי נִיג ו דַגִּמִירְסִין בִּשְׁמֵנְלִינִינִי יְנִי נַה ו סַנְרֵם קוֹצְקַר אוֹל קוֹיַדן וֹ קוִימִטִיג בּילו גומוש מסְקַל לַר ו מִסְקְלִי בִּילוֹ אוֹל קוֹדֶשׁ נִיג אָשְׁם נֵא : ואת דא נְנֵי כִּי יַזִיּקְלִי בּוֹרָרִי אוֹל קוֹרְשׁ דון אוֹדִכִּין ו דַא בַּשִׁינְנִיי פֿאַהוֹנִ אַלְמְאַיּנְטִן אַסְמוּנִא וּ בַוֹנְטִין אַנִּי כּנֵין נַע וּ בֿאוּך כּנֵין כּפַּלַנִי אַמְטִין אוּסְמוּנאַן קוֹצְקַרִי בִּילן אוֹל אָשֶׁם נִיג ו דַבּוֹשְׁמִילָר אַנֵר: ואם דַאָנַר נֵ׳ן בִּי יַנִיקְלִי בּוֹלְטַה וּ דַקִּילְטַה בִּירְנִי ג׳וּמְלַא מִצְוֹהְ לַרִינְדְן יְנִי נִיג בִּי קִילִינְמַסְלַרְדִיר וּ דַבִּילְמַדִי דַגוּנַחְלִי בּוּלְדִי ו דַצִבְּסִין ג׳וּרוּמִין: והביא דַגִּמִירְסִין קוּצְקַר סַגְּלַם אול קוירו ו קייסיג בילו אָשֶׁם נַה אול כהן נַא ו דַכַּפְּרָה אָטִסין אוּסְטוּנַא אול פהן ו ינְלִישְׁלִינִי אוצון כִּי יַנִילְדִי ו דַאוֹל בִילְמַדִי דַבּוֹשְׁמִילִר אַגר : אשם אָשָׁם דִיר אוֹרן גונַחְלִי בּוּלְמַא גונַחְלִי בּוֹלְמַא גונַחְלִי בּוֹלְתִי יִנָּי בּוֹלְתִי דִּר : וידבר דַסוֹזּלְדִי יְנָיּ משה גא דמה : נפש ג׳ן כּי יוִיקְלִי בּוּלְסָאוֹ דַמַנְסָא מַנְמַק יְיִּי נַאוּדַמִנְסָא דוֹסמונא אַמַנמְקא ו יַא קול בִילו קויְנִנְנָא וַיִא זוּלום בִּילוֹ יֵא דומופ אַלְרִי אִיסָא בוֹסְמִינְבֵן : אַנְ הַאַ מַפְּכָא מִס בּוּלְדְנִנִי וּ בַמִּנְסֵא אַנְּב בַאַנְהְ אִמְסַב יַּלְנְנִי וּ ביר אוצון ג'ומלאדן כי קילר אול אָדְם ו יויִקלי בולמה אַלר בילן : והיה דבולגי

ויכרא

דַבּוּלְנֵי כִּי יַזִיּקְלִי בּוּלְסַה דָפִּשְּׁמֵן בּוּלְסָא ו דַקִּיְמֵּרְסִין אוֹל זוּלוּמְנִי כִּי זוּלוּמְלַרִי ו יַא אוֹל דוּטוּפּ אַלְּמָלְי כִּי דוּטוּפּ אַלְרִי ו יַא אוֹל אַמַנִּטְנִי כִּי אַמַנַּטְ קוּוּלְיִי בִּירְנִסִינָּא ו יַא אוֹז טֵס בּוּלְנַנְי כִּי טַפְּטִי : או יַא ג׳וּמְלַאדֵן כִּי אַנְּט קּוּוּלְרִי אַנְיג אוֹצוּן יַלְנַנְנָה וֹ דַאוֹדְסִין אַנִּי בָּשִׁי בִּילַן ו רַבִּשִׁינְנִי כִּי אַמְסַר אוֹל פַהָן וֹ הַבְּשִׁינְנִי בִּי אוֹל אַנִי בִּשְׁמַן בּוֹלְנֵן גוֹנוְנְדֵא : ואת דַא אוּסְטוּנָא ו אַנִּי בִּילון אַנְים קוֹצְקר אוֹל קוּיְדֵן ו הִייַסִיג בִּילַן אָשְׁם בּּוּלְנֵי נִי גוֹיְרָסִין יְיוֹי בָּה ו סַנְּלַם קוֹצְקר אוֹל קוּיְדֵן ו הִייַסִיג בִּילַן אָשְׁם בּּוּלְנֵי נִיג דַבּוּשִׁכִּילִר אַנִּר ו בִּיְּכָּיה אַנִּים אוּצוּן ג׳וּמְלַאדְן בִּי קוֹיֵר ו גוּנְחָלֵי זּי יְנִי נִיג דַבּוּשִׁכִילִר אַנִּר וּ בִּיְרָסִי אוּצוּן ג׳וּמְלַאדְן בִּי קוֹיֵר ו גוּנְיְנָדִי בּילִן אָשְׁם בּוּלְמַה אַנִיל בָּהְן גוֹנִיבְּר וֹ בּנִים בּוֹלְכֵּה אַנִילְר אַנְינִיה בּילִן אִינְים בּוֹיְבְילִר אַנִּיל בַּיְלַן אָשְׁם בּוּלְמַה אַנִילְר אַנִילְר אַנִּיל בִּוֹנְים בּיּיִבְילִר אַנִין אוֹנִי בִּילְם בּיּלְנִים בּילִים אוֹנִייל בִּילְן אַמְיַבְילִי וּ בּיִבְּיִם בּוֹיבְילִי וּ בּיִבְּיילִי וּ בִּי בִּילְם בּוּבְּנִילְיך בּי אַנְיבְּיל בִּי אוֹנִים בּיוֹילְייִים בּיוֹים אַיִּיבְיל וּ בּיִּבְיילִר אַנִייִבְיּי בִּי בִּילְם אִינִים בּילְם אַנִיילְי בּשְׁיבְילִר אַנִיים בּילְנִים בּיוֹיבְילְים בּיּילְם בּוּבְּיבִילְיך בּוּעְנִילִי בּיּילְם בּוּבְּיבִילְים בּיּילְם בּייִּבְיּים בּיּילְם בּיּבְּים בּילְנִי בּיי בּיְנִילְייִיים בּיּבְּילְים בּי בְּיבְּים בּי בִּילְם בּי בּילְם בּיּילְיבִיה וּי בּיּילְן בּי בִּילְיבְן בּי בִּיבְּים בּילְנִיים בּי בּילְן בּי בּילְנִייִילְייִין וּ בּי בִּילְנִייִייִי בּי בּילְּיבְיי בּיילְן בּיי בִּיבְּיבְיי בּי בִּיבּי בּילְנִיי בִּי בּייבְּים בּי בּילְנִי בּי בִּילְים בּי בּילְים בּיי בּילְיי בּיי בּילְנִיין בּיי בְּיבְּים בּיי בְּייִבְּיי בִּיי בּי בְּבְּים בּייבּיים בּי בּייִּילְים בּיּילְים בּי בּייבְּים בּייייבְייים בּי בּייבְים בּיי בּייבְּים בּיוּינְייִיים בּי בּיי בּייבְיי

פרשת צו

וירבר יני אָל משָה לֵאמר : דַסוֹיְלַדִי יְיָ משָה גַא דִמָּה : בּסוֹיְלַדִי יִי משָה גַא דַמַה : צו סִימִּרְלְּגִין אַהָרן גַא דַאוֹגְלַנְלְרֵינָא דָמָה ו פּוּדִיר חוּרָסִי אוֹל גַצָא אוֹל אַרְיִר אוֹל שוֹיָה אוֹצֵגִי אוּסְמִינָא פּיִאוֹל מִוְבֶּחַ אוּסְמוּנָא ו גִימְלָא אוֹל גַצָּא אוֹל בָּהָן בִּיסִי אוּפְּרַק ו דַבִּיסִי פּוּנְצַב כִּיִיסִין אָמָי אוּסְמוּנָא ו דַאִירְסִין אוֹל פּהֹלְנִי ו כִּי אוֹרְמֵר אוֹל אוֹם אוֹל עוֹלְנִי אוֹל מִוְבֶּחַ אוּסְמוּנָא ו דַאִּירְסִין אוֹל פּוּלְנִי ו כִּי אוֹרְמֵר אוֹל אוֹם אוֹל עוֹלְנִי אוֹל מִוְבֶּחַ אוֹסְמוּנָה ו דַסוֹיְסִין אַנְּרָר אוֹל מוֹבְּהַ אוֹלְנִי אוֹל מִוְבָּחַ אוֹל מוְבָּחַ אוֹסְמוּנָה ו דַמוּיְסִין אַנְּבַּר בִּיִיסִין אוֹל מִוְבָּחַ אוֹסְמוּנָה ו דִמִּיְסִין אוֹל מִוְבָּחַ אוֹל מוֹנְבָּח ו בִּילְוּ אַרְבִיּלְר אוֹל מוֹנְהַין אוֹל מִנְּמָה וֹן ז בִינְשִׁמִירְסִין אוֹל מִוְבָּח אוֹל מִוְבָּח אוֹל מוֹנְבָּח וּ בִּילְוּ אוֹל שִׁלְּלָיי נִינֹ ו אַרְמִיה נִינִי ו בּוֹלְנִי אַבּוּדְיר הוֹרָסִי אוֹל עוֹלְנִי ו בִיבּין אוֹל מִנְמָה לַוְנִי אוֹל מִנְמָה נִינִ ו דִירְשְׁמִיּרְסִין אוֹסְמוּנָה אוֹל עוֹלְנִי ו בִיבּי אוֹלְלְלֵיי נִיוֹ אוֹל עוֹלְנִי ו וֹבוּים אַנְמִים לַּרְנִי בּירִסִי אוֹלְנְלְייִי בּירְסִי אוֹנְלְלְשְׁמִירְסוֹן אוֹל מִנְחָה נִיג ו יוּבוּקלשְׁמִירְמָוֹ אוֹל מְנְתָּה נִיג ו יוּבוּקלשְׁמִירְסוֹן אוֹל מִנְתָה נְינִ ב בִּיבְים אוֹלְלְלֵיי נִי גֹּג אַלְרִינָּא וּ בִּילְן אוֹל מִנְתָּה נְיִנֹ א אוֹל מִנְתָה נְינִי ב וֹהרים דַאִייִרְסִין אַנְלְּלְי אָבוֹן אוֹל מִנְיְתָּה נְיִג וֹ בוּי בּיִרְי אוֹל מִנְתָּה נִיג וּ בּיִב וּ אִלְיִינָּא מוֹל מִנְתָה נִיג ו וּבּים בִיִּיְם אוֹל מִנְיְרָן אוֹל מִנְיְתָּה נְיִג וּ הִיים דְּיִי אִינִינְן אוֹל מִנְתָה נִיג ו הרים דְּאִייִים אַוֹל מִנְתָּה נִינִי בּילן אוֹנִינְרָן אוֹל מִנְתָּה נְיִנֹי בּי בִיים אוֹנִינְנִי בְּן אוֹל מִנְיתָּי בְּי בִּיל מִבְּחָ מוֹבְּחָ מוֹי בְּי בִּי בְּיִים אוֹי בְּיִי בְּיִי אִינְי בְּיִים אוֹנִייְ וּיִי בְּיִים בְּיִבְי בְּיִים מִּיְנִי עִי בּיִבְי בְּיִים מִינְי בְּי בְּיְי אָּנְי שִּנְיתִים בְּיִי אִינְי עִינְיי מִיוֹם בְּיִבְי בְייִי בְּיִי אִינְיְיי וּיְיִי י בְּיִי אִינְיי בְּיִּי בְּיִי אִי

בינינדו ו בא ניומרא אול פסיינני ו פן אול מנחה איסטינא ו דמושטסין אול ימוֹבֶּחַטַאוֹ קוֹקוֹסִי קבּוּרְלִיְקנִיג טוּטְסוּסי יָנִי גַה : והנותרת דאול בַּלְנַנְנִי צֵּנְבְןוּ ששטונלר אַבְּרן דַאוּגְלנִלְרֵי מַצְלָר אַשׁלְסִין אַיְרוּכִסִי יהֵדָה ו אַוְבַּרִינְדָה אֹהֶּל מוֹער ניג אַשַׁסִינְלַר אַנִי : לא פּישְׁמַכִּין דְמֵין וּ פַּיְלַרִינִי וַרְרִים אַנִי אוֹמְלוּ קַרְבְּנְלַרִימְדֵן וְקוֹהָשִׁי קּוֹרֶשׁ לַרָנִיג דִיר אול ו חַמְאת גוִבִּי דַאְשְׁם גִּיבִּי : כל גיומְלֵא אָרְכָּכּ כִּי אֵוֹגְלַנְלַרִינְדַה אַהַרו נִיג אַשׁסִין אַנִי ו רַסִימִי דונְינִיג דַווּרְלַרִיגִיוְגַהו אוֹטְלוּ קָרְבָּנְלְרִינְדֵן זְיָי נִיגוּ הַר כִּים כִּי טִיְיַכַה אָלַרְגַה אַיִרוּכִםי בּוּלְסִין : וידְבר יַבְּכוֹיְנִיִרִי יְיִי מֹשֶׁה נָא הָמַה : זה בּוּדִיר קְרְבְּנִי אַהְרן נִיג דַאוֹגְלַנְּלְרִינִיג ו בּי יובוּקלשְטִירְגוִלֵר לֹי נַה וּ סִילִינְגן גוּנְבוּן סוּגְרֵא בִּי סִילְטַא אַנִי וּ אונוּנְגִ׳י פִיי אוֹל בַּבִּיצְנִיג אוֹבַ מִנְחָה דַיִּים ו יַיִּרִמִיםן אַרְטַה דָא ו דַיַרִימִיםון עַבְשַמְרַא : על פוַה אוצונה יַג בִּילו קילינְסִן ו קוורולְגַנְנִי גִפִירְטְגִין אַנִי ופִּישְׁמַבְּלַרִין) ינשְׁמִירְגִין דִילִימְלָר מִנְחָה בִינִי ו יובוקלשִמִירְגִין קוֹקוּסִינֵא קבּוּלְלִיְקְנִיג ינָי גַה : הן דאול סילינגן כהן אורנינא אוגלנלרינדן קילסין אני ו בסימי רוניניג יני בּאוֹ הַבְּמִיל מוּטַמִּיְסִוֹ : וירבר דַבוּוְלַדִי יְנִי משָׁה נַא דֵמַה : רבר כוּוּלָגִין אַהַרוֹ נַה דאוּגְלַנְלַרִינַא דִמָה ו בּוּרִיר תּוֹרְסִי אול חַמָאת נִיג ו נָא יִהְדַה כִּי סייוּלְכַה אוֹל עוֹכָה ו סייוּלְסִין אוֹל חַטְאת אַלְרִינַה יוֹנָי נִיג ו קוֹרֶשִׁי קוֹרֶשׁ לַרְנִיג דיר אול : הכהן אול כהן אול חַמְאת אַמְכַּן אַנִי אַשַׁסִין אַנִּי וּ אַיְרוּכְסִי יִרְדַה אַשַּלְסִין וּאַוְבַּרִינַדה אֹהָל מוֹעֵר נִיג : כל הַר נֵא כִּי מִייְסַה אָטִינַה אַיְרוּכְסִי בולכין ו באַנר סִצְלְסָא קַנִינָדן אול אוּפָרַק אוּסְטוּנַא ו כִּי סַצִילְסַה אוּסְטוּנַה ו יובְגִין אַיְרוּכְסִי יִרְדַה : וכל דַצורַפּ סַגִיט בִּי פִּישְׁסַה אִיצִינְדָא סִינְדִירִילְּסִין ו דאגר בּקיר סִנִימְרַה בּישְׁסַא ו דַקִירִילְסִין דַיְיַסְלְסִין סוְבַלֵּר בִּיַלן : כּל הַר אָרְכַּב כּהֵן לַרְבָה אַשַׁסִין אַנִי ו קוֹרָשִׁי קוֹרְשָׁי לֹרְנִיג דִיר אוֹל : וכל דַהַר חַבְאת בּי גִירְגִיזִילְסַא קַנִינַרן אהָל מועד גַה ו בְּפְרֵת אִטְמַגַה קוֹדָשׁ רַה אַשַלְמַסִין ו אוטְקַה בּוּיְרוּרוּלְסָן :

וואת דבוריר תורסי אול אָשְׁם ניג ו קוֹרִשׁי קוֹרָשׁ לַרְנִיג דִיר אוֹל:

193

במקום נא וְרָרָה כִּי סוְיְסַלֵּר אוֹל עוֹלָה נִי ו סויְסִינְלַר אוֹל אָשָׁם נִי ו דַא קניני פַּצְּסִין אוֹל מִוֹבֶּחַ אוּסְפוּנהַ צוּפְצוּרְרַה : ואת רא ניומְלא ינִיני יובוקלשְטִירְסִין אַנְדֵן ואוֹל קוּיְרָקְנִי דַאוֹל יַנְנִי אוֹל יַבְּקוֹ אוֹל קַרִינְנִי : ואת דַא אָבִי אוֹל בּוֹנְרַכְּלְרָנִי ו דַאוֹל יַנְנִי כִּי אוּסְמְלַרִינַה ו בִּי אוֹל אָמִיוְלֵר אוּסְמוֹנַה ו דאול אַרְשִיקְסִינִי אוֹל בַּגִיר אִיִּסְשוֹנַה ו אוֹל בּוֹגְרַכְּלֵר בִּילֵן בִּשַּרְסִין אַנִי : והקשיר רַמוּמַמְמִכּין אַלַרָנִי אוֹל כּהֵן אוֹל מִוְבֵּחַמַה ו אוֹמְלוּ קְרָבּן יְנָי נַה ו אָשְׁם דִיר אוֹל : כל הַר אָרְכַּבּ פּהָן לַרְיַה אַשַׁסִין אַנִי ו אַיְרוּכְסִי יִרְבה אַשַּׁלְסִין ו קוֹדֶשִׁי קוֹרָשׁ לַרְנִיג דִיר אוֹל : כחמאת נְצִיכִּי הַמְאת עַלֵי אָשְׁם כִּיר תּוֹרָה אַלַרְגַה וּ אול כהן כי כפרה אִמְסַה אַנִיג בִּילֵן וּאָגַר בּוּלְסִין : והכהן דאול כהן אול יובוּקְלַשְׁאִירְגַן בִּישִׁינִיג עוֹלְסִינִי ו מִרִיסִי אוֹל עוֹלְנִיג אוֹ כֹּהַן נֵא כַּי יובוְקלַשְׁמִירְרִי אַנִיואַנֵר בּוֹלְכִין: וִכַל דַהַר מִנְחָה בִּי פִּישְׁסַה תַּנְדְוּרְרָה ו דַהַר מִלְינְנַן בּנְנִירַדָה יַאַ טַוּהַ אִיצִינְדָא אוֹל כּהֵן נַא אוֹר יוּבוּלְלַשְׁטִירְנַן אַנִיואַנַר כּוֹלְסִין בּ וכל דַהַר מִנְחָה קַרִישִׁילְגַן וַג בִּילַן יָא קורו ו ג׳וּמְלַא אוֹגְלנְלַרִינִא אַהְהן ניג בּוּלְסִין ו הַר בִּישִׁיגַא ַקְרָרָשִׁי גִיבִּי : וואת דַבּוּדִיר תוֹרָסִי אוֹל שְׁלָמִים קָרְבְּנִינִיג ו כִּי יוֹבוּקְלַשְׁמִירְכַא יֹנִי גַה : אם אָגַר שוֹבּוְרְלִיכּ אוֹצוּן יובוּקְלַשְׁמִירְכַה אַני ו דַיוּבוּקְלַשְׁמִירְסִין אוֹל שוּבּוּרְלִיבּ קְרְבָּנִי בִּילַן קּלִין מַצְּלֵר קרישׁילְנִנְּלַר יַג בִּילַן ו דַיוּפָּקא מַצְלַר סִילִינְגְנַלר יַג בּיַלן ו דַקווּרוּלְגן אוֹוגי ו קּלינְלֵר קרישׁילְננְלַר יַג בִּילֵן : על חָמֵץ בַּלִין אוֹמְמַבְּלַר בִּילֵן יוּבוּקּלַשְׁמִירְכִין כְּרְבְּנִינִיוּ שובורְלִיבּ שְׁלְמִים קְרְבָּנִי נִיג יַנִינַא : והקריב דיובוּקְרַשְׁמִירְּסִין אַנְדַן בִּירֵר אַבְּמָב הַר פִיר סוי ְקִרְבְּן אַבְּמִגיְנְדֵן ו אַיִיְרְמַק ינִי נַא ו אוֹל פּהֵן גַה אוֹל סַצְקּוֹ אול שְלְמִים קַנִינִי אַגַר בּוּלְסִין : ובשר באמי שובּוּרְלֹבּ קרְבָּנִינִיג שׁלְמִים לַרִינֵיג - ו לָרְבּן אִמְבַן גונוְגַרה אַשׁלְסִין ו קונִמַסִין אַנְדַן אִרְטַנַה הֵגִין : ואם דאֵגַר אַנְיַים יַא ג׳וֹמַרְמְלִיק אָמַה שְׁלָמִים קּרְבָּנִי וֹ יוּבּוֹקְלַשְׁמִירְדִיגִי גוּנְדָה קְרְבָּנִינִי אַשַלְסִין דַסַבַּחִינַדַרָאוֹ אוֹל קַלְנַן אַנַדַן אַשַּלְסִין : והנותר דאול קּלְנַן אוֹל שַׁלָמִים אָמִינְדָן ו אוֹל אוצִוּנְנִיי גוּנְדָה ו אוֹמְקָה בּוּיְדוּרוּלְמִין : ואם דָאנֵר אַשְּׁלְמָא אַשִּׁלְמָה אִמִינְדִן שְׁלְמִים כְּרְבָּנִינִיג אוֹל אוצונְנְיִי גוּנְרֵה מַקְבוּל בּוּלְמַכְמִירּוּ 375

אוֹל יוּבוּקְלַשְׁטִירְגַן אַנִי קַרְבַּן בַיִילְמֵז אַגַרו בַיִם ופסולו בוֹלִיר ו דַאוֹל גַ׳ן אול אַשַנּן אַנְדַן גוּנַחִינִי צָבַר: והכשר דאול אָשׁנּן אַנְדַן גוּנַחִינִי צָבַר: והכשר פונְרֵר שׁיִינִיג אַשַׁלְמַכִין ו אומְקַה בּוּיְרוּרוּלְכִין ו באול קוָדְשׁ אָמִי הַר מִמְיוֹ פּישִׁי אַשַּׁכִין לְּרָשׁ אָמִינִי : והנפש דַאוֹל נֵ׳ן פִּי אַשַּׁכָא אָמ ו אוֹל שְׁלְמִים בָּרְבָּנִינְדֵן כִּי יֹנְי נַא ו דַמוּנְדַרְלִיגִי אוֹלְטָא אוּזַרִינְדֵא ו דַבָּמִילִר אוֹל גַ׳ן אוֹלוּסְלַרִינְדַן: וגפש דַנ׳ן כִּי מִיְיסָא ג׳וּמְלָא מוּנְדַרְגַה ו מוּנְדַרְלִיגִינַא אָדָם נִיג ו יַא מוּנְדַרְ הווֹרָגַא ו יַא ג׳וּמְלַא מוּנְדַר אָיַרְנְצִיגַה ו דְנִיזְלִי אַשַׁכַה אָטִינְדַן אוֹל שְׁלְמִים בְּרְבְּנִינִיג בִּי יָנִי גַא ו בִּסִילִיר אוֹל גַ׳ן אולוּסְלַרִינְדַן : וירבר דַסוּוְלַדִי ינָי משה גא המה : דבר פוזלגין אוגלנלרינא ישראל ניג המה ו הץ יגין אוגויןניג בָּא קוֹינִיג נִא אֵצְבִּינִיג אַשְּׁמַגִיז : וחכב דַיַנִּי וְבַלְה נִיג דַיַנִּי מְדַבְּה נִיג קילינסיון הַר אִישְׁכַּה אַנְצֵא אַשַׁמַה אַשַּמָנִיז אַנִי : כי זיַרא הַר אַשַנַן כִּימְסַא יַג אוֹל תּוֹנְרְדֵוֹ וַ כִּי יוּבוּיְקַלַשְּׁמִירִיר אַנְדֵן אוֹמְלוּ קָרְבֵּן יְיִיֹי גַא ו בָּסִילִיר אוֹל ביולי אַשַבוּן ב׳ן אולוּסְלַרִינְבוּן : וכל בהץ קו ני אַשַׁמַגִיז ו ג׳וּמְרַא אומורַשְלַרְנְזָרָא ו נָא קוּשְׁטֵּן נָא דַתּוּוּרְרַן : כל ג׳וּמְלַא גַ׳ן כִּי אַשַׁסַה הִץ רבר סוֹזְלְנִין אוֹנְלַנְלְרֵינִא יִשְׁרָאֵל נִיג ו אוֹל יובוּלְלְשְׁמִירְנֵן שְׁלְמִים לְרְבָּנִינִי יָיָּי בהו גמירסין קרבניני יוו בה קרבנינדן שלמים לרניג : יריו קרבנינ גמירסינלר אוֹמְלוּ בְּלְבְּנַלְרין יני ניג ו אול יוני אול מוש בִּילֵן גִמיְרְכִין אַני ו אול מושני סללמגא אני אוגונא סללמק אַרִרינא יני ניג : והקטיר דמוטיסין אול פּהן אול יגַנִי אול מובחטהודבורסין אול טוש אהרן גה דאוגלנלרינא: ואת דא אול אוג בּוּטְנִי והֵיגִיז בּוֹטַרִיפּ סַלְלַמַקאול בּהוֹגַהוֹקְרְבְּנָלְרֵינְדְן שׁלְמִיְםלריַגִּיזִניְגִּיהמקריב אול יובוּקלשְׁטִירְגוּן אול שְׁלְמִים ַקנִינִי ו דַאוֹל יַנְנִי ו אוֹגְלַנְּלַרִינְרֵן אַתְרן נִיג ו אַגַר בּוּלְסִין אוֹל אוֹג בּוּט בּוְגַה : כי זירַא אוֹל אוגונא סַלְלַמַק טוּשִינִי דַא אול יוקרי סַלְלַמֵּק בּוּטוּנִי ו אַלְרִים יַאנִינְרֵן אוֹגְלַנְלַרִינִיג יִשְׁרָאֵל נִיג ו שְׁלְמִים קָרְבְּנְלַרִינְדֵן וְ דַוֹרָדִים אַלַרְנִי אַהַרן אול כּהֵן גָה וּ באוּגְרַנְלַרִינַה רַסִימִינַא דוּנְיַנִיגוּ יאנינְרוּ אוֹגְלַנְלְרִינִיג יִשְּרָאֵל נִיג: זאת בּוּרִיר בּוּיוּכְּלִיכּ בַּאִיי אָרְרוּ נִיג דבּוּיוּבְּלִיכּ

פַּאיי אוֹגְלַנְלַרִינִיג ו אוֹמְלוּ לְּרְבְּנְלָרִינְדֵן יְיָי נִיג ו יוּבּוּקְלַשְּׁמִירְגוּ גוּנְדֵן אַלַרְנִּ בּּ הִימְרְלִיכּ רִסִימִי דַווֹר לִרִינִה סִיְלְרִינִּי גוּנְדֵן אַלַרְנִי וּ נַאנִנְדֵן אוֹגְלַנְלַרִינִיג יִשְּׂרָאֵל נִיג אוֹמִיְרְלִיכּ רְסִימִי דַווֹר לַרִינַה : גוּנְדֵן אַלַרְנִי ו נַאנִנְדָן אוֹגְלַנְלַרִינִיג יִשְּׂרָאֵל נִיג אוֹמִיְרְלִיכּ רְסִימִי דַווֹר לִרִינַה : זאת בּוּדִיר אוֹל תוְרה ו עוֹלְגַה מִנְחְגָה ו דַחַמְאַת כַּה דַאִּשְׁם גַה ו דַאוֹל שְּלְמִים כְּרְבְּנִינַה : אשר בּי סִימִרְלַדִי יִיִּי משְׁה גַא בַּרִינִא סִינִי נִיג ו סִימִרְלַדִי יִיִּי גוּנְדֵא אוֹגְלִנְלַרִינֵא יִשְּרָאֵל נִיג ו יוּבוּקְלַשְּמִירְמַגְה מִינִי נִיג ו סִימִרְלַדִינִי יִיְּי גַה יַבּנִינָדא סִינִי נִיג:

וירבר דַכוּוְלַרִי יִּי מְשֶׁה נַא דִמַה : קח אַלְגִין אַהְרן נִי רַא אוּלְבִינִי בִּירְנַסִינַה ו דַאוֹל אוּפְּרַקְלְרְנִי דַא אוֹל כִּילְמַבּ יַגִינִי ו

ראול הַשְאת בּוּגַסִינִי רַא אָבִּי אוֹל קוּצְקרְלָרְנִי וּ דַאוֹל מַצַּלֵר מִלְלִיסִינִי ואת דא ג׳וְמְלֵא אוֹל גַ׳מַעְמְנִי מוֹפְּלַגִין וּ אָשִׁיגִינה אֹהָל מוֹעֵר נִיגּ: ויעשׁ דַקּילְרִי משֶׁה גַצִיבִּי סִימַרְלַרִי יִינִי אַגַר ו דַמוּפְּלַנְרִי אוֹל גַ׳מַעם אַשִּיגִינַה אָהָל מוער ניג : ויאמר דַאַיְמְמִי משה אוֹל נֵ׳מַעִמְקַה ו בּוּדִיר אוֹל סוז כִּי סִימַרְלַדִיי מוּער נִיג : ויאמר דַאַיְמְמִי משה אַהַרן נִי דַא אוּגְלַנְלַרִינִי ו דַיוּבְרִיי מֹשֶׁה אַהַרן נִי דַא אוּגְלַנְלַרִינִי ו דַיוּבְרִיי אַלְרָנִי סוּכְלַר בִּילַן : ויתן דוֹרָדִי אוּסְטוּנָא אוֹל גוּלְמַבְּנִי ו דַקוּשָׁדִי אַנִּי אָנְצְקִיר בִּילֵן וֹ דַגִייְרִירְרוּ אַגַר אוֹל שׁוּשְׁמֵנְנוֹ וַ דְנִירִי אוֹסְמוּנָא אוֹל אָפּר נִי וּ דַקושַרי אַנִי בַגִּי בִּילַן אוֹל אָפּוֹר נִינ ו דַבַּגְלֵרי אַנִי אַנִיג בִּילַן : וישם בַרקוּיִרי אוְסטוּנֵא אוֹל חשָׁן נִי ו דַוְרָדִי אוֹל חשָׁן נַא אוֹל אוּרִים נִי דַאוֹל תּוּמִים נִי : דַקוֹיְרִי אוֹל כַרִיק נִי בַּשִׁי אוֹסְטוּגַא ו דַקוֹיִרי אוֹל כַרִיק אוּסְטוּנַא בּרְשִיסִינַה יוּיְלַרִינִיג אוֹל אַלְמִין צִיצַב נִי אוֹל קוֹרֶשׁ מַנִּ׳ינִי ו נִצִיכִּי סִימַרְלַדִי יֹנִיּ משָה גַא : ויקח דַאַלְרָי משָה אוֹל סִילְמַבּ יגִינִי ו דַּסִילְרִי אוֹל מִשְּבְּנְנִי בא גיומלא ני כּי אַנְבע ו באַרוּכָסי אַמְאָי אַלַרְנִי : ויוּ מוֹבֹת אוּסְמוּנא יִרִי בֹרַמְלֵר ו דַיִּילְרִי אוֹל מִוְבֹּחַ נִי דָא גִּיוּמְלֵא מֵּ־מְלֵרִינִי באול לוּלְנֹינִי בא מַסְלְבּנִינִי אַיְרוּכְסִי אִמְטַנַה אַלְרָנִי : ויצק במוֹכְמִי יַנִינְרָן אול סילמַבְנִיג ו בַּשִׁי אוּסְטוּנַא אַהַרן נִיג ו דַסִילְרִי אַנִּי אַיְרוּכְסִי אַטְמַנַה אַנִּי : ויקרב דַיוּבוּקלשְׁטִירָרִי משֶׁה אוֹגְלִנְיִין אַהְרן נִיג ו דַּגִייִרִיי אַלְרָנָה גוּלְמַבְּלֵר וּ קושרי.

לְּישׁרִי אַלַרְגִי אִינְצְקִיר בִּילַן ו דַגִּיִירִירִי אַלַרְגַא אוֹבְּסֵב קבוּלְְרֵר ו גִצִיכִּי סִימַרְלַדִי ייני משָה גַא : ויגש דיוּבוּקלשְטִירְדִי אוֹל חַטְאת בוֹנְסִינִי ו דַמַיִרי אַהַרן דַאוֹגְלַנְלַרִי קוֹלְלַרִינִי בַּשִׁי אוּכְטוּנָא אוֹל חַטָאת בּוּגַסינִיג : וישחש דַכוּיְרִי דַאַלְרִי משֶׁה אוֹל קּנְנִי וּ דַוֹרְרִי בּוּצְקַקְלַרִי אוּסְטוּנַא אוֹל מִוְבַחַ נִיג צוּפְצוּבְרַה בּרְמֵגִי בִּילֵוֹ וַ דַפַצְמִי אוֹל מִוְבֵּחַ נִי ו דַאוֹל בַּוְנִי מוֹבָמִי מִמָּלִינֵא אוֹל מִוְבֵּחַ נִיג ו בַאַרוּכְסִי אִמְטִי אַנִי כְּפָּרַת אַמְמַבְּבֵּא אוּסְמוּנֵא : ויקח דַאַלְרִי ג'וּמְלֵא אוּל יַנְנִי כָּי אוֹל קַרִין אוּסְטוּנַא ו דַא אוֹל בַּגִירְנִיג אַרְטִיקְסִיכִינִי ו דַא אָבִּי אוֹל בּוֹנְרַבְּּלַרְנִי דא יגַלרִינִי ו דַמוּמַמְטְמִי משָׁה אוֹל מִוְבֵּחַמַה : ואת דַאוֹל בּוגַנִי דָא מִרִיכִינִי בא אֹאָנִי בא שׁוֹנִינִי פּוּיְרוּלְנִי אוֹמְלֵבוּ ו אַבוּלְרוּ אִיכֹנִי ו נֹאִפּי סִמְּלְנִי יְנֵיּ משה גא : ויקרב דיובוקלשִמיְרְדִי אוֹל עוֹלָה קוֹצְקרִינִי ו דַמַיִרִילַר אָהַרּן דַאוֹנְלנְלֵרִי קוֹלְלַרִינִי בַשִׁי אוֹסְשוֹנַא אוֹל קוֹצְקּרְנִיג : וישׁחם דַסוּיְרִי דַסַצְטִי משה אול קַנְנִי וַ אוֹל מִוְבֵּחַ אוּסְטוּנֵא צוּפְצוּכְרַה : ואת דַאוֹל קוֹצְקּרְנִי בּוֹבוּמְלְדִי בּוּבוּמְלֵרִינְגַ׳ה ו דַמוּמַאְמִי משָה אוֹל בַּשְׁנִי דַאוֹל בּוּבוּמְלַרְנִי דַאוֹל בַריוֹ יַנִינֵי : ואת דאול בַרינְנִי דַאוֹל אַיִּקְבַּרְנִי יוּבְדִי כוּבְלַר בִּילֵן ו דַמוּמַמְמִי משָה הַבְּמִיל אוֹל קוֹצְקַרְנִי אוֹל מִוְבַּחַמַהו עוֹלָה דִיר אוֹל קוֹקִיסִינַה קבּוּלְלִיְקְנִינו אוֹמְלוּ לְרְבָּן דִיר אוֹל ינָי נַא ו נִצִיבִּי סִימַרְלִדִי יָנְי מֹשֶׁה נַא : ויקרב דַיוּבוּקַרַשְׁמִיְרָדִי אוֹל אָבִינְנִי קוֹצְקּרָנִי אוֹל מָלוּאִים קוֹצְקַרִינִי ו דַמַיִּדִילַר אַהְרן דַאוֹרָי לְיַלְרִי בִּשִׁי אוֹסְמוּנֵא אוֹל קוֹצְקּרְנִיג : וישׁחִם דַכּוּיְרִי דַאַלְרִי משה קנינדן ז דורדי ימשני איסמונא אול אוג קולגיניג אַהַרן ניג ו דַבַּשׁ בַּרְמִגִי אוסְטוּנָא אוֹל אוֹג קוֹלוּגִיג ו דַבַּשׁ בַּרְסַגִי אוּסְטוּנַא אוֹל אוֹג אַיגִינִיג : ויקרב ביוּבוּקְלַשְׁשִׁירִדִי אוֹגְלַנְּלַרִין אַהַרן נִיג ו בוֹרָדִי מֹאָה אוֹל קַנְּדֵן יִמְשַׁגִּי אוּכְמוּנֵא אול אוֹג לוּלַלְרִינִיג ו דַבַּשׁ בַּרְמַגִי אוּסְטוּנֵא אוֹל אוֹג לוּלְלֵרִינִיג ו דַבָּשׁ בַּרְמַגִי אוסְטוּנֵא אול אוֹג אַיַּקְלַרִינִיג ו דַסַאָאַי מֹשֶׁה אוֹל קַנְנִי אוֹל מִוְבֵּחַ אוּסְטוּנַא צוּפְצוּבְרַה : ויקח דַאַרְרִי אול יַגְנִי דַאוֹל קוּיְרוּקנִי ו דַא ג׳וּמְלַא אול יַגְנִי כִּי אול קריני דאול אוג בוטני : וססל באול מצלר סלליסינדן כי אלדינה יני 23

73

ניג ו אַלְרי בּיר קלין מִצָה ו דַבִּיר קלין יְגְלִי אוֹמְמַבּ ו דַבִּיר יוְפַקְה וּדַקוֹיְריּ אול יַגְלַר אוּסְמוּנַא רַאוֹל אוֹג בּוֹט אוּסְמוּנַה : ויתן דוֹרָדי אוֹל ג׳וּסְלֹנָי אַבוּצְלַרִי אוּכְמוּנֵא אַהַרן נִיגִּו דַאַבוּצְלַרִי אוּסְמוּנֵא אוֹגְלְנְלַרִינִיגֹּו דַסַּלְלַרִי אַלְרְנִי אוגונא סַלְלַמַק אַלְרִינַה יֹנָי נִיג : ויקח דַאַּדְרִי משֶׁה אַלַרְנִי אַבוּצְּלַרֵי אוּסְטוּנְדּן וּ רַטוּטְטַאַי אול מִוְבַּחַטַה אול עוּלָה אוּסְטוּנא ו מְלוּאִים רִיר אַלַר קוקוּסְיַנּה קבוּלְלִיקְנִיג ו אוֹבְלוּ בְרָבְּנְרִיֹר אוֹל יָנֶר נֵא : ויקח דַאַלְרִי משָׁה אוֹל מוֹשְׁנִי דַבַּלְלַרִי אַנִי בַלְּכַמַקּ אַלְרִינָהְ ינָי נִיג ו קוֹצְקַרִינְדָן אוֹל מְלוּאִים נִיג ו משֶׁה נָא בּוֹלְרִי פַּוְגַה ו נִצִיבִּי סִימִרלַרִי יִי משָה נֵא : ויקח דַאַלְרִי משָה יִנִּינְבוּן אוֹל סִילְמַבְנִיג וּ דַאוֹל קַנְדַן בִּי אוֹל מִוְבֵּחַ אוּסְמוּנֵא וֹ דַּםַצְמִי אַהְרן אוּסְמוּנֵא אוּפְּרַקְלַרי אוּסְמוּנֵא ו דַאוֹגְלַנְלַרִי אוּסְמוּנֵא דַאוּפְּרַקְלַרִי אוּסְמוּנֵא אוֹגְלַנְלַרִינִיג בִּירְנַסִינֵה וּ דַאַירוּכְסִי אָטְמִנ אַהְרן נִי אוּפְּרַקְלַרִינִי ו דַא אוֹגְלַנְינִי דַא אוּגְלַנְלַרִינִיג אוּפְּרַקְלַרִינָי בּירְגַּסִינַה : ויאמר דַאַיְמְמִי משֶה אַהַרן גַא דַאוֹגְלַנְלַרִינַה ו בִּישִירִיגִיז אוֹל אִשְנִי אִשִינִינְרָה אֹהֶל מוּעֵר נִיג ו רַאַנְרַה אַשַנִיז אַניּ-ו רַאוֹל אוְטְמַבְּנִי כִּי אוֹל מְלוּאִים כֵּלְלֵיסִינְרָא ו גִצִיכִּי סִימַרְלָרִים הַמַה ו אָהְרן דַאוֹּגְלַנְלַרִי אַשַּׁסִינְלַר אָני : והנותר דאול קלְנְנְנִי אוֹל אִמְרַן דַאוֹל אוֹמְמַבְּמֵן וֹ אוֹמְקָה בּוּיְדוּרוּנְיוֹ : ומפתח אשינינְרַן אֹהֶל מוֹעֵר נִיג צִיקְמֵגִיז יָרִי גוּנְלֵר הַמַם בּוּלְגַן גתּוּנַה דָגִין מִלְאִים -גוּנְלַרִי גִיז וֹ זִירַא יִרִי גוּנְלַר מוּלְרוּרִים אַלִּישְׁמִירִיר קוֹלוּגִיוְנִי : כאשר גִּצִיבִּי קילְרִי אוּשְׁבּוּ גוּנְרַה סִימַרְלַרִי יֹנִי קילְמָנַה בַּבְּרָה אִמְמַה אוּסְטוּגִיוְנֵה : ופתח רַאָשׁיגִינָרה אֹהָל מוּעֵר נִיג אומורוּגִיז גוּן רַגַּצַה יָרִי גוּנְלַר זֹ רַסַקְּלַגִּיז סַקְּלוּבִין יניג דאולְמַגיז ו זִירָא עַלֵּי סִימַרְלַנְרִים : ויעש דַקּילְרִי אַהַרן דַאוּגְלַנְלָרִי וּ ביומְלֵא אול כווְלַרְנִי בִּי כִימִרְלַרִי יִנִי קוֹלוּ בִּילוְ מֹשָׁה נִיג :

פרשת שמיני

מַבּייִינְגִיי גוּנְרָה ו צַקִירָדִי משֶׁה לָאַרָרוֹ וּלְבָנִיו וּלִזְקְנִי יִשְׂרָאל : רַאִרְיִּלְרִיוֹ מַ רָיִּרוּ בִּיוֹם הַשְׁמִינִי כְּרָא משֶׁה לָאַרְרוֹ וּלְבָנִיוֹ וּלְזִקְנִי יִשְׂרָאל : רַאִרְיִּעְלְרִיוֹ

9599900

ישַׁרָאַל נָיג : ויאמר דאַיְאָשִי אַדְרוּ גאַ אַלְנִין אוונה בּיווב בּלַפִין סינִירְנִיג תַּטְאת קה וַ דַקוֹצְקַר עוֹלְגַה בַּנְכַמְלרַנִיוֹ דַיִּיבוּקְלַשְּטִירְנִין אַלְהִינַה יֹנֵי נִיג : ואל באוּלְלִנְלַרִינִא יִשְּׂרָאֵל נִיג סוּוְּלַנִין בַּמָה ו אַלִינִיז אוּלַגִין אַצְּכִּיַלְרְנִיג חַמְאַת קהו דַבּיזוֹב דַקוֹי בִּירַר יַשַּׁרְלֵר בַּנְלְמַלֹר ניִ עוֹלְנֵה : ושור דַאוֹגוּז דַקוֹצְקַר שְּׁלְמִים לַרְצַה וּ בְּרָבַוֹ אִאְמַה אַלְּרִינַה זֹינִ נִיגו דַמִנְחָה קרִישִׁילְגַן יַג בִּילו וּ בִּי סִחִי בּוּגוּן בְּבוֹרִי יֹנָי נִיג אַשְבַרַה בּוּלוּנִיר סִיוְנֵה : ויקחו דַאַּרְרִילַר נְנֵי כִּי סִימַרְלַרִי משֶׁה אַלְרֶינֵא אֹהֶל מוער נִיג ו דָיוֹבוּקלַשְׁמִילֵר ג׳וּמְלֵא אול גַ׳מַעַט ו דַטוּרְרִילֵר אַלְדִינא יוֹי נִיג : ויאמר דַאַיְמְמִי משָה בּוּדִיר אול סוז כִּי סִימַרְלֵּדִי יּנְיָּ קיליגיו ו דַאַשְׁבַּרה בּוּלְסִין סִיוְגַה בְּכוֹרִי יֹנְי נִיג : ויאמר רַאַיְמְטִי משֶׁה אַבִּרן גַאַ ו יוּבוּקלַשְקון אוֹל מִוְבֶּחַ קה תַקּילְגִין חַטְאתִינְנִי רַעוּלְנְנִי ו דַבַּפְּרָה אִמְבִין גַנְדִיג אוצון דַאוֹל אוּלוּם אוצון ו דַקּילְגִין אוֹל אוּלוּסְנִיג קְרְבְּנִינִי ו דַבַּפְּרָה אַמְבִין אַלַר אוצון ו נִצִיבִּי כִימַרְלַרִי יוֹנָי : ויקרב דֵיוּבוּקְלַשְׁמִי אַהַרן אוֹל מִוֹבֶתַ קַה ו דַסוּיְרִי אוֹל חַמְאַת בִּיזוֹבוּנִי כִּי אַנִיג : ויקריבו דיוּבוּקְלַשְּׁמִירְרָיַלֵּר אוֹגְלַנְלָרֵי אַהַרוֹ נִיג אוֹל קַנְיִי אַנַר ו דַמַנְצְמִי בַּרְמֵגִינִי אוֹל קַנְגַהוֹדֵוֹרְרִי פּוּצְקּקֹלְרִי אַסְטוּנַה אוֹל- מִוֹבֶּחַ נִיג ו דַאוֹל קַנְנִי טוֹכְטִי ו בִּנְיַיִטִינַא אוֹל מִוְבַּחַ נִיג : ואת דאול יַנְנִי דַאוֹל בּוֹנְרַבְּלַרְנִי דַאוֹל אַרְטִיקְסִינִי אוֹל בַּנִיְרָן אוֹל חַטְאת דַן ו מושמים אול מוֹבּנוּמָא ו נֹצִיבּי סִמֹרְלַרִי יינָ מֹשֶׁר נַא : ואת באול אָטְנִי דַאוֹל מָרִינִי ו בּוּיְרוּרְדִי אוֹמְקַה אַבוּלְדֵן צִיבַרִי : וישׁחם בַּסוּיְרִי אוֹל עוּלְנִי ו דֵיָרִישְׁמִיְרְדִילֵר אוֹגְלַנְלָרֵי אַהֶרן נִיג אַגַר אוֹל קַנְנִי דַםַצְמִי אַנִּי אוֹל מִוְבָּח אוּסְמוּנֵא צוּפְצוּבְרָה : ואת דָאוֹל עוֹלְנִי וֵרִישְׁטִיְרְדִילֵר אַנֵר בּוּבוּמְלַרִינְנֵ׳ה וּ דַאוֹל בּאָנִי דַמוּמִמְמִי אוֹל מִוְבֵּח אוּסְמוּנַא : וירחץ ביוּבְדִי אוֹל קריְנְנִי בַאוֹל אַיִקְלרָנִי ו דָטוּטָטְאִי אוֹל עוֹלָה אוּסְטוּנַא אוֹל מִוֹבְּחַטֵה : ויקרב דיוּבוּקְלַשְּׁמִיְרְרִי קַרְבָּנִין אוֹל אוּלוּסְנִיג ו דַאַלְדִי אוֹל דַוְטָאת אוֹלְגִינִי כִּי אוּלוּסְנִיג ו דַסוֹיִדִי אַנִי דַרַקְאַמִירְרִי אוֹל עוּלְנִי נִיבִּי : זיקרב ביובוּקְלְשְׁמִירְרִי אוֹל עוּלְנִי בַקּילְרִי אַנִי שֶׁרַעַמְצָא : ויקרב דִיוּבוּקְלַשְׁמִירְדִי אוֹל מִנְחְנִי וֹ דַמוֹלְרוּרְרִי אַבוּצִינִיאַנְרַן ו במימימי אול מוֹבֶּח אוכְטוּנָא וּ עוֹלְסִינְהַן בַּשְּׁקַה אוֹל אַרְטַה נִיג : וישחש רסוידי

דַכוּידָי אוֹל אוֹנִייְנִי דַאוֹל קוֹצְקּרְנִי ואוֹל שְׁלְמִים בְּרִבְּנִינִ בִּי אוֹלְיִסְנִיג וּ דַוֹרִישְׁמִירְדִילֵר אוֹגְלְגְלָרִי אַהַרן נִיג אוֹל קּוְנִי אַנֵר ו דַסַצְאַי אוֹל מִוְכֵּח אוּסְטוּנַא צופּצוּכְרַא: ואת דאול יגלרני אול אוגויורן דאול קואקררן ו אול קויריקני דאול יפּבּנְגי דאול בונרפלרני ו דאר איקסיסין אול בנירניג (אייררילר) :וישימו בקוירילר אול יגלרנו אול טושלר אוכטונה ו דפושמשי אול יוגלרני אול מובחשה: ואת דאול טושלרני ראול אוג בוֹמְנִי ו סַלְלַדִי אַהַּרוֹ סַלְלָמֵק אוֹגוּנַה יְנֵיּ נְנִג ו נְצִיבּי סִוֹמְרְלֵדִי יְנֵיּ משה נא: וישא בבושרהי אהרן קוללריני אול אולוכקה באקנישלרי אלרני ראַבְבַל אַנְדִי קּילְמַקְמַן אוֹל חַמָּאת נִי רַאוֹל עוֹלְנִי רַאוֹל שְׁלְמִים לַרְנִי : ויכא בּגְלְרִי משָׁה דַאַבְּרוֹ אֹהֶל מוֹעֵר נֵא ו דִצִיקְמִילַר דַאַלְגִישְׁלְרִילֵר אוֹל אוּלוּסִנִי וּ ראַשְבַּרָה בּוֹלְרָי בְּבוֹרִי יָנִי נִיג ג׳וּטְלֵא אול אולוּסְקַה : ותצא דִצִּיקְטָּי אוֹט אַלְרִינְרַן יני ניגו דַיַקְמִי אול מִוְבֶּחַ אִיְּסְמוּנָא ו אוֹל עוֹלֶה נִי ראוֹל יַנְרַנְיִ ו דַגוֹּרְהִי ג׳וֹבְילֵא אוֹל אַיִּרוֹם דַקִּיצְקִירְרִינֵלר ו דַמוּשְׁמִילֵר יוּזְלַרִי אוּסְמוּנֵא : ויקחו דַאַלְרִילֵר אוֹגְלַנְלְרִי אַבֶּרוֹ נִיג נָרָכ דַאָבִירוּא הַר כִּישִׁי מַחְתָּסִינִי ו דַוַרְרִילֵר אַלַרְרָה אוֹם ו דָקוירְילֵר אוּסְטוּנֵא טוּאָסוּ ו דיובוּקלַשְאִירְדִילַר אַלְדִינַה ינוֹ נִיג יִט אוט ו כּי מימרלמרו אַלְרנא : ועגא בֹּגוּלמׁי אוּם אֹלְרוֹרוֹ ונְיָּ נוֹר בֹוּלמׁי אַלְרוֹי וּ ראולרילר אַלוינא יני ניג : ויאמר דאימטי משה אַהרן גא ו אוּלרירנא בי סווְלַדִי יוֹיָ דְמַה ו יובוְקלַרִימָא אַיְרוּכְסִי בּוֹלִירְמִין ו דַאַּלְרִינַא ג׳ומְלָא אוֹל אולוסנינ חורמאליירייון דסוכמו אחרן : ויקרא דצקירדי משה מישאל ני בא לבלו ני אונלנלבין עויאל אנצסי ניג אהרן ניג ו באימטי אלרגא ו יובוקלשיגיו בן דיריגוז קרדשבריייוני ו יאנינדן יוזלריניג אול קורש ניג ו אבולדן ציברי ויקרכו דייכיקלשְאִילַר דַבַּלְדִרְרִייַר אַלְינִי נוֹלְמַבְּלְרִי בִּילַן ו אַבוּלְדֵן אִינְרִי ו נֹצִיכֹּי כוּוֹלְבִי מֹחֵב : ויאמר בֹאִימֹאִי מִחֵּב אַבְרוֹ נַבְּוֹ בֹאַ-ְעוֹר נֹאַ בֹאִיפִּמֹּב אוֹרְלְנְלֵנִינָא ו בַּאַלְנִינִיוֹנִי אַאַכּינִיז ו באוֹפּנַקלְנִינִיוֹנִי יִרְמְמֵנִיז ו באוּלְמֵנִיז ו דג׳וּמְלָא אוֹל גַ׳מַעט אוּסְטוּנַה עַצוּבְלַנְטַסִין וַ דַקּרָבְשְׁלַרִיגִיז ג׳וּמְלַא נַ׳מַעסִי יִשְׂרָאָל ניג צילְמֵנִיז אוּלְמֵיא אוּלְכִיז ו זיִרא סִילְמֵבּ ינִי יְנָיּ נִיג אוּסְמִיגיוְדֵּה ו דַּלִּילְרִילֵר גימל כווינצא

359

Batha

סוזינצא משה ניג : ויהבר דַפוּוְלַדִי יני אַהרן גַאַ דָמָה : יין ינִי שַׁרַפּ נָא אַכְבּי שַׁרַפּ אִיצְתַּגִין וּ בָּא מוֹ נֵא רַאוֹנְלַנְלַרִיג בִּיְרְגַנָא ז גּוְרְרִינִינִיוֹבוֹ פּוּרוּן אֹהֶל מועד גא דאוּלְמִנִיז ו אומוּרְלִיכּ בַּפִים דַוּוּרְלַרִיגִיזְגַה : וֹלהבריל דַעַנֵיֹי אַייִרְפַּגַה גָּלִיפּ אַרַפִּינַא אול קורֶשׁ נִיג רַאַרַפִינַה אול יָגִילְנִיג ו דַאַרַפִינַא אול מונְדַרְנִיג דַאָרַסִינַה אול אַרוּבְנִיג : ולהורות דַאוֹגְרַשְׁמַנַא גָּלִיפּ אוֹגְרַנְּלַרִינַא יִשְׂרָאֵל נִיג ו ג׳וּמְלֵא אוֹל רַכִּימְלַרְנִי וֹ כִּי כוּזְלַדִי ינִי אַלְרָגַה קוֹלו בִּילַן משָה ניג : וירבר דַכוּוְלַדִי משֶה אַהַרן גָא ו דַאֶּלְעוֹר גַא דַאִיתְמֶר אוֹגְלַנְלַרִינַא שול קַלְנַנְלַרְנַה ו אַלִיגִיז אוֹל בַקלנון מִנְחָה ני אוֹמֵלוּ כְּוְבְּנְלֵרִינְרֵן יְיֹי נִיגו רַאַשְׁנִיז אַנִי מַצְּלֵר יַנִינַא אוֹל מִוְבֵּחַ נִיג וֹ זִירַא קֹדָשִׁי קוֹרָשׁ לַרְנִיגְרִיר אוֹל: ואכלתם דַאַשׁינִיז אַנִי אַיְרוּכְכִי יִרְבָה ו זִירַא חַקְּקִיג דַחַקְּקִי אוֹגְרַנְיגְנִיגְרִיר אוֹל ו אוֹמְלוּ קָרְבְּנְלַרִינְרוּ יְנִיּ נִיג וּ בִּי עַלִי כִימְרְלַנְדִים : ואת דאול כַלְלַמַק טושִינִי דַא אול אַיִירְמַק בּוּטוּנִי אַשַּנִיז מָמִיז וַרְדָה ו מָן דַאוֹגְלַנְלַרִיג דַקִיזְלַרִיג בְּירְנַגַה ו בּי חַקּקיג דַחַקְקּי אוֹגְלַנְיַרִינְנִיג וָהֵילְדִילֵר וּ שְּלְמִים מֶרְבַּנְלַרִינְדֵן אוֹגְלַנְלַרִינִיג יִשְּׂרָאֵל ניג : שוק אול אַיִירְמַק בּוּטוני דַאוֹל סַלְלַמַק טוֹשִׁינִי אוֹטְלוּ קַרְבְּנְלַרִי בִּילֵן אוֹל יַגְלַרְגִיג גִּטִירְכִינְלַר ו כַּלְלַמַגה סַרְלַמַק אַלְרִינָה יָנִי נִיג ו דַבּוֹלְסִין סַגַה דאוְגַלְנְלַרִינִא בּירְנֵגֵא אוּמוּרְלִיכּ רָסִימְנֵא ו נִצִיכִּי סִימִרְלַרִי יני : ואת דא אול חַמָאת אוּלַגִינִי אַרַמַא אַרַארִי משָׁה ו דַמוּנָא בּוּיְדוּרוּלְרִי ו דַעצוּבְלַוְרִי אָלְעִזְר אוּכְטוּנַא דאיתְמֶר אוּסְמוּנַא אוֹל קּלְנָן אוֹגְלַנְלַרִי אַהָרן נִיג דִּמַה : מרוע נוצון אַשַּמֵרִיגִיז אול תַפְאת נִי אוֹל פוּרָשׁ אוּרְנִינְרַהַ ו כִּי קוֹרָשִׁי קוֹרְשׁ לַרְנְיגִרִיר אוֹ ו דַאַנִי וְרֶרִי פּיוָנָה ו פּוֹטַרְמַגַה גוּנַחִין אול גַ׳מַעַמנִיג ו בְּפָּרַת אִמְטַנֵּא אַלַר אוִצין אַלְרִינָא פּנָי נִיג : הן מונה קני גִירְגִיזִילְמִרִי אול קוֹרֶשׁ גַא אִיצְבֵּרִי ו אַשְּׁמָא אֲשׁנִיוֹ אַני אַירוּכְסִי וַרְדֵא וּ נִצִיבִּי סִימַרְלָרִים: וידבר דַכוֹוְלַדִי אַהַרן מֹשֶׁה גַא וּ מונה בוגון יובוקלשְטִירְרִילֵר חַטְאת לרָניִ רָא עוֹלַלְרִינִי אַלְרִינָא יוֹנִי נִיג ו רַאוּצְרַרִילַר מַנָה בּוֹלֵר נִיבִּי רָאשׁכֶם אֵירִים נַחְטָאת אַטִינִי בּוֹנוֹן וַיְבְשִׁי בּוֹלִירְמִי י אָדִי עַנַיִימְלַרִינִא יָנָי נִיג : וִישמע דַאִשִימְמִי מֹשֶׁה ו דַיַּכְשִׁי בּוְּלְרִי גוּוְלַרִינַא : וולכנ במונבני ינה מאו לא באבון לא במא אקרדע :

ישמיני

דברו בוולגיו אוולרלרינא ישראלין דמא בודור או בונינט בי ייסיווג'ומלא או תוורדן כי או יר אוורינא : כל הר צטל מירנקליו ואייריגייני איירסגין מִירְנַקְנִין גַּוְוְשֵׁמֵבּ גָּמִירָנִנִי רוּוַרְדַא אונוּ אַשַאנִיוּ : אך אַמְמָא גּוּוְשִׁמְבּ גָּטִירַנְלֶרְרֵן וַמִירַנִקּ צַמַלְלִילַרְרַן בּוּלַרִי וָמָיָסִיזוֹאוֹ דַיִוֹנִי פַּרְזַם כִּי גִּוֹוְשֶׁר צַמְמַא תַמַם אַיְירִי מִירְנַקְלִי הַגִּיל הִירִן מוּרְבֵר הִיר אוֹ כִּיוַא: ואת וָאוֹ אַרַא טַװּשַׁן (בּירְפִי) נִי פַרְזַם כִּי אוֹ גִוּוְשֵׁר אַמְמַא תַּמַם מִירְנַק אַיִירְמַזוֹמִירְבַרִרִיר אוֹ סִוָּא: ואו פוושני פַרְנַם כִּי גווְשֶׁר אַמְמֵא תַּמִם טִירְנַק אַיִירְמַדִי וְמוּרְדֵרְדִיר או סיווא : ואת ואו הינוירי (חוירני) פַרְוֹם כִּי צַמֵּל מִירְנַּלְיֹדֵיר וַחַמֵּם אַיִּרִיר אַיירְמַסִין טִירְנַקנו אַמְמֵא או גווְשִׁמָא גווִשׁמָז ו מוּרְדַרְדִיר או סִיזַא : מבשרם אַאְלַרִינְרַן יִמֶגִיז הַלְשְׁלֵרִינִא בַנְמִנִיז הְרָם דִיר אוֹלַר סִיוֹא : את בונו וְיָסֵיזּ ג׳ומְלַא דַן בּי סוב דַא הַר שִי בִּי אונא קנט ופול סוְבַדְא דְגִיזְלֵרֶדְא דַאוֹנְלֶרְרָא אופרי ייביז : וכל נחר שיי כי יוק אונא קנט ופול הגיולרהא אוונלרהא ג׳וְמְלֵא קוֹוְלַמַסִינְרַן סוּכְנוּן וְג׳וּמְלֵא אוֹ דִירִי נֵ׳ן דֵן כִּי סוּכָרַא ו מִיקְרוּח דִיר אולר בווא : ושקץ ומקרוח אולְסוּנְכַר בִיוֹא וּ אִמְלַהַנְבון וִמִיבִיז וַבְּשְׁלֵרְינִי אִיְקְרַח אִירַסִיז : כל הַר שִׁייִ בִּי יוֹק אוֹנֵא קַנַט וָפוּל סוּבְרַרְדֵא מִיקְרוּח דִירּ או סיוא : ואת ובולרי איקרח אידיגיו או קושטן אשלְסַסינְלַר סִיקרוּח דִיר אולר ו או קרְשַׁלְנִי (קָרָא קּישְׁנִי) וָאִירוֹנִינִי וִדְנִיז קּרְשַׁלִינִי : ואת וַאַק בַּבְּנִי וֹא צּוֹלְלְנִי נִינְסִינְנִיא : את הַר קוֹזְנִינִי נִינְסִינְצַא : ואת וֹדְוֹא קוֹשׁוּנִי וֹבַיֵי קושִינִי וַקּיקוּ קושִׁינִי וַדוּגַנִי גִ'נְסִינִצֵא: ואת בואינִי קושׁינִי וַקרא בּסַקני וְטוּיִנִי : ואת וַקוֹגוּנִי וַקשִׁיקְצִי קוּשִׁינִי וַרַחַמִינִי : ואת וַלְגַּלְנְנִי וַבַּלִּיּק צִינִי גְיָנִסִינְצֵא וָהִוּד הוּרִינִי וִשֶּׁפֵּרָנִי : כל וָהָר אוּצַן קוּשׁוּ דוֹרִם אַיַיִק אוֹוָרִינַא נָזָן וּ מִיקְרוּת דִיר אוֹ סִיוָא : אך סַלְט בּונוּ יִיִסִיז ג'וּמְלַא דוֹגוּיְרדוּגוּנְדן אוֹ קוּשְׁנִיג או גון דורט אַייַק אוורינא ו כּי אונא דיוֹלַר יוֹהַרִיבוּן אַייַקלרִינַא ו בּאַירָטַא אולר אילן או יר אוורינא : את בולרני אולר בן אשניו או אבירטבני בּינֹפֹנְנְגְּא ו נֵגְינְרְגִינִינִּי צִינְפִינִאַא ו נֹשׁרְצוּלְנִי נִינְכִּינְגָּא וֹשׁרְבִּנִי צִינְפִינְגַא וגיומלא דוגורדוגו או קושון ניג כי אונא דורט אייקלרו מיקרות דיר או כיוא

שמיני

פיוא : ולאלה ובולר גא טִיפּ מונְדֵר אולוּרְסִיוֹ ו דֵּר דִנְגון לֵשְׁלֵרִינְאַ מוּנְדֵר אולסין או עַבְשַׁמַא דָגִין : ובל נאו מַשְׁיֵין) קּלְרִירֵן לַשְּלַרִינְדַן ייִקּסִין אוּרְבַּלַרִינִי וֹכוּנְרֵר אוּלְסוּן אוּל עַבְשַׁמַא דָגִין : לכל דֵּר דַיִּיוַן בִּי אוֹ צַמֵּל פִירְנַקְּלִי דִיר וַתַּמַם אַיִּרְמַק אַיִּרְמִיוֹר וָגוּוְשֵׁמֵבּ גִּמִירְמִיִיר מוּרְבַרְלַר דִיר אוֹלֵר סִיזֵא ו הַר דְגָגן לַשְּׁלֵרִינֵא מוּנְדַר אוּלְסִין או, אַבְשַׁמַא רָגין : וכל ג'וְמְלֵא גַזן מַבּּנְּלַרִי אוורינאוגיומלא גינוורדא אונון דורם אייק אוורינאודורמלר או לר פיואו גיומלא או טִיְיגוּ לִשְׁלְרִינֵא וֹמוּנְרַר אוֹלְכוּן אוֹ אַכְשַׁמֵא רָגִין : והגשא נאו בקלריבן הַשְּׁלֵרינִי יִיקּסִין הוַבַּלַרינִי דַמוּנְבַר אוּלְכוּן אוֹ אַכְשַׁמַא קַבַר ו קַרְמְּלֵר דִיר אוֹלַר סָיוָא: ווה וַבּוּרוּר סִיוָא אוֹ חָרָם דוּנְגַן דֵא אוֹ קוֹוְלַגן אוֹ וֵר אווִרינַא אוֹ גַּלִינְצִיכּ וֵאוֹ סִיצֵן וֵאוֹ קַפְּלִי בַּגַא גִינְסִינְגֵ׳א : והאנקה וֵסִיבְרִי סִיצֵן וֹגוּנֵשׁ בָּלִרָסִי וֵיִלְרִיז בָּלֵרָסִי וַבָּרְשָׁן בָּלָסִי וֵגוּז מוֹבַּא : אלה בּוּלַר או חַרַמְלֵר סיוא ג׳וּסְלֵא אוֹ דוּנְגַן בא ו ג׳וּסְלַא אוֹ דְנָגָן אוֹלַרָא אוּלְרִיכְּלְרִינְרֵן כוּרָא חַרַם אוֹלְסוּן אוֹ אַכְשְׁמַא הָגִין : וכל והר שִׁייְ כֵּי דוּשֶׁרְמָא אוּזִרֵינאַ אוּלַרְרוּן אוּלְרוּבְּלַרִינְרֵן סוֹרֵא חַרֵם אוּלְסוּן וּ ג׳וּמְרֵא אַגַץ עָבְרַנְּלִיקְרַן (סַגִּימְמַן) יַכוּר אָכְפַפּ יַא דָרִי יַא צוּוַל הַר עַבְרַנְלִיק כִּי יַפּוּלוּר אִישׁ אוֹלַר אִילֵן ו כוּבְלַרֵא גּיְרְגִיזִילְסִין הַחַרַס אוּלְסוּן אוֹ אַכְשַׁמַא קַרַר וִפַּק אוֹלוּר : וכל וְנִיוּסְלא סַקְסִי קפי בי דוּשֶׁרְכֵּא בִּירִיסִי אוֹלַרְדֵן אִיצְרִיסִינֵא ו הַר גַא בּי אִיצִרִיסִינָּרא חַרַם אולסון דאו קפני קיריניז: מכל ג'ומְלַא או וַימֵבּ דַן כִּי אַשַׁלִיר כּי גַלִירְסָא אוּזָרִינֵא סוּב בַרָם אולור ו וג'ומְלא אִיצְבִי כִּי אִיצִילִיר הַר קפ אִיצִינְרַא חַרַם אולור : וכל והר שיי בי דושרמא אוזריגא לשלרינדן דעם אולור פורון נאוצק בוזוילְסוּן חַבְמְלֵר דִיר אוֹלֵר ו נַחַבִמְלַר אוֹלְסוּגְלֵר סִיוַא : וכי מֶב צָשְׁמֵא וִכַּרְנִיץ דְּוֹוְשִׁירִילְבֵּיִכִּי כּוּבְנוֹן אוֹלוּר חֶלֵל ו וַהְנָגָן לֶשְׁלֶרִינָא דַרַם אולוּר : וכי דַאֵנַר - דוּשֶׁרְכָּא לֶשְׁלֶרִינְרַן הַר דוּרְלוּ אָבִילְמִיש אַבִּין אוּזָרֵינֵא כִּי אָבִּילְכָא ו הָלַל רִיר או : וכי וֹאָנֵר וָרִילְפַא פוב תורום אוֹוָרִינַא וֵדוֹשִׁרְפַא לַשְּׁלְרִינְדַן אוּוְרֵינַא ו חָרָם דִיר אוֹ סִיוֹא : וכּי דאנר אולוּרְסא אוֹ טוּווּרְדְן בִּי חַלֵּל דִיר אוֹ סִיוָא ימוא ו או דינו לשיוא חרם אולסון או אַכְשׁמִא קרר : והאכל ואו יוו לשינהן

שמיני

יִשְׁינְרֵן יָקְסִין אוּרְבּלֵרִינִי דַמוּנְדֵר אוּלְסוּן אוֹ אַכְשַׁמָא קַרַרוֹוָאוֹ קּלְּרִינִי דְמוּנְדֵר אוּלְסוּן אוֹ אַבְשַׁמָא קַרָרוֹוִאוֹ הַלְּרִינִי דַמוּנְדַר אוֹלְכוּן אוֹ אַבְשַׁמַא קַרָר: וכל וְגִיּוִישְׁא אוֹ דִוּנְגן שִׁייְ בִּי רוּנִייר אוֹ זִרְ אוּזְרִינַא וֹ מִיְקרוּח שֵׁיִי דִיר זִילְמָסִין : כל הַר גַּוֹן פוּרְסֵ אַיִּק אוֹזִיינַא וֹ גִינְקְנִא אוֹ דוּנְגַן בַּן בִּי דוֹנִיִיר אוֹ הַיְרָבְּ וֹמִיסִיז אוֹלְרִי זִירַא מִיְקרוּח שֵיִי דִיר אוֹלְר: אוֹל אִיךְ בִּיּיְרָ גִיזִּי בִּיִרִי אַלְּלִינִיז אוֹלְר: זִירָא מִיְּקרוּח שִייִ דִיר אוֹלְר: אוֹלְבְּ גִּיזִּי בִּיִר אַקּלִינִיז אוֹלְר: זִירָא קּוּדוּשְׁאִים בּן ו זִחְרְס אִיְּקְיִים אוֹלְרִי וְדָּא קְּבְּיִרְיִן אוֹ זִירָא קְבִּירְיִין אוֹ זִירָא קּוּדוּשְׁאִים בּן ו זִחְרְס אִיְּקְיִיִים זּיִּלְרִינִין אוֹ זִירָא קּוְדוּשְׁאִים בּן ו זִחְרָס אִיְקְמָיִסִיז גַּיְלְרָינִיזִי גִּיִּרְאָּלְר זִירָא קוּדוּשְׁאִים בּן ו זִחְרָס אִיְקְמִים זְּיִילְ אוֹ קִימִילְרַיִין אוֹ זִירָא קּוּדוּשְׁאִים בּן ו זִחְרְסִין וֹנִין מִיּסִירִין וּ אוֹלְמָא סִינִא מַנְרִינְאוֹ בִירְנִיא קּוּדוּשְׁאִים בּן זּ זֹחְרְסִין וְיִרְא קּוּדוּשְׁאִים בּן זּי זְיִרָּא קּנִילְרְיִין מִיסִירִין וּ אוּלְמָא סִינִא מַנְינִיא גַּיִבְלָּא זוֹ נְיִינְא אַרְרִּשְׁאִים בּן זּיִינְמִילְרַיִּין כּיּבְלַרְרָאוֹ וֹ נִינִילְנִין מִּיּלְרְיִין מוֹן בּיִרְכְּנִיּא אוֹ בּוֹלִין וּ וּאִרְמָנִא אוֹ בְּלִיין גִּין מִוֹן וְנִייִבְּין מִיּיִינְן וּ אוֹ מִיּמִילְרִייִן מִּיּיִנְיִין בְּיִּיְיִין מִיּיִינִין וּ אוֹבְיִינִיא אוֹרְנִין אוֹן וּבְּיִייִין מִיּלְרָייִן זְּעִרְיִיִּיִן אוֹ בּיּיִיִּיִן וּיִּין מִיּיִיִין וּ בּיּבְיִינִין אוֹיִנְיִין אוֹין וְיִיּיִיּיִין מִּיּיְלְיִייִּיִין אוֹיִין אוֹיִין מִיּיִין וְשִּיּבְּייִין זְּיִין בְּיוֹי בְּיוֹיִין זְּיִין מִּיְיִיִין אִין מִיּיְיִין זְיִין מִיּיְיִין וְּיִין בְּיִיּיִין זְּיִין בְּיִּיִין בְּיוּיִי בִּיוּ בְּיוּיִין זְּיִין בְּיִיוֹיִין בְּיוּיִין בְּיוֹי בְּיוּיִיִּייִין מִּיּנְיוֹין זְּיִין בְּיוֹין מִּיוֹיִין בְּיוֹייִין מִּיוֹין מִין בְּיִייִין מִּיוּיִין זְּיִין מִּיּיִיןּיִין בְּיוֹי מִיוּיִין זְּיִין מִיּיִיּיִין זִּיִין מִּיּיִין זְיִין מ

פרשת תזריע

בן יַדְרֵבְּרִ יִיָּ אֶל משָׁה לָאמר: דַסוּזְלַדִי יִיּיָ משָׁה גַא הַמַה: דבּר סוּזְלַנִין אוּלְנִלְרִינָה ישִׁרְאֵל נִיג דֵמַה ו כַּאָין כִּי אוּרְלוּק אִטְסַה לִיגִינִיג מוּנְדַר בּוּלְסִין יְדִי גוּנְלַר ו גוּנְלַרִי גִיפִּי אַרְסִיזְלִיגִינָג נִדְה לִיגִינג מוּנְדַר בּוּלְסִין: וביום דַאוֹל מִכְּיִזִינְגִי גוּנְדַה ו סוּנִאַלְנִסִין אַנִּילְר ו אוֹמוּרְסִין קַנְּלַרִי בִּילְן מִמִיזּ לִיגְינִיג ו הִיץ קוֹדָשׁ גַה מִיְימַסִין ו דַאוֹל מִקְרָשׁ קּה גִּלְמַסִין ו תַּמַם בּוּלְנִנִינָה דִּגִין נִיְיג גוּנְלַרִי מִמִיז לִינִינִג ואם דַאנִנְינָה דִּגִין נִידְּג וּיִבְּר לִיגִינִיג ואם דַאָּנֵר מִישִׁינִין הַנְּלָר אוֹמוּרְסִין הַנְּלָרִי אוֹצוּן ו בַאַלְמִיִישׁ אַלְמִי גוּנְּלָר אוֹמוּרְסִין הַנְּלָרִי אוֹצוּן ו בַאַלְמִיִישׁ אַלְמִי גוּנְלָרִי מִמִיזליִגייִנִּג אוֹגוּלְנִה יִא קּוֹינְנָה ו גִּשְּירִסִין בּמֹלְנִינָה וּ בִּמִילִּין אַנְיִנְינָה גוּנְלָרִי מִיִּילִינִיג אוֹצוּן ו דָאַלְמְמִישׁ אַלְמִי גוּנְּלָּר אוֹמוּרְסִין הַנְּלָרִי אוֹצוּן בַּאַלְמְּמִישׁ אַלְמִי גוּנְלָרִי מִּמִיזליִיגינִיג אוֹנִילְה יִא קּוּינְה ו גִּשְּירִיכִּוּ בּוּלְנִינָה גוּנְלְרִי מִּמִין בּוֹלְנִינָה גוּנְלְרִי מִים בּוּלְנִינָה וּ בְּבּיל בִיּי בְּבְּיִינִיה וּ בְּיִינִיה וּ בְּיִינִיה וּ בְּבִּיל בִיּינִיה וּ בְּיִינִיה וּ וּנְבְּילִיה וּ בְּלִין וּ בַּבְּלְמִיתִישׁ אַלְמִי גוּנְלְרִי מִימִין לְּנִינְרָה בּוּלְנִינָה וּ בְּעִינִיה אִּינִינְיה וּ בְּיִינִיה אוּנִינִיה אוּנִין וּ בְּבּלְטִין וּ בִּבּלְימִין גוּינְנִינִי בְּיִינְנְינִי בְּיִם בּוּלְנִינָ ה וְנִינִינָה וּיִּין בְּיִינִינְייִי בְּיִינִינְייִינִי בְּיִי מִינִינִי בְּיִים בּוּלְנִייְ בָּה בּוּלְנִייִר בּייִינְנִיי בּי מִייּילְינִינִי בּי בְּיִיים בּיּילִיי בְּיִים בּיּים וּי בּוּלְנִייִי בְּיוּל בִייִינְייִי בְּיִילְייִי בְּיִייִינְייִי בְּיִינִייְייִי בְּיִייִי בְּיִיים בּיּילְייִי בְּיִייִים בּיוֹינְיינָי בְּייִיים בְּיִיים בּיוּים בּילְנִינִים בּילְבְייִים בּייִיים בְּייִינְייִים בּייִייְנִייִים בּייִייִים בּייִייִייִים בּייִייים בּיייים בְּינִייים בּייִייִים בּייִיים בּייִיים בּייִייִים בְּיִייִים בְּיייִים בּייִייִּייִים בּיּנִינְים בּיּיִיים בּייִיייים בְּיִייִייִי

חזריע

מוער ניג אול פּהֵן גַה : והקריבו דיובוּקלשְׁשִירְסִין אַנִי אַלְרִינֵא יְיָּי נִיג דַפַּפְּרָה אַמְסִין אוּסְשׁוּנֵא וּ דַארוב פּוּלְסִין צוּקרַגיִנְדֵן קַנְלְרִינִיג וּ פּוּרִיר תּוֹרְסִי אוֹל דוֹגוּרְגנְנִיג וּ אַרְפַּנְי יַא מִישִׁינִי : ואם דאגר יִשְׁמַסֵא קּיבַמִי קוּיְגַה וּ דַאַלְסִין אָבִי קוּמְרוֹלֵר יֵא אָבִי פַּלְלָרִין פּוֹגוּרְצִינְנִיג פִּיְרְנִי עוֹלְנָא דַבִּירְנִי תַשְׁאת קַה וּ דַבּפְּרָה אָמִין אוּסְשוּנֵא אוֹל פּהֵן דַמִמִין אוֹלוּר:

וירבר בסוולרי יני משה גא דאהרו גה המה : ארם אָדָם בּי בוּלְסַא טַרִיסִינָרָה אָטִינִיג שִׁישִיק יַא אוינו יַא אַק לֵבָא ו דַבּוּלְכַא מָרִיסִיְנַדה אָטִינִיג כַסְשַלִינִי ג׳וּוֹם דִרְדְלִיבְּנִיג ו דַגְּשִׁירִילְסִין אַהְרן אוֹל פהן גַהוֹיֵא בִּיְרְסִינָה אוֹגְלַנְיַרִינְרַן אוֹל כֹהוֹ לַרְצָה : וראה דַגוּרְסִין אוֹל כֹהוֹ אול כַסְטַלִיקְנִי מִרִיסִינְרַה אוֹל אָטְנִיג ו דַסֵץ בַסְטַלִיקְטַה דְגִישִירִילְרִי אִיסֵא אַק ו דַגורוּמִי אוֹל כַסְשַלִּיִקְנִיג מַרַן אִיסָא מֵרִיסִינָרן אָשִינִיג ו כַסְשַלִיגִי ג'וּזַם הַרְדְלִיבְּנִיג אול ו דגורסא אָנִי אול פּהון דַמונְדַר אַמִסִין אַנִי : ואם דאָגַר אַק הָכָּא אָסַה אוֹל מַרִיסִינְרַה אָטִינִיג וּ דַמָּרַן דוֹגוֹל אָסַה גורוֹמיִ אוֹל מֵרִידַן וּ דַסַצִי דָגִישִׁירִילְמַדִי אָמַא אָק וּ דַבּּבְּלַמִין אוֹל כּהָן אוֹל כַמְמַלִיקְגִי יְדִי גוּנְלַר : וראהו דַגוּרְסִין אָנִי אוֹל כּהָן אוֹל יְרִינְגִי׳ גוּנְּדָה ו דַמונָא אוֹל כַסְטַלִיק טוּרְדִי הִיגְּבִינְראו יילָמַדִי אוֹל בַסְמַלִיק מָרִידַהוּ דַבּבְּלַסִין אָנִי אוֹל פֹהֵן יְדִי גוּנְכֵר אָבִינְגִיי : וראה בגורסין אול כהן אני אול ידיוניי גונדה אבינגיי ברס ו במונה סונובכן אול כסשליק ו דיילמדי אול כסשליק מרידה ו ממיו אִשְׁסִין אַנִי אול כּהָן אוינוֹ דיר אוֹל ו דִיוּבְּסִין אוּבְּרַקְלַרִין דַמָּמִיז בּוֹלְסִין : ואם דַאֵּגַר יִיִּקְמַה יוֹלְטַה אוֹל אויוו מֵרִירַה ו גורוּנְגַנִינְדֵן סוֹנְרַה אוֹל בּהֵן נָה מִמִיז בּוּלְגַנִינְדֵן כוֹנְרָה וֹ דַגורוּנְסִין אָבִינְגִיי בֶּרַטְ אוֹל בֹהֵן גַה : וראה דַגוּרְסַא אוֹל בּהֵן בַמוּנֹא נִילְבִי אוּל אויוּן מֹרִיבַה ו בַמוּנְבַר אֹאַסִין אַנִּי אוּל פְנֵּוֹ גְיוֹס בֹּרְבְּלִיכְּבִיר אול : נגע בַּקְּמֵלִינִי געוַם דֵּרְדְּלִיבּ בִּי בּוֹלְמַה אָדָם דֶה וֹ דַגְמֵירִילְסִין אול בּהֵוֹ גַה : וראה דַנוּרְסִין אול בּהון דַמוּנַה אַק שִׁישִׁיבּ מִרִידַה ו דַאוֹל דְנִישְׁמִי אַקּ סין ו דַכַּגְלִינִי צִיִּי אָטְנִיג שִׁישִׁיק דַה: צרעת אָסְבִּיְרְגַן ג׳נַוֹם דִרְרְלִיבְּּרִיר אול מֹנִיםׁינְבַה אָמִינִיג ו בּמוּנְבר אֹמִסון אני אול פְנֵוּן ו בּבּלְמַסוּן אני זובא מוּנֹבר **ل**، ل

תזרע

דיר אול: ואם ראגר נילפא נילפא אול ג'נום דררליב פרידה ו דיבכא אול גיוום הרדלים גיומלא מריכין אול בסמליקניג ו בשינהן דאיקלרינה הגין ו ג'ימלא גורובינא גוולריניג צול פהן ניג: וראה דגורסין אול פהן דמינה יפסי אול ג'ווַהַ דְּרְרִיבּ ג'וּמְלֵא אָמִינִי וּ מִמִיז אָמְכִין אוֹל כַכְּמַלִּיְקְנִי וּ ג'וּמְלַאִסִי הגישרי אַק ו הַמֵּי דִיר אַוֹל : וביום דגורוּנְגַן גוּנְדָא אַנְדַה צִיִּי אָט ומוּנְרַר פוּלְבִיוּ : וראה דָגוֹרְסִין אוֹל בֹהָן אוֹל צְיִי אָמְנִי דַמוּנְבר אַמְסִין אַנִי וּאוֹל ציי אם מונדנדיר אולוג'ווס דרדליפדיר אול ו או נא פי קימסה אול ציי אם דַרְגִישִירִילְרִי אִיפַא אַקקהו דָגִלְסִין אוֹל כַּהָן גַה : וראה דגורסין אַנִי אוֹל פהן ו דמונה דגישירילרי אול בספליק אַק בקה ו דַמְמִיז אִמְסִין אול כּהו אוֹל בַּכְטַלִּיקְנִי טִמִיוּ דִיר אול : וכשר דאמ פִּי בּוּלְכַה אַנֵרה מֵרִיכְינִדה צִיבְן ראוגלסה : והיה רבולסה ורינדה אול ציבנניג אק שישים ו יא אַק לבא ו קירסיוג׳ א דגורונסין אול כהן גא : וראה דגורסין אול כהן דמונא גורומי אַלְצַק אוֹל מִרִידַן ו דַכַצִּי דִגִּישְׁרִי אַק ו דַמוּנְדַר אִמְסִין אַנִּי אוֹל פּהָן בַּסְטַלִיגִי ג׳וּזָם הַרְּדְרִיפּ גִיג דִיר אוֹל ו צִיבּנרַה יִילִרי : ואם רַאָּגַר גוֹרְסַה אַנִּי אול פהן ו דַכונה יוקטיר אַנְדַה אַק סִץ וּ דָאַלְצַק דוגוּלְרִיר אוֹל פַרְיַדָּוֹ דָאוֹל כונוּבְּבַּן ו דַבּבְּרַסִין אַנִי אול כהו יִדִי גוּנְלַר : ואם דאַנר יִיּלְמַה יִיּלְמַה מַרִירַה דַמוּנְדַר אַמְסִין אוֹל כּהֵן אַנִי בַסְמַלִּיק דִיר אוֹל : ואם דַאָּבַר יִרִינְדָא - פוּרְכַה אוֹל אַק לַבָּא יִילְמַכַה ו כּויוּגִי אוֹל צִיבַנְנִיגְרִיד אוֹל וַדְמִמִיז אַמְסִין אָנִי אול כהן : או יַא אַט בּי בּוֹלְסַה מֵרִיסִינְרא כּוְיַגְנִי אִטְנִיג וּ דַבּוּלְסַה צִיְיִי סְנִי אוֹל בּוּיְגַנְנִיג בִּיוְצֵא קִירְמִיוִי׳ַא לִבּאויַא אַק : וראה דגורְמִין אַנִי אוֹל פהן ו במונא הגישירילהי אק סיי ב בה ו בגורומי ברן אול מרידן ו גיוום בַרְרָּרִיְבִּיבִרר אוֹל ו בּוּיוּבְּרַא יֵיִירְרִי ו דַמוּנְּדַר אִמְסִין אַנִּי אוֹל בּהֵן ו כַמְּמַלִינִי ג׳וּיַם הַרְרְלִיכּ נִיג דִיר אוֹל : ואם דַאַנֵר גוֹרְסָה אַנִי אוֹל כּהֵן ו דַמוּנַה יוַקְמִיר הַבָּא רא אַק סִץ ו רַאַלְצַק דוֹנוּל דִיר אוֹל מִרִיבוֹ ו באוֹל סונוּבְּבּן ו בַבּּבְּלְסִין אַנִי אוֹל פהן ידי גונלב : וראהו בגורסין אני אול פהן אול ודינגיי גונדה ו אנר יַיִּלְמַה יִיּלְמָא מֵרִירָא ו רַמוּנְרַר אַמְבִין אוֹל כּהָן אַנִי ו כַּסְמַלִּינִי גְ׳וּוַם הַרְרְלִיבְּניְנְרִירִּ אורה

הודיע

אול : ואם דאנר ירינרא פורסא אול לבא ז יילמסא מרירה ראול כונובפן ו שישיגי אול פויובניג דיר אול ו דַמִּמִיז אַמְסִין אַנִי אוֹל פהן ו זירא פּוּיוּגִי אוֹל פּוּיוּבְנִיגְדִיר אוֹל : ואיש דַבְּישִׁי יַא בַּמִּין בִּי בּוֹלְכֵה אַנְדָה כַּסְשַלִיק ו בּשְׁמֵא נֵא סַבּלְרַה : וראה דַגוֹרְסִין אוֹל כּהֵן אוֹל כַסְמַלִּיקְנִי ו דַמוּנַה גוֹרוּמִי מָרַן אוֹל מַרִירַן רַאַנְרָה סַרִי סִין אִינְנֵיא ו דַמוּנְרֵר אֵמְסִין אַנִּי אוֹל כּהַן נָתְק. דיר אול ג׳ווַם דִרְדִירְלִיגִי אוֹל בַשְׁנִיג יַא אוֹל בַקְלַנִינְדִיר אוֹל : וכי דַאָגר גוֹרְסַא אוֹל כּהָן בַסְשַלִיגִין אוֹל נָתָק נִיג וַ דַמוּנַה דוּגוּלְדִיר גורוּמִי מָרַן אוֹל מריבו דַקַרַה בין יוּקְמִיר אַנְדַה ו דַבָּבְּלְבִין אול בּהָן כַסְמַלִיגִין אוֹל נָתָק נִיג וְדִי גוּנְלֵר : וראה דַגוּרְסִין אוֹל בּהֵן אוֹל כַסְטַלִּיְקְנִי אוֹל וִדִינְגִי גוּנְרַה ו דַמוּנַה ייִלְמַרִי אוֹל גָתֶק ו דַבּוֹלְמַרִי אַנְדַה כַרִי בַין ו דַגורומִי אוֹל גָתֶק נִיג ו מַרַן דוגול אול מרידן: והתגלח דמיבש אולסין דאול נָתֶק ני מיבש אירְמַסִין ו דַבֶּבְּלְסִין אוֹל בּהֵן אוֹל נָתֶק נִי ו יְדִי גוּנְלֵר אָבִּינְגִיי בִּרַשׁ : וראה דַגוּרְסִין אול פהן אול נָתָק נִי אול וַרִינְגִי גוּנְדַח ו דַמוּנַה יַוֹלְטַרִי אוֹל נָתָק מַרִידָה וו דַגורָמִי מֵרַן דוּגוּל אוֹל מֵרִידָן ו דַמָּאִיו אַמְסִין אַנִי אוֹל פּהֵן ו דַיוּבְסִין אוּפְּרַקְלְּדִין בא מִמְיז בּוּלִיר : ואם דאָגר יוּלְמַא יוֹלְמַא אוֹל נְתָק מִרִּירָא וּ מַמִּיז בּוּלְגִּינָרן סוּגָרַא : וראהו דגוּרְסִין אַנִי אוֹל כּהַן ו דַמוּנַה יַיִּלְרִי אוֹל נָתָק מַרִירָא ו אַרַאמַסִין אוֹל פַהָּן אוֹל סַרִי סַצְקַה ו מוּנְדַר דִיר אוֹל וּ ואם באנר בובינדא שורדי אכה אול נהקן ו בקרה פץ בישאי אפא אנדהו אונלדי אול נָתַק מַמִין דִירָ אוֹלוּ דַאַרוּב אַטְסִין אַנִי אוֹל פּהֵן: ואיש דַבִּישׁי נָא כַמִּין בִּי בוֹלְסָה מֵרִיסִינְדָה אָמְלְרֵינִיג לַבַּאלַר אָק לֵבַּאלַר : וראה דַגוֹרְסִין אול בּהֵן דַמוּנאַ מִרִיסִינְרָא אָמְלְרִינִיג אַק לבָאלַר מונוּבְבַנְלְר ו בהַק דִיר אוֹל ו יִילְרִי מַרִירָה מִמִיז דִיר אוֹל : ואיש דַבִּישִׁי בִּי יוּרְקוּלְסַא כַצִי בַשִׁינִיגו קַרְחַ דִיר אול מִבְיוֹ דִיר אוֹל : ואם דאַנֵר יוּזְלַרִינִיג כַּיַרְפִּינְרַן יוִרקוּלְכַה בַּצִי הַשׁיִיג וּ גְבָת דִיר מָמִיוְרָיר אוֹל : וכי דַאָנַר בּוֹלְכַה אַרְטְבַה יֵא אַלְרִירָה בַּסְמֵלִיק אַק -קיִרימִיוִיגִ'א וַ יִיִּלִי מּוּרָנָן ג'וּזַם דִּרְרָלִיכּ אוֹלוַיִּיְרִדִינַרה יא אַלְרִינַרה : וראה בגוֹרְמוּ אַנְּה אוֹר בַנוֹן בַמוּנָא שִׁישִׁינִי אוֹל כֹכֹכֹלְינְנִיג אֵל לִנְנִמוּוִהְא ו אַרְנִינְבא הָא אלרינרא

المات المات

99

אַלְרִינְרֵאוֹ גוֹתִי גִיבִּי ג׳וּזַם הַרְדְלִיבּ מֵרִיסִינִיג אָמִנִיג : איש ג׳וּזַם הַרְדְלִיבּּרִיר אול מונדר דיר אול ו מונדר אַאָמא מונדר אַמְסִין אַני אוֹל פּהָו ו בַּשִׁינְדַה דיר כַּסְמַלִיגִי : והצרוע באול ג׳וּוַם הַרְּדְלִי כִּי אַנְרֵא אוֹל כַסְמַלִיק ו אוּפְּרַקְלַרִי בּוֹלְסִינְלֶר יִרְטִילְגַנְלֶר ו דַבַּשִׁי בּוֹלְסִין אַצִיק ו דַמִיִיק נִי סַרְסִין ו דַמוּנַדְרְרִיר מונְדַרְרָיר צַקִירְסִין : כל ג׳וּמְלֵא גוְנַלֵר בִּי אוֹל כַסְשַׁרִיק אַנְדַה ו מונְדֵר בּוֹלְסִין מונְדַר דִיר אוֹל ו יַלְגִיז אוֹמוּרְסִין וֹ אַבוּלְדֵן - צִיכַרִי אוֹמוּרָשִׁי : והבגד רַאוֹל אוּפְרַק בִּי בּוֹלְסֵא אַנְרַת בַסְמֵּלִיגִי ג׳ וּזַס דַרְדְלִיבּ נִיֹגויון אוּפְרַקְמַא יַא אוּסְבּוּלוּ אוּפְרַקְדַה: או נא בּוֹיְבַה יַא אַרְקוּכְרַא אוֹסְבּוּלוּדֵן דֵיוֹנְדַן ו יַא מִּרִירַה יַא ג׳וּמְלֵא מִרִי אִישִינְרֵא : והִיה דַבּוּלְפַה אוֹל כַּפְשֵׁלִיק יַשִּילְנֵיא יַא קיְרְמִיוִיצֵיא ו אוּפְּרַקְמַה יַא מָרִירָא יָא בּוֹיָרָה יָא אַרְּקּוּבְרָא ו יַא ג׳וּמְלֵא מֵרִי מַקְמְרַה ו כַּסְאֲלִיגִי ג׳ יַּזִּס דְרְדְּלִיבְּנִיג דיר אול ז בגורונְסִין אול פּהָן גַא : וראה בגורְסִין אול פּהָן אול בַסְמַלִּיְקְנִי וּ דַבֶּקְלַסִין אוֹל כַסְשַלִיקְנִי יְרִי גוּנְלֵר : וראה דגורסין אוֹל בַּסְשַלִּיקנִי אוֹל יִרִינְגִי גוּנְרַה וֹ כִּי יֵילְרִי אוֹל כַסְמַלִיק אוּפְּרַקְמַה וֹ יָא בּוְיַרָה יַא אַרְקוּבַרְא יַא מֵרִידָה וֹ ג׳וֹמְלֵא גַה בִּי קוּלִינִיר אוֹל מֵרִי אִישְׁבַה ו סִיְזַלְמִינִי ג׳נִוֹם הַרְדִיֹּלִיבְּרִירָ אול בַסְטַלִיקן מוּנְדַרְרִיר אוֹל : ושרף דַבּוּידְוּרְסִין אוֹל אוּבְּרַקְניִ יֵא אוֹל בּוְיְנִיּ בּוֹלְסָא אַנְרָא אוֹל כַסְשַלִיקוּ זִירָא סִיזְלַמִינִי ג׳וַיִּם הַרְרְלִיבְּרִיר אוֹל וּ אוֹמְקא פּוּיְדוּרוּלַסִין : ואם דַאָגַר גורְסַא אוֹל כּהֵן דַמוּנַא יֵיִלְמֵּדִי אוֹל כַּסְמַלִיִק וּ אופְרַקְטַא יַא בּוֹיַרָאיַא אַרְקוּבְרָא ו יַא ג׳וּמְלֵא מִרִי טַקמְרָה: וצוה דְּכִיּמַרְלַכִין אוֹל כּהֵן דַיוּבְסִינְלַר גִנִי כִּי אַנְדָא אוֹל בַסְטַלִיק ו דַבּבְּלַסִין אַנִי יִדִי גוּנְלֵר אָבִינְגִי בַרַט : וראה דגוּרְסין אוֹל כּהֵן יוּבוּלְגַנִינְרון סוְגַרא אוֹל כַּסְאִילִיקְנִי דַמוּנֵא דְגִישְׁמַרִי אוֹל כַסְטֵּלִיה הַנְבְּינִי ו דַאוֹל כַסְצַלִיק יַיִלְמַרִי ו מונְדַרְרִיר אוֹל אושַקה פוירורנין אני ו פְּחָתֶת דיר אול ו אַרְטִינֶרה יא אַלְדִינְרַה : ואם דאָנֵר גורסא אול כהן ו דמונא סונובבן אול בסטלית ו יובולגנינדן סוגרא אול ו דירשקין אני אול אופרקטן יא אול מרידן ו יא אול בוידן יא אול אַרְקוּבְדוֹ: ואם דאגר גורונסא דגין אופרקטה יא בוידה יא אַרְקוּבְרָא יא ג׳וּמְלא טְרִיּ א דלת בשמח מקמרה

323500

מַּקְּמְרָהוּ יַיִּלְגַן הֵרְדְיִר אוֹל ואוֹשְׁקָא פּוּיְדוּרְגִין אַנִּי נִגִּי כִּי אַנְדֵא אוֹל כַקְּמַלִּיק : זהבגר דַאוֹל אוּפְּרָק יַא אוֹל בּוּיְ יַא אוֹל אַרְקוֹב יַא ג׳וּמְלַא אוֹל מֵרִי מַקּים פִּי יובְּטַג וּ דַבּּסְטַא אַלַּרְדַן אוֹל כַסְטַלִינִינִיג ג׳וּזַם דַרְדְלִיפְנִיג אוֹל יוּן אוּפְּרַקְנִיג בּוּלְסִין : זאת בּוּהִיר תּוֹרְסִי כַסְטַלִינִינִיג ג׳וּזַם דַרְדְלִיפְנִיג אוֹל יוּן אוּפְּרַקְנִיג וּ מַאוֹל אוֹל בּוּיְנִיג יַא אוֹל בּוּיְנִיג יַא אוֹל בּוּיְנִיג יַא אוֹל בּוּיְנִיג יַא אוֹל בּוּיְנִיג יַא מונְדֵר אִשְּׁמַגה אַנִי :

פרשת מצורע

בר ניורבר בפוולדי יני משה גא במה : זאת בי בוּלְסִין תוֹרְסִי אול ג'ווַם דַרְרָלְבְּנִיג מַמִיז בּוֹלַגֵ׳ק גוּנוּנְדָה ו דַבְּמִירִילְכִין אול כּהֵן גַה : ויצא דַצִּיקְסִין אול כּהֵן אַכוּילְרַן צִיכַרִי דַגוּרְסִין אוֹל כּהֵן ו דַמוּנַה אוֹגַלְרָי כַסְמֵּלִיגִי אוֹל ג׳וּזַם הַרְרָלִיפְּנִיגוּ אוֹל ג׳וּזַם הַרְרָלִי וצוה דַסִימַרְלַסִין אוֹל כּהֵן דַאַלְסִין מִמִיז בּוֹלַגַיִק אוצון אָכִּי יַבַּנִי הַלָּל קוּשְׁלַרִי ו דַבּלְוּי אַגַצִי דַּקִירִימִוּי אִירַוְבָּלִי יִיפָּבּ דַאָזוֹב : וצוה דַבִּימַרְלַבִין אוֹל כּהֵוֹ ! דַכּוּיְכִין אוֹל בִּיר קוּשְׁנִי צוֹרַפּ כַּגִימְרַה וֹ טַמְלִי כוּבְלַר אוּסְטוּנַה : את וַבּן לוּשְׁנִי אַלְסִין אַנִי ו דַאוֹל סַלְנִי אַנִצְנִי ו דַאוֹל קוֹיְרָמִזִּי אִירַנְּבְּלִי יִיפִּבְּנִי דאול אזוב ני ו דמנצפין אַרְרָנִי דאול יַבַּן קוּשְׁנִי ו קַנִינַה אול סויְיוּלְנַן קוּשְׁנִיג אול שַשְיִי סוּכְלַר אוּסְשוּנא : והוה דַסַצְסִין אוֹל מָשִיוְלַנְנֵן אוֹסְשוּנַא ו אול ג׳ווֹם ברְרְלִילִיקִשׁן יְדִי בּרַשְׁרֵר ו דַמִּמִיו אִמְסִין אַנִי ו דַיִבּירְכִין אוֹל יַבּּוֹ קושני יוזו אוקטונא אול דוּוְנִינ : וכבם דיוּכְסִין אול מִמִיז בּוּלְגוּן אוּפְּרַקְלֹרִינִיוּ יַדְמִירַשׁ אִידְמִין ג׳וּמְלֵא סָצִיגִי דַיִּוּבוּנְמִין בוּבְלַר בִּילַן דַמִּמִיז בּוּלְמִין ו דַאַנְרַן פוגרה גַּיְכִין אוֹל אַבוּלְגַה ו דַאוֹמוּרְסִין צַרִירִינְדַן צִיכַרי יְרִי גוּנְלֵר : והיה דַבּוֹלְגֵי אוֹל וֵדִינְגִי גוּנְדַה מִירש אִידְסִין ג׳וְמַלֹא סַצִיגִי ו בַּשִׁינִי דַא סַקּלִיגִי דַא פּירְפּקְלַרִין גוֹוּלְרִינִיג ו דא ג׳וּמְלֵא כִצִינִי מִירִשׁ אִירְסִין ו דִיוּבְסִין אופּרַקְלַרִינִי ביובְסִין אָטִיגִי סוב בִּיצָן דַמֵּמִיז בּוּלְסִין: וביום דאול בַבִּיזנָגִי גונדָה אַלְסִין 32K

כוצורן

אָבִי בַנְלַם קּוּיִלַר ו דַבִּיר טִישִׁי קוּן בִּיר יַשַּׁר בַנְלַם ו דַאוּץ פַּאוִי אונגִיג וו אווַבּ קַרִישִירְגוּן יֵג בִּילַן ו דַבִּיר מִיִבּ יַג : והעמיד דַמיִרְגוּיוְסִין אוֹל ו מָמִיז אַמְבַן בּהֵן אוֹל מַמִיוְלַנְנַן בִּישִׁינִי דַאַלַרְנִי ו אַלְרִינַה יֹנִי נִיג אִשִּׁינִינָדא אֹהֶל מועד ניג : ולקח דאל סין אול כהן אול ביר קיני ו דיובוקלשְמִירָכִין אַנִי י אָשָׁם גַא ו דַאול פִייִבּ יַנְגִי ו דַפַּלְלַפִין אַלַרְנִי פַלְלַפֵּק אוֹגונַא יֹנְי נִיג : ושחם דַּכוּיָכן אול קוְינִי גַא יִרְדַה כִּי כוֹיַר אוֹל חַטְּאת נִי דַאוֹל עוֹלְנִי יִרִינְדַה אול קורש גיג ו כִּי חַמָּאת גִיכִּי דִיר אול אָשָׁם אול כּהָן גַה ו קוֹרָשִׁי קורָשׁ לַרְנִינְדִיר אוֹל : ולקח דַאַלְסִין אוֹל כּהָן כַנִינְרַן אוֹל אָשָׁם נִיג ו רַוַרְסִין אוֹל פהן יִמְשׁנִי אוּסְשוּנַא אוֹג קוּלַגִינִיג אוֹל מֵּמִיז בּוּלְגַנִיג דַבַּשׁ בַּרְמֵנִי אוּסְשוּנֵא אוג קולוניג ו דַבַשׁ בַּדְמַגִי אוּסְמוּנה אוֹג אַיִּנִיג: ולקח דַאַלְסִין אוֹל כּהֵן אול סֵיִכּ יַגְרַן ו רַקוּיְסִין אוֹל כוֹל אַכוּצוּ אוְסְטוּנַא (אוֹזוּניג) אוֹל כּהָן נִיג : ומבל דַמַנְצְסִין אול כהן אוג בַּרְמַגִינִי אוֹל יַנְדַן כִּי אוֹל כוֹל אַבוּצוּ אוּכְמוּנַא ו דַבַּצְפִין אוֹל יַנְדַן בַּרְטֵגִי בִּילַן יְדִי בִּרַמְלֵר אַלְדִינַא יֹנְי נִיג : ומיתר דַקּלְנַן אור יַנְדַן כִּי אַבוּצוּ אוּסְטוּנַא ו וַרְסִין אוֹל כֹהַן יִמְשַׁנִי אוֹסְטוּנַא אוֹג קוּלַגִינִיג אול מִמִיז בּוֹלְגנִנְיג ו דַבָּשׁ בַּרְמַגִי אוּסְטוּנַא אוֹג . קולונִיג ו דַבַּשׁ בַּרְמַגִּי אוֹכְמוּנֵא אוֹג אַנִינִיג ו אוֹל אָשֶׁם לַנִינִיג יִרִי אוּסְמִינֵא : והנותר דאוֹל בּלְנֵנְנִי יַנְדֵן בִּי אַבוּצוּ אוּסְטוּנִא אול בּהוֹ נִיג ו וֹנְסִין בַּשִׁי אוסְטוּנַא אול מִמִיז בּוֹלְנַנְנִיג ו דַבּבְּבְרָה אִמְסִין אוּסְמוּנַא אוֹל כּהַן אַלְרִינַא יֹנְי נִיג : ועשה דַקּילְכִין אוֹל כַהֵּן אוֹל הַטְאת נִי ו דַבַּבְּרָה אַטְכִין אוֹל הַהָּן אוֹל הַטְאת נִי ו דַבַּבְּרָה אַטְכִין אוֹל הַנְיוֹ בּוּלְנֵן מוּנְדַרְלִיגִינְדַן ו דַאַנְדַן כוּגָרָא פוּיְפִין אוֹל עוֹלְנִי : והעלה דַצִיגַרְפִין אוֹל כֹהֵן אור עולני ו דאול מנְחָה ני אול מוֹבֵח כַה ו דַבַּפְּרָה אַמְסִין אוסטינה אול בהו דממיו בולסין: ואם דאנר יַרְלִי אַכַה אוֹל דַקּיבִמִי יִמְמֵוּ אַכַה ו דַאַלְסִין בִּיר קון אשם נא סַלְלַמַקּקא בַפּרָה אַמִמָא אוסמונא וראונדא ביר פּאיי אווב קרישילנן ינ בילן מנְחָנָה דַמִייכ יג : ושתי דַאָבִי קומְרוּלֵר יא אַבִּי בַּלְרַרִין בוגוּרְצִנְנִיג נגא בי יִמְסַה קוּבַמִי ו דַבּוּלְסִין בּיִרְסִי חַמְאת דַאוּל בִּירְסִי עוּלְה : והביא בומירטון אַלְנִי אוֹר פַבּיוֹינְיִי נוּנְבא מִמִיוּ בּוּלְנָיִנֹא אוֹל בּבוֹ נֹא ו אִמְינִינֹא 578

תוריע

אֹהֶל מוֹעֵר נִיג אַלְדִינַה יִנִי נִיג : ולקח ַ דַאַלְכִין אוֹל כַהֵּן אוֹל אָשְׁם קוֹיִנִּי בַאוֹל בֵיוּבּ יַגְנִי ו דַבַּרְלָסִין אַלַרְנִי אול פַהַן כַלְלַמַק אַלְרִינֵא יְנִי נִיג : ושחש דַכוּיְכִין אוֹל אָשָׁם בַּוִיונִי ו דַאַלְכִין אוֹל כַהַן כַנִינְרַן אוֹל אָשָׁם נִיג ו דַוֹרֵכִין יִמְשַׁגִּי אוּכְטוּנַא אוֹג קוּלַגִינִיג אוֹל מִמִיז בּוֹלְגַנְנִיג ו דַבַּשׁ בּּרְמַגִי אוסטונא אוג קורוניג ו רַבַּשׁ בַּרְטַגִי אוּסְטוּנַא אוֹג אַיַגִיניג : ומן דאול ינְדן קוְיִסִין אול כּהָן אול כול אַבוצו אוסטונא (אוווניג) אול כהָן ניג: והוה דַסְצְּסִין אול כּהָן אול אוג קולוניג בַּרְמַגִי בִּילֵן אול יגַרַן ו כִּי אול סול אַכוצו אוּסְמוּנַא וּ יְרִי בִּרַמְלַר אַלְּרִינַה יוֹנִי נִיג : ונתן דֵוֹרְסִין אוֹל כּהֵן אוֹל יַנְּדֵן פי אבוצו אוקטונא ו יִקשׁגִי אוקטונא אוֹג קולַגִינִיג אוֹל מָמִיז בּוֹלְגַנְנִיג רַבַּשׁ בּרָמַגִי אוּסְטוּנִא אוֹג קוֹלוּנִיג דַבֵּשׁ בּרְמַגִי אוּסְטוּנַא אוֹג אַיִגִינִיג ו אוֹרְנִי אוּסְטוּנַא קניניג אול אָשָׁם ניג : והנותר דאול קלנני אול ינדן כי אבוצו אוסטונא אל פּהַן נִיג ו וַרְסִין בַּשִׁי אוּסְטוּנַא אוֹל מֵמִיז בּוֹלְגנְנִיג ו בַפְּרָה אִמְטַה אָסְטִנַא אַלְרִינַה יָיני נִיג : ועשה דַקִילְסִין בִּירְנִי אֹל הוּרְלַרְדֵן ו נָא בַּרַלַרִינְדֵן אֹל פוגורצנניג נדן כי יִמְכַה קובמי : את נגא כי יִמְכַה קובמי אל כִּירְנִי חַמְאתו דאל בּיְרָנִי עוֹלָה אל מִנְחָה אָסְמִנֵה וּ דַכַפְּרָה אִמְסִין אל כּהֵן אל מִנְחָה אַסְמַלִּיגִי גִינִים אַכְּמָנִיג בִּי אַנְדַה כַסְמַלִּיגִי גִינִים הַרָּרִיבְּנִיג ו דַבִּימְנִיג בִּי יִמְאַסָה קובִמי מִמִיז בּוּלְגַנִינַדה : וידבר בסוּוְלִדי יני משה גא דַאַהַרן גא הַמַה : כי כִּי גַּרְסַגִיז יִרִינַה בְּנַעַן נִיג כִּי מֵן וַרִיְדִירְמִן ביוְנַה מושוּבְלִיקְקָה ו דֵוֹרָפַם כַּבְּשֵׁלִי גִין ג׳ווַם דִרְרְלִיבְנִיג ו אַוֹינְרַה וֵרִינִיג שוּשוּבְלוּגִיגִיוָנִיוֹ : ובא דגלפין פים פי אַניג אול אַב ודאַגלְטִסין אול פהן גא המה ו בַסְטַלִיק גִיבִּי גורונְדִי מַגַה אִוְבַה : וצוה דַסִימַרְלַסִין אול כֹהֵן דַבּוּשַׁמְסִינְלֹרַ אוֹל אֵוְנִי ו בּלְמִסִינְדֵן בּוּרוּן אוֹל כּהֵן גוּרְמַנַה אוֹל כַסְשֵלִיקְנִי ו כִּי מוּנְדֵר בּוּלְמַנֵי ג'וּמְלַא כִּי אול אונה ו דאנדן סונרה גלסין אול כהן גורְמַנַה אול אוני : וראה דגורְסִין אול בַסְשַׁלִיקְנִי ו בַמוּנַה אוֹל בַסְשַלִיק דוּבַרְלַרִינְבַה אוֹל אִוּנִיג יוֹל יוֹל יַשִּׁילְלַר יא קריכוזילר ו דגורומלירי אַלצַק אול רוברדן : ויצא דציקטין אול כהן אול אורן אשיגינה אול אוניג ו בפבלפין אול אוני ידי גונרד : ושב בקומיםיו

12

אול כהן אול ידיניי גונדה דגורסא ו דמונה יילדי אול בסטליק דוכדלרינדה אול אַוְגיג : וצוה דַסִימַרְלַסִין אוֹל כַּהֵן דַסוּבוּרְסִינְלַר אוֹל שַשְׁלַרְנִי כִּי אַלַרְדַה אול בסטליק ו דַמַשְׁלַסִינְלַר אַלַרְנִי שַׁחַרְרָן צִיכִרי מוּנְדַר יִרְגַה : ואת דָּאוֹל אָנְגִי קִירְסִין אִיצְבַּרְמִין צוּפְצוּבְרַה ו דַמוּבְּסִינְלֵר אוֹל מוֹפְּרַקְנִי כִּי קּירְדִילֵר ו שַתַרְרַן צִיכַרִי מוּנְרַר יִרְגָה : ולקחו דַאַלְסִינְלַר אוֹנְגַה טַשְׁלַר ו דַבָּמִיְרְסִינְלַר אוֹרְנִינֵה אוֹל מַשְׁלַרְנִיג וֹ דַאוֹגְנֵה מוֹפְּרֵק אַלְסִין דַסִילַסִין אוֹל אַוְנִי : ואם דַאֵנֶר קיִמְבָה אוֹל בַסְמֵלִיק דַבִּימְסַה אוֹל אִוְרָה ו סובוּרְגַנִינְדַן סוְגַרָה אוֹל מַשְׁלַרְנִי וּ דַקירְנַנִינְרַן כוּנְרַה אוֹל אַוְנִי וּ דַכִּילְנַנִינְרַן כוֹנְרָה : ובא דַגַלְכַא אוֹל פּהָן דגורְבַא ו דַמוּנַה יַיִּלְרִי אוֹל בַסְשַׁלִיק אוֹל אֵוְרָה ו סִיְזְלַמִיגִ׳י ג׳וּזַם הַרְרְלִיבְּרִיר אול ו אול אוְרָאו מוּנְרַרְרִיר אול : ונתץ דַיִּקְמִין אול אַוְנִי פִיְשׁלַרִינִי דַאוֹל אַגַצְלַרִינִי דַא ג׳וּמְלַא טוּפַרָגִין אוֹל אָוְנִיג וֹ דַצִיגַרְסִין שַׁחַרְדַן צִיכַרִי - מוֹנְדַר יִרְנַה : והבא בַאול בּלְנַן אול אַוְגַא ו ג׳וּמְלַא בּּכְּלַגַן גוּנְלַרְרַא אַנִי ו מוּנָרַר בּוֹלְסִין אוֹל עַכְשַׁם גַא הָגִין : והשוכב הַאוֹל יַמְקַן אוֹל אָוְדָה יוּבְסִין אופְּרַקְלַרִינִי דַאוֹּל אַשַׁגַן אוֹל אַוָרָה ו יוּבְסִין אוּפְּרַקלַרִינִי : ואם דַאָגַר גִּרְטַה גַּלְפַה אוֹל כּהָן וּ דַגוֹרְפַא דַמוּנַה יִילְמֵרִי אוֹל כַסְמַלִּיק אוֹל אֵוְדַא סִילַנָנִינְדַן סוֹנְרֵא אוֹל אֵוְגִי וֹ דַמָמִיז אַמְסִין אוֹל כַהָן אוֹל אָוְגִי ו זִירַא אוֹגְלְרִי אוֹל כַסְטֵלִיק: ולקח דַאַלְסִין סַצְּמַגַא אול אַוֹנִי אָבִּי קוּשְׁלַר ו דַסַלְוִי אַגַּצִי דַקִירִימְוִי אִירָגְבְּּלִי ייפב ראוזב : ושחש - דַסוְיִסין אוֹל בִּיר קוֹשׁנִי צוֹרַפּ סִגִישַקה ו מַשְלי סוּבְלֵר אוסטונא : ולקח דאלסין אול סלוי אַנציני דאול אווב ני דאור קורמייי אירוברי ייפבני דאול יבן קישני ו דבונצסין אַרָני קנינא אול סויולגן קושניג דאור מַטְלי סוּבְלַרְגַה ו דַסַאְסִין אוֹל אַוְגַה וְדִי בִּרַטְלַר ו וחטא דַסָּאָסִין אול אוני קני בילן אול קושניג ו דאול מַמְלי סוּבְלַר בּילַן ו דאול יַבּן קושוּ פילן ו דאול סלוי אַנְצִי בִילן דַאַזוֹב בִּילוֹ דַאוֹל קירִימִזִּי אִירַנְבְּלִי יִפָּבּ בִּילוֹ : ושלח דיבירסין אול יבן קושני שחרדן ציברי ו יוזו אוסטינא אול דוזניג ו דַבֶּפְרָה אָטְסִין אוֹל אָב אוצון דַטָּמִיוֹ בּוֹלְסִין : זאת בּוּדִיר אוֹל תּירָה הַר בַּסְטַרִינִינֵא אוֹל ג׳נוֹם דִרְדְלִיכְנִיג דַנָתָק קה : ולצרעת דְג׳וּוֹם דִרְדְלִינְינֵא אוֹל אופרקניג.

מצורע

אוּפְרַקניג הַאָּוְגַא : ולשאת דַשִּׁישִּׁיק בַּא דַאוֹיוּוְגַא דַלְבַּא גַא : להורות אוֹגְרַשְּׁסֵגַה אוֹל מונְרַר וַקִּישְנִי דַאוֹל מִמְיזּ וַלְקִישְׁנִי וֹ בּוּדִיר תוֹרָסִי אוֹל ג׳ וַּיִם הַרְדְלִיבְּנִיג : וירבר דַסוּוְלַדִי יִנִי משֶׁה גַא דַאָהַרן גַא הַמַּה -: דברו פוּוְלַגִין 10

אוֹגְלְנְלַרִינַה יִשְּׁרָאֵל נִיג דַאַיְטִינִיז אַלְרגַה וֹ כִּישִׁי כִּישִׁי כִּי בּוֹלְסַה אַקְקּוֹ אָמִינְרוֹ זָב דִיגִין מוּנְרַרְרִיר אוֹל : ווֹאת רבוֹ אוֹלֵא מוּנְדַרְלִיגִי זָב לִיגִינְרֵן ו מובן אַקְמִי אִיסָא אִייְפְּרִי אִמִינְרון אַקְמֵגי ו יַא קוֹיוּ אַקּפּרִי אִיסָא אוּסְטוּנַא אַמְינִי אַקְמֵגיינְרַן וּ-מוּנְרַרְרִיר אוֹל : כל גייִמְלֵא אוֹל טוֹשַׁבּ בִּי יַטְסַא אוּסְטוּנַא אול זָב מוּנְדֵר בּוּלְסִין וֹ דָג׳וּמְרָא אוֹל כַגִישׁ בִּי אוֹשוּרְכַא אוּסְשוּנֵא מוּנְדר בולְבוֹן: ואיש דַבִּישִׁי בִּי מִיְיַסַה מוֹשֵׁגִינָה ו יובְבִין אוֹפְרָקְלַרִינִי דֵיובוֹנְנִבִין סיבלר בילו ו במונדר בוֹלְסִין אוֹל עַכְשֵׁם גַא הַגִּין: והיושב באוֹל אוֹטוּרְנוֹ אול סגים אוסטונא ו כי אוטורסא אוסטונא אול זָב ויובסין אופּרקלריגי דיובונסין סובלר בילון דמונדר בולסין אור עבשם גא דגין : והנוגע דאול מִייְגַן אָמִינַה אול זָב נִיג ו יוּבְסִין אוּפְּרַקְלַרִינִי דִיוּבוּנְסִין סוּבְלַר בִּיצֹן ו דַמוּנְדֵר בוּלְסִין אוֹל עַבְשַׁם גַא הָגִין : וכי דבי טובוּרְסַא אוֹל זְב מִבְּיוֹגָא וּ דִיוּבְכִּין אוּפְרַקְלַרִינִי וֹ דֵיובוּנְנִסִין סוּבְלַר בּיַלוו דַמוּנְדֵר בּוֹלְסִין אוֹל עַכְשֵׁם גַא דִגִין : וכל דג׳ומלא אול בּינִב בּי אַמִּלנִסָא אוֹסְמוּנֵא אוֹל זָב ו מוּנְדַר בּוּלְסִין: וכֹל דג׳וּמְלֵא אוֹל מִייְגַן ג׳וּמְלַגַא בִּי בּוּלְכַא מִיבִּינַא ו מוּנְדֵר בּוּלְסִיו אוֹל עַכְשֵׁם בא דגין ו דאול פושרגן אַלרני ו יובְסִין אוְפַרְקְלֵרינִי דַיוּבוּנְסִין סוּבְלֵר בִּיַלוּוּ דַמוּנְרַר פּוֹלְסִין אוֹל עַכְשַׁם נַא רֵגִין: וכל דַג׳וּמְלַא כִּי מִייִכַא אַנֵר אוֹל יָב וּ בּינְינֹים, וְיֹבְשְׁבִי כִּיבְּלֵר בִּילַן ו בַיוּבְטוּן אוּפְּבַיְנִין ו בווּבוּנְסוּן סוּבְרֵר בּילּן דַמוּנְדֵר בּוֹלְסִין אוֹל עַכְשַׁם גַא דֵגִין : וכלי דַצוֹרַפּ סַנִים כּי מִיְיַסַה אָגֵר אוֹל זָב סִינְדִירִילְסִין וּ דַג׳וּמְלֵא אַגֵץ בַּנִים נִיְקּלְסִיו בוּבְלֵר בִּילֵן : וכי דַאָנִר שׁמָיון בֹּלְינְסֹא אוֹלְ זָב אַלְבַסִיוֹבון ו בַסִיסִון אוונגא וֹבָּו דוּנְלַב מִּמִוּוֹלְיִנִּוּ אוּצוּן ו ביובְקין אוּבְּרַקְרִינִי דַיוּבְקין אָשִׁינִי שַשְׁלִי כוֹבְלֵר כִּיֹלֵן דַמִּמִיז בּוּלְסִין : זביום באול מביוויני, נינבא אלמון אווינע אבי פומרולבו וא אבי בקלניו דומילאנירו דַגְלְכִין אַלְרִינָא יְנִי נִיג ו אָשִינִינָא אַבָּל מוֹעֵר נִיג ו דֵוֹרְסִין אַלְרָנִי אוֹל כַּבֵּוֹ נַה: ועשה

מצורע

נעשה דַקילְסִין אָלַרְנִי אוּל בּהָן ו בִּירְנִי חַטְאת דַאוֹל בּירְנִי עוֹלְה ו דַבַּפְּרְה אַמְסִין אוּסְטוּנִא אוֹל כּוֹזִן ו אַרְרִינָא יוֹי נִיג זְבְלִינְינָדן : ואִישׁ דַבְּישׁי כִּי ציָקְסָא אַנְדַן מוּבְּמַגִּי אוּרְלוּקניִג ו דִיוּבְסִין סוּבְלר בִּילַן ג'וּמְלַא אַמִנְג' ו דַמּנְצַדר בּוּלְסִין אוֹל עַרְשַׁם גַא דִגִּין ; וּבָּל דַג'וּמְלַא אוּפְּרַק דַג'וּמְלַא מֵרי בֵּי בּוּלְסַה אוּקְשוּגַא שוּבְּמַגִּי אוּרְלוִּקנִיגַ וּ דִייּבוּלְכֵין בּיבָלַר בִּילֵן וּ דַמוּנְדַר בּוּלְכִין אוֹל עַכְשָׁם נָא הַנִין ; ואשה דַכַמִין כִּי יַמְסֵא כִּישִׁי אַנִי מִוּכְּמַנִין אוּרְלוּקְנִיג ו הַיוּבוּנְּסִינְּלַר בוּבְלַר בִּילַן ו דַמוּנְדֵר בּוּלְסִינְלַר אוֹל עַכְשַׁם נַא דִגְין : וֹאְשַׁה דַכַּמִין בִּי בּוּלְכַה זָבָה ו אַן בּוּלְכַה אַקּאָנִי אָפִינְדן ו יָדִי נִּוּנְלֵר בּוּלְכִין יִירָקּלִינִי אוּצוּן ו דַג׳וּמְלַאַ אוֹל מִייָגַן אַגַרו מִוּנְדַר בּוֹלְסִין אוֹל עַכְשָׁם גַאַ דִגַין : ובל דַהָּר גא כּי יַמְסַה אוּסְמוּנַא נְדֵה לִיגִינְרָה ו מוּנְדֵר בּוֹלְסִין וַדְהֵר נָא כִי אוֹמוּרְסַא אוּכְטוּנַא ו מוּנָדַר בּוֹלְכִין : וכל דָג׳וּמְלֵא אוֹל טִייַנַן טוּשַנִינַא ו יוּבְּבִּין אוּפְרַקְלַרִינִי וּ דַיוֹבוּנְסִין כוּבְלַר בִּילֵן וּ דַמוּנְדֵר בּוֹלְסִין אוֹל עַכְשֵׁם גַא דֵגִין : וכל דג ומְלֵא אול טִיְיגַן הַר דּוּרְלִי סַגִימְקַה בּי אוֹשוּרְסַה אוּכְטוֹנַא וַ יוּכְסִין אופרקלריני ביובונסין כוכלר בילן במונבר בולסין אול עכשם נא בנין : ואם דאָגר אול טושַב אוסטונה אָסַה אול ו יַא אול סגיש אוסטונה ו כִּי אוֹל אושוריַרִיר אוּסְשוּנאַ שִּוּנִינְרָא אַגרו מוּנְדֵר בּוֹלְסִין אוֹל עַכְשַׁם נַא דְּגִין : וֹאסָ דָאָגַר נַמְמַה יַמְכַה כִּישִׁי אַנִי דַבּוּלְכַא נִרָה לִיגִי אוּסְשׁינַא ו דַמוּנְדַר בּוּלְכִיו יִדִי גוּנְלַר ו דַג'וּמְלֵא אוֹל מושׁב כִּי יַמְסַה אוּסְמוּנַה מוּנְדֵר בּוֹלְסִין : ואשה דַבַּמִין בִּי אַקְסָא אַקְקָנִי קַנִינִג צוֹק גוּנְלַרַ ו דוּגול וַבְּמִי בִּילַן נִדְּה לִינִינִג ו יא בּי אַקְּסָא זִייַרע נְרָה לִיגִי וַקִּימִי אוּסְמוּנַא וּ ג׳וּמְלֵא גוּוְלַרִינְרָא אַקְּקנִינִיג מונְדַרְלִיגִינוֹג ו גונְלָרִי גִיבִּי נִרְהּ לִינִינִיג בּוֹלְסִין ו מוּנְדַרְיר אוֹל : כל ג׳וּמְלֵא אול פושב פּי יַאְסַה אוּסְטוּגַא ג׳וּמְלַא גוּנְלַרִינְדא זַבְּלִיגִיג ו פושני גיפּי נִדְה ליגיניג בוּלְסִין אַגַר ו דַג׳וּמְלָא אוֹל סַגִים כִּי אוֹשוּרְסַא אוּסְטוּנַא ו מוּנְדַר בּוּלְסִין מוּנְדַרְלִיגִי גִיבִּי נִדָה לִיגִינִיג : וכּל דֵג׳וּמְלַא אוֹל מִיְיגַן אַלַרְגַה מוּנְדַר בּוֹלְסִין וּ דיובְסִין אוּפְרַקְלַרִינִי דַיוּבוּנְסִין סוּבְלַר בִּילַן ו דַסוּנְדַר בּוּלְסִין אוֹל עַכְשֵׁם גא בנין : ואם באדר שמיו ללינסא אלמנינבן ו בסוסין אווופע יני הירקר באנבן סוגרא

מצורע

פוּנְרֵא מִמְיוֹ פּוֹלְסִיּן: וביום דַאוֹל סְבִּיוִינְגִי גוּנְדָה אַלְסִין אוֹוּנֵא אָבִּי בּעַלְרִין פּוֹגוּרְצִינְנִיג ו דַגִּמִירְסִין אַלַרְנִי אוֹל פּהַן נֵה וּאָשִינְינָה אַהָּל מוֹעֵר נִיג : ועשה דַקּוּלְסִין אוֹל פּהַן אוֹל פּהָן אוֹל פּיִרְנִי חַאָאת רַאוֹל פִּיְרְנִי עִינְה וּ דַבּפְּרָה אִשְּסִין אוּסְמִינַה אוֹל פּהָן אַלְרִינֵא יֹנָי גיג ו זַבְּלִיגִינְדֵן מוּנְדְרְלִיקְלַרִי נְיִג וֹ אוֹנְבְרִי נִיג מוּנְדַרְלִיקְלַרִי נְּיָן ו באוֹלְסֵסִינְלֵר מִּינְרָלִיקְלַרִי בִּילַן ו מוּנְדֵר אִשְבּנְלַרִין יִשְּׂרָאֵל נִיג מוּנְדַרְלִיקְלַרִי נְיִן וּ דְאוֹלְסַסִינְלֵר מִינְרָ מוֹנְדֵר אִשְבּנְלַרִי בִּילן ו מוּנְדֵר אַמְבּנְלַרִי בְּיֹלְ מוֹנְדֵר הוֹיְכָל אוֹל זָב נִיג ו דַבְּיִם פִּי צִיקְסֵה אַנְּדְוֹ מוּבְּלִיקְנִיג ו מוּנְדֵר מִשְּבָּנִי אוֹרְלִיקְנִיג ו מוּנְדֵר בִּיִל אוֹל זָב נְיִג ו דַבְּישׁינִא פִי יִמְּסָה מוּנְדֵר כַמִין בִּילָן : וחרוח דְאוֹל נִדְה יִבְקְלִיגִּינְרַה וּ דִּאוֹל זְב זְבְלִיגִּינְרָה וּ בִּילֵן : וחרוח דְאוֹל נִדְה יִבְקְלִיגִּינְרַה וּ דִּאוֹל זְב זְבְלִיגִינְרָה וֹ בִּילְוֹ בְּיִבְי בִּיוֹן בִּילְנִיךְה וּ בִּילְן : וחרוח דְאוֹל נִדְה יִבְּלְיגִינְרַה וּ בִּילְן : וחרוח דְּאוֹל נִבְּה יִבְקְלִיגְנְיבְה וּ בִּילֵן : וחרוח דְּאוֹל נִבְ הִי יִמְסָה מוּנְדֵר כַמִין בִּילְן : בְּיִל וֹ בִּילִיה וּבִּים בִּי יִמְסָה מוּנְדֵר כַמִין בִּילָן : בּילִי שִּינִב וּ וּ בִּילִינְיבָה וּ בִּילְן : בּיישִינֵא בִּי יִמְּסָה מוּנְדֵר כַמִין בִּילָן :

פרשת אחרי מות

יו נייברר יני וויביקל אַלְרִינָה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִיה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְרִינִה אַלְּאַלְרִינִה אַלְּרִינִה אַלְּאַלְרִינִה אַלְרִינִי אַלְרִי דִיר אַלְרִי דִיר אַלְרו דיִיּבְסְין סִּבְּלַר בִּילַן אָטִין אַלְּרִינִי אִיּבְּין אַלְּרִינִי אַלְרִי דִּיִּין אַלְּיִין אַלְּיִין אַיִּין אַלְּרִינִי אִין אַלְּרִינִי אַלְרִי דִּיִּין אַלְּיִין אַיִּי אִלְיְעִרינִי אִיּיְלְיִין אַיְּיִין אַלְּיִין אַלְיִין בּיּלְן מִייִּין אַלְּיִין אַלְיִין בּיִּלְיִין אַלְּיִין בִּילִן אַלְיִין אַיִּי אִיּבְּין סִיּבְּין בִיּיְלְיוֹ אַלְּיִין אַבְּין בִּילְיִי אַלְיִּין אַיְּאִין בִּילְן בִּילְין אַלְּיִין אַיּאִין אִיּאִן בְּילִין אַלְּיִין אַיִּי אִיּבִין אַבְּין אַיְיִי אִבּין אַיְּיִין אַיְיִין אִייִּין אַיּיִין אַיְּיִין אַיְּיִין אִייִין אַיּיִין אַיּיִין אַיּיִין אַיּיִין אַיּיִין אָּיִין אַיִּין אַיִּין אַיִּין אַיְיִין אַיִּין אַיִּין אַיִּין אָּיִין אַיִין אַיִּין אַיִּין אָּיִין אַיִּין אַיִּין אָּיִין אָּיִין אָּיִייִין אָּיִין אָּיִין אָּיִייִין אָּיִייְיִּין אָּיִין אָּין אָּין אָּין אַין אָּין אָּין אָּין אָּין אָּין אָּין אָּין אָּין אָּין אָּין אָּין אָּין אָּין אָּין אִין אָּין אִין אָּין אִין אָּין אָּין אִין אָּין אִין אִין אָּין אִין אָּין אִין אָּין אִין אַן אָּין אִין אָּין אִין אָּין אִין אָּין אִין אָּין אִין אָּין אִין אָּין אִין אָּין אָּין אִין אָּין אִין אִין אָּין אִין אָּין י

אחרי מ'ת

אַדֵּרן אַבִּי אוֹל אוּלַקְלַר אוּסְמוּנַא צִנְּלַר ו בִּיר צֵבּ יְיָי נַא דַבִּיר צֵבּ עֵוְאוֵל גַה : והקריב דיובוקלשטירסין אַהרן אול אולַקגי בי ציקטי אוסטונה אול צָב ינינגה ו דַקילְסִין אַנִי חַטְאת : והשעיר באול אוַרַק כִּי צִיקְטִי אוּסְטוּנַא אוֹל צֵב עוָאוֵל גַה וּ מוּרְגוּװּלְכִין דִירִי אַלְדִינַה יְנִי נִיג פַפְּרָה אִטְמַה איִכְמוּנָא ו יִבּיִרְמַנַה אַנִּי עַנָאוֹל גַא אוֹל יַבּנְנַה: והקריב דיוּבוּקְלַשְׁטִירְכוּן אַהֶּרן אוֹל הַבְּנָה: פּי אַנִיג ו דַבַּפְּרָה אִמְסִין גִנְּדִי אוצון דַאָני אוצון ו דַכוּיְכִין אוֹל חַמְאת בּוּגַסִינִי בּי אַנִיג : ולקח דאַלְסִין דוֹלוּ אוֹר מַחְתָה נִי כּוֹז אוֹטְלַר אוּכְטוּנְדֵן אוֹל מִוְבֵּחַ ניג אַלְרִינְדַן יֹנֶי נִיג וּ דַרוֹלוּ אַבוּצְלַרִינִי מוּמְכוּכִין אומְיַמְלַרְנִיג אִינְנֵא בֿגירָגייָסין איצְפַרָטִין אוֹל פּּרְבַנַה : ונתן בוֹנָסִין אול מימְסני אול אוֹם איסְטוּנַה אַלְרִינָא יֹנָי נִיג ו דַיַפְּסִין בּוּלוּטִי אוֹל טוטְסוּגוִג אוֹל קפַקְנִי כִּי אוֹר שַׁחַרְתְּלִיכּ אוסְטוּנֵא דַאוּלְמַסִין: ולקח דַאַלְסִין בּוְנִירָן אוֹל בּוּגַנִיג ו דַסַּאָסִין בּרְמֵוּנָי בּילן אַלְרִינָא אול קפַקניג גון דוגושי נא ו דַאַלְרִינָא אול קפַקניג סצְקין יְרִי בּרמְלַר ו אוֹל קַנְדֵן בַּרְמַגִּי בִּילָן: ושחם דַכוּיְסִין אוּלַגִין אוֹל חַמְאת נִיג בִּי אולוסְנִיג וֹ דַנִירְגוִיוְסִין בַנִיגִי אִיצְבַרְטִין אוֹל פַרְרַנַה וַ דַקּילְסִין בִנִיגִי נִצִיבִּי קּוְלְרִי קנינא אול בונניג ו דסצסין אני אול קפק אוסטונא ו דאַלרִינא אול בקפק ניג: יכבר דַכַּפְּרָה אִמְסִין איל קוּרָשׁ אוצון מונְדַרְלִיקְלַרִינְדַן אוּנְכִנְלַרִינִיג ישְׁרָאֵל ניג ו דַמַנְמַקְלַרִינְדַן ג׳וּמִלָא יַוִיקְלַרִינַה ו דַעְלֵי קִילְסִין אֹהֶל מוּעָר נַה ו אוֹל מוֹרְשֵנוֹ בִּירְנַלֵרִינָא ו אוֹרְשַׁכִּינְרַה מוּנְדַרְלִיקְלֹרִינִיג : וכל דַהִּץ אָדְם בּוֹלְשַׁכִין אהל מוער דה ו גירריגינדא בּפְרָה אִמְמַנה קֹדָשׁ דה ציקקנינא דְנִין ו דַבְּפְרָה אַמְסִין גִנבֵי אוצון דאָני אוצון וּ בַרְינֹאָלְא בְּוֹבְיי אוצון יִשְׁרָאֵל נִיג : ויצא דאַלְסִין אוּל מִוְבִּחַ קַה ן כִּי אַלְרִינִא יְנִי נִיג ו דַבַּבְּרָה אִמְסִין אַנִיג אוּצִין וּ דאַלְסִין קּנִינְּדֵן אוּל בּוּגַנִיג דַקּנִינָרן אוּל אוּלַקְנִיג וּ דַוֹרְסִין בּוּאַקּקּלְרִי אוּסְטוּנַא אוֹל מִיִּבְּוֹ נִיג צוּפִּצִיכְרָה : והוּה דַסַצְסִין אוּסְטוּנַא אוֹל קּנְיָנָן וּ בּרְמֵגיּ פּיבוֹ יִבִּי פֹנַמְלֵר ו בַמִּלִּיוּ אֹמִׁלָּוּן אַנִּי בַאִינוֹכִלּי אַמְׂבוּן אַנִּי כּוּנְבַּבְּלִּיִּל לְבִינְבוּן אוֹגְלְנְלַרִינִיג ישׁרָאֵל נִיג : וכלה רַבִּישִׁמִירְסִין בּפַּרָה אִמְּמַבְּעֵן אוֹל קוֹרָשׁ נִי ו דא אהל מוער ני דאול מובח ני ו דיובוקלשמירסין אול מירי אולקני : וסמך 778 רמיסין

375

אדרי מית

דַטִיבון אַהַרן אָבִי קוּלְלְרִינִי פַשִׁי אוּסְטוּנַא אול טִירִי אוּלַקנִינ ו דַאִיְקּרַר אִטְבִין אוּסְטוּנָא וּ ג׳וּסְלַא יַזִיקְּרָינִי וּ דַוֹלְסִין אַלַרָנִי פַּשִׁי אוּסְטוּנַא אול אוּלַכְנְיִי וּ דַּיִבּיְרְטִין קוֹלוּ בִּילַן חַרִיר בִּישִׁינִיג אוֹל יַבַּנְגַא : ונשא הַפּוּשַרְטִין אוֹל אוּלַקּ אוןי אוּסְטוּנֵא ג׳וּמְלַא גוּנַדְעַרִינִי וַרַן יִרְנַא ו דַיִּבּיְרִסִין אוֹל אוּלַקְנִי יַבּנְנֵא : ובא דַּגִּלְכִין אַהֶּר מוער גַּה ו דַצִשְׁכִין אול כִּיכַיי אוּפְרַקלרָנִי כִּי גּיְידִי גִּירְדִיגִּינְּבַא אול קוֹרֶשׁ גאַ ו דַקוּיְסִוּן אַלַרְנִּנִּ אַנְרַה : ורחץ דַיוּבְסִין אָטִינִי סוּבְלַר בִּילוּ אַיְרוּפְּטִי וִרְבֵא וֹ דַגִּיִיסִין אוּפְרַקְלַרִינִי וֹ דַצִיְקְסִין דַקּוֹלְסִינִי בא שׁיּכְּטִין אול אולוּכְנִיג וּ דַבַּפְּרָה אָמַכִּין גַנְדִי אוצוּן דאול אולום אוצוּן : ואת דא יְגִין אוֹל הַשְּׁאַת נִיג שוּמַשְּׁסִין אוֹל מִוְבֵּחַשְּׁה: והמשלח באוֹל יִבִּירְגַן אוֹל אוּלַקְנִי עַזְאוֹל גַא ו יוּבְסִין אוּפְרַקּלַרִינִי ו דַיוּבְסִין אָטִינִי סוּבְלַר בִּילַן ו דַאַנְדַן סוּגְרַא גּירְפִין אול אַכוּלְגַה : ואת דא אוֹל חַטְאת בּוּגַסִינִי דַא אוֹל חַטְאת אוּלַגִינִי וּ כִּי גִיְרְגִיזִילְדִי קַנְּלָרִי כַּפֶּרָה אָמִמַגַא קוֹרָשׁ דַה צִיגרְסִין אַכוּלְדַן צִיכַרִיו דַבּבוּידְוּרְכִּינְלַרַ אוֹשְׁקָא ו מִרִילַרִינִי דַא אִשְלַרִינִי דַא מְוַבְּּלַרִינִי : והשרף דַאוֹל פּוּיְדוּרְרַגּן אַלַרְנִי יוּבְּסִין אוּפְּרַקְלַרִינִי וּ דִיוּבְסִין אָמִינִי סוּבְלַר בִּילַוּ ו דַאַנְדַן סוֹגרְהַ גָּלְסִין אוֹל צַבּוְּלגַה : והיתה דַבּוּלְסִין סִיְזַגַה רַסִימִינַה דוּנְיַנִיג ו אוֹל יְדינְגִיי עַיְבַה אונוּנְדַה אול עַיְנִיג ו קּיְיַנגיז ג׳נְלַרִיגִיזְנִי ו דַהָּץ אִישׁ קִילְמַגִּיז ו אוֹל יִרְלִי דאול גַרִיפ אול דִירִילְגַן אוֹרְטַגִיזְרה : כי פִּי אוּשְׁבוּ גוּנְדֵא בַּפְּרָה אָטֵר אוּפְּסוּנּיוֹדַע שׁמִינִא שִמְּמִנֵא כִיוֹנִי ו ג׳וּמִלא וֹוֹלְלַנִינִיוֹדֵן אַלְנִינִא יָנָי נִיג מֹמִינ בולוגיו : שבת שבתון דיר אול סיונה ו דַקיינגיו בּ׳נְלריגיוני ו דַסִימִי דוּנְיַנְיְג : וכפר דַבְּפְּרָה אִמְסִין אוֹל בּהֵן בִּי סִילְסֵא אַנִי דַבִּי דוֹלְרוֹרְסַא קוֹלוּנִי פּהַנְלִיפּ אַמְּמָנֵא אַמַּסִינִיג וַרִינָא ו דַגְיִיִסִין אול כִּיסַיִי אוּפְּרַקְלַרַנִי אוֹל קוֹדֶשׁ אפּרַקלַרִינִי : וכפר דַכַּפְּרָה אַמְסִין מִקְרָשִין אוֹל קוֹרֶשׁ נִיג ו רַא אֹהֶל מוִער נִי ו באוֹל מִוֹבֶּתַ נִי כַּפְּרָה אָמְסִין ו דאוֹל בּהִנְלַר אוּצִין ו דַג׳ומְלא אוּלוּסִי בּנְינִיג וּ בַּפְּרָה אִמְּסִין : והיתה דַבּוֹלְסִין בּוּ סִיוְגַה דַסִימִינֵא הוֹנְלְנְיִר אוֹצִין וִשְּׁרְאֵל נִיג ג׳וּמְלַהי נִינְרָה אוֹצִין וִשְּׁרְאֵל נִיג ג׳וּמְלַא יַזִיּקְלְרִי וּ בִּיר אוֹצִין וִשְּׁרְאֵל נִיג ג׳וּמְלַא יַזִיּקְלְרִינְוּ וּ בִּיר

וידבר

אחרי מות

יורבר דַכּוֹוְלַרִי יִנְי משָה גַה דַמַא : דבר פוּוְלַנִין אַהַרן בַּאּ דאוגְלנַלַרִינַא ו דַג׳וּמְלָא אוֹגְלנְלַרינַא ישְׁרָאל נִיג ו דַאַיְטְקין אַלַרְנַהּ בּוּדִיר אוֹל כוּז בִּי סִימַרְלַדִי יִנִּי הַמַה : איש בִּישִׁי בִּישִׁי גַ׳מַעִּמִנְּדֵן יִשְּרְאֵל ניג ואָגַר סויְכָא אוֹגוּוּ יַא קוֹי יַא אָצְכִּי אַבוּלְרַה ו יַא פִּי כויְכָה אַבּוּלְרַן צִיבַרִי זּ ואל דאשיגינא אהל מוער ניג גִּטִירְמַפַא אַנִי יוּבוּקלשְטִירְמַנַא ו קּרְבּּן יינָ נַא ו אַלְרִינַא מְשְׁבּנִינִיג יֹנָי נִיג ו כַּן פַאיִילִיר אוֹל בִּישִׁינַא כַן מוּבְמִי ו דַבָּסִילְסִין אוֹל בִּישִׁי אוֹרְטַסִינְדַן אוּלוּסְלַרִינִיג: למען אַנִיג אוּצוּן בּי בִּשִירְנַיְלַר אוּגְלַנְלַרי ישְׁרָאֵל נִיג קַרְבְּנְלַרִינִי ושׁחִיםלריני) בּי אַלַר קַרְבּן ושׁחִיםהּ אָמַיְדִירְלַר יוזר אוסטונא אול מוּוְנִיג ו דַגִּמִירְסִינְלַר אַלְרָנִי יֹנְי נָא ו אַשִׁינִינָא אָהֶל מוּעֵר ניג אול כהו נא ו בַקרבו אִשְבּיִלר אַלרני קרְבָּני קרְבָּני הַלְמִים לַרְנִיג יוֹנְי בַה וזרק דַבַּצְּׁכִין אוּל כּהֵן אוֹל קַנְנִי מִוְבֵּחַ אוֹכְטוּנַא יַנִי נִיג ו אַשִּׁינִינְדָה אהֶל מוער ניגו דַמושַמְקין אול יַנְנִי קוֹקוּסִינַא קַבּוּלְלִיקְנִיג יְיָי נַא : ולא דַקּרְבָּן אִמְטַמִּסִינְלֵר אַרְמִיק קְרָבַּנְלַרִינִי שִׁיְמַנְלַרְגַא (אוּלק סיפּמלרינא) ו כִּי אַלֵּר אַזְיִרִירְלֵּר אַרְרָלֵרינְבוּן ו אוּמוּרְלִיכּ רַסִים בּוּלְסִין בּוּ אַלַרְגַא דַווּרְלַרִינָא: ואלהם דַאַלַרְגַה אַיְטְקון בּישׁי בּישׁי נַיְמַעִטִינְדַן יִשְּרָאֵל נִיג ו יָא אוֹל נַרִיפְּטֵן כִּי דִירִילְסַא אוֹרְטְלְרֵינְדֵא וַ אִנֶּר צִינִרְסַה עוֹלָה יֵא קרִבּן : ואַל דאַשִינִינַה אהָל מוֹעֵר ניג בּאִירְמַסֵה אַנִי וּ קִילְמַגַא אַנִי יָנְי גַה וּ דַבּּסֵילְסִין אוּל בִּישִׁי אוּלוּסְלַרִינְּדַן יּ ואיש דַכּישׁי בִּישִׁי גַ׳מַעַמִינְרַן יִשְּׁרָאֵל נִיג ו יַא אוֹל גַּרִיפְּמַן אוֹל מִיִרילְגַן אוֹרְטַלְרִינְרָה ו אָגַר אַשַּׁסַה הִץ קוֹ ו דֵוֹרִימִין כִישִׁימִימִנִי אוֹל קַנְנִי אַשַּׁגַן גַ׳נְגַה רַכּּכַרְמִין אַנִי אוּרְמַסִינְרַן אוּלוּסִינִיג : כי זִירַא נַיְנִי אוּל אַמְנִיג אוֹל קּן אִיצִינְרַא דיר אול ו דַמֵן וְרָרִים אַנִי סִיזְגַה אוֹל מִוֹבֵּחַ אוּסְטוּנַא וְ כַּפְּרָה אַטְּמָגַה גַ׳ נְלַרִינִיוּ אוצון ז בי אול קן אור ב׳ן אוצון כפרה אשר : על אניג אוצון איטשים אוֹנְרַנְלְרִינֵא יִשְׁרָאֵל נִיג ו הִץ נֵ׳ן סִיְזֵרן אַשְׁמַסִין ַקוֹ ו דַאוֹל נַרִים אוֹל מִירִילְבֵּן אוּרְטַגיוִרָה אַשַּׁמִּסִין קון : ואיש דַבִּישִׁי בִּישִׁי אוֹגְלַנְלְרִינְדַן יִשְּׂרָאֵל נִיג ו דאוּל גריפטן אול טירילגן אורטלרינדה ו כּי אוֹלְסַה אַוּוּן כּיִיכּנִיג יא קרשניג כּי אַשׁלִירוּ במוכמא עווני ו ופטון אני מופֿרל בינו : בי וובא דור בוטלא סורור

אחרי מות

לְנִי נִינִי בִּילֵן וַקְּרִישִׁיקּן דִיר אוֹל ו דַאַיְמְמִים אוֹגְלְנְלְרִינֵא יִשְׂדָאֵל נִיג ו חִיץ מִנְיִיג קַנִינִי אַשְּׁמֵנִיז ו זִירֵא ג'ַנִי הַר מִנְנִיג קַנִיִּדִיר אוֹל ו גוּיְמְלֵא אַשְּבְּצִירֵרִי מִנְנִיג קַנִיִּדִי אוֹל ו גוּיְמְלֵא אַשְּבְּצִירֵרִי מִנְנְיִג קַנִּיִּנִי נִיאִּבְּיוֹ וּנִרָב וִיְרְלִיַדְן בִּמִיּנְנִי וֹכִל דַהַּר גוֹן בּיִיּבְסִין אִיּפְּרָינִי דִיוּבְסִין אִמִּינִי כוּבְלַר בִּילֵן ו דַמוּנְרֵב וִיְרִלִידְן גוֹלְמַין ווֹאם בּיִלְּחִינִי אוּפְּרַקְּרֵרִינִי יוּבְמַסָּה בּוֹלְסִין : ואם דַאָּגַר אוּפְּרַקְרֵינִי יוּבְמַסָּה בּיּלְסִין אוֹל עַבְשׁם גִּא דִגִין דִּאָמִיז בּוֹלְסִין : ואם דַאָּגַר אוּפְּרַקְרֵינִי יוּבְמַסָּה דִאָּמִין יוּבְּמַסָה ו גוּנַחִינִי צַבָּר :

ירבר דַפוֹזְלַרִי יֹנְי מֹשָׁה גַה הַמַה: דבר פוֹזְלַגִין אוּגְלַנְלַרִינַה יִשְׁרָאֵל נִיג דַאַיִּטְקוֹן אַלַרְגַה ו מִנְמִין יוֹנַ מִנְרִיגִיז : כמעשה אִישִי גיבי מיסיר וַרִינִיג בִּי אוֹשִּרְרוּגִיז אַנְרַה קִילְמַגִּיז ו רַאִישִׁי גִיבּי בְּנַען וִרִינִיג כּי מוֹ נְפִירַיִירִירְמוֹ סִיזְנִי אוֹרֵנִא קִילְמֵנִיז ו דַקנוּנְלְרֵי בּילוֹ נִיְמֵנִיז : את שַׁרַעְמְלֵרִימְנִי קּוּלִיגִיז ו דֵא רַסִימְלַרִימְנִי סַקְלְגִיז גוְמַגַא אַלַר בִּילוּ ו מַנְמִין יני מַגְרִיגִיז : ושמרתם דַסַקלְגִיז רַסִימְלַרִימְנִי רָא שַרַעְמְלַרִימְנִי וּ בּּי קיְלְסַה אַלַרְנִי אוֹל אָדָם דַמִירִילִיר אַלַר בִּילַן) מִנְמִין יָיי : איש בִּישִׁי בִּישִׁי הִיץ יובונינה שניניג ו יובוקלשְׁמֵנִיז אַשְּׁבַּרְמִמֵנִא עַיִּפּ ו מִנְמִין עַיִּי : ערות י אַיִבּיני בעני עייבּן אַנגִניג אַצְּסַנִין אַנגְרִיר אול ו אַצְטַנין עייבּיני ערות עייבין בַטִינִינִיג אַשִּנְנִיג אַצְמַנִין וּ עִייבִי אַשְּנְנִיגר אוֹל : ערות עויבין קיז קרבשיגניג קיזי אַשַּנִניג יַא קיזי אַנְנִיג וֹגַרַב דוגְנַנִי אַנְנִיג (אשנדו) גַרַכּ דוֹגְנָנִי צִיכַרִינִיג וֹגִייריםינדן) ו אַצְמַגִין עַייִפְּלַרִינִי: ערות עַיִיבִּין קיניניג אוגְלוּגנְיג ו יַא קינִינִיג קיִזיגְנִיג ו אַצְכַּונִין עֵייִפְּלַרִינִי ו זִירַא עַיִיבִּיגְרִיר אַיִּבֶּין קִיזִינִיג כַּמִינִינִיג אַאַנְיג ו דוֹגורְמְאַקנִי אַאַנְיג סַאיִילִיר ו קיז קרְדַשִּׁיגִּדִיר אוֹל ו אַצְמַגִין עַיִּבִּינִי : ערות עַיִּבִּין קיז קַרְדַשִּׁינִיג אַאַגְנִיג אַצְמַנִין וֹ זִירָא יוּבוּגִי אַשַּגְיִיג דִיר אול : ערות עַיִיבִּין קוֹ קַרַרְשִׁינִיג אַנַנְנִיג אַבְּמַנִינוֹ זִירַא יוּבוּנִי אַנַנְנִיג דִיר אוֹל : ערות עייִבִּין קַרְרַשִׁינִיג אַמַנְנִיג שַּאְמַגִין וּ (יעני) כַמִינִינַא יוּבוְּקלַשְּמֵגִין וּ יִנְגַצְגְרִיר אוֹל : ערות שֵייִבִּין בּלִינִינְנִיג אַצְּמַגִיןוּ כַּמִינִי אוֹגְלוּגְנִיג דִיר אוֹלו אַצְמָגין עִיִבִּין : ערות עַיִּבִּין בַטִינִינִיג קַרְרַשִּׁינְנִיג אַצְמַנִין וּ עַיִיפִּי קַרְרַשִּׁינְנִינְרִיר אול : ערות עיִיבִּין כַטִּינְנִיג בַלוֹנִנִיר אַבְּמִנְוּן נִא לּנִיר אוֹלְנִנִיר נִא לוֹנִוּ לוּנִיר אַלְמִנִּין אַבְּמִנְין נִא לוֹנִוּ אוֹלְנִיר נִא אלר

אַלַר ו אירַנְצִילִיבְמִיר אוֹל ואלמק) : ואשה דַכַמִין קיז קַרְרַשִׁי אוְסְטוּנָא אַלְמֵגִין ו כּוּנְרֵש אִירְמֵגַא אַנַרָ סַנְלִיגִינְרַא ו אַצְמְנַא אַיִּפִיגִי אַנִיג אַלְרִינַא : דַבַמִינְגַה גִּדָה לִיגִינְרַה מוּנְדַרְלִיגִינִיג וּ יוּבוְּקלשְמַגִין אַצְמַגַא עַיִיבִּינִי ואל דַכַמִינִינַה דוֹסְשוֹגְנִיג (קרשירבּיגניג בַנְּדִיגְיַנְרא) וַרְמַנִין יַשוּבוּגְנִי אוּדְלוּק וַיְרָמַנָאוֹ מוּנְרֵר בּוֹלְמַנָה אַנִיג בִּילֵן : ומורעה דאוּרְלוּגִינְרַן וַרְמַגִּין אַמִּשׁ הַן בּצִירְטַגֵּא מֹלֶךְ בַּא ו דִיגִיל אִמְטַגִּין מִגְּרִינְנִיג אַרִינִי ו מֵנְּמִין דארְבַּבּ בִּיצוֹ יַשְּׁמֵגִין יַשִּוּבוּן בַשִּינְנִיג ו אִיְקְרַחְלַיּקְטִיר אוֹל : ובכל דַהִיץ תוּוְרַגַה וִרְמַגִין יַמוּבוּגְנִי ו מוּנְדֵר בּוּלְמֵגַה אַנִיג בִּילוֹ ו דַכַמִין מוּרְמִסין אַלְרִינֵא תוּנַרְנִיג קוֹשוּלְמַגַה אַנֵר ו אִיקְרַחְלִיקְאַיִר אוֹל : אל מוְנַדר בּוֹלְמַגִּיזּ ג׳ומְלַא בּוּלֵר בִּילָן ו זִירַא ג׳ומְלַא בּוּלֶר בִּילָן מוּנְדַר בּוּלְרילֵר אוֹל אוּלוּסְלֵר וּ בּי מון סורוַדִירְמון אַלְרִיגִיזְרֵן: ותשמא דמונדְרַ בּוּלְרִי אוֹל יֵר ו דַסַגִינְדִים גונַחִינִי אוּסְטוּנָה ו דַקוּסְטוּ אול ֵיר אוטוּרִיוֹצְלַרִינִי : ושמרתם דַסַקלנִיז סִיז רַסִימַלְרִימְנִי דַא שַׁרַעַמְלַרִימְנִי וּ דַקִּילְמֵנִיז ג׳וּמְלַא אוּשְבוּ אִיקְרַחְלִיקּלַרְרַוּ וּ אוֹל יַרְלִי רַאוֹל גַרִיפּ אוֹל מִירִילְגַן אוֹרְמַגִּיְזְרַה : כי זִירָא ג׳וּמְלֵא אוּשְׁבוּ אִיְקרַחְלִיקְנִיוּ קילְרִילַר בּישִׁילַרִי אוֹל יִרְנִיג בִּי סִיוְדֵן בּוּרוּן ו דַמוּנְדַר בּוֹרְרִי אוֹל יִר : ולא דקיִסְמַסִין אוֹל יֵר סִיןְנִידִא וּ מוּנְדַר אָטְבַנִיגִיזְדַה אַנִי וּ נִצִיבּ קּוֹּבְטוּ אוֹלְ אולוסְנִי בּי סִיוְרֵן בּוּרוּן : בִּי סַחִי הַר בִּיְמְבִּי וגיזלי) קוּלְסַה ו ג׳ומְלא אושְבוּ אַיקרַהְלִיקּלְרָדֵן ו בָּסִילִרְלַרַ אוֹל הִילְגַן גַ׳ְנַלר אוֹרְטַסִינְדַן אוּלוּסְלַרִינִיג : ושמרתם דַסַקְלַגִיז סַקְלובוּמְנִי ז קִילְמַמֵּגַה קַנוּנְלַרִינָרו אוֹל אִיקְרַחְלִיקְלַרְנִיג בִּי קוּלִינְרִילֵר אַלְרִיגִיזְגַה ו דַמוּנְרַר בּוֹלְמֵגִיז אַלַר בִּילֵו ו מִנְמִין יֹנָי מִנְרִיגִיז :

פרשת קרושים

רכר פוּלְגִין ג׳וּמְלָה בְּי מִנְרִינְי יִינִי מְשָׁה נָא הַמַה יִנִי מְשָׁה נָא הַמַה יִנִי מְשָׁה נָא הַמַה ה דבר פוּלְגִין ג׳וּמְלָא גַ׳מַעִמִינָה אוֹגְלְנְלְרִינְגִי יִשְׁרָאֵל נִיגּ וּ דְאִימְּקִין אַלְרְגַה אַיְרוּכְסִילֶר בּוּלוּגִיז וּ בִּי אַיְרוּכְפִי דִיר מֵן יֹנְי מִנְרִיגִיו : אישׁ

קרושם

פר בישו אנסינדן דאמפינדן קורקוניו ו דא שַבּת לַרִימְנִי כַרְלַנִיוּ ו בִּישׁי מַנְרִינְיוֹ : אל קַיִירִילְמַגִיז אוֹל יוֹקְלִיקְלַרְגַאוֹ דַמוֹבְּמֵא אַבַּק קִילְמַגִיז אוֹזוּגִיוְגַה ו מַנְמִין עני מַנְּרִינִיז : ובי דַאַגַר קּרְבָּן אַמְּסַנִיז שְׁלְמִים קּרְבְּנִינִי יִנְי גַּה מוֹרַרִינִיז אוצון קַרְבָּו אָטִינִיז אַנִי : ביום קַרְבָּן אִשְׁלְבִין דַסַבַּהִינְרֵן וּ דַאוֹל קַלְנֵן אוֹל אוצוּנְגִ׳י גוּנְגַה הַגִּין וּ אוֹטְקַה כּוּיְדורוּלְסִין : ואם דאַנר אַשַּלְמַה אַשַּלְסַה אוֹל אוצונְג׳י גוּנְדַה ו בַּיַמְדִיר ופסוּלדיר) אוֹל שְּלְמִיםו קבול בולמקטיר : ואוכליו בא (ביריפ) אַשַבְצִילֵרי גוּנַחִינִי צִבּר ו זִירא קרָשִין יני ניג וִגִיל אַמְטִי ו דַבּבּסִילִיר אוֹל גַ׳ן אוּרוּסְלַרִינְדַן : ובקצרכם באורנגיגיוָדה אורגין יִריגיוְנִיג ו טַבוּסְמֵגין קייִרין טַרְלוּבִיגְנִיג אוֹרְמֵגַה ו דַבַּשַׁנִין אוֹרַגִיגְנִיג בַּשַׁקְלַמַגִּין : וכּרמך דבּוֹרְלַלִיגִיגְנִי צִּמִימְלַמַגִּין דַרַנָּסִין בּוֹרְלַלִיגִיגְנִי צוֹפְלַמַגִין וּ יַרְלִוגַה דַגַרִיפְּבָּה בָּמִישְׁבִּין אַנַרְנִי מִנְמִין יְנִי מִגְרִיגִיז: לא בִירְסִיוְלִיק אַמְאַנוּ וּ דַשַּנְמַגִּיוּ דַאַלְדַמַנִיוּ כָּמְכַא קַרְשִׁיַדְכִּיִסִינִי : ולא דַאַלְדַמַנִיוּ כָּמְכַא אַרִים בּילַן יַלְגַנְגַהוּ דָיִגִיל אָמְמַנִין אַרִין מֵנְרִינְנִיגּ ו מִנְמִין ינְיָ : לא זוּלוּמְלַמַנִין דוֹקְשׁוּגְנִי דַדוּשוּפּ אַלְּמַגִּין וּ קּוֹנְמַסִין יַנִיגַא חַקּי אִירְגַשִּיגְנִיג אִרְשַנַה דִּגִין : לא אַרְבַּמַגין בּגירָנִי ו דַאַלְרִינַה סוֹקיּרְנִיג וִרְמַגין סוּרונְמַבּ ו דַקוֹרְקְקוֹן מִנְרִינְדן מִנְּמִין יני : לא קילְמַגִיז קיגְגִירְלִיק שַׁרַעַמְשַׁמַה ו יוּז אַמְמַגִין יוּזְלַרִין יַרְלִינִיגּ בַּבְיִילַמַגִין יוּיְלַרִין זִנְגין נִיג ו דוֹנְרוּלִיק בִּילַן שַׁרַעט אִמְבִּין דוֹכְטוּגַה : יירוֹמֵגְיוֹ קַמְבָּז אוּלוּסִינְדַה ו פוּרְמַגִּין קַנִּי אוּסְפוּנָא דוֹסְטוּנְנִיג ו מִנְמִין יֹיָי : לא דושְמַן מוּמְמַגין קַרְדִשִׁיגְנִי גוֹגְלִיגְדָה ו אוֹנִימְלַמַה אוֹגוּמְלַגִין דוֹסְמוּגְנִי וּ פּי צַבְּמַניִיְסִין אוניג אוצון ג׳ורום: לא אוץ אַלְמַנִין דַבִּין קובְמַנִין אוֹגְינְ אוֹנְינָ אוּלְיּסִינְנִיג וּ בַּכּוֹנִין אַרְּצִרְדְּאִינְנִי צִנְּדִיג נִי נִיבִּי וּ מִנְּמִין יְנָי : את בֹּכִימְלַרִימְנִי פַקְלַנִיז ו תּוֹרִיגְנִי קוֹשְׁמַנִין אָבִי גִינְסְו מַרְלוּבוּגִנִי פּאְמַנִין אָבִּי גִינְסְו דַאוּבְּרַק אַבּׁי נִ, וֹנֹם הְּתֹמֹנִי ו צִּילְמַטִּין אוּסְמוּנִא : ואיִשׁ בַּנִּיְהִי בִּי וֹמְסַבַּ כַּמִיְנִנִּי מוּבְּמֵנִין אוּרְלוּקְנִיג ו דַאוֹל קרבש אוֹלוּפּ בּלְשִׁינְנֵן בִּשׁנַה ו דִיוּלוּנְמַה יולוּנְמַדִי ו יא אַזְטַרִיק בּיחַטִי וִרִילְטִרִי אַגַר ו טַפַּטִישׁ בּוֹלְסִין ו אוּלְמַסִינַלַר ו אָגַר אַזְם בּוֹלְסִין יוֹ אָזְטַרִי אָהָל מוּעָד בּוֹלְמַרִי אִיסֵא : והביא דָבְּטִירְסִין בּשְׁמַנְלִינִינִי יְיִי גַא ו אַשִּינִינָה אֹהֶל מוּעָד

קדושים י

נינ ו קוצקר אשם גא : וכפר דַבַּפְּרָה אִמְכִין אַניג איצון אול כּהָן קוצקר פִילַן אול אֲשֶׁם נִיג אַלְדִינַה יָנִי נִיג וּ יַוִיגִּי אוצון כִּי יַזִּיְקְלִי בּוּלְדִי וּ דַבּוּשִׁמִילְר אַנֵר וּ יַזּיגִינְדֵן בּּי יִזִיּקְלִּי בּוֹלְרִי : וכי אַנַר גַּלְּסַגִּיז אוֹל יִרְגַה דַטִּיבְּסַגִּיז הַר דוּרְלִי יִמִּישׁ אַגַצִי ו דַאַקְלַפְּלִי <u>סִינִיז אַקְלַפִינִי יַענִי יִמִישׁינִי ו אוץ יִיְלַלַר בּוּלְסִין</u> פִיְזְנָה ְעָרֵל ַלַר גִיבִּיו אַשַּׁלְמַכִּין וַ ובשנה דַאוֹל דוֹרְדְוּנְגְיי ייִלְרַא בּוֹלְסִין ג׳וְמְלֵא יֵמְישִׁי קְדָשׁ וּ מַכְמוֹכְלֵד יֹנָי נֵא : ובשנה דַאוֹל בִּשִׁינְגִ׳י יִילְדָה וּ אַשׁנִיז וִמִישִׁינִי אַרְמָמִירְמַגַה סִיוָגַה מַבְּסוּלוֹנִי מִנְמִין ינָי מִגְרִינִיז: לא אַשַּׁמַגִּיז אול קן ביצן ו קושְנשׁלִיק אִמְכַנִיוֹ וּ דַבּוּלומָקה בַּקְמַנִיוֹ : לא קוֹרְשׁלַמִנִיוֹ אַמְרְפִּין בַּשִׁיגִיְונִיג וּ דַצַיְפַּמָגִין בָּנָרין דָכִי כַּקַלִיגְנִיג : ושרם דַיַיִּרְמְמַק אוֹלוּ ארצון וַרְמַגִיז אַמִיגִיזְרַה וּ רַמַמְגַלִי נַאִזִי וּ וַרְמַגִיז אווּגִיזְרַה וּ מִנְמִין יּנְיּ : אל יָגִיל אָמְּמָנִין קוּזִיגְנִי אַוְדִירְמַגַּה אַנִי וּ דַאַזְמַסִין אוֹל יִיר חַלְּקִי וּ דַשּוֹלְמַסִין אוֹל יָר זִינַאַלִים : את שַׁבְּת דַרִימְנִי בַקְּרַנִיז ו דַמִקְרָשִׁימְדַן מוֹרְמִנִיז ו מִנְמִין יַנְיי אל קיירילְמַגִיז אול קַמְצִילַרְגַה רַאוֹל בִּילְרָגִי ז אִיְזְלַמַגִּיז מוּנְרַר בּוֹלְמַגַה אַלַר בִּילַן ו מִנְמִין ינְי מִגְרִיגִיז : מכני אַלְרִינְדֵן פִּירְנִיג מוּרְגִין ו דַסִיִּילַגִּין יוִּזֹלֵרין קַרְמְנִיג ו דַקוּרְסְקִין מִנְּרִיגְדַן מִנְמִין יָנִי : וכי דַאָגַר מִירִילְסַהּ בִּירְגַנָה גַּרִיפּ יִרִיגִיזְדַה ו קִיִּנְמַגִּיז אַנִי : כאזרח יִרְלִי גִיבּי סִיזְדֵן בּוּלְסִיןּ כִיְנָנָה ו אוֹל גַרִיפּ אוֹל מִירִילְגַן בִּירְנָגִיוְגַה ו דַבּוֹנִין אַנִי גַנְדיְגָנִי גִיבִּי ו זִירָא גַריפְלַר אַדיגיז מִיסִיר וַרִינְדַה ו מִנְמִין ינָי מִגְריגִיז : לא קלְמַגִּיז קיגְנִיְרְלִיק שַרַעִמְרָה ו אוּלְצַרָה מִסְקל רָא רַטַרָזִירָה : מאוני דוֹנְרוּ מֵרָזִי דוֹנְרוּ מֵשׁ וּ דוגרו בּביץ דרוגרו בייב בולסין סיוְגַה ו מִנְמִין יָנְיּ מִגִרינִיז ו בִּי צִינַרְרִים סִיוְנִי מִיסִיר וַרִינְרַן : ושמרתם דַסַקְלַנִיז ג'וּמְלֵא רַסִימַלְרִימְנִי ו דַא ג'וּמְלֵא שַרַעַמְלַרִימְנִי וּ דַקִילִינִיוּ אַלַרְנִי וּ מִנְמִין יוֹי

אולוסי אול יִרְנִיג מוֹפַּלְסִינְּלֶר אַנִי מֵשׁ בִּילֵן : ואל דֵמה אוּלְנְלָרִינָה יִשְׂרָאֵל אולוסי אול יִרְנִיג מוֹפַּלְסִינְּלֶר אַנִי מֵשׁ בִּילֵן : ואל דֵמה אוֹלְנַלְרִינָה יִשְׂרָאֵל
אולוסי אול יִרְנִיג מוֹפַּלְסִינְּלֶר אַנִי מֵשׁ בִּילֵן : ואני דַמּה אוֹלְרוּרוּלְסִין וּ
אול קרושם

אול וגיולי זירגן בּישִׁידַה ו דַבָּכַרְמִין אַנִי אוֹרְטַסִינְדן אוּלוּסִינִיג ו בּי אוְרלוּגיְנַרן נְרָדִי כּוּלֶךְ בָּאוּ מִינְרַר אָמִמַב איצון מִקְרָשִימְנִי וּ דַיִנִיל אַמְטַנַה אַרוּכְסִי אַרימְנִי : ואם דאָנַר אוּרְטָטָא אוּרְטָסַלַר אוּלוּסִי אוֹל וִרְנִיג גווְלְרֵינִי אוֹל כִּישִׁירַן ו אַצִּיקְטַן ורגנינהה אורלוגינהן מולה פא אולדורממפה אני : ושמתי בקורמין מן כִישִׁימִימְנִי אוֹל כִּישִׁינֵה דַמִשְׁבְּחָכִינַה ו דַבָּכַרְמִין אַנִי דַא ג׳וּמְלַא אוֹל אַזְּגְּלְלְרִנִי שַּרְטִינְהַן וּ אַזְמַגַה אַרְרִינְהַן אוֹל מוֹלֶךְ נִיג ו אוֹרְמַסִינְּהַן אוּרוּסְלֵרִינִיג : והנפש דאול ג׳ן כּי בוירילְפַה אוֹל בּמְצִילַרְנַה דאוֹל בּיְלַרַג׳ יבְּלְרַג אַוְמֵה אַרְטְלֵרינְדוּן וּ דורימין בישימימני אול ג׳נגה ודבסרמין אני אורשסינדן אולוסיניג: והתקדשתם דאַירוּכְסִי פוּלוּגִיז דַפוּלוּגִיז אַירוּכְסִילַר ז כִּי מִנְמִין ינְי מִגְרִיגִיז : ושמרתם בַּפַעְלְנִיוּ בַּפִימְלְנִימְנִי וַ בַּלּיִלִינִיוּ אַלְרָנִי וֹ מִנְמִין יְנָי אַיְרוּכִפִי אִמִּיוֹאֵי סִיזְנִי : בי בי בישי בישי בי קרגפה אַשַּסיני יא אַנסיני אוּלְטָא אוּלְרורוּלְסיוּו אַמַסין יַא אַנַסין קרְנִדִי ו קּנְלְדִי בַשִּׁינַא אוֹלָא : ואִישׁ דַבִּישִׁי בִּי נואָפְּלִיכּ אַטְסַה כַמִינִי בִּילֵן בִישִׁינִיג ו בִּי נוֹאָבְּלִיבּ אַטְסַה ו כַמינִי בִּילן אַרָקרַשִׁינִיג אַ אולמה אולרורולסין ו אול נואפליב אמפן בישי דאול נואפליב אמכן במין: ואיש דַבּישׁי כּי יַמְכַה כָמִינִי בּילַן אַמַ־ִינִיג וּ עַיִיבִּין אַמַסִינִיג אַצְמִי וַ אוֹלְמֵא אולְרורולְסִינְלֵר אַבִּיסִי וַ לַנְלְרִי כַשְׁלַרִינַא אולא : ואיש בַבּּישִׁי בִּי יָמְסֵא בּלִינִי ו בִּילֵן אוּלְמֵא אוּלְרורוּלְסִינְלַר אָבִּיסִי ו אִיְקְרַחְלִילַר בִּשְׁלַרִינַה אולא: ואוש דבישי פו ימָסַה אִרְפַּבּ בִּילן יַמוּבוּן בַמִּינְנִיג ו אִיקְרַחְלִיק קילרילר אביסין אולמה אולרורולסינלרו קנלרי בשלרינא : ואיש דבישי בּי אֹלְסַה כֹמִינְנִי דַא אַנֹמִינִי אִירְנְאִילִיפֹּרִיר אוּץ ו אוֹמְלֵח כּוּיְרוּרְטִינִלָּר אַנִּי בַפּּירְנִי אַלְרְדֵן וֹ דָבּוּלְמַסְין אִירַנְצִילִיכּ אוּרְטֵנִיוְדָה : ואישׁ דַבִּישׁי כּי וִרְסַה יַטּוּבין תנורנה ו אולמה אולדורולטין ו דאול תוורני באולדרניו: ואשה דכמין כי יוכוקלשפה ג'ומלא תוורניג בירינא דורם אייקלפ ימפנא אני ו באולדרגין אול בַּמִינְנִי דַאוֹל תּוּנַרְנִי ו אוּלְמֵה אוּלְרוּרוּלְסִינִלֶר לַנְּלֵנִי בַּשְׁלֵרִינַא: ואִישׁ דַבִּישִׁי בי אלבה קיו ברדשיני ו קייו אטסיניג יא קייו אנסיניגו דגורפה עייביני דאול גורפה אניג עוביניו איבוגיליפטיר אול ו בבפילפיולר גוולרינאה אוגלילריניג אולוסלריניג

קרו שים כ

אוּלוּסְלְרֵינִיג וֹ עִייִבִּין הִיז ַקְרַרַשִינִיג אַצְּאָי וֹ גוּנַחִינִי צִבְּסִין : ואישׁ דַבִּישִׁי פּי יַשְׁסַה נִדְה בַשִּינְנִי דַאַצְּסֵא עַיִיבִּינִי וֹצוֹנְרַגִּינִי אַצְאָי וֹ רַאוֹל אַצְאָי כַּוְלַרִינִיגּ צוֹגְרַגִינִי וּ דַבָּסִילְסִינְלַר אָבִּיסִי אוֹרְשַּׁסִינְדָן אוּלוּסְלְרֵינִיג : וערות דַעַיִּיבִּין קיז בּרְרַשִׁינִיג אַמַנְנְיָג אַצְמַגין ו כִּים כִּי יובוּגִינִיג אַיִיבּינִי אַצְמִי ו גוַנּחְלַרין צִבְּּכִינְלר: ואיש דַבּישִׁי בּי וַשְּׁסַה וִגְגַצַסְנֵנִי ו אַגַצַסִינִי אַעִיי אַיִּיקוֹ ו וַיִייְקְלַרִין צִבְּסִינְלַר ו מַרְרִימְלַר אוֹלְסִינְלַר : ואיש בַבִּישִׁי בִּי אַלְסַח בַּרְבַשִּׁינִיג כַמִּינִינִי נְדָה גִיבִּי דיר אוֹלו קַרְרַשִׁינִיג עֵייִבִּינִי אַצְמִי וֹ מַחְרִימְלֹרַ בּוֹלְסִינְלֵר : ושמרתם דַסַקְלַגִיז ג׳ומְלֵא רַסִימַלַרִימְנִי וּ דַא ג׳ומְלָא שַׁרַעַמְלַרִימְנִי דַקּוּלִיגִיז אַלַּרְנִי וּ דַקּוּכְמַסִין סִיְונִי אוֹל יֵר ו כִּי מֵן גַּמִירַיִּירִיְרִמְן סִיוְנִי אוֹרְגַא אוֹמוּרְמַגַא אַנְרַה : ולא דיורומגיז רַסִימְלְרֵי בִּילַן אול אוּלוּסְנִיג בִּי מֵן סוּרַיְרִירְמִן אַלְרִיגִיזְּדֵן וּ זִירַא ג׳ומְלַא בּוּלַרְנִי קִילְרִילַר וּ דַבּזְרִים אַלַרְרַן : ואמר דַאַיְמְמְטִים סִיזְגַה וּ סִיזּ מֶרַסְלַגִיז וֹרְלַרִיגִי דַמָּן וַרַיִים אַנִי סִיוְגַה מִרַסְלַמַגַה אַנִי וֹ יִר אַגַי מוּרְגַן סוּם דַבֵּל ו מִנְמִין יוֹנָי מִגְרָיגִיז ו כִּי אַיִירְדִים סִיזְנִי אוֹל אוּלוּסְלַרְדֵן: והבדלתם דַאַיִירִיגִיז אַרַבִינה אול הַלַל תּנַוְרגִיג מוְנַדֵר הוֹיר גַא ו רַאַרַסִינַה אוֹל מוּנְדֵר קוּשְׁנִיג הַלַל נַא ו דַאִירַנְצִי אִמְמַנִיז בַיְנְרֵינִיוְנִי חוּוְרְבִּילַן דַא קושׁ בִּילַן יא ג׳ומלא בּילובי קרְמִילְדַר אול יִר אוּסְמוּנַה ו בִּי אַיִירְדִים סִיזְגַה מוּנַדר אַטְמַגַה : והייתם דַבּוּלוּגִיוּ מַגַּה אַיְרוּכְסִילֵר וּ בִּי אַיְרוּכְסִידִיר מִן יֹנְיּ וּ בִּיּ בּוּלְכַה אַלַרְרָה קַמְצִי יַא בִּילְרְגִי ו אוֹ מָה אוֹלְרוּרוּלְסִינְלֵר ו שָשׁ בִּילֵן טַשְּׁלְכִינְלֵר אַלַרְנִי לַנְּלָרִי בַּשְׁלַרִינֵא :

ם פרשת אמור

כא ניאבֶּר ייני אָל משָה אָמוֹר אָל הַבּהְנִים בְּנִי אַהְרן וְאָמַרְהְ אַלְהָם אֵל בּהְנִים בְּנִי אַהְרן וְאָמַרְהְ אַלְהָם אוֹל בּהֵן לְרָנֵה אוֹגְלְנְלַרִינִא אַהְרן נִיג רָאִיִמְקוֹ אַלְרְנֵה וֹ אוֹלוּגַה מוּנְדֵר בּוֹלְמַמְיוֹ אוֹל בּהָן לְרָנֵה אוֹגְלְנְלַרִינִא אַהְרן נִיג רָאִימִקוֹ אַלְרְנֵה וֹ אוֹלוּגַה מוּנְדֵר בּוֹלְמַמְיוֹ אוֹלוּנֵה אוֹגִלְנְלַרִינִא אַהְרן נִיג רָאִימִקּן אַלְרְנֵה אוֹלוּנֵה מוּנְדֵר בּוֹלְמַמְיוֹ

210%

אורוכלביינְדַה : כי פּי אַנְצַק יַקִינִינַה אול יובוק אַנֵר וּ אַנְסִינַה יַא דַאַמַּסִינַהּ וּ יא אוּגְלִינֵא יַא קיִנִינה יַא קַרְרַשִׁינַה : ולאחותו בַקּיוֹ קַרְרַשִׁינַא אול פּוֹיְ וִיּוּ אול יובוק אַנֵר ו בִּי בּוּלְטַרִי קוֹנֵ׳א נַהּ ו אַנַר שִׁיִפּ מוְנַדר בּוּלְסִין : רָא כיינְדר בּוֹלְכַּסִין קוֹנֵא וְשִׁיִפּ כַמִינִינַהוּ אוּלוּסָלִריְנַדה וּ יִגִיל בּוֹלְמַנָה אוֹל : לא יוּרְקְמַסְינּלָר יוּרְמַק בַשְׁלַרִינְדָה ו דַבּנרִין דַכִי בַּקּלְלַרִינִיג מִירַשׁ אָרְמַסְינְלַר ו דָאִמְלַרִינְדַה צִיזְמָסִינְלַר צִיְזִמַק : קרושים אַיְרוּכְסִילֵר בּוּלְסִיגְלַר אָיִזְמַסִינּלָר צִיִזִמָק : קרושים דיגיל אַשְׁכַסִינְלַר אַדִין פָגְרִילַדִינִיג וּ זִירֵא אוֹמְלוּ קָרְבְּנְין יוֹיָ נִיג קּרְבְּנִין מִגְרִילַרִינִיג אַלֵר יוּבוּקְלַשְׁמִירַיִיְרִילֵר וּ הַבּוּלְסִינְלַר אַיְרוּכְסִי : אשה זוֹנָה בַּמִינְנִי נא בוזוק יולְלוּנִי אַלְטַכִינְלַר נָא כורוּלְגַן כַמִינְנִי אָרִינְרַן אַלְטַכִּנְלַר וּ זִירַא אַיְרוּכְסִיִדִיר אוֹל מִגְּרִיסִינַא : וקרשתו דַאַיְרוּכְסִי אִטְבִין אַנִי ו זִירַא הְּרְבָּנְיֻן שַּנְרִינְנִיג אוֹל יוּבוּנְקלַשְׁטִירַייְדִיר ו אַיְרוּכְסִי בּוּלְסִין בַּגַה ו פִּי אַיְרוּכְסִידִיר מֵּן יֹיָי אַיְרוּכְסִי אֵטִיְוְצִי סִיוְנִי : ובת דַקיוִי כּהֵן כִּישִׁינִיג אֵגַר כַּשְׁלַסַא אַוְּסֵּגַה וּ צַּמַסִינִיג כּהַנְלָיגִינִי אוֹל בְּסוּל אָמַיְדִיר ו אוֹמָקה פּוּיְדוּרוּלְמִין : והכהן דַאוֹל בויוב כהו ברדשלרינדו ו כי קייילפה בשי איסטינה אול סילמב יני ו והולרורפה כּוֹלוֹנִי גִייְמֵגַה אוֹל קוֹבֶשׁ אוּפְּרַקְלָרִנִי וּ בַּשִׁינִי אַצְמַסִין דַאוּפְּרַקְּלַרִינִי יִרְאַמֵּסִין : ועל דַהִיץ אוֹלִי גַ׳ן גַא גַּלְמַסִין וּ דַכִי אַמַסִינַה נָא אַנִסִינַה מוּנְדֶר בּוֹלְמַסִין: ומן באוֹל מִקְּרָשׁ בוֹן צִילְמַסִין וּ בַינִיל אַמְּמַסִין מִקְרָשׁין מִגְּרִיסִינִיג וּ זִיַּרָא פֹינִי סִילְמַבּ יגִינִיג מַנְרִיסִינִיג אוּכְטוּנַה דִירו מִנְמִין ינָי : והוא דַאוֹרוּכַשִּינְנִי בּוֹילִיקְלַרִי בִּילוֹ אַלְסִין : אלמנה מוּרְנִי נֵא כּוּרוּלְגַנְנִי נִא בּוּזוּק יוֹלְלְוּנִי נִאִּזוּנְנִיוּ בּוּלַרְנִי אַלְּמַסִין וֹ כִּי אַנְצַק בּוְי קּיוְנִי אוּלוּסְלַרִינְרַן אַלְסִין אווונא כָמִין נֵא : ולא ביגיל אִמְמַכִּין אוּרְלוּגִינִי אוּלוּכְלַרִינְדַה ו כִּי מֵנְמִין יוֹיָ אַיִרוּכְכִי אֵמִיוֹצִי אַנִי : נידבר דכוולרי יני משה גא דמה : דבר כוולגין אַרון גַה דמה ו בּישִׁי אוּרְלוּגִינְרֵן דַווְּרַלִרִינֵה כִּי בּוּלְכַה אַנְרֵה חִילְרֵא יוּבוּקְלַשְׁמַכִין יובוּקְלַשְׁמִירְמַה קּרְבְּנִין פוּרִיִםינִיג : כי כי ג׳וְמְלֵא כִּישׁי כִּי אַנְרַה חִילְלֵא יובוְקלְשׁמַסִין יו סוקור כִישׁי נא מופל נא אַבְּסִיבּ בּוֹבוּמְלִי נֵא אַרְמִיקּ בּוֹבוּמְלִי : או יַא בִּישִׁי בִּי בּוּלְסַה אַנְרָה בִינִיק אַיַּק יָא צוֹלְק קול : או יא קַמְפוּר יָא אִינְגִיא וכוּרִיר) יָא אַק מושכן

אמור

כב וידבר דסוילרי יני משה גא דמה : דבר סוילגין אַהרן גַה

דאוּגַלַנְלַרִינָא דַאַיִירִילְסִיְנַלֵר קוּדָשׁ לַרִינְדֵן אוֹגְלַנְלַרִינִיג יִשְׂרָאָל נִיג ו דֵוְגִיל אַמְמַסִינְּלַר אַיְרוּבְסִי אַרְמְנִי ו בִּי אַלַר קוֹרֶשׁ אַפַּיְריְרַלַר מַנַה מִנְמִין יַּנְי : אמר אַיְמְקוּן אַלַרְגַה דַווּרְלַרִינִיזְנָה ו ג'וּמְלַא בִּישִׁי כִּי יוּבוּמְלַשְׁטַא ג'וּמְלַא אוּרְלוּגִנְוְדֵןוּ אוֹל קוֹרֶשׁ לַרְגַה בִּי קוֹרֶשׁ אָשַרְלֵר אוֹנְלַנְלַרִי יִשְׂרָאֵל נִיג ענָי גַה ו דַמוּנְדַרְלִיגִי פּוּלְסֵא אוּסְטוּנַה ו דַבּּסִילִיר אול גֵ׳ן אָלְדִימְדֵן ו מִנְמִין יְיֹי : איש בּישִׁי פּישׁי אוּרְלוּגִינְרֵוֹ אַהָּרוֹ נִיג ו רַאוֹל ג׳ווַם דְרְרָלִי יַא זָב ו קוָרְשׁ לַרְרַן אַשְׁמַסִין וּ דַגינְצָא בּי מַמִיז בּוּלִיר ו דַאוֹל מִיְיגַן ג׳וּמְלַא מוּנְדַר גַיְנְלִיגַא ו יַא בִּישִׁי גַא בִּי ציקְכַה אַנְדַן טוּבְּמַנִי איִרְלוּקְנִיג : או יא בּישִׁי גַא בִּי מִיְיַמַא ג׳ומְלֵא קוֹלְגַנְגַא בִּי מוּנְדָר בּוֹלִיר אַנִיג בִּילַן ו יַא אָדְם גַה בִּי מוּנְדָר בּוֹלִיר אַנִיג בִּילַן ג׳וּמְלָא מוּנְדַרְלִיגִינַה : נפשׁ בֹ׳ן כִּי מִיְיַסָה אַנֶר וּ דַמוּנְדַר בּוֹלְכִין אוֹל עַכְשַׁם נא הנין ו דַאַשַׁמַסִין אול קוֹרָשׁ לַרְרַן וֹכִּי אַנְצַק יוּבְסַה אַטִינִי סוּבְלַר בִּילַן : דַבַּמְכָה אוֹל קוַיש דַמָּמִיז אוֹלִיר דַאַנְדֵן כוֹנְרָה אַשַׁכִין אוֹל קוֹרֶשׁ לַרְדַן וּ זִירַא אַשִירִיר אול : נכלה נְבֶלְה נִי נִא מְרֵפָּה אַשַּמָסִין מוּנְדֵר בּוּלְמֵנָה אַניג בּילַן וּ מִנְמִין יִנְיי : ושמרו דַסָּקְלַסִינְלֹרָ סַמְלוֹבוּמְנִי וּ דאַ צִבְּמְסִינְלֶרְ אַנִיג אוצהן גונח ביורו מוני באודרלר אניג בילן אבר יגיל אישמלר אני ו מנמין יני נא אִינְנְמִי אַהְּמַמִין לּעִיבְּה נִי : וכנו בַּנְיוֹ אֵינֶר כֹמִין אִלְסֵׁנִ נִּוֹ סְׁמֵּין אִינִּינִסׁ, אִמִּיוֹאַ אַלְנִי : וכּלְ בַנִּיּוֹ יִפִּנְיֹ, אַהְּמַמִין לּוַנְיְּה נִיוִימִּוּסְפּּיִנִּי כִּיוֹן נִינִּ אלגנין

J.DK

אַלְנִנִין צַּרְצַסִינִיג אול אַשַׁסִין אַנְדַן ו דַדוֹנְנִנִי אַנִינִיג אַלַר אַשַּׁסִינְלַר אַשִּׁינִדן : ובת בקוני כהן גיג כי בולסה ים כישינה ו אור אַיִּרְמַגִּינַרן אול קורָש לַרְנִיג אַשַּׁמַבִין: ובת בַקִינִי פּהָן נִיג בִּי בּוּלְכַה בוּל יַא כוּרוּלְגַן ו דַאוּרְלוּק בּוּלְמַסֵה אַנֵר ו דַקּוְ מְסַה אִוינַה אַמַּסִינִיג נַשְּׁלִיקְרִי גִיבִּי ו אוּמְמַגִּינְדֵן אַסַסִינִיג אַשַׁכִין וַ דַהִין יַבּנְגִ׳י אַשַׁמַכִין אַנְדַן : ואיש דַבּישִׁי בִּי אַשַּׁכַה קוֹדָשׁ נִי יַגְלִישְׁלִיק : בַּיַלוֹ ו דַאַרְטְטִירְסִין בּשִׁינְגִיִסִיו אוּסְטוּנַה ו דַוֹרְסִין בּהֵן גַא אוּל קוֹרֶשׁ נִי ולא דיגיל אִטְמַסִינְלַר קוֹדֶשׁ לַרִין אוֹגְלְנְלְרֵינִיג יִשְּׁרָאֵל נִיג ו נְנֵי כִּי אַיִּרְסַלַר יְנָי גַה : והשיאו דְצִבְּשַׁרְטְסִינְלַר אָלַרְגַא פִּשְּׁמַנְלִיק ג׳וּרוּמוּנִי ו אַשְׁגִּנְלַרִינְדָה קוֹדֶשׁ לַרִין דַן ו כִּי מֵנְמִין יוֹיְ אַיְרוּכְסִי אָטִיוְצִי אַלְרִנִי : וידבר דַסוּזְלַרִי יִיִּי משָׁה גַא דָמַה : דבר סוּזְלַגִין אַהַרן גַא דָאוֹגְלַנְלַרִינַה ו דַג׳וּמְלֵא אוּגְלַנְלַרִינַה יִשְׂרָאֵל נִיג דַאַיְמְקוֹן אַלַרְנַה ו בִּישִׁי בִּישִׁי גַ׳מַעִמִינְרָן יִשְׂרָצֵ נִיגו דַאוּל גַרִיפְּטַן יִשְׁרָצֻ דַאוּבִי יובוּ זַלְשַטְיְרַסוֹ קְרְבְּנִינִי וגוּמְלָא אִיְנַיִיטִלְרָיִנִּי דְגִיוּמְלָא ג׳ומַרְשְׁלִיְקְלַרִיגִי ו כִּי יובוּקְלַשְׁמִירְםַלַר יְיִ נִאַ עוּלְגָה : לרצונכם כִּילַגִיגִיז בּיבוֹ סַגְלַם אָרְבַּבּ נִי סִיגִירְדֵן קּוֹיְלַרְדַן יֵא דַאָּצְבִּילַרְדַן : כל ג׳וּמְלַא בִּי יאיש : ואיש בוּלְבֵי יובוּלְלֵי יובוּלְלֵי יובוּלְלִישְׁמִירְמַגִּיז וּ זִירַא קבוּלְלִיְקְקֵא בּוֹלְמֵז מִיזְגַא דַבִּישׁי בִּי יובוּקְלַשְׁאִירָסַה שְׁלְמִים קַרְבָּנִינִי יֹנֵי נֵא ו אָיִרְמַנָה אִינְייֵט יַא ג׳וֹמֵרְטְלִיקְקַהַה ו סִיגִירְהַוּ יַא קוֹיְדֵן ו כַגְּיָם בּוֹלְסִין קבּוּלְלִיקְקַה ו הִץ חִירְגִי בּוּלְמַסִין אַנַרה : עורת סוקור נִי נֵא קיִרקנִי נֵא סַקְּטְנִי נֵא סִינִירְלִינִי נֵא אויווֹלוני גא אַנְרִי בּוּבוּמְלִינִי ו יוּבוּקְלַשְׁמִירְמֵגִיז בּוּלַרְנִי יִנָּי בַא ו דאומְלוּ בְּוֹרְבָּו וֹרָמֵניוֹ אַלְרָבן אוֹל מִוְבַּחַ אוִסְטוּנַא יוֹי נָא : ושור דאוגוז יַא קוֹי אַרְטִיקְלִי יַא אַבְּסִיבְּלִי וּ ג׳וֹמַרְטְּלִיק קִילְנִין אַני וּ דַאִינְיִשְׁבַּה קבוּל בּוּלְמַסְטִיר : יומעוד דאָזילְבוֹ נִי נִא יַנְצִילְנוֹ נִי נָא אוזוּלָגוֹ נִי נִא בּוְצֵילִגוֹ נִי ווּכוּקְלַשְׁמִירְמָנִיוּ יֹנְי נַא דיריגיודה קילמניו : ומיד בקולונדן ים אולוסניג יובוקלשמירמגיו קרבנין בְּיִרִינִינִיג גִיוֹמְלֵא בּוּלַרְדֵן ו זִירֵא בִּי צִיפַּלְמַקְלֵרִי אַלְרָרַה חִיְלְלִי אַכְרְרַה ו כַבּוּל בּוֹלְמַסְלַרְדִיר סִיוֹנֵה : וידבר דַסוּוְלַדִי יֹנִי משה גַא דִמַה : שור אוגוּז אַ קוֹי יאַ אִצִכִּי כּי דוֹגְסָה ו דַבּוֹלְסִין יִדִי גוּנְלֶר אַנַסִינִיג יַאנִינַדְא ו דַאוֹל פַבּיוִינְגִיי גוּנְדֵן דַאוֹמֵא גֵא קבול בּוּלִיר קְרְבְּנְגַה ו אוֹמְלוּ קְרְבַּן יִנִי נֵא : ושור דאוגוו

אבור .

רַאגוּז יָא קוֹיְ וּ אַנִּי רָא בַּלְּסִינִי וּ סוְיַמַגִּיז בִּיר גוּנְרָה : וכי רַאַנַר קְּרְבְּן אִטְבִּגִּיז שׁוּפּוּרְלִּבּ קְרְבְּנִנִי יְיֹי נַא וּ מוּרַדִיגִין אוּצוּן קְרְבָּן אָטִיגִיז : ביום אוֹל גוּנְרַה אַשְׁלְּכִין וּ קַלְּרִירְמַגִּיז אַנְרַן אֵרְמַגַּה רֵגִין וּ מִנְמִין יִיָּי : ולא רֵינִיל אָשְׂמַגִּיז בַּרְלְנִי וּ דַאַיְרוּכְסִי בּוּלִירְמִין אוֹרְמַסִינְרַה אוֹנְלַנְלַרִינִיג יִשְׁרָאֵל נִיג וּ בּוּלְמַה סִיזְנִי : המוציא אוֹל צינֵרְנֵן סִיזְנִי מִיסִיר יֵרִינַרְןוּ בּוּלְמַה סִיזְנִה מִצִּיִנְה וּ מִנְמִין יִּיְּי :

יידבר בסוְלַנִין אוּגְלַנְלִרינַה : דבר כוּוְלַגִין אוּגְלַנְלִרינַה (בַּנָּ) וידבר בסוְלַגִין אוּגְלַנְלִרינַה ישְׁרָאֵל נִיג בַאִיְשְׁקִין אַלְרַגָה ו וַעַדַלַרִי ייָי נִיג בִּי צַקִירְגַיְסִיּזּ

אַלְבִי צַקִירְמַקְלַרִי קוֹרָשׁ נִיג ו בּוּלַרְדִירְלַר אַלַר וַערַלַרִים: ששת אַלְמִי גוּנְלַר קילינסין איש ו באול ודינג׳י גונבה שבת שבתון צקירטק קורש ו הין איש קילְמַגִיז ושַבְּת דיר אוֹל יִנְי נַא ו ג׳וּמְרַא אוֹמוּרַשְּלֵרִינְיז דַא : אַלה בּוּלֵרְרִירְלַר וערַבַרי יני ניג צַקירְמַקְלַרִי קוֹרֶשׁ נִיג וּ כִּי צַקירְגַיְסִיז אַלַרְנִי וַערַלַרִינְרָה: בחרש אול בורונגי עיבה און דורְדוּנג׳י גונוְנַדה אול עיניג ו אול אָבִּי עַכְשֵׁם אַרְסִינַה וּ פָּכַח יִיָּי גַאַ : ובחמשה דאון בַשִּׁינְגִיי גוּנוּנְדַה אוּשְׁבּוּ עַיְנִינ ו חַגִּי אוֹל מַצְלַרְנִיג יֹנְי גַא ו יִדי גוּנְלַר מַצְלַר אַשְׁגִיז : ביום אול בּוּרוְנְנִי גוּנְרַה צָקִירְמַקּ קוֹדֶשׁ בּוֹלְסִין סיוְגַהו הִץ קוּלְלוּק אִישִׁינִי קיְלְמַגִיז: והקרבתם דֵיוּבוּקְלַשְּׁטִירִיגִיז אומלו קרבו יני נוגלר ו אול ידינגי גוננה צקירמק קודש ו הין קוללוק אישיני קילמגיז : וידבר דכוולדי יור משה גא דמה : דבר פוולגין אוגלנלרינה ישראל ניג באישהין אלרנה וכי גלסגיו אול ירגה וכי מן וריידירמן סיונה באורסגיו אורגיני ו דגטיריגיו אילב אורגיגיוניג פולטסיני אול פהן גא ו והניף דַכַּלְלַכִין אוֹל כּוּלְבֵינִי אוגונה יני כוֹרַרִיגִיז אוצוןו כַבַּחִינְרַן אל שַבָּת ניג ו סלְלַסִין אַנִי אוֹל כּהָן: ועשיתם דַקוֹלִיגִיז כַּלְלַגַן גונוגְיְזַרָה אוֹל כּוּלְטֵנִי פַנְלַם קוֹי בִיר יַשַּר עוֹלְנָה יוֹי גַא : ומנחתו רמנחכי אבּי פַאיי אונניג ו אוב קרישילגן יג בילן ו אוטלו קרבון יוי גה קוקיכי קבולליקניג ו דְּקוּיכִנִי יני שרפ דורְדוּנְנִי פַאִי אוֹל כִייכְּנִיג : ולחם דַאִבְּמַב נֵא קוורולן אַרִישׁ נִא אַרְפַּה

7 2%

בַשִּׁי אַשַּׁמֵנִיז ז נַנְּדִיסִינַה דֵנִין אישכו גונניג ז נִמִּירנּנִיגִיוֹלְא בֵנִין כַּרְבָּנִין מַנְרִיגִיוְנִיג ו דַּסִימִי דוּנְיַנִיג דַוּיִרְלַרִיגִיוְנַה וג׳וּמְלַא אומורַשְׁלַרִינִיז דַא : וספרתם דַסַנַגִּיז אווּגִיזְגַה סַבַּּחִינְרַן אול שַבְּת נִיג ו גָּטִירְגַן גוּנוּגִיזְרָן אול סַלְלַמַק פּוּלְשַׁפִינִי וּ יְרִי שַׁבָּת לַר הָּמְמְלֶר בּוּלְסִיְנַלר : עד בַבּחינַה דְגִין אוּל יִדִינְגִ׳י שַבְּת נִיג ו סַנִגִיז אַלְלִי גוּן ו דַיוּבוְּקַלַשְּׁמִירִינִיז יַנִי מִנְהָה יָנִי גַה : ממושבותיכם אופירשלריגיודן גפיריניו כּלְלַמַק אוּמְמַגִינִי אָבִי ו אָבּי אוּלוּשִׁי אוֹנְנִיג ו אוֹבּ בּוּלְסִינְלֵר חָמֵץ פִּישְׁסִינְלַרו אִילְיכְּלַר קְרְבָּנִי יִנְי גַה : והקרבתם דיובוּקְרַשְּׁמִירִינִיז אול אושְמַבּ בִּילַן וּ יְדִי סַגְּלַם קוְיְלַר בִּירַר יַשַּרְלַר וּ דַבּוּגַה סִיגִיר בַּלַסִי בִּירוּ דַקוֹצְקְרָלֵר אָבָיָ וּ בּוּלְְסִינְלָר עוֹלְה ייִי נֵה וּ דַמִנְחָה לַרִי דַקוּיְמָקְלֵרִי אוֹשְׁלוּ כְּרְבַּן קיקוסי קבילְלִיקְנִיג יני נא : ועשיתם דַקּילִיגִיז אוּלַגִין אָצְבָּילַרְנִיג בִּירְנִי חַמְאַתּ קה ו דאבי קוילר בירר ישרלר שלמים קרבנינה : והניף דסללסין אול כהן אַלַרְנִי אוּל אִילְבָּלֵר אוּמְמֵנִי בִּילוּ סַלְלַמֵק אוֹגוּגַא יֹנֵי נִיג ו אָבִּי קוּיְלַר בִּילֵן ו קוֹרָשׁ בּוֹלְסִינְלֵר יָיָ גַה וּ אוֹל כּהָן גַה : וקראתם דַצַקִירִיגיִז גְנְדִיסִינְרָא אוּשְׁבָּוּ גונניג צַקירָמַק קורָשׁ בּוּלְמִין סִיוֹנָה וּ הָץ קוּלְלוּק אִישִׁינִי קּוּלְמַנִיז וּ רַסִימִי דוּנְינִיג ג׳וּמְלֵא אוֹמוּרַשְׁלַרִינִיז רַא דּווּרְלַרִינִיז : ובקצרכם דאוּרְנַנִיגִיזְרָה יַרִיגִיוְנִיג אוֹרַגִינִי ו שַבוּסְמַנִין קּייִרין מַרְלוֹבוּגְנִיג אוֹרְגַנִיגְרה ו דַבּשׁגִין אוֹרַגִיגְנִיג בַשַּקלמַגין ו יַרְלִיגַה דַגַריפְּבָּה בִּמִישְׁבִין אַלְרָנִי ו מַנְמִין ינִי מִגְרִינִיו : וידבר דַסוֹלַרִי יוֹי משָׁה גַא רָמַה : רבר סוְילַגִין אוגְלַנְרֵינַא יִשְׂרָאֵל נִיג רָמָה ו אול וְרִינְגִיי עִיְרַה בִּירִינְרַה אול עיִנִיג וו בּוּלְסִין סִיוְגַה שַׁבְּתוֹן ו סַנִינְמַקְלִיגִי קיצְקירְמַקְנִיג צַקִירָמַק קוֹרָשׁ : כל הץ קוּלְלוּק אִישִׁינִי קילְמַנִיז ו דַיוּבוּקְרָשְׁמִירִינִיז אוֹמְלוֹ קָרָבְּן יִנָּי נַה : וירבר דַכוּוְלַרִי יִנְי מֹשֶׁה נַה דִּמַה : אך מַבּ אוֹנוּנְדָה אוּשְבּוּ יְדִינְנְיִי עִינִיג ו אוֹל יוּלְמִקְלַר גונוּ דִיר אוֹל יצַקיְרְמַק קוֹרְשׁ בּוּלְסִין סִוְנַה וּ דַקּיִינָנִיוּ צַיְנְלִיינִיוְנִי וּ דַיִוּבוּקְלַשְׁאִירִינִיוּ אוֹשְׁלוּ קְרְבַּן יְיִי נַה : וכל דַהָּץ אִישׁ קִילְמַנִיז נִנְרִיסִינְרָה אוּשְׁבּוּ נִוּנְנִיג ו זִירַא גוּנוּ יוּלְמַקְלַרְנִינְרִיר אול ו בַּפְרָה אִמְמַנָה אוִסְמוּנִיוְנַה ו אַלְרִינַה יְנֹי מִנְרִינִיו נִיג : כי בִּי נִינְמַלֵא אול נ׳ן אנר קיניססה ננדיסינדה אישבו גונניג ו בּסִילִיר אולוְסַלְרִינְדֵן : וכל ר'וסלא

בר

אמור אמור

תניושלא אול נין כי קולפה הץ איש גנריסונדה אישכי נונינ ו דקיים אפרסין אול ביפְנִי אוֹרְטַפִּינְרַן אוּלוּכִינִינ : כל יִדְץ אִישׁ בְּוּלְמַנִיז ו רַכִּימֵי דוּנְיֵנָינּ חווהַלַרִינִיזְנָה וּ ג'וּמְלַא אושוּרְשַלַרִינִיז רַא :שבת שַׁבְּת שַׁבְּתוֹן דִיר אוֹר פִיזְנָה יוּ דַיּקּוְינַנִיוֹ נַ׳ נְּלַרִיגִיוְנִי וּ פוּקּוִוּיְנַרָה אול עיְנִיג עַבְשַׁם רַאוֹ עַכְשַׁם דֵּן עַכְשַׁם בַא בּגִין יו שַבְת אִירִיגִיז שַפַּתִּינִיז נֵי : יוירבר דַבּווְלַרִי יְנִי מֹשֶׁה נַא דָפַה : רבר יפוןלנין אוגקנלרינה ישראל ניג דמה ואון בשונג׳י גונינהה אושבן ידינג׳י אַיְנִיג יו חַגִּי אוֹל כַבְּלַרְנִיג יו יְרִיצוּנְלַר יֹיִי גַה : ביום אוֹל בּוּרונְגָיי גוּוְדַה צַּקּירָמַק קוֹרָשׁ וּ הִין קוּלְלוּק אִישִׁינִי קילְמַנֵּיִז : שבעת יִרי בּנְבָלר יובוקלישְׁשִירִיגִיז אוֹמְלוּ בְּקּרָבּן ינוֹ בַה ו אול בִבּיוִינְגִי׳ גוּנְרַה צַקּירָבַק קוֹרְשׁ פּוֹלְכִיּוֹ פּיוְנַה וּ דַיוּבוּקְלַשְׁטִירִיגִיז אוֹמִלוּ לְתְבֵּן יָנִי נַה מִיִילְמַבְּמִיר אוֹל וּ הִץ קּוּלְלוּק אִישִׁינִי קּילְבָּונִיז : אלה פּוּלַחָרִירְלַר וַדַלַרִי יִנָּי נִיג יו בִּי אַקִירְנִיְבִיִם אַלַרְנִי צַקּירְבַיְ פורש ניג יו יויבוקלשטירמנה אוטרו קחבן יני בחועולה דמנחה שלמים דקיימקלר ו בּמִין גוּנְנוּג צוּנוּנְרַה : מלבר שַבַּת לַרִינְרַן בַשְׁלַה ינֵי ניג ו ירַבּּלְשִׁישְׁבַרינִיזְרֵן פּשְׁקָה וֹ פִּי זֶרְכַגִּיוֹ יוֹנְ גַה : אך מֵב און בִּשִׁונְגִיי גינונְנָחה אוֹל יְרִינְגִיי אַנָּיג ו טוּפְּלַבַנִינִייְעַרה אול יְרָנִיג מַבְּסוּלוֹנִי ו יחַג אָטִיגִיז יַחְבֵּין יַּיְיֹי נִיג יְיִדִּי גונְלַר וּ אוֹל בוחונְגִיי גונְרָה שַׁבְּתוֹן וֹ דַאוֹל בַבְּיוֹינְנִי גונְרָה שַׁבְּתוֹן: זלקחתם ראַליגיז אוזיגיוְגַה אול בורוְנג׳י גונְדֵן יבורון וו וסבהליקו יִמִישִין ביְילִי אַנַצְנִיג בוּרְמַלֵר יַפְּרַקְלַרִינִי בַפּוּטַגוּן קַלִיוּ יַפְּרַקלִי מִרַכְּנִיג הַמַּלְלַרִין אוֹזְנְנִיג ו דַּהָיִינִיגיז אַלְרִינַה יָּיָּי נִיג יְרִי גוּנְלַר : וחגתם דַחַג אָפִיגִיז אַנִי חַג יְיִּי נֵה יְיִדִי גּוּנְלַר יִלְרַה וּ רַסִימִי הוּנְנָנִיג דַוּיִרְלַרִיגִיוְגַה וּ אוֹל יְרִינְנְיִי שַיְרַה הַג אָטִינִיז אַנִי : בככות סבה לרבא אוטורוגיז ידי נונכר ו ג'ומלא אול ירלי ישראל הה אושוּהְסוּנְלַר סָבָּה לַרָרָא : למען אַניג אוצון בִּילְנַיָלַר דווּרְלַרִיגִיז בִּי צַלְשְּׁלַרְרָא אושוּרְגוּיוְדִים יִשִּׁרָאֵל אוֹגְלַנְלַרִינִי וּ צִינִרנִּנִימְרַה מִיכִיר מִיכִיר וִיינְבַןּוּ מִנְּמִין ייָ פּגְרִינִיז : וידבר דַפּוּוְלַדִי משָה וַדַלִרין יוֹיִי נִיג וּ יִישִּרְאַל אוגלנַלְרִינַא: וירבר דַסווְלַרִי יִיָּי משָה גַא דָמַה : צו סִימַרְלַנִין יִשְּרָאֵל אוֹדְלְנְּלְרִינָה ו בַאַרְּשִׁינְּלֶר שַנַה מָמִיו וִיתוּן וֹגִי אִיסְבּנְגִיהַן אַיְלְבָא יריקקה

389

コージを

יְנִילְקָה וַנְיִירְמָה צִירַק דַייִם : מהיין צִיבַרְמִין פַּרְדַסִינַה אוֹל שַחַדַּהְּלִיכְּנִיג אהל מוער בא ו ירַשְּׁמִירָסִין אַנִי אַהַרו עַכְשַׁם דַן אַרְטַנַה דֵגִין אַלְרינַה יְיֹי נִיג - בַּיִּים וּ דְּסִימִי דוּנְיַנִיג דַווּרְלְרִיגוִזְגַה : על אול טָמִיז אַלְמִין שַמִידן אָסְאָנֵא ו יַרַשְּׁטִירְסִין אוֹל צִירַקְלַרְנִי אַלְרִינַה יָנִי נִיג דַיִים : ולקחת דאַלגִין אווכ בּפּישִירְגִין אַנִּי אוֹן אַבִּי קלין אַבְּמַבְּרַרוּ אָבִי אוּלוֹשִׁי אוּנְנִיג פּוּלְסִין אוּל בִּיר קלין אַבְּמַב : ושמת דקוֹינין אַלרַנִי אָבִי יַרַשְׁמִירְמַבְּלַר ו אַלְמִי אוֹל יַרַשְׁמִירְמַבְּנִיו אוֹלַ מַמִין מַרַפִּיו אוּסְמוּנַה אַלְּרִינַה יָנִי נִיג : ונתת דַוֹרְגִין אוֹל יַרַשְׁמִירְמַק אוסמונה מִמִין מִמְיֵין ו דַבּוּלְסִין קּרְבּנְנַה מוּמְסוּנַה אוֹמְלוּ קְרַבּן יְיָי נַה : ביום אוֹל שַׁבְּת גוּנְרֵה אוֹל שַבְּת גוּנְרָה יַרַשְׁמִירְסִין אַנִי אָלְרִינַה יֹנְי נִיג רַיִים ו קטינְדַן אוֹגְלַנְלַרִינִיג ישְׁרָאֵל נִיג שַׁרְטִי דוּנְיֵינִיג : והיתה דַבּוֹלְטִין אַהַרן גַה דַאוֹגְלַנְלְרֵינֵא ו דַאַשַׁסִינְלָר אַנִי אַיְרוּכְסִי וֹרְדָה ו בִּי קוֹרֶשִׁי קוֹרֶשׁ לַרְנִיגְדִיר אוֹל אַנַר וּ אוֹמְלוּ ,ַלְרְבָּנְלַרִינְּרֵן יֹנָי נִיג וַ הַבִּימֵי דונְינִנִיג : ויצא דבּיְקְמִי אוֹנְלִי יִשְׂרָאֵל כַמִינְגִיג וּ דַאוֹלְ אוֹגְלִי מִיסִירְלִי בִּישִינִיג וּ יִשְּׂרָאֵל אוֹגְלַנְלַרִינִיג אַרַסִינְדַן וּ : דַּמַלַשְּאָילֵר אַבּוּלְרָה וּ אוֹגְלוּ אוֹל יִשְּרָאֵל כַמִּינְנִיג וּ דַאוֹל יִשְּרָאֵל כִּישִׁי ויקב דַבּלְגִילִי אַשְּׁמִי אוֹגְלִי אוֹל יִשְּׂרָאֵל בַּמִינְנִיג אוֹל שֵׁם נִי דַקּרְגַרִי וַ דַגִּמִירְרִילֵר אַנְי משָה גַאוֹדַאַמִי אַנַסִינֵיג שְׁלְמִית קִינִי רְבְרִי נִיג שֶׁבְמִינְרַן דְןּבְּנִיג : ויניחוחו הַקוֹיִרילַר אַנִי סַקְּלוּבְדָה וֹ בֵּיֵין אוּלְמֵנֵה אַלַרְנֵה וֹ בּוּיְרוּגֵי בִּילַן יְנִי נִיג : וירבר דַכווּלַדִי יני משָׁה גַא דָמַה: הוצא צִינִרְנִין אוֹל קרְנַבְצִינִי אַבוּלְדִן צִיכַרִי ו דַמַיַיִסְינּלְר בּיִּקְּרָא אוֹל אָשִׁמְבַּנְרַר קוֹלְלַרִיני בַּשִּׁי אוּכְטוּנא ו דָטוּפַּלַסִינְלָר אַנִי גוּמְרָאר נִיג סוּוְלַגִין הַמַה ו בּישִׁי בּאוּגְלַנְלַרִינַא יִשְּׂרָאָר נִיג סוּוְלַגִין הַמַה ו בּישִׁי בּישׁי בּי בַּרְנַםְא מִּנְרִיסִינִי דַצִּבְּּסִין גונחִינִי : ונקב דַבַּרְנוֹצִי אַרִין יוֹי ניג ו אוּלְמֵה אוֹלְדוּרוּוְלְסִין ומֵשׁ אַמְּכָה מֵשׁ אַמְסִינְלַר אִסְמַנא ג׳וּנְיֵלָא אול ג׳מַעם נִצִיבִּי גַרִיפּ שַלַּי יִרְלִי וּ בּרְנַגַנִינְדָה שִׁם נִי אוּלְרוּרוּלְסִין: ואישׁ רַבִּישִׁי בִּי אוּרְסַה מוֹדֶסוֹן אַנִי וֹ גִין אוּרְנִינַה בִּין : ומִבה דַאוּרוּבְצִי גַיִּנִין תּנַּוְרְנִיג רוְסְאִינֵא ו נִצִיבִּי קּוְלְרִי עַלֵּי קּילִינְסְין אַנֶּר : שבר בינִיק אוְרְנֵינָה סינִיק ו אמור

79

גוז אוֹרְנִינָה גוז רִישׁ אוְרְנִינָה דִישׁ וּ נִצִיכִּי וְרָסַה חִילְלֵא אָדְס גַה עַלִּי וִרִיקְסִין אַנִּי וּ דָאוּרוּבְצִי אִלְרוּרוּלְסִין : נמכה דַאוּרוּבְצִי הַנַּוְרְנִי אוֹדְסִין אַנִּי וּ דָאוּרוּבְצִי אָדְס נִי אוֹלְרוּרוּלְסִין : משׁפֵּט בּיר שַׁרַעָט בּוֹלְסִין סִיוְגַה וּ גִּצִיכּ גַּרִיפּ עַלֵי וִיְלִי בּוֹלְסִין וּ כִּי מִנְּמִין מֹשְׁה אוֹגְלְנְלַרִינֵא יִשְּׂרָאֵל נִיג וּ דַצִיגַרְדִילַר אוֹל יִינְ מִאָּרְנִי וִ יִיבר דַכּוּוְלַרִי משׁה אוֹגְלְנְלַרִינֵא יִשְּׂרָאֵל נִיג וּ דַצִיגַרְדִיילַר אוֹל יִיג מְשָׁה בִּילֵן וּ דַאוֹּגְלַנְלַרִי יִשְּׂרָאֵל נִיג פּימִרְלַרִי יִשְּׁרְאֵל נִיג פּילְרִילֵר נִצִיכִּי סִימִרְלַרִי יִייְי מִשְׁה גַּה :

פרשת בהר

י טַגיְנָרָה סִינִי נִיג דָמַה : דבר כוּוְלַגִין אוֹגְלְנְלַרִינֵה יִשְּׁרָאֵל , טַגיְנָרָה סִינַי נִיג דָמַה : דבר כוּוְלַגִין אוֹגְלְנְלַרִינֵה יִשְּׁרָאֵל ניג דַאַיְמְקוֹן אַלַרְגַה וּבִּי גַּלְּכַנִיז אוֹל וִרְגַה בִּי מִן וַהְיִדירְמִין כִיזְנַה וּ דַשְּׁמִטְה בוּרָסִין אוֹל יֵר שִׁמְטָה יִייּ בַה : שש אַלְטִי יִילְרַר סַצְקוֹן טַרְרוּבוּנְנִי ו דַאַרְמִיּ יִלְלַר בּוּשַׁגִין בּוֹרְלַלִיגִיגְנִי דַשׁוֹפְּלַגִין מַכְסוּלִינִי : ובשנה באול יְדִינְגִי יִילְרַה יּ שַבְּתוֹן בּוּלְסִין יִרְגַה שִׁמְשָׁה יֹנָי גַהו שַרְלוֹבוּגְנִי סַצְמַנִין ו דַבּוֹרְלַלִיגִינְנִי בּוּמַמָגין : את אורַגיִּנְנִיג פִּייִגִינִי אוּרְמַגין וּ דַא דַנְלַרִין בּוְּרַלַלַרִיגְנִיג צוּפְּלַמִגִין וּ שְׁמִמְה יִילִי בּוֹלְסִין יִרְגַה : והיתה דבּוֹלְסִין שְׁמִמְסִי אוֹל יִרְנִיג סִיוְנָה אַשַמֵּנָה ו סַנָּה דַקוּלוּגַה דַקּרָבַשִּינֵה ו דַאִירְנַסְנַא דַמוּסִפִּיְרְנָא אול טירילְגִנְלַר בּירְגַנָה : ולבהמתך דתוורינה דבּיִיבַה בּי יִריגְדַהו בּוֹלְכִין ג׳וּמְלֵא מַבִילִי אַשַׁמַגַה : וספרת דַסַנִגִין אוווּגַה וִדִי שְּׁמִמְה לַרִין יִילְלַרְנִיג וּ יְרִי יִילְלַרְ יְרִי בִּרַטְּלֵר ו דַבּוּלְרַלֵר סַגַּה גוּנְלַדִּי יִדִי שְׁמִשְׁה צַׁרִינִיג אוֹל יִילְלַרְנִיג קורָק מוֹקוּז יִיל רָא : והעברת רֵגְצִירְנִין קיצְקִירְמַק שוֹפֶּר בַּסִינִי אוֹל יְדִינְנִי׳י עַיְרַה י אונוּנְרַה אול עִינִיג אול כפורים גונוּנְרַה ו נִצִירִיגִיז שוֹפְר בִּסִינָּי ג'וְמַלֵא וִרִיגִיזְדַה : וקרשתם דַאַיְרוּכְּסִי אִטִיגִיז אוֹל אַלְלִינְגִ'י יִילְנִיג מֵנָה כִּנְהוּ רַצַּקירִיגִיז מַעַפְּלִיק יִרָרה ג׳וּמְלַא אוֹמוּרִיוִצִילֵרִינָהו יוֹבֶּל דִיר אוֹל בוּלְסִין סִיזגְהוּ דקיטיגיז

בהר

דַקּוְטִיגִיז דַר בִּישִׁי טוּצוּבְלִיגִינַה ז דַהַּר בִּישִׁי מִשְׁבְּחַמִינָה קּיְמִיגִיז : יובל יובל דיר אול יילי אול אַלְרִינְגִיי יִילְנִיג בּוֹלְסִין כִיוְגַחוּ בַצְּמַגִיז דַאוֹרְמַגִיז פִּיבְּלַרִינִיו דַצוֹפְּלְמָגִיז דַנַלַרִינִי ואמברלרגאז : כי פּי יוֹבֵל דִיר אוֹל ו קוֹדֶשׁ פּוּלְמִין כִיזְגַה ו אוֹל מַרְלוֹבְרַן אַלִיפּ אַשַׁגִיז כִיִיבּ מַכְכוּלוּנִי : בשנת יִילִינְרַה אוֹשְׁפּוּ יוֹבֶל נִיג ו קַוְמִינִיז הַר בִּישִׁי מוּמוּבְלִיגִינַה: וֹכִי רַאַגַר כַמְכַגִּיז כַמוּב דוֹכְטוּנָה ו יַא כַטִּין אַלְמַה כַטִּין אַלְכֵג קולוּנְדֵן דוֹכְעוּג נִיג ו קוֹינַמַגִּיז כִּישִׁי קַרְרַשִׁינִי : במכפר בַנִי בִּילֵן יִילְלַרְנִיג אוֹל יוֹבֵל דֵן כוֹנְרָה ו כַמִּין אַלְנִין ניובלגא דגין קִמִינְדן דוֹכְטוּגְנִיג ו בַנִי בִּיַלן מַבְכוּל יִילְלַרִינִיג בַטְסִין כַגַּה : לפי צוּקְלִיגִינַה גוֹרַה אוֹל יִילְלַרְנִיג צוֹק אַבְּבִין בַמוּבוּנִי ו _דאָזְלִיגִינַה גוֹרַה אוֹל יִילְלַרְנִיג אוֹ אָמְבִין כַמוּבוּנִי וֹ זַיֵרא כַנִין מַבְּכוּלְלַרְנִיג אוֹל בַמִייר כַגַה זלא דַבְינָמַגִיז בּישִׁי בַּרְדַשִׁי דַבָּינִי ו דַקוּרְקְקוֹן מִגְרִינְדֵן ו זִירַא מִנְמִין ינִי בַּוּרִינִיז : ועשיתם דַקִירִינִיז רַכִימְלַרִימְנִי וַ דֵא שַׁרַעְמְצַרִּמְנִי בַקְלַנִיז דַקּוּלִינִיז צַלַרְנִי ו דַאוֹשוּרוֹרְכָּז אוֹל יִר אוּכְטוּנַא אָמִינְלִיכּ בִּילַן : ונתנה דְוִירִיר אוֹל יִר בּיכוּלוּנִי וֹ דַאַשַׁרְסִיז פּוֹיְגוּנְצַה וֹ דַאוֹפוּרוּרָסִיז אָמִינְלִיכּ בִּילֵן אוּכְפוּנַא : וכי דַאָגַר אַיְשְׁכַנִיז נִא אַשְּׁלִים אוֹל יְרִינְנִי׳ יִירְבָה ו מוּנה כַּצְמַסַק דְטוּפְּלַמַסַק בַּכִילִימִיוְנִי : וצויתי דַכִּיכַרְלַרְמָן אַרְגִישִׁימִנִי כִיוְנָה : וצויתי דַכִּיכַרְלַרְמָן אַרְגִישִׁימְנִי כִיוְנָה ו אוֹל אַלְמִינְגִיי יִילְרַהוּ דַקּיַלר אוֹל מַבִילְנִי וּמִמורו אוּץ אוֹל יִילְלַרְנַה :וזרעתם דַכַצַרְכִיז אוֹל כִבִּיזִינְגִיי נִילְבַא וּ בַאַשְׁרְסִיז אוֹל סַבִּילָרוֹ אַסְבִּינִי וּ אוֹל בוֹלוּוֹיְנְגִי יִילְנַה בִּיוֹ יִרִישְׁבּינְנִ׳א גַּלָגַ׳כּ מַכְּסוּלִי ו אַשַׁרְסִיז אָבְבּיגִי : והארץ דאול וֵר כַמִּילְמַסִין בַּשְׁבַּמְרַה (אוֹמוּרליבּ) זִירַא מֵנִימְדַר אוֹל יִר ו אַנְגֵ׳א גַרִיפְּלֶר גַיִבּי דַאוֹמורַשְׁלֶר גַיבִּי (אוֹמוּרירסיז) סִיוֹלֵר בּיוְרַגִּמָא : ובכל דַגוּ מְלֵא הִרְיָנַדה מומוּבְלוּגִיגִיוֹנִיג ו בִילְכָלִיק וְרִיגִיז בי בּי זוּגוּרְשְלַכַא קּרְדִשִׁיגוּ דַכַּמְכַה מוּמוּבְלוּגִיְנַדֵּן וּ דַגַּלְכַה כִילַם אָטִיבָצִיכִי אוֹל יובוּק אַנֶר ו דַכִּילַם אִידְכִין כַמְּקנִיג קַרְרַשִׁינִיג : ואיש דַבּישִׁי בִּי בּוּלְמַסָה אַגַר בִילַם אִמִּבְצִי ו דֵוִמְסַה קוֹבַטִי ו דֵוִרישְׁכַא בִילַסְּדִינְיְגְיֵא וחשב דַהַפַבּ אִמְסִין יִילְלָרִין כַפוּבוּנִיג ו דַקּוְטַרְסִין אוֹל אַרְטְקוֹנִי אוֹל בִּישִׁינֵה בּי כממי

95

סַטְמִי אַנֵר ו דַקּיִמְסִין שושוּבְּלוּנִינֵה : ואם דַאֵנַר וִשְׁמַסָה קוֹבִמִי קִימִרְמֵנָה אַנֵר ו דבוֹלְסִין סַמוּבוּ קוֹלוּנְדָה אוֹל סַמִין אַלִּיוְצִינִיג אַנִי ו אוֹל יוֹבֵל יִילִיגַה דְגִין ו דַצִיְקְסִין יוֹבֵל רָא ו דַקִיִּמְסִין שוּמוּבְלוּגִינֵה: ואישׁ דַבִּישִׁי בִּי סַמְסַה אוֹמירַשׁ אָבוּ קללי שַחַרְבַהּוֹ דַכּוֹלְסִין וַעדַסִי כִילַסְלִיגִינִיג ו הַמַּם בּוֹלְגַנִיבה דִנִין יִילִי פַטוּבוּנִיגוּ בּיר ייִל בּוּלְסִין וַערַסִי כִילַסְלִיגִינִיג : ואם דַאָגַר כִילַס אוֹלוּנְמַסֵא דוֹלְגִינְנִ׳ה אַנַר תַמַם יִיל ו דַמוּלֵם בּוּלְסִין אוֹל אָב בִּי אוֹל שַתְרַדָה בִּי וַר אַנר קלהו בַשְבַּמַרה ואוֹמוּרְלִיבּוֹ סַמִין אַלִּיוְצִינַה אַנִי דַוּיְרַלַרִינַהוֹ כִילַם צִיקְמַסִין יוֹבֵל דַה: ובתי דַאוּלְרִי אוֹל סַלַלְרְנִיג ו כִּי יוּקְמִיר אַלְרַנֵה קַכַה צוּפְצוּבְרַהו פוּזוּ בִּילַן אוּל וַרְנִיג סאיִילִירו כילַסְלִיק בּוּלְסִין אַגַר וּ דַיוֹבֵר דַה כִילֵס צִיְקְסִין : וערי דַשַׁחַרְלַרִי אוֹל לֵוָי לִיכַרְנִיג ו אִוְלַרִי שַׁחַרְלַרִינִיג מושוּבְלִיקְרִינִיג ו אומוּרְלִיכּ בִּילַסְלִינִי בּוּלְסִין לִוְי לִילַרְגַה : ואשר דאָגַר כִילַם אִירְסֵא בִּיְרְסִי אוֹל לֵוֹי לַרְדֵן ו דַכִּילַם צִיקְסִין כשובו אוניג דשחרי טוטובלוגיניג יובל בא ו זירא אולרי שחרלריניג אול לני לַרְנִיג ו אוֹלְדִיר טוּטוּבְּלִיקְלֵרִי ו אוֹרְטַסִינְרַה אוֹגַלְנַרִינִיג יִשְּׁרְאֵל נִיג : ושרה דַמַּרְלוֹבוּ סַלַסִינִיג שַתַּרְלַרִינִיג סַמִילְמַסִין וּ כִּי מוּמוּכְלוּגִי דוּנְיַגִינְרִיר אוֹל אַלַרְגַה: וכי באַנַר זוגוּרְמְלַסַא דִינְדַשׁ בַּרְדַשִׁיגוַדַמִיּסָה כוֹלוּ בִּירְגַנַהוֹ דַקוּנְמְנְנְרִירְגִין אַנִּי וֹנֵבּ וֹנִיפְּנִי וֹנִבּ מוּכַפִּינְנִי וּ דַנְּאִינְסִוּ בּּינְנַנִּה : אַךְ אַלְּמַנִּיוּ 'וִאנִינְּדוּ פֿאיִיז נֹא פַמִּלָא פַּאוֹנִי וּ דַקּוֹרְקְקּקוֹן מִגְּרִינְּדֵן וּ דַגֹּאִינְסִין בּוֹרְדָשִׁיג בִּיְרַנִּרְה : את אַכְצְּגְּנִי וֹנְמַנְיוֹ אַנְר פֿאִיוֹזִגא ו בא פֿוֹבִיא וֹנְמַנִין אַשִּׁינְנִי : אני מִנְמִין יְּנָי מִנְּרִינִיוּ ו פִּי צִיגַרְדִים סִיוְנִי מִיסִיר וֵרִינָדן ו וַרְמַה סִיוְנָה בְּנַעֵן וֵרִינִי ו בּוֹלְמַה סִיוְנָה פִּנְרוֹגַה : וכי הַאַגר זוּגוְרְמְלַכַא דִינְרַשׁ קַרְרַשִׁיג בּי בּיְרַגַּגה דַכַמִיּלְכַה כַּגָהוקוּלְלַנְמִנִין אַנִי קוּלְלוּיִין קוּלְנִיג : כשכיר אִירְגַט גִיבִּי מוּסַפִיר גִיבִּי בּוּלְסִין בִּירְנַנָהוֹ יִילִינַה דָנִין אול יוֹבֵל נִיג קוּלְלוּק אָמְסִין בִּירְנֵגַה: ויצא דַצִיקְסִין יַאנִינְדַן אוֹל דַאוֹגְלַנְכַרִי בּיְרַגִּסִינָה וּ דַקִּיְמְסִין מִשְּׁפְּדָסִינָה וּ דַשוּמוּבְרוּגִינָה אַשַּׁלֵרִינִיג קִיְמְסִין : כי זִירַא קוּלְלַרִימְדִיר אָלַרו כִּי צִיגַרְדִים אַלַרְנִי מִיסִיר וַרִינְדַן ו סַמִילְמַכִינְלַר סִמִילְמַנִין קוּלְנִיג : לא אַרְבָּרָנְמָגִין אַגַר ווֹר בִילֵן ו דַקוֹרְקְקוֹ מִגְרִינְדֵן : ועברה דַקוּלוּגְ דַקרַבְשִׁיג

בהר

פִּי בּוֹלְסַלַר טַנָה ו יַאנִינְדֵן אוֹל אוּלוּסְלַרְנִיג כִּי צוּפְצוּרְרַלַריגיוְגַה ו אַלַרְדָן כַמִין אַרִינִיז קול דַקרבשׁ : וגם דַדִּגִין אוֹגְלַנְיַרִינְרן אוֹל אוטוּרַשְׁלַרְנִיג אוֹל פִירִילְגַנְלַר בּירָגַגִיוְגַהו אַלַרַדּן כַטִין אַלִיגִיז ו דַמִשְׁבְּחָה לַרִינְדן בִּי בּירְגַגיוְגַה ובִּי דוֹגוְרְכַלֵּר יִרִיגִייְוַרַהּוֹדַבּוֹלְסִינְלַר סִיוְגַה טוּטוּבְרִיקְקַה: והתנחלתם באוּלִיש אַלִיפ אוּלַשִּירְסִיז אַלַרְנִי אוֹגְלַנְלֵרינִיוֹנְהַה אַרְטִיגִיוְבוֹ מֵרַסְלַמַנָה טוּטוּבְּלִיק ו דוּנְיַנַה הַנִין אַלַרְנִי קוּלְלַנִיגִיז ו בַקרַדְשְׁלָרִיגִיוְגַה אוֹגְלַנְלָרִי יִשְׂרָאֵל נִיג פִּישִׁי קַרְדַשִׁינַה ו אִרְכְּלַנְמַגִין אַנֵר זור בִּילֵן: וכי דַאַגַר וְמָםַא קַבַּאָ גַריְפָנִיג יַא אוֹטוּרְשָׁנִיג בִּיְרְגַגַה ו דַזוּגוּרְטְלַסַא קַרְדַשִׁיג כִּי בּיְרַגִַּּכִינַה ו דַכַּטִילְּכָה גַּרִיפְּבָּה אוֹטורַשְּׁקַה בִּי בִּיְרְגַנַה ו יַא גוּי מִמֶּלִינִא משְּׁפְּחְסִינַה גָרִיפּ נִיג : אחרי סַמִילְגַנִינְרוֹ כוֹנְרָה כִילַסְלִיק בּוֹלְסִין אַגַרו בִּירְסִי קּרְרַשְׁלֵריְנַרוֹ בִילַם אִידְכִין אָני : או יַא אַגַצַכִּי יַא אוֹגְלִי אָנַצַכִּינִיג כִילַם אִיְּדְכִין ו יַא יוּבוּגִּינְּדֵן מִנִינִיג מִשְׁפַּחְסִיְנָדן כִילַם אִירְסִין אַנִי ו יַא יִמְבח קובִמִי דַאוֹזי כִילַם אוּלִינְסִין : וחשב דַחָסַפּ אָמְסִין סַמִין אַלִּיוְצִיסִי בִּילַן וּסַמִילְרִיגִי יִיִּדִינְדֵן אַנַרוּ יִיּלִינַה דִגִּין אוֹל יוֹבֶל נִיג וּ דַבּוֹלְסִין גוּמוּשִׁי סַמּיבוּנִיג סָנִי בִּילַן יִילְלַרְנִיג ו גוּנְלַרִי גִיבִּי אִיְרְגַמְנִיג בּוֹלְסִין בּוְרַנִסִינַה: אם אָבַר דַגין צוֹק קַלְהִי אָסַה יִילְלַרְדֵן ו אַלַרְגָה גוֹרַה קַוְשַׁרְסִין כְיַלֹס דִיגִינִי גוּטוּשִׁינְדֵן סמובוניג : ואם דַאָנַר אַז קַלְרִי אָסַה יִילְלַרְרַוֹויִילִינַה דִגִּין אול יובר נִיגו דַחָּסַכּ אַבְר יִילְלַרִינַה גוֹרָה דָקוֹשִׁרְסִין בִילַם לִיגִינִי בּ כשביר אִירְנַם נִיבִּי יִילְבֹן יִילְגַה בּוּלְסִין בִּירְגַסִינַה ו אָרְבְּלַנְמַסִין אַגֵּר וור בִּיַלן גוּוְלֵרִיגְצַה : ואם דַאָּגַר בְיַלָם אוֹלְמַמָא פּוּלֵר בּיַלוֹ ו דַצִיקְסִין יִילִינְדַה אוֹל יוֹבֵל נִיג ו אוֹל דַאוֹגְלַנְלַרֵי בּיְרְגַּסִינֵא : כי זירַא מַגַה אוּגְלַנְלַרי יִשְּׁרָאָל נִיג קוּלְלַר ו קּיְלְרַיִּימְרִיר אַלַר ו

בי ציגורים אַלְרָני מִיסִיר וַרִינְדַן וּ מִנְמִין יוֹנְי מַגְרִיגִיז : משׁ וְרְמַגִיז אוזוּגִיזְנָה אַפַּקְלַרו דַפָּמָל דַמַצִּבָה מוּרְגוּזְמַגִיז אוזוּגִיזְנָה דַפּוּשַׁמַק את שַבָּת לַרִימִנִי בַּקְלַגִיז וּ דָמִקְרָשִימְרַן קוֹרְקוּגִיז וּ מִנְמִין יוֹי מַנְמִין יוֹי מַגְרִיגִיז את שַבָּת לַרִימִנִי בַקּלַגִיז וּ דָמִקְרָשִימְרַן קוֹרְקוּגִיז וּ מִנְמִין יוֹי מַנְמִין יוֹי :

DX

בחקתי

בְּחַקוֹתִי תַּלֵכוּ וְאָת רִיצְוֹתֵי תִּשְׁמְרוּ וַנְעִשׁיתָם אֹנְתם: אַנֵר רַבִּימְלַרִים ロン בּיבוֹ יוּרוּסַגִּיז בַא נִיצְוָה לַרִימְנִי סַקלַסִגִּיז וֹ דַקִילְסַגִּיז אַלַרְנִי : ונחתי דַוָרִימִין יַמְנוּרְלְרִיגִיזְנְי וַבְּשְּׁלַרִינְדָה ו בַּוִירִיר אוֹל יֵר מַבְסוּלִינִי ו דַאַנְצֵי אוֹל שוּוְנִיג וִרִיר יִמִישִׁינִי: והשיג דַיִּמֵר סִיזְגַה אָנְדִיר בַּסְמַק בַּג בּוֹזוּמוּגַא קַדַרו דַבַּג בּוזוּמוּ יִמֵר אוּרְלוּק סַצְמַכַּה קַרַר וַרַאַשַּׁרְסִיז אַשִּׁיגִיזְנִי טוּיְגוּנְצַה וּ רַאוֹטוּרוּרְסִיז אָמִינְלִיכּ בּילַן יִרִיגִיזְרָה: דורימין אָכַנְלִיבּ אוֹל וִרְבַה ו דַיַּמִּרְסִיוֹ דַבּוֹלְמֵז קַלְמְרַמִיוִצִי ו דַא בִּכַרְמִין יַמֵּן כִּייבְנִי אוֹל וַרְבוֹן בַקִילִיץ גִצְאַוֹ וֵרִיגִיוְנַה : ורדפתם בַקּובַרְסִיז דוּשְׁמַנְלַרִיגִיוְנִיוּ בשושרבר אַלְריגיוְגַה קיליץ בּילַן : ורדפו בקוברכר סיוְבן בש בישי יוּוְנִי וּ דיוז בּישׁי כִיזְבן שוּמַנְנִי קוּבַרְלֶר ו בַשוּשַׁרְלֵר דוּשְׁמַנְלְרִינִיז אַלְרִיגִיזְגַה ו קילִיץ בִילֵן : ופניתי - דַבַיִירִילִיְרִמִין ביִזְנַה ו דַיֵילְרִירִימִין בּוֹנָי ו דַבּוּבּיְשִירְמִין בּיזְנִי ו דַמּוּרְגוּזוּרְמִין שַׁרְמִימְנִי בּּיְרַנִגִּיזְנָה : ואכלתם דַאַשַּׁרְסִיז אַסְבּינִי אָסְבּיְרַגְנְנִי וּ דאַכְבִּינִי אַלְרִינְדָן ינִינִיג צִינִרִיפּ מוֹבַּרְסִיז : ונתתי רֵוִירִימִן מִשְׁבְּנִימְנִי אוֹרְאַגִיוְדַתוּ דַכור אִירְמֵז מוַרַרִים סוּזגָי : והתהלכתי דַיורורְמִין אוּרְטַגִיוְרָה ו דַבּוּלוּרְמִין סיוֹגַה מִנְרִיגַה ו דַסִיז בּוֹלוּרְסִיז מַגַה אוּלוּכְקַה : אני מִנְמִין יְנִי מֵנְרִינִיז ו כִּי ציגַרְדִים בִיוְנִי מִיכִיר וִרִינְדֵן בּוֹלְמַקשֵן אַלַרְגַה קוּלְכֵר ו דַסִינְדִירְדִים צוּמַצֵּלֵרִין בּוּוּנְסַגִּיזְנָיג וּ בַיוּרוּמְמִים סִיזְנִי מִיכּ בּוּיְ בִּילֵן : ואם בַאִּגַר אָהִימְסַנִיז סַגַּהוּ דַקילְמַכַנִיז ג׳וּמְלֵא אוּשְׁבּוּ מִצְיָה לַרְנִי: ואם דַאגר רַסִימְלַרִימְני כור אָמְכַנִיז ו דאָגַר שרעמְלַרְמְנִי כוֹר אִירְסַא גַ׳נִינִיזּ ו קִיְלְמַמַקּה ג׳וּמְלַא מִצְוְהְ לַרִימְנִי ו בּוּזְמַגַה סיו שַׁרְמִיקְנִי : אף דַגִּין מֵן קּילַרְמִין מוּנִי סִיְנַה ו דַמַיִין אִמַרְמִין אוּקְמוּגִיזְגַה הַלְשְרַמֵּק וֹ אוֹל כובוק כַבְּמְלִיקְנִי וֹ דַאוֹל קיוָדִירְמַא כַבְמְלִיקְנִי וֹ מַבוּכִיוְצִילֵר גוְיִלַרְנִי דַסִיוְלַמִיוְצִילַר גַ׳ נְנִי ו דַסַצַרְסִיוַ בּוֹשְׁקַה אוּרְלוּגִיגִיוְנִי ו דַאַשַרְלַר אַנִי דושְׁמַנְלְרִינִיוּ: ונתתי בורימין בישימימני ביזרה ו הקירילרסיו אלרינה דושמנלריגיוניג ו דארְבְּלַנִירְלֵר סִיזְגָה דּוּשְׁמַנְלַרִיגִיז ו דַקַצַרְסִיז דַבּוֹלְמִזַ קוּבוּכְצִי סִיזְנִי : יאם דָאָנֵר בּוּלְרְנַה הָנִין אָשִׁימְמַסְנִיז מָנָה ו דַאַרְמְטִירִימִין אונומְלְמַה סִיוְנִי ו יְרִי בּוֹבְּ יַזִיּקְלָרִינִיז אוצון : ושברתי דָסִינְדִירִימִן אוֹבְּמַמְלִינִין קוּבִמִינְיְזְנִיג ו דֵּוֹרִימִין כ גובלריגיוני

בחקתי

גוֹבְּלַרִיגִיוְנִי הַמִיר גִיבִּי ו בא וִרִיגִיוְנִי בַּקִיר גִיבִּי : ותם דַבִּימֵר בּוֹשְׁקַה קּיבַמִּיגִיזו דַוּרָכַּזוֹ וַרִיגִיז בַּלְכוּלִינִי 1 דַאַגַצִי אוֹל יִרְנִיג וַרְכַזוֹ יִמִישִׁינִי : ואם דַאָּגַר יוּרוּסַגִיז בּירְנַמָה אוצור בִּילֵן ו דַבִּילַמַסַנִיז אַשִּימְמַה מַגַה ו דַאַרְמְאִירִימִין אוּסְטוּגִיוְגַה אוּרְכֵּוֹקוֹ יְרִי ַקוָּבַ יַנִיקְלַרִיגִיז אוּצוּן: והשלחתי דַבּנִירִימִן סִיוְגַא אול סוּוּגָיג בִּייִגִינִיוֹ בּמוּב אָמַר סִיזְנִי וּ דַבְּסַר תּוֹנִייּגִיזְנִי וּ דַאַז אָמַר סִיזְנִי וּ דוֹבן בּוֹלִירְכַר יוֹלְלַרִיגִיז: ואם דַאָנֵר בּוּלֵר בִּילֵן אוּגוּטְלַנְמַסַגִּיז מַגַה ו דַיוּרוּסַגִיז בִּירְגַמַה אוצור בִּילַן : והלכתי דיורוּרְמִין דַגִּין מֵן בִּירְגַנִיזְנַה אוצור בִילַן ו דאור בִיזְנִין כִין וּ דַגִּין מֵן וּ יַרי קוֹב יַוֹיָקלַרִיגִיז אוצון: והבאתי דגמירימן אוּסְטוּגִיזְבַה קיליץ אוץ אַליוְצִי אוצון שַׁרְמָנִיג ו בַמוֹפַלַגִיְרָסִיז שַׁחַרְלַרִיגִיְזגַה ו דַיִבִּירִימָן אוֹלָט אוֹרְטַגִיזְדַהוּ דַוֹרִילִיסִיזּ קולוגה דושְׁמַנְנִיג : בשברי סִינְדִירְגַנִימְדַה סִיזְגַה אוֹשְמַב קוּוַמִינִי ו דַפִּישִׁירְרַלַר און כַמִינְלֵר אָבְּמָגִיגִיוָנִי בִיר הַנְרוּרְדַה ו דַקְיִמִריְרַלַר אָשׁוּגְיִזנִי צִבִּי בִּילן ו דַאַשַׁרְסִיז דַמּוֹיְמַוְסִיז : ואם דַאָּגַר מוַנִיג בִּילֵן אֵשִימְמַבוּיז מֵגַה ו דֵיוּרוּבַוּיז בִּירְגַמַה אוּצור בִּילֵן : והלכתי דַיוּרוּרְמִין בִּיְרגַגִיוְגַה אוצור קחֵרִי בִּילֵן ו דַאוּגוּמְלַרְמִין כִיוְנִי דַגִין מֵן ו יִרִי קוֹב יַזִיקְלַרִיגִיז אוצון : ואכלתם דַאַשַׁרְסִיז אוֹגְלַנְיַרִיגִיוְנִיג אָטִינִי דַקּוְזַלְרִיגִיוְנִיג אָטִינִי אַשַּׁרְסִיז : והשמרתי דַקוִיפ אִירַרְמִין בְּטָה לַרִיגִיוְנִיוּ דַבֶּכַרְמִין קוּיַשׁ סִיפַּמְלַרָיגיוְגִי : דוַרִימִין גוֹבְדַלַרִיגיוְנִי ו גוֹבְדַלַרִי אוּסְטוּנַא אַיַרְנְצִילְכְּלַרִיגִיוְנָיג וּ דַכוֹר אִיָּדֵר בִּילַגִים כִּיזְנִי : ונחתי דַוֹנִימִין שַׁחַּרְלַרִיגִיוְנִי וֹבַוּוּ דורן אַמַרְמִין מִקְרָשׁ לַרִיגִיוְנִי ו דָקבוּל אַמְמַבְיר קרְבְּנְלַרִיגִיוְנִיג קוֹקוּכִינִי:והשמתי דַוָרַן אָמַרָמִין מֵן אוֹל וַרְנִי ו דַּמַגְמוּנְלְנַרְלַר אוּסְמוּנֵה דוּשְׁמֵנְלַרִיגִיז ו אוֹל אוֹמוּרְנִגְנְלַר אַנְדַה : ואתכם דַפִּיזְנִי כַצַרְמִין אוֹלוּסְלַרְדַה ו דַסוּבוּרִימִין אַרְדִיגִיזְדַן קּילִיץ ו דַבּוֹלִיר וֹרִיגִיזוַהוֹ ו דַשַּׁחַרְכִרִינִיז בּוֹלְרַלֵר חַרב : אז אוֹל וַכְמַמָּה חַמַם אָמֵר אול זר שְׁמְשַׁלַרִינִי וגינְמָלָא גוּנְלַרִינָדה וַרַנְלִיגִינִיג ו מִיוֹ יְרִינְיבוּ אוֹלִיפ רוּשְׁמֵּנְלַרִיגִיונִיג ו אול וַבְּשְׁשֵׁה שִּנְץ בּוֹלָר אוֹל יֵר ו דַתַּמִם אִמֵר שְׁמִשְׁה לַרִינִי : כל ג׳וְמְלַא גוּנְלַרִינְּדַה וָרַנְלִינִינִיג מִינִץ בּוּלַר ו נִנִי בִּי מִינְץ בּוּלְמֵרִי שְׁמִשְׁה לַרִינִיוְדַה ו אומוּרְגנִינִיוְדַה איקטונה : והנשארים ראול קלנולר סיובה ו דגמירמן ימשקליק גוגוללרינבה ו ירלרינדה

בחקתי. כה

וֹרְלַרִינְדָהַה דוּשְׁמַנְּלַרִינִיג ו דַקּוּבַר אַלַרְנִי אַוֹיִי שׁוּבוּלְרָגוֹ יַפְּרַקְנִיג ו דַקּצַרְלַר קּילִץ קאָמָסִי גִיפִּי ו דַמוּשַׁרְלַר דַבּוּלְמֵזּ קיבוּבְאַי: וְכשׁלוּ דַסוּרוּנִיְרְלַר הַר בִּישִׁי קּרְדַשׁי בּילֵן וְ שִּישְבִּי אַלְדִינְדֵן קִילִיצְנִיג ו דַקוּבוּבְצִי יוֹקְשִיר ו דַבּוֹלְמֵז סִיוְגַה שוּרְמַק אַלְרִינַה רוֹשְׁמַנְּלֵרִיגִיוְנִיג : והנשארים באול ַקלְגַנְּלַר כִיוְבֵה וּ צוּרִירְלַר גוּנַחְלַרִי בִּילֵן נ וֹרְלַרִינְדָה דּוּשְׁמַנְלַרִיגִיזְנִיג ו דַדַגִין גוּנִּחְלֵרִי בּילו אַמַרִינִיג בּיְרגַלַרִינַה צוּרִירְכַר ו והתורו דאיקרר אַשַרְלר גוּנחְלַנִינִי וּ בא גוּנַחִין אַשַּלרינִיג וּ שַנְּמַקְלַנִי בּילַן כִּי מַנְדִילַר מַגַר ו דַדַגִין נְנִי כִּי יוּרוּדִילַר בִּירְגַמַה אוּצוּר בִּילַן : אף דַנִין מֵן יוּרוּרְמִין בִּירָנַלְרִינָה אוּצוּר בִּילַן ו דַגִּמִירִימִן אַלַרְנִי וְרִינָה דוּשְׁמֵנְלַרִינִיג ו דַאוֹל וַבְמְטַה כִינִינִיר אול יבוק גוגוללרי ו דאול ובשמה בישמירירלר גונחלריניג ג׳ורומוני : וזכרתי דַבַגִיניִרְמִין שַרְּמִימְנִי כִּי נַעֲקוֹב כִּילוֹ ו דַרַגִין שַרְמִימְנִי כִּי יַצְחְק כִּילוֹ ו דַרַגִין שַׁרְמִימְנִי כִּי אַבְרְדְּםְ בִּילֵן סגַינִירְמֵין וּדַאוֹל יִרְנִירָא סַגִינִירְמִין : והארץ דַאוֹל יִר בָּמִישִׁילִיר אַלַרְבון ו דַתַּמֵם אַמֵר שְׁמִמְה לַרִינִי וֹבון בּוּלְנגִינְבה אַלַרְבון ו בַאַלַר בּימְמִירִירְלֵר גוּנַהְרִינִיג ג׳וּרוּמוּנִי ו אַנִיג אוּצוּן בִּי שַׁרַעַמְלַרִימְנִי כוֹר אִמְמִילַר וּ דא אַנִיג אוצון כּי רַסִימְלַרִימִני כור אִירְמִי גַ׳נְלָרִי : ואף דַרַגִין בּוּנִי קוֹרַרְמִין בּוֹלְגנִלְרִינְרָה וַרִינְדָה דּוּשְׁמַנְלַרִינִיג וּ בִּי בוֹר אָמְמַמְדִיר אַלְרָנִי דַאָּיַרְנְמַמְּדִיר אַלְרְנִי מַבוּסְמֵנָה אַלְרָנִי ו בּוּזִמַנָה שַׁרְמִימְנִי בּיִרנּלְרִינָה ו זִירָא מִנְמִין יְיָּי מִנְּרִילִרִי : ווכרתי דַבַּגִינִירְמִן אַלַרְגַהְ שִׁרְשִׁין בּוּרְוּגְּיַלַרְנִיג וֹבִּי אִיגַרְרִים אַלַרְנִי מִיפִיר זָהִינְדַן גוּוְלַרְנְצֵא אול אולוּסְלַרְנִיגַ וּ בּוֹלְמַגַה אָלַרְגַא מֵנְרִינַה וּ מִנְמִין יְנִי : אלה בּוּלַרְדִירְלַר אוֹל רַסִימְלַר דאור שַרַעמְלַר דאול תוּרְלַר ו כִּי וֹוְדִי יְנָי אוֹזוּנִיג אַרַסִינַה ו דַאַרַכִּינַה אוֹנְלַנְלַרִינִיג ישְרָאֵל נִיג ו טַנִינְרַה סִינֵי נִיג קוֹלוּ בִּילַן משָׁה נִיג :

בן וידבר דַכּוּזְלַרִי יְיני משָה גַה דִמַה : דבר סוְזַכְּגִין אוֹגְלַנְּלָרִינַה יְשְּׁרְאֵל
נִיג דַאַיְטְקון אַלְרָנַה וּ כִּישִׁי כִּי אַיִירְסַה אָנְייֵט וּ קוַיִסִיג בִּילַן גַיְנְלַר יְיַּיְּ
נֵה : והיה דַבּוּלְסִין קִייָסִיג אוֹל אָרְכַּב וּ יגִירְמִי יִשְׁטֵן דַאַלְטְמִישׁ יַשְּׁקַה דְנִין וּ
דְבּוּלְסִין קִייַסִיג אַוֹלִי מִסְקָל גוּמוּש וּ מִסְקְלִי בִּילֵן אוֹל קוֹרָשׁ נִיג : זֹאם דַאָּגַר
מִישִׁי אָסָה אוֹל וַדְבּוּיְסִין הַיִּיְסִיג אוֹמוּז מִסְקל : זֹאם רָאָגר בָּשׁ יְשְׁטֵן רְינִיְרְמִיּ
מִישִׁי אָסָה אוֹל וַדְבּוּיְסִין הַיִּיְסִיג אוֹמוּז מִסְקל : זֹאם רָאָגר בָּשׁ יְשְׁטֵן רִינִיְרְמִיּ

רשבה

בדקתי

בישְבַּה הַגִּין וַ דַבּוֹלְסִין קִייַסִיג אוֹל אִרְבַּבּ יִגִירְמִי מִסְקְל ו דַמִּישִׁיגַה אוֹן מִסְקְל : יאם דאנר פיר עירו רבש ושבה הגיו ורבולסין קייסיג אול ארבב ובש מסקל גומושׁ ו דַמִישִׁיגַה קייַסִיג אוץ מְסָקָל גוּמָשׁ : ואם דַאַגַר אַלְמְמִישׁ יַשְׁמַן דַיוּגַרִי אַנַר אַרְכַּב אִיסַאוֹדַבּוֹלְסִין קּנִיסִיג און בִשׁ מִסְקּל וֹדַמִישִׁיגַה און מִסְקּל : ואם דַאַנַר זַרְלִי אָפַה אוֹל קײַסִינְבוֹ ו בַשוּרְגוּוְסִין אָנִי אַלְרִינַה אוֹל כּהֵן נִיג ו בַקּיִיְסְלָסִין אַנִי אול כהן ו גגא גורה כּי וִשְׁסַה קוּבשִי אוֹר אִנְייַשׁ אַשְבַנְנִיג ו קיַיִסְלַסִין אַנִי אוֹר כּהֵן: ואם דַאָנַר תוּוֹרַ אִיסַא כִּי יוּבוּקְלַשְאַיִריּרְלֵר אַנְרַן הָרָבֵּן יְיִי נַת ו הֵר נֵא כִּי וַרְסַה אַנְבֵן יני נַה בּוֹלְסִין קוֹרָשׁ : לא תַּבְדִיל אִידְמַסִין אַנִי בַרגִישְׁמִסִין אַנִי יַבְשִׁינִי יַמַנְנָא יַאַ יַמַנְנְי יַבְשִׁיגַה ו רַאֵיַר הַגִּישְׁטִירְמַה הֵגִישְׁמִירְכַה תוּוַרְנִי תוּוָרְנַה ו דַבּוֹלְסִין אוֹל דַדְגִישְׁמִירְטַסִי בּוֹלְסִין קוֹדֶשׁ : ואם באַר ג׳וּסְלֵא מוּנְדֵר תּנִור אִיסַאו פּי יובוּקְלַשְׁמִירְמַסְלַרַ אַנְדֵן קָרְבַּוֹ יָיָי נַה ו דַמוּרְגוּזִסִין אוֹל תוּוַרְנִי אַלְרִינַה אוֹל כּהַן ניג : והעריך דַקיַיִסְלְסִין אוֹל כּהֵן אַנִי וֹ גַרַכּ יַבְשִׁינִי גַרַכּ יַמַנְנִי קייַסִיגְצַה אִי כּהֵן עלי בּוֹלְסִין: ואם דַאָגַר בִילַס אִיְרְמֵא בִינַס אִירְסַא אַנִי וּ דַאַרְמְמִירְסִין בִּשִׁינְגִ׳יסִינִי קיַיִּםיג אוּסְשוּנה : ואיש דַבּישׁי בִּי קוֹרָשׁ אִמְסַה אֵוִינִי קוֹרֶשׁ ינֹיְי גַה ו דַקייַכְלַבִין אַנִי אול בהן ו גרב יבשיני דגרב ימני ו נקדר בי קיבלפה אני אול בהוועלי פורפין:ואם באַנר אור קודשׁ אַשְּבּּוֹכִילַם אִידִרְבַא אוֹינִיובאַרִמְאִיְרְכִּיוֹבְּשׁינְנִיִי גוּמוּשׁ קיִיסִיג אוּכִּמוּנְהוּ דַבּוֹלְסִין אַנִיג: ואם דַאַנַר מַרְלוּבִינְדֵן מוֹשוּבְלוֹגִינִיג קוֹרָשׁ אַמְסַה כִּישִׁי יני נַהו דַבּוּלְמִין הִייַסִיג אוּרְרוּגִינַה גוַרהן אוּרְלוּגִי סַבּונִיג אַרְפַּרְלַרְנִיג ו אַלְלִי מִפַּקל בּוּמוּשְׁבַּה: אם אָנֶר יִילִינְרַן אוֹל יוֹבֵל נִיג קוֹרָשׁ אִמְבַה שַׁרְלוֹבִינִי ו קייַסִיגְצַה שוּרְסִין : ואם בַאָּנֶר אוֹל יוֹבֵל דֵן סוֹנְרָה קוֹרֶשׁ אִמְסַה מַרְלוֹבִינִי ו דַחַסַפּ אַמְסִין אַנֵּר אוֹל פֹהֵן אוֹל גומושני ואול קלנו יילְלַרְגַה גורה ויילינה הגין אול יובל ניג ו דאפקסילסין קייסינְרַן: ואם באנר כילם אידמא כילם אידמא אול שרלובני אול קוהש אִמְבּן אַני ו באַרְמְאִירְסִין בשינג׳י גומוש קייִסִיג אוסטונה דמוקם בולסין אַנַר: ואם דאגר כִילַם אידְמַסא אול פַּרְלוּבְנִי וֹ דָאָנֵר סַמְסַה אוֹל שַרְלוֹבנִי אוֹגְנָה בִּישִׁיגַה וֹ כִילַם אוּלְמַסֵין אַרְמִיק : והיה בּבוּלְפִין אוֹל פַּרְלוֹב וַעבִפִי צִיקְקוֹינְבָה יוֹבַל בה קוֹנִשׁ ינִי גַה ו פַּרְלוֹבוּ גִיבּי אוֹל חֵוֹכִם ניג

בחקתי

כט

ניגופהן גַה בּוּלְסִין מומוּבְלוּגִי : ואם באגר סַאִין אַדַן שַרְלוּבִיניִ כִּי רוּגוּלּ,שַרְלוּבִינְהַן שושובלוגינין ו קורש אַשְּׁסָה יני גה : וחשב דַחַסַפּ אִשְּׁסִין אַנַר אוֹל פהן ו קררין אול קייסיגניג ו ייליגה הגין אול יובל ניגו הורסין קייסיגני אול גוּנְרַה קוֹרֶשׁ יִנְי גַה: בשנת יִילִינְרַה אוֹל יוֹבֵל ניג ַקִּימְסִין אוֹל מַרְלוֹב ו אַנֵר כִּי סַטִין אַלְרִי אַנִי יַאִנִינַרן ו אַנֵר כִּי אַנִיג טוטוּבְלוּגִי אוֹל יִרְנִיג ו וכל דַג׳וּמְלֵא קייַסִיג ו בּוּלְסִין אוֹל קוֹרָשׁ מִכְקְלִי בִּילַן וּ יְגִירְמִי צגירָרַב בּוּלְסִין אוֹל מִסְקל : אך מִב מוּנְגוּץ בִּי מוּגְגוּץ בּוֹלְסַה יֹנֶי בַה אוֹל תוורדה ו קורש אַמְמַסִין כּישִׁי אַנִי ו גַרַכּ אוגוז גַרַכּ קוֹיְ יִינִי נִיג דִיר אול : ואם דַאָנַר אוֹל מוּנָרַר תּוּירְרֵן אִיסַאוֹדַכִילַם אִידְסִין קּיַיסִיג בִּילַן ו דַאַרְטְמִירְכִין בּשִׁינְגִיסִינִי אוּסְמוּנַא ו דַאָנַר כִילַם אוּלוּנִמַםַא דַכַּמִילְסִין קּיַיִּסִיג בּילו : אך מַב ג׳וּמְלַא חָרֶם ו בּי חָרֶם אַמְסַה בִּישׁי יְיף גַה ג׳וּמְלַאחַן בּי אַנִיג ו אָרָם דַן יַאַ תּוּנַרָדן יַאַ שַׁרְלוּבוּנְדַן טוּטוּבְלוּגִינִיג ו בְּטִילְמַסִין דַכִּילְם אולונמסין ו הר הרס ו קורשי קורש לרניגריר אול יי בה : כל הר הַרָם כִּי הָרֶם בּוּלְסַה אוֹל אָדָם דֵן כִילִם אוּלוּנְמַסִין ו אוֹלְמַה אוּלְרוּרְוּלַסִין : ובל דג׳וֹמְלֵא אוֹנְדַסִי אוֹל יִרְנִיג ו אוּרְלוּגִינְדַן אוֹל יִרְנִיג יִמִישִׁינְדַן אוֹל אַנְצְנִיג ו יוֹיָ נִיג דִיר אול קוֹרָשׁ יִיָּי גַה ; ואם דַאָנַר כִילַם אִירְמַא כִילַם אִיְרְסֵא בִּישִׁי אוְנַדַסִינַדן וַ בִּשִׁינְגִיסִין אַרְטְאִירְסִין אוּסְמוּנֵה : וכל דֵג׳וּמְלַא אונְדַם סִיגִירְנִיג יַא קּוְינִיג וַ הַר נִא כִּי נִצְּסַה מִיבּינַה אוֹל מַיַּקְנִיג ו אוֹל אונונגיי ו בולסין קוריש יני בה : לא אַראמַסין אַרְסִינה יַכְשִׁיניג יִמְנְגַה דַבְגִישְׁמִירְמַסִין אַנִי ו דַאָנַר בֵגִישְׁמִירְמַה בֵגִישְׁמִירְסַה אַנִי ו דַבּוּלְסִין אוֹל דַרָגִישְׁבַגִי בּוֹלְסִין קוֹרָשוֹכִילַם אוֹלוּנְמַסִין : אלה בּוּלַרְדִירְלַר אוֹל מִצְּוְלֵר וּ בּי סִימַרְלַרִי יִנְי משָה גַה אוֹגְלַנְלַרִינַה יִשְׂרָאֵל נִיג ו מַגִּינְרַה סִינִי נִיג : אֶּלֶה הַמִצְוֹת צְשֶׁר צְוָה ינִי אֶת משָה אָל בְנִי ישִׁרְאַל בִּדְּרָ סִינִי : בְרוּךְ יְיָי לְעוֹלָם אָמֵן וְאָמֵן וּ

חית

12709